An action plan to enhance the transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals

by

LITABA EFRAIM KOLOBE

submitted in accordance with the requirements for the degree of

DOCTOR OF PHILOSOPHY

in the subject

NURSING

at the

UNIVERSITY OF SOUTH AFRICA

SUPERVISOR: PROF L ROETS

MARCH 2024

DECLARATION

Name: Litaba Efraim Kolobe

Student Number: 36640085

Degree: Doctor of Philosophy in Nursing

I proclaim that this thesis, **AN ACTION PLAN TO ENHANCE TRANSFER OF LEARN-ING OF PAIN MANAGEMENT COMPETENCIES OF NURSES IN SAUDI ARABIAN TEACHING HOSPITALS**, is my own work and that all sources used or cited have been recognized and acknowledged using a comprehensive list of references.

This thesis has also been subjected to reliable software testing to check its originality.

I further confirm that this has never been submitted at any institution of higher learning elsewhere before, including submission to Unisa for any degree purposes.

Goure

Signature:

Date: 18/02/2024

DEDICATION

I dedicate this work to:

Proverbs 1:7: The fear of the LORD is the beginning of knowledge. I would like to express my sincere gratitude to Almighty God, Allah, for carrying me and my supervisor through this process.

My wife (Pauline Kolobe) and our children, thank you for your patience in waiting and supporting the time I spent in this study process.

ACKNOWLEDGEMENTS

I would like to express my sincere gratitude to the following individuals and organisations for their invaluable support and assistance in the completion of this study:

- My supervisor, Professor Lizeth Roets, for her unwavering guidance, support, and encouragement throughout the research process. Her invaluable insights and continuous feedback have been instrumental in shaping the quality of this research.
- Dr. Nasser Tawfeeq, Labour and Delivery Anaesthetist Consultant, thank you for supporting this research to KAIMRC and IRB guidelines throughout the process of this study.
- Dr Liezel Korf, my statistician, for her support in analysing the data of this study and support in the accurate interpretation.
- The Nursing Leadership, KAMC and KASCH, Riyadh, Saudi Arabia, for supporting the study to be carried out in the respective nursing care areas.
- The KAIMRC and IRB committee members, for ensuring the study process followed the institutions' guidelines.
- I extend my gratitude to all participants who generously dedicated their time and made valuable contributions to this study.
- My entire family, all friends, and colleagues at work thank you for your enormous support, understanding, and curiosity in encouraging this process. Your love and belief in my abilities sustained me through the highs and lows of this research journey.
- Professor Phil Botha, my editor, for this meticulous proofreading of my thesis, which has contributed to its quality.

AN ACTION PLAN TO ENHANCE THE TRANSFER OF LEARNING OF PAIN MANAGEMENT COMPETENCIES OF NURSES IN SAUDI ARABIAN TEACHING HOSPITALS

| STUDENT NUMBER: | 36640085 |
|-----------------|---------------------------------|
| STUDENT: | LITABA EFRAIM KOLOBE |
| DEGREE: | DOCTOR OF PHILOSOPHY IN NURSING |
| SUPERVISOR | PROFESSOR LIZETH ROETS |

ABSTRACT

Purpose: The purpose of this study was to develop an action plan that would facilitate the enhancement of the transfer of learning (TOL) of pain management competencies amongst nurses at teaching hospitals in Saudi Arabia.

Methods: The study employed a five-phase explanatory sequential mixed method design. The study commenced with quantitative data collection and analysis, followed by qualitative data collection and analysis. Nurses and clinical facilitators completed a self-administered questionnaire to assess pain assessment resources in Phase 1. Phase 2 collected data and analysed nurses' characteristics and learning styles with a self-administered questionnaire. Phase 3 described hospital teaching methods, the learning content, and the learning climate using clinical facilitator data from self-administered surveys with closed and open-ended questions. In Phase 4, Phases 1, 2, and 3 data and a literature review were used to develop an action plan to enhance nurses' pain management competencies. In Phase 5, a purposively selected panel of clinical facilitators and nurses validated the action plan to enhance the transfer of learning of pain management competencies of nurses within the context.

The Systemic Model of Transfer of Learning by Donavan and Darcy (2011) was the study's theoretical framework. This model suggests individual and organizational performance as the link between the transfer of learning and work environment characteristics. The model is based on the idea that four factors influence the transfer of learning in the workplace: 1) trainee characteristics (the nurses), 2) training design (learning content), 3) training transfer climate, and 4) work environment.

The findings:

The action plan was validated and approved to enhance the transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals. The action plan encompassed the specified actions, which were to motivate nurses to further their studies, make appropriate and relevant pain management tools accessible to the nursing team in every clinical area, develop a practice-oriented content-specific short pain management training program, develop a pain management short program that accommodates all learning types, incorporate different teaching approaches to accommodate diverse learners and facilitators in the training of pain management, develop strategies to motivate nurses to participate in the short training program and motivate nurses to apply the knowledge gained in the training program into practice.

Conclusion:

The action plan was prepared by incorporating the insights of twelve e-Delphi panellists consisting of clinical facilitators and registered nurses. The presence of two teaching hospitals in Saudi Arabia will increase the likelihood of adopting and implementing the action plan that will help nurses effectively transfer their knowledge of pain management and enhance their competencies.

Key concepts: Action plan, Competencies, Pain management, Transfer of learning.

TABLE OF CONTENTS

| DEDICATIONiii |
|---|
| ACKNOWLEDGEMENTSiv |
| ABSTRACTv |
| TABLE OF CONTENTS vii |
| LIST OF TABLES xviii |
| LIST OF FIGURES xxii |
| LIST OF ANNEXURES |
| LIST OF ACRONYMS AND ABBREVIATIONS xxiii |
| CHAPTER ONE: OVERVIEW OF THE STUDY1 |
| 1.1 INTRODUCTION AND BACKGROUND TO THE RESEARCH PROBLEM 1 |
| 1.2 STATEMENT OF THE RESEARCH PROBLEM |
| 1.3 RESEARCH AIM/PURPOSE4 |
| 1.4 RESEARCH OBJECTIVES |
| 1.5 RESEARCH QUESTIONS4 |
| 1.6 THEORETICAL FRAMEWORK |
| 1.7 DEFINITIONS OF KEY THEORETICAL CONCEPTS |
| 1.7.1 Action plan6 |
| 1.7.2 Clinical facilitator6 |
| 1.7.3 Competency6 |
| 1.7.4 Learning6 |

| 1.7.5 | Transfer of Learning (TOL)7 |
|--------|---|
| 1.7.6 | Nurse7 |
| 1.7.7 | Pain management7 |
| 1.8 KE | Y OPERATIONAL DEFINITIONS7 |
| 1.8.1 | Clinical facilitator7 |
| 1.8.2 | Competency7 |
| 1.8.3 | Nurses |
| 1.9 RE | SEARCH METHODOLOGY 8 |
| 1.9.1 | The Research Paradigm8 |
| 1.9.2 | Research design8 |
| 1.9.3 | Research setting11 |
| 1.9.4 | Population13 |
| 1.9.5 | Sample and sampling methods13 |
| 1.9.6 | Data collection methods and procedures16 |
| 1.10 N | IEASURES TO ENHANCE THE QUALITY OF THE STUDY |
| 1.10.1 | Validity and reliability17 |
| 1.10.2 | Rigour18 |
| 1.10.3 | Inference quality19 |
| 1.11 C | DATA ANALYSIS |
| 1.11.1 | Phases 1, 2 and 3: Quantitative data analysis19 |

| | 1.11.2 | Phase 5, Qualitative data analysis | 19 |
|---|-----------------|--|------|
| | 1.11.3 | Integration of the findings | 20 |
| | 1.12 E | ETHICAL CONSIDERATIONS | 20 |
| | 1.12.1 | Respect for human dignity and self-determination | 20 |
| | 1.12.2 | Informed consent | 20 |
| | 1.12.3 | Confidentiality and anonymity | 20 |
| | 1.12.4 | Researcher-Participant Relationship | 21 |
| | 1.12.5 | Protecting the rights of the institution | 21 |
| | 1.12.6 | Scientific integrity of the research | 21 |
| | 1.13 (| ORGANISATION AND STRUCTURE OF THE THESIS | 21 |
| | 1.14 (| CHAPTER SUMMARY | 23 |
| С | HAPTER | TWO: LITERATURE REVIEW | 24 |
| | 2.1 IN | TRODUCTION | 24 |
| | 2.2 TH | IE CONCEPT TRANSFER OF LEARNING (TOL) | 26 |
| | 2.3 SY DARCY | STEMIC MODEL OF TRANSFER OF LEARNING BY DONOVAN (2011) | AND |
| | 2.4 DII 28 | MENSIONS OF THE SYSTEMIC MODEL OF TRANSFER OF LEAR | NING |
| | 2.4.1 | Trainee characteristics | 28 |
| | 2.4.2 | Training design | 32 |
| | 2.4.3 | Training transfer climate | 38 |

| | 2.4.4 | 4. Workplace environment characteristics4 | 10 |
|---|-------|---|----------------|
| 2 | .5 l | _EVELS OF TRANSFER OF LEARNING4 | ł0 |
| | 2.5.′ | 1. Level 1: Non-specific or general transfer4 | ł0 |
| | 2.5.2 | 2 Level 2: Application transfer4 | ¥1 |
| | 2.5.3 | 3 Level 3: Context transfer 4 | ¥1 |
| | 2.5.4 | 4 Level 4: Near or specific transfer 4 | ¥1 |
| | 2.5.5 | 5 Level 5: Far or general transfer 4 | 11 |
| | 2.5.6 | 6 Level 6: Displacement or creative transfer4 | 12 |
| 2 | .6 - | TYPES OF TRANSFER OF LEARNING4 | 12 |
| | 2.6.′ | 1 Positive transfer4 | 12 |
| | 2.6.2 | 2 Negative transfer 4 | 13 |
| | 2.6.3 | 3 Zero transfer 4 | 13 |
| | 2.6.4 | 4 Horizontal transfer 4 | 13 |
| | 2.6.5 | 5 Vertical transfer4 | ł3 |
| | 2.6.6 | 6 High-road transfer4 | 14 |
| | 2.6.7 | 7 Low-road transfer4 | 14 |
| 2 | .7 F | PAIN MANAGEMENT4 | 1 5 |
| | 2.7.′ | Assessment4 | 1 6 |
| | 2.7.2 | 2 Nursing diagnoses | 71 |
| | 2.7.3 | 3 Planning7 | 73 |

| 2.7 | 7.4 | Implementation | 74 |
|--------|------------|--|----|
| 2.7 | 7.5 | Evaluation | 81 |
| 2.7 | 7.6 | Documentation | 82 |
| 2.8 | со | NCLUSION | 82 |
| CHAP | FER | THREE: RESEARCH DESIGN AND METHODOLOGY: PHASE 1, 2 A | ND |
| 3: MET | HOI | DOLOGY AND DATA GATHERING | 84 |
| 3.1 | INT | | 84 |
| 3.2 | RE | SEARCH PARADIGM | 85 |
| 3.3 | RE | SEARCH DESIGN | 86 |
| 3.3 | 3.1 | Mixed method research (MMR) approach | 87 |
| 3.4 | ΤH | E SETTING | 92 |
| 3.5. | PO | PULATION | 93 |
| 3.5 | 5.1 S | ite population | 93 |
| 3.5 | 5.2 S | ite sampling | 93 |
| 3.5 | 5.3. F | Population of professional registered nurses and clinical facilitators | 94 |
| 3.6 S | Samf | PLING | 94 |
| 3.6 | 6.1 | Proportional stratified probability sampling | 94 |
| 3.7 | INC | CLUSION CRITERIA | 95 |
| 3.8 | SA | MPLE SIZE | 96 |
| 3.8 | 3.1 | Response rates: Phases 1, 2 and 3 | 97 |
| 3.9 | DA | TA COLLECTION INSTRUMENTS: PHASES 1, 2 AND 3 | 99 |

| | 3.9.1 | Questionnaires |
|----|---------------|--|
| | 3.9.2 | The characteristics of the developed questionnaires |
| 3. | 10 | PRE-TESTING OF THE QUESTIONNAIRES |
| | The follov | results of the pre-tests of each of the developed questionnaires were as vs. 104 |
| | 3.10. | 1 Questionnaire 1: pre-test (professional nurses' Phase 1): 104 |
| | 3.10. | 2 Questionnaire 2: Phase 2 (professional nurses) |
| | 3.10. | 3 Questionnaire 3: Phase 3 (clinical facilitators) 115 |
| 3. | 11 | VALIDITY AND RELIABILITY OF THE QUESTIONNAIRES 117 |
| | 3.11. | 1 Validity118 |
| | 3.11. | 2 Reliability |
| 3. | 12 | ETHICAL CONSIDERATIONS122 |
| | 3.12. | 1 Protecting the rights of the study participants 122 |
| | 3.12. | 2 Protecting the rights of the institution124 |
| | 3.12. | 3 Scientific integrity of the research 125 |
| 3. | 13 | DATA COLLECTION: PROCESS FOLLOWED IN PHASES 1, 2 AND 3 125 |
| 3. | 14 | DATA ANALYSIS126 |
| | 3.14. | 1 Descriptive statistical analysis126 |
| | 3.14. | 2 Qualitative data analysis (qualitative enhancement) |
| 3. | 15 | CONCLUSION |

| CHAPTER | FOUR: PHASE 1, 2, AND 3: DATA ANALYSIS AND INTERPRETATION |
|-----------------|--|
| 4.1 IN | TRODUCTION |
| 4.2 PH | IASE 1 |
| 4.2.1 | Demographic characteristics (N = 385)133 |
| 4.2.2 F | Resources available to conduct pain assessment (N = 385) 140 |
| 4.2.3 | Thematic content analysis175 |
| 4.3 PH | IASE 2 |
| 4.3.1 | Demographic characteristics 176 |
| 4.3.2 pain m | Respondents' characteristics identified to enhance transfer of learning of nanagement competencies (N = 384) |
| 4.3.3 manag | Learning styles identified to enhance nurses' transfer of learning of pain gement competencies (N = 384) |
| 4.3.4 | Thematic content analysis185 |
| 4.4.1 | Demographic characteristics (N = 47) |
| 4.4.2 F | Resources available to conduct pain assessment (N = 47) |
| 4.4.3 | Teaching approaches employed in pain management education (N = 47) 225 |
| 4.4.4 | Learning content included in pain management education $(N = 47) \dots 234$ |
| 4.4.5 | Transfer of learning climate within the hospital nursing care |
| 4.4.6 | Thematic content analysis |
| 4.5 CC | DNCLUSION |

| СНА | PT | ER | 5: | PHASE | 4: | LITERAT | URE | REVIEW | ON | ACTION | PLAN |
|------------|------|------|---------------|--------------|----------|---------------|----------|--------------|-------|----------|-------|
| DEV | ELO | OPN | IENT | | | | | | | | 284 |
| 5.′ | 1 | INT | ROD | UCTION | | | | | | | 284 |
| 5.2 | 2 | AN | ACTI | ON PLA | N | | | | | | 285 |
| 4 | 5.2. | 1 | Туре | es of action | on plai | าร | | | | | 286 |
| : | 5.2. | 2 | Princ | ciples of | action | plan develo | pment | | | | 287 |
| : | 5.2. | 3 | Step | s in deve | eloping |) an action p | olan | | | | 291 |
| : | 5.2. | 4 | Appli | ication o | f the S | ystemic Mo | del of | Transfer of | Learr | ning | 293 |
| : | 5.2. | 5 | Deve | elopment | t of the | draft actior | n plan . | | | | 296 |
| 5.3 | 3 | THE | EFIR | ST DRA | FT AC | TION PLAN | ۱ | | | | 310 |
| 5.4 | 4 | CO | NCLL | JSION | | | | | | | 329 |
| СНА АСТ | | ER S | SIX: F .AN | PHASE (| 5: ME1 | THODOLOG | GY, VA | LIDATION | I PRO | CESS AND | FINAL |
| 6.´ | 1 | INT | ROD | UCTION | | | | | | | 330 |
| 6.2 | 2 | ME | THOE | DOLOGY | / | | | | | | 331 |
| | 6.2. | 1 | The | Delphi te | echniqu | | | | | | 331 |
| | 6.2. | 2 | Adva | antages o | of e-De | elphi technic | que | | | | 333 |
| | 6.2. | 3 | Critic | cisms an | d limita | ations of the | e Delph | ii technique | ə | | 334 |
| | 6.2. | 4 | The ' | Validatio | n Instr | ument | | | | | 335 |
| | 6.2. | 5 | Ρορι | ulation | | | | | | | 338 |
| | 6.2. | 6 | Sam | ple | | | | | | | 338 |

| 6.2.7 | Pre-testing of the e-Delphi validation instrument |
|---------------------------|---|
| 6.2.8 | Ethical considerations |
| 6.2.9 | Data gathering |
| 6.2.10 | Data analysis |
| 6.2.11 | Trustworthiness |
| 6.2.12 | Validity |
| 6.2.13 | Reliability |
| 6.3 FIN | IDINGS |
| 6.4 RC | OUND 1 |
| 6.4.1 | Biographical data |
| 6.4.2 | Action statement 1: Motivate nurses to further their studies $(N = 12)$. 352 |
| 6.4.3 tools a | Action statement 2: Make appropriate and relevant pain management ccessible to the nursing team in every clinical area (N = 12) |
| 6.4.4 pain m | Action statement 3: Develop a practice-oriented content-specific short anagement training program (N = 12) |
| 6.4.5 accom | Action statement 4: Develop a pain management short program that modates all learning types (N = 12) |
| 6.4.6 accom (N = 12 | Action statement 5: Incorporate different teaching approaches to modate diverse learners and facilitators in the training of pain management 2) |
| 6.4.7 in the s | Action statement 6: Develop strategies to motivate nurses to participate short training program (N = 12) |

6.5.1 Action statement 1: Motivate nurses to further their studies (N = 12). 387

| 6.5.7 | Action statement 7: Motivate nurses to apply the knowledge gained in the | Э |
|----------|--|---|
| training | g program in practice (N = 12) | 9 |

6.6 FINDINGS FROM ROUND THREE 401

6.6.2 Action statement 1: Motivate nurses to further their studies (N = 10). 401

| 6.7 T | THE VALIDATED FINAL ACTION PLAN | 409 |
|-----------------|---|-----------------------------|
| 6.8 C | CONCLUSION | 416 |
| CHAPTE | ER SEVEN: CONCLUSIONS, RECOMMENDATIONS, AND L | MITATIONS |
| ••••• | | |
| 7.1 I | INTRODUCTION | 417 |
| 7.2 (| CONCLUSIONS | 418 |
| 7.2.1 asse | 1 Objective 1: Identify and describe the resources available to essments. | conduct pain 419 |
| 7.2.2 style | 2 Objective 2: Identify and describe nurses' characteristics es that enhance the transfer of pain management competencies. | and learning 420 |
| 7.2.3 facili | 3 Objective 3: Explore the teaching approaches employed by itators during pain management education of nurses. | y the clinical |
| 7.2.4 and | 4 Objective 4: Describe the learning content regarding pain management. | assessment |
| 7.2.5 nursi | 5 Objective 5: Describe the transfer of learning climate within t ing care areas. | he hospitals' 422 |
| 7.2.6 to en | Objective 6: Develop and validat an action plan that can be inhance the transfer of learning of pain management competenci 422 | mplemented es of nurses. |
| 7.3 F | RECOMMENDATIONS | 423 |
| 7.3.1 | 1 First step: Diffusion | 423 |
| 7.3.2 | 2 Second step: Dissemination | 423 |
| 7.3.3 | 3 The third step: collaborative Implementation | 424 |
| 7.3.4 | 4 Recommendations for further research | 424 |

| ANNEXURES | | |
|------------|-------------|-----|
| REFERENCES | | |
| 7.5 | SUMMARY | 425 |
| 7.4 | LIMITATIONS | 425 |

LIST OF TABLES

| Table 1.1 Table 1.2 | The five nursing care divisions of hospitals A and B Accessible Population | 12 13 |
|------------------------|--|----------|
| Table 1.3 | Population and sample in each stratum of the participating hospitals | 14 |
| Table 1.4 | Organisation and structure of the study | 22 |
| Table 2.1 | Organisation and structure of the study | 24 |
| Table 2.2 | Pain rating assessment tools and patient populations to as- sess | 55 |
| Table 3.1 | Organisation and structure of the study | 84 |
| Table 3.2 | Population and proportional samples | 97 |
| Table 3.3 | Phase 1: Sample size ($N = 385$) and response rate. | 98 |
| Table 3.4 | Phase 2: Sample size (N = 384) and response rate. | 98 |
| Table 3.5 | Phase 3: Sample size (N = 47) and response rate | 99 |
| Table 3.6 | Questionnaire 1 and 3 section B modifications | 105 |
| Table 3.7 | Questionnaire 2: modifications | 109 |
| Table 3.8 | Questionnaire 3 sections C, D, E and F: modifications | 116 |
| Table 4.1 | Organisation and structure of the study. | 131 |
| Table 4.2 | Nurses' nationality (N = 385) | 135 |
| Table 4.3 | The nurses' country of origin and their highest education qualification (N = 385) | 138 |
| Table 4.4 | Distribution of respondents within wards ($N = 385$). | 139 |
| Table 4.5 | Availability of the systematic pain assessment guide (N = 385) | 142 |
| Table 4.6 | Availability of pain rating assessment tools for patients who can self-report their pain ($N = 385$) | 146 |
| Table 4.7 | Availability of pain rating assessment tools for patients who cannot self-report their pain ($N = 385$) | 149 |
| Table 4.8 | Availability of pain rating assessment tools for elderly pa- tients with dementia or cognitive impairment ($N = 385$) | 154 |
| Table 4.9 | Availability of human resources ($N = 385$) | 158 |

| Table 4.10 | Availability of other types of support (N = 385) | 163 |
|------------|--|-----|
| Table 4.11 | Availability of the publications and electronic resources (N = 385) | 167 |
| Table 4.12 | Availability of the organisations that specialise in pain man- agement (N = 385) | 172 |
| Table 4.13 | Availability of the policies for pain management (N = 385) | 174 |
| Table 4.14 | Clinical Facilitators' nationality (N = 47) | 188 |
| Table 4.15 | Country of origin and highest education qualifications (N = 47) | 191 |
| Table 4.16 | Clinical facilitators' nursing wards of work (N = 47) | 192 |
| Table 4.17 | Availability of the pain assessment guides $(N = 47)$ | 194 |
| Table 4.18 | Availability of pain rating assessment tools for patients who can self-report their pain ($N = 47$) | 198 |
| Table 4.19 | Availability of pain rating assessment tools for patients who cannot self-report their pain ($N = 47$) | 202 |
| Table 4.20 | Availability of the pain rating assessment tools for elderly patients with dementia or cognitive impairment ($N = 47$) | 207 |
| Table 4.21 | Availability of the human resources ($N = 47$) | 210 |
| Table 4.22 | Availability of the patient support $(N = 47)$ | 213 |
| Table 4.23 | Availability of the publications and electronic resources ($N = 47$) | 217 |
| Table 4.24 | Availability of the organisations that specialise in pain man- agement ($N = 47$) | 222 |
| Table 4.25 | Availability of the policies for pain management (N = 47) | 224 |
| Table 4.26 | Employment of the teaching approaches $(N = 47)$ | 227 |
| Table 4.27 | The physiologic pain indicators $(N = 47)$ | 236 |
| Table 4.28 | The behavioural pain indicators $(N = 47)$ | 240 |
| Table 4.29 | The pain screening (N = 47) | 243 |
| Table 4.30 | Patient's self-report of pain (N = 47) | 245 |
| Table 4.31 | The proxy-reported pain ($N = 47$) | 247 |
| Table 4.32 | The WILDA systemic pain guide $(N = 47)$ | 248 |
| Table 4.33 | The valid and reliable pain rating scales (N = 47) | 252 |
| Table 4.34 | The acute types of pain types $(N = 47)$ | 255 |
| Table 4.35 | The chronic pain types (N = 47) | 259 |
| Table 4.36 | The factors related to the impact of pain on activities of daily living (ADLs) and quality of life (QOL) ($N = 47$) | 263 |
| Table 4.37 | The strategies to plan for pain management (N = 47) | 266 |
| Table 4.38 | The pain intervention strategies implemented in $(N = 47)$ | 270 |
| Table 4.39 | The nursing actions for pain management evaluation (N = 47) | 272 |
| Table 4.40 | The transfer of learning climate within the hospital nursing care $(N = 47)$ | 275 |
| Table 5.1 | Organisation and structure of the study | 284 |
| Table 5.2 | Integration of Phases 1, 2, 3, and literature support | 298 |
| Table 5.3 | First draft action plan with embedded validation tool | 311 |
| Table 6.1 | Organisation and structure of the study | 330 |
| Table 6.2 | Sample of e-Delphi panellists. | 339 |
| Table 6.3 | Panellists' age (N = 12) | 347 |

| Table 6.4 | Panellists' nationality (N = 12) | 348 |
|------------|--|-----|
| Table 6.5 | Distribution of panellists within wards $(N = 12)$ | 349 |
| Table 6.6 | Panellists' positions (N = 12) | 350 |
| Table 6.7 | The duration of the panellists' current positions ($N = 12$) | 350 |
| Table 6.8 | Action Statement 1: Motivate nurses to further their studies $(N = 12)$ | 352 |
| Table 6.9 | Action statement 2: Make appropriate and relevant pain | |
| | management tools accessible to the nursing team in every clinical area ($N = 12$) | 360 |
| Table 6.10 | Action statement 3: Develop a practice-oriented specific short pain management training program ($N = 12$) | 371 |
| Table 6.11 | Action statement 4: Develop a pain management short pro- gram that accommodates all learning types | 374 |
| Table 6.12 | Action statement 5: Incorporate different teaching approaches to accommodate diverse learners and facilitators in the training of pain management ($N = 12$) | 378 |
| Table 6.13 | Action statement 6: Develop strategies to motivate nurses to participate in the short training program ($N = 12$) | 381 |
| Table 6.14 | Action statement 7: Motivate nurses to apply the knowledge gained in the training program into practice ($N = 12$) | 383 |
| Table 6.15 | Responsible person(s) to develop a policy to motivate nurses to improve their nursing qualifications ($N = 12$) | 387 |
| Table 6.16 | Time frame required to present the policy and negotiate the implementation to the MNGHA) (N = 12) | 388 |
| Table 6.17 | Responsible person(s) to include the policy in all hospitals' policies to motivate nurses to improve their nursing qualifications ($N = 12$) | 389 |
| Table 6.18 | Time frame required to include the policy in all hospitals' policies after approval of the action plan by MNGHA (N = 12) | 389 |
| Table 6.19 | Responsible person(s) to ensure that an electronic format of the pain assessment tools for inclusion in the electronic pa- tient record system to be accessible to the nursing team in every nursing care area ($N = 12$) | 390 |
| Table 6.20 | Time frame required to present the policy and negotiate the implementation to the MNGHA) ($N = 12$) | 391 |
| Table 6.21 | Responsible person(s) to involve nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessments ($N =$ 12) | 392 |
| Table 6.22 | Time frame required to involve the nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessment in all nursing care areas ($N = 12$) | 393 |
| Table 6.23 | Responsible person(s) to ensure that internet-based re- sources are accessible to patient and family to provide sup- port about pain management ($N = 12$) | 394 |

| Table 6.24 | Responsible person(s) to ensure that internet-based re- sources on pain management publications, electronic mate- rials, and organisations that specialise in pain management are made accessible to the nursing team in all nursing care | 394 |
|------------|--|-----|
| Table 6.25 | areas (N = 12) Time frame required to provide for the inclusion of the spe- cifically oriented pain management training in the program | 395 |
| | (for method 3.1) (N = 12) | 000 |
| Table 6.26 | Time frame required to ensure that the learning types are shared and included within the training program (N = 12) | 396 |
| Table 6.27 | Time frame within which time should different teaching approaches be part of the teaching program before implementation ($N = 12$) | 397 |
| Table 6.28 | Responsible person(s) to develop strategies that will moti- vate nurses to participate in the short pain management training program related to their pursing care areas $(N = 12)$ | 398 |
| Table 6.29 | Responsible person(s) to facilitate the implementation of the methods to motivate nurses to apply their knowledge in practice ($N = 12$) | 399 |
| Table 6.30 | Time frame required to implement the methods to motivate nurses to apply their knowledge in practise ($N = 12$) | 400 |
| Table 6.31 | Responsible person(s) to develop a policy to motivate nurses to improve their nursing qualifications ($N = 10$) | 402 |
| Table 6.32 | Time frame required to present the policy and negotiate the implementation to the MNGHA) (N = 10) | 403 |
| Table 6.33 | Responsible person(s) to include the policy in all hospitals' policies to motivate nurses to improve their nursing qualifications ($N = 10$) | 403 |
| Table 6.34 | Time frame required to include the pain assessment tools in the electronic patient record system ($N = 10$) | 404 |
| Table 6.35 | Responsible person(s) to involve nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessments (N = 10) | 405 |
| Table 6.36 | Time frame required to involve the nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessment in all pursing care areas $(N - 10)$ | 406 |
| Table 6.37 | Responsible person(s) to ensure internet-based resources on pain management publications, electronic materials, and organisations that specialise in pain management should be accessible to the nursing team in all nursing care areas (N = 10) | 407 |
| Table 6.38 | Time frame required to implement the methods to motivate nurses to apply their knowledge in practice ($N = 10$) | 408 |
| Table 6.39 | The validated action plan to enhance the transfer of learning of pain management competencies of nurses in Saudi Ara- bian teaching hospitals | 409 |

LIST OF FIGURES

| Figure 1.1 | Structural overview of the research design | 9 |
|-------------|---|-----|
| Figure 1.2 | Map of Kingdom of Saudi Arabia | 11 |
| Figure 2.1 | Systemic Model of Transfer of Learning | 28 |
| Figure 3.1 | The phases of the Explanatory Sequential Mixed Method ap- | |
| - | proach. | 91 |
| Figure 4.1 | Gender of nurses (N = 385) | 133 |
| Figure 4.2 | Age of nurses $(N = 385)$ | 134 |
| Figure 4.3 | Nurses' Highest Education Qualifications distribution (N = | |
| - | 385) | 137 |
| Figure 4.4 | Top four applications of what was learned before $(N = 384)$ | 177 |
| Figure 4.5 | Learner types ($N = 384$) | 179 |
| Figure 4.6 | Top three motivations to learn $(N = 384)$ | 181 |
| Figure 4.7 | Top three motivations to apply in practice $(N = 384)$ | 182 |
| Figure 4.8 | Top five learning styles of preferences $(N = 384)$ | 184 |
| Figure 4.9 | Gender of clinical facilitators ($N = 47$) | 187 |
| Figure 4.10 | Age of clinical facilitators distribution $(N = 47)$ | 188 |
| Figure 4.11 | Clinical Facilitators' Highest Education Qualifications distri- | |
| | bution $(N = 47)$ | 190 |
| Figure 5.1 | Systemic Model of Transfer of Learning | 294 |
| Figure 5.2 | Application of Systemic Model of Transfer of Learning as | |
| | aligned from the citation by Donovan and Darcy | 295 |
| Figure 6.1 | A copy of a part of the Google Form illustrating a draft action | |
| | plan with the embedded validation instrument | 337 |
| Figure 6.2 | Highest level of nursing education $(N = 12)$ | 348 |

LIST OF ANNEXURES

ANNEXURE 1 Recruitment letter: Request of names list of registered nurses and clinical resource nurses from Nurse Managers (gatekeepers) KAMC

512

| ANNEXURE 2 | Recruitment letter: Request of names list of registered | |
|-------------|---|-----|
| | nurses and clinical resource nurses from Nurse Managers | |
| | (gatekeepers) KASCH. | 513 |
| ANNEXURE 3 | Information letter and consent form | 514 |
| ANNEXURE 4 | Questionnaire 1: Phase 1 (professional nurses) | 518 |
| ANNEXURE 5 | Questionnaire 2: Phase 2 (professional nurses) | 527 |
| ANNEXURE 6a | UINISA Ethical approval | 538 |
| ANNEXURE 6b | Ethical extension UNISA REC | 539 |
| ANNEXURE 7a | Nursing Services Permission to conduct research | 540 |
| ANNEXURE 7b | IRB approval SP 18/036/R | 541 |
| ANNEXURE 7c | IRB Annual Extension SP 18/036/R. | 542 |
| ANNEXURE 7d | IRB 6 Months Extension SP 18/036/R. | 543 |
| ANNEXURE 8 | Questionnaire 3: Phase 3 (clinical facilitators) | 544 |
| ANNEXURE 9 | Invitation for participation in the pre-test for the validation | |
| | of the action plan | 565 |
| ANNEXURE 10 | Recruitment letter: Round 1 (Delphi round 1) | 567 |
| ANNEXURE 11 | Storage and management of data collected | 569 |
| ANNEXURE 12 | Draft 1: Action plan with embedded validation tool on | |
| | Google Forms | 570 |
| ANNEXURE 13 | Recruitment letter: Round 2 e- Delphi. | 600 |
| ANNEXURE 14 | Gatekeeper letter: Request to recruit Delphi panellists for | |
| | round 1 | 602 |
| ANNEXURE 15 | Draft 2: Action plan with embedded validation tool | 604 |
| ANNEXURE 16 | Gatekeeper letter to recruit panellists for round 2 | 617 |
| ANNEXURE 17 | Findings: during Round 2 from Action Statements 1 to 7 | 619 |
| ANNEXURE 18 | Draft 3: Action plan with embedded validation tool. | 634 |
| ANNEXURE 19 | Recruitment letter round 3 | 645 |
| ANNEXURE 20 | Gatekeeper letter to recruit panellists for round 3 | 647 |
| ANNEXURE 21 | Findings: during Round 3 from Action Statements 1 to 7 | 649 |
| ANNEXURE 22 | Validated Action Plan | 656 |
| ANNEXURE 23 | Certificate of Language Editing | 662 |

LIST OF ACRONYMS AND ABBREVIATIONS

- AACN American Association of Critical Care Nurses
- ADLs Activities of daily livings
- ADPIE Assessment, Diagnosis, Planning, Implementation, Evaluation
- ATC Around the clock
- BPI The Brief Pain Inventory
- BPS Behavioural Pain Scale
- **CAM-** Complimentary medicine and alternative medicine

| CBAHI | Central Board for Accreditation of Healthcare Institutions |
|-----------|--|
| СВТ | Cognitive behavioural therapies |
| CNE | Center of Nursing Education. |
| CNPI | Checklist of Nonverbal Pain Indicators |
| COLDSPA | Character, onset, location, duration, severity, pattern, and associated factors |
| СРОТ | Critical Care Pain Observation |
| СОХ | Cyclooxygenase enzyme |
| CRIES | Crying, Required oxygen, Increased vital signs, Expression, Sleep- lessness |
| CRN | Clinical Resource Nurse |
| CSPMS UK | Core Standards for Pain Management Service, United Kingdom |
| EMLA | Eutectic Mixture of Local Anaesthetics |
| FLACC | Face, Legs, Activity, Crying, Consolability |
| FPS | Wong-Baker FACES Pain Rating Scale |
| FPS-R | Revised FACES Pain Rating |
| IMI | Irish Management Institute |
| JCI | Joint Commission International |
| KAIMRC | King Abdullah International Medical Research Center |
| KSA | Kingdom of Saudi Arabia |
| LF-MPQ | Long -Form of McGill Pain Questionnaire |
| NIC | Nursing Interventions Classification |
| NIPS | Neonatal Infant Pain Scale |
| NMDA | N-Methyl D-Aspartate |
| NOPPAIN | Non-communicable Patients Assessment Instrument |
| NPASS | Neonatal Pain Agitation and Sedation Scale |
| NSAIDS | Nonsteroidal anti-inflammatory drugs |
| OLD CARTS | Onset of pain, location of pain, characteristics of pain, aggravating factors of pain, relieving factors, timing and severity |
| OPQRSTUV | Onset of pain, provoking/palliating, quality, region/radiation of pain, timing/treatment, understanding impact on one and values |
| PACSLAC | Pain Assessment Checklist for Seniors with Limited Ability to Com- municate |
| PAINAD | Pain Assessment in Advanced Dementia Scale |
| PIPP | Premature Infant Pain Profile |
| | |

| PPQ | Vani Thompson Paediatric Pain Questionnaire |
|--------|--|
| PQRST | Provocation and palliation symptoms, quality of pain, region and ra- diation of pain, severity of pain and timing |
| QOL | Quality of life |
| RNAO | Registered Nurses' Association of Ontario |
| WILDA | Words to describe pain, intensity, location, duration, and aggravat- ing/alleviating factors |
| SF-MPQ | Short-Form of McGil Pain Questionnaire |
| SPSS | Statistical Package for the Social Sciences |
| TENS | Transcutaneous electrical nerve stimulation |
| TOL | Transfer of learning |
| QUESTT | Question the child/caregiver, use pain rating tools, evaluate behav- iour, sensitise parents, take cause of pain into account and take pain into account |
| VAS | Visual Analogue Scale |
| VDS | Verbal Descriptor Scale |
| VRS | Verbal Rating Scale |

CHAPTER ONE: OVERVIEW OF THE STUDY

1.1 INTRODUCTION AND BACKGROUND TO THE RESEARCH PROBLEM

As primary caregivers, nurses spend more time with patients than other healthcare providers. Patient outcomes and the well-being of every individual patient are therefore affected by the quality of nursing care they receive. Nurses are responsible for ensuring patient safety, including assessing and managing patients' pain to promote optimal pain management, using interventions based on individual patient's needs. Every nurse, therefore, has an essential role and responsibility towards pain management (Matzo & Sherman 2015:474; Cooper & Gosnell 2015:473; Arnstein 2010:5-10).

Pain management does not always form part of the curriculum of undergraduate programmes, or when included, there is limited input or emphasis on it (Dongara, Nimbalkar, Phatak, Patel & Nimbalkar 2017:31; Latchman 2014:11). This might be one reason for the low priority given by nurses to pain management (Baird 2015:153; Ferrel, Coyle & Paice 2015:136; Carter & Simons 2014:25). To overcome this barrier, some teaching hospitals established centres of nursing education that included pain management experts, such as pan team members or specialist nurses responsible for training other professional nurses (Drake & Williams (2017:9). Moreover, Dongara et al. (2017:25), Pretorius, Searle and Marshall (2015:378), Krokmyrdal and Andenaes (2015:792), Tse and Ho (2014:10); Ekim and Ocakci (2013:e226) also mention that other centres of nursing education provide ongoing education to improve the knowledge, skills and attitudes towards pain management. In two hospitals in Riyadh, Kingdom of Saudi Arabia, pain team members teach pain management twice monthly in a four-hour workshop in the Centre of Nursing Education (CNE). These pain management teaching courses are formally conducted at the CNE and in the respective units as unit-based in-service education opportunities.

Despite all training opportunities available, pain management teams and clinical facilitators still experience persistent challenges and barriers related to pain management skills, possibly due to challenges in the transfer of learning of pain management competencies of nurses. Even though pain management education is highly prioritised in the mentioned two specific Saudi Arabian teaching hospitals, it appears as if nurses are not compliant with applying what they learned despite repetitive educational efforts and the availability of guidelines about pain management (Gordon Rees, McCausland, Pellino, Sanford-Ring, Smith-Helmenstine & Danis 2008:509).

Patient satisfaction surveys are considered the most valid determinant of patient satisfaction and are used by hospitals to measure quality care and evaluate whether that service provides adequate pain management care (Khoie, Tabrizi, Khorasani, Rahimi & Marhamati 2017:1; Karabulut, Aktas, Gürcayir, Yilmaz and Gökmen 2015:17). Pain reassessment also contributes to patient satisfaction (Glowacki 2015:38; DeVore, Clontz, Ren, Cairns & Beach 2016:e23).

Some surveys indicated that patients were not satisfied with pain management in both the study hospitals, as patients indicated that the nurses did not evaluate the effectiveness of analgesia given to them (Acute Pain Service Quality Improvement Pain Data 2017). An average of only 40% of patient satisfaction regarding pain reassessments after interventions was recorded (Acute Pain Service Quality Improvement Pain Data 2017).

Poor pain management might be due to a possible challenge with the transfer of learning of pain management competencies of nurses (DeVore et al. 2016:e23). To conquer the pain management theory-practice gap, educators, supervisors, and clinical facilitators may play a major role in teaching nurses about pain management and the appropriate scales to use for conducting the assessment. They need to understand the characteristics of their students, thus enabling them to apply information obtained during training in order to improve pain management (Keefe & Wharrad 2012:e70; Donovan & Townsend 2011:32). The transfer of learning in pain management education, therefore, plays an essential role in improving nurses' competencies (Long 2013:225; Tse & Ho 2014:9). Educators should comply with the principles of the transfer of learning. These principles refer to (1) selecting the correct teaching approach, (2) identifying the characteristics and learning style of the students, (3) choosing the learning content, and (4) ensuring a learning climate (Donovan & Darcy 2011:123; Botma, Van Rensburg, Coetzee & Heyns 2015:499; Botma & MacKenzie 2016:105).

The literature regarding the transfer of learning justifies the need to develop an action plan to enhance the transfer of learning of nurses' pain management competencies.

1.2 STATEMENT OF THE RESEARCH PROBLEM

The effects of poor pain management contribute to patient dissatisfaction, prolonged institutionalization, readmissions, and increased costs (Hardin & Kaplow 2016:279; Arnstein 2010:48). Patients with inadequate pain management have a reduced capacity to perform physical and complex cognitive tasks due to direct distraction of pain and or related to associated fatigue, sleep deprivation, depression and anxiety (Cardenas & Hooton 2015:8; Glowacki 2015:35).

Therefore, pain management education for health care professionals is highly prioritised internationally to improve patient outcomes and patient satisfaction. In countries such as Brazil, the Ministry of Health created a national program for professional education and assistance with pain and palliative care (De Freitas, De Castro, Castro & Heineck 2014:808). At the same time, in the United Kingdom, the Core Standards for Pain Management Service (CSPMS) prioritises professional education in pain management (CSPMS UK 2015:137).

Although pain management education is also highly prioritised in the two Saudi Arabian teaching hospitals, "Hospital A" and "Hospital B" nurses still demonstrate poor pain management competencies (Eid, Manias, Bucknall & Almazzrooa 2014:e34; Alqahtani & Jones 2015:47). A factor, amongst others, might be the challenges about the transfer of learning of pain management competencies. An action plan to enhance the transfer of learning of pain management competencies of nurses might contribute to better pain management competencies and ultimately to better patient outcomes.

1.3 RESEARCH AIM/PURPOSE

This study aimed to develop an action plan to enhance the transfer of learning (TOL) of pain management competencies of nurses in Saudi Arabian teaching hospitals.

1.4 RESEARCH OBJECTIVES

The objectives of this study were to:

- 1) Identify and describe the resources available to conduct pain assessments (Phases 1 and 3).
- 2) Identify and describe nurses' characteristics and learning styles that enhance the transfer of pain management competencies (Phase 2).
- 3) Explore the teaching approaches employed by the clinical facilitators during the pain management education of nurses (Phase 3).
- 4) Describe the learning content regarding pain assessment and management (Phase 3).
- 5) Describe the transfer of a learning climate within the hospitals' nursing care areas (Phase 3).
- 6) Develop an action plan to contribute to enhancing the transfer of learning of pain management competencies of nurses (Phases 4 and 5).

1.5 RESEARCH QUESTIONS

Central question: What action plan can be developed within Saudi Arabian teaching hospitals to enhance the transfer of pain management learning?

The following subsidiary-questions further structured the research study:

- 1) What are the available resources for pain assessment?
- 2) Which characteristics and learning styles of nurses enhance the transfer of pain management competencies?
- 3) Which teaching approaches are employed by the clinical facilitators during the pain management education of nurses?
- 4) What is the learning content regarding pain assessment and management?

- 5) What is the transfer of a learning climate within the hospitals' nursing care areas?
- 6) Can an action plan to enhance the transfer of learning of pain management competencies of nurses be developed?

1.6 THEORETICAL FRAMEWORK

This study adopted the Systemic Model of Transfer of Learning, as developed by Kontoghiorghes (2004) and adapted by Donovan and Darcy (2011:125). Donovan and Darcy (2011:125) indicated that transfer occurs within a specific system where each factor influences the transfer of learning.

This model suggests individual and organizational performance as the link between the transfer of learning and work environment characteristics (Donovan & Darcy 2011:124). The model is based on the idea that four factors influence the transfer of learning in the workplace: 1) trainee characteristics (the nurses), 2) training design (learning content), 3) training transfer climate and 4) work environment as illustrated by Donovan and Darcy (2011:125), Botma and MacKenzie (2016;105), Ma, Bai, Bai, Ma, Yang and Li (2018:2).

This model is appropriate within this study context as all four factors are applicable if an action plan to enhance the transfer of learning of pain management competencies of nurses is to be developed. The nurse is an individual with specific characteristics that include cognitive ability, personality and motivation to learn and transfer that knowledge. These characteristics influence the transfer of learning of pain management competencies. Another aspect is the pain management educational design that includes principles of learning, sequencing, learning content and material consistent with the clinical requirements, which may also influence the transfer of pain management competencies. The nurses need to apply their acquired competencies in pain management in a favourable transfer climate supported by supervisors, clinical facilitators, and co-workers to enhance the transfer of learning. To learn to transfer successfully, nurses need the resources and opportunities to apply their new skills and abilities to the workplace. The Systemic Model of Transfer of Learning is significant for this study as nurses are primarily responsible for managing pain.

1.7 DEFINITIONS OF KEY THEORETICAL CONCEPTS

1.7.1 Action plan

An action plan is a formal or informal strategy to help the learner to change in doing things differently by setting a goal, creating a vision, thinking about a timeframe, considering boundaries and resources, alternatives, options, implementation and consequences of change to help the learner and influence others visible to them (Cranton 2016:131).

1.7.2 Clinical facilitator

A clinical facilitator is described by Tollefson, Bishop, Jelly, Watson and Tambree (2011:viii) as a skilled or experienced nurse who is knowledgeable and confident, employed within assigned clinical areas, supernumerary to the ward and solely responsible for clinical education and support.

1.7.3 Competency

Competency is a process that continually verifies an individual's capacity to integrate knowledge (cognitive ability), skills (psychomotor capabilities), and attitudes (affective inclinations) reflected in the quality of clinical practice that benefits others, which can be evaluated by professional standards and be developed and enhanced through professional training and reflection (Avis 2016:123; Still, Sarwer & Blankenship 2014:188).

1.7.4 Learning

According to Kolb (2015:49), learning is the process whereby knowledge is created through experience transformation.

1.7.5 Transfer of Learning (TOL)

Transfer of Learning is the application of knowledge, skills and attitudes obtained in learning situations in practice (Donovan &Townsend 2011:6).

1.7.6 Nurse

In most countries, a nurse is either registered, licensed or qualified and is a person equipped and authorised to: "(1) engage in the general scope of practice, promotion of health, prevention of illness, and care of physically ill, mentally ill, and community settings; (2) carry out healthcare teaching; (3) participate as a member of the healthcare team; (4) supervise and train nursing and healthcare auxiliaries; and (5) be involved in research" (Oulton & Caldwell 2008:565).

1.7.7 Pain management

Pain management is the process or techniques of providing medical or nursing care that prevents and provides treatment or interventions that alleviate, reduce or control the levels of pain an individual experiences for the longest possible time (DeWit & Kumagai 2013:134)

1.8 KEY OPERATIONAL DEFINITIONS

1.8.1 Clinical facilitator

Clinical facilitator in this study context refers to a "clinical resource nurse" (CRN) employed within the assigned clinical areas, supernumerary to the ward and solely responsible for clinical education and the support of nurses.

1.8.2 Competency

Competency in this study refers to nurses' knowledge, skills and attitudes pertaining to the comprehensive assessment and management of pain.

1.8.3 Nurses

In this study, the word "nurses" refers to expatriate and Saudi registered nurses, all registered with the Saudi Council for Health Specialities and working in medical, surgical, oby-gynae, cardiac and paediatric wards with experience in providing pain management.

1.9 RESEARCH METHODOLOGY

1.9.1 The Research Paradigm

The study was done within the pragmatist paradigm using a mixed methods research methodology. Polit and Beck (2021:587), Johnson and Christensen (2012:32), and Creswell and Clark (2011:41) assert that the pragmatist or pluralist paradigm is associated with mixed methods research and assumes that the focus is on the consequences of research, on the primary importance of the question asked rather than the methods, and on the use of multiple methods of data collection to inform the problems under investigation. This means that the research design should be planned and conducted based on what will best help to answer the research question of the study (Minton & Lenz 2019:93; Johnson & Christensen 2012:32). Mixed methods research enables the researcher to avoid the limitations of a single approach as both qualitative and quantitative in a single study are complementary (Polit & Beck 2021:587) and deemed appropriate (refer to Section 3.3.1).

1.9.2 Research design

An explanatory sequential mixed method design was adopted for the study. The explanatory sequential design started with the collection and analysis of quantitative data, which has the priority for addressing the study's questions, followed by the subsequent collection and analysis of qualitative data as suggested by Polit and Beck (2021:594), DeCuir-Gunby and Schutz (2017:86), and Creswell (2015:37-38). The research was conducted in different phases, where different approaches and research techniques surfaced, as illustrated in Figure 1.1.



Figure 1.1 Structural overview of the research design

1.9.2.1 Phase 1: (Quantitative)

In Phase 1, the resources available for pain assessment were assessed using a selfadministered questionnaire to gather data from nurses and clinical facilitators.

1.9.2.2 Phase 2: (Quantitative)

In Phase 2, the nurses' characteristics and learning styles were assessed using a selfadministered questionnaire.

1.9.2.3 Phase 3: (Quantitative)

In Phase 3, the teaching approaches employed, the learning content and the learning climate within hospitals were described from data obtained from the clinical facilitators using self-administered questionnaires with closed and open-ended questions.

1.9.2.4 Phase 4:

In Phase 4, the combined data from Phases 1, 2, and 3, as well as a literature review, were utilised to develop an action plan to enhance the transfer of learning of pain management competencies.

1.9.2.5 Phase 5 (Qualitative)

During Phase 5, the developed action plan to enhance the transfer of learning of pain management competencies of nurses was validated by a purposively selected panel of clinical facilitators and nurses to seek agreement on the actions, the responsible persons as well as the time frame included in the plan to ensure quality, faithful interpretation and applicability within the specific context (Baldwin 2018:20, DeCuir-Gunby & Schutz 2017:190; Bazeley 2013:408; Cohen, Manion & Morrison 2011:404).

1.9.3 Research setting

A research setting is defined by Polit and B,eck (2021:42), Richards (2021:58) and Grove, Gray and Burns (2015:38) as the specific places where information is collected or the location in which a study is conducted.

The Kingdom of Saudi Arabia (KSA) is a sovereign Arab state in Western Asia, also called the Middle East and is divided into 13 regions bordered by Jordan and Iraq to the north, Kuwait to the northeast, Qatar, Bahrain and the United Arab Emirates to the East, Oman to the southeast and Yemen to the south. The Gulf of Aqaba separates the KSA from Israel and Egypt (see Figure 1.2).

Riyadh is the capital city, and within the capital city, there are four teaching hospitals. The study was conducted in two purposively selected teaching hospitals, as access to the other two hospitals was impossible due to the researcher's work permit control.



(http://annamap.com/saudi-arabia/saudi-arabia-map.jpg)

Figure 1.2 Map of the Kingdom of Saudi Arabia
The two teaching hospitals cater to all Ministry of National Guard employees, their families, and civilians in need of medical care. In the backyard of the two hospitals is King Saud bin Abdulaziz University for Health Sciences, which collaborates with the two hospitals for student training with faculties of nursing, medicine, pharmacy, dentistry, science and health professionals and a postgraduate training centre for all health care providers. In the same vicinity is the King Abdullah International Medical Research Center (KAIMRC) for all medical and nursing research (see Figure 1.2). For continued nurse professional development, the two hospitals use one nursing education centre erected in their backyard.

Even though there are other nursing divisions, it was decided to conduct the study in five nursing care divisions that mainly met the highest eligibility criteria of the study, namely 19 (nineteen) medical wards, 9 (nine) surgical wards, 9 (nine) paediatric wards, 4 (four) cardiac wards and 7 (seven) obs-gynae (obstetric-gynaecologist) wards with professional nurses and clinical facilitators ("clinical resource nurses") working in the selected hospitals.

In Hospital A, there are 15 (fifteen) medical wards, 8 (eight) surgical wards, 4 (four) cardiac wards, 1 (one) paediatric cardiac ward and 5 (five) obs-gynae wards, while in Hospital B, there are 4 (four) medical wards,1 (one) surgical ward and 8 (eight) paediatric wards (see Table 1.1).

| Nursing Care Divi- sions | Wards in Hospital A | Wards in Hospital B |
|-----------------------------|---------------------|---------------------|
| 19 Medical wards | 15 | 4 |
| 9 Surgical wards | 8 | 1 |
| 9 Paediatric wards | 1 | 8 |
| 4 Cardiac wards | 4 | 0 |
| 7 Obs-gynae wards | 5 | 2 |

| Table 1.1 | I The five | -nursing ca | re division | s of ho | spitals A | A and B |
|-----------|------------|-------------|-------------|---------|-----------|---------|
|-----------|------------|-------------|-------------|---------|-----------|---------|

1.9.4 Population

Ngulube (2021:299), Patel (2018:85) and Jha (2014:182) refer to the population as the entire group, person or object that is of interest to the researcher. In other words, the population meets the variable the researcher is interested in studying. The study population was all 1462 professional registered nurses (Hospital A = 1041 and Hospital B = 421) working in the five nursing care divisions (see Table1.2) as well as the 47 clinical facilitators (Hospital A = 32 and Hospital B = 15) working in the two teaching hospitals. Table 1.2 below indicates the population size of nurses and clinical facilitators ('CRNs') in both hospitals' five nursing care divisions.

| | Hospi | tal A | Hospi | tal B |
|-----------------------------|------------------------------|--|------------------------------|--|
| Nursing Care Di- visions | Population size of nurses | Population size of Clini- cal facilitators | Population size of nurses | Population size of Clini- cal facilitators |
| 1. Medical wards | 459 | 14 | 135 | 5 |
| 2. Surgical wards | 282 | 8 | 35 | 1 |
| 3. Paediatric wards | 20 | 1 | 216 | 8 |
| 4. Cardiac wards | 166 | 4 | 0 | 0 |
| 5. Obs-gynae wards | 114 | 5 | 35 | 1 |
| Total | 1041 | 32 | 421 | 15 |

 Table 1.2
 Accessible Population

1.9.5 Sample and sampling methods

A sample is a subset of the population that is selected for a study (Polit & Beck 2021:802; Grove et al. 2015:46). Sampling can be defined as the process of obtaining a small number of participants for research to draw conclusions regarding the whole population (Ngulube 2021:300) because it is generally impractical to study an entire population.

1.9.5.1 Phases 1, 2 and 3: Quantitative

A probability sampling method, namely a proportionate stratified sampling method (Messinger & Guadalupe-Diaz 2020:343; Dillman, Smyth & Christian 2014:76), was used to select the registered nurses as well as the clinical facilitators from the five nursing care divisions within the two participating hospitals. The rationale for choosing stratified random sampling in Phases One, Two and Three was to enhance representatives and reduce sample error (Maxfield & Babbie 2018:214; Leedy & Ormrod 2015:179; Polit & Beck 2014:180; Dillman et al. 2014:76). Each of the five nursing care divisions had equal representation as the sample size from each nursing care division stratum was determined at a 95% level of confidence as recommended by Polit and Beck (2021:387) and Babbie (2020:206). To calculate the size of each stratum and the sample of this study, the Raosoft 2011 sample size calculator was used to calculate the margin of error and confidence level, as Zizile and Tendai (2018:229) recommend. Table 1.3 indicates the population, the sample size in each stratum, and the total size of all five strata, calculated at a 95% confidence level of each nursing care division of each participating hospital. The rationale for using a 95% level of confidence in each nursing care division in both hospitals was to draw valid and reliable data by drawing a proportional sample size, as supported by Privitera (2022:42) and Jha (2023:143).

| | Medic wards | al | Surgi ward | cal s | Paed ward | liatric s | Card ward | iac s | Obs-gynae | wards | Total Popula | tion |
|------------------------------|----------------|------------|---------------|------------|--------------|--------------|--------------|------------|------------|------------|-----------------|------------|
| Nursing care di- vision | Hospital A | Hospital B | Hospital A | Hospital B | Hospital A | Hospital B | Hospital A | Hospital B | Hospital A | Hospital B | Hospital A | Hospital B |
| Population size of nurses | 459 | 135 | 282 | 35 | 20 | 216 | 166 | 0 | 114 | 35 | 1041 | 421 |

| Fable 1.3 Population and sar | ple in each stratum of the | participating h | nospitals |
|------------------------------|----------------------------|-----------------|-----------|
|------------------------------|----------------------------|-----------------|-----------|

| Sample size of nurses | 210 | 101 | 163 | 33 | 20 | 139 | 117 | 0 | 89 | 33 | 281 | 202 |
|--|-----|-----|-----|----|----|-----|-----|---|----|----|-----|-----|
| Population size of Clinical facili- tators | 14 | 5 | 8 | 1 | 1 | 8 | 4 | 0 | 5 | 1 | 32 | 15 |
| Sample size of Clinical facilita- tors | 14 | 5 | 8 | 1 | 1 | 8 | 4 | 0 | 5 | 1 | 30 | 15 |

Hospital A mainly had nurses working in adult patients' wards, while Hospital B was more for paediatric wards. There are also adult patients' wards.

A recruitment letter (see Annexure 1) was e-mailed to the nurse managers who acted as gatekeepers, together with the letter to the gatekeeper (see Annexure 2) of each ward in the participating hospitals. The nurse manager was requested to select the participants and distribute the recruitment letters via e-mail to nurses and clinical facilitators who have complied with the following criteria:

- Nurses who attended at least one pain management workshop within the past 3 years.
- 2. Nurses who attended ward in-service training about pain management in the past 12 months.
- 3. Participants who were willing and comfortable to be interviewed in English.
- 4. Clinical facilitators who educate nurses about pain management in the five nursing care divisions.

The researcher randomly selected the sample in each stratum from the volunteers who provided contact details to the nurse managers (gatekeepers).

1.9.5.2 Phase 5, Qualitative

The Delphi technique is defined as a structured group communication process in research within the social sciences to identify degrees of consensus on practice issues within a group of diverse, knowledgeable advisors (Polit & Beck 2021:236; Lloyd & Stirling 2015:3). It was used in Phase 5. A purposive non-probability sampling was used to select a panel of two clinical facilitators (one from each hospital) and ten nurses (five from each nursing care division in both hospitals) to ensure representativeness (Polit & Beck 2021 268; Mellenbergh 2019:23). The participants were selected based on their consent to participate, their expertise in pain management, and their active participation in pain management either as a nurse or as a facilitator.

1.9.6 Data collection methods and procedures

1.9.6.1 Administering the data collection instrument

In Phases 1, 2 and 3 (refer to Sections 3.9, 3.10, 3.11 and 3.13 for detailed discussion), self-designed questionnaires were used (see Annexures 4, 5 and 8) with closed-ended and open-ended questions at the end of the last section of each questionnaire.

As mentioned, a memo was sent to gatekeepers (nurse managers of units) through emails to recruit potential participants (see Section 3.13) who comply with the inclusion criteria (see Annexures 1 & 2). From the list of volunteers and contact details received from the nurse managers, the researcher randomly selected the sample and delivered the questionnaire to the respondents in person. After completing the questionnaires, the respondents placed them in a closed box provided in each ward, and they were fetched afterwards by the researcher.

In Phase 4 (refer to Section 5.2.5 for detailed discussion), an instrument was not used as the data obtained in Phases 1-3 and the literature review were used to develop the draft action plan.

In Phase 5, the Delphi technique was used. Following the five phases, the Delphi technique, using open-ended questions, was implemented, as advised by Grima,

Sood and Özen (2023:28), and Botma, Greeff, Mulaudzi and Wright (2010:253-254). The panel of two clinical facilitators and ten nurses validated the developed action plan as a representative group. The structured tool with open-ended questions (see Annexure 10), called a validation tool, was used to give the panellists space to provide their written opinions (Grima et al. 2023:28; Botma et al. 2010:253). They used it to make suggestions for improvement. The developed action plan, with the incorporated assessment tool, was circulated via e-mail to each panel member. They were allowed ten working days to assess the plan and provide recommendations and feedback. After receiving feedback, the researcher collated and analysed the feedback and resent a revised draft to the panellists. The process was continued until a 75% agreement was reached.

1.10 MEASURES TO ENHANCE THE QUALITY OF THE STUDY

The study addressed method-specific quality criteria in the quantitative and qualitative components of mixed methods research to help achieve external consequences, as Yardley and Bishop (2008, cited in Bishop 2015:8) recommend.

1.10.1 Validity and reliability

In Phases 1, 2 and 3, the principles of validity and reliability were applied to minimize errors that might arise from measurement problems in the research study.

1.10.1.1 Reliability

Reliability is "the consistency, stability, and repeatability of observations or measures" (Gray and Grove 2021:458; Curry & Nunez-Smith 2015:176). The reliability of the survey questionnaire was assessed to ensure stability and internal consistency. Testand-retest for reliability of the questionnaires was used to measure changes in people over two weeks, as recommended by Rubin (2020:99) and Cairns (2019:177). A total of 10% of the total sample of nurses and clinical facilitators for each hospital was pretested with subjects from other nursing divisions with characteristics similar to those of the main study (see Section 3.10 for detailed discussion). The actual survey results were then compared and correlated with the initial results found in the pilot study and expressed by the "correlation coefficient r". The correlation above .70 or .80 was deemed acceptable stability (Rubin 2020:99; Rubin & Babbie 2014:219). Cronbach's alpha (α) coefficient test was used for assessing internal consistency using the R Statistical Software ("R") as recommended by Rey, Pena and Neto (2020:71). Inter-item correlation reliability was examined based on the correlation matrix of all items on the scale, corrected item-total correlation, and alpha if an item was deleted to improve scale reliability (Hogan 2019:107).

1.10.1.2 Validity of instrument

The concept validity of an instrument refers to the extent to which an instrument actually reflects or is able to measure the construct being examined (Khakshooy & Chiappelli 2018:35). Face and content validity were used to validate the questionnaire, as embraced by Gray and Grove 2021:463. The face and content validity were evaluated by the supervisor and a scientific committee that provided input and feedback for modifications as supported by Shrotryia and Dhanda (2019:4) (refer to Section 3.11 for detailed discussion). The questionnaires were pre-tested for refinement before being utilised in the main investigation, as Jönsson and Prins. (2019:63) recommend.

1.10.2 Rigour

In Phase 5, rigour or trustworthiness, as described by Lincoln and Guba (1985) cited in Polit and Beck (2021:569) and Liamputtong (2013:25-27), was adopted. Trustworthiness is the criterion for evaluating the rigour of qualitative research (Grundström 2018:75; Rubin & Babbie 2014:485). The criteria credibility, dependability, confirmability, transferability, and authenticity, as described by Polit and Beck (2021:322-323), were utilised (see Section 6.2.11 for detailed discussion).

1.10.3 Inference quality

Inference quality refers to the believability and accuracy of deductively derived conclusions from a mixed-method study (Polit and Beck 2021:605). Phases 1, 2, and 3 data and a literature review were combined in Phase 4 to develop a draft action plan to enhance transfer learning of nurses' pain management competencies. In Phase 5, a developed and validated action plan, based on the panel members' consensus, resulted in the study's outcome.

1.11 DATA ANALYSIS

This is a five-phase design in which quantitative data analysis was followed by qualitative data analysis as proposed by DeCuir-Gunby and Schutz (2017:186)

1.11.1 Phases 1, 2 and 3: Quantitative data analysis

The quantitative data in Phases 1, 2, and 3 were analysed using the Statistical Package for the Social Sciences (SPSS) version 25. The data were explored by conducting descriptive analyses and variances of response to each item, as explained by Polit and Beck (2021 366). The open-ended questions in Phases 1, 2 and 3 were coded, and a thematic content analysis was done to draw valid inferences as mentioned by Gottschalk (20,20: 57) (see Section 4.2.3, 4.3.4 and 4.4.6, where these are discussed in detail).

1.11.2 Phase 5, Qualitative data analysis

The developed action plan was scrutinised and assessed by the panellists who took part in the Delphi technique. All comments and recommendations received during each round were thematically analysed using a six-phase guide as described by Clarke and Braun (2013:121) and Maguire and Delahunt (2017:3354). This consisted of Step 1, becoming familiar with the data; Step 2, generating initial codes; Step 3, searching for themes; Step 4, reviewing the themes; Step 5, defining the theme; and Step 6, writing up the themes as a final refinement of the themes and subthemes to interact and relate to each other to create the drafted action plan. The recommendations were implemented in a new draft action plan (see Section 6.2.10 for detailed discussion). All Likert scale items were also analysed and incorporated in the new draft version of the action plan.

1.11.3 Integration of the findings

Merged integration occurred after both the quantitative Phases 1, 2 and 3 and qualitative Phase 5 data collection and analysis were completed, as described by Curry and Nunez-Smith (2015:10). The merged integration contributes to the final agreedupon action plan developed to enhance the transfer of learning of pain management competencies of nurses (DeCuir-Gunby & Schutz 2017:119).

1.12 ETHICAL CONSIDERATIONS

The principles of ethics and moral issues were adhered to as described by Polit and Beck (2021:131--148) and Ngulube 2019:76). They include the following items.

1.12.1 Respect for human dignity and self-determination

The information letter (see Annexures 3 and 9) informed the respondents that participation is voluntary and that they could withdraw at any time without fear of penalty.

1.12.2 Informed consent

An information letter (see Annexure 3) and consent letter (see Annexure 9) were provided to ensure the participants knew all information relevant to the study and their participation. The participants were asked to give informed written consent before participating in the study.

1.12.3 Confidentiality and anonymity

All information and data collected were portrayed confidentially. Confidentiality and anonymity were maintained by pledging to the participants that their information would not be disclosed to their leadership, colleagues or anybody else and their identity would not be published in the research report or thesis. Complete confidentiality was ensured by keeping the completed questionnaires in locked storage to prevent unauthorised access to the data.

1.12.4 Researcher-Participant Relationship

The researcher maintained rapport and respect with the participants while negotiating their participation in the study by assuring them that their participation would not be used against them in any way, ensuring the relationship was not exploited (Polit & Beck 2021:144).

1.12.5 Protecting the rights of the institution

Permission to conduct the research was first sought from the Research Ethics Committee of the Department of Health Studies, Unisa (see Annexures 6a and 6b), followed by application from the King Abdullah International Medical Research Center (KAIMRC) Review Committee and Institutional Review Boards (IRB) of the two selected Saudi Arabian teaching hospitals. The Nursing Administration, KAIMRC and IRB reviewed the proposed study and monitored it to ensure that ethical standards were applied, such as the privacy and confidentiality of the hospitals (see Annexures 7a, 7b, 7c and 7d for IRB approval Memo SP 18-036-R).

1.12.6 Scientific integrity of the research

Polit and Beck (2021:147) and Iphofen (2017:97) emphasise the responsibility of the researcher to monitor the integrity of the research proposal, results and publications. The researcher maintained honesty by avoiding duplicating any other work or any other form of misconduct, such as fabrication, falsification, dishonesty and plagiarism. There was voluntary participation with no compensation.

1.13 ORGANISATION AND STRUCTURE OF THE THESIS

The structure of the study is illustrated in Table 1.4.

Table 1.4 Organisation and structure of the study

| Organisation and structure of the study | | | | | | |
|---|---|--|--|--|--|--|
| Chapter | Chapter outline | Chapter content | | | | |
| number | | | | | | |
| Chapter 1 | Overview of the study | Contains the introduction, background of the study, the problem statement, research pur- pose and objectives, research question, theo- retical framework, key theoretical and opera- tional concepts, the research design and meth- odology and ethical considerations. | | | | |
| Chapter 2 | Literature review | Consists of the literature review related to: 1) Systemic Model of Transfer of Learning by Donovan and Darcy, 2) Transfer of learning and 3) Pain management and tools. | | | | |
| Chapter 3 | Research design and meth- odology | Illuminates the overarching research design Phase 1, 2 and 3 (quantitative phases): methodology and data gathering | | | | |
| Chapter 4 | Data analysis and interpreta- tion | Presents the data analysis and interpretation of the findings from Phases 1 to 3. | | | | |
| Chapter 5 | Phase 4 | Includes a description of Phase 4 of the study: a) Literature review on action plan development b) Development of the draft action plan | | | | |
| Chapter 6 | Phase 5 | Outlines and describes Phase 5 of the study (qualitative phase): a) Methodology b) Validation of the action plan: c) The validated action plan | | | | |
| Chapter 7 | Conclusion, recommenda- tions, and limitations | Deals with the conclusion, recommendations and limitations of the study. | | | | |

1.14 CHAPTER SUMMARY

This chapter provided an overview of the study, the methodology followed, and the intended outcome. Chapter 2 will review the literature on the Systemic Model of Transfer of Learning by Donovan and Darcy, as well as the transfer of learning, pain management, and tools as described in Figure 2.1.

CHAPTER TWO: LITERATURE REVIEW

2.1 INTRODUCTION

A thorough literature search was conducted to prepare and present the literature review. The purpose of the review was to obtain relevant literature that supports the problem statement and could be used as the basis for developing the questionnaires used in Phases 1, 2, and 3 of the study, as illustrated in Table 2.1.

Table 2.1 Organisation and structure of the study

| | Organisation and structure of the study | | | | | | |
|-------------------|---|---|--|--|--|--|--|
| Chapter number | Chapter outline | Chapter content | | | | | |
| Chapter 1 | Overview of the study | Contains the introduction, background of the study, the problem statement, re- search purpose and objectives, research question, theoretical framework, key the- oretical and operational concepts, the re- search design and methodology and ethi- cal considerations. | | | | | |
| Chapter 2 | Literature review | Consists of the literature review related to: 1) Systemic Model of Transfer of Learning by Donovan and Darcy, 2) Transfer of learning and 3) Pain management and tools. | | | | | |
| Chapter 3 | Research design and methodology | Illuminates the overarching research design. Phase 1, 2 and 3 (quantitative phases): methodology and data gathering. | | | | | |
| Chapter 4 | Data analysis and inter- pretation | Presents the data analysis and interpre- tation of the findings from Phases 1 to 3. | | | | | |

| Chapter 5 | Phase 4 | Included a description of Phase 4 of the study: a) Literature review on action plan development. b) Development of the draft action plan. |
|-----------|---|---|
| Chapter 6 | Phase 5 | Outlines and describes Phase 5 of the study (qualitative phase): a) Methodology. b) Validation of the action plan: c) The validated action plan. |
| Chapter 7 | Conclusion, recommen- dations, and limitations | Deals with the conclusion, recommenda- tions and limitations of the study. |

The following databases were searched via the Unisa Library: Academic Journals, Academic Search Premier, BioMedCentral, CINAHL, Cochrane Library, eBook, EM-BASE, Global Health, Health Source: Nursing/Academic Edition, Google Books, Google Scholar, OVID, Medline, Nursing & Allied Health Database, Pubmed Central, PsychINFO, PsyARTICLES, Routledge publishers, South African Journal, Wiley Online Library, and Unisa Institutional Repository. Books published between 2010 and 2018 were included in the review, while articles published between 2013 and 2018 were searched. Additional searches included Clinical Key, Academia.edu and Research gate.net.

The search in the review was limited to the English language. It included keywords such as pain behavioural cues, transfer of learning, training transfer, Systemic Model of Transfer of Learning, pain assessment, comprehensive pain assessment, pain nursing assessment, pain interventions, pain rating scale, pain nursing diagnosis, pain pharmacological interventions, and pain non-pharmacological interventions.

The chapter consists of three main segments: (1) The concept of transfer of learning, (2) The Systemic Model of Transfer of Learning by Donovan and Darcy (2011), and (3) Pain management and pain management tools nurses use for assessment of pain. All searched items were saved in Mendeley's library for easy access and sorting of similar keywords.

2.2 THE CONCEPT TRANSFER OF LEARNING (TOL)

The concept of transfer of learning is widely discussed and defined within the extant literature and was initially penned by Edward Thorndike and Robert Woodworth in 1901, who were interested in how individuals would transfer learning from one context to another (Findlay 2015:34; Snowman & McCown 2015:372). Transfer of learning can be described as the degree to which trainees effectively apply knowledge, skills and attitudes gained in a training context to an application context (Babu & Gayathri 2018:101; Ma et al. 2018:2; Botma & MacKenzie 2016:105; Daffron, Moore & Chicovsky 2015:50; Aluko & Shonubi 2014:641; Shalin 2014:36; Salvendy 2012:517). The term transfer of learning is interchangeably used as "transfer of training", "transfer of knowledge", "training transfer", "learning transfer", or "transfer" (Daffron & North 2011:2; Aluko & Shonubi 2014:641; Seel 2011:3337). According to Dirani (2017:104), Aluko and Shonubi (2014:641), and Kaiser, Kaminski and Foley (2013:5), transfer of learning relates to adult education, vocational or professional training, or workplace education. Transfer of learning involves three processes: (i) learning or mastering the knowledge and skills gained during a training program, (ii) using the new knowledge and skills within the practice, and (iii) maintaining the changed behaviour over time (Aluko & Shonubi 2014:641).

Within a profession such as nursing, transfer of learning is about theory-practice integration, which is the process of application of acquired theoretical knowledge, such as knowledge about pain management and the application of this knowledge in the real world (Botma et al. 2015:499; O'Connor 2015:235). Four hierarchical domains of knowledge, namely, demonstration of knowledge, comprehension of knowledge, application of knowledge, and analysis of knowledge, all play a role in the transfer of learning in the workplace. The Systemic Model of Transfer of Learning discussed in 2.3 describes how the four domains of knowledge influence the transfer of learning.

2.3 SYSTEMIC MODEL OF TRANSFER OF LEARNING BY DONOVAN AND DARCY (2011)

Various frameworks or models related to the transfer of learning have been adopted and adapted in the literature to refer to the transfer of learning. The most quoted and used models are Baldwin and Ford's (1998) "Transfer of Training Model"; Holton, Bates, and Ruona's (2000) "Learning Transfer System Inventory (LTSI) model"; Kirkpatrick's (1996) "Training Evaluations Framework"; and Donavan and Darcy's (2011) "Systemic Model of Transfer of Learning". The first three models mentioned are considered traditional transfer of learning models as they ignore the systemic or holistic factors that affect the transfer of learning (Donovan & Darcy 2011:124; Dirani 2017:101). For the following reasons, the Systemic Model of Transfer of Learning was deemed the most appropriate to be used in this study: (i) This model views learning as systemic in nature and explores factors within a specific system where each factor is interrelated to each other to influence the transfer of learning back to the workplace (Donovan & Darcy 2011:124). (ii) Originally, this model was utilised in human resource development but has been adapted and utilised in higher nursing education institutions to achieve its principal motivation, which is to promote the integration of theory with practice (Donovan & Darcy 2011:125; Botma et al. 2015:499; Botma & MacKenzie 2016:105).

The Systemic Model of Transfer of Learning, as developed by Kontoghiorghes (2004) and adapted by Donovan and Darcy (2011:125), is based on research conducted in the Irish Management Institute (IMI), an executive education institution in the Republic of Ireland (Donavan & Darcy 2011:125). Donovan and Darcy (2011:125) indicate that the transfer of learning in the work environment occurs within a specific system where each factor influences the transfer of learning. This model suggests that individual and organizational performance is the link between the transfer of learning and work environment characteristics (Donovan & Darcy 2011:124). Similar to any other model of transfer of learning, the primary objective is to improve the performance of the individual and or organisation by producing desirable behavioural and or organisational change through the transfer of learning (Donovan & Darcy 2011:123). In this study context, it

can imply that the model's objective will be to improve the pain management competencies of nurses to ensure effective pain relief of patients within the healthcare institution through the transfer of learning. The Systemic Model of Transfer of Learning is described within several dimensions.

2.4 DIMENSIONS OF THE SYSTEMIC MODEL OF TRANSFER OF LEARNING

The Systemic Model of Transfer of Learning designates four dimensions of factors that influence transfer of learning in the workplace namely, (1) trainee characteristics, (2) training design, (3) training transfer climate, and (4) workplace environment characteristics (Donovan & Darcy 2011:125; Ma et al. 2018:2; Botma & MacKenzie 2016;105; Schneider 2014:68). Each of the four factors influencing transfer of learning is illustrated in Figure 2.1.



Figure 2.1 The Systemic Model of Transfer of Learning

Source: Adopted from Donovan and Darcy (2011), cited in Donovan and Darcy (2011:125).

2.4.1 Trainee characteristics

Trainee characteristics refer to how prepared the trainees are to attend training programmes to acquire knowledge, how motivated they are to apply their learning, and how positive they are to improve their performance (Suleiman, Dassanayake & Othman 2016:22; Nazli, Sipon, Zumrah & Abdullah 2015:56; Landy & Conte 2010:321; Donovan & Darcy 2011:123). Although there are several factors influencing the transfer of learning under trainee characteristics, the Systemic Model of Transfer of Learning stipulates factors such as ability, personality, and motivation (Donovan & Darcy 2011:123; Wen & Lin 2014:123; Madagamage, Warnakulasooriya & Wickramasuriya 2014:13; Daffron & North 2011:36).

The "ability factor" of a trainee is an important characteristic that influences the trainee, in this study context, the nurse, to understand, comprehend information, process the information, and then have a positive attitude towards learning and be motivated to learn new material or skills (Daffron & North 2011:36).

2.4.1.1 Ability

An ability-related element of trainee characteristics influences the transfer of learning as it includes both the cognitive and psychomotor abilities that trainees need to possess to be able to learn the content of the training (Wen & Lin 2014:115; Nazli et al. 2014:56). The ability of any trainee, including a nurse, to apply knowledge, skills, and attitudes are influenced by their readiness to learn, self-efficacy, motivation to learn and the motivation to transfer that learning (Landy & Conte 2010:321; Ma et al. 2018:4). The nurses` ability to transfer their knowledge about pain management is enhanced by their general mental ability, general intelligence (cognitive ability), and psychomotor ability (Madagamage, Warnakulasooriya & Wickramasuriya 2014:61; Landy & Conte 2010:321). Nurses or trainees with higher cognitive abilities and experiences after training will demonstrate higher performance for information processing and the ability to retain knowledge (Nazli et al. 2015:56; Wenzel and Cordery 2014:23). Psychomotor ability refers to nurses' utilisation of their physical energy to learn and practice the skills in the clinical area or practising of skills during a learning program (Pillitteri 2014:1015; Bastable 2014:447). The abilities of the nurses or trainees that exert effectiveness in applying knowledge and skills acquired in a training program at the workplace are determined by their characteristics.

2.4.1.2 Personality

Personality can be defined as the intrinsic organisation of an individual's mental world that is stable and consistent over time and in different situations (Gandevani 2017:34). The nurses' (as trainees) personality characteristics that increase the transfer of learning are factors such as personality traits, work-related attitudes, individual goal intentions, and utility expectations (Tonhäuser & Bürker 2016:148; Wenzel & Cordery 2014:14). The personality characteristics of trainees that primarily influence the transfer of learning are intellectual curiosity, conscientiousness, extraversion, openness to experiences, agreeableness, and neuroticism (Laguna & Purc 2016:2; Tonhäuser & Bürker 2016:145; Wenzel & Cordery 2014:13). The trainees, in this case nurses, with more positive personality characteristics during training, are better able to explore, to be flexible, to accept and to adopt new skills (Laguna & Purc 2016:2; Nazli et al. 2015:56). The nurses' (trainees') reaction after attending training as well as their characteristics influence the effectiveness of the transfer of training (Nazli et al. 2015:56). The "motivation factor" of the nurses (trainees) to learn and to transfer learning are directly influenced by their ability and personality.

2.4.1.3 Motivation

The concept of motivation is defined as "the process whereby goal-directed activities are initiated and sustained" (Cook & Artino Junior 2016:997; Andrade & Evans 2013:24). Motivation involves internal processes such as goals, beliefs, views and expectations that shape and direct trainees' behaviours that enhance the transfer of learning (Andrade & Evans 2013:24). The "trainee motivation" can be defined as the extent to which the trainees, or nurses, are interested in attending training, learning from training, and applying the skills and knowledge acquired in training back to the workplace (Landy & Conte 2016:264; Andrade & Evans 2013:24; Landy & Conte 2010:325), thus apply their knowledge about pain management in practice. The motivation factors for the transfer of learning include motivation to learn and transfer, as illustrated in Figure 2.1.

Motivation to learn

Motivation to learn is comprehensively defined as a specific desire of the trainee to join and learn the content of the training program to obtain knowledge and skills (Nafukho, Chakraborty, Johnson & Cherrstrom 2017:333; Kirwan 2016:93; Lee, Lee, Lee & Park 2014:2839). Motivated trainees (nurses) are more likely to perform better because of the increased transfer of learning (Nafukho et al. 2017:349). The nurses who are motivated to learn, take responsibility for their learning and practice self-regulated learning strategies (Hoyle 2015:42; Andrade & Evans 2013:24). The higher motivation of trainees or nurses to learn and participate in education programs influences their transfer of learning and competency (Nafukho et al. 2017:347; Wen & Lin 2014:117; Schneider 2014:88; Kasurkar, Cate, Vos, Westers & Croiset 2013:57). Participation in a training program is guided by self-motivation, self-efficacy, and expectations set by management (Nafukho et al. 2017:347; Tonhäuser & Bürker 2016:142; Daffron et al. 2015:54; Wen & Lin 2014:117; Schneider 2014:88; Kasurkar, Cate, Vos, Westers & Croiset 2013:57). The trainees, or nurses, who attended a course (pain management) intending to gain knowledge or skills which are interesting ("training efficiency") or useful to them ("training relevance"), tended to display higher levels of transfer of learning (Nafukho et al. (2017:349). Nurses or trainees who are provided with choices and inputs to attend a training program, such as pain management, may indicate a high transfer of learning ability (Schneider 2014:88; Saks & Haccoun 2010:297; Gegenfurtner, Könings, Kosmajac & Gebhardt 2016:293). This is typical of the participation of nurses in designing pain management training programs or if they are autonomous in deciding whether to attend. Nurses or trainees motivated to learn about pain management may develop "motivation to transfer" in their practice areas.

Motivation to transfer

Motivation to transfer learning is defined as the trainees' desire to apply and connect back to their everyday work routine their newly acquired competencies from a particular training program (Reinhold, Gegenfurtner & Lewalter 2018:3; Govaerts, Knydt & Dochy 2017:8; Banerjee, Gupta & Bates 2017:609; Kirwan 2016:93; Snoek & Volman 2014:95; Wen and Lin 2014:117). Donovan and Darcy (2011:123) illustrated that motivation to transfer learning is integrated and influenced by factors related to trainees' personality, age, situational variables such as pre-training situations, self-efficacy, valence, and job or career variables that affect what motivates them to learn new skills and knowledge. Nafukho et al. (2017:333), Madagamage et al. (2014:13), Wen and Lin (2014:86), and Donovan and Darcy (2011:124) indicate that many beliefs influence motivation to transfer learning, but in particular, pointing out three main beliefs:

- A belief that a particular act will precede a particular outcome and suggesting that behaviour is followed by desirable outcomes ("Vroom expectancy theory").
- The logic is that trainees want to be treated fairly, and if they feel that by transfer of learning, they can gain equity through pay, promotion, or any other kind of rewards, then they will transfer learning ("equity theory").
- A person's intention is a function of attitude, subjective norms, and behaviour control, which influence willingness ("theory of planned behaviour").

Significant predictors that enhance motivation to transfer are trainees' self-efficacy factor, positive learning transfer climate, job motivation, satisfaction, organisational commitment, and motivation to learn (Nazli & Khairudin 2018:126; Mohanty, Dash & Dash 2017:234; Singh 2017:3; Khan, Mufti & Nazir 2015:200). The nurses' or trainees' characteristics and motivation factors should be taken into consideration by the trainer, such as the nurse educator or facilitator designing and planning the pain management training program in order to enhance transfer of learning (Daffron et al. 2015:60; Salvendy 2012:517).

2.4.2 Training design

Training design "refers to the relationship between the activities engaged in on the training programme and the actual job requirements of the trainee" (Donovan & Darcy 2011:123). It is important that the educator or facilitator carefully prepare the training program by assessing the needs of the trainees that will address knowledge and skills issues (Czaja & Sharit 2016:1; Munkvold & Kolãs 2015:132; Wan 2013:13). The trainer or facilitator, like the nurse educator or facilitator, should plan the learning objectives

and outcomes, materials to use, content and specific instructional strategies that make it easier for trainees to apply the new knowledge and skills (Kirwan 2016:16; Czaja & Sharit 2016:1; Munkvold & Kolãs 2015:132; Wan 2013:13; Kaiser et al. 2013:34). The nurses trainees should have opportunities to practice skills and give feedback related to the programme (Schneider 2014:68; Cartaxo & Simões 2014:0460; Kim & Callahan 2013:188). A transfer-oriented educational design is highly recommended as it consists of various elements and sub-elements that influence the trainees (as nurses) to introduce change in work performance (Donovan and Darcy 2011:123; Leberman & McDonald 2016:22; Wenzel and Cordery 2014:7). Factors such as principles of learning, sequencing of training content and training content are considered as part of the transfer-oriented educational design (Donavan & Darcy 2011:123; Wenzel & Cordery 2014:7).

2.4.2.1 Principles of learning

Principles of learning are referred to as laws of learning applicable to the learning process that provide additional insight into what makes the trainees learn and the teaching effective during the training program (Lopez 2010:116). Adult education principles of learning should be applied using a variety of methods of delivery and involvement with the trainee (Daffron et al. 2015:61). Specific methods of training should be used based on how nurse participants learn best and methods to increase the incidence of learning transfer should also be used. As discussed in earlier dimensions of the Systemic Model of Transfer of Learning, it is obvious that adult learning is influenced by an individual's motivation to learn. The principles of learning have great value for enriching human life in all its spheres and may enhance the transfer of learning. Four basic learning principles to be incorporated during the design and delivery of training programs that influence the transfer of learning are identical elements, general principles, stimulus variability, and conditions of practice (Schneider 2014:69). The consideration of the four principles of learning during training design may maximise the transfer of learning as follows:

• Identical elements learning principles are those stimuli and response elements of learning that are similar to the actual training and transfer settings and which

increase the retention of both motor actions and behaviours (Nafukho et al. 2017:331; Schneider 2014:69). Examples during pain management training may include using simulation training to imitate real-life practice such as case studies; games or role-plays; preparing the training surroundings similar to the workplace; practising the tasks the same as at the workplace; and preparing the equipment is the same as used in the workplace (Nafukho et al. 2017:331; Schneider 2014:69; Werner & DeSimone 2012: 80; Daffron & North 2011:52).

- Teaching through general principles is when the design of the course content includes theory and general rules that underlie fundamental elements of a concept to be learned rather than a specific skill set for each situation (Schneider 2014:69; Werner & DeSimone 2012:81). Examples during pain management training may be: videos, case studies, tutorials, discussion groups, demonstrations, presentations, handouts, simulation, action learning, role-play in the educational environment, the level of learner centeredness, active engagement with learning material that is applicable to the practice and job requirements as well as the atmosphere in which learning occurs (Nazli et al. 2015:57; Kirwan 2016:9; Botma & MacKenzie 2016:105; Schneider 2014:69).
- The stimulus variability principle is described as the learning principle whereby the nurse educator or clinical facilitator employs and engages the trainees in several different practice situations to strengthen their understanding, as suggested by Nafukho et al. (2017:331); Schneider (2014:69); Werner and DeSimone (2012:81); and Daffron and North (2011:52).
- Various conditions of practice include decisions by nurse educators or clinical facilitators to use strategies of learning that promote appropriate response at an appropriate time (Nafukho et al. 2017:331; Schneider 2014:69). Examples during pain management training can be to divide training into sections either as a whole or part training; providing feedback, and over-learning which is the process to provide trainees with the opportunity to practice beyond the mastery of a task (Nafukho et al. 2017:331; Schneider 2014:69).

It is vital that the nurse educator or clinical facilitator consider the principles of learning for effective teaching during a training program by determining the sequence of the training content.

2.4.2.2 Sequencing of the training content

Sequencing of the training content is the decision to group and to determine the learning content that is logical and arranged rationally according to learning objectives that will enhance the transfer of learning (Czaja & Sharit 2016:134; Carliner 2015:96). There are nine known approaches to sequence training content, including chronological, topical, whole-to-part, part-to-whole, known-to-unknown, unknown-to known, step-by-step, part-to-part-to-part and general-to-specific sequencing (Alvord 2010:82). For instance, in a pain management training program the first topic or assignment may be designed by (a) using fading prompts or cues, (b) using step by step prompts, and (c) breaking or segmenting down the activities into manageable portions (Allen & Cowdery 2015:321).

The two basic patterns of sequencing strategies that can be applied to enhance the transfer of learning are topical and spiral sequencing (Czaja &Sharit 2016:134). With topical sequencing, a topic or task is taught to some target level of depth of understanding or competence before the subsequent task is initiated (Czaja and Sharit 2016:134). In a pain management training program, the first topic may focus on fundamental concepts of pain before presenting a topic on how to assess pain (Bement & Sluka 2015:146; Bement, St. Marie, Nordstrom, Christensen, Mongoven, Koebner, Fishman & Sluka 2014:452; Fishman, Young, Arwood, Chou, Herr, Murinson, Watt-Watson, Carr, Gordon, Stevens, Bakerjian, Ballantyne, Courtenay, Koebner, Djukic, Koebner, Mongoven, Prasad, Singh, Sluka, St. Marie & Strassels 2013:975).

Within spiral sequencing, the trainees have to master several interrelated tasks by passing through all the tasks until the target level of learning is achieved (Czaja and Sharit 2016:134). In a pain management training program, the fundamental concepts

of pain may be interrelated with how to assess pain (Bement & Sluka 2015:146; Bement et al. 2014:452; Fishman et al. 2013:975).

The literature indicates that educators or clinical facilitators should design the training program by maintaining the following sequence:

- Conduct needs assessment for underdeveloped skills, insufficient knowledge or inappropriate worker attitudes, such as a gap between nurses' knowledge regarding pain assessment and the application of their knowledge to patient care (Joly, Coranado, Bickford, Leider, Alford, Mickeever & Harper 2018:4; Kizza, Muliira, Kohi, Nabirye 2016:23; Ufashingabire, Nsereko, Njunwa & Brysiewicz 2016:21; Aziato & Adejumo 2013:3; Van der Akker, Branch, Gustafson, Nieveen & Plomp 2012:63).
- Develop training objectives, for instance, to enhance nurses' levels of knowledge towards pain assessment and apply what they learn in pain management training programmes (Kizza 2016:26; Ufashingabine et al. 2016:25; Aziato & Adejumo 2013:5; Van der Akker et al. 2012:63).
- Design a curriculum that covers pain management core competencies such as the multidimensional nature of pain, pain assessment and measurement, management of pain, and context of pain management (Herr et al. 2015:317; Doorenbos, Gordon, Tauben, Palisoc, Drangsholt, Lindhorst, Danielson, Spector, Ballweg, Vorvick & Loeser 2013:1534; Van der Akker et al. 2012:63).
- 4) Select instructional strategies and training methods that will enhance the transfer of learning. In this state, in pain management, these may comprise immersive simulation games, questions, demonstrations, role-playing, problem-solving, job aids, positive feedback, focus groups, post-training coaching and action planning, selfpaced study, a recorded portion of the lecture (Kaiser et al. 2013:13; Sankaranarayanan & Sindhu 2012:89; Van der Akker et al. 2012:63; Daffron & North 2011:56).
- 5) *Design an evaluation approach*, for example, effectiveness of pain management training similar to (i) formative evaluation by asking nurses' feedback attending the pain management training to evaluate the education they received, and (ii)

summative evaluation to assess the value the pain management education has to the transfer of learning (Dolin, Deierlein & Evans 2018:54; Griffin 2014:14; Hatzipanagos & Rochon 2012:76; Van der Akker et al. 2012:63).

- 6) Conduct the training, in the case of pain management education to nurses it will be related to core competencies in pain assessment and pain management (Czarnecki & Turner 2018:597; Herr et al. 2015:319; Doorenbos et al. 2013:1536; Fishman et al. 2013:975; Van der Akker et al. 2012:63).
- 7) Measure the results of the effectiveness and efficiency of training outcomes. In this study, it will be pain management content on how much the nurses learned and can apply from what they learned in their nursing areas (Orey 2015:116; Schneider 2014:25; Van der Akker et al. 2012:63).

When designing the training content that will positively influence the transfer of learning, the pain management educator and clinical facilitator should integrate the learning objectives and outcomes into nurses' clinical practice (Beek, Dawson & Whelan 2017:7; Botma & MacKenzie 2016:106; Kim & Callahan 2013:188).

2.4.2.3 Training content

The training content may be defined as the information delivered during continuous education to develop trainees' skills, knowledge, and attitudes related to specific competencies (Janet 2016:36; Manichander 2016:79). Training content in this context is pain management learning that needs to be delivered to nurses. Content such as the fundamental concepts of pain, pain assessment and measurement, management of pain, and the role of the nurses in pain management must be addressed (Herr et al. 2015:319; Doorenbos et al. 2013:1536; Fishman et al. 2013:975). To improve the transfer of learning, the training content should be congruent with work requirements and be recognised as relevant by the trainees (Donovan & Darcy 2011:130; Cartaxo & Simõnes 2014:0460; Schneider 2014:86). If the training content is relevant and addresses the needs of the trainees, the transfer of learning is maximised (Ma et al. 2018:4; Schneider 2014:87; Daffron et al. 2015:55). To enhance the transfer of learning, the educator or clinical facilitator must have the knowledge and skills to deliver the

training content (Donovan & Darcy 2011:130; Freitas & Silva 2017:311). The educator or clinical facilitator who presents training on pain management must know about pain management, have professional experience in pain management, have knowledge of learning styles and be aware of the teaching approaches that enhance the transfer of learning of pain management competencies to nurses (Sankaranarayanan & Sindhu 2012:88; Donovan & Darcy 2011:130).

The content learned in the training program can further be successfully transferred if the work environment supports the trainees.

2.4.3 Training transfer climate

The training transfer climate is described as the characteristics of the relationship between the work environment and the motivation to transfer learning (Donavan & Darcy 2011:123; Srimannarayana 2016:263; Landy & Conte 2016:277; Qureshi, Bhutto & Memon 2015:106). A favourable or positive training transfer climate is considered multidimensional, as stated by Donavan and Darcy (2011:130), Singh (2017:4), Frasard and Prasuhn (2016:130), Srimannarayana (2016:264), and Daffron and North (2011:4). Some of those dimensions are the following:

- There must be supervisory and managerial support for new skills, such as reinforcing learning on the job.
- Peers or co-workers must support the development of new skills.
- The perceived utility or value outcomes of training must include that it will provide opportunities to apply skills learned in training.
- It must have a career or job utility, referring to planning or exploration by trainees to achieve their learning goals.
- There must be training accountability.
- There must be an opportunity to practice.
- Trainees must be able to perform or use learning at the workplace, an opportunity to use new skills and knowledge.
- There must be intrinsic and extrinsic rewards for using new skills and knowledge, a transfer design, and positive personnel outcomes.

• There should be no adverse personal outcomes but organisational commitment through behaviour, systems and processes.

According to various authors (Ma et al. 2018:4; Govaerts et al. 2017:5; Beek et al. 2017:7; Singh 2017:4; Qureshi, Bhutto & Tunio 2017:361; Frasard & Prasuhn 2016:144; Botma & MacKenzie 2016:105; Bansal & Thakur 2013:54; Lee at al 2014:2846) supervisors or managers should support the trainees who attend training programs with the following:

- Resources and opportunities to apply the learning.
- Encourage trainees to attend trainings, seminars, conferences, discussion forums or presentations; participate in goal-setting activities; champion and reinforce new ideas to influence the transfer of learning.
- Provision of adequate support to trainees to apply learned theory to the specific tasks at the workplace.
- Practice a good communication style about training programs and create times and places to learn at the workplace.

The peers or co-workers should also support the trainees after their training by setting goals, adopting coaching-mentoring relationships, and providing positive feedback and encouragement concerning what was learned in the training (Donovan & Darcy 2011:129; Ma et al. 2018:4; Qureshi et al. 2017:361; Frasard & Prasuhn 2016:129).

The hospital's nursing leadership, for example, the associate executive director of nursing, may contribute by valuing the training programs and putting in place policies that support the empowerment of the competencies of their employees (Donovan and Darcy 2011:129; Snoek & Volman 2014:92; Daffron & North 2011:108; Lee et al. 2014:2846; Bansal & Thakur 2013:53). In two Saudi Arabian teaching hospitals, the associate executive directors of nursing, as well as the director of nursing education, support the pain management training programs. This is evidenced in the Administrative Policy and Procedures (APP) 1430-07 (2017:5), which mandates regular education for nurses.

The transfer climate of training can be enhanced through a workplace with a conducive environment that will promote the transfer of what was learned in the training program.

2.4.4. Workplace environment characteristics

Workplace environment characteristics refer to workplaces that allow trainees to practice new skills acquired during training (Donovan & Darcy 2011:123). The nurses who believe that their work environment is supportive and conducive to learning will demonstrate high performance in learning efforts (Donavan & Darcy 2011:124; Nazli et al. 2015:56). For instance, encouragement and motivation of nurses by supervisors, peers, and co-workers and the organisation to apply what was learned during pain management training will create a supportive and positive work environment (Suleiman et al. 2016:23; Yasin, Nur, Ridzwan, Bekri, Arif, Mahazir & Ashikin 2014:184).

In essence, the Systemic Model of the transfer of learning is significant for this study as nurses who attend training programs for pain management are largely responsible for their learning and applying their knowledge in different nursing units. Other factors to be considered to enhance trainees' ability to apply what they learned after training are the different learning transfer levels.

2.5 LEVELS OF TRANSFER OF LEARNING

Different levels of the transfer of learning are described in the literature. Haskell's taxonomy identified six levels of the transfer of learning, namely, (1) non-specific transfer, (2) application transfer, (3) context transfer, (4) near transfer, (5) far transfer, and (6) displacement or creative transfer (Mishra 2016: 314; Sangster 2016:21); Schneider 2014:54; Kaiser et al. 2013:7). The six levels of the transfer of learning are briefly described below:

2.5.1. Level 1: Non-specific or general transfer

Non-specific transfer implies that all learning is transferred because the mind recalls prior learning, which is applied in a daily work environment (Leberman & MacDonald 2016:29; Mishra 2016:314; Sangster 2016:21; Cox 2013:139). For example, nurses

know what pain assessment is because it is part of what is done during their daily practice.

2.5.2 Level 2: Application transfer

Application transfer refers to the application of knowledge that was learned during a specific learning situation that is similar to a specific work environment (Leberman & MacDonald 2016:29; Sangster 2016:21). An example is when nurses understand what pain assessment is and can assess the pain of patients experiencing pain.

2.5.3 Level 3: Context transfer

Context transfer is described as the application of what one has learned under slightly different situations where the learning situation and the work context are not related to the learning content to be transferred (Sangster 2016:21; Leberman & MacDonald 2016:29; Schneider 2014:55). This is, for example, when nurses who learned about pain assessment and management in clinical settings can use their experience to assess and manage pain in home settings.

2.5.4 Level 4: Near or specific transfer

Near or specific transfer occurs when the knowledge domains and settings from learning situations are highly similar to the work environment (Leberman & MacDonald 2016:29; Sangster 2016:22; Snowman & McCown 2015:373; Schneider 2014:55; Kaiser et al. 2013:28; Cox 2013:139). Nurses who learned skills about assessing pain in a training program are examples of near or specific transfer for someone who has already learned and practised a similar skill to assess pain at the bedside.

2.5.5 Level 5: Far or general transfer

Far or general transfer occurs when applying knowledge and skills to a work context that is entirely different from the original setting of learning situations (Leberman & MacDonald 2016:29; Sangster 2016:22; Snowman & McCown 2015:373; Kaiser et al. 2013:7; Cox 2013:139). For example, nurses who learned about electrical conduction

during physical science studies will be able to connect their understanding of the process of nociception of pain when learning about the physiology of pain.

2.5.6 Level 6: Displacement or creative transfer

Displacement or creative transfer results in the discovery of new concepts because of the interaction of the newly perceived similarity between the new work situation and the old learning situation (Leberman & MacDonald 2016:29; Sangster 2016:22; Schneider 2014:55). In this study context, it can be when a nurse conducted a research study on pain management and created and published new pain models to contribute to the knowledge of pain management.

In addition to the classification of the *levels* of the transfer of learning according to Haskell's taxonomy, it may also be identified by *types* of the transfer of learning.

2.6 TYPES OF TRANSFER OF LEARNING

Depending on the learning situation faced by the learners, the following types of the transfer of learning can be identified: (1) positive transfer, (2) negative transfer, (3) zero transfer, (4) horizontal or lateral transfer, (5) vertical transfer, (6) high road transfer and (7) low road transfer (Hale & Stanney 2015:783; Kaiser et al. 2013:7; Ahmad, Mustafa, Gessler & Spöttle 2013:199). These types of transfer of learning are briefly explained:

2.6.1 Positive transfer

Positive transfer is the desired type of transfer of learning that helps trainees to function effectively as learning from a previous context complements a current context (Babu & Gayathri 2018:102; Blumberg 2014:190; Kaiser et al. 2013:7; Yasin et al. 2014:180; Ahmad, Razzaq, Mustafa, Ahmad, Gessler & Spöttle 2013:201). For example, nurses may learn how to document pain assessment in a new electronic medical record after being accustomed to a different electronic medical record. This means the learning occurred in one context but improves performance in some other context. Contrary to positive transfer, the transfer of learning may be negative.

2.6.2 Negative transfer

Negative transfer is when the prior learning or previous experiences interfere with or impact negatively on learning and transfer in the work situation due to either contrary experiences, expectations, or connotations between the two (Babu & Gayathri 2018:102; Snowman & McCown 2014:372; Kaiser et al. 2013:7). An example can be that of nurses who worked in hospital settings where pain management was not taught or prioritised this may hinder their competencies of pain management when working in hospital settings that prioritise pain management training and practices.

2.6.3 Zero transfer

Zero transfer denotes the fact that the learning or training in one situation does not influence the work situation. This means no transfer effect from the learning situation to the work situation (Babu & Gayathri 2018:102; Snowman & McCown 2015:372), such as when the nurse attended pain management training but still did not demonstrate change or improvement in their usual practices.

2.6.4 Horizontal transfer

Horizontal or lateral transfer resembles learning new skills or performing new tasks in work situations that differ significantly from the original learning situation (Landy & Conte 2016:278; Hale & Stanney 2015:783; Burns & Dobson 2012:346). Nurses might learn from their peers how to assess and manage pain during preceptorship, mentoring, or exchange of information at the workplace. The opposite criterion of horizontal transfer is vertical transfer.

2.6.5 Vertical transfer

Vertical transfer refers to applying previously acquired knowledge and skills from a learning situation in the same product to identical work situations and new knowledge or skills, building hierarchically (Landy & Conte 2016:278; Hale & Stanney 2015:783). In this context, nurses who learned about pain management during their prelicensure

or undergraduate training may be able to manage patients experiencing pain as registered nurses.

2.6.6 High-road transfer

High-road or mindful transfer refers to applying knowledge and skills over longer periods to work situations that look somewhat different from the learning situations (Hale & Stanney 2015:783; Snowman, McCown & Biehler 2012: 357). High-road transfer involves a trainee being able to reflectively think (mindful abstraction) about what was learned and then deliberately abstract from the original context to connect it to the work contexts (Snowman & McCown 2015: 374; Kaiser et al. 2013:7). Nurses who might have learned about electrical conduction during physical science will be able to understand the process of nociception of pain when thinking about the physiology of pain. To achieve high-road transfer, the trainee should be assisted by encouraging cognitive understanding, purposeful and conscious analysis, mindfulness, and application of strategies across disciplines (Hale & Stanney 2015:783; Snowman & McCown 2015: 374; Kaiser et al. 2013:7).

2.6.7 Low-road transfer

Low-road or reflexive transfer refers to a situation in which a previously learned skill or idea is automatically retrieved from memory and applied in a similar current task (Hale & Stanney 2015:783; Snowman & McCown 2015: 374; Snowman et al. 2012: 357; Kaiser et al. 2013:7). In this case, it may apply to nurses working at the bedside who become educators or clinical facilitators of pain management to support other nurses.

Training and learning play a significant role in developing nursing knowledge, skills, and behaviours to apply during care in clinical areas (Almasi, Bavani & Mohammadpour 2018:242), for example, enhancing nurses' "pain management" competencies.

2.7 PAIN MANAGEMENT

Pain management can be defined as the process of providing medical or nursing care that prevents and provides treatment or interventions that alleviate or reduce types of pain disorders for the longest possible time (DeWit & Kumagai 2013:134; Fritz & Chaitow 2011:10; Sun 2011:34; Leifer 2011:160; Fishman, Ballantyne & Rathmell 2010:678; Pillitteri 2010:1125).

An interprofessional consensus summit engaged healthcare experts to categorise core competencies for pain management. Four domains were identified, namely: (1) the multidimensional nature of pain, (2) pain assessment and measurement, (3) management of pain, and (4) the context of pain management (Bement & Sluka 2015:146; Bement et al. 2014:452; Fishman et al. 2013:972). The core competencies serve as foundations for developing comprehensive pain management curricula across all health professionals designed to advance care that effectively responds to pain (Bement & Sluka 2015:146; Bement et al. 2014:452; Fishman et al. 2014:452; Fishman et al. 2014:452; Fishman et al. 2014:452; Fishman et al. 2013:972).

Each of the four domains of the pain management core competencies for nurses are briefly described below:

- 1) The multidimensional nature of pain domain focuses on the concepts and complexity of pain (Fishman et al. 2013:976). In this regard, it is the responsibility of nurses to know the pain and understand its mechanisms (Pickering, Zwakhalen & Kaasalainen 2018:104; Royal College of Nursing (RCN) 2018:29).
- 2) The pain assessment and measurement domain relates to how pain is observed, assessed, and measured (Fishman et al. 2013:976). For example, it is the role of nurses to have knowledge and skills to assess patients' pain accurately and reassess it by using valid and reliable pain scales (Lim, Han, Lee, Lee, Kim, Yun, Park, Park, Choe, Ryoo, Lee, Cho, Zang & Choi 2015:228; Washington & Leaver 2015:226; Nicol, Bavin, Cronin, Rawlings-Anderson, Cole & Hunter 2012:66).
- 3) The Management of pain domain focuses on relieving pain (Bement & Sluka 2015:146). This also illustrates that nurses have a significant role in applying their

knowledge and skills regarding collaborative approaches and effective pain management options for each patient to ensure effective pain management (Ojong, Ojong-Alasia & Nlumanze 2014:316; Van Griensven, Strong & Unruh 2014:4).

• 4) The Context of pain management domain focuses on the role of the clinicians of different professions to apply the competencies of the first three domains and in the context of varied patient populations, settings, and care teams (Bement et al. 2014:452). It is, therefore, the role of the nurse to be competent in knowing the concept of pain, skills in pain assessment, and effective pain management skills depending on the context of the patient's care to be able to care for the patient effectively (Van Griensven et al. 2014:4; Ojong et al. 2014:319).

Within this study context, the core competencies for pain management by nurses will focus on their essential responsibilities in providing comfort to all patients by systematically implementing the steps of the nursing process that comprise (1) Assessment, (2) Diagnosis, (3) Planning, (4) Implementation, (5) Evaluations (acronym, ADPIE), and (6) Documentation (Krau & Overstreet 2017:433; Baird 2016:154; Ackley & Ladwig 2014; Van Griensven et al. 2014:4; Elsevier 2014:162; DeWit & Kumagai 2013:128; DeWit & Williams 2013:52; Registered Nurses' Association of Ontario (RNAO) 2013:7; Thomas 2011:17). The first step in the nursing process is portrayed in terms of "assessment" of pain.

2.7.1 Assessment

Assessment is defined as the initial phase of operation in the nursing process, whereby information available about the individual patient, family, or community is gathered using subjective or objective data (Black 2017:219; Stanhope & Lancaster 2012:424). Another definition of pain assessment is that it is a comprehensive, interactive, systematic, and collaborative process of gathering and synthesising data (symptoms of pain) by involving the patient, the family members, nurses, physicians, and other health professionals (RNAO 2013:7; Dougherty & Lister 2015:10; Song, Eaton, Gordon, Hoyle & Doorenbos 2015:461; DeWit & O'Neil 2014:52; Keogh 2012:6, Arnstein 2010:64). The data obtained form the basis for the identification and application of correct individualised pain management. If a patient's primary complaint is pain, the assessment will guide the nurse in assessing the pain (Marx, Walls & Hockberger 2013:36).

Comprehensive pain assessment can be seen as a thorough interview with a patient informed by a detailed investigation of the patient's pain history, a physical examination or diagnostic test, biopsychosocial aspects of the patient's pain experiences, the prudent use of a standardized pain scale and psychometric inventories; clarification of the goals of interventions; and discussing options with the patient and family (Paice 2015:2; Ebert & Kems 2011:47, Vadivelu, Urman & Hines 2011:206); Alexander, Corrigan, Gorski, Hankins & Perucca 2010:376).

Within this context a comprehensive pain assessment that comprises the following steps will be regarded as acceptable: (1) screening of the pain; (2) obtaining the patient's self-report of pain; (3) proxy-reported pain; (4) using systematic pain assessment guide tools for pain history taking; (5) using pain rating assessment tools appropriate for diverse patient populations; (6) conducting a specific physical assessment of a patient with pain (7) observing for physiological indicators of pain; and (8) observing for behavioural indicators of pain (Acton 2013:36; Pasero & McCaffery 2011:123; Ebert & Kems 2011:47; Arnstein 2010:64). Pain screening therefore forms the initial step of pain assessment.

2.7.1.1 Pain screening

Pain screening is defined as the first step that precedes a comprehensive and systematic pain assessment whereby patients who can communicate are asked about the presence or absence of pain (Brown, Edwards & Seaton 2017:55; Moore 2013:102; Registered Nurses' Association of Ontario (RNAO) 2013:19; Ebert & Kems 2011:47; Arnstein 2010:61). The Joint Commission International (JCI), the Saudi Central Board for Accreditation of Healthcare Institutions (CBAHI), and Registered Nurses' Association of Ontario (RNAO) recommend that patients' pain should be screened during a healthcare visit, on admission to hospital, after a change in medical status, each time
vital signs are measured, and before, during and after an invasive procedure (JCI 2017:91; CBAHI 2015:66; RNAO 2013:20).

If pain is present during screening by the nurses, then they will proceed with a comprehensive and systemic assessment of the patient's pain experience using an appropriate pain scale (JCI 2017:99; CBAHI 2015:67; RNAO 2013:20). For patients who may verbalise their pain, it is essential to obtain a "self-report of pain".

2.7.1.2 Obtaining the patient's self-report of pain

Patients' self-report or subjectivity of pain is defined as what the patients say or verbally indicate about their pain (Van Griensven, Strong & Unruh 2014:26; RNAO 2013:25; Pasero & McCaffery 2011:21). It is considered the only truly direct and most accurate, reliable measure of pain. The subjectivity of pain includes the sensory component and the psychological, cultural, and emotional elements of the experience (Czarnecki & Turner 2018:218). It is deemed to be the "golden standard" for the existence and intensity of pain (Rababa 2017:1; Drake & Williams 2017:10; Hadjistavropoulos, Herr, Prkachin, Craig, Gibson, Lukas & Smith 2014:1220; American Association of Critical-Care Nurses (AACN) Practice Alert 2014:81; Daher, Versloot and Costa 2014:1; Howard and Liossi 2014:1; Nair & Neil 2013:3). The nurse must obtain a self-report of pain from the patient, believe and act on the patient's report of pain, and use pain scales to obtain pain ratings from patients to avoid underestimating their pain (Pasero and McCaffery 2011:15; D'Arcy 2011:42). According to Ebert and Kems (2011:48) and Pasero and McCaffery (2011:21) no objective measures of pain exist, as the sensation of pain is an entirely subjective phenomenon, meaning the nurse must accept a patient's pain history and self-report of pain as the primary source of pain assessment.

The comprehensive key elements of pain assessment that nurses should obtain from a patient's self-report of pain are a description of the quality of pain, the intensity and or severity of the pain, the location of the pain, the duration of the pain, the aggravating and alleviating factors of the pain; the patient's pain goals, thus telling the nurse the acceptable pain level, previous interventions used, the effects or impact of pain on daily activities, and on the physical, mental, and social wellbeing of the patient (Joint Commission International [JCI] 2017:99; Kisser-Larson 2017:90; Saudi Central Board for Accreditation of Healthcare Institutions [CBAHI] 2015:69; RNAO 2013:20; Ebert & Kems 2011:48; D'Arcy 2011:42; Pasero & McCaffery 2011:50; Vadivelu et al. 2011:371; Arnstein 2010:65). The key elements needed from a patient to conduct a comprehensive pain assessment (Czarnecki & Turner 2018: 218; Burns 2015:320; RNAO 2013:21; D'Arcy 2011:59) are briefly discussed below:

- Description of pain: This involves the quality of the pain, referring to the specific physical sensations associated with pain (Czarnecki & Turner 2018:218). The patient must be asked to describe their pain in their own words, such as whether it is aching, stabbing, causing tenderness, tiring, numbing, feeling dull, throbbing, gnawing, burning, deep, shooting, nag-ging, squeezing, radiating, lancinating, tingling, jolting (as in electrical), causing numbness; creating pressure, cramping, distending, or stretching (Paice 2015:4; Rose 2013:144; D'Arcy 2011:43). The patient's language literacy level should be assessed as patients use different terms according to their ability and if necessary, an interpreter must be involved during the assessment (interview) of the patient (Perry, Potter & Ostendorf 2015:318).
- The intensity of pain: The intensity can be assessed by rating the pain using a valid pain self-reporting scale with numbers between 0 and 10 to represent the intensity of the pain, 0 is no pain, and 10 is the worst pain imaginable as the standard metric for pain intensity (Lim et al. 2015:230; Paice 2015:4; Arnstein 2011:65; D'Arcy 2011:43). According to Pasero and McCaffery (2011:86), patients and families should be given an education on how to use a pain rating scale as follows: to show the pain rating scale to the patient and family and explain its primary purpose; to explain the parts of the pain rating scale; to verify that the patient understands the broad concept of pain; to ask the patient to

practice using the pain rating scale with the present pain; to set goals for comfort and functioning or recovery by asking the patients what pain rating would be acceptable.

- Location of the pain: The patient must be asked to point to the specific painful area on the body (D'Arcy 2011: 42; Arnstein 2010:66). The site of the patient's pain or discomfort should be inspected for discolouration or swelling and palpated for a change of temperature, an altered sensation, painful area, areas that trigger pain, and range of motion (Perry et al. 2015:319; D'Arcy 2011:42, Arnstein 2010:66). The assessor, or in this context the nurse, should ask about the radiation of pain from the pointed area to another area of the body (Paz & West 2013:20; D'Arcy 2011:42; Pasero & McCaffery 2011:50).
- The duration of pain: The patient must be asked about the history of the pain, thus about the onset and pattern of pain over time; how it has changed since then; how long the pain lasts; and to find out whether the pain is continuous, intermittent, pulsatile, chronic or has a breakthrough nature (Nicol et al. 2012:67; D'Arcy 2011:43; Arnstein 2011:66; Vadivelu 2011:58).
- Aggravating and alleviating factors of pain: The patient must be asked to describe the factors that make the pain better or worse (Paice 2015:7; Paz & West 2013:20; D'Arcy 2011:44). This will help to refine the nursing diagnosis, care plan, and the interventions needed.
- The patient's pain goal: The nurse must ask the patient what level of pain rating would be acceptable or satisfactory to maintain quality of life, thus identifying a pain rating that will exist without interfering with the activities that the patient needs or wishes to perform (Cheatle & Fine 2017:112; Ebert & Kems 2011:49; Pasero & McCaffery 2010:86). For example, the nurse may set the comfort-function goal by working with a patient and identify a pain rating of 3 for ambulation as more acceptable.
- *Previous pain interventions used:* The nurse must ask the patient or family to describe previous pain experiences and methods used to manage

pain effectively and record the patient's response to the interventions (Perry, Hockenberry, Lowdermilk, Wilson 2014:377; Ebert & Kems 2011:48; Haugen & Galura 2011:53).

 Physical, psychological, social, spiritual, or existential domains of pain and suffering: The nurse must assess the impact of pain to maintain a patient's quality of life. For example, a patient with chronic or cancer pain may be assessed for (i) the presence of physical pain and its impact on functional activities, such as impaired mobility; (ii) the psychological aspects of pain, such as fear, sadness, depression, and anxiety; and (iv) the social impact of pain such as poor relationships with the family or community (Bruera, Higginson, Von Gunten, Morita 2016:92; Yarbro, Wujcik & Gobel 2014:76; Werth, Jr 2013:31).

Despite the self-report of pain that is deemed to be the golden standard of pain assessment, some patients' conditions create challenges for nurses to obtain self-reported pain, and therefore, "proxy-reported pain" becomes an alternative.

2.7.1.3 Proxy-reported pain

Proxy-reported pain is an alternative to self-report that occurs when caregivers, parents, or family members are involved in reporting pain on behalf of patients who are unable to report their pain (Stone & Walker 2017:72; Hla, Hegarty, Russell, Drake-Brockman & Ramgolam 2014:1127, D'Arcy 2013:19; Moore 2013:132).

Patients who cannot self-report their pain result from communication barriers related to different degrees of cognitive impairment, intellectual disabilities, being comatose, being on the verge of dying, being sedated, being a neonate, being a child intimidated or frightened by the hospital environment preventing them from communicating clearly about their pain as they are not familiar with typical pain behaviours or changes (Weissman-Fogel, Roth, Natan-Raav & Lotan 2015:915; RNAO 2013:26; Pasero & McCaffery 2011:123). Asking the family members to contribute to the pain assessment

implies enquiring about typical behaviour, changes in behaviour, and cognition (Yarbro et al. 2014:74), which then becomes important

2.7.1.4 Systematic pain assessment guide tools for pain history taking

Systematic pain assessment guide tools are condensed mnemonic tools used to guide the nurse to interview the patient when taking pain history during pain assessment (D'Arcy 2011:59). It is recommended to use pain assessment guide tools to help guide nurses through the interview portion of the pain assessment (Urman and Vadivelu 2011:4-1; Arnstein 2010:65; Pasero & McCaffery 2011:50). It is important to use and follow either one of the following frequently used pain assessment guide tools:

- "WILDA" for *words* to describe pain *intensity*, *location*, *duration* and *aggravating* or *alleviating* factors (Rose 2013:144; Alexander et al. 2010:376; Arnstein 2010:71).
- "PQRST" stand for *provocation* and *palliation* of symptoms, *quality* of pain, *re-gion* and *radiation* of pain, *severity* of pain and *timing* (Stefan & Rodriquez-Galindo 2014:144; Urman and Vadivelu 2011:4-1; Chila 2010:260).
- "OPQRSTUV", an adapted pain assessment acronym for the onset of pain, provoking/palliating, quality, region/radiation of pain, severity of pain, timing/treatment, understanding/impact on you and values (Cooper & Gosnell 2015:71; RNAO 2013:21).
- "OLD CARTS" is another mnemonic for pain assessment similar to "OPQRST" for onset of pain, location of pain, characteristics of pain, aggravating factors of pain, relieving factors, timing and severity (Ballweg, Brown, Vetrosky & Ritsema 2018:396; Burns 2015:320; Quinlan-Colwell & D'Arcy 2012:43).
- "QUESTT", an adapted "QUEST", is specifically for the assessment of children's pain that stands for *questioning* the child/caregiver, *using* pain rating tools, *evaluating* behaviour, *sensitise* parents, *taking* the cause of pain into account, and

"taking the cause of pain into account" (Gladston, Emmanuel & Prasad 2016:55; Dolan & Holt 2013:271).

- "COLDSPA" abbreviates the character, onset, location, duration, severity, pattern, and associated factors (Weber & Kelly 2014:20; Weber, Kelly & Sprengel 2014:8).
- The "McCaffery Initial Pain Assessment Tool" guide is an interview form that guides the patient or health care professionals in completing an initial assessment of the verbal patient (Forbes & Watt 2016:233; Weber & Kelly 2014:153). It includes diagrams of the human body to help patients locate the pain they experience as well as questions to prompt the patient to describe the location, intensity, constant quality, onset, causes, effects, contributing factors of the pain, other comments, and plan (Rebeiro, Jack, Scully & Wilson 2013:298; Pasero & McCaffery 2011:50).

These pain assessment guide tools are used with pain measurement tools to rate the pain while assessing patients experiencing pain.

2.7.1.5 The use of different pain rating assessment tools

Pain measurement tools are pain scales used to rate patients' pain levels during pain assessment (Roberts, Custalow & Thomsen 2017:21; Pasero & McCaffery 2010:49). Most of the pain rating scales used are commonly categorised into four groups, namely: (1) Unidimensional pain scales; (2) Bidimensional pain scales; (3) Multidimensional pain scales and (4) Impact of pain tools (Teo, Johnson, Pandanaboyana & Windsor 2016:2).

 Unidimensional pain tools: These are pain scales used to assess one aspect of pain, such as the severity of pain (Teo, Johnson, Pandanaboyana & Windsor 2016:2; D'Arcy 2013:24; Weigelt 2012:1624; Urman & Vadivelu 2011:4-3), for example, the visual analogue scale, the numeric rating scale, the verbal descriptor scale, and the Wong-Baker pain scale.

- 2) Bidimensional pain tools: They are pain scales that combine two aspects of pain, such as pain intensity (how much pain hurts) and pain-related interference with activities (how much a patient suffers) (Teo et al. 2016:2). For example, the Shoulder Pain and Disability Index (SPADI) scale was used to assess shoulder pain and dysfunction (Vrouva, Batistaki, Koutsioumpa, Kostopoulos, Stamoulis & Kostopanagiotou 2016:315).
- 3) Multidimensional pain tools: These are pain scales used to assess and provide more complex information about a patient's pain, that is, more than two aspects of pain including affective, sensory, quality and character of pain, satisfaction with pain control, and impact of pain on various indicators (Teo et al. 2016:2; Bruckenthal & Quinlan-Colwell 2012:62; Weigelt 2012:1624; Urman & Vadivelu 2011:4-4; Waldman 2011:359). Examples are the McGill Pain Questionnaire (MPQ) and the Brief Pain Inventory (BPI) (Gregory & Richardson 2014:1; Davies & D'Arcy 2013:24).
- Impact of pain tools: These are pain tools used to evaluate the quality of life (QOL), the level of disability, and the effects of pain on mental and emotional states (Teo et al. 2016:2). An example is the Brief Impact Questionnaire (BPIQ) (D'Arcy 2013:32).

Pain assessment using scales must be used according to patients' age groups, such as neonates, preverbal toddlers, children, and adult patients (Chotolli & Luize 2015:111; Vael & Whitted 2014:302). Clinical conditions must also be taken into account, such as unconscious patients, cognitively impaired, intubated patients, sedated patients, non-verbal patients, elders with advanced dementia and critically ill patients (Upadhyay, Cameron, Murphy & Battistella 2014:367; Weber & Kelly 2013:149). The pain rating assessment tools are based on either self-reporting or observing behaviour (Nair & Neil 2013:3). In this study, the researcher decided to describe pain scales that were frequently reviewed as valid, reliable, and feasible for clinical utility. The rationale

for choosing these pain tools is that there is no consensus regarding which scales are preferable and mandatory (Andersen, Munsters, Vederhus & Gradin 2018:2).

The pain scales used according to the age group and ability of the patients to rate their pain (see Table 2.2) are (1) pain rating assessment tools for patients who can self-report their pain, (2) pain rating assessment tools for patients who cannot self-report their pain, and lastly (3) pain rating assessment tools used for elderly patients with dementia or cognitive impairment. It is important for nurses to use appropriate pain scales consistently and specific to the age group and ability of the patients to rate their pain, as this makes it possible to compare the pain ratings and to minimise confusion for both patients and nurses (Pasero& McCaffery 2011:68).

| A. Pain rating assessment tools for patients who can self-report the pain | |
|--|---|
| Population of patients | The pain rating tools used |
| 1.1 Children and | Wong-Baker FACES |
| adolescents able to | Numerical Rating Scale (NRS) |
| verbalise their pain. | Poker Chip Tool (Pieces of Hurt Tool) |
| | Varni-Thompson Paediatric Pain Questionnaires (PPQ) |
| 1.2 Adults able to | Revised FACES Pain Rating (FPS-R) |
| verbalise their pain. | Numerical Rating Scale (NRS) |
| | Verbal Rating Scale (VRS) |
| | Visual Analogue Scale (VAS) |
| | Verbal Descriptor Scale (VDS) |
| | Pain Questionnaires and dairies: |
| | The Brief Pain Inventory (BPI) |
| | Long-Form and Short-Form of McGill Pain Questionnaire (LF-MPQ & SF-MPQ) |
| B. Pain rating assessment tools for patients who cannot self-report the pain | |
| 2.1 Premature and | Premature Infant Pain Profile (PIPP) |
| neonates. | Neonatal Infant Pain Scale (NIPS) |
| | Crying, Required oxygen, Increased vital signs, Expression, Sleeplessness |
| | (CRIES) scale |
| | Neonatal Pain, Agitation and Sedation Scale (N-PASS) |

| 2.2 Children and | Face, Leg, Activity, Crying, Consolability (FLACC) | |
|---|---|--|
| adolescents unable to | The COMFORT-Behaviour pain scale | |
| verbalise their pain. | | |
| 2.3 Adults unable | Critical Care Pain Observational Tool (CPOT) | |
| to verbalise their pain. | Behavioural Pain Scale (BPS) | |
| C. Pain rating assessment tools used for elderly patients with dementia or cognitive impairment | | |
| 3.1 Elderly pa- | Abbey Pain Scale | |
| tients with dementia or | Checklist of Nonverbal Pain Indicators (CNPI) | |
| cognitive impairment. | Doloplus-2 | |
| | Non-communicative Patient's Assessment Instrument (NOPPAIN) | |
| | Pain Assessment Checklist for Seniors with Limited Ability to Communicate | |
| | (PACSLAC) | |
| | Pain Assessment in Advanced Dementia Scale (PAINAD) | |

A: Pain rating assessment tools for patients who can self-report the pain

Self-report pain rating assessment tools are used to rate pain intensity, which is only one effective aspect of the patient's pain experience (Lim et al. 2015:226; Nicol et al. 2012:68). The self-report pain scales reviewed in this study are those that are common with high clinical utility, and it is evident that they are highly validated, reliable with feasibility for the criteria of pain assessment. In children and adolescent patients who can self-report their pain, it is recommended to assess pain with the following pain scales:

1) Wong-Baker FACES pain rating scale (FPS) is originally for use with children but also validated for use in adults and cognitively impaired populations (Tsui & Suresh 2015:82; Marmo & D'Arcy 2013:37; Pasero & McCaffery 2011:59). The scale consists of six animated smiling happy/sad faces or tears presented with the 0 to 10 metric that includes:(1) face 0 means not hurt; (2) face 1 hurts a little bit; (3) face 2 hurts a little more; (4) face 3 hurts even more; (5) face 4 hurts a whole lot; and (6) face 5 hurts worst to be explained to the patient (Forbes & Weight and the state of the scale of

Watt 2016:233; Marmo & D'Arcy 2013:37). The patient is asked to choose the face that best describes own pain at that moment (Tsui & Suresh 2015:82; Pasero & McCaffery 2011:66; D'Arcy 2011:78).

- 2) Numerical Rating Scale (NRS) is a self-report pain assessment ordinal scale used for cognitively intact adults and children aged seven or more who may self-report pain and have the mathematical skills required for pain rating using the numeric rating scale (Paice 2015:7; Tsui & Suresh 2015:82; Howard & Lioossi 2014:2). The scale consists of an 11-point set of numbers from 0 to 10 where 0 means "no pain", and 10 means the "worst possible pain" presented graphically as a horizontal or vertical line (Fell et al. 2018:138; Czarnecki & Turner 2017:223; Ismail, Ghafar, Shamsuddin, Roslan, Kaharuddin& Muhamad 2015:289; D'Arcy 2011:57). Patients must point or mark a number on the scale to indicate their pain levels (Lim et al. 2015:226; Pasero & McCaffery 2011:56; D'Arcy 2011:57).
- 3) The Poker Chip Tool (Pieces of Hurt Tool) is a self-report scale used for children of the age group 4-7 years and consists of four red chips that represent pain intensity ("pieces of hurt") without precise quantification (De Freitas, De Castro, Castro & Heineck 2014:816). The child is asked to choose a number of chips that indicate how much pain is experienced from four red poker chips ("pieces of hurt") placed on a flat surface in a horizontal line, with one chip indicating a little hurt and all four chips indicating the most hurt a child could ever have (Thirion, O'Riordan & Stormorken 2015:2; Pillitteri 2014:1120; Nair & Neil 2013:4).
- 4) For children between 4 and 9 years, their pain can also be assessed using multidimensional questionnaires (RNAO 2013:83). An example of a widely used pain questionnaire is the Varni-Thompson Paediatric Pain Questionnaire (PPQ), which is a patient self-report and parent perspective pain tool used for comprehensive chronic pain assessment in children aged 5 to18 years (Hochberg, Silman, Smolen, Weinblatt & Weisman 2015:878; McClain &

Suresh 2011:67; Bruera & Portenoy 2010:137). For example, the nurse may assess a child with juvenile rheumatoid arthritis by completing the questionnaire and asking about pain intensity, location by marking the body diagram on the form, sensory dimension of the pain experience, and affective impact on the daily lives (McClain & Suresh 2011:67; Hochberg et al. 2015:878). It includes a 100 mm horizontal line colour-coded visual analogue rating scale anchored by happy and sad faces for present and worst pain (Hochberg, Silman, Smolen, Weinblatt & Weisman 2015:878; RNAO 2013:83; McClain & Suresh 2011:66; Bruera & Portenoy 2010:137).

Adult patients are commonly assessed using scales such as:

- 1) The revised FACES Pain Rating (FPS-R) scale was adapted from the Wong-Baker FACES scale (FPS) for self-rating and is focused more towards older adults who may not have well-developed verbal skills to explain how their pain symptoms make them feel (Czarnecki & Turner 2017:223; Paice 2015:7). The scale has 0 to 10 metrics to represent a visual description of the six oval cartoon adult-like faces, with the absence of smiles and tears arranged with increasing looks of distress from "no pain" to "worst imaginable pain" (Paice 2015:7; Marmo & D'Arcy 2013:37). The patient does not require reading or writing ability and is asked to point the face that best describes own pain (RNAO 2013:81; Pasero & McCaffery 2011:68; D'Arcy 2011:78). The appropriate number that represents the face chosen by the patient will be recorded (Tsui & Suresh 2015:82; Wright 2014:88).
- 2) The Verbal Rating Scale (VRS) is a categorical ordinal linear tool that comprises four or five-point levels of a list of gradually ascending or descending verbal adjectives used by the patient to describe different levels of pain with an assigned number for ease recording (Fell, Lunnen & Rauk 2018:138; Czarnecki & Turner 2017:223; Kliger, Stahl, Haddad, Suzan, Adler & Eisenberg 2015:539). During pain assessment, the patients are asked to select or mark the adjective that best describes their level of pain intensity; the commonly used

descriptors are "no pain," "mild pain," "moderate pain"; and "severe pain" (Fell, Lunnen & Rauk 2018:138; Czarnecki & Turner 2017:223; Kliger, Stahl, Haddad, Suzan, Adler & Eisenberg 2015:539; Karcioglu 2017:e2; Leslie, Johnson, Thomas & Goodwin 2011:476).

- 3) The Verbal Descriptor Scale (VDS) consists of adjectives anchored with a number that reflects the description of pain ranked in order of severity such as no pain, mild pain, moderate pain, severe pain, very severe/extreme pain (Czarnecki & Turner 2017:222; Forbes & Watt 2016:233; D'Arcy 2011:56). Pain is assessed using this scale for the patient that understands the meaning of the words and who is asked to select the word that best describes the present pain experienced (Pasero & McCaffery 2011:94; Arnstein 2010:63). The assessor records the number anchored to the words described by the patient (Czarnecki & Turner 2017:222).
- 4) The Visual Analogue Scale (VAS) is a self-report scale comprising scores from 0–10, or a 10 cm or 100 mm horizontal or vertical line, with verbal descriptive pain anchors whose extremes are labelled as "no pain" at the left endpoint and "worst imaginable/possible pain" or a comparable term at the right endpoint or far-right; is recommended for children of age seven to adults (Tsui & Suresh 2015:82; Fell et al. 2018:138; Paice 2015:6; Nair & Neil 2013:4). The patients are asked to make a mark of their pain intensity on the line between the two endpoints of the pain scale (Pasero & McCaffery 2011:55; Kliger et al. 2015:540; Wariaghli, Allali, Berrada, Idrissi, Hmamouch, Abouqal, Hajjaj-Hassouni 2013:104).

Multidimensional pain scales such as pain diaries and questionnaires are used for assessing pain in speciality adult patient populations suffering from chronic and cancer pain (Davies & D'Arcy 2013:24; Bruckenthal& Quinlan-Colwell2012:62; D'Arcy 2011:63). The following pain diaries and questionnaires scales are mostly used in a clinical setting to assess pain for adult patients:

- 1) The Brief Pain Inventory (BPI) is a 16-item self-administered validated questionnaire that may be completed by the assessor used to assess pain in cancer patients (Forbes & Watt 2016:233; Karcioglu 2017:e2; Marmo & D'Arcy 2013:39). The BPI scale consists of questions used to gather information about pain severity using a numeric rating scale of 0 to 10 to rate the pain at its worst and least for the past 24 hours, impact or degree of interference of pain on daily function, location of pain, pain medications and amount of relief in the past 24 hours or the past week from current pain management (RNAO 2013:82; Bruckenthal& Quinlan-Colwell2012:62; Kibel 2012:507; D'Arcy 2011:66; Pasero & McCaffery 2011:52).
- 2) The Long Form of McGill Pain Questionnaire (LF-MPQ) is a self-report measure developed by Melzack 1975, that comprises 78 pain descriptors used to assess pain in the four major domains of pain: (1) sensory in terms of time, space, pressure, heat, and brightness, (2) affective in terms of tension, fear, and autonomic properties, (3) evaluative and (4) miscellaneous sensory (Boyle, Boerresen & Jang 2015:797; Ferreira, De Andrade & Teixeira 2013:211). To assess pain, the pain rating index (PRI) is extracted using the sum of the rank values of the words chosen by the patient to obtain a score separately for the sensory, affective, evaluative, and miscellaneous words, and provide a sum total score ranging from 0 to 78 (Benuto & Leany 2015:268; Marmo & D' Arcy 2013:39; Turk & Melzack 2011:47; Urman and Vadivelu 2011:40-2).
- 3) The Short Form of McGill Pain Questionnaire (SF-MPQ) is a shorter version of the Long Form of McGill Pain Questionnaire, and its main components consist of 15 descriptors of pain divided into two subscales: (1) Sensory subscale consists of 11 descriptors, and (2) Affective subscale consists of 4 descriptors (Benuto & Leany 2015:268). Each descriptor is ranked on an intensity scale of 0 = none, 1 = mild, 2 = moderate, and 3 = severe (Waldman 2011:197). The tool also contains a visual analogue scale (VAS) and a Present Pain Intensity scale (PPI) (D'Arcy 2011:65; Waldman 2011:197).

B: Pain rating assessment tools for patients who cannot self-report the pain

In premature babies and neonates, some of the best-known pain scales used to assess pain are:

- The Premature Infant Pain Profile (PIPP) scale is best known and used to measure behavioural expression of pain in ventilated premature neonates of gestational age between 28 and 38 weeks (Andersen et al. 2018:2; Yaripoor, Khalili, Joobakhsh, Talebiyanpour & Almasi 2016:42). Pain is assessed using seven behavioural indicators of pain that includes: three behavioural changes (facial actions: brow bulge, eye squeeze, and nasolabial furrow); two physiological (heart rate and oxygen) indicators; and two contextual (gestational age and behavioural state) (Desai, Nanavati, Jasani & Kabra 2017:288; Emergency Nurses Association [ENA] 2013:122). Each indicator is scored on a 4-point scale consisting of scoring points of 0, 1, 2, or 3. Scores are summed across the seven indicators with a maximum of 21 points, thus a total of 21/21 (Marko & Dickerson 2017:40; Yaripoor, Khalili, Joonbakhsh, Talebiyanpour & Almasi 2016:42).
- 2) The Neonatal Infant Pain Scale (NIPS) measures pain behavioural cues in full-term and preterm infants and consists of six behavioural indicators that include: (1) facial expressions; (2) cry; (3) breathing patterns; (4) arms; (5) legs; (6) and state of arousal by taking pain measurement before, during and after a painful procedure and scored in one-minute intervals (Andersen et al. 2018:2; Yaripoor et al. 2016:42; James, Nelson & Ashwill 2013:70). Each item is scored on a 2-point scale, scoring points of 0 and 1, except for the cry indicator, which is scored on a 3-point scale scoring of 0, 1, or 2. The summed scores of all six indicators are a maximum of ten points; thus, they are 10/10 (Andersen et al. 2018:2; Marko & Dickerson 2017:28; RNAO 2013:89; Gardener, Carter, Enzman-Hines & Hernandez 2011:240).

3) The "Crying, Required oxygen, Increased vital signs, Expression, Sleeplessness" (CRIES) scale. CRIES is observer-rated, which means an assessor observes to recognise, note, rate, and report the behaviours as described by Kulshrestha and Bajwa (2021:102) pain assessment tool used in neonates experiencing pain, specifically pain responses in the postoperative period (Behrens & Beinert 2014:51; Leifer 2013:286). The tool is used to observe and rate three behavioural and two physiological categories based on the CRIES acronym that includes: (1) crying due to pain characterised by a high-pitched cry; (2) requires oxygen for saturation less than 95%; (3) increased vital signs' values of heart rate and mean blood pressure; (4) expression of grimace on the face due to pain; (5) sleeplessness related to the hour preceding the recorded score (Malakian, Dehdashtiyan, Aramesh, Aletayeb & Ghazanfari 2017:6881; Andersen et al. 2018:2; Marko and Dickerson 2017:26; Leifer 2013:286). Each item is scored on a 3-point scale scoring 0, 1, or 2, thus: (1) crying is zero points for no crying and crying that is not high pitched, one point if is high-pitched cry consolable, or two points if it is high-pitched cry inconsolable; (2) requires oxygen for saturation less than 95% is zero point if when no change in oxygenation is noted and no supplemental oxygen required, one point if less or equal to 30% of oxygen is required, or two points if more than 30% of oxygen is required; (3) increased vital signs' values of heart rate and mean blood pressure is zero point if the values are less than or equal to baseline, one point if the values increase by less than or equal to 20%, or two points if values increase more than 20%; (4) expression of grimace on the face is zero point if no grimace present, one point if grimace alone and two points if a child has been awake constantly (Marko and Dickerson 2017:24; Pillitteri 2014:1120; Fishman, Ballantyne & Rathmell 2010:682). The point values are added from all assessment categories ranging the scores from 0 to 10; thus, the maximum total score is 10/10 (Malakian et al. 2017:6881; Andersen et al. 2018:2; Marko and Dickerson 2017:26; Pillitteri 2014:1120).

4) The Neonatal Pain, Agitation and Sedation Scale (N-PASS) was developed to measure pain in term and preterm infants who are experiencing prolonged postoperative pain and or pain during mechanical ventilation (Desai et al. 2017:292; Hall & Anand 2014:897; Leifer 2013:286). N-PASS consists of five items to assess: (1) crying/irritability; (2) behavioural state; (3) facial expression; (4) extremities/tone, vital signs such as heart rate, respiratory rate, blood pressure; (5) and oxygen saturation graded 0, 1 or 2 for pain/agitation, and 0, -1, or -2 for sedation (Hummel 2017:175; Marko & Dickerson 2017:30; Oakes 2011:28). A score greater than +3 indicates pain, and a score of more than -3 indicates sedation (Desai et al. 2017:289; Hummel 2017:175).

Children and adolescents who cannot verbalise their pain are assessed with the following behavioural pain measurement scales:

- 5) The FLACC (Face, Leg, Activity, Crying, Consolability) Scale is a behavioural scale used to assess pain in patients with challenges of self-reporting that include children in the intensive care unit, preverbal children, children with cognitive impairment, pediatric postoperative pain, and critically ill intubated adults (Varndell, Fry & Elliot 2016:11; Vael & Whitted 2014:302; Coté, Lerman & Anderson 2013:913; Nair & Neil 2013:4). FLACC scale incorporates five behavioural categories to assess pain that comprises: (1) face; (2) legs; (3) activity; (4) crying and (5) consolability (Hummel 2017:177; Marko & Dickerson 2017:44). Each item is scored on 3-point scale scoring points of 0, 1, 2 scores which are summed across the five items with a total of 10/10 score (Hummel 2017:177; Marko & Dickerson 2017:44; McGuire, Kaiser, Haisfield-Wolfe & Iyamu 2016:6).
- 6) The COMFORT-Behaviour pain scale (COMFORT-B) is an observer-rated measure derived from the original COMFORT scale and is used with critically ill children ages 0 to 18 years for assessment of pain and sedation in intubated and ventilated children in intensive care environments (Hockenberry & Wilson 2018:145; Andersen et al. 2018:2; Hughes, Breatnach, van Dijk, Magner & Paul

2014:A534; Carter & Simons 2014:100; Nair & Neil 2013:4). The COMFORT-B scale consists of six behavioural items and items of ventilation to be indicative of pain and distress that includes: (1) alertness; (2) calmness or agitation; (3) respiratory response/distress (in ventilated patients); (4) crying (in a non-ventilated child); (5) physical movement; (6) muscle tone; (7) and facial tension; (Carter & Simons 2014:100; Hughes et al. 2014:3; Coté et al. 2013:913; Edwards & Coyne 2013:94). Each dimension is scored on a Likert scale from 1 to 5, and the scores are added to yield a measure of sedation and pain (Coté et al. 2013:913; Nair & Neil 2013:4). The sum of scores varies from 6 to 30 to indicate a patient in pain and the maximum total is 35/35 (Edwards & Coyne 2013:94).

In adult patients who cannot verbalise their pain, the following pain scale may be used to rate the pain:

 Critical Care Pain Observational Tool (CPOT) is used to assess and manage pain in adult non-conscious patients, critically ill ventilated or non-ventilated in intensive care units (Damico, Cazzaniga, Murano, Nattino & Molin 2016:259; McGuire et al. 2016:5; Storsveen & Hall-Lord 2016:3; Asadi-Noghabi, Gholizadeh, Zolfaghan, Mehran & Sohrabi 2015:277; Stites 2013:73; Pasero & McCaffery 2011:144). The scale measures a patient's pain level by observing four behavioural categories, namely: (1) facial expressions, (2) body movements, (3) muscle tension, and (4) compliance with the ventilator for ventilated patients and or vocalisation for those who are not ventilated or extubated patients (D'Arcy 2013:53; D'Arcy 2011:76; Pasero and McCaffery 2011:145). Each domain has three behaviours which are scored on a 3-point scale of 0, 1 or 2 for a possible total score ranging from 0 (no pain) to 8 (maximum pain), that is an 8/8 total score (Damico et al. 2016:259; Asadi-Noghabi et al. 2015:277; D'Arcy 2013:53; Rose, Haslam, Dale, Knechtel & McGillion 2013:247; RNAO 2013:94). 2) The Behavioural Pain Scale (BPS) is the scale used for assessing pain in uncommunicative, critically ill, sedated, and intubated intensive care unit patients (Damico et al. 2016:257; McGuire et al. 2016:3; Paice 2015:7). The scale is used to evaluate three behavioural domains that include: (1) facial expressions; (2) upper limb movements; (3) and compliance with mechanical ventilation as behavioural indicators of pain (Suzuki 2017:2; D'Arcy 2013: 55; RNAO 2013:93; Stites 2013:72). Each domain of the scale is scored on a point 4 scale scoring of 1, 2, 3, or 4; scores are summed across three domains with a maximum total score of 12 points, that is 12/12 (D'Arcy 2011:75; Pasero and McCaffery 2011:146).

C: Pain rating assessment tool used for elderly patients with dementia or cognitive impairment

Elderly patients with cognitive impairment or dementia are characterised by memory loss, personality changes, and loss of other functions such as judgement, abstract thinking, and language skills, resulting in barriers to pain assessment due to imprecise verbalisation of their pain (Lichtner, Dowding, Esterhuizen, Closs, Long, Corbett & Briggs 2014:2; Miller 2012:112). Observational pain assessment scales are used to assess pain in elderly patients with cognitive impairment or dementia (Malara, De Biase, Bettarini, Ceravolo, Di Cello, Garo, Praino, Settembrini, Sgrð, Spadea & Rispoli 2016:1224; Hadjistavropoulos et al. 2014:1217; Takai, Yamamoto-Mitani, Chiba & Kato 2014:440; Takai, Yamamoto-Mitani, Ko & Heilemann 2014:237).

The best-known observational pain assessment scales used to assess pain are (1) Abbey pain scale, (2) the Checklist of Nonverbal Pain Indicators (CNPI), (3) Doloplus-2, (4) Pain Assessment Checklist for Seniors with Limited Ability to Communicate (PACSLAC), (5) the Non-communicative Patient's Assessment Instrument (NOP-PAIN); and (6) Pain Assessment in Advanced Dementia Scale (PAINAD)(Rizzo 2018:636; Corbett, Achterberg, Husebo, Lobbezoo, De Vet, Kunz, Strand, Constantinou, Tudose, Kappesser, De Waal & Latenbacher 2014:6; Hadjistavropoulos et al.

2014:1219; Lichtner et al. 2014:5; Yarbro, Wujcik & Gobel 2014:74; Tuck & Melzack 2011:269). Firstly, the Abbey pain scale and its use to assess pain is briefly described.

1) The *Abbey Pain Scale (ABBEY)* contains six items that measure acute, chronic and acute-on-chronic pain intensity in people with late-stage dementia (Rizzo 2018:636; Paice 2015:12; Pickering & Gibson 2015:106). The six behavioural items used to rate pain intensity are (1) facial expression, (2) change in body language, (3) vocalisation, (4) behavioural change, (5) physiological change, and (6) physical change (Hadjistavropoulos et al. 2014:1220; Takai et al. 2014:239; Pasero & McCaffery 2011:141; Tuck & Melzack 2011:269). Each behavioural item is evaluated on a 4-point scale (absent = 0, mild = 1, moderate = 2, severe = 3, and the scores are added to provide a total assessment of pain intensity ranging from "no pain" (total score is 0–2), "mild pain" (total score 3–7), "moderate pain" (total score is 8–13) to "severe pain" (total score is more than 14) and the maximum total score is 24/24 (Atee, Hoti, Parsons & Hughes 2017:139; Tuck & Melzack 2011:269).

2) **The Checklist of Nonverbal Pain Indicators (CNPI)** measures pain behaviours in cognitively impaired older adults unable to validate the presence of pain or quantify pain by self-report methods (McGuire et al. 2016:4; D'Arcy 2013:51; Miller 2012:112). CNPI comprises an observation list of six pain behavioural items that include (1) nonverbal vocalisations, (2) facial grimacing or wincing, (3) bracing, (4) rubbing, (5) restlessness, and (6) vocal complaints (Lichtner et al. 2014:10; D'Arcy 2013:51; Pasero & McCaffery 2011:128). Each item is scored on a dichotomous scale (1 = present, 0 = not present), both at rest and on movement, scoring from 0 to 6 points for each situation and a combined total of 12 points. Thus, the highest score is 12/12 (Paice 2015:13; Lichtner et al. 2014:10; D'Arcy 2013:51).

3) The **Doloplus-2** is a comprehensive multidimensional pain assessment scale for assessing pain in nonverbal older adults with cognitive impairment (Rostad, Utne, Grov, Puts & Halvorsrud 2017:26). The tool is based on behaviour change with three subscale dimensions that consist of 10 items: (1) two psychomotor reactions (washing and dressing, mobility), (2) five somatic reactions, (protection of sore areas, sleep

pattern, somatic complaints, protective body postures adopted at rest, and expression); (3) and three psychosocial reactions (problem behaviour, social life, and communication) items (Rizzo 2018:636; Rostad et al. 2017:26; Hadjistavropoulos 2014:1220; Pasero & McCaffery 2011:141). Each item is scored on a 4-point scale scoring 0, 1, 2, or 3, and the total score of 10 items ranges from 0 to 30 (Rostad et al. 2017:2; Fry, Arendts & Chenoweth 2016:1282).

4) The **Non-communicative Patient's Pain Assessment Instrument (NOPPAIN)** is based on proxy reports of pain intensity in demented and cognitively impaired patients (Macintyre & Schug 2015:20; Yarbro et al. 2014:74; Pasero & McCaffery 2011:142). NOPPAIN is composed of four main sections for assessing pain behaviour in demented and non-communicative patients that are: (1) observed daily activities, (2) information about assessing the presence or absence of six pain behaviours, (i) facial expressions, (ii) pain-related words, (iii) rubbing, (iv) bracing, (v) pain noises, and (vi) restlessness), (3) pain location, and (4) pain behaviours observed and pain intensity scored using a pain thermometer (Hadjistavropoulos 2014:1220; Pasero & McCaffery 2011:142).

5) The *Pain Assessment Checklist for Seniors with Limited Ability to Communicate (PACSLAC)* uses direct observation and familiar caregiver information to assess both common and uncommon pain behaviours (Pace, Treloar & Scott 2011:92; Pasero & McCaffery 2011:130). PACSLAC involves observation of 60 items that include behaviour during movement, eating and sleeping as well as mood and changes in social interactions and are divided into four subscale items that are: (1) facial expressions, (2) activity or body movements, (3) social or personality or mood and (4) physiological or eating or sleeping or vocal (Rizzo 2018:638; Hadjistavropoulos 2014:1221; Miller 2012:113). The nurse has to indicate with a checkmark which items on the PACSLAC occurred during the period of interest. The patient must be observed over time to enable observation of often subtle changes in behaviour, generate a total pain score and sum all subscale totals (Hadjistavropoulos 2014:1221; Miller 2012:113). 6) The *Pain Assessment in Advanced Dementia Scale (PAINAD)* is used to assess pain in patients with advanced moderate to severe dementia who cannot verbally communicate (D'Arcy 2013:63; Guccione, Avers & Wong 2011:402). PAINAID consist of 5 behavioural categorical items that include: (1) breathing; (2) negative vocalisation; (3) facial expression; (4) body language; (5) consolability with three response modalities (Bryant & Nix 2016:391; Fry et al. 2016:1282; Pasero & McCaffery 2011:128). Each of the five behavioural domains is scored from 0 to 2 with a range of total score of 0 to 10; the maximum score is 10/10, and the nurse has to observe the patient for 5 minutes (Fry et al. 2016:1282; Malara et al. 2016:1223; Varndell et al. 2016:11).

The interview about the patient's pain history should be accompanied by a general physical assessment followed by a specific physical assessment of the painful region or area (Waldman 2011:42; Urman & Vadivelu 2011:4-6; Arnstein 2010:70; Nagelhout & Plaus 2010:1243).

2.7.1.6 Specific physical assessment of a patient with pain

Yarbro et al. (2014:76) state that physical assessment includes a head-to-toe examination, and the assessor should observe the patient and identify overt signs of pain. Haanpää (2014:203), Yarbro et al. (2014:76), Benson, Raja, Fishman, Liu and Cohen (2011:22), Smith (2013:22), Vadivelu et al. (2011:62), Urman & Vadivelu (2911:4-6), and Arnstein (2010:70) add that a comprehensive pain assessment includes the specific physical assessment of the painful site that consists of the use of subsequent assessment techniques such as:

- 1. *Inspection* of the painful lesion, including skin abnormalities and colour changes, muscle changes, and presence of oedema.
- 2. *Gentle palpation* should follow inspection of the affected painful area to observe the changes in pain intensity due to noxious and non-noxious stimuli.
- 3. Gentle percussion and occasional auscultation of painful areas similar to a chest, abdomen, or vascular structures.

4. Range of motion assessment should be assessed by both passive and active range of movement to assess the severity of pain.

Patients subjectively report their pain during assessments, but if they have challenges with self-reporting, pain may be dependent on proxy reporting of pain.

2.7.1.7 Observations of physiological indicators of pain

The physiologic pain indicators are mostly used when self-reported pain is impossible and cannot be used as one part of complete pain assessment (Swearingen 2016:39, James et al. 2013:321, Arnstein 2010:69). If the patients are unable to report their pain using customary self-report assessment tools or if there is no proxy-reporting of pain, nurses should assume that pain is present by using the physiological indicators of pain (Stannard & Krenzischek 2018:34; Ignatavicius & Workman 2016:33).

During pain assessment, the nurse should also observe any changes such as increased heart rate, increased respiratory rate, elevated blood pressure, desaturation, apnoea, diaphoresis, skin pallor, flushing, cyanosis, constriction or dilatation of pupils, increased intracranial pressure, vomiting (Marko & Dickerson 2017:12; Gladston, Emmanuel & Prasad 2016:56; Howard & Liossi 2014:1, Chen & Chen 2015:105; Arbour & Gèlinas 2014:512; Pasero & McCaffery 2011:25, Arnstein 2010:69). Nurses are required to use physiological parameters with caution as very few researchers support vital signs as being relevant indicators of pain since other factors may elevate them or pain may be present without increased vital signs (Stannard & Krenzischek 2018:34; Pasero & McCaffery 2010:28). The rationale for observing physiological indicators of pain is related to the origin and duration of pain that may stimulate the sympathetic nervous system and influence the physiologic responses.

It is also essential that during pain assessment, nurses use behavioural tools to observe patients' behaviours that determine the presence of pain (Stannard & Krenzischek 2018:34; Ignatavicius & Workman 2016:33; Potter, Perry, Stockert & Hall 2013:972).

2.7.1.8 Observations of the behavioural indicators of pain

Behavioural indicators of pain are those behaviours used to determine the presence of pain in patients who cannot communicate their pain and to guide treatment using behavioural tools for pain assessment (Stannard & Krenzischek 2018:35; Fishman et al. 2010:785).

The populations of patients who may be unable to self-report are neonates, infants, preverbal children, adults with cognitive impairment such as nonverbal patients, people with advanced dementia, with intellectual disability, aphasic confused patients, disoriented patients, critically ill or unconscious persons, mechanically ventilated patients and persons who are terminally ill (Marko & Dickerson 2017:21; Baird 2016:154; RNAO 2013:25; Pasero & McCaffery 2011:123; and Arnstein 2010:69). It is extensively recommended that if a person is unable to self-report, one should rely on pain-related behavioural indicators or utilise behavioural pain scales by observing the following: (i) facial expressions, (ii) verbalisations and vocalisations, and (iii) body movements (Baird 2016:154; Hadjistavropoulos et al. 2014:1221; Arbour & Gelinas 2014:506; Howard & Liossi 2014:2; RNAO 2013:25; and Pasero & McCaffery 2011:123). These behavioural indicators of pain are discussed below:

- The facial expressions acknowledged for pain assessment include: a slight frown, a frightened face, grimacing, a wrinkled forehead, closed or tightened eyes, a distorted expression, rapid blinking, squinting, wincing, mouth opening, brow-raising, brow lowering, cheek raising, eyelids tightening, nose wrinkling, lip corner pulling, chin raising, and lip-puckering, and clenching the teeth (Marko & Dickerson 2017:23; Arbour & Gélinas 2014:516; Hadjistavropoulos et al. 2014:1220; Pasero & McCaffery 2011:125).
- The *vocalisations* for pain assessment include sighing, moaning, groaning, grunting, whining, whimpering, crying, screaming, yelling, being verbally abusive, asking for help, or making specific sounds such as gasping or noisily breathing

(Arbour & Gélinas 2014:512; Hadjistavropoulos et al. 2014:1220; Pasero & McCaffery 2011:125; Arnstein 2010:69).

The body and limb movements for pain assessment comprise rigidness, fidgeting, rubbing the hurt body part, increased pacing, floppiness, stiffness, spastic positioning, taking on a fetal position, tenseness, having a rigid body part, gesturing to or touching a part of the body that hurts, protecting a body part, favouring a body part, guarding a part of the body that hurts, moving the head down, holding arms down, clenching fists, shaking or trembling, refusing to move, avoiding certain body positions, curling up, walking with a gait, and undergoing mobility changes such limping (Arbour & Gélinas 2014:512; Hadjistavropoulos et al. 2014:1220; Pasero & McCaffery 2011:125; Arnstein 2010:69).

2.7.2 Nursing diagnoses

The nursing diagnosis is the description nurses give after completing an assessment whereby they analyze objective and subjective data about the patient and the drug (Black 2017:220; Stanhope & Lancaster 2012:424). According to DeWit and Kumagai (2013:131), the nursing diagnosis for pain is "pain related to" which gives a cause. In addition, Ackley, Ladwig and Makic (2017:3), Asadi-Noghabi, Gholizadeh, Zolfaghari, Mehran and Sohrabi (2015:276), De Wit and O'Neil (2014:50), and Arnstein (2010:89) declare that the nursing pain diagnosis is the step after:

- pain assessment by the nurse's clinical judgement or analysis of information gathered during pain assessment,
- clustering or organising the related information,
- identifying actual or potential problems and
- choosing the appropriate pain nursing strategies that will help implement effective pain management.

Acute and chronic pain are considered two primary nursing diagnoses that describe the pain (White, Duncan & Baumie 2013:121; Silvestri 2012:395). Acute pain may be

related to tissue injury secondary to surgical intervention as evidenced by restlessness, pallor, elevated systolic blood pressure, pulse, and respirations, dilated pupils, abdominal pain, procedural pain, acute postoperative pain, acute incisional pain, cramping, joint pain, perianal pain, and so forth (DeWit & Kumagai 2013:753; White et al. 2013:121; Beevi 2012:340; Haugen & Galura 2011:596; Arnstein 2010:90). Chronic pain may be related to chronic physical and psychological disability, biological, chemical, or physical psychological injuring agents as evidenced by hopelessness, self-care deficit, anxiety, fatigue, deficient knowledge, ineffective coping, ineffective role performance, avoidance of activities, activity intolerance, impaired physical mobility, a disturbed sleep pattern, a disturbed body image, powerlessness, sexual dysfunction, or impaired social interaction (DeWit & Kumagai 2013:753; White et al. 2013:121; Beevi 2012:340; Haugen & Galura 2011:596; Arnstein 2010:90).

Nursing diagnoses should include nurses' observations by clinically judging and involving the patients and family in assessing the impact of pain on how their pain interferes with their activities of daily living (ADLs) and quality of life (QOL). The nurses have to check the repercussions of pain on patients' level of functioning and QOL (Barros & Albuquerque 2014:107). The patients or family members should be asked about the impact of pain on their recovery as pain, mostly chronic pain, may impair the functioning regarding physical, psychological, and social abilities (Washington & Leaver 2015:227; Owen 2014:261; Poretsky & Liao 2013:762; Ebert & Kems 2011:49; Arnstein 2010:67; Bennett 2010:181). The presence of impaired functioning that affects patient's quality of life (QOL) and their activities of daily living (ADLs) to be assessed are:

The *physical impact* of pain is indicated by a limited ability to perform physical exercises and domestic chores, deterioration, muscle stiffness, difficulty in walking, poor participation in social activities and difficulty in getting up or sitting down, reduced grip strength, fatigue, and disability (Yiengprugsawan & Steptoe 2018:1054; Dueñas, Ojeda, Salazar, Mico & Failde 2016:459; Dziedzic & Hammond 2010:64; Tiran 2010:129).

- The *psychological impact* of pain entails reduced executive functioning (mental processes) due to the focus on their pain, comorbidity with depression, emotional stress, anxiety, agitation, loss of memory, low moods, feelings of frustration, reduced attention, sleep disturbances, and impaired cognitive ability (Gill 2018:88; Dueñas et al. 2016:462; Owen 2014:261; Dziedzic & Hammond 2010:64; Tiran 2010:129).
- The social impact of pain, in essence, has subsequent strained social relationships, financial burdens, frequent absenteeism, frequent medical leaves, loss of jobs, early retirement, reduced efficiency, productivity, and ability to carry out social roles as consequences (Barros & Albuquerque 2014:107; Bruckenthal & Quinlan-Colwell 2012:23; Dueñas et al. 2016:462; Bennet 2010:181; Dziedzic & Hammond 2010:64).

Following the assessment of pain, it is important to have a planning phase to act on the identified pain-related problems of the patient.

2.7.3 Planning

The planning phase is defined as the identification of goals and outcomes ("that are objective, realistic, and measurable") that are patient-oriented and provide time frames (Black 2017: 222; Stanhope & Lancaster 2012:424). Planning for pain management in nursing is defined as setting nursing care goals that indicate actions to relieve or control pain (De Wit & Kumagai 2013:131). Ackley et al. (2017:7), Cooney (2016:446), Collins (2015:4), and Arnstein (2010:92) declare that planning is to determine how to relieve or reduce pain by specifying the desired objectives. Patient-centred planning includes:

- a) Patients and care providers' realistic and measurable goals to achieve satisfactory pain relief.
- b) The expected outcomes for resolution of the problem to enhance comfort and function.

- c) Identifying related nursing interventions appropriate to meet comfort, psychosocial and spiritual needs that include drugs, nondrug and environmental interventions.
- d) The person responsible for implementing the interventions.
- e) Participation of the patient and family members through individually tailored education.
- f) Information on treatment options for pain interventions.

The planning that addresses the patient's pain management needs will be implemented to achieve the set goals that will promote comfort for the patient.

2.7.4 Implementation

The implementation phase occurs after a plan of care is developed and is defined as the phase in which nurses follow through on the decided plan of action specific to each patient to intervene in specific patient problems such as pain (Black 2017:225; Stanhope & Lancaster 2012:424). The implementation of nursing actions for pain management includes (1) nonpharmacological and (2) pharmacological interventions as determined by nurses, and a combination of both is often the best-recommended intervention, and (3) patient and family education about pain (Kisser-Larson 2017:90; Cooney 2016:446; Pachana & Laidlaw 2014:885; Svendsen & Bjørk 2014:e19; White et al. 2013:122; Vadivelu et al. 2011:374; Arnstein 2010:96). The non-pharmacological interventions are implemented mainly by nurses and patients as follows:

2.7.4.1 The non-pharmacological pain interventions

The non-pharmacological pain interventions may be defined as those therapies that do not involve administering medicines or any other active substance but can lessen the pain experience and are adjuncts of pharmacological interventions (Gélinas, Arbour, Michaud, Robar & Côté 2015:308; Toth & Moulin 2013:349; Vaajoki 2013:1). The non-pharmacological interventions cannot replace pharmacological treatment in cases of severe and chronic pain (Boxwell 2010:240). Nurses can suggest, implement

or teach nonpharmacological strategies to the patients or family to help decrease a patient's pain to enhance pain relief strategies categorised as (1) conventional medicine, (2) complementary or alternative medicine (CAM), and (3) integrative approaches (Kiser-Larson et al. 2017:91; Saha, Brüning, Barcelona, Büssing, Langhorst, Dobos, Lauche & Cramer 2016:1; Perry et al. 2015:319; D'Arcy 2011:171; Urman & Vadivelu 2011:42-4; Arnstein 2010:151). Non-pharmacological pain interventions improve the quality of life of care available to patients (Hall, Leach, Brosnan & Collins 2017:47).

Conventional medicine for pain relief is recommended as part of the non-pharmacological approach to managing pain, as discussed below.

1) Conventional medicine for pain relief

Conventional non-drug therapies for pain relief are methods classified as physical or psychosocial (cognitive or behavioural) modalities (Arnstein 2010:151). (1) Physical therapies mainly instructed by nurses to the patient for self-management activities include positioning for comfort, alignment, and healing; good health practices, such as diet, exercises, temperature regulation, having ample sleep; graded activity such as therapeutic exercises; heat and cold compressions, breathing techniques; pacing activities while using proper body mechanics; avoiding pain triggers and transcutaneous electrical nerve stimulation (TENS) (Rizzo 2018:640; Yilmaz, Karakaya, Baydur & Tekin 2018:7; Cooney 2016:446; Hökkä, Kaakinen & Pölkki 2014:3; RNAO 2013:84; Skeel & Khleif 2011:562). (2) Psychosocial therapies for pain relief focus on patients' cognitive behaviours. Cognitive-behavioural therapies (CBT) for pain management emphasise the role of thoughts, emotions, and behaviours in influencing pain (Kisser-Larson 2017:91; Cooney 2016:447; Jankovic & Peng 2015:245; Pachana & Laidlaw 2014:890). (i) Cognitive therapies teach patients to use specific strategies that act on their thoughts or feelings such as distraction activities, therapeutic humour, distracting conversations that interest the patient, reading, listening to music, relaxation techniques useful for patients with pain, guided imagery techniques, and active coping training (Rizzo 2018:640; Hökkä et al. 2014:3; Ebert & Kerns 2011:454; Vadivelu et al. 2011:533; Arnstein 2010:152). (ii) *Behavioural approaches* focus on applying changes in the way the patient lives with pain, and these may include operant therapies that reinforce adaptively and extinguish maladaptive pain behaviours, group therapies and work-hardening or functional restoration programs (Hökkä et al. 2014:3; Ebert & Kerns 2011:454; Vadivelu et al. 2011:533; Arnstein 2010:153). Complementary and alternative approaches may also be used as strategies to relieve pain.

2) Complementary and alternative (CAM) approaches to pain relief

The term "complementary" relates to medicine used together with mainstream medicine. At the same time, "alternative" refers to medicine that is used in place of conventional or "mainstream" medicine (Bowman, Davis, Ferguson & Taylor 2018:81). Complementary medicine and alternative medicine (CAM) comprehensively apply to several modalities or additional therapies such as natural products, mind, and body used in conjunction with recognised mainstream conventional or medical practices (Hall et al. 2017:51; Hall, Griffiths & McKenna 2015:137; D'Arcy 2011:175; Arnstein 2010:153). Czarnecki and Turner (2018:505) indicate that nurses have historically embraced complementary therapies, but recently, other healthcare professionals have recognised their value for pain management. According to Holliday-Welsh, Gessert and Renier (2018:109), Rizzo (2018:640), Felix, Ferreira, Cruz and Barbosa (2017:7), Hall et al. (2015:140), Vadivelu (2011:3), and Arnstein (2010:154), the current exponential growth of the adopted four domains of complementary pain interventions used are: (1) mind-body medicine (for example, meditation, imagery, prayer, art, music, hypnosis, cognitive-behavioural therapy); (2) biologically-based therapies (for example, herbal supplements, diets, and vitamins); (3) manipulative and body-based practices (for example, massage therapy, chiropractic, osteopathic medicine) and (4) energy therapies (e.g., Reiki, Magnetic therapy, and therapeutic touch).

Alternative pain interventions include homoeopathy, traditional Chinese medicine, acupuncture, acupressure, and naturopathy) (Hall et al. 2015:140; Vadivelu 2011:3; Arnstein 2010:154). Management of pain in neonates uses complementary interventions such as sucrose prior to a painful procedure, swaddling or facilitated tucking to provide containment, healing touch, Reiki, therapeutic touch, non-nutritive sucking with pacifiers, maternal skin-to-skin or kangaroo care, breastfeeding, reduction in stimuli, calming music, sound, movement, and acupressure (Buonocore & Bellieni 2017:125; Marko & Dickerson 2017:101; Mason 2015:250; RNAO 2013:84; Hall & Anand 2014:898; Gleason & Devaskar 2012:441; Miller 2012:117; Fanaroff 2011:6; Boxwell 2010:240).

3) Integrative approaches to pain relief

Integrative medicine is defined as the interventions that utilise a combination of complementary therapies with conventional therapies for comprehensive care that attends to biopsychosocial and spiritual needs, especially when pain is severe or chronic (Saha et al. 2016:1; Hökkä et al. 2014:3; Arnstein 2010:160). Pain relief provided in a comprehensive and integrative manner is an intervention that targets the patient's pain from the domain of the mind (mental), body (physical), spirit, and social interactions (Saha et al. 2016:2; Arnstein 2010:160). Physical (body) therapy is aimed at employing treatment approaches that include reducing pain triggers, massage, self-massage, applying heat or ice, body mechanics, positioning, nutritional supplements, exercises, pacing activities, yoga, sleep hygiene, and cutaneous stimulation (Perry et al. 2015:319; RNAO 2013:86; Arnstein 2010:161). The mental (mind) therapies are approaches that consist of relaxation techniques, imagery, self-hypnosis, pain diary, journal writing, distracting attention, reducing fear, anxiety, stress, sadness, and helplessness, and learning more about one's own pain or pain relief (Perry et al. 2015:319; RNAO 2013:86; Arnstein 2010:162). Spiritual therapy comprises prayer, meditation, self-reflection, meaningful rituals, humour, and energy works (Perry et al. 2015:319; RNAO 2013:86; Arnstein 2010:162). The interventions that target social interactions are functioning at the highest level possible, with improved communication, optimizing family relations, talking about one's stress with others, problem-solving, volunteering, and using support groups (RNAO 2013:86; Arnstein 2010:162).

2.7.4.2 The pharmacological pain interventions

Pharmacological pain interventions are the most common form of pain management relating to the use of pain medications that target the sensory dimension of pain (Vaajoki 2013:1; Boltz 2012:251). Nurses are encouraged to administer the prescribed pain medications in all patient populations and to know that pain medications are chosen based on what is known about each patient's condition, cognitive abilities, age, type of pain, location of pain, duration of pain, and severity of pain (Perry et al. 2015:328; Boltz 2012:251; Arnstein 2010:115). The World Health Organisation (WHO) recommends that nurses know and use pharmacological and non-pharmacological pain interventions and administer analgesics using three-stepped approaches (the analgesic step ladder) for effective pain relief. These are: Step 1 = non-opioids, step 2 = weakopioid analgesics and Step 3 = strong opioids, as recommended by the World Health Organisation; administer analgesics around the clock (ATC) considering half-lives of the analgesics and to anticipate and manage side effects of analgesics as stated by Davies (2012:94) D'Arcy (2011:93), and Arnstein (2010:116). The pharmacological pain interventions depend on three specific types of analgesics widely used for pain relief: (1) Nonopioid analgesics, (2) Opioid analgesics, and (3) Adjuvant analgesics (Perry et al. 2015:328; Pasero & McCaffery 2011:179; Arnstein 2010:122). The administration of these analgesics may be through different routes such as oral, intravenous such as patient-controlled analgesia, intramuscular, neuraxial such as epidural analgesia, and intrathecal, peripheral nerve blocks, and topical applications (Fishman 2012:835; Urman & Vadivelu 2011:21-2).

1) Nonopioids analgesics

Mild to moderate pain (1-3/10 to 4-6/10 pain score) for acute and chronic pain is managed effectively by administering non-opioid analgesics considered as first-line simple analgesics (Step I of the analgesic ladder), and these include (i) acetaminophen (Tylenol) and aspirin (Marko & Dickerson 2017:75; Dagenais & Haldeman 2012:144; Davies 2012:94; Pasero & McCaffery 2011:181; Arnstein 2010:122); (ii) non-steroidal anti-inflammatory drugs (NSAIDs) that inhibit cyclo-oxygenase enzyme (COX), i.e., both COX-1 and COX-2, for example, ibuprofen, diclofenac, indomethacin, naproxen, ketorolac, meloxicam and specific COX-2 inhibitors such as celecoxib (Marko & Dickerson 2017:75; Lewis, Bucher, Heitkemper, Harding, Kwong & Roberts 2017:11; Davies 2012:96; D'Arcy 2011:95; Pasero & McCaffery 2011:187; Arnstein 2010:123).

2) Opioid analgesics

Step II of the analgesic ladder medications comprises administering the weak opioids used for moderate to severe pain (4-6/10 to 6-10/10 pain score). Common examples are codeine, oxycodone, hydrocodone, tramadol, and tapentadol (Dagenias & Haldeman 2012:159; Fishman 2012:834; D'Arcy 2011:96; Pasero & McCaffery 2011; Arnstein 2010:117). For severe pain (7-10/10 pain score), patients are given strong opioid medications for pain relief (Step III of the analgesic ladder) and frequently used medications are morphine, meperidine, fentanyl, hydromorphone, and methadone (Marko & Dickerson 2017:83; Lewis et al. 2017:113; Stannard, Coupe & Pickering 2013:15; Dagenais & Haldeman 2012:159; Fishman 2012:835; D'Arcy 2011:96).

3) Adjuvant analgesics

Adjuvant analgesics or co-analgesics for additive pain relief are medications not classified as pain medications but which have pain-relieving effects as they can provide analgesic effects, opioid-sparing effects, and prevent neuropathic symptoms (D'Arcy 2011:139; Vadivelu et al. 2011:384; Arnstein 2010:147). The adjuncts broadly used are (i) antiepileptics (e.g., gabapentin, carbamazepine, phenobarbital and pregabalin); (ii) antidepressants (e.g., amitriptyline and duloxetine); (iii) skeletal muscle relaxants (cyclobenzaprine and tizanidine and orphenadrine); (iv) antispasmodic (e.g., baclofen); (v) benzodiazepine (e.g., lorazepam and midazolam); (vi) N-Methyl-D-Aspartate (NMDA) antagonists/blockers (e.g., ketamine); (vi) alpha-2 agonists, for example, clonidine and dexmedetomidine or precedex (vii) local anaesthetics, for example, topical analgesia like lidocaine patch 5%, Eutectic mixture of local anaesthetics (EMLA), local infiltration analgesias such as lidocaine, intermittent or continuous infusion of ropivacaine or bupivacaine by epidural or regional blocks, and intravenous lidocaine (Marko & Dickerson 2017:91; Lewis et al. 2017:117; Fishman 2012:837; D'Arcy 2011:140; Vadivelu 2011:385; Pasero & McCaffery 2011:704; Arnstein 2010:149).

2.7.4.3 Patient and family education about pain management

The responsibility of a nurse in pain management involves patient and family education that helps patients better understand and participate in their care and make wellinformed decisions (Joint Commission International [JCI] 2017:173; Kiser-Larson 2017:91; Saudi Central Board for Accreditation of Healthcare Institutions [CBAHI] 2015:94; van Griensven et al. 2014:4). Schaller (2018:14), Kiser-Larson (2017:91), Lewis et al. (2017:124), Pasero and McCaffery (2011:86), and Arnstein (2010:147) recommend that pain management nursing education to patients and family members be considered. This should consist of:

- Patient and family education should be given on how to use pain rating scales.
- The importance of reporting their pain must be stressed.
- Specific instructions must be given on non-pharmacological and pharmacological interventions, and the importance of taking the prescribed pain medications must be stressed.
- The safety issues of keeping pain medications in a secure place to avoid easy access for others, the adverse effects of pain medications, and the need to report them must be emphasised.
- The options for pain management and the right to choose must be explained.
- Pain and its interventions before and after surgery must be explained.
- Educational pamphlets for further instructions must be provided.
- Barriers to effective pain management, such as a fear of addiction, a fear of tolerance, concerns about side effects, fear of injections, and forgetting to take analgesics, must be discussed.

After implementing the pain interventions, evaluating the effectiveness of the nursing interventions applied to optimise the management is essential.

2.7.5 Evaluation

The evaluation phase is the phase that includes monitoring whether patient outcomes, as related to the nursing diagnoses, are met (Stanhope & Lancaster 2012:424). In pain management, evaluation is the step to assess the outcomes and adequacy of the nursing care plan for pain relief measures applied (Kiser-Larson 2017:90; Collins 2015:4; Song et al. 2015:461; White et al. 2013:130; Pasero and McCaffery 2011:102; Arnstein 2010:71). Evaluation is also an action to regularly reassess pain after interventions for effects and appropriateness of the pain interventions and to determine whether the interventions should be modified (Kiser-Larson 2017:90; Collins 2015:4; Macintyre & Schug 2015:19; Song et al. 2015:461; White et al. 2013:130; Pasero and McCaffery 2011:102; Arnstein 2010:71). The core purpose for pain reassessment is to evaluate the progress toward achieving and maintaining the desired pain management goals and outcomes met, level of improvement, patient safety issues, and evidence of any adverse effects (Pasero & McCaffery 102; Arnstein 2010:71). The aim is to evaluate if the patient still verbalises pain and discomfort, requesting analgesics, and rate the pain to the lowest level as compared from at onset of pain by using a pain scale. The widespread risks of adverse effects to observe after administration of analgesics are related to opioids such as sedation, respiratory depression, nausea, vomiting, constipation, confusion, dysphoria, dry mouth, orthostatic hypotension, urinary retention, and pruritus (Cole & LoBiondo-Wood 2014:406; RNAO 2013:86; Stannard et al. 2013:12; D'Arcy 2011:143; 2010:71). Arnstein (2010:71) suggests that during pain reassessment the patient should be requested to rate the pain intensity on the same pain scale used before the intervention and to estimate a per cent reduction in the pain intensity or to describe the amount of relief, for example, no relief = 0%, minimal effect = 10%, some effects = 25%, good effect = 50%, excellent effect = 75% and complete effect = 100%.

All pain management care provided should be "documented" in patients' medical records (Pasero and McCaffery 2011:102).

2.7.6 Documentation

Documentation of pain management should be accurate and should include all measures to control pain through all phases of the nursing process to provide clear communication about patient care (Black 2017:219; Brown et al. 2017:60; Song et al. 2016:461; DeWit and Kumagai 2013:132; RNAO 2013:40; Arnstein 2010:71). The use of electronic medical records and standardisation of pain documentation are recommended as these are more comprehensive, they ensure that related material can be safely saved and promote easy access to the information when needed by other professionals (Wang, Shen, Hong-Jun-Zhang, Li & Ji 2017:650; Pasero & McCaffery 2011:102).

2.8 CONCLUSION

In this chapter, some subsidiary questions which surfaced from the literature review were discussed:

- 1) What are the available resources for pain assessment?
- 2) Which characteristics and learning styles of nurses enhance the transfer of pain management competencies?
- 3) What is the learning content regarding pain assessment and management?
- 4) How is the transfer of learning climate within the hospitals' nursing care areas?

Based on these questions, the concept of the transfer of learning was explained, including the four dimensions of the Systemic Model of the Transfer of Learning by Donovan and Darcy. Furthermore, the concept of "transfer of learning" was reviewed to describe how the application of learning can be enhanced. Nurses' core competencies for pain management were reviewed as guided by systematically implementing the phases of the nursing process that comprise assessment, diagnosis, planning, implementation, evaluation, and documentation. Chapter 3 will elaborate on the methodology, data gathering, and interpretation of the research's first, second, and third phases.
CHAPTER THREE: RESEARCH DESIGN AND METHODOL-OGY: PHASE 1, 2 AND 3: METHODOLOGY AND DATA GATHERING

3.1 INTRODUCTION

In Chapter 3, the overarching research design of the explanatory sequential mixed method approach is described, with the latter part of the chapter focussed on Phases 1, 2 and 3, which will address the applicable population, sampling, data gathering technique, validity and reliability, data gathering and ethical aspects, as is illustrated in Table 3.1.

| Organisation and structure of the study | | | | |
|---|--------------------------------------|--|--|--|
| Chapter number | Chapter outline | Chapter content | | |
| Chapter 1 | Overview of the study | Contains the introduction, background of the study, the problem statement, research purpose and ob- jectives, research question, theoretical framework, key theoretical and operational concepts, the re- search design and methodology and ethical consid- erations. | | |
| Chapter 2 | Literature review | Consists of the literature review related to: 1) Systemic Model of Transfer of Learning by Donovan and Darcy, 2) Transfer of learning, and 3) Pain management and tools. | | |
| Chapter 3 | Research design and methodol- ogy | Illuminates the overarching research design Phases 1, 2 and 3 (quantitative phases): methodology and data gathering. | | |
| Chapter 4 | Data analysis and interpretation | Presents the data analysis and interpretation of the findings from Phases 1 to 3. | | |

Table 3.1 Organisation and structure of the study

| Chapter 5 | Phase 4 | Included a description of Phase 4 of the study: a) Literature review on action plan development b) Development of the draft action plan. c) |
|-----------|---|---|
| Chapter 6 | Phase 5 | Outlines and describes Phase 5 of the study (quali- tative phase): a) Methodology b) Validation of the action plan: c) The validated action plan. |
| Chapter 7 | Conclusion, recommendations, and limitations. | Deals with the conclusion, recommendations and limitations of the study. |

3.2 RESEARCH PARADIGM

A research paradigm is a set of assumptions about entities of a worldview or framework, epistemological stance, ontological, methodological or system of shared beliefs held by a community of research practice (Holton & Walsh 2017:212; Attri 2023: 86; Cypress 2021:14; Slutskiy 2021:4; Hall 2020:20). There are various commonly cited philosophical paradigms such as positivism, interpretivism, critical research, and pragmatism (Jacobsen 2020:103; Wa-Mbaleka & Rosario 2022:19; Marchiori 2018:278). The underlying philosophical paradigm for this study was based on the pragmatic paradigm.

A pragmatic paradigm is "a philosophical belief system that developed at the start of the 20th century out of the work of Charles Sanders Peirce, William James, John Dewey, and George Hebert Mead, which holds no allegiance to a particular set of rules or theories but rather suggests that different tools may be useful in different research contexts" (Leavy 2022:14; Gray & Grove 2021:76). Pragmatism is defined as a North American philosophical tradition that views reality as characterised by indeterminacy and fluidity and assumes that people are active and creative and that meaning emerges through practical actions to solve problems (Bryan & Charmaz 2019:658).

In contrast to positivism, interpretivism, and the critical approach, pragmatism embraces mixed methods as the third research community that works best for answering research questions (Harris & Muvuka 2023:28; Flick 2022:620). Thus, it is often associated with mixed methods research and assumes that the focus is on the following: the consequences or the outcomes of action that the researcher desires of research, on the primary importance of the question asked rather than the methods, on the use of multiple methods of data collection to inform the problem under study, and thus it is pluralistic and oriented toward "what works" and practice is what is important or "valid" for those people under study (Creswell & Clark 2011:41; Newton, Da Silva & Berry 2020:4 Johnson & Christensen 2019; Polit & Beck 2021:585; Kawachi, Lang & Ricciardi 2020:336; Various 2018:276).

The benefits of applying the pragmatic paradigm and embarking on the integration of quantitative (QUAN) and qualitative (QUAL) approaches enabled this study to address the research questions, value the participants and avoid the limitations of a single approach (Auzer 2017:6; DeCuir-Gunby & Schutz 2017:24; Creswell & Clark 2011:25; Flick 2022:21; Guralnick, Auer & Poce 2022:547; _Flynn 2022; 434; Onwuegbuzie & Johnson 2021:2; Tashakkori & Teddlie 2021:322; Stacey 2019:202; Regis 2018:115).

3.3 RESEARCH DESIGN

A research design is defined as all decisions the researcher made in "planning" to provide an overall structure for the procedures to pursue in conducting the research project, including decisions about measurement, sampling, procedures for collecting, analysing, interpreting, and reporting data in research studies (Creswell & Clark 2011:53; Houser 2021:144; Grove & Gray 2018:501; Rentala 2018:260). There are five major approaches to research, namely: (1) quantitative, (2) qualitative, (3) mixed methods research, (4) arts-based research, and (5) community-based participatory research (Leavy 2022:18). The research methods are specific techniques the researcher use to structure a study to select cases, gather and analyse information systematically, and report on results (Leavy 2022;144; Khosrow-Pour, Clarke, Jennex,

Antti, Kame, Lee, Kisielnicki, Gupt, Van Slyk, Wang & Weerakkody 2022:456). A "mixed method research design" was the appropriate approach and beneficial for this study.

3.3.1 Mixed method research (MMR) approach

Mixed method research is defined as the collection and analysis of quantitative (QUAN), "(designed to collect numbers)" and qualitative (QUAL), "(designed to collect words)" data, and then integrating the findings by drawing inferences from both approaches (Gray & Grove 2021:37; DeCuir-Gunby & Schutz 2017:2; Leavy 2017:9; Curry & Nunez-Smith 2015:4; Jokonya 2016:2; Subedi 2016:571).

The basic and primarily used mixed methods designs are (1) convergent parallel designs, (2) explanatory sequential designs, (3) exploratory sequential designs, and (4) embedded designs (DeCuir-Gunby & Schutz 2017:87; Jokonya 2016:3; Creswell 2015:6; Curry & Nunez Smith 2015:292; Creswell & Clark 2011:70; Gavin Ware & Johns 2018:37). The explanatory sequential mixed method design was found appropriate to achieve the study purpose to enhance the transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals.

An explanatory sequential mixed-method approach was pursued based on the following core characteristics the study encompasses, as stated by Creswell (2021:25), Creswell (2015:3), and Creswell and Plano (2011:5):

- (i) The collection and analysis of quantitative and qualitative data in response to research questions.
- (ii) The use of rigorous qualitative and quantitative methods.
- (iii) The combination or integration of quantitative and qualitative data using a specifically explanatory sequential mixed-method design and interpretation of this integration.

3.3.1.1 Justification for using mixed-method research approaches

Conducting mixed-method research has advantages as well as challenges. There are different rationales for using mixed-methods approaches. The reasons are related to the advantages associated with mixed-method approaches. The following advantages, as explained by Onwuegbuzie and Johnson (2021:163), DeCuir-Gunby and Schutz (2017:3), and McKim (2017:203), inspired the researcher to utilise a mixed-method approach:

- It permitted the examination of a complex problem, such as an action plan, to enhance the transfer of learning of pain management competencies of nurses within a single study. It enabled the researcher to provide evidence to triangulate or corroborate findings with multiple sources of evidence, such as the quantitative survey data of Phases 1, 2, and 3 that were merged to develop a draft action plan that was validated in the qualitative phase employing the e-Delphi technique.
- The design allowed the researcher to maximize the strengths of both quantitative and qualitative approaches while minimising their weaknesses, as both methods complement each other and allow for a more robust analysis.
- It further allowed for the use of multiple data-gathering instruments or tools. In this study context, questionnaires were used in the quantitative Phases 1, 2, and 3, consisting of closed-ended and open-ended questions for qualitative enhancement. In Phase 5, the e-Delphi technique (qualitative) was used to gather data using an applicable assessment tool to validate the action plan.

The challenges that are well-known when employing mixed-method research and that were challenging in this study were the following:

 The researcher needed to master quantitative and qualitative research methods and the specific skills requirements (Bergin 2018:186). With the thesis supervisor's assistance, the researcher could master and overcome the potential challenge. Extensive time, resources and effort for data collection and analysis were required (Cameron & Golenko 2023:255; Gray & Grove 2021:393; Onwuegbuzie & Johnson 2021:163). It was time-consuming to use both quantitative and qualitative approaches together in one study as the researcher needed to develop experience and skills in both methods and still comply with the university deadline to complete the study. The researcher obtained support from the supervisor and the statistician to address these challenges. The researcher worked according to a timetable that compels daily activities and responsibilities. The important rationale for the mixed-method approach for this study was that the qualitative phase was utilised to validate the quantitative findings used as the basis for the draft action plan, thus avoiding the deficiencies of any single approach.

The specific research design adopted was an explanatory sequential mixed method design to assist with developing an action plan to enhance the transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals.

3.3.1.2 Explanatory Sequential Mixed-Method Design

An explanatory sequential mixed-method design is explicated as a design that begins with a quantitative phase and then conducts a second qualitative phase to explain the quantitative results (Richards, Hemphill & Wright 2023:201; Creswell 2015:38; DeCuir-Gunby & Schutz 2017:86).

In this context, the first phase started with the collection and analysis of quantitative data, which has the priority for addressing the study's objectives, namely:

- Identify and describe the resources available for pain assessment (Phase 1 and 3, quantitative).
- 2) Identify and describe nurses' characteristics and learning styles that enhance the transfer of pain management competencies (Phase 2, quantitative).
- 3) Explore the teaching approaches the clinical facilitators employ during nurses' pain management education (Phase 3, quantitative).

- Describe the learning content regarding pain assessment and management (Phase 3, quantitative).
- 5) Describe the transfer of learning climate within the hospitals' nursing care areas (Phase 3, quantitative).
- 6) Develop an action plan to enhance the transfer of learning of pain management competencies of nurses (Phase 4, merged quantitative Phases 1, 2, and 3; and Phase 5 qualitative).

The data from Phases 1 to 3 were analysed and interpreted, and with the literature review conducted in Phase 4, the data were used to develop the first draft of the action plan.

The e-Delphi technique, a qualitative data-gathering technique, was used in Phase 5 to validate the draft action plan in various rounds until consensus was reached, as suggested by Polit & Beck 2021:236; DeCuir-Gunby and Schutz (2017:86), Leavy (2017:263), Subedi (2016:576), Creswell (2015:37), and Creswell and Clark (2011:71).

The diagrammatic representation and summary of the five-phase explanatory sequential mixed method study are presented in Figure 3.1.



Figure 3.1 The phases of the Explanatory Sequential Mixed Method approach

3.4 THE SETTING

A research setting is the specific physical, social and cultural site or location in which the researcher conducts the study that may be natural, partially controlled or highly controlled (Polit & Beck 2021:42; Richards 2021:58; Salmons 2016:90; Grove, Burns & Gray 2013:709).

The Kingdom of Saudi Arabia (KSA), wherein the study was conducted, was founded in 1932 by Ibn Saud and is a sovereign Arab state in Western Asia, also called the Middle East. KSA has a total area of 2,150,000 square kilometres. The country is Western Asia's fifth-largest sovereign state and the second-largest oil producer and exporter. The KSA has 13 regions bordered by Jordan and Iraq to the north, Kuwait to the northeast, Qatar, Bahrain and the United Arab Emirates to the East, Oman to the southeast and Yemen to the south. The Gulf of Aqaba separates the KSA from Israel and Egypt (see Figure 1.2).

This study was conducted in Riyadh, the capital city of KSA, which the Municipality of Riyadh manages. Within the capital city, four teaching hospitals offer nursing programs.

To retain and maintain their practice licences from the Saudi Commission for Health Specialities (SCFHS), all nurses must undergo professional ongoing education and earn the required credits of CME (continuing medical education) hours. One of the opportunities to gain CME credits can be attending pain management workshops.

PHASES 1, 2, AND 3: QUANTITATIVE RESEARCH

Quantitative research was conducted in three phases to achieve the objectives and address the research problem. Quantitative research was decided on due to the advantages as described by Creswell (2015:15):

 Quantitative research provides an opportunity for generalisation and precision.
 In these phases, the researcher gathered data about resources available to conduct pain assessment in Phases 1 and 3, nurses' characteristics and learning styles relevant to the enhancement of the transfer of pain management competencies in Phase 2 and teaching approaches employed during pain management education of nurses as well as the learning content regarding pain assessment and management.

(ii) The data obtained in quantitative research can be drawn from closed-ended questions (in this study, questionnaires, see Annexures 4, 5 and 8, were used to gather data).

3.5. POPULATION

A population is defined as a group of elements or individuals, objects, or substances that meet certain criteria for inclusion in a given universe about which the researcher could get information about the phenomenon under research (Ngulube 2021:299; Polit & Beck 2021:260; Leavy 2017:8; Gómez-Galán 2016:29).

3.5.1 Site population

The study was conducted in the Kingdom of Saudi Arabia (KSA), which is divided into 13 regions (refer to Section 1.9.3 for a detailed discussion). Riyadh is the capital city, and within the capital city, there are four teaching hospitals. Due to work permit restrictions, the researcher was only allowed access to two of these teaching hospitals.

The two teaching hospitals, named hospitals A and B, cater to all Ministry of National Guard employees and their families and civilians in need of medical care. The King Saud bin Abdulaziz University for Health Sciences collaborates with these two hospitals to train nursing students. A total number of 47 nursing wards existed in these two hospitals, and they constituted the nurses who participated in this study.

3.5.2 Site sampling

The researcher purposefully selected five nursing care divisions within the two teaching hospitals. A proportional stratified random sample of five nursing care divisions was made (see Table 3.1) as the hospitals did not have equal numbers of wards. There were, in total, 19 (nineteen) medical wards, 9 (nine) surgical wards, 8 (eight) paediatric wards, 5 (five) cardiac wards, and 6 (six) oby-gynae wards.

3.5.3. Population of professional registered nurses and clinical facilitators

A total of 1,462 professional registered nurses and 47 clinical facilitators were working in the five nursing care divisions (see Table 3.3, 3.4, and 3.5) of hospitals A and B in the 19 medical wards, nine surgical wards, eight paediatric wards, five cardiac wards, and the six oby-gynae wards.

3.6 SAMPLING

Sampling refers to a process by which a manageable number of individual cases or a sample is selected from a larger population for participation in a study (Ngulube 2021:300; Polit & Beck 2021:261; Leavy 2017:268; Rubin & Babbie 2014:380). Probability sampling, a *proportional stratified random sampling* method, was used to select the sample of registered professional nurses and clinical facilitators (see Table 3.2).

3.6.1 Proportional stratified probability sampling

A probability sampling method is considered the gold standard of sampling where everybody the researcher is interested in has an equal chance of being selected to participate in the study (Messinger & Guadalupe-Diaz 2020:343; Beins 2017:124). A *proportional stratified probability sampling* as described by Privitera (2022:42), Jha (2023:143) and Sharma (2022:260), was used to select the registered nurses and clinical facilitators from the five nursing care divisions.

A *proportionate stratified sampling* is when members of a population are classified into strata (groups), and the number of units from each stratum is directly equivalent to the size of the population in that stratum (Jha 2023:143; Privitera 2022:42; Sharma 2022:260; Polit & Beck 2021:268; Pajo 2017:149)

The reasoning for using a proportional stratified probability sampling method for this study was to recruit samples that were representative of the population to reduce the

potential for human bias, as suggested by Privitera (2022:42), Kumar (2019:157), Pecora (2018:52). All the professional registered nurses and clinical facilitators, according to each of the strata, formed the sampling frame that was provided to the nurse managers who did the sampling, taking into consideration the inclusion criteria (see Section 3.11) as suggested by Privitera (2022:42), Jha (2023:143) and Sharma (2022:260) Having calculated beforehand the stratified proportional random sample sizes in the two hospitals, as suggested by Privitera (2022:42), Jha (2023:143) and Sharma (2022:260) provided a significant justification for using probability sampling in the quantitative Phases 1, 2 and 3 in detail.

The advantages of proportional stratified probability sampling, as highlighted by Hall (2020:103) and Mellenbergh (2019:23), motivated the use of this method, namely:

- It was possible to select many professional registered nurses and clinical facilitators to reduce sampling error and ensure an accurately representative sample of the total population.
- Each professional registered nurse and clinical facilitator of the five nursing care divisions eligible for inclusion had an equal chance to be selected.

3.7 INCLUSION CRITERIA

Inclusion criteria are defined as the specific characteristics that define the eligible study population that the researchers will use to answer their research question (Polit & Beck 2021:261; Patino & Ferreira 2018:84)The inclusion criteria for professional registered nurses and clinical facilitators were:

- 1. Nurses must have attended at least one pain management workshop within the past three years.
- 2. Nurses must have attended in-service pain management training within the past 12 months.
- 3. Clinical facilitators must be responsible for pain management training nurses in the five nursing care divisions.

 4. Participants who agreed to be interviewed in English and were comfortable with it.

3.8 SAMPLE SIZE

A sample is defined simply as the number of participants (*n*) in a given research study or subsets of a larger population where data are collected to draw conclusions about the population from which it has come (Polit & Beck 2021:261; Paternoster & Bachman 2018:8; Leavy 2017:268; Leedy & Ormrod 2015:389). A total of 1462 professional nurses and 47 clinical facilitators were sampled (see Table 3.2)

The sample size for Phases 1, 2, and 3 was drawn from professional registered nurses and clinical facilitators working in the five nursing care divisions in Hospital A and Hospital B. For each nursing care division, the sample sizes were calculated using Rao Soft to determine a 95% confidence level to ensure the proportional sample size of each stratum (see Table 3.2) (Polit & Beck 2021 268; Mellenbergh 2019:23). A confidence level is the estimated probability that a population parameter lies within a given confidence interval (Dhinu 2021:127; Babbie 2020:206). Moreover, the larger the sample, the less sampling error, and adopting a 95% level of confidence, the study was limited to the margin of error to a 5% confidence interval (Ares & Varela 2018:541)

The samples were drawn from the total population of 1041 professional registered nurses and 32 clinical facilitators in Hospital A, and 421 nurses and 15 clinical facilitators in Hospital B were first stratified into five strata per five nursing divisions.

Table 3.2 indicates the population and the proportional allocation of sample size in each stratum calculated using the Raosoft sample size calculator at a 95% confidence level of each nursing care division.

| | Medic wards | al | Surgio wards | al | Paedia wards | ıtric | Cardia Wards | | Oby-g wards | ynae | Tota | |
|--|----------------|------------|-----------------|------------|-----------------|------------|-----------------|------------|----------------|------------|------------|------------|
| Nursing care divi- sion | Hospital A | Hospital B | Hospital A | Hospital B | Hospital A | Hospital B | Hospital A | Hospital B | Hospital A | Hospital B | Hospital A | Hospital B |
| Popula- tion size of nurses | 459 | 135 | 282 | 35 | 20 | 216 | 166 | 0 | 114 | 35 | 1041 | 421 |
| Sample size of nurses | 210 | 101 | 163 | 33 | 20 | 139 | 117 | 0 | 89 | 33 | 281 | 202 |
| Popula- tion size of Clini- cal facili- tators | 14 | 5 | 8 | 1 | 1 | 8 | 4 | 0 | 5 | 1 | 32 | 15 |
| Sample size of Clinical facilita- tors | 14 | 5 | 8 | 1 | 1 | 8 | 4 | 0 | 5 | 1 | 30 | 15 |

Table 3.2Population and Proportional sample sizes

3.8.1 Response rates: Phases 1, 2 and 3

Regardless of the intended numbers reflected in Table 3.2, the number of participants who volunteered to participate was less than anticipated in Phases 1 and 2 (see Table 3.3 and Table 3.4).

3.8.1.1 Phase 1

Four hundred and twenty-three questionnaires were distributed to the sample of volunteered professional nurses, and 385 questionnaires (Questionnaire 1, Annexure 4) were received back, with an overall response rate of 91%. This suggests a very good response rate as, according to Flynn (2022:242) and Lohr (2021:Ch 8.8) a 70% response rate is considered very good and maintains representativeness. The response rate varied between the different types of nursing wards, as illustrated in Table 3.3.

| Type of Unit | Number of | Number of | Number of re- | Response rate |
|--------------|-----------|----------------|---------------|---------------|
| | wards | questionnaires | spondents | % |
| | | sent out | /Sample size | |
| Medical | 19 | 140 | 122 | 87.1% |
| Surgical | 9 | 163 | 163 | 100 % |
| Paediatric | 8 | 54 | 45 | 83.3% |
| Cardiac | 5 | 34 | 32 | 94.1% |
| Oby-gynae | 6 | 32 | 23 | 71.9% |
| Total | 47 | 423 | 385 | 91% |

Table 3.3 Phase 1: Sample size (N = 385) and Response rate

3.8.1.2 Phase 2

384 professional registered nurses out of the 423 who received Questionnaire 2 (see Annexure 5) completed the questionnaires, thus also achieving a good response rate of 90.8%, as was explained (see Section 4.3). The distribution of the respondents within the different strata is illustrated in Table 3.4

Table 3.4 Phase 2: Sample size (N = 384) and Response rate

| Type of Unit | Number of | Number of | Number of re- | Response rate |
|--------------|-----------|----------------|---------------|---------------|
| | wards | questionnaires | spondents | |
| | | sent out | | % |
| Medical | 19 | 140 | 122 | 87.1% |
| Surgical | 9 | 163 | 163 | 100 % |
| Paediatric | 8 | 54 | 46 | 85.2% |
| Cardiac | 5 | 34 | 31 | 91.2% |
| Oby-gynae | 6 | 32 | 22 | 68.8% |
| Total | 47 | 423 | 384 | 90.8% |

3.8.1.3 Phase 3

A total of the 47 questionnaires were distributed by the gatekeepers to 47 clinical facilitators, and all were received back, resulting in a 100% response rate (see Table 3.5).

| Type of Unit | Number of wards | Number of questionnaires send out | Number of re- spondents | Response rate % |
|--------------|--------------------|---|----------------------------|--------------------|
| Medical | 19 | 19 | 19 | 100% |
| Surgical | 9 | 9 | 9 | 100% |
| Paediatric | 8 | 9 | 9 | 100% |
| Cardiac | 5 | 4 | 4 | 100% |
| Oby-gynae | 6 | 6 | 6 | 100% |
| Total | 47 | 47 | 47 | 100% |

Table 3.5 Phase 3: Sample size (N = 47) and Response rate

3.9 DATA COLLECTION INSTRUMENTS: PHASES 1, 2 AND 3

A research instrument can be defined as a tool used to collect data and measure and assist with the analysis of the responses on the related study variables (Wang 2018:263; Kearney-Nunnery 2016:89; Valencia-Go 2016:48). The questionnaire was the data-gathering technique of choice in Phases 1 to 3.

3.9.1 Questionnaires

Three different questionnaires, a specific one to gather data in each of the three phases, were developed to enable the researcher to gather relevant data that would be useful for developing an action plan to enhance the transfer of learning of pain management competencies. Questionnaire one (see Annexure 4) was developed to gather data (information about resources available for pain assessment) from registered nurses between 14 December 2019 and 13 March 2020. Questionnaire 2 (see Annexure 5) was developed to gather data from registered nurses about nurses' characteristics and learning styles to enhance the transfer of pain management competencies between 14 December 2019 and 13 March 2020 December. Questionnaire 3 was used to obtain data about resources available to conduct a pain assessment, teaching approaches employed during pain management education of nurses, learning content regarding pain assessment and management and the transfer of learning climate within the hospital nursing care (see Annexure 8) from clinical facilitators between 12 December to 17 March 2020.

The advantages of using questionnaires in research as explained by various authors (Sharma 2022:209; Polit & Beck 2021:235, Ackini & Saunder 2015 363, Zickermann 2014:6) motivated the use of questionnaires:

- Large amounts of data could be collected within three months between 12 December 2019 and 17 March 2020.
- It was affordable as the questionnaires were self-administered, and large numbers of participants were reached easily in a cost-efficient way to distribute all questionnaires to all respondents as the researcher was the sole distributor to those who voluntarily agreed to participate without reward.
- The quality of the data collected was proofed to be valid and reliable as the questionnaires underwent pre-testing that yielded validity and reliability before they were used in the main study (LoBiondo-Wood & Haber 2014:284).
- The privacy and anonymity of respondents were ensured with the use of questionnaires. No personal identifiers such as names, addresses, telephone numbers, or hospital badge numbers were requested to maintain confidentiality with which the respondents answered the questions (Chen &Terken 2022:189)

- The questions in all three questionnaires were practical to participants' daily nursing practice pertaining to pain management, providing uniformity in a predetermined order and covering the study objectives.
- The respondents could use their own time and space, and it was explained that it would only take an average of 25 minutes to complete. The average time was calculated during pre-testing.

Each questionnaire had the characteristics associated with the research questions and the objectives of this study.

3.9.2 The characteristics of the developed questionnaires

All three questionnaires were developed after a thorough literature review was conducted. The questionnaires were completed in English as the researcher considered the factors of linguistic competency and literacy and because English is the official language amongst all healthcare professionals in the two hospitals.

3.9.2.1 Questionnaire 1:

The questionnaire used in Phase 1 was completed by professional registered nurses to identify the resources available for pain assessment. It encompassed two sections. Section A included only close-ended questions, and Section B consisted of closedended questions with one open-ended question.

- Section A: The five close-ended questions focussed on the biographical information of the professional registered nurses.
- Section B: Fourty four "yes" or "no" dichotomous questions were included. All close-ended questions were content-related, identifying whether resources to conduct pain assessments were available. For qualitative enhancement, one open-ended question allowed for any comments (see Annexure 4).

3.9.2.2. Questionnaire 2:

Questionnaire 2, developed for Phase 2, was completed by professional registered nurses to identify their characteristics and learning styles to enhance the transfer of pain management competencies. It encompassed three sections: Sections A and B included only close-ended questions, and Section C included closed-ended questions and one open-ended question.

- Section A: Five close-ended questions about the biographical information of the professional registered nurses.
- Section B: Thirty-four closed-ended questions to rate on a scale of 1 to 10, then ranking from high to low only three statements from the rated questions. The questions were divided into four main questions about trainee characteristics identified to enhance the transfer of learning of pain management competencies.
- Section C: Fourteen closed-ended questions to rate learning styles identified to enhance the transfer of learning of pain management competencies on a scale from 1 to 10, then ranking from high to low only three statements from the rated questions. For qualitative enhancement, one open-ended question allowed for any comments (see Annexure 5).

3.9.2.3. Questionnaire 3:

The questionnaire used in Phase 3 *was* completed by clinical facilitators to identify resources available to conduct a pain assessment, explore the teaching approaches employed by the clinical facilitators during pain management education of nurses, describe the learning content regarding pain assessment and management, and describe the transfer of learning climate within the hospitals' nursing care areas. The questionnaire consisted of five sections with Sections A to D, close-ended questions, and Section E close-ended questions, with one open-ended question (see Annexure 8).

- Section A: Five close-ended questions about the biographical information of the clinical facilitators.
- Section B: Fourty-four questions about resources available for pain assessment (see Annexure 8).
- Section C: Fourteen questions about teaching approaches employed during pain management education of nurses (see Annexure 8).
- Section D: Forty-one questions based on the nursing process to describe learning content regarding pain assessment and management (see Annexure 8).
 Section E: Eleven questions to describe the transfer of learning climate within the hospital nursing care areas (see Annexure 8).

All three questionnaires were pre-tested and refined where necessary before data gathering commenced (Arai 2022:81; Zahoor 2021:81).

3.10 PRE-TESTING OF THE QUESTIONNAIRES

All questionnaires were pretested following the motivations for pre-testing as described by different authors (Gularso & Okti Purwaningsih 2023:382; Jönsson & Prins. 2019:63; Leavy 2017:264). The pre-tests were done to:

- Verify that respondents understood the wording of each questionnaire in the same way.
- Identify problems such as unclear questions, unfamiliar words, ambiguous syntax and lack of an appropriate answer.
- Consider the practicability of the main study to modify any issue related to questionnaire wording, layout and question sequencing.
- Determine how much time will be spent to complete the questionnaires.

After ethics approval to conduct the study was received from the Health Studies Research Ethics Committee, Department of Health Studies, Unisa REC-012714-039 (see Annexure 6a), REC-240816-052 (see Annexure 6b); and Nursing Services Permission to conduct research (see Annexure 7a), KAIMRC and the Institutional Review Boards (IRB) approval study number SP 18/036/R (see Annexure 7b); IRB Annual Extension SP 18/036/R (see Annexure 7c); and IRB 6 Months Extension SP 18/036/R (see Annexure 7d) of study hospitals A and B. The pre-tests were done between 16 July to 23 September 2019.

Non-sampled nursing divisions with similar characteristics to those of the main study, namely, emergency care and critical care, were used for pre-testing as recommended (Kline 2023:38, Sharma 2022:401).

To recruit the participants to participate in the pre-tests, a memo, accompanied by the information letter, was e-mailed to the emergency care and critical care unit nurse managers (refer to Annexures 1 & 2). The managers then provided the researcher with a list of names of volunteers. The researcher then followed the process discussed in Section 3.13 to collect the pre-testing data from the conveniently selected volunteers.

After completing the questionnaires, the respondents placed them in sealed boxes provided in each ward, and the researcher fetched them the following day.

The results of the pre-tests of each of the developed questionnaires were as follows.

3.10.1 Questionnaire 1: pre-test (professional nurses' Phase 1):

The pre-test was completed by 99 professional registered nurses in Hospital A and 40 in Hospital B on 23 August 2019 and retested on 07 September 2019.

The researcher worked with the statistician to run reliability statistics for Questionnaire 1. The pre-tested questionnaire was analysed by means of "R Statistical Software" ("R") (refer to Section 3.11 for a detailed discussion). The outcomes of the pre-tests of Questionnaire 1 were assessed by the statistician and supervisor, who gave feedback regarding the items to adapt. The modifications of words and phrases of some questions were done according to the feedback from the pretesting respondents. The questions in Section B of Questionnaire 1 were the same as in Section B of Questionnaire 3 for assessing resources available for pain assessments. Therefore, the changes implemented for Section B of Questionnaires 1 and 3 were similar. The following changes in Section B of Questionnaires 1 and 3 were implemented, as illustrated in Table 3.6. Changes were implemented before the final data gathering for Phases 1 and 3 commenced.

| Questionnaire Section B | Resources available to conduct a pain as | ssessment |
|----------------------------|--|---|
| | Please answer all the questions. Please indicate if you agree or disa- gree by marking the statement that de- scribes your choice with a tick ($$). | Please answer all the questions. Please choose "Yes" or "No" with a tick ($$) as your response to ALL state- ments below that describe your choice. |
| Item no | Words and phrases for modifications. | Modifications included. |
| 1 | | |
| 1.1 | The use of QUEST (question the child, use pain rating tools, evaluate behav- iour, sensitise parents, and take action) approach is used to assess pain. | QUEST (question the child, use pain rating tools, evaluate behaviour, sen- sitise parents, and take action) ap- proach is available to assess pain. |
| 1.2 | The use of WILDA (words to describe pain, intensity, location, duration, and aggravating or alleviating factors) ap- proach is a comprehensive guide tool used to assess pain. | WILDA (words to describe pain, inten- sity, location, duration, and aggravat- ing or alleviating factors) approach is available to assess pain. |
| 1.3 | The use of PQRST (provoking/pallia- tion factors, quality of pain, region of pain, severity and timing) approach is a comprehensive guide tool used to as- sess pain. | PQRST (provoking/palliation factors, quality of pain, region of pain, severity and timing) approach is available to assess pain. |
| 1.4 | The use of OPQRSTUV (onset of pain, provoking/palliating, quality, | OPQRSTUV (onset of pain, provok- ing/palliating, quality, region/radiation |

 Table 3.6
 Questionnaires 1 and 3 Section B modifications:

| region/radiation of pain, severity of | of pain, severity of pain, timing/treat- |
|--|--|
| pain, timing/treatment, understand- | ment, understanding/impact on you |
| ing/impact on you and values) ap- | and values) approach is available to |
| proach is a comprehensive to assess | assess pain. |
| pain. | |
| The use of COLDSPA (character, on- | COLDSPA (character, onset, location, |
| set, location, duration, severity, pattern | duration, severity, pattern and associ- |
| and associated factors) approach is a | ated factors) approach is available to |
| comprehensive to assess pain. | assess pain |
| The Wong-Baker FACES pain scale is | The Wong-Baker FACES pain scale is |
| used to rate pain children who can re- | available to rate pain in children who |
| port their pain. | can report their pain. |
| The Numeric Rating Scale (NRS) is | The Numeric Rating Scale (NRS) is |
| used to rate pain in children and adults | available to rate pain in children and |
| who can report their pain. | adults who can report their pain. |
| The Verbal Analogue Scale (VAS) is | The Verbal Analogue Scale (VAS) is |
| used to rate pain in adults who can re- | available to rate pain in adults who |
| port their pain. | can report their pain. |
| The Verbal Descriptor Scale (VDS) is | The Verbal Descriptor Scale (VDS) is |
| used to rate pain in adults who can re- | available to rate pain in adults who |
| port their pain. | can report their pain. |
| The Brief Pain Inventory (BPI) is a | The Brief Pain Inventory (BPI) is a |
| questionnaire used to assess pain in | questionnaire available to assess pain |
| cancer patients. | in cancer patients. |
| Pain rating assessment tools of pa- | Pain rating assessment tools of pa- |
| tients who cannot self-report their pain | tients who cannot self-report their pain |
| included below are available in your | included below are available in your |
| context to be used to rate pain during | context to rate pain during pain as- |
| pain assessments: | sessments: |
| A CRIES (Crying, Required oxygen, In- | A CRIES (Crying, Required oxygen, |
| creased vital signs, Expressions, | Increased vital signs, Expressions, |
| Sleeplessness) pain scale is available | Sleeplessness) pain scale is available |
| to be used to rate pain in premature | |
| | region/radiation of pain, severity of pain, timing/treatment, understand- ing/impact on you and values) ap- proach is a comprehensive to assess pain. The use of COLDSPA (character, on- set, location, duration, severity, pattern and associated factors) approach is a comprehensive to assess pain. The Wong-Baker FACES pain scale is used to rate pain children who can re- port their pain. The Numeric Rating Scale (NRS) is used to rate pain in children and adults who can report their pain. The Verbal Analogue Scale (VAS) is used to rate pain in adults who can re- port their pain. The Verbal Descriptor Scale (VDS) is used to rate pain in adults who can re- port their pain. The Verbal Descriptor Scale (VDS) is used to rate pain in adults who can re- port their pain. The Brief Pain Inventory (BPI) is a questionnaire used to assess pain in cancer patients. Pain rating assessment tools of pa- tients who cannot self-report their pain included below are available in your context to be used to rate pain during pain assessments: A CRIES (Crying, Required oxygen, In- creased vital signs, Expressions, Sleeplessness) pain scale is available to be used to rate pain in premature |

| | and neonates during pain assess- | to rate pain in premature and neo- |
|-----|---|---|
| | | hates during pair assessments. |
| 3.2 | The Neonatal Pain, Agitation and Se- | The Neonatal Pain, Agitation and Se- |
| | dation Scale (N-PASS), is pain scale | dation Scale (N-PASS), is available to |
| | available to be used to rate pain in | rate pain in premature and neonates |
| | premature and neonates during pain | during pain assessments. |
| | assessments. | |
| 3.3 | The Neonatal Infant Pain Scale (NIPS) | The Neonatal Infant Pain Scale |
| | is pain scale available to be used to | (NIPS) pain scale is available to rate |
| | rate pain in premature and neonates | pain in premature and neonates dur- |
| | during pain assessments. | ing pain assessments. |
| 3.4 | A FLACC (Faces, Legs, Activity, Cry- | A FLACC (Faces, Legs, Activity, Cry- |
| | ing, and Consolability) is behavioural | ing, and Consolability) pain scale is |
| | pain scale is available to be used to | available to rate pain for patients who |
| | rate pain for patients who cannot ver- | cannot verbalise their pain during pain |
| | balise their pain during pain assess- | assessments. |
| | ments. | |
| 3.5 | The COMFORT-Behaviour pain scale | The COMFORT-Behaviour pain scale |
| | (COMFORT-B) is available to be used | (COMFORT-B) is available to rate |
| | to rate pain for patients unable to ver- | pain for patients unable to verbalise |
| | balise their pain during pain assess- | their pain during pain assessments. |
| | ments. | |
| 3.6 | The Critical Care Pain Observational | The Critical Care Pain Observational |
| | Tool (CPOT) is used to assess and | Tool (CPOT) is available to assess |
| | manage pain in adult non-conscious, | and manage pain in adult non-con- |
| | critically-ill ventilated or non-ventilated | scious, critically-ill ventilated or non- |
| | patients in intensive care unit. | ventilated patients in intensive care |
| | | unit. |
| 3.7 | The Behavioural Pain Scale (BPS) is | The Behavioural Pain Scale (BPS) is |
| | used for assessing pain in uncommuni- | available for assessing pain in uncom- |
| | cative, critically-ill, sedated and intu- | municative, critically-ill, sedated and |
| | bated patients in intensive care units. | intubated patients in intensive care |
| | | units. |
| | | 1 |

| 4 | Pain rating assessment tools used for elderly patients with dementia or cogni- tive impairment included below are available in your context to be used to rate pain during pain assessment: | Pain rating assessment tools for el- derly patients with dementia or cogni- tive impairment included below are available in your context to rate pain during pain assessment: |
|-----|---|---|
| 4.1 | The Abbey Pain Scale (ABBEY) is used to measure acute, chronic and acute-on chronic pain intensity in pa- tients with late-stage dementia. | The Abbey Pain Scale (ABBEY) is available to measure acute, chronic and acute-on chronic pain intensity in patients with late-stage dementia. |
| 4.2 | The Checklist on Nonverbal Pain Indi- cators (CNPI) is used to measure pain behaviours in cognitively impaired older adults. | The Checklist on Nonverbal Pain Indi- cators (CNPI) is available to measure pain behaviours in cognitively im- paired older adults. |
| 4.3 | The Pain Assessment in Advanced De- mentia Scale (PAINAD) is used to as- sess pain in patients with advanced moderate to severe dementia. | The Pain Assessment in Advanced Dementia Scale (PAINAD) is available to assess pain in patients with ad- vanced moderate to severe dementia. |
| 4.4 | The Non-communicative Patient's As- sessment Instrument (NOPPAN) is pain scale used to assess pain in de- mented and cognitively impaired pa- tients. | The Non-communicative Patient's As- sessment Instrument (NOPPAN) pain scale is available to assess pain in de- mented and cognitively impaired pa- tients. |
| 5 | Human resources included below are available in your context to be used to conduct pain assessments: | Human resources included below are available in your context to conduct pain assessments: |
| 6 | Patient support as a resource available in your context to be used to conduct pain assessments: | Patient support as a resource included below is available in your context to conduct pain assessments: |
| 7 | Publications and electronic resources available in your context to be used to conduct pain assessments: | Publications and electronic resources available in your context to conduct pain assessments: |

| At the end of | Would you like to add any comments? |
|---------------|-------------------------------------|
| the question- | |
| naire: | |

3.10.2 Questionnaire 2: Phase 2 (professional nurses)

The same professional registered nurses who completed Questionnaire 1 also pretested Questionnaire 2. The pre-testing of data collection of the questionnaire was undertaken on 23 August 2019, and the retest was done on 7 September 2019. Ninetynine registered nurses working in emergency care and critical care in Hospital A and forty in Hospital B participated in the pre-testing, as was said.

At the end of the pre-testing, the researcher worked with the statistician to run the reliability statistics of Questionnaire 2. The pre-tested Questionnaire 1 was analysed also utilising "R Statistical Software" ("R") (refer to Section 3.11 for a detailed discussion). The outcomes of the pre-tests of Questionnaire 2 were also assessed by the statistician and supervisor, who gave feedback regarding the items to adapt. The modifications of words and phrases of some questions implemented in Sections B and C were done according to the feedback from the pretesting respondents. The questions were adapted, as illustrated in Table 3.7, as follows:

Table 3.7 Questionnaire 2: Modifications

| | Words and phrases for modifications | Modifications included |
|---------------|--|--|
| Questionnaire | Characteristics that enhance transfer of | Characteristics seen to enhance trans- |
| Section B | learning of pain management compe- | fer of learning of pain management |
| | tencies. | competencies. |
| Item no | Words and phrases for modifications. | Modifications included. |

| Question 1 | Which of the following describes the ability you possess to apply what you have learned? | How do you apply what you have learned? Rate yourself on a scale from 1 (this does not describe me at all) to 10 (this describes me perfectly). ①②③④⑤⑦⑧⑩ |
|------------|---|---|
| 1.1 | Remember the pain management infor- mation from past experience. | I remember the pain management in- formation from past experience. |
| 1.2 | Learn during pain management learn- ing/training sessions. | I learn during pain management learn- ing/training sessions. |
| 1.3 | Concentrate well during pain manage- ment learning/training sessions | I concentrate well during pain manage- ment learning/training sessions. |
| 1.4 | Understand the content of information taught during pain management learn- ing/training sessions. Please motivate your answer: | I understand the content of information taught during pain management learn- ing/training sessions. |
| 1.5 | Choose an appropriate pain interven- tion strategy for every individual pa- tient's pain level. | I choose an appropriate pain interven- tion strategy for every individual pa- tient's pain level. |
| 1.6 | Think rationally to assess a patient ex- periencing pain. Please motivate your answer: | I think rationally to assess a patient experiencing pain. |
| 1.7 | Perform an accurate pain assessment. | I perform an accurate pain assess- ment. |
| 1.8 | Assess pain on time. | I assess pain on time. |
| 1.9 | Reassess pain after interventions. | I reassess pain after interventions. |
| 1.10 | Apply the knowledge by orientating new colleagues on how to assess and manage pain. | I apply knowledge by orientating new colleagues on how to assess and man- age pain. |

| | Please motivate your answer: | Please choose from High to Low ONLY |
|---------------|--|---|
| | | 3 (three) statements from above ques- |
| | | tions 1.1 to 1.10 that best describe you. |
| | | Please write only the statement ques- |
| | | tion number in the box below: |
| | | 1 st |
| | | 2 nd |
| | | 3 rd |
| Questionnaire | Learning styles that enhance transfer of | Learning styles seen to enhance trans- |
| Section C | learning of pain management compe- | fer of learning of pain management |
| | tencies. | competencies. |
| Item no | Words and phrases for modifications. | Modifications included. |
| Question 2 | Which of the following best describes | How does the following describe you as |
| | you as a learner? I am a/an: | a learner? |
| | | Rate yourself on a scale from 1 (this |
| | | does not describe me at all) to 10 (this |
| | | describes me perfectly). |
| | | 123456789 |
| | | |
| 2.1 | Self-directed learner. | Self-directed learner. (Learner taking |
| | | charge of his/her learning). |
| 2.2 | Inquisitive thinking learner. | Inquisitive thinking learner. (Learner |
| | | who is inclined to ask questions or ea- |
| | | ger for knowledge). |
| 2.3 | Curious thinking learner. | Curious thinking learner. (Learner hav- |
| | | ing desire to learn or know more about |
| | | something) |
| 2.4 | Enthusiastic thinking learner. | Enthusiastic thinking learner. (Learner |
| | | showing interest or excitement about |
| | | learning and doing something). |

| 2.5 | Truth-seeking learner. | Truth-seeking learner. (Learner who asks challenging questions or asks truth, reasons and evidence about something he/she is learning). |
|------------|---|---|
| 2.6 | Organised thinking learner. | Organised thinking learner. (Learner who is able to think carefully to plan about something to learn or do). |
| 2.7 | Hard-working in enquiring learner. | Hard-working in enquiring learner. (Leaner putting efforts in doing a lot of work to know about something). |
| 2.8 | Self-confidence thinking learner. | Self-confidence thinking learner. (A leaner who believes in oneself and knows about one's own ability to learn and do something) |
| 2.9 | Creative learner. | Creative learner. (Learner having cour- age to try to learn new things, feeling to produce ideas, like to be the first do something and open to share his/her experience with others). |
| | For every yes answer ticked, please provide a motivation for choosing the specific aspect that describes you as a learner: | Please choose from High to Low ONLY 3 (three) statements from above ques- tions 2.1 to 2.9 that best describe you: Please write only the statement ques- tion number in the box below: 1st 2nd 3rd |
| Question 3 | Which of the following describes how you are motivated to learn? | How does the following describe how you are motivated to learn? Rate your- self on a scale from 1 (this does not de- scribe me at all) to 10 (this describes me perfectly). 1-2345678910 |

| 3.1 | Choosing to attend a pain manage- ment training program encourages learning. | Attending pain management training programs motivates me to learn. |
|------------|---|---|
| 3.2 | Your initiatives without the assistance of others encourage learning of pain management. | The ability to take initiative without the assistance of others motivates me to learn. |
| 3.3 | Learning of pain management skills rel- evant to working area encourages learning. | Learning of pain management skills rel- evant to my working area motivates me to learn. |
| 3.4 | The efforts that result in learning new knowledge about pain management encourage learning. | Gaining new knowledge about pain management motivates me to learn. |
| 3.5 | Knowing the desired goals for learning pain management encourages learn- ing. | Knowing the desired goals for learning pain management motivates me to learn. |
| 3.6 | Your own goals about learning pain management encourage learning. | My own goals about knowing pain man- agement motivate me to learn. |
| | For every yes answer ticked, please provide a motivation for choosing the specific aspect that describes how you are motivated to learn: | Please choose from High to Low ONLY 3 (three) statements from above ques- tions 3.1 to 3.6 that best describe you: Please write only the statement ques- tion number in the box below: 1st 2nd 3rd |
| Question 4 | Which of the following describes how you are motivated to apply in practice what you have learned? | How does the following describe how you are motivated to apply what you have learned in practice? Rate yourself on a scale from 1 (this does not describe me at all) to 10 (this describes me perfectly). |

| | | Û23€56789₩ |
|-----------|---|--|
| | | |
| | | |
| | | |
| | | |
| | | Please choose from High to Low ONLY |
| | | 3 (three) statements from above ques- |
| | | Please write only the statement ques- |
| | | tion number in the box below: |
| | | |
| | | 1 st |
| | | 2nd |
| | | |
| | | 3'' |
| Section C | Learning style that enhances transfer | Learning styles seen to enhance trans- |
| | of learning of pain management com- | fer of learning of pain management |
| | petencies | competencies. |
| Item no | Words and phrases for modifications. | Modifications included. |
| | Please tick (\checkmark) in the appropriate box | How does the following describe your |
| | how you prefer to learn about pain | learning styles? |
| | management. | Rate yourself on a scale from 1 (this |
| | | does not describe me at all) to 10 (this |
| | | describes me perfectly). |
| | | Û23 4 56789® |
| 1 | By just reading through the information. | By reading through the information. |
| 3 | By just listening to the information such | By listening to the information such as |
| | as during a lecture. | during a lecture <u>.</u> |
| | | Please choose from High to Low ONLY |
| | | 3 (three) statements from above items |
| | | 1 to 14 that best describe you: Please |

| | write only the statement question num- ber in the box below: |
|--|---|
| | 1 st 2 nd 3 rd |
| Please add any comment or sugges- tion on how pain management training can be improved to ensure positive outcomes to you as the learner: | Would you like to add any comments? |

3.10.3 Questionnaire 3: Phase 3 (clinical facilitators)

Questionnaire 3, developed for Phase 3, was pre-tested by clinical facilitators who volunteered to participate, as mentioned. The pre-testing of data collection was undertaken on 15 July 2019, and it was retested on 30 July 2019. Three clinical facilitators in Hospital A and two in Hospital B, working in emergency care and critical care, participated in the pre-testing.

At the end of the pre-testing, the researcher worked with the statistician to run the reliability statistics of Questionnaire 3 (see Section 3.11). The pre-tested Questionnaire 3 was also analysed by means of "R Statistical Software" ("R") (refer to Section 3.11 for detailed discussion). The outcomes of the pre-tests of Questionnaire 3 were also assessed by the statistician and supervisor, who gave feedback regarding the items to adapt.

The questions in Section B of Questionnaire 3 were the same questions in Section B in Questionnaire 1. Consequently, the changes applied for both Section Bs of questionnaires 3 and 1 were similar (see Table 3.5). The modifications of words and phrases of some questions implemented in Sections C, D, E and F were also done

according to the feedback from the pretesting respondents. The questions were modified as illustrated in Table 3.8 as follows:

| Ques- tionnaire Section C | Teaching approaches employed during pain management education of nurses. | |
|------------------------------------|---|--|
| | Words and phrases for modifications. | Modifications included. |
| | Please answer all the questions. Please indicate if you agree or disagree by marking the statement that describes your choice with a tick (\checkmark). | Please answer all the questions. Please choose "Yes" or "No" with a tick ($$) as your response to ALL statements below that describe your choice. |
| Ques- tionnaire Section D | Learning content regarding pain assessment and management. | |
| | Words and phrases for modifications. | Modifications included. |
| | Please answer all the questions. Please indicate if you agree or disagree by marking the statement that describes your choice with a tick (\checkmark). | Please answer all the questions. Please choose "Yes" or "No" with a tick ($$) as your response to ALL statements below that describe your choice. |
| Ques- tionnaire Section E | Transfer of learning climate within the hospital nursing care. | |
| | Words and phrases for modifications. | Modifications included. |
| | Please answer all the questions. Please indicate if you agree or disagree by marking the statement that describes your choice with a tick (\checkmark). | Please answer all the questions. Please choose "Yes" or "No" with a tick ($$) as your response to ALL statements below that describe your choice. |

Table 3.8 Questionnaire 3 Sections C, D, E and F: Modifications

| At the end of Section E | | Would you like to add any com- ments? |
|----------------------------------|--|--|
| | Words and phrases for modifications. | Modifications included. |
| Ques- tionnaire Section F | Please answer the following question as honestly as possible. | Removed |
| Item no | Words and phrases for modifications. | Modifications included. |
| 1 | What is your general view about the knowledge, skills and attitude of nurses in your hospital about pain manage- ment? Which teaching strategy did you find most appropriate to teach pain manage- ment? Please motivate why you indicate the specific teaching strategy. | Removed |
| 2 | Please provide any suggestion on what you will include if you need to prepare the training program on pain manage- ment for the nurses. | Removed. |
| 3 | Please describe how the hospital envi- ronment is supportive to nurses to be taught about pain management and the application of their knowledge and skills. | Removed. |
| 4 | Please provide any suggestions on what can be done to improve the pain man- agement knowledge and skills of nurses in your hospital. | Removed. |

3.11 VALIDITY AND RELIABILITY OF THE QUESTIONNAIRES

The pre-testing of the questionnaires was done to ensure the data quality, thus testing the validity and reliability (Mitra 2023:26; Chin, Caputo, Lin & Hu 2022:82).

3.11.1 Validity

Validity refers to the extent to which an instrument can accurately measure what it is intended to measure (Kumar 2019:206; Khakshooy & Chiappelli 2018:35; Beins 2017:130; Leavy 2017:113) To enhance the validity of this study, respondents self-administered the questionnaires in their own time and space without any interventions from others. Kumar (2019:206), Taylor (2017:285) illustrate that several different forms are used to develop and assess the validity of research questionnaires, such as face validity, content validity, criterion validity and construct validity. To ensure that the study research questionnaires measure what they were purported to measure, *face validity* and *content validity* were used to validate the questionnaires as embraced by Gray and Grove (2021:463).

3.11.1.1 Face validity

Face validity or superficial appearance of a questionnaire is referred to as whether on the face of the questionnaire, the questions appear to respondents to be relevant and measure the underlying construct (Gray and Grove 2021:463; Gravetter & Forzano 2018:59). In this study, the questionnaires were evaluated by the supervisor, statistician, and the scientific committee of the Department of Health Studies to evaluate whether they were measuring what needed to be measured before being pre-tested by the respondents.

The questionnaires and the items were factually designed, taking into account the thorough literature review specific to the topic of the study conducted in books between 2010 and 2018 and journals, dissertations, electronic documents, and articles from 2013 to 2018 to investigate avenues and address the objectives of the study (see Chapter 2). This helped the researcher to develop the three questionnaires (refer to Section 3.9).

The supervisor, the statistician, and the mentioned scientific review committee assessed and gave feedback and determined if the instruments measured what they were intended to measure. In this manner, the items of the questionnaires were determined to have face validity as they were well-thought-out and appropriate to measure the intended construct.

3.11.1.2 Content validity

Content validity is defined as the degree to which items in a research instrument reflect the content domain or universe of the construct it aims to measure (Management Association, Information Resources 2018:425; Steyn 2017:2; Taylor 2017:285; Taherdoost 2016:29; Tiko 2015:16). The three questionnaires were evaluated for content validity (Rea 2022:87). The content of the three questionnaires derived from the literature review (refer to Chapter 2) and consultations with the supervisor, the statistician, and a scientific review group to ensure that the study objectives could be met.

Reliability was ensured before application to the main study as it is a prerequisite for validity (Gray and Grove 2021:458).

3.11.2 Reliability

Reliability is defined as a measure that refers to the consistency, stability, and repeatability or reproducibility of observations, measures, or data collected using the same methodology on more than one occasion across different but related test items or by different individuals (Gray and Grove 2021:458; Beins 2017:130; Leavy 2017:113; Curry & Nunez-Smith 2015:176). In this context, the reliability of the survey questionnaires in Phases 1, 2 and 3 was assessed to ensure *stability* and *internal consistency*.

3.11.2.1 Stability reliability

Stability reliability, also referred to as *test-retest reliability* (repeating the same test to the same subjects on a second occasion), was used for this study (Pallant 2020:6; Rubin 2020:99; Cairns 2019:177). Test-retest reliability implies that when the questionnaire is administered to a sample twice on two different occasions, and the scores are then compared to measure the consistency or stability among scores or data collected it would yield similar results (Polit & Beck 2021:317; Pallant 2020:6; Rubin 2020:99; Beins 2017:130). The rationale for adopting test-retest reliability was to
ensure that the survey questionnaire consistently replicates the result more than once in the same situation and population.

To measure the test-retest reliability of questionnaires, a reliability coefficient is computed and designated as "correlation coefficient r," ranging from 0.00 to 1.00 (Polit & Beck 2021:316). To deem a questionnaire stable, the acceptable correlation between two sets of responses to the instruments should be higher than .70, considered adequate, or above .80, considered for preferable test-rest-reliability coefficients (Polit & Beck 2021:319; Rubin 2020:99; Shougaard, de Thurah, Bech, Hjollund & Christiansen 2018:2). A higher test-retest-reliability coefficient indicates a smaller measurement error of the questionnaire (Polit & Beck 2021:319; Shougaard et al. 2018:2; Leppink & Pérez-Fuster 2017:159).

In this study, the researcher and the supervisor ensured all three questionnaires were tested through test-retest for reliability to indicate whether the same results were obtained with the repeated administering of the same data collection methods using similar study respondents. All three questionnaires were re-tested after two weeks (see Section 3.10) by the original respondents, as recommended by Polit and Beck 2021:319 and Rubin (2020:100). The statistician helped to estimate the test-retest re-liability of each questionnaire by correlating the scores obtained from two successive measurement procedures for the same group of pre-testing individuals at two different times using the R Statistical Software (R) for data analysis (Rubin 2020:100; Gravetter & Forzano 2018:63). The test-retest reliability results of the three questionnaires are discussed below.

3.11.2.1 Internal Consistency

Internal consistency or homogeneity refers to the degree to which multiple items on a single scale measure the same attribute or dimension/trait (Polit & Beck 2021:320; Rubin 2020:100). To determine the internal consistency, Cronbach's Alpha (α) coefficient is calculated (Polit & Beck 2021:321). The Cronbach's alpha (α) value ranges from 0-1, where 0 is no internal consistency and 1 is the maximum internal consistency

(Taber 2017:1287; Tiko 2015:18). Cronbach's alpha (α) quantifies the degree to which items on an instrument are correlated with one another and the scale used in a clinical setting should have a minimum $\alpha = 0.90$, while as low as $\alpha = 0.70$ is regarded as a satisfactory level as stated by Information Resources Management Association (2021:1090) and Blaga (2020:122)

In this study, the data analysis program, the R Statistical Software (R), was used to determine the internal consistency of the questionnaires in Phases 1, 2, and 3. The statistician did data analysis. The items of the questionnaires were analysed by calculating Cronbach's Alpha (α) coefficient test as recommended by Polit and Beck (2021:320), and Grove, Gray and Faan (2019:271). The internal consistency of the three developed questionnaires in Phases 1, 2, and 3 was calculated based on the Cronbach's alpha (α) coefficient mentioned above.

In the end, the results of *test-retest reliability and internal consistency* computed during pre-testing for the three questionnaires were as follows:

- Questionnaire 1: The Cronbach's alpha coefficient score the first time was 0.927; at time two, it was 0.933. The high alphas were indicative of greater internal consistency, an excessive number of questions, or possible redundant questions as they were greater than 0.9 (Ngulube 2021:304; Gorrie, Goodall, Rush & Ravenscroft 2019:6). Therefore, modifications of words and phrases were made based on comments from the respondents, the supervisor, and the statistician's feedback (see Table 3.5).
- Questionnaire 2: Cronbach's alpha coefficient score of Questionnaire 2 at time one was 0.945, and at time two it was 0.945. The high alphas were indicative of greater internal consistency, a large number of questions, or possibly redundant questions as the score was greater than 0.9 (Ngulube 2021:304; Gorrie et al. 2019:6). Despite these very high coefficient scores, modifications of words and phrases were made as recommended by the respondents, the supervisor, and the statistician's feedback (see Table 3.6).

 Questionnaire 3: The first time, the Cronbach score was 0.934, and the second time it was 0.977. Both also showed an internal consistency of more than 0.9. Some words and phrases were modified as recommended by the respondents and the supervisor, and based on the statistician's feedback (see Table 3.7).

The analysis did provide information on which items needed rewording or even needed removal according to Cronbach's alpha coefficient scores as suggested by Ngulube (2021:304). After all necessary modifications were made to the questionnaires, they were finally administered for data collection (see Tables 3.5, 3.6, and 3.7).

3.12 ETHICAL CONSIDERATIONS

Ethics in research is defined as a system of moral values that is concerned with the degree to which the research procedures adhere to professional, legal, and social obligations to the study participants (Polit & Beck 2014:380; Profetto-McGrath, Polit & Beck 2010:409).

The researcher considered the ethical principles and adhered to three main domains of responsibilities related to ethical principles: (1) protecting the rights of the study participants, (2) protecting the rights of the institution, and (3) scientific integrity of the research (Polit & Beck 2021:133).

3.12.1 Protecting the rights of the study participants

To adhere to ethical standards of research and comply with the King Abdullah International Medical Research Centre's (KAIMRC) institutional review board's (IRB) policy and guideline requirements, the researcher protected the rights and welfare of study participants by ensuring the following: (1) respect for human dignity and self-determination; (2) assuring informed consent; (3) maintaining confidentiality and anonymity; and (4) protecting participants from harm (Kazdin 2021:426; Polit & Beck 2021:133).

• Respect for human dignity and self-determination

The information leaflet (see Annexures 3 and 9) informed the participants that they could withdraw from the research at any time without fear of penalty. This was in line with Kazdin (2021:426), Polit and Beck (2021:134), and Leavy (2017:35), who state that the researcher must respect and treat participants as autonomous agents by informing them about a proposed study and allowing them to choose to participate voluntarily or not.

Informed consent

Informed consent is defined as "the provision of information to participants about the purpose of the research, its procedures, potential risks, benefits, and alternatives, so that the individual understands this information and can make a voluntary decision whether to control and continue to participate" (Liamputtong 2013:39). To influence voluntary willingness and participation in the study, an information leaflet and consent letter (see Annexures 3 and 9) were provided to ensure that the participants were aware of all information relevant to the study (Kazdin 2021:426; Polit & Beck 2021:137; Arifin 2018:30). The participants were informed about their rights to withdraw from the study without penalty.

The participants were asked to give informed consent voluntarily before participating in the study (Polit & Beck 2021:137; Arifin 2018:30; Leavy 2017:36).

• Confidentiality and anonymity

Respecting the respondents' right to privacy in this study involved protecting their anonymity and keeping their information confidential (Kazdin 2021:426; Polit & Beck 2021:142). Confidentiality means a trustful action of the researcher to safely manage information or data shared by participants by ensuring that the data are kept private from others (Grove & Gray 2023:493; Polit & Beck 2021:141). All information and data collected were portrayed confidentially. Confidentiality was maintained and assured to all respondents in all study phases by preventing anyone outside the study from accessing the information and data other than the researcher. Confidentiality is linked to anonymity. Anonymity may be defined as an aspect of privacy to protect the research participants by keeping them unknown or unspecified to others (Grove & Gray 2023:493; Polit & Beck 2021: 141).

The self-administered questionnaire kept respondents anonymous as no names or personal details were requested on the questionnaire. The judgements and opinions of the panel members were kept confidential as recommended by Harris and Mavuka (2023:191), and Polit and Beck (2021:141). The researcher did not share the names and details of the panellists with any of the panellists involved. The findings of the study were reported without identifying study participants.

• Protection from harm

The researcher protected the participants from discomfort and harm by ensuring a comfortable environment around them (Grove and Gray 2023:493; Polit & Beck 2021:133).

Researcher-Participant Relationship

Privacy was maintained throughout the study. Study participants entered into a special trusting relationship with the researcher, and it was crucial that this relationship not be exploited (Harris & Mavuka 2023:191; Polit & Beck 2021:135).

3.12.2 Protecting the rights of the institution

Permission to conduct the research was first sought from the Research Ethics Committee of the Department of Health Studies, Unisa REC-012714-039 (see Annexure 6a), REC-240816-052 (see Annexure 6b); and Nursing Services Permission to conduct research (see Annexure 7a); KAIMRC and the Institutional Review Boards (IRB), approval study number SP 18/036/R (see Annexure 7b); IRB Annual Extension SP 18/036/R (see Annexure 7c); and IRB 6 Months Extension SP 18/036/R (see Annexure 7d) of the two selected Saudi Arabian teaching hospitals as recommended by Pozgar (2019:82), Salhan (2011:30) and DePoy and Gitlin (2011:149). Care was taken not to infringe on the institutions' rights by scheduling meetings to suit respondents at three free times to avoid disturbing their routine patient care and allow for as much attendance as possible.

3.12.3 Scientific integrity of the research

The researcher maintained honesty by avoiding duplication of work and misconduct such as fabrication, falsification, dishonesty, and plagiarism. Halstead and Billings 2023:93, Polit and Beck (2021:147), and Iphofen (2017:97) emphasise that the researcher is responsible for monitoring the integrity and honesty of his or her research protocols, results and publications. There was voluntary participation with no compensation. The data were honestly reported, and raw data were kept safely in a locked place accessible only to the researcher.

3.13 DATA COLLECTION: PROCESS FOLLOWED IN PHASES 1, 2 AND 3

Data were collected between 4 December 2019 and 17 March 2020 (refer to Section 3.8). A gatekeeper letter, as well as the information letter for respondents (see Annexures 1 and 2), were sent via email to the nurse managers who acted as gatekeepers of the five nursing care divisions, namely medical, surgical, paediatric, cardiac and oby-gynae wards to assist with the sampling to ensure that the volunteers comply with the inclusion criteria of the study. The managers then provided the researcher with a list of names and telephone numbers of professional registered nurses and clinical facilitators who volunteered to participate after they had read the information letter that was forwarded to them by their nurse managers (see Annexures 1, 2 and 3).

The researcher telephonically contacted the selected volunteers to arrange a convenient time and place to deliver the information letter, the consent letter (see Annexure 3), and the questionnaire and envelope that could be sealed. This was to avoid disturbing the respondents' daily routine by scheduling the meetings to suit them. The volunteers were allowed to read the information letter again and then required to sign the consent form if they agreed to participate. The completed questionnaires, in sealed envelopes, were put back in the sealed box provided in each ward (see Annexure 7). The researcher collected them from each unit at the end of a week, after working hours, between 4 and 6 pm. This was done until all respondents' questionnaires were collected.

Data collected were analysed after each phase.

3.14 DATA ANALYSIS

After data collection, the statistician used Statistical Package for the Social Sciences (SPSS) version 25 to enter the data for analysis. The researcher and the statistician verified the completeness of the data through data cleaning to resolve quality issues that could have been found in the data, as described by Karishree, Sudha, Soundarya and Ramya (2023:10) and Van der Loo and De Jong (2018:183). The closed-ended questions were analysed using descriptive statistical analysis showing frequencies, mean, and standard deviations. At the end, each questionnaire had open-ended questions for qualitative enhancement. The open-ended responses were coded openly by the researcher and analysed thematically to draw valid inferences, as described by Anderson, Taylor, Taylor and Virues-Ortega (2022:742).

3.14.1 Descriptive statistical analysis

Statistics, as a branch of mathematics, are used for data collection, analysis, and interpretation of data, and are classified mainly as descriptive or inferential statistics and are used to summarise, organise and describe a sample of data to understand the fundamental properties of the variables being studied (Polit & Beck 2021:366; Sharma 2022:16).

Descriptive statistics measures are categorised into (1) frequency distributions; (2) measures of central tendency such as means; (3) measures of dispersion such as standard deviations; and (4) correlation coefficients, cross-tabulations and standard-ized scores (Sharma 2022:291; Leavy 2017:111; Leedy & Ormrod 2015:386; Rubin & Babbie 2014:662).

In this study, the descriptive statistical analyses used were: (1) the frequency distributions, (2) the mean, (3) the standard deviations [SD], and (4) percentage values related to the relevant objectives of each quantitative phase to (1) Identify and describe the resources available to conduct pain assessments (Phase 1, quantitative); (2) Identify and describe nurses' characteristics and learning styles that enhance the transfer of pain management competencies (Phase 2, quantitative); (3) Explore the teaching approaches employed by the clinical facilitators during pain management education of nurses (Phase 3, quantitative); (4) Describe the learning content regarding pain assessment and management (Phase 3, quantitative); and (5) Describe the transfer of learning climate within the hospitals' nursing care areas (Phase 3, quantitative) (Creswell & Clark 2011:206, Cohen et al. 2011:622).

1) Frequency distributions

Frequency distributions are statistical procedures that involve systematic presentations of values of variables displayed on graphs or tables from lowest to highest, together with a report about the count of a number or percentage of how many times each value occurred (Polit & Beck 2021: 368). In this study, the frequency distributions are presented in graphic and table format for the biographical information of the respondents (see Chapter 4).

2) The mean

The mean is a measure of central tendency that uses a single value to represent the arithmetic average of all the scores to represent a distribution of values of the variables in the data set (Polit & Beck 2021:371; Leavy 2017:111). To calculate the mean, the number of all scores in the data set are added together to make the total sum of scores, and then the total sum of scores is divided by the number of scores being added (Polit & Beck 2021:371; Weir & Vincent 2020:41). The mean is symbolised as *M* or \bar{x} (Polit & Beck 2021:841), as is the case in the description of the research findings (see Chapter 4).

3) Standard deviations [SDs]

The standard deviations [SDs] are measures of dispersion or variation that summarise the average deviation of values from the data set's mean (Kaplan & McCune 2018:217; Polit & Beck 2014:219; Rubin & Babbie 2014:672). "It indicates how values vary about the mean of the distribution" (Polit & Beck 2021:372). The standard deviations were calculated around the mean of the data set collected in the three quantitative phases of the study.

4) The percentage values

Descriptive statistics included percentages for categorical ordinal and nominal data that form part of each frequency (Polit & Beck 2021:366; Taylor 2017:335). Percentage values are calculated by dividing the frequency by the total number of eligible counts with the sample (Taylor 2017:335). Percentage values are included in all frequency distributions in Phases 1, 2, and 3 of this study (see Chapter 4).

Even though Phases 1, 2, and 3 were quantitative, at the end of the last section of each questionnaire, there was a non-compulsory open-ended question for qualitative enhancement. Thus, the data were open-coded and followed the qualitative thematic content analysis steps.

3.14.2 Qualitative data analysis (qualitative enhancement)

Qualitative data analysis is defined as the formal interpretation of collected data to create order and meaning by exploring the data by coding, indexing, retrieving or sorting to search for patterns, themes, relationships and significant statements (Singh & Ramdeo 2020:238).

All open-ended questions in the three questionnaires used in Phases 1-3 were opencoded, and a thematic analysis was done to enhance the quantitative data obtained. Content analysis can be defined as a systematic analytical method that statistically analyses the narrative data by identifying prominent patterns, themes, or biases (Singh & Ramdeo 2020:238; Leavy 2017:146; Leedy & Ormrod 2015:275).

There are different step-by-step methods described to do a content analysis (Singh & Ramdeo 2020:237; Vaismoradi 2013:400). A thematic content analysis according to the steps described by Clarke and Braun (2013:121) and Maguire and Delahunt (2017:3354) was followed to analyse the data. These steps include (1) "familiarising oneself with the data, (2) generating initial codes, (3) searching for themes, (4) reviewing themes, (5) defining and naming the themes, and (6) producing the report."

1) Step 1: Familiarisation with the data

The written narrative comments from all open-ended questions were repeatedly read until the researcher became immersed and familiar with the entire content of the data sets (Braun & Clarke 2006:16). Relevant and similar ideas were highlighted in different colours before starting with the coding process as described in the literature (Maguire & Delahunt 2017:3355; Clarke & Braun 2013:121).

2) Step 2: Coding

Similar ideas were grouped and coded.

3) Step 3: Searching for themes

The identified codes were grouped to ensure that similar ideas were together. Similar ideas, therefore, formed the themes. The selected themes were further refined into themes specific to the research questions.

4) Step 4: Reviewing themes

The identified themes were formed, refined, and then reduced into manageable themes. Those themes that were not fitting were discarded and refined until additional refinements were not adding anything substantial.

5) Step 5: Defining and naming themes

The final step identified the categories grouped to form the themes. The direct quotations that led to the categories were included in the table to present the data (see details in Sections 4.2.3, 4.3.4, and 4.4.6).

6) Step 6: Writing up

This is the analysed data report, which is presented to answer the research questions and the objectives of the study at the end of each phase.

3.15 CONCLUSION

This chapter described the research design and methodology of the study. Phases 1, 2, and 3 were used to gather the information from the respondents.

Chapter 4 will discuss the data analysis and interpretation of findings from Phases 1 to 3.

CHAPTER FOUR: PHASE 1, 2, AND 3: DATA ANALYSIS AND INTERPRETATION

4.1 INTRODUCTION

This chapter presents the data analysis and interpretation of Phases 1, 2, and 3. Each phase will be discussed separately as the findings were aimed at addressing the objectives of each phase, and sequentially, the phases inform each other, as illustrated in Table 4.1.

| Organisatio | n and structure of | the study |
|-------------------|---|--|
| Chapter number | Chapter out- line | Chapter content |
| Chapter 1 | Overview of the study | Contains the introduction, background of the study, the prob- lem statement, research purpose and objectives, research question, theoretical framework, key theoretical and opera- tional concepts, the research design and methodology, and ethical considerations. |
| Chapter 2 | Literature re- view | Consists of the literature review related to: Systemic Model of Transfer of Learning by Donovan and Darcy, Transfer of learning and Pain management and tools. |
| Chapter 3 | Research de- sign and meth- odology | Illuminates the overarching research design. Phase 1, 2 and 3 (quantitative phases): Methodology and data gathering. |
| Chapter 4 | Data analysis and interpreta- tion | Presents the data analysis and interpretation of the findings from Phases 1 to 3. |

Table 4.1 Organisation and structure of the study

| Chapter 5 | Phase 4 | Includes a description of Phase 4 of the study: Literature review on the action plan development. Development of the draft action plan. |
|-----------|--|--|
| Chapter 6 | Phase 5 | Outlines and describes Phase 5 of the study (qualitative phase): Methodology Validation of the action plan: The validated action plan. |
| Chapter 7 | Conclusion, recommenda- tions, and limi- tations. | Contains the conclusion, recommendations, and limitations of the study. |

All but one respondent completed both Questionnaire 1 for Phase 1 and Questionnaire 2 for Phase 2. The one respondent completed only Questionnaire 1 for reasons unknown. Thus, 385 completed Questionnaire 1 and 384 completed Questionnaire 2.

In the descriptions of the findings discussed, the following common symbols for the presentation of data apply:

- N = Upper case N represents the total sample included, as suggested by Lynch (2013: xi) and Beins (2012:76).
- n = refers to the size of subsamples (Beins 2012:76).
- F = refers to frequencies within the sample set under discussion and can refer to either N or n (Beins 2012:76).
- f = Lower italicised f refers to percentages.

4.2 PHASE 1

The objective of Phase 1 was to identify and describe the resources available to conduct pain assessments in the context of Saudi Arabian teaching hospitals.

4.2.1 Demographic characteristics (N = 385)

The demographic characteristics of the respondents (N = 385) are described according to gender, age, nationality, highest education qualification and nursing care areas.

4.2.1.1 Gender of nurses (N = 385)

As illustrated in Figure 4.1, 7.0% (n = 27) of nurse respondents were male and 93.0% (n = 358) were female. Nursing is globally found to be a female-dominated profession where male nurses are underrepresented (Badu, Abalo, Bam, Agyemang, Noi, Badu and Peprah (2019:9). According to the WHO (2020), Saudi Arabia has 79% female nurses and 21% male nurses, with a global distribution of 89% female and 11% male nurses (WHO 2020:41). The dominance of female respondents is therefore acceptable within this study.



Figure 4.1 Gender of nurses (N = 385)

4.2.1.2 Age of nurses (N = 385)

The mean age of respondents was 34.86 years. The youngest respondent was 24 years old, and the eldest was 60 at the time of data gathering. The standard deviation was 7.725, and the distribution of registered nurses' ages was a positively skewed attribute, as the mean age was 34.86, thus greater than the median age of 33 (see Figure 4.2). It was appropriate to inquire about the age of nurses in the context of inquiring about the experience and skill mix that enhance the transfer of learning after pain management training or sessions. Globally, young professional nurses are under 35 years, while older professional nurses are aged 55 or above (WHO 2020:40). This suggests a positive outcome as the respondents' ages show that there was a high proportion of the young nursing workforce with fewer ageing nurses in both the study hospitals.



Figure 4.2 Age of nurses (N = 385)

4.2.1.3 Nationality of nurses (N = 385)

The nationality of the 385 respondents who participated in this study is depicted in Table 4.2.

| Nationality | n = | <i>f</i> = % |
|---------------|-----|--------------|
| Filipino | 208 | 54.0 |
| Malaysian | 131 | 34.0 |
| Saudi | 19 | 4.9 |
| South Korean | 8 | 2.1 |
| Indian | 6 | 1.6 |
| South African | 5 | 1.3 |
| British | 2 | 0.5 |
| Jordanian | 2 | 0.5 |
| Slovak | 2 | 0.5 |
| Czech | 1 | 0.3 |
| Spanish | 1 | 0.3 |
| Total | 385 | 100 |

Table 4.2: Nurses' nationality (N = 385)

The respondents' nationality was important to know due to their diverse education and possibly their knowledge and experience about the resources needed to do pain assessment. As illustrated in Table 4.2, nurses of eleven (11) different nationalities participated in the study.

The majority of the respondents, namely 54.0% (n = 208), were from the Philippines, 34% were Malaysians (n = 131), and 4.9% (n = 19) were from Saudi Arabia. Five (f = 1,3%) South Africans participated, with two (f = 0.5%) participants each from the United Kingdom, Jordania and Slovakia. Only one (f = 0,3%) Czech and one (f = 0,3%) Spanish nurse participated. Of interest is that the vast majority of respondents

(n = 366; f = 95.1%) were from outside Saudi Arabia. The study findings had an even greater representation of foreign participants than those mentioned by Al Yami and Watson (2014:11), namely 66% foreign and 34% Saudi nurses working in Saudi healthcare facilities.

It is mandatory for nurses working in Saudi Arabian hospitals, despite their nationality, to register with the Saudi Commission for Health Specialities (SCFHS) to be licenced to practice. To renew their licences to practice with the SCFHS, they were expected to engage in continuing professional development (CPD) activities that are accredited. One such programme is the pain management training programme that can be done to meet re-registration requirements and practical competencies (Aljohani 2020:2; Alkhazim & Althubaiti 2016:18).

Due to the diverse nationalities of participants, but also that of the patients, the language used during pain assessments could have been a challenge to the pain management outcome, as was mentioned in the study findings of a study conducted by Van Rosse, de Bruijne, Suurmond, Essink-Bot and Wagner (2015:9).

4.2.1.4 Highest education qualifications (N = 385)

Nurses undergo different types of education that can affect their competencies within nursing practice. The highest academic qualifications of participants are important as they may contribute to motivating participation in continuing nursing education, with participants having different levels of skills and knowledge to contribute to quality patient care, different problem-solving skills, and have trained to function in diverse environments (Suliman & Aljezawi 2018:529; Kamariannaki, Alikari, Sachlas, Stathoulis, Fredelos & Zyga 2016:233). The highest education qualifications of 385 respondents are illustrated in Figure 4.3.



Figure 4.3 Nurses' Highest Education Qualifications distribution (N = 385)

Two hundred and ninety (290) out of 385 (f = 75.5% n = 290) respondents had a bachelor's degree, 91 (f = 23.6%; n = 91) a diploma, with only 4 (f = 1%, n = 4) having obtained a master's degree (refer to Figure 4.3). These findings are consistent with the global nursing community and national trends, which show that various countries have programs on the rise with baccalaureate degree-prepared nurses for education as an entry into the nursing profession. These countries are, among others, the United States, the United Kingdom, Denmark, Norway, Ireland, Australia, New Zealand, Canada, Philippines, and Mexico (Institute of Medicine (IOM) 2011:565). In Saudi Arabia, the Saudi Arabian Ministry of Health (MoH) and SCFHS have stipulated the minimum requirement of a bachelor's degree for entry into nursing practice (Alboliteeh, Magarey & Wiechula 2018:76). Table 4.3 illustrates the highest educational qualification of the participants as per the country of origin.

| Nationality | Total nu | mber | Master's | s degree | Bachelor's d | egree | Diploma | |
|---------------|----------|--------------|----------|--------------|--------------|--------------|---------|--------------|
| rtationality | n = | <i>f</i> = % | n = | <i>f</i> = % | n = | <i>f</i> = % | n = | <i>f</i> = % |
| Filipino | 208 | 54.0 | 0 | 0 | 208 | 54.0 | 0 | 0 |
| Malaysian | 131 | 34.0 | 2 | 0.5 | 41 | 11.2 | 88 | 22.8 |
| Saudi | 19 | 4.9 | 0 | 0 | 19 | 4.9 | 0 | 0 |
| South Korean | 8 | 2.1 | 0 | 0 | 8 | 2.1 | 0 | 0 |
| Indian | 6 1.6 | | 0 | 0 | 6 | 1.6 | 0 | 0 |
| South African | 5 | 1.3 | 0 | 0 | 2 | 0.5 | 3 | 0.8 |
| British | 2 | 0.5 | 0 | 0 | 2 | 0.5 | 0 | 0 |
| Jordanian | 2 | 0.5 | 0 | 0 | 2 | 0.5 | 0 | 0 |
| Slovak | 2 | 0.5 | 2 | 0.5 | 0 | 0 | 0 | 0 |
| Czech | 1 | 0.3 | 0 | 0 | 1 | 0.3 | 0 | 0 |
| Spanish | 1 | 0.3 | 0 | 0 | 1 | 0.3 | 0 | 0 |
| Total | 385 | 100.0 | 4 | 1.0 | 290 | 75.3 | 91 | 23.6 |

Table 4.3:The nurses' country of origin and their highest education qualifica-
tion (N = 385)

The only respondents with master's degrees were two from Malaysia (f = 0.5%) and two from Slovakia (f = 0.5%). All respondents from the Philippines had a bachelor's degree (f = 54.0%; n = 208). Respondents from Malaysia (f = 11.2%; n = 41), Saudi Arabia (f = 4.9%; n = 19), South Korea (f = 2.1%; n = 8), India (f = 1.6%; n = 6), as well as 2 (f = 0.5%) participants each from Britain, Jordan, and South Africa, had a bachelor's degree. Only 1 (f = 0.3%) each from the Czech Republic and Spain had a bachelor's degree. The only diploma-prepared nurses were from Malaysia (f = 22.8; n = 88) and South Africa (f = 0.8%; n = 3). The study findings were consistent with those of Roets, Botma, and Grobler (2016:428), who stated that most nurses were diplomaprepared in some countries, such as South Africa. Degree-prepared nurses, however, are mostly equipped with high levels of clinical reasoning skills and can manage patients with acute or chronic pain, render quality care and improve patient safety (Deng 2016:97; Roets et al. 2016:429).

4.2.1.5 Nursing wards/divisions where the registered respondents worked (N = 385)

It was important to know the nursing wards of the five nursing care divisions where the respondents worked to access the available resources within the different wards, which could enable nurses to conduct pain assessments. One hundred and sixty-three (f = 42.3%) respondents were from surgical wards, 122 (f = 31.7%) from medical wards, 45 (f = 11,7%) from paediatric wards, 32 (f = 8.3%) from cardiac wards, while 23 (f = 6.0%) were from obs-gynae wards (refer to Table 4.4). A high number of respondents in this study were from surgical and medical wards, where pain management is crucial and where nurses value the management thereof, as suggested by Alzghoul and Abdullah (2016:157). Nurses from surgical and medical nursing divisions, therefore, emphasized the high availability of resources to conduct pain assessments in their wards.

| Nursing Wards | n = | <i>f</i> = % |
|---------------|-----|--------------|
| Surgical | 163 | 42.3 |
| Medical | 122 | 31.7 |
| Paediatric | 45 | 11.7 |
| Cardiac | 32 | 8.3 |
| Obs-gynae | 23 | 6.0 |
| Total | 385 | 100.0 |

| Table 4.4: Distribution o | respondents within | wards (N = 385) |
|---------------------------|--------------------|-----------------|
|---------------------------|--------------------|-----------------|

4.2.2 Resources available to conduct pain assessment (N = 385)

This section indicates the findings from 385 nurse respondents about Section B of the questionnaires that focussed on the resources available to conduct pain assessments (see Annexure 4). It was important to identify the availability of resources to conduct pain assessments to enhance the transfer of learning of pain management competencies within various wards.

The various systematic pain assessment guide tools for pain history taking were assessed for availability as resources to conduct pain assessments.

4.2.2.1 Systematic pain assessment guides for pain history taking. (N = 385)

Five systematic pain assessment guides used for pain history taking were selected for inclusion in this study, namely: (1) QUEST (question the child, use pain rating tools, evaluate behaviour, sensitise parents and take action); (2) WILDA (words to describe pain, intensity, location, and aggravating or alleviating factors); (3) PQRST (provok-ing/palliation factors, quality of pain, region of pain, severity, and timing); (4) OPQRST (onset of pain, provoking/palliating, quality, region/radiation of pain, the severity of pain, timing/treatment, understanding/impact on you and values); (5) and COLDSPA (character, onset, location, duration, severity, pattern, and associated factors). Table 4.5 illustrates the findings from the 385 respondents regarding the systematic pain assessment guide tools available in their wards. The tools are discussed from the highest to the least available.

4.2.2.1.1 Availability of the WILDA pain assessment guide (N = 385)

As illustrated in Table 4.5, WILDA was indicated as the tool most accessible within the various wards, with 97,4 % (n = 375; N = 385) of respondents indicating its availability. The second most accessible tool, according to the respondents, was PQRST with 69.1% (n = 266; N = 385) availability, followed by OPQRSTUV with 41.9% (n = 161; N = 384), COLDSPA with 41.8% (n = 161; N = 385), and QUEST was the least

available with only 9.4% (n = 36; N = 385) (see Table 4.5). The WILDA tool was reported to be predominantly accessible, as illustrated in Table 4.5.

| | Nursi | ing Wa | rds | | | | | | | | | | | | | | | | | | | | | |
|----------------------------|-------|--------|-----|------|-------|------|-----|------|-------|---------|-----|------|-----|------|-----|------|-------|--------|-----|------|-------|------|-----|------|
| Systemic | Surgi | cal | | | Medio | cal | | | Pae | diatric | | | Car | diac | | | Obs | - gyna | е | | тоти | AL. | | |
| pain assess- ment guide | n = 1 | 63 | | | n = 1 | 22 | | | n = - | 45 | | | n = | 32 | | | n = 2 | 23 | | | N = 3 | 85 | | |
| toois | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | |
| | F = | % = | F = | % = | F = | % | F = | % = | F = | % = | F = | % = | F= | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = |
| WILDA | 158 | 96,9 | 5 | 3,1 | 122 | 100 | 0 | 0 | 44 | 97,8 | 1 | 2,2 | 32 | 100 | 0 | 0 | 19 | 82,6 | 4 | 17,4 | 375 | 97,4 | 10 | 2,6 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| PQRST | 113 | 69,3 | 50 | 30,7 | 78 | 63,9 | 44 | 36,1 | 44 | 97,8 | 1 | 2,2 | 18 | 56,3 | 14 | 43,8 | 13 | 56,5 | 10 | 43,5 | 266 | 69,1 | 119 | 30,9 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| OPQRSTUV | 61 | 37,4 | 101 | 62,0 | 54 | 44,3 | 68 | 55,7 | 22 | 48,9 | 23 | 51,1 | 13 | 40,6 | 19 | 59,4 | 11 | 47,8 | 12 | 52,2 | 161 | 41,9 | 223 | 58,1 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| COLDSPA | 60 | 36,8 | 103 | 63,2 | 59 | 48,4 | 63 | 51,6 | 17 | 37,8 | 28 | 62,2 | 13 | 40,6 | 19 | 59,4 | 12 | 52,2 | 11 | 47,8 | 161 | 41,8 | 36 | 9,4 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| QUEST | 22 | 13,5 | 141 | 86,5 | 1 | 0,8 | 121 | 99,2 | 11 | 24,4 | 34 | 75,6 | 1 | 3,1 | 31 | 96,9 | 1 | 4,3 | 22 | 95,7 | 36 | 9,4 | 349 | 90,6 |

Table 4.5: Availability of the pain assessment guides (N = 385)

In both the medical and cardiac wards, the WILDA tool was reported to have been 100% available to all respondents. Respondents from paediatric wards indicated WILDA to be 97,8% (n = 45; F = 44) available. In surgical wards, it was 96.9% (n = 163; F = 158) and in obs-gynae wards, it was 82,6% (n = 23; F = 19) (see Table 4.5). As Fink and Gallagher (2019:231) suggest, WILDA assists the nurses in asking questions during pain assessment to persons who can self-report their pain; thus, its availability in the wards that are mainly for adults.

4.2.2.1.2 Availability of the PQRST pain assessment guide (N = 385

PQRST, used to conduct pain assessment in the paediatric population, as explained by Fink and Gallagher (2019:231), was primarily accessible within paediatric wards, namely a 97.8% (F = 44; n = 45) availability. This tool was reported to be 69,3% (n = 113; F = 163) available in surgical wards, 63,9% (F = 78; n = 122) in medical wards, 56,5% (n = 13; F = 23) in obs-gynae wards, and 56,3% (F = 18; n = 32) in cardiac wards (see Table 4.5).

4.2.2.1.3 Availability of the OPQRSTUV pain assessment guide (N = 385)

As explained by Cash, Glass, Fraser, Corcoran and Edwards (2019:59), as well as Jufri, Permana and Widagdo (2019:13), the OPQRSTUV tool is used to guide the assessor to conduct pain assessment within paediatric wards. Thus, it was primarily available in paediatric wards with a 48.9% (F = 22; n = 45) availability. In obs-gynae wards, the tool was reported available with 47,85 (F = 11; n = 23), in medical wards (44,3%; F = 54; n = 122), in cardiac wards (40,6%; F = 13; n = 32), and surgical wards with 37,4% (F = 61; n = 163) (see Table 4.5).

4.2.2.1.4 Availability of the COLDSPA pain assessment guide (N = 385)

Dains, Baumann, and Scheibel (2015:1) recommend COLDSPA as a useful tool, using mnemonics to gather information for the process of pain assessment. Respondents from obs-gynae wards reported the tool to be 52.2% (F = 12; n = 23) available, 48,4% (F = 59; n = 122) from medical wards, 40,6% (F = 13; n = 32) from cardiac wards,

37,8% (F = 17; n = 45) from paediatric wards, and 36,8% (F = 60; n = 163) from surgical wards reported the tool to be available (see Table 4.5).

4.2.2.1.5 Availability of the QUEST pain assessment guide (N = 385)

QUEST, used to conduct pain assessment in paediatric areas as stated by Brand (2019:314) and Gladston et al. (2016:55), was primarily available within paediatric wards with 24.4% (F = 11; n = 45) availability. The tool has been reported available with a 13,5% (F = 22; n = 163) response in surgical wards, 4,3% (F = 1; n = 23) in obsgynae wards, 3.1% (F = 1; n = 32) in cardiac wards, and 0,8% (F = 1; n = 122) in medical wards (see Table 4.5).

The pain rating assessment tools available to assess the pain of patients who can selfreport their pain were also assessed for their availability in the various nursing wards.

4.2.2.2 Pain rating assessment tools for patients who can self-report their pain (N = 385)

The respondents indicated the availability of the following five chosen validated and reliable pain rating assessment tools for patients who can self-report their pain to rate pain during pain assessments: (1) the Wong-Baker FACES pain scale; (2) the Numeric Rating Scale (NRS); (3) the Verbal Analogue Scale (VAS); (4) the Verbal Descriptor Scale (VAS); and (5) the Brief Pain Inventory (BPI) questionnaire. The availability of these tools in the different wards, as indicated by the 385 respondents, is illustrated in Table 4.6. The pain rating assessment tools are discussed from the highest availability to the least available.

The numeric rating scale (NRS) was indicated as the pain rating scale predominantly available within the five nursing divisions with a 95,3% (n = 367; N = 385) availability as was indicated by respondents (see Table 4.6). Wong-Baker FACES followed with 82,3% (n = 316; F = 384) availability, VAS with 37,9% (n = 146; N = 385), VDS with 35,6% (n = 137; N = 385), and lastly BPI with 17,1% (n = 66; N = 385) availability (see Table 4.6). This finding is supported by the study findings of Fischer, Hosie, Luckett,

Agar and Phillips (2019:492), who report in their scoping review that clinicians used NRS on patients' self-report of pain due to availability thereof.

The NRS for pain rating was reported as available in various wards, as portrayed in Table 4.6.

4.2.2.2.1 Availability of the Numeric Rating Scale (NRS) (N = 385)

In paediatric and cardiac wards, the Numeric Rating Scale (NRS) was reported to have been 100% available to all respondents. This finding is interesting as Lim et al. (2015:230) indicate that NRS is very useful as a self-reporting assessment tool for patients who can self-report their pain (see Table 4.6), implying that it is expected not to be available in paediatric wards. Respondents from medical wards reported NRS to be 95,5% (n = 122; F = 117) available, followed by surgical wards with 95,1% (F = 155; n = 163), and oby-gynae wards 78,3% (F = 18; n = 23) (see Table 4.6).

| | Nurs | sing V | /ards | | | | | | | | | | | | | | | | | | | | | |
|--|-------|--------|-------|------|-----|-------|--------|------|--------|---------|--------|------|--------|-------|--------|------|--------|--------|--------|------|-----|------|-----|------|
| Pain rating assessment tools for | Surç | gical | | | Mec | lical | | | Pae | ediatri | с | | Ca | rdiac | | | Ob | s- gyn | ae | | тот | AL | | |
| pain | n = ' | 163 | | | n = | 122 | | | n = | 45 | | | n = | 32 | | | n = | 23 | | | N = | 385 | | |
| | Yes | | No | | Yes | | No | | Ye | 3 | No | | Ye | 3 | No | | Yes | 6 | No | | Yes | | No | |
| | F = | % = | F = | % = | F = | % | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = |
| The Numeric Rating Scale (NRS) | 155 | 95,1 | 8 | 4,9 | 117 | 95,9 | 5 | 4,1 | 45 | 100 | 0 | 0 | 32 | 100 | 0 | 0 | 18 | 78,3 | 5 | 21,7 | 367 | 95,3 | 18 | 4,7 |
| The Wong-Baker FACES pain scale | 140 | 85,9 | 23 | 14,1 | 85 | 69,7 | 37 | 30,3 | 40 | 90,9 | 4 | 9,1 | 31 | 96,9 | 1 | 3,1 | 20 | 87,0 | 3 | 13,0 | 316 | 82,3 | 68 | 17,7 |
| The Verbal Analogue Scale (VAS) | 52 | 31,9 | 111 | 68,1 | 60 | 49,2 | 62 | 50,8 | 13 | 28,9 | 32 | 71,1 | 14 | 43,8 | 18 | 56,3 | 7 | 30,4 | 16 | 69,6 | 146 | 37,9 | 239 | 62,1 |
| The Verbal Descriptor Scale (VDS) | 49 | 30,1 | 114 | 69,9 | 55 | 45,1 | 67 | 54,9 | 12 | 26,7 | 33 | 73,3 | 13 | 40,6 | 19 | 59,4 | 8 | 34,8 | 15 | 65,2 | 137 | 35,6 | 248 | 64,4 |
| The Brief Pain Inventory (BPI) tool | 24 | 14,7 | 139 | 85,3 | 27 | 22,1 | 95 | 77,9 | 8 | 17,8 | 37 | 82,2 | 3 | 9,4 | 29 | 90,6 | 4 | 17,4 | 19 | 82,6 | 66 | 17,1 | 319 | 82,9 |

Table 4.6: Availability of pain rating assessment tools for patients who can self-report their pain (N = 385)

4.2.2.2.2 Availability of the Wong-Baker FACES scale (N = 385)

The Wong-Baker FACES scale is used to rate pain in children who can report their own pain (Fink & Gallagher 2019:231; Alizadeh, Paymard, Khalili & Hejr 2017:2). Its availability is illustrated in Table 4.6. The scale was reported to be available with 96,9% in cardiac wards (F = 31; n = 32). Paediatric wards followed with 90,9% (F = 40; n = 44) availability, obs-gynae wards, with 87.0% (F = 20; n = 23) surgical wards with 85,9% (F = 140; n = 163), and medical wards with 69,7% (F = 85; n = 122) (see Table 4.6).

4.2.2.2.3 Availability of the Verbal Analogue Scale (VAS) (N = 385)

Kliger, Stahl, Haddad, Suzan, Adler and Eisenberg (2015: 538) define the Verbal Analogue Scale (VAS) as a scale that can be used to rate the pain in adults who can report their pain. Not in any of the different wards was the scale reported to be available to more than 50% of respondents. It was most available in medical wards (f = 49,2%; F = 60; n = 122), followed by Cardiac wards (f = 43,8%; F = 14; n = 32), surgical wards (f = 31,9; F = 52; n = 163), obs-gynae wards (f = 30,4%; F = 7; n = 23) with for obvious reasons the least available in paediatrics wards (f = 28,9%; F = 13; n = 45) (see Table 4.6).

4.2.2.2.4 Availability of the Verbal Descriptor Scale (VDS) (N = 385)

The Verbal Descriptor Scale (VDS) is also known to be used to rate pain in adults who can self-report pain (Booker & Haedtke 2016:65). The scale was most reported to be accessible in medical wards (f = 45,1%; F = 55; n = 122), cardiac wards (f = 40,6%; F = 13; n = 32), obs-gynae wards (f = 34,8%; F = 8; n = 23), paediatrics wards (f = 26,7%; F = 12; n = 45), and the least available in surgical wards (f = 30,1%; F = 43; n = 163) (see Table 4.6).

4.2.2.2.5 Availability of the Brief Pain Inventory (BPI) (N = 385)

Another pain rating tool, the Brief Pain Inventory (BPI) questionnaire, was reported to be available, as is demonstrated in Table 4.6. This tool is used to rate pain in cancer

patients who can self-report their pain (Alizadeh-Khoei, Sharifi, Akbari, Fadeyevatan & Haghi 2017:565). The BPI was reported to be not more than 22% available in the various wards. It was available in medical wards (f = 22,1%; F = 27: n = 122), in paediatric wards (f = 17,8%; F = 8; n = 45), in obs-gynae wards (f = 17,4% F = 4; n = 23), in surgical wards (f = 14,7%; F = 24; n = 163), and in cardiac wards (f = 9,4%; F = 3; n = 32) (see Table 4.6).

The researcher requested that respondents indicate the availability of pain rating assessment tools for patients who cannot self-report pain.

4.2.2.3 Pain rating assessment tools for patients who cannot self-report their pain (N = 385)

The selected pain rating assessment tools of patients who could not self-report their pain to enable pain assessments were assessed within wards: They were (1) a CRIES (Crying, Required Oxygen, Increased vital signs, Expressions, Sleeplessness); (2) the Neonatal Pain, Agitation and Sedation Scale (N-PASS); (3) the Neonatal Infant Pain Scale (NIPS); the FLACC pain scale (Face, Legs, Activity, Crying, and Consolability); (5) the COMFORT-Behaviour pain scale (COMFORT-B); (6) the Critical Care Pain Observational Tool (CPOT); and (7) the Behavioural Pain Scale (BPS). Table 4.7 il-lustrates the availability of these different tools.

A FLACC pain scale was indicated as the most available scale within the different wards, with 96,1% (n = 370) availability, and CRIES, with 65,2% (n = 251) available. The scales reported to be less than 50% available in wards were COMFORT-B (f = 30,1%; n = 116), BPS (f = 27,8%; n = 107), NIPS (f = 24,7%; n = 95), CPOT (f = 21,8%; n = 84), and the least available was N-PASS (f = 21,0%; n = 81).

Respondents reported the availability of these scales in different wards, as illustrated in Table 4.7.

| | Nursi | ng Wa | rds | | | | | | | | | | | | | | | | | | | | | |
|---|-------|-------|-----|------|-------|------|-----|------|-------|---------|-----|------|-------|------|-----|------|-------|--------|-----|------|-------|------|-----|------|
| Pain rating assessment tools of pa- | Surgi | cal | | | Medi | cal | | | Pae | diatric | | | Card | diac | | | Obs | - gyna | e | | тоти | AL. | | |
| tients who | n = 1 | 63 | | | n = 1 | 22 | | | n = - | 45 | | | n = 3 | 32 | | | n = 1 | 23 | | | N = 3 | 85 | | |
| report their pain | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | |
| | F = | % = | F = | % = | F = | % | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = |
| FLACC pain scale | 153 | 93,9 | 10 | 6,1 | 120 | 98,4 | 2 | 1,6 | 44 | 97,4 | 1 | 2,2 | 31 | 96,9 | 1 | 3,1 | 22 | 95,7 | 1 | 4,3 | 370 | 96,1 | 15 | 3,9 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| CRIES pain scale | 107 | 65,6 | 56 | 34,4 | 63 | 51,6 | 59 | 48,4 | 45 | 100 | 0 | 0 | 21 | 65,6 | 11 | 34,4 | 15 | 65,2 | 8 | 34,8 | 251 | 65,2 | 134 | 34,8 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| COMFORT- B pain scale | 44 | 27,0 | 119 | 73,0 | 39 | 32,0 | 83 | 68,0 | 14 | 31,1 | 31 | 68,9 | 11 | 34,4 | 21 | 65,5 | 8 | 34,8 | 15 | 65,2 | 116 | 30,1 | 269 | 69,9 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| BPS | 34 | 20,9 | 129 | 79,1 | 41 | 33,6 | 81 | 66,4 | 16 | 35,6 | 29 | 64,4 | 9 | 28,1 | 23 | 71,9 | 7 | 30,4 | 16 | 69,6 | 107 | 27,8 | 278 | 72,2 |

| NIPS | 41 | 25,2 | 122 | 74,8 | 31 | 25,4 | 91 | 74,6 | 11 | 24,4 | 34 | 75,6 | 6 | 18,8 | 26 | 81,3 | 6 | 26,1 | 17 | 73,9 | 95 | 24,7 | 290 | 75,3 |
|--------|----|------|-----|------|----|------|----|------|----|------|----|------|---|------|----|------|---|------|----|------|----|------|-----|------|
| | | | | | | | | | | | | | | | | | | | | | | | | |
| СРОТ | 31 | 19,0 | 132 | 81,0 | 30 | 24,6 | 92 | 75,4 | 11 | 24,4 | 34 | 75,6 | 8 | 25,0 | 24 | 75,0 | 4 | 17,4 | 19 | 82,6 | 84 | 21,8 | 301 | 78,2 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| N-PASS | 32 | 19,6 | 131 | 80,4 | 27 | 22,1 | 95 | 77,9 | 11 | 24,4 | 34 | 75,6 | 5 | 15,6 | 27 | 84,4 | 6 | 26,1 | 17 | 73,9 | 81 | 21,0 | 304 | 79,0 |

4.2.2.3.1 Availability of the FLACC pain scale (N = 385)

FLACC pain scale is described as the most reliable and accessible in nursing for rating the pain in patients who cannot self-report their pain (Matsuishi, Hoshino, Shimojo, Enomoto, Kido, Hoshino, Sumitani & Inoue 2018:7) and its availability is as illustrated in Table 4.7. This pain scale was reported to be mostly always available in medical wards (f = 98,4%; F = 120; n = 122), followed by the availability in paediatric wards (f = 97,4%; F = 44; n = 45), in cardiac wards (f = 96,9%; F = 31; n = 32), in obs-gynae wards (f = 95,7%; n = 22; F = 23), and in surgical wards (f = 93,9%; F = 153; n = 163) (see Table 4.7).

4.2.2.3.2 Availability of a CRIES pain scale (N = 385)

A pain scale known to be used to assess pain in neonatal patients, namely CRIES (Andersen et al. 2018:4), was reported to be available in different wards (see Table 4.7). Since CRIES are well-known for use in neonatal patients, it was not surprising to find it to be reported to be 100% available in paediatric wards (see Table 4.7). Surgical wards followed with 65,6% (F = 107; n = 163) availability, cardiac wards with 65,6% (F = 21; n = 32), obs-gynae wards with 65,2% (F = 15; n = 23); and medical wards with 51,6% (n = 63; F = 122) (see Table 4.7).

4.2.2.3.3 Availability of the COMFORT-Behaviour pain scale (N = 385)

The third most reported pain scale, the COMFORT-B, is well-known to be used in paediatric intensive care areas to enable the assessment of pain in children who cannot self-report their pain (Seixas-Moizes & Wichert-Ana 2017:10) and its availability is as indicated in Table 4.7. Paediatric intensive care units were not selected as part of the study, and as it is the place recommended to utilise COMFORT-B, it is not strange that this was also reported to be not more than 34% available in any of the wards. Despite not being explicitly recommended for other wards, the scale was readily available in some wards namely, obs-gynae wards (f = 34,8%; F = 8; n = 23), with cardiac wards (f = 34,4%; F = 11; n = 32), medical wards (f = 32.0%; F = 38; n = 122), paediatric

wards (f = 31,1%; F = 14; n = 45), and reported with the least availability in surgical wards (f = 27,0% (F = 44; n = 163) (see Table 4.7).

4.2.2.3.4 Availability of the Behavioural Pain Scale (BPS) (N = 385)

Kotfis et al. (2017:70) emphasise the use of BPS to assess uncommunicative, critically ill, sedated, and intubated patients in intensive care units who cannot report their pain. This scale was available according to the respondents (see Table 4.7). Because intensive care units, where BPS is to be used, were excluded in this study, its availability did not exceed 35%. It was, however, available in paediatric wards (f = 35,6%; F = 16; n = 45), medical wards (f = 33,6%; F = 41; n = 122), obs-gynae wards (30,4%; F = 7; n = 23), cardiac wards (28,1%; F = 9; n = 32), and in surgical wards (f = 20,9%; F = 34; n = 163) (see Table 4.7).

4.2.2.3.5 Availability of the Neonatal Infant Pain Scale (NIPS) (N = 385)

Andersen et al. (2018:4) and Desai, Aucott, Frank and Silber-Flagg (2018:262) explain NIPS as a valid and reliable scale to rate pain in patients who cannot self-report their pain, such as premature babies and neonates. Respondents reported its availability as illustrated in Table 4.7. The NIPS was available in obs-gynae wards with 26.1% availability (F = 6; n = 23), in medical wards with 25,4% (F = 31; n = 122), in surgical wards with 25,2% (F = 41; n = 122), in paediatric wards with 24,4% (F = 11; n = 45), and 18,8% in cardiac wards (F = 6; n = 32) (See Table 4.7).

4.2.2.3.6 Availability of the CPOT pain scale (N = 385)

The CPOT, described by Emsden, Schäfer, Denhaerynck, Grossmann, Frei, Kirsch (2020:12) and Kotfis et al. (2017:70), is a tool used to assess pain in non-conscious adults, critically ill ventilated patients, and non-ventilated patients mainly in intensive care units. The CPOT pain scale is primarily used in intensive care units, implying the rationale for its lowest availability as respondents from cardiac wards reported the scale to be 25.0% (F = 8; n = 32). In medical wards, it was 24.6% (F = 30; n = 122); in

paediatric wards, it was 24,4% (F = 11; n = 122); in surgical wards, it was 19,0% (F = 31; n = 163), and in obs-gynae, it was 17,4% (F = 4; n = 23) (see Table 4.7).

4.2.2.3.7 Availability of the N-PASS (N = 385)

According to the literature, the N-PASS should be available in neonatal intensive care units (Hillman, Tabrizi, Carson & Aucott 2015:5; Desai et al. 2017:287). For obvious reasons, as illustrated in Table 4.7, this scale was reported to be less than 50% available in various wards as it is more used in neonatal intensive units. Therefore, it was reported to be available with 26,1% in obs-gynae wards (f = 26,1%; F = 6; n = 23), paediatric wards with 24,4% (F = 11; n = 45), medical wards with 22,1% (F = 27; n = 122), surgical wards with 19,6% (F = 32; n = 163), and cardiac wards with 15,6% (F = 5; n = 32 (see Table 4.7).

The pain rating assessment tools for elderly patients with dementia or cognitive impairment to conduct pain assessments are for discussion.

4.2.2.4 Pain rating assessment tools for elderly patients with dementia or cognitive impairment (N = 385)

The four validated and reliable pain rating tools to rate pain for elderly patients with dementia or cognitive impairment selected were (1) the Abbey Pain Scale (ABBEY), (2) the Checklist on Nonverbal Pain Indicators (CNPI), (3) the Pain Assessment in Advanced Dementia scale (PAINAD), and (4) the Non-communicative Patient's Pain Assessment Instrument (NOPPAIN) pain scale (see Annexure 4). The four pain tools' availability is illustrated in Table 4.8 and discussed from the highest availability to the least available.

The availability of the pain rating assessment tools in different wards is illustrated in Table 4.8, as reported by the respondents.

| | Nurs | sing Wa | ards | | | | | | | | | | | | | | | | | | | | | |
|--|-------|---------|------|------|-------|-------|-----|------|-------|---------|-----|------|--------|-------|-----|------|--------|---------|-----|------|-------|------|-----|------|
| Pain rating assessment tools for el- | Surg | jical | | | Med | lical | | | Pae | diatric | | | Ca | rdiac | | | Ob | s- gyna | e | | TOTA | AL. | | |
| derly pa- tients with | n = ′ | 163 | | | n = ' | 122 | | | n = 4 | 45 | | | n = | 32 | | | n = | 23 | | | N = 3 | 85 | | |
| dementia or cognitive impairment | Yes | | No | | Yes | | No | | Yes | | No | | Yes | 5 | No | | Yes | 6 | No | | Yes | | No | |
| | F = | % = | F = | % = | F = | % | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = |
| CNPI | 41 | 25,2 | 122 | 74,8 | 35 | 28,7 | 87 | 71,3 | 11 | 24,4 | 34 | 75,6 | 9 | 28,1 | 23 | 71,9 | 6 | 26,1 | 17 | 73,9 | 102 | 26,5 | 283 | 73,5 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| NOP-PAIN | 31 | 19,0 | 132 | 81,0 | 31 | 25,4 | 91 | 74,6 | 9 | 20,0 | 36 | 80,0 | 4 | 12,5 | 28 | 87,5 | 5 | 21,7 | 18 | 78,3 | 80 | 20,8 | 305 | 79,2 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| PAINAD | 29 | 17,8 | 134 | 82,2 | 30 | 24,6 | 92 | 75,5 | 8 | 17,8 | 37 | 82,2 | 2 | 6,3 | 30 | 93,8 | 6 | 26,1 | 17 | 73,9 | 75 | 19,5 | 310 | 80,5 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| ABBEY pain scale | 29 | 17,8 | 134 | 82,2 | 23 | 18,9 | 99 | 81,1 | 8 | 17,8 | 37 | 82,2 | 3 | 9,4 | 29 | 90,6 | 4 | 17,4 | 19 | 82,6 | 67 | 17,4 | 318 | 82,6 |

Table 4.8: Availability of the pain rating assessment tools for elderly patients with dementia/cognitive impairment (N = 385)

4.2.2.4.1 Availability of CNPI scale (N = 385)

The CNPI scale was reported as the scale most available within the various wards, with 26,5% (n = 102) of respondents indicating its availability. NOPPAIN was available to 20,8% (n = 80) of respondents; PAINAD to 19,5% (n = 75), and ABBEY to 17,4% (n = 67) (see Table 4.8). According to Paice (2015:17), the CNPI is a pain rating assessment tool that is available to observe and measure pain behaviours in elderly patients with dementia or cognitive impairment during nursing care delivery. It was reported to be available as indicated in Table 4.8. The availability of the CNPI was reported not to be more than 28% in the different wards. In medical wards it was reported to be available with 28,7% (F = 35; n = 122), in cardiac wards with 28,1% (F = 9; n = 32), in obs-gynae wards with 26,1% (F = 6; n = 23), in surgical wards with 25,2% (F = 41; n = 163), and in paediatric wards it was available with 24,4% (F = 11; n = 45).

4.2.2.4.2 Availability of the NOPPAIN (N = 385)

As illustrated in Table 4.8, respondents reported that the NOPPAIN was available. However, in some of the wards, the participants disagreed about its availability. This pain rating assessment tool is used to observe pain behaviours in patients with dementia or cognitive impairment during the delivery of nursing care (Paice 2015:17). NOPPAIN was one of the least reported tools to be available with only 25,4% (F = 31; n = 122) of the respondents indicating its availability in medical wards; 21,7% (F = 5; n = 23) in obs-gynae wards; 20,0% (F = 9, n = 45) in paediatric wards; 19,0% (F = 31; n = 163) in surgical wards; and 12,5% (F = 4; n = 32) available in cardiac wards.

4.2.2.4.3 Availability of the PAINAD scale (N = 385)

The PAINAD scale, known for its use to rate pain rate in elderly patients with advanced dementia to resolve problematic behaviour and improve the quality of life (Malara et al. 2016:1224; Fry et al. 2016:1288), was available in all the wards (see Table 4.8).
This scale was indicated to be available in obs-gynae wards with only 26,1% (F = 6; n = 23) of the respondents, 24,6% (F = 30; n = 122) in medical wards, 17,8% (F = 29; n = 163) in surgical wards; 17,8% (F = 8; n = 45) in paediatric wards; and 6,3% (F = 2; n = 32) in cardiac wards (see Table 4.8).

4.2.2.4.4 Availability of the Abbey Pain scale (ABBEY) (N = 385)

As explained by Paice (2015:12), the Abbey Pain scale is used to rate and assess pain in patients with late-stage dementia or cognitive impairment. Table 4.8 illustrates that this scale used in patients with advanced dementia or cognitive impairment was reported to be the least available in most of the wards. In medical wards it was 18,9% (F = 23; n = 122) available, 17,8% (F = 8; n = 163) in surgical wards; 17,8%; (F = 8; n = 45) in paediatric; in obs-gynae wards 17,4% (F = 4; n = 23) in obs-gynae wards, and 9,4% (F = 3; n = 32) cardiac wards.

The availability of human resources is essential in conducting pain assessments to enhance the transfer of learning of pain management competencies of nurses.

4.2.2.5 Human resources (N = 385)

Human resources refers to the number of people employed in an organisation. They are regarded as a significant asset in terms of diverse talents and essential characteristics with roles, skills, and abilities to accomplish the set organisational goals (Johnson & Davey 2019:92). The respondents indicated the availability of the following human resources to conduct pain assessments (see Annexure 4): (1) the registered nurse with pain management training, (2) the clinical facilitators, (3) the pain nurses working in acute or chronic pain services, (4) the pain nurse specialists, (5) the nurse educators, (6) the pain management physicians, (7) the ward nurse managers, (8) and the nurse supervisors. The availability of the different cadres of human resources within wards is illustrated in Table 4.9 and discussed from the highest to the least available. The registered nurses with pain management training were the human resource most available within the wards, with 92,7% (n = 357; N = 385) of respondents indicating their availability (see Table 4.9). The nurse educators were reported to be 89,4% available (n = 344; N = 385), the pain nurses working in acute or chronic pain services 87,8% (n = 337; N = 384), the pain management physicians 87,3% (n = 337; N = 385), the clinical facilitators 85,5% (n = 329; N = 385), the pain nurse specialists 83,1% (n = 320; N = 385), the ward nurse managers with pain management training 81,6% (n = 314; N = 385) and the nurse supervisors with pain management training 76,6% (n = 295; N = 385) (see Table 4.9). The respondents reported the availability of these human resources in different wards, as illustrated in Table 4.9.

Table 4.9: Availability of human resources (N = 385)

| | Nurs | sing V | /ards | 6 | | | | | | | | | | | | | | | | | | | | |
|---|-------|--------|-------|------|-------|------|----|------|-----|---------|----|------|-----|------|----|------|-------|--------|----|------|-----|-------|-----|------|
| | Surg | gical | | | Med | ical | | | Pae | ediatri | С | | Car | diac | | | Obs | - gyna | ae | | | TOT | AL | |
| Human Resources | n = ' | 163 | | | n = ′ | 122 | | | n = | 45 | | | n = | 32 | | | n = 2 | 23 | | | N = | 385 | | |
| | Yes | | No | | Yes | | No | | Yes | 5 | No | | Yes | 6 | No | | Yes | | No | | Yes | | No | |
| | F = | % = | F | % = | F = | % | F | % = | F | % = | F | % = | F | % = | F | % = | F = | % = | F | % = | F = | % = | F | % = |
| | | | = | | | | = | | = | | = | | = | | = | | | | = | | | | = | |
| Registered nurse with pain manage- | 150 | 06.0 | _ | 2.4 | 111 | 01.0 | 11 | 0.0 | 27 | 02.2 | 0 | 17.0 | 24 | 06.0 | 4 | 2.4 | 20 | 07.0 | 2 | 12.0 | 257 | 02.7 | 20 | 7.2 |
| ment training | 100 | 96,9 | 5 | 3,1 | | 91,0 | 11 | 9,0 | 37 | 02,2 | 0 | 17,0 | 51 | 96,9 | | 3,1 | 20 | 87,0 | 3 | 13,0 | 307 | 92,7 | 20 | 7,3 |
| Nurse educators | 149 | 91,4 | 14 | 8,6 | 112 | 91,8 | 10 | 8,2 | 33 | 73,3 | 12 | 26,7 | 28 | 87,5 | 4 | 12,5 | 22 | 95,7 | 1 | 4,3 | 344 | 89,4 | 41 | 10,6 |
| Pain nurses working in acute or chronic | 151 | 92,6 | 11 | 6,7 | 106 | 86,9 | 16 | 13,1 | 35 | 77,8 | 10 | 22,2 | 26 | 81,3 | 6 | 18,7 | 19 | 82,6 | 4 | 17,4 | 337 | 87,8 | 47 | 12,2 |
| | 454 | 00.0 | 10 | 7.4 | 407 | 077 | 45 | 40.0 | 00 | 00.0 | 0 | 00.0 | 00 | 74.0 | 0 | 00.4 | 40 | 00.0 | | 47.4 | 000 | 07.0 | 10 | 407 |
| Pain management physicians | 151 | 92,6 | 12 | 7,4 | 107 | 87,7 | 15 | 12,3 | 36 | 80,0 | 9 | 20,0 | 23 | 71,9 | 9 | 28,1 | 19 | 82,6 | 4 | 17,4 | 336 | 87,3 | 49 | 12,7 |
| Clinical facilitators | 148 | 90,8 | 15 | 9,2 | 104 | 85,2 | 18 | 14,8 | 31 | 68,9 | 14 | 31,1 | 26 | 81,3 | 6 | 18,8 | 20 | 87,0 | 3 | 13,0 | 329 | 85,5 | 56 | 14,5 |
| Pain nurse specialists | 143 | 87,7 | 20 | 12,3 | 99 | 81,1 | 23 | 18,9 | 33 | 73,3 | 12 | 26,7 | 27 | 84,4 | 5 | 15,6 | 18 | 78,3 | 5 | 21,7 | 320 | 83,1 | 65 | 16,9 |
| Ward Nurse managers with pain man- | 129 | 79 1 | 34 | 20.9 | 106 | 86.9 | 16 | 13.1 | 38 | 84.4 | 7 | 15.6 | 24 | 75.0 | 8 | 25.0 | 17 | 73 9 | 6 | 26.1 | 314 | 81.6 | 71 | 18.4 |
| agement training | 120 | 10,1 | | 20,0 | 100 | 00,0 | 10 | 10,1 | 00 | 0-1,-1 | ľ | 10,0 | 27 | 10,0 | | 20,0 | | 10,0 | Ŭ | 20,1 | 014 | 01,0 | / 1 | 10,4 |
| Nurse supervisors with pain manage- | 130 | 79.8 | 33 | 20.2 | 99 | 81.1 | 23 | 18.9 | 32 | 71.1 | 13 | 28.9 | 19 | 59.4 | 13 | 40.6 | 145 | 65.2 | 8 | 34.8 | 295 | 76.6 | 90 | 23.4 |
| ment training | | 10,0 | | ,_ | | 51,1 | | ,0 | | ,. | | _0,0 | | 55,1 | | ,0 | | 50,2 | Ŭ | 5.,0 | | . 0,0 | | , ' |

4.2.2.5.1 Availability of the registered nurse with pain management training (N = 385)

Registered nurses with pain management training are nurses empowered to be competent to conduct pain assessments (Yoo, De Gange, Kim and Oh 2019:10; Fitzgerald, Tripp and Halksworth-Smith 2017:55), and thus, it could be expected that they would be available in all wards where patients might experience pain. Their availability in the different wards is illustrated in Table 4.9. It is good to state that these nurses were reported to be nearly always available in surgical wards (f = 96,9%; F = 158; n = 163) and in cardiac wards (f = 96,9%; F = 31; n = 32). Medical wards followed with the availability of 91,0% (F = 111; n = 122), obs-gynae wards with 87,0% (F = 20; n = 23), and paediatric wards with 82,2% (F = 37; n = 45) (see Table 4.9).

4.2.2.5.2 Availability of nurse educators (N = 385)

Nurse educators' availability is crucial because they effectively pursue evidence-based interventions that empower nurses' pain knowledge and attitudes towards conducting pain assessments (Shoqirat, Mahasneh, Singh & Hadid 2019:6; Chow & Chan 2015:371). Respondents believed that nurse educators were between 73,3% and 95,7% % available in the different wards (see Table 4.9). It was reported that nurse educators were nearly always available in obs-gynae wards, with 95,7% (F = 22; n = 23), in medical wards, with 91,8% (F = 112; n = 122), in surgical wards, with 91,4% (F = 149; n = 163), in cardiac wards with 87,5% (F = 28; n = 32), and in paediatric wards with 73,3% (F = 33; n = 45) (see Table 4.9).

4.2.2.5.3 Availability of the pain nurses working in acute and chronic pain services (N = 385)

It is comforting to know that pain nurses working in acute or chronic pain service areas were available to assist with pain assessments (see Table 4.9). These nurses are described as registered nurses who specialise and are employed in pain management nursing services to support other nurses and in helping patients deal with acute and chronic pain (Khatib & Razvi 2018:50; Rockett, Vanstone, Chand & Waeland 2017:1240). As illustrated in Table 4.9, the pain nurses were reported primarily available in surgical wards (f = 92,6% (n = 151; F = 163), followed by medical wards (f = 86,9%; F = 106; n = 122), obs-gynae wards (f = 82,6%; F = 19; n = 23), cardiac wards (f = 81,3%; F = 26; n = 32), and lastly paediatric wards (f = 77,8%; F = 35; n = 45).

4.2.2.5.4 Availability of pain management physicians (N = 385)

Pain management physicians are anaesthesiologists who collaboratively work with the nurses and play an essential role in conducting pain assessment and managing pain (Jai, Bakshi & Thota 2020:555; Deni, Greco, Turi, Meani, Comotti, Perotti, Mello, Colnaghi, Pasculli, Nardelli, Landoni & Beretta 2019:586). Their availability was reported as illustrated in Table 4.9. Most respondents reported these physicians to be 92,6% available in surgical wards (f = 92,6%; F = 15; n = 163). In other nursing wards, namely in medical wards, they were 87,7% available (F = 107; n = 122), in obs-gynae wards 82,6% (F = 19; n = 23), in paediatric wards 80,0% (F = 36; n = 45), and in cardiac wards they were 71,9% available (F = 23; n = 32) (see Table 4.9).

4.2.2.5.5 Availability of the clinical facilitators (N = 385)

Clinical facilitators are skilled registered nurses responsible for the facilitation of teaching and learning of the theoretical and clinical components in a clinical setting to deliver quality care (Sweet & Broadbent 2017:35; Phillips et al. 2017:4344), for example, reinforcing the learning of pain assessment. As is illustrated in Table 4.9 the respondents in surgical wards reported the clinical facilitators with 90,8% availability (F = 148; n = 163), respondents from obs-gynae wards reported 87,0% availability (F = 20; n = 23), 85,2% (F = 104; n = 122) in medical wards; 81,3% (F = 26; n = 32) in cardiac wards; and 68,9% (F = 31; n = 45) in paediatric wards.

4.2.2.5.6 Availability of pain nurse specialists (N = 385)

Pain nurse specialists collaborate with nurses to assess pain (Rocket et al. 2017:1237). Thus, these specialists were indicated to be variably available in all the

wards (see Table 4.9). The pain nurse specialists were available in all wards in different percentages: in surgical wards (f = 87,7%; F = 143; n = 163), in cardiac wards (f = 84,4%, F = 27; n = 32), in medical wards (f = 81,1%F = 99; n = 122), in obs-gynae wards (f = 78,3% F = 18; n = 23), and in paediatric wards (f = 73,3%; F = 33; n = 45) (see Table 4.9).

4.2.2.5.7 Availability of the ward nurse managers with pain management training (N = 385)

Another role player in pain management is the nurse manager, who has pain management training and is very competent in pain management (Duffield, Gardener, Doubrovsky & Wise 2019:1538). These nurse managers were reported to be 86,9% available in medical wards (f = 86,9%; F = 106; n = 122) (see Table 4.9). In paediatric wards, they were reported to be 84,4% available (F = 38; n = 45). In surgical wards, they were seen as having 79,1% availability (F = 129; n = 163), in cardiac wards, 75,0% (F = 24; n = 32), and in obs-gynae wards, 73,9% (F = 17; n = 23) (see Table 4.9).

4.2.2.5.8 Availability of the nurse supervisors with pain management training (N = 385)

Nurse supervisors with pain management training, those empowered to be competent to provide post-training supervisory support to the nurses on how to conduct pain assessments, are an important resource in the management of pain (Mahama & Ninnoni 2019:3). Their availability in the wards where patients need pain management is essential, and the respondents indicated their availability (see Table 4.9) to be in medical wards 81,1% (F = 99; n = 122), in surgical wards 79,8%; (F = 130; n = 163), in paediatric wards 71,1% (F = 32; n = 45), in obs-gynae wards 65,2% (F = 15; n = 23), and in cardiac wards 59,4% (F = 19; n = 32). It is interesting to note that supervisors were the least available in cardiac wards. This finding is supported by Bradley-Ingle (2018:34), who indicated a great shortage of nurse supervisors despite significant demand for supervision services.

4.2.2.6 Other types of support (*N* = 385)

The support by others, namely (1) patients, (2) pain management support groups, (3) patient pain management hotlines, and (4) patient pain management websites to support patients in pain, were identified as also being important resources to manage pain (Herr, Coyne, Ely, Gélinas & Manworren 2019:214). Peer support groups are especially beneficial to patients in pain management (Tolley, Michel, Willams & Renschler 2020:9). The mentioned resources were identified as available by respondents, as illustrated in Table 4.10. The respondents perceived other patients, other people or patients with pain to be 83,1% available (n = 320) to support pain management (Herr et al. (2019:214). It is important to note that this is the respondents' opinion and not that of the patients themselves, as they were not part of this study. The respondents identified pain management support groups to be 76,4% (n = 294) available, patient pain management hotlines 63,4% (n = 244), and the patient pain management websites 58,7% available (n = 226) (see Table 4.10). Table 4.10 indicates the availability of these other types of support in various wards, as reported by the respondents.

| | Surgi | cal | | | Med | ical | | | Pae | diatric | | | Card | diac | | | Obs | - gynae | e | | тот | AL. | | |
|---|-------|------|-----|------|-------|------|-----|------|-------|---------|-----|------|-------|------|-----|------|-------|---------|--------|------|-------|------|-----|------|
| Patient sup- ports | n = 1 | 63 | | | n = ′ | 122 | | | n = 4 | 45 | | | n = 3 | 32 | | | n = 2 | 23 | | | N = 3 | 85 | | |
| | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | |
| | F = | % = | F = | % = | F = | % | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F= | % = |
| Patients or (other people or patients with pain) | 142 | 87,1 | 21 | 12,9 | 95 | 77,9 | 27 | 22,1 | 37 | 82,2 | 8 | 17,8 | 26 | 81,3 | 6 | 18,8 | 20 | 87,0 | 3 | 13,0 | 320 | 83,1 | 65 | 16,9 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Pain man- agement sup- port groups | 127 | 77,9 | 36 | 22,1 | 98 | 80,3 | 24 | 19,7 | 28 | 62,2 | 17 | 37,8 | 22 | 68,8 | 10 | 31,3 | 19 | 82,6 | 4 | 17,4 | 294 | 76,4 | 91 | 23,6 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Patient pain management hotlines | 108 | 66,3 | 55 | 33,7 | 78 | 63,9 | 44 | 36,1 | 23 | 51,1 | 22 | 48,9 | 20 | 62,5 | 12 | 37,5 | 15 | 65,2 | 8 | 34,8 | 244 | 63,4 | 141 | 36,6 |
| | | | | | | | | | | | | | | | | | | | | | | | | |

Table 4.10: Availability of other types of support (N = 385)

| Patient pain management | 89 | 54,6 | 74 | 45,4 | 77 | 63,1 | 45 | 36,9 | 26 | 57,8 | 19 | 42,2 | 20 | 62,5 | 12 | 37,5 | 14 | 60,9 | 9 | 39,1 | 226 | 58,7 | 159 | 41,3 |
|----------------------------|----|------|----|------|----|------|----|------|----|------|----|------|----|------|----|------|----|------|---|------|-----|------|-----|------|
| websites | | | | | | | | | | | | | | | | | | | | | | | | |

4.2.2.6.1 Availability of "peer" patients or (other people or patients with pain) (N = 385)

Patients or other people with pain have been recognised as resources that play a role as "peer" support groups in patients with cancer pain for exchanging information about pain experience, pain management, gaining recognition, emotional support, and caring for others (Huber, Muck, Maatz, Keck, Enders, Maatouk & Ihrig 2018:5). The respondents believed that other patients with pain were available to the patients in surgical wards (87,1%, F = 142; n = 163), in obs-gynae wards (87,0%; F = 20; n = 23) in paediatric wards (82,2% (F = 37; n = 45), in cardiac wards (81,3%; F = 26; n = 32), and in medical wards (77,9%; F = 95; n = 122) (see Table 4.10).

4.2.2.6.2 Availability of the pain management support groups (N = 385)

Pain management support groups help to empower patients with persistent pain and to inform them on how to employ self-pain management strategies (Clauw, Essex, Pitman & Jones 2019:185; Hylands-White, Duarte & Raphael 2017:36). Table 4.10 illustrates the respondents' views about the availability of these groups. The availability of pain management support groups was indicated to be 82,6% (F = 19; n = 23) available in obs-gynae wards, 80,3% (F = 98; n = 122) in medical wards, 77,9% (F = 127; n = 163) in surgical wards; 68,8% (F = 22; n = 32) in cardiac wards; and 62,2% (F = 28; n = 45) in paediatric wards (see Table 4.10).

4.2.2.6.3 Availability of the pain management hotlines (N = 385)

Patient pain management hotlines are services utilised by patients, families, and care providers comprising a multidisciplinary team to support such patients with traumatic or postoperative pain (Rhame, Le, Horner, Thomas, Foreman, Kreitzer & Ngwenya 2019:353). The respondents agreed that these hotlines were available, as indicated in Table 4.10). Pain management hotlines were understood by the respondents to be available in surgical wards (f = 66,3%; F = 108; n = 163), obs-gynae (f = 65,2%; F = 15; n = 23), medical wards (f = 63,9%; F = 78; n = 122), in cardiac wards (f = 62,5%;

F = 20; n = 32), and also in paediatric wards (f = 51,1%; F = 23; n = 45) (see Table 4.10).

4.2.2.6.4 Availability of the pain management websites (N = 385)

Lastly, patient pain management websites are online resources important in taking care that patients with persistent pain stay updated about self-management skills training such as assessing and managing their pain conditions (Yeh, Lee & Chou 2019:30; Devan, Perry, van Hattem, Thurlow, Shepherd, Muchemwa & Grainger 2019:1593). This finding, as is illustrated in Table 4.10, indicated the patient pain management websites that respondents in medical wards could access were 63.1% available (F = 77; n = 122), for cardiac wards, it was 62,5% (F = 20; n = 32), in obs-gynae wards it was 60,9% (F = 14; n = 23), in paediatric wards 57,8% (F = 26; n = 45), and in surgical wards 54,6% (F = 89; n = 163).

The perceptions about publications and electronic resources were also evaluated for availability to enable the nurses to assess pain to enhance the transfer of learning of nurses' pain management (see Annexure 4, subsection 7).

4.2.2.7 Publications and electronic resources (N = 385)

Publications and electronic resources are important to enhance the transfer of learning of pain management competencies of nurses (Zitzmann, Matthisson, Ohla & Joda 2020:3). The selected publications and electronic resources identified to be relevant in this study were (1) e-learning modules indicated to be 94,8% (n = 365) available, (2) best clinical practice guidelines for pain assessment 93,2% (n = 359), (3) electronic flow sheets 91,9% (n = 354), (4) a pain toolkit 86,8% (n = 334), (5) clinical updates or journals 74,8% (n = 288), (6) printed reference books 68,8% (n = 265), (7) the fact sheets 51,4% (n = 198), (8) videos on pain management 50,4% (n = 194), and (9) e-newsletters to be 44,4% (n = 171) available (see Table 4.11). These selected publications and electronic resources were reported to be available by respondents in different wards, as portrayed in Table 4.11.

| | Nurs | sing W | /ards | 5 | | | | | | | | | | | | | | | | | | | | |
|-----------------------------------|-------|--------|-------|------|-------|------|----|------|-----|---------|----|------|-----|------|----|------|-----|--------|----|------|-----|------|-----|------|
| | Surç | jical | | | Med | ical | | | Pae | ediatri | с | | Car | diac | | | Obs | s- gyn | ae | | TOT | AL | | |
| Publications and electronic re- | n = 1 | 163 | | | n = 1 | 122 | | | n = | 45 | | | n = | 32 | | | n = | 23 | | | N = | 385 | | |
| sources | Yes | | No | | Yes | | No | | Yes | 3 | No | | Yes | 5 | No | | Yes | 3 | No | | Yes | | No | |
| | F = | % = | F | % = | F = | % | F | % = | F | % = | F | % = | F | % = | F | % = | F | % = | F | % = | F | % | F | % |
| | | | = | | | | = | | = | | = | | = | | = | | = | | = | | | | | |
| E-learning modules | 157 | 96,3 | 6 | 3,7 | 113 | 92,6 | 9 | 7,4 | 43 | 95,6 | 2 | 4,4 | 30 | 93,8 | 2 | 6,3 | 22 | 95,7 | 1 | 4,3 | 365 | 94,8 | 20 | 5,2 |
| Best clinical practice guidelines | 156 | 05.7 | 7 | 4.2 | 444 | 01.0 | 11 | 0.0 | 44 | 01.1 | 4 | • • | 20 | 00.6 | 2 | 0.4 | 22 | 05.7 | 4 | 1.2 | 250 | 02.2 | 26 | 6.0 |
| for pain assessment | 100 | 95,7 | ľ | 4,3 | 111 | 91,0 | | 9,0 | 41 | 91,1 | 4 | 0,9 | 29 | 90,6 | 3 | 9,4 | 22 | 95,7 | ľ | 4,3 | 309 | 93,Z | 20 | 0,0 |
| Electronic flowsheets | 151 | 92,6 | 12 | 7,4 | 113 | 92,6 | 9 | 7,4 | 40 | 88,9 | 5 | 11,1 | 29 | 90,6 | 3 | 9,4 | 21 | 91,3 | 2 | 8,7 | 354 | 91,9 | 31 | 8,1 |
| Pain Toolkit | 139 | 85,3 | 24 | 14,7 | 108 | 88,5 | 14 | 11,5 | 40 | 88,9 | 5 | 11,1 | 27 | 84,4 | 5 | 15,6 | 20 | 87,0 | 3 | 13,0 | 334 | 86,8 | 51 | 13,2 |
| Clinical updates or journals | 130 | 79,8 | 33 | 20,2 | 89 | 73,0 | 33 | 27,0 | 33 | 73,3 | 12 | 26,7 | 18 | 56,3 | 14 | 43,8 | 18 | 78,3 | 5 | 21,7 | 288 | 74,8 | 97 | 25,2 |
| Printed reference books | 118 | 72,4 | 45 | 27,6 | 83 | 68,0 | 39 | 32,0 | 32 | 71,1 | 13 | 28,9 | 17 | 53,1 | 15 | 46,9 | 15 | 65,2 | 8 | 34,8 | 265 | 68,8 | 120 | 31,2 |
| Fact sheets | 78 | 47,9 | 85 | 52,1 | 69 | 56,6 | 53 | 43,4 | 20 | 44,4 | 25 | 55,6 | 18 | 56,3 | 14 | 43,8 | 13 | 56,5 | 10 | 43,5 | 198 | 51,4 | 187 | 48,6 |
| Videos on pain management | 87 | 53,4 | 76 | 46,6 | 61 | 50,0 | 61 | 50,0 | 24 | 53,3 | 21 | 46,7 | 11 | 34,4 | 21 | 65,6 | 11 | 47,8 | 12 | 52,2 | 194 | 50,4 | 191 | 49,6 |
| E-newsletters | 72 | 44,2 | 91 | 55,8 | 60 | 49,2 | 62 | 50,8 | 15 | 33,3 | 30 | 66,7 | 14 | 43,8 | 18 | 56,3 | 10 | 43,5 | 13 | 56,5 | 171 | 44,4 | 214 | 55,6 |

 Table 4.11: Availability of the publications and electronic resources (N = 385)

4.2.2.7.1 Availability of the e-learning modules (N = 385)

The e-learning modules are online instructional modules that deliver content on how to conduct pain assessment via computer, internet, multimedia or video clip, and others (Ómgreen, Meyer & Buhl 2019:487). E-learning modules empower nurses with skills to conduct pain assessments, as confirmed by Watt-Watson, McGillion, Lax, Oskarsson, Hunter, MacLennan, Knickle and Victor (2019:42) and are very important as part of the professional development of nurses. They were reported to be available by 96,3% (n = 157; F = 163) of respondents in surgical wards; by 95,7% (F = 22; n = 23) in obs-gynae wards; by 95,6% (F = 43; n = 45) in paediatric wards; by 93,8% (F = 30; n = 32) in cardiac wards; and by 92,6% (F = 113; n = 122) in medical wards (see Table 4.11).

4.2.2.7.2 Availability of the best clinical practice guidelines for pain assessments (N = 385)

The best clinical practice guidelines for pain assessments are recommended to be used to enhance accurate methods to assess pain (Chisholms-Burns, Schwihammer, Malone, Kolesar, Bookstaver, and Lee 2019:526). In surgical wards, these guidelines have been reported to be available with 95,7% (F = 156; n = 163) of the respondents, as supported in the study by Smeland, Twycross, Lundeberg and Rustøen (2018:596). In obs-gynae wards, 95,7% (F = 22; n = 23), 91,1% (F = 41; n = 45) in paediatric wards, 91,0% (F = 111; n = 122) in medical wards, and 90,6% (n = 29; F = 32) in cardiac wards (see Table 4.11).

4.2.2.7.3 Availability of electronic flowsheets (N = 385)

To document clinical data of patient progress, such as pain assessment, electronic flowsheets are suitable computer software with sections to complete a patient's health information (Kronenberger & Ledbetter 2020:194; Stuebe, McKenzie, Tucker, Tully, Bryant & Verbiest 2018:1). Both in the surgical wards and medical wards, the

electronic flowsheets have been reported to have had a 92,6% availability (see Table 4.11). Respondents from obs-gynae wards reported the electronic flowsheets to be 91,3% (n = 21, F = 23) available, 90,6% (n = 29; F = 32) from cardiac wards, and 88,9% (n = 40; F = 45) from paediatric wards (see Table 4.11).

4.2.2.7.4 Availability of the Pain Toolkit (N = 385)

Another publication and electronic resource, the pain toolkit is an information booklet or internet application available for patients that suffer from pain and used to explain pain in simple language and provide support with skills on how to assess and selfmanaging their pain (Findley, Ryan, Cartwright & Martin 2019:2). The respondents reported that the pain tool kit was the most available in paediatric wards (f = 88,9% (F = 40; n = 45), followed by medical wards (f = 88,5%; n = 108; F = 122), obs-gynae wards (f = 87,0%; n = 20; F = 23), surgical wards (f = 85,3%; n = 139; F = 163), and in cardiac wards (f = 84,4%; n = 27; F = 32) (see Table 4.11).

4.2.2.7.5 Availability of the clinical updates or journals (N = 385)

Clinical updates or journals, for example, IASP online pain clinical updates, are indicated to be used to support patients in pain management and education on self-management strategies of pain (Amris, Jones & Williams 2019:4). Furthermore, Kim (2019:1) indicated another example of clinical updates or journals, such as nursing journals that can be shared to support patients in pain management. The indicated clinical updates or journals were reported to be most available in surgical wards (f = 79,8%; F = 163; n = 130), followed by obs-gynae wards (f = 78,3%; F = 23; n = 18), paediatric wards (f = 73,3%; F = 45; n = 33), medical wards (f = 73,0%; F = 122; n = 89), as well as in cardiac wards (f = 56,3%; F = 18; n = 32) (see Table 4.11).

4.2.2.7.6 Availability of the printed reference books (N = 385)

Respondents identified the printed reference books to be available, as portrayed in Table 4.11. These books are intended to locate information or a discipline-specific

subject such as pain management (Santos, Machado, Ribeiro, Neto, Ribeiro & Menezes 2018:326). As portrayed in Table 4.11, printed reference books were indicated to be primarily available in surgical wards (f = 72,4%; F = 118; n = 163), followed by paediatric wards (f = 71,1%; F = 32; n = 45), medical wards (f = 68,0%; n = 83; F = 122), obs-gynae wards (f = 65,2%; F = 15; n = 23), and lastly cardiac wards (f = 53,1%; F = 17; n = 32).

4.2.2.7.7 Availability of the fact sheets (N = 385)

Fact sheets are pieces of paper or electronic documents that provide essential information and prepare patients to manage their pain (Atkinson, Armbruster & Evans 2021:382; Alotaibi et al. 2018:531). Respondents that agreed that facts were available to conduct pain assessments were 56,6% (F = 69; n = 122) in medical wards, 56,5% (F = 13; n = 23) in obs-gynae wards, and 56,3% (F = 18; n = 32) in cardiac wards, 47,9% (F = 78; n = 163) in surgical wards, and in paediatric wards it was with 44,4% (F = 45; n = 20) (see Table 4.11). This finding is supported by IASP (2017:2), which says that pain fact sheets play a part in assessing pain.

4.2.2.7.8 Availability of videos on pain management (N = 385)

Videos on pain management are specific videos that promote the use of pain management strategies or observational pain assessment approaches (Elmali & Akpinar 2017:40; Ammaturo, Hadjistavropoulos & Williams 2017:1895). The respondents from surgical wards reported the videos on pain management to be 53,4% available (F = 87; n = 163), 53,3% (F = 24; n = 45) in paediatric wards, and 50,0% (F = 61; n = 122) in medical wards (see Table 4.11). In the other wards, videos on pain management were reported to have less than 50% availability: 47,8% in obs-gynae wards (F = 11; n = 23) and 34,4% in cardiac wards (F = 11; n = 32) (see Table 4.11).

4.2.2.7.9 Availability of E-newsletters (N = 385)

E-newsletters which are used as updates that provide information by subscribing through e-mails, for example, "PAIN IASP" specifically give information on how to assess pain (Rozario 2018: E6). They were indicated available in various wards (see Table 4.11). Even though the respondents believed that e-newsletters were available, they were not reported to be available more than 43%. The availability of newsletters was reported to be available with 49,2% (F = 60; n = 122) in medical wards, 44,2% (F = 72; n = 163) in surgical wards, 43,8% (F = 14; n = 32) in cardiac wards, 43,5% (F = 10; n = 23) in obs-gynae wards, and with 33,3% (F = 15; n = 45) in paediatric wards (see Table 4.11).

The findings indicated the availability of organisations that specialise in pain research, treatment, clinical practice, and education to enhance the transfer of pain management learning.

4.2.2.8 Organisations that specialise in pain management (N = 385)

Internationally some organisations specialise in pain management that includes, research, treatment, clinical practice, and education to support pain management strategies, for example, Pain Societies, IASP, and the WHO (Bhandari, Goddard, Campbell, Sangster & Stevens 2019:5; Yamitsky & Keefe 2016:1). Of the 385 respondents, 78.45% (n = 302) indicated that organisations that specialise in pain management were available to them (see Table 4.12).

| | Nurs | sing W | /ard | S | | | | | | | | | | | | | | | | | | | | |
|---|-------|--------|------|------|-----|-------|----|------|-----|---------|----|------|-----|-------|----|------|-----|--------|----|------|-----|------|-----|------|
| Organisations that special- | Surg | jical | | | Me | dical | | | Pae | ediatri | с | | Car | rdiac | | | Obs | s-gyna | ae | | тот | AL | | |
| ise in pain re- search, treat- ment, clinical practice, and education | n = 1 | 163 | | | n = | 122 | | | n = | 45 | | | n = | 32 | | | n = | 23 | | | N = | 385 | | |
| | Yes | | No | | Yes | 6 | No | | Yes | 5 | No | | Yes | 6 | No | 1 | Yes | ; | No |) | Yes | | No | |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n = | % | n = | % |
| | 128 | 78,5 | 40 | 21,5 | 96 | 78,7 | 26 | 21,3 | 35 | 77,8 | 10 | 22,2 | 24 | 75,0 | 8 | 25,0 | 19 | 82,6 | 4 | 17,4 | 302 | 78,4 | 83 | 21,6 |

 Table 4.12: Availability of organisations that specialise in pain management (N = 385)

Respondents indicated the availability (access to these organisations) in obs-gynae wards (f = 82,6%; F = 19; n = 23), in medical wards (f = 78,7%; F = 96; n = 122), in surgical wards (f = 78,5%; F = 128; n = 163) in paediatric wards, in surgical wards (f = 77,8%; F = 35; n = 45), and in cardiac wards (f = 75,0%; F = 24; n = 32) to support with pain management (see Table 4.12). The policies for pain management were identified for availability as they guide the nurses on how to assess and manage pain to enhance the transfer of learning of pain management of nurses (see Annexure 4, subsection 9).

4.2.2.9 Policies for pain management (N = 385)

Policies are principles or guidelines that govern activities adopted and proposed by an organization expected to be followed by its members (Mosby 2013:1415). The awareness of the availability of these policies for pain management is important to standard-ise and guide the nurses on how to assess and manage pain (Bonkowski, De Gange, Cade and Bulla 2018:184; Garcia, Bonilla, Kraychete, Flores, de Valtolina & Guerreo 2017:395), in any healthcare institution. Policies on pain management availability are reflected in Table 4.13. Within wards, 99,7% (n = 384; N = 385) of the respondents indicated that organisations specialising in pain management were available to them (see Table 4.13). Respondents from medical wards, paediatric wards, cardiac wards, and obs-gynae wards reported the policies for pain management to have had 100% availability and 99,4% availability in surgical wards (F = 162; n = 163) (see Table 4.13).

| | Nurs | ing Wa | ards | ; | | | | | | | | | | | | | | | | | | | | |
|----------|-------|--------|------|-----|-------|-----|----|---|-----|----------|----|---|-----|------|----|---|-----|--------|----|---|-------|------|-----|-----|
| Policies | Surg | ical | | | Medi | cal | | | Pae | ediatrio | c | | Car | diac | | | Obs | s-gyna | ae | | тот | AL | | |
| | n = 1 | 63 | | | n = 1 | 22 | | | n = | 45 | | | n = | 32 | | | n = | 23 | | | N = 4 | 1 | | |
| | Yes | | Nc |) | Yes | | Nc |) | Yes | 5 | No |) | Yes | ; | Nc |) | Yes | ; | Nc |) | Yes | | No | |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n = | % | n = | % |
| | 162 | 99,4 | 1 | 0,6 | 122 | 100 | 0 | 0 | 45 | 100 | 0 | 0 | 32 | 100 | 0 | 0 | 23 | 100 | 0 | 0 | 384 | 99,7 | 1 | 0,3 |

Table 4.13: Availability of the policies for pain management (N = 385)

The narrative responses of respondents who added additional comments were opencoded and thematically analysed.

4.2.3 Thematic content analysis

Only eight of the 385 respondents answered the open-ended questions and wrote narratives. The narratives addressed only two aspects, namely (1) additional resources needed to motivate pain management and (2) hindering factors to pain management. Participants believed that staff needs to be motivated by acknowledging good pain management skills, as was indicated by responses such as:

"Motivate staff and provide feedback".

"[Award a certificate] for example [an] outstanding in BEST Care certificate to staff monthly".

These findings are supported by Kee, McCrate, McLennon, Wall and Jones (2017:141), who report that the awareness of additional pain resources and feedback were motivating factors for pain management competency.

A participant commented about a hindering factor to pain management:

"pain assessment has too many mnemonics, which can hamper interest".

4.3 PHASE 2

In Phase Two, the objective was to identify and describe the nurses' characteristics and learning styles that enhance the transfer of learning in the context of two Saudi Arabian teaching hospitals.

4.3.1 Demographic characteristics

As explained, only one respondent did not complete the second questionnaire. Therefore, 384 out of the 385 respondents participated in Phase 2. After comparing all the demographic information from the two data sets, it was clear that the individual who did not complete the questionnaire was a female nurse from the Philippines with a diploma at the highest level of education who worked in a cardiac ward.

4.3.2 Respondents' characteristics identified to enhance transfer of learning of pain management competencies (N = 384)

Three hundred and eighty-four (N = 384) respondents responded on the characteristics identified to enhance the transfer of learning of pain management competencies. These characteristics were divided into (1) the application of what was learned before, (2) learner types, (3) motivations to learn, and (4) the motivations to apply knowledge in practice, as explained by Donavan and Darcy (2011:125). Respondents **rated** characteristics relevant to them on a Likert scale where "one" implied "this does not describe me at all" and "ten" to "this describes me perfectly". According to the characteristics identified to describe them best, the respondents identified the three most relevant ones. The three were then rated from high to low: **first, second, and third relevant**. The characteristics rated as most relevant were based on the weighted frequencies and percentage ratings obtained.

Applying what was learned before was the respondents' top-rated identified characteristic to enhance the transfer of learning of pain management competencies.

4.3.2.1 Application of what was learned before (N = 384)

Applying what was learned before refers to applying the experience, knowledge, and skills gained by respondents during pain management training and lessons in their clinical practice (Thompson, Johnson, Milligan & Briggs 2018:2155). In this study context, applying what nurses have learned before refers to the transfer of learning of pain

management competencies. The respondents judged every mentioned example of applying knowledge on a scale of 1–10 to determine the top four rated ways of applying knowledge. The top-rated description of how knowledge learned before is applied is illustrated in Figure 4.4. The **four methods to explain how what was learned be-fore was applied (application of what has been learned)** are discussed below.



Figure 4.4: Top four applications of what was learned before (N = 384)

The **first rated** aspect indicated by respondents as descriptive of how they apply what they have learned before was that they could rationally think to assess a patient experiencing pain with a weighted frequency of 311 (80.99%, N = 384) (see Figure 4.4). According to Yue, Zhang, Zhang and Jin (2017:94) and Johansen and O'Brien (2016:42), nurses are knowledgeable executors who can use their critical thinking, cognitive, and decision-making skills during their clinical practice skills related to pain management, thus supporting the study findings. This implies that cognitive abilities enable the nurses to apply their rational thinking to assess a patient experiencing pain.

The **second** highest-rated aspect illustrative of how the respondents applied what they have learned was that they **could concentrate well during pain management learning or training** with a weighted frequency of 300 (78,13%; N = 384) (see Figure 4.4). This means that the nurses' ability to focus well on the information given during pain management learning/training sessions enhanced the transfer of learning of pain management competencies. Manwere, Chipfuwa, Mukwamba and Chironda (2015:5) state that continuing education with necessary information organised by hospitals significantly impacts nurses' knowledge of pain management.

The **third** aspect was that they could **reassess pain after interventions**, with a weighted frequency of 290 (75.52%; N = 384). Respondents' ability to apply what they learned before was, according to participants, proven by their ability to reassess pain after pain interventions as supported by the study findings of other researchers (Jack-lyn 2019:14; Mazara, Zareel, Gharib & Aljazzazi 2016:1; El Rahi, Zaghloul & Murillo 2017:113).

Respondents rated learning during pain management learning or training sessions fourth. The idea that they were positively influenced by what was learned before was chosen by respondents with a weighted frequency of 288 (75%; N = 384) (see Figure 4.4). It was demonstrated by the study of Aloitabi, Higgins, Day and Chan (2018:531) that paediatric pain management educational programmes for nurses and other health professionals could reinforce their comprehensive content and quality of competency.

4.3.2.2 Learner types (N = 384)

The learner type is a characteristic known to influence the transfer of learning. Respondents, therefore, had to describe themselves as learners by rating the learning styles (how the style relates to them) described in the questionnaire. In this way, the learning styles of the respondents could be identified as their impact on the transfer of learning specifically related to pain management competencies. The top-rated description of how respondents described what type of learners they were is illustrated in Figure 4.5. The **three learner types** that best describe the respondents are discussed below.



Figure 4.5: Learner types (N = 384)

The learner type identified as most applicable to the respondents with a weighted frequency of 395 (102.86%; N = 384) was the **curious thinking learner type** (see Figure 4.5). Di, Danxia and Chun (2019:495) and Yang et al. (2018:46) indicate that nurses with this learning style are high-order thinking learners, enabling them to be adventurous and keen to explore new things about pain management in their workplace.

The **self-directed learner** was rated second with a weighted frequency of 298 (77,60%; N = 384) (see Figure 4.5), revealing that a majority of respondents indicated they were self-directed learners. Lemmetty and Collin (2019:61) demonstrate in their study that nurses with a self-directed learning style can take responsibility in their

workplaces for their learning practices, such as pain management and show their autonomy, self-guided and self-managed observations that enable them to be flexible to apply what they have learned before.

The **third-rated** learner type indicated by the respondents as their learning style described them as **hard-working enquiring learners** with a weighted frequency of 258 (67.19%; N = 384) (see Figure 4.5). As confirmed by Burger and Trehan (2018:131), most of the respondents were working hard by doing assigned work, listening attentively during pain management education sessions, revising the work given and being able to put in a considerable amount of practice to be successful in learning and apply skills of what was learned before.

4.3.2.3 Motivation to learn (N = 384)

The motivations referred to in this context are how the respondents described their desire to join and learn the study content to enhance the transfer of learning of pain management competencies (see Annexure 5). Respondents described how they were motivated to learn by rating the aspects described in the questionnaire. The top-rated descriptions of how respondents described what motivated them to learn about pain management are illustrated in Figure 4.6. **The three top-rated** motivations to learn are discussed in detail below.



Figure 4.6: Top three motivations to learn (N = 384)

Figure 4.6 portrays learning pain management skills relevant to their working areas as the most relevant motivation to learn (rated first), with a weighted frequency of 545 (141,92%; N = 384). According to Young, Goldbold and Wood (2018:172) and Aktas and Karabukut (2016:128), nurses were motivated to learn the essential pain management skills that were adapted to the context of their clinical practice. Nurses' knowledge and attitudes were reported to be influenced positively by receiving a brief and targeted educational program specific to managing pain in complex cases (Keen, McCrate, McLennon, Ellis, Wall and Jones 2017:143).

Respondents indicated their second-highest rated motivation to learn was attending pain management training programs with a weighted frequency of 522 (135,93%; N = 384) (see Figure 4.6). The respondents' motivation to learn by attending pain management educational training programmes was shown in the study by Peterson, Carlfjord, Schaller Gerdle and Larrsson (2017:20). Furthermore, Chaghari, Saffari, Ebadi and Ameryoun (2017:31) suggested that to motivate nurses to attend and learn pain management training programs they must be provided with a self-development

opportunity that includes the creation of informal learning opportunities, tangible support, the provision of time, information, resources, and reward strategies.

Gaining new knowledge about pain management was rated third with a weighted frequency of 450 (117.19%; N = 384) (see Figure 4.6). Respondents indicated that increasing knowledge was their driving force to learn during pain management training programs, as supported by Lin, Chen and Liu (2017:3556).

4.3.2.4 Motivation to apply knowledge in practice (N = 384).

Motivation to apply knowledge in practice is a characteristic enhancing the transfer of learning. Respondents' ratings (see Annexure 5) also enabled them to describe what motivates them to apply their knowledge in practice. Therefore, the respondents identified their top-rated motivations to apply knowledge in practice, as illustrated in Figure 4.7. The **three top-rated** characteristics that describe what was **motivational in applying knowledge in practice** are reflected in Figure 4.7.



Figure 4.7: Top three motivations to apply knowledge in practice (N = 384)

The first ranked quality indicated by respondents was that they have **the desire to successfully perform a pain management skill** with a weighted frequency of 651 (169.53%; N = 384). The need for nurses to perform pain management skills was shown to be influenced by the extrinsic motivation that improves their autonomy, competence, and the relatedness to apply in practice what they have learned (Zainuddin 2018:77; Nafukho, Alfred, Chakraborty & Johnson 2017:333).

Rated second was **that they aimed to increase their work performance in pain management** with a weighted frequency of 523 (136.1979%; N = 384). In their study, Olusadum and Anulika (2018:55) uphold that motivation impacts an individual's performance to achieve the organisation's objectives, such as performance about pain management practice of what they have learned. This indicated that the nurses' intentions to improve their pain management performance motivated them to apply what they had learned.

Receiving positive feedback about their performance was rated third with a weighted frequency of 466 (121.35%; N = 384) (see Figure 4.7). Nafukho et al. (2017:334) indicate that nurses are motivated to transfer what they have learned about pain management if there is a supportive climate of transfer of learning from peer support, their supervisors, and top management at the workplace. These things positively influence nurses to practice what they have learned, such as pain management (Nafukho et al. 2017:334).

4.3.3 Learning styles identified to enhance nurses' transfer of learning of pain management competencies (N = 384)

The concept "learning styles" refers to an individual's preferred ways of absorbing, processing, comprehending, and retaining new information and skills (how one learns) (Torlone & Vryonides 2016:63). Respondents rated their preferred learning style from the list of fourteen learning styles mentioned as relevant to them to enhance the transfer of learning of pain management competencies (see Annexure 5). The learning

styles rated as most preferred were based on the weighted frequencies and percentage ratings obtained from their ratings and rankings.

Five top-rated learning styles identified by the 384 respondents are portrayed in Figure 4.8.



Figure 4.8: Top five learning styles of preferences (N = 384)

Rated **first** was that transfer of learning happens when they can watch the activity, **such as during a demonstration,** with a weighted frequency of 409 (106.51%; N = 384) (see Figure 4.8). Thus, most respondents were considered visual learners who preferred to watch demonstrations and visual representations of presented material, including pictures, diagrams, and flow charts, as mentioned by other authors (Mangold, Kunze, Quinonez, Taylor & Tenison 2019:3).

Respondents indicated that learning by **reading through the information (**rated **sec-ond**) was a relevant style of learning with a weighted frequency of 338 (88.02%; N =

384) (see Figure 4.8). This learning style was indicated by Mangold et al. (2019:3) for learners who prefer to learn using written text and spoken explanations.

The learning style, which was rated third, with a weighted frequency of 296 (77.08%; N = 384), was learning **by listening to the information, such as during a lecture** (see Figure 4.8). Listening as a learning style represents auditory (aural) learners who prefer to learn from written and spoken explanations (Mangold et al. 2019:3).

The fourth style was that they preferred learning **by solving different pain management real-life problems** with a weighted frequency of 196 (51.04%; N = 384) (see Figure 4.8). Problem-based learning includes the ability to solve the problem by applying what has been learned and the associated metacognitive abilities (Siagan, Saragih and Sinaga (2019:336), which applies to these respondents in this study.

The respondents' judgement of their learning style rated **fifth** was indicated **by participating in a group discussion** with a weighted frequency of 196 (51.04%; N = 384) (see Figure 4.8). According to Mangold et al. (2019:3), nurses who prefer to participate in groups were considered to be active learners (Mangold et al. 2019:3). This was relevant to some of the respondents in this study, suggesting that they preferred to actively discuss the concepts of pain management in groups that enhanced their competencies.

4.3.4 Thematic content analysis

Seven out of 384 respondents wrote seven different narrative responses to answer the open-ended question. The narratives revealed two themes, namely, (1) active learning strategies during pain management training and (2) effective communication is essential to facilitate pain assessment. The following quotes illustrate some participants' learning styles that needed to be adopted during and after pain management training: "I hope there will be more lectures about pain management which will be not [too] long but good with information [and] time to practice it clinically, such as group discussions".

These findings are supported by the recommendations in the study by Blau and Shamir-Inbal (2017:79), who emphasise using active learning, such as individual reflection with peer feedback, collaboration, self-regulation, and teamwork during training to enhance the transfer of learning.

The participants commented on effective communication:

"Communication barrier is discouraging effort[s] to bring the gap between learning to the real context".

"By culture, people perceive the scales differently, personal experience and what [the] patient said [about] pain is so subjective that can be confusing".

Dithole, Thupayagale-Tsheneagae, Akpor and Moleki (2017:4) alluded to the use of effective communication between nurses and patients as crucial to facilitating quality patient care, including pain assessment.

4.4 PHASE 3

The objectives of Phase 3 were to identify the resources available to conduct a pain assessment, explore the teaching approaches employed during pain management education of nurses, and describe the learning content regarding pain assessment and management as well as the transfer of learning climate within the hospitals' nursing care areas (see Annexure 8).

4.4.1 Demographic characteristics (N = 47)

The demographic characteristics of the respondents (N = 47) entailed gender, age, nationality, highest education qualification, and nursing care areas.

4.4.1.1 Gender of the clinical facilitators (N = 47)

The clinical facilitator respondents were predominantly female (97.87%; n = 46) and only 2,13% (n = 1) male (see Figure 4.9). This is not a strange finding as more female nurses are evident globally (Badu et al. 2019:9). The gender distribution within Saudi Arabia can specifically be attributed to few male nurses graduating and joining the nursing workforce (Punshon et al. 2019:28; Alboliteeh et al. 2017:7).



Figure 4.9 Gender of clinical facilitators (N = 47)

4.4.1.2 Age of clinical facilitators (N = 47)

The respondents' mean age was 41.6 years. The youngest respondent was 29 years old, and the eldest was 61 at the time of data gathering. The age of clinical facilitators is relevant as it can be related to skills mix, relationships, and expertise in clinical facilitation roles to support other clinical nursing staff from a culturally, academically and linguistically diverse group (Bostick, Norman, Sharma, Toxopeus, Irwin & Dhillon 2021: 24; Lin, Del Fabbro, Needham, Sidwell & Shaw 2021:4). The standard deviation

in age was 7.626 and the distribution of clinical facilitators' age was negatively skewed since the mean age was 41, less than the median age of 42 (see Figure 4.10). This suggests that clinical facilitators' age, experiences, and continued professional development are essential in gaining expertise in clinical practice (Thomas and Kellgren 2017:232).



Figure 4.10 Age of clinical facilitators distribution (N = 47)

4.4.1.3 Nationality of clinical facilitators (N = 47)

The respondents' nationalities are as illustrated in Table 4.14.

Table 4.14: Clinical facilitators' nationality (N = 47)

| Nationality | n = | <i>f</i> = % |
|---------------|-----|--------------|
| Filipino | 19 | 40.4 |
| Malaysian | 9 | 19.1 |
| South African | 7 | 14.9 |
| Saudi | 5 | 10.6 |

| Jordanian | 2 | 4.3 |
|-------------|----|-------|
| Czech | 1 | 2.1 |
| Egyptian | 1 | 2.1 |
| Indian | 1 | 2.1 |
| Irish | 1 | 2.1 |
| Singaporean | 1 | 2.1 |
| Total | 47 | 100.0 |

It was important to describe the respondents' nationality to demonstrate the diversity among the clinical facilitators, such as cultural, academic, and linguistic backgrounds that may enhance the transfer of learning of pain management competencies of nurses (Yanaprasasart & Lüdi 2018:835). Table 4.14 highlights the ten (10) diverse nationalities of clinical facilitators who participated in the study. Nineteen (f = 40.4%) were Filipino, 9 were Malaysians (f = 19,1%), 7 were South African (f = 14.9%), 5 (f = 10,6%) were Saudi; 2 (f = 4.3%) were Jordanian, while only one (f = 0.5%) participant was from the Czech Republic, Egypt, India, Ireland and Singapore. Notably, 89.4% (n = 42) of respondents were non-Saudis. The diversity of respondents' nationality characteristics enabled them to apply their clinical skills, education experiences, and communication skills effectively with others from a similar background to enhance the transfer of learning of pain management competencies of nurses (Day & Beard 2019:280).

4.4.1.4 Education Qualifications (N = 47)

Clinical facilitators' role is to facilitate the development of students' critical thinking skills and support those nurses who have to facilitate student learning and who have to apply their knowledge, in this study context, pain management skills to clinical settings (Phillips, Duke & Weerasuriya 2017:4344). Clinical facilitators can undergo different levels and types of education, including pain management training that can be

applied in practice (Suliman & Aljezawi 2018:529). Thus, a clinical facilitator is a nurse who completed a nursing education qualification such as a bachelor's degree in nursing, postgraduate diploma, honours degree, master's degree and or a diploma in nursing (Phillips et al. 2017:4348), and someone with a diverse background and with core competencies in pain management. The qualifications the respondents attained in this study were master's degrees (n = 4; f = 8,5%, Bachelor's degrees (n = 38; f = 80,9%), and Diplomas in nursing (n = 5; f = 10,6%) (see Figure 4.11).



Figure 4.11 Clinical facilitators' Highest Education Qualifications distribution

(N = 47)

Academic qualifications are important to motivate participation in supporting learning and enhancing registered nurses' and students' clinical education, such as pain management relevant to clinical practice (Ogbolu Scrandis & Fitzpatrick 2017:9; Sweet & Broadbent 2017:35). The respondents' highest educational qualifications varied largely between their countries of origin (see Table 4.15).

| Nationality | Total ber | num- | Master | s degree | Bachel gree | or's de- | Diplor | na |
|---------------|--------------|--------------|--------|--------------|----------------|--------------|--------|--------------|
| | n | <i>f</i> = % | n | <i>f</i> = % | n | <i>f</i> = % | n | <i>f</i> = % |
| Filipino | 19 | 40.4 | 1 | 2.1 | 18 | 38.3 | 0 | 0 |
| Malaysia | 9 | 19.1 | 0 | 0 | 8 | 17.0 | 1 | 2.1 |
| South African | 7 | 14.9 | 0 | 0 | 3 | 6.4 | 4 | 8.5 |
| Saudi | 5 | 10.6 | 2 | 4.3 | 3 | 6.4 | 0 | 0 |
| Jordanian | 2 | 4.3 | 0 | 0 | 2 | 4.3 | 0 | 0 |
| Czech | 1 | 2.1 | 1 | 2.1 | 0 | 0 | 0 | 0 |
| Egyptian | 1 | 2.1 | 0 | 0 | 1 | 2.1 | 0 | 0 |
| Indian | 1 | 2.1 | 0 | 0 | 1 | 2.1 | 0 | 0 |
| Irish | 1 | 2.1 | 0 | 0 | 1 | 2.1 | 0 | 0 |
| Singaporean | 1 | 2.1 | 0 | 0 | 1 | 2.1 | 0 | 0 |
| Total | 47 | 100. 0 | 4 | 8.5% | 38 | 80.9% | 5 | 10.6% |

 Table 4.15:
 Country of origin and highest education qualifications (N = 47)

Of the respondents who owned a master's degree, two (f = 4,3%) were originally from Saudi Arabia, and one each from the Czech Republic and the Philippines. Of the bachelor's degree qualified, 18 (f = 38.3%) were from the Philippines; eight (f = 17%) were from Malaysia; three each from Saudi Arabia and South Africa; two (f = 4.3%) from Jordan, while one each from Egypt, India, Ireland, and Singapore. The diploma-prepared participants' origins were four (f = 8.5%) from South Africa and one (f = 2.1%) from Malaysia (see Table 4.15). This finding concurs with the study of Jayasekara, Smith, Hall, Rankin, Smith, Visvanathan and Friebe (2018:122), which found that clinical facilitators in this study context have diverse educational preparedness, enabling them to apply their pain management skills.
4.4.1.5 Nursing wards worked in (N = 47)

The clinical facilitators predominantly perform the role of promoting clinical competency and enhancing the transfer of learning of pain management competencies (Rafii, Ghezeljeh and Nasrollah 2019:1411). Table 4.16 indicates the various nursing wards where the 47 respondents were responsible for facilitation and mentoring.

| Nursing wards | n = | <i>f</i> = % |
|---------------|-----|--------------|
| Medical | 16 | 34.0 |
| Surgical | 15 | 31.9 |
| Paediatric | 7 | 14.9 |
| Obs-gynae | 5 | 10.6 |
| Cardiac | 4 | 8.5 |
| Total | 47 | 100.0 |

 Table 4.16: Clinical facilitators' nursing wards of work (N = 47)

Sixteen (f = 34.0%) respondents were responsible for medical wards, 15 (f = 31,9%) for surgical wards, seven (f = 14,9%) for paediatric wards, five (f = 10.6%) for obsgynae wards, and four (f = 8,5%) for cardiac wards (see Table 4.16). These findings show that most clinical facilitators were in medical and surgical wards, as demonstrated in the study by Rafii et al. (2019:1411).

4.4.2 Resources available to conduct pain assessment (N = 47)

Respondents were also asked which resources were available to conduct pain assessments in the nursing wards for which they are responsible (see Annexure 8, Section B). The same tools discussed earlier (see Section 3.9) are relevant. It was important to redo the exercise by asking the nurse facilitator respondents the same questions (see Annexure 8, Section B) to confirm what the nurse respondents identified as the available tools to assess the pain. The availability of the same tools identified by nurses and the 47 clinical facilitators in this phase is important. It would allow the facilitator to teach nurses how to implement the tool in practice, thus facilitating the transfer of learning about pain management utilising different tools.

Different systematic pain assessment guide tools for pain history taking were assessed for the availability of resources to conduct pain assessments.

4.4.2.1 Systematic pain assessment guides for pain history taking (N = 47)

The 47 respondents reported the availability of the five systematic pain assessment guides, namely (1) WILDA, (2) PQRST, (3) OPQRSTUV, (4) COLDSPA, and (5) QUEST for pain history taking during pain assessment as illustrated in Table 4.17.

| | Nur | sing W | ards | | | | | | | | | | | | | | | | | | | | | |
|----------------------------|-------|--------|------|------|-----|-------|-----|------|--------|----------|--------|------|--------|--------|--------|------|--------|-------|--------|------|-----|------|----|------|
| | Med | lical | | | Sur | gical | | | Pae | ediatrio | ; | | Ob | s-gyna | e | | Ca | rdiac | | | тот | ſAL | | |
| Pain assess- ment guide | n = ' | 16 | | | n = | 15 | | | n = | 7 | | | n = | : 5 | | | n = | 4 | | | N = | 47 | | |
| tools | Yes | | No | | Yes | | No | | Yes | 5 | No | | Ye | S | No | | Yes | 5 | No | | Yes | | No | |
| | F = | % = | F = | % = | F = | % | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | n | % | n | % |
| WILDA | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 6 | 85,7 | 1 | 14,3 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 45 | 95,7 | 2 | 4,3 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| PQRST | 9 | 56,2 | 7 | 43,8 | 13 | 86,7 | 2 | 13,3 | 6 | 85,7 | 1 | 14,3 | 2 | 40,0 | 3 | 60,0 | 3 | 75,0 | 1 | 25,0 | 33 | 70,2 | 14 | 29,8 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| OPQRSTUV pain | 5 | 31,3 | 11 | 68,7 | 4 | 26,7 | 11 | 73,3 | 3 | 42,9 | 4 | 57,1 | 1 | 20,0 | 4 | 80,0 | 3 | 75,0 | 1 | 25,0 | 16 | 34,0 | 31 | 66,0 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| COLDSPA | 5 | 31,3 | 11 | 68,7 | 3 | 20,0 | 12 | 80,0 | 4 | 57,1 | 3 | 42,9 | 1 | 20,0 | 4 | 80,0 | 1 | 25,0 | 3 | 75,0 | 14 | 29,8 | 33 | 70,2 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| QUEST | 4 | 25,0 | 12 | 75,0 | 2 | 13,3 | 13 | 86,7 | 4 | 57,1 | 3 | 42,9 | 1 | 20,0 | 4 | 80,0 | 1 | 25,0 | 3 | 75,0 | 12 | 25,5 | 35 | 74,5 |

Table 4.17: Availability of the pain assessment guides (N = 47)

The nurse facilitator respondents confirmed what the nurse respondents indicated as available (see Table 4.25). WILDA was reported to be available by the most (f = 95,7%; n = 45), followed by PQRST (f = 70,2%; n = 33) availability; OPQRSTUV (f = 34,0%; n = 16), COLDSPA 29,8% (n = 14), and QUEST 25,5% (n = 12) (see Table 4.25). Fink and Gallagher (2019:231) iterated that the WILDA tool was predominantly used to assess pain. Both the clinical facilitator respondents and nurse respondents similarly reported WILDA to be available predominantly and QUEST to be the least available (see Tables 4. 5 and 4.17).

4.4.2.1.1 WILDA assessment guide (N = 47)

WILDA was reported to be 100% available to all respondents in the medical, surgical, and cardiac wards (see Table 4.17). In paediatric wards, it was 85,7% (F = 6; n = 7) available, and in obs-gynae wards, 80,0% (F = 4; n = 5).

4.4.2.1.2 The PQRST pain assessment guide (N = 47)

Thirteen respondents (n = 15, f = 86,7%) reported the **PQRST** tool availability in surgical wards, followed by 6 in paediatric wards (n = 7; f = 85,7%), 3 (n = 4; F = 75,0%) in cardiac wards, 9 in medical wards (n = 16; F = 56,2%), and 2 in obs-gynae wards (n = 5; f = 40.0%) (see Table 4.17).

4.4.2.1.3 The OPQRSTUV pain assessment guide (N = 47)

OPQRSTUV, according to the literature, should be available in paediatric wards (Cash et al. 2019:59, Jufri et al. 2019:13). Three out of the seven respondents responsible for the paediatric wards reported that it was available, three out of the four respondents in cardiac wards; five out of the 16 in medical wards, four out of 15 for surgical wards, and one out of the five in obs-gynae wards (see Table 4.17).

4.4.2.1.4 The COLDSPA pain assessment guide (N = 47)

The **COLDSPA** tool was reported to be available by four (n = 7; f = 57,1%) respondents in paediatric wards, five (n = 16; f = 57,1%) respondents in medical wards, one (n = 4; f = 25,0%) in cardiac wards, three (n = 15; f = 20,0%) in surgical wards as well as one (n = 5; f = 20,0%) in obs-gynae wards (see Table 4.17).

4.4.2.1.5 The QUEST pain assessment guide (N = 47)

Four of the seven clinical facilitators reported that QUEST was available in paediatric wards (see Table 4.17). In medical wards, four out of 16 facilitators indicated the tool to be available, one out of four in cardiac wards, one out of five in obs-gynae wards, and only two out of 15 facilitators in surgical wards (see Table 4.17).

The patients who can self-report their pain were assessed with the identified pain rating assessment tools available in the various wards.

4.4.2.2 Pain rating assessment tools for patients who can self-report their pain (N = 47)

Five valid and reliable pain rating assessment tools were selected for patients who can self-report their pain to determine their availability within the nursing wards. Table 4.18 illustrates the identified pain rating assessment tools reported by respondents to be available, namely (1) the NRS (n = 45), (2) the Wong-Baker FACES pain scale (n = 43), (3) the VAS (n = 11), (4) the VDS (n = 11), and (5) the BPS (n = 9). Studies indicated that NRS is the most suitable pain scale for self-reporting pain assessment for cognitively intact patients (Kang & Demiris 2018:19). Hence, in this study context, it was the highest available tool in the nursing wards (see Table 4.18).

As illustrated in Tables 4.6 and 4.18, the 47 clinical facilitator respondents and 385 nurse respondents provided similar information about the scales available to assess pain from patients who can self-report pain. The two groups of respondents indicated

NRS to be available the most, while the BPI scale was the least available (see Table 4.6 and 4.18). The availability of these scales, as reported by the 47 respondents, is indicated in Table 4.18.

| Pain rating as- | Nu | rsing | Wai | ds | | | | | | | | | | | | | | | | | | | | |
|---------------------------------------|--------|-------|--------|------|--------|--------|--------|------|--------|--------|--------|------|--------|-------|--------|------|--------|-------|--------|------|-----|------|----|------|
| sessment tools | Me | dical | | | Su | rgical | | | Pae | ediatr | ic | | Ob | s-gyn | ae | | Ca | rdiac | | | ТО | TAL | | |
| for patients | n = | : 16 | | | n = | : 15 | | | n = | 7 | | | n = | 5 | | | n = | 4 | | | N = | = 47 | | |
| who can self- | Ye | S | No | | Ye | S | No | | Yes | S | No | | Ye | S | No | | Ye | S | No | | Ye | S | No | |
| pain | F = | % = | F = | % = | F = | % | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | n | % | n | % |
| Numeric Rat- ing Scale (NRS) | 15 | 93,7 | 1 | 6,3 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 45 | 95,7 | 2 | 4,3 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Wong-Baker FACES pain scale | 14 | 87,5 | 2 | 12,5 | 14 | 93,3 | 1 | 6,7 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 43 | 91,5 | 4 | 8,5 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Verbal Ana- logue Scale (VAS) | 3 | 18,8 | 13 | 81,2 | 4 | 26,7 | 11 | 73,3 | 2 | 28,6 | 5 | 71,4 | 1 | 20,0 | 4 | 80,0 | 1 | 25,0 | 3 | 75,0 | 11 | 23,4 | 36 | 76,6 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Verbal De- scriptor Scale (VDS) | 2 | 12,5 | 14 | 87,5 | 4 | 73,3 | 11 | 26,7 | 2 | 28,6 | 5 | 71,4 | 2 | 40,0 | 3 | 60,0 | 1 | 25,0 | 3 | 75,0 | 11 | 23,4 | 36 | 76,6 |

Table 4.18: Availability of pain rating assessment tools for patients who can self-report their pain (N = 47)

| Brief Pain In- | 3 | 18,8 | 13 | 81,2 | 3 | 20,0 | 12 | 80,0 | 1 | 14,3 | 6 | 85,7 | 2 | 40,0 | 3 | 60,0 | 0 | 0 | 4 | 100 | 9 | 19,1 | 38 | 80,9 |
|----------------|---|------|----|------|---|------|----|------|---|------|---|------|---|------|---|------|---|---|---|-----|---|------|----|------|
| ventory (BPI) | | | | | | | | | | | | | | | | | | | | | | | | |

4.4.2.2.1 The Numeric Rating Scale (NRS) (N = 47)

In cardiac, paediatric, and surgical wards, the NRS was reported to have been available to all respondents working in these wards (see Table 4.18). The scale was available in medical wards to 15 out of the 16 respondents and in gynae wards to four out of the five respondents (see Table 4.18). Adams, Murname, Adams, Elfenbein, Chang, Sannon, Gay & and Choudhury (2018:2) indicate the importance of the availability of NRS but also emphasize the role the clinical facilitators play in educating the nurses on how to use the scale.

4.4.2.2.2 The Wong-Baker FACES scale (N = 47)

The Wong-Baker FACES scale, used to rate pain in children who can self-report their pain intensity, was available (see Table 4.18). All respondents from paediatric wards (n = 7) and cardiac wards (n = 4) reported the scale as available. In surgical wards, all but one respondent (n = 15; F = 14) indicated the scale to be available. In medical wards, 14 of the 16 respondents and four of five in obs-gynae reported its availability (see Table 4.18).

4.4.2.2.3 The Verbal Analogue Scale (VAS) (N = 47)

In paediatric wards, two respondents (n = 7) indicated the scale to be available, four in surgical wards (n = 15), one in cardiac wards (n = 4), one in obs-gynae wards (n = 5), and three in medical wards (n = 16) (see Table 4.18).

4.4.2.2.4 The Verbal Descriptor Scale (VDS) (N = 47)

Four of the respondents in surgical wards (n = 15) reported the VDS to be available, two in obs-gynae wards (n = 5), two in paediatric wards (n = 7), one in cardiac wards (n = 4), and two in medical wards (n = 16) reported availability (see Table 4.18).

4.4.2.2.5 The Brief Pain Inventory (BPI) tool (N = 47)

BPI is commonly available to assess pain within wards caring for patients with cancer (Kang & Demiris 2018:8). In this study, two out of the five facilitators responsible for obs-gynae wards reported its availability, three out of 15 in surgical wards, three out of 16 in medical wards, and one out of 7 in paediatric wards (see Table 4.18).

Some patients cannot provide a self-report of pain verbally; hence, the clinical facilitators were asked to indicate the availability of the pain rating assessment tools for patients who cannot self-report their pain.

4.4.2.3 Pain rating assessment tools for patients who cannot self-report their pain (N = 47)

The seven existing nonverbal pain assessment tools considered the most valid and reliable pain rating assessment tools for patients who could not self-report their pain were to be identified for availability within wards. Of the 47 respondents, the pain rating assessment tools identified to be used to rate pain for patients who cannot self-report their pain were (1) the FLACC pain scale (n = 45), (2) the CRIES pain scale (n = 36), (3) BPS (n = 17), (4) NIPS (n = 13), (5) CPOT (n = 12), (6) COMFORT-B pain scale (n = 11), and (7) N-PASS (n = 11). As illustrated in Tables 4.7 and 4.19, the facilitator respondents and nurse respondents indicated the FLACC Scale and CRIES to be the most available, while both participant groups confirmed NPASS to be the least available. Regarding the availability of BPS, NIPS, CPOT, and COMFORT-B, the two groups of respondents supplied diverse opinions (see Tables 4.7 and 4.19).

| Pain rating | Nur | sing V | Vard | S | | | | | | | | | | | | | | | | | | | | |
|------------------------|-----|--------|------|------|-----|-------|-----|------|-----|---------|-----|------|-----|--------|-----|------|-----|------|-----|------|-----|------|----|------|
| assess- | Med | dical | | | Sur | gical | | | Pae | diatrio | 2 | | Obs | s-gyna | ae | | Car | diac | | | TO | TAL | | |
| for patients | n = | 16 | | | n = | 15 | | | n = | 7 | | | n = | 5 | | | n = | 4 | | | N = | 47 | | |
| who cannot self-report | Yes | ; | No | | Yes | i. | No | | Yes | ; | No | | Yes | ; | No | | Yes | | No | | Yes | 5 | No | |
| their pain | F = | % = | F = | % = | F = | % | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | n | % | n | % |
| FLACC pain scale | 15 | 93,8 | 1 | 6,2 | 15 | 100 | 0 | 0 | 6 | 85,7 | 1 | 14,3 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 45 | 95,7 | 2 | 4,3 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| CRIES pain scale | 9 | 56,2 | 7 | 43,8 | 13 | 86,7 | 2 | 13,3 | 6 | 85,7 | 1 | 12,3 | 5 | 100 | 0 | 0 | 3 | 75,0 | 1 | 25,0 | 36 | 76,6 | 11 | 23,4 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| BPS | 5 | 31,3 | 11 | 68,7 | 5 | 33,3 | 10 | 66,7 | 2 | 28,6 | 5 | 71,4 | 2 | 40,0 | 3 | 60,0 | 3 | 75,0 | 1 | 25,0 | 17 | 36,2 | 30 | 63,8 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| NIPS | 4 | 25,0 | 12 | 75,0 | 3 | 20,0 | 12 | 80,0 | 3 | 42,9 | 4 | 57,1 | 2 | 40,0 | 3 | 60,0 | 1 | 25,0 | 3 | 75,0 | 13 | 27,7 | 34 | 72,3 |
| | | | | | | | | | | | | | | | | | | | | | | | | |

 Table 4.19: Availability of pain rating assessment tools for patients who cannot self-report their pain (N = 47)

| СРОТ | 2 | 12,5 | 14 | 87,5 | 4 | 26,7 | 11 | 73,3 | 3 | 42,9 | 4 | 57,1 | 1 | 20,0 | 4 | 80,0 | 2 | 50,0 | 2 | 50,0 | 12 | 25,5 | 35 | 74,5 |
|------------------------------|---|------|----|------|---|------|----|------|---|------|---|------|---|------|---|------|---|------|---|------|----|------|----|------|
| | | | | | | | | | | | | | | | | | | | | | | | | |
| COM- FORT-B pain scale | 2 | 12,5 | 14 | 87,5 | 4 | 26,7 | 11 | 73,3 | 3 | 42,9 | 4 | 57,1 | 2 | 40,0 | 3 | 60,0 | 0 | 0 | 4 | 100 | 11 | 23,4 | 36 | 76,6 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| N-PASS | 3 | 18,7 | 13 | 81,3 | 2 | 13,3 | 13 | 86,7 | 2 | 28,6 | 5 | 71,4 | 2 | 40,0 | 3 | 60,0 | 2 | 50,0 | 2 | 50,0 | 11 | 23,4 | 36 | 76,6 |

4.4.2.3.1 FLACC pain scale (N = 47)

The FLACC pain scale was reported to be available by all respondents in the surgical (n = 15), obs-gynae (n = 5), and cardiac wards (n = 4) (see Table 4.19). Fifteen of the 16 respondents responsible for the medical wards indicated it to be available, and six out of seven indicated it to be available in paediatric wards (see Table 4.19).

4.4.2.3.2 CRIES pain scale (N = 47)

All respondents from obs-gynae wards (n = 5) reported the CRIES pain scale as available. In surgical wards, 13 respondents (n = 15) indicated the scale to be available, six in paediatric wards (n = 7), three in cardiac wards (n = 4), and nine in medical wards (n = 16) (see Table 4.19).

4.4.2.3.3 The Behavioural Pain Scale (BPS) (N = 47)

The BPS was reported to be available by three out of the four respondents in cardiac wards, two out of five in obs-gynae wards, five out of 15 in surgical wards, five out of 16 in medical wards, and two out of 7 in paediatric wards (see Table 4.19). This finding suggests that the BPS was less reported to be available as the scale was intended to be available in intensive care units (Emsden et al. 2020:12).

4.4.2.3.4 The Neonatal Infant Pain Scale (NIPS) (N = 47)

The NIPS is intended to be available within neonatal intensive care units (Egede, Valstar, Torres & Sharkey 2019:471). Table 4.19 illustrates the availability of NIPS as reported by respondents. The scale was reported to be available by three out of the seven respondents from paediatric wards, two out of five from obs-gynae wards, four out of 16 from medical wards, one out of four from cardiac wards, and three out of 15 from surgical wards.

4.4.2.3.5 Critical Care Pain Observational (CPOT) (N = 47)

The CPOT scale is meant to be available within intensive care units (Emsden et al. 2020:12; Kotfis et al. 2017:70). In this study context, two respondents in cardiac wards (n = 4) indicated the scale to be available, three in paediatric wards (n = 7), four in surgical wards (n = 15), one in obs-gynae (n = 5), and two in medical wards (n = 16) (see Table 4.19).

4.4.2.3.6 The COMFORT-B pain scale (N = 47)

In paediatric wards, three out of the seven respondents reported the COMFORT-B pain scale to have been available, two out of five in obs-gynae wards, four out of 15 in surgical wards, and two out of 16 in medical wards (see Table 4.19). Studies demonstrated that the COMFORT-B is predominantly available within paediatric wards (Saelim, Chavananon, Ruangnapa, Prasertsan & Anuntaseree 2019:157). In this study's findings, it was primarily indicated to be available by respondents in paediatric wards (see Table 4.19).

4.4.2.3.7 The N-PASS (N = 47)

The N-PASS was the least reported to be available, as portrayed in Table 4.19, and two out of the four facilitators responsible for cardiac wards reported it to be available, two out of five in obs-gynae wards, two out of 7 in paediatric wards, three out of 16 in medical wards, and two out 15 in surgical wards.

4.4.2.4 Pain rating assessment tools for elderly patients with dementia or cognitive impairment to rate pain (N = 47)

For elderly patients with dementia or cognitive impairment, the clinical facilitators reported the availability of the four identified valid and reliable tools, as illustrated in Table 4.20. The pain rating assessment tools used for the elderly patients with dementia or cognitive impairment identified to be available were (1) the CNPI (f = 29,8%; n =

14), (2) the NOPPAIN (f = 14,9%; n = 7), (3) the ABBEY (f = 12.8%; n = 6), and (4) the PAINAD pain scale (f = 12%; n = 6) (see Table 4.20). As indicated in Tables 4.8 and 4.20, the nurse facilitator respondents agreed with the nurse respondents as they furnished similar information about the CNPI availability to assess the pain of elderly patients with dementia. The two groups, however, provided different opinions about the availability of the other three remaining scales (NOPPAIN, ABBEY, and PAINAD) (see Tables 4.8 and 4.20). Amongst the two groups of respondents, nurse facilitators provided information that indicated the PAINAD tool to be the least available, while the ABBEY tool was the least available according to the nurse respondents (see Tables 4.8 and 4.20). The nurse facilitator respondents reported the tools' availability in the different wards as illustrated in Table 4.20.

Table 4.20: Availability of the pain rating assessment tools for elderly patients with dementia or cognitive impairment (N =

| Pain rating assess- | Nu | rsing \ | Nard | S | | | | | | | | | | | | | | | | | | | | |
|---------------------------|--------|---------|------|------|--------|--------|-----|------|--------|--------|--------|-----|--------|--------|--------|------|--------|-------|--------|------|-----|------|----|------|
| ment tools for elderly | Me | dical | | | Su | rgical | | | Ра | edia | tric | | Ob | s-gyna | ae | | Ca | rdiac | | | тот | AL | | |
| patients | n = | : 16 | | | n = | : 15 | | | n = | : 7 | | | n = | : 5 | | | n = | - 4 | | | N = | 47 | | |
| with de- mentia or | Ye | S | No | | Ye | S | No | | Ye | s | No | | Ye | S | No | | Ye | S | No | | Yes | | No | |
| cognitive impairment | F = | % = | F = | % = | F = | % | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | n | % | n | % |
| CNPI | 6 | 37,5 | 10 | 62,5 | 5 | 33,3 | 10 | 66,7 | 0 | 0 | 7 | 100 | 1 | 20,0 | 4 | 80,0 | 2 | 50,0 | 2 | 50,0 | 14 | 29,8 | 33 | 70,2 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| NOPPAIN | 3 | 18,7 | 13 | 81,3 | 2 | 13,3 | 13 | 86,7 | 0 | 0 | 7 | 100 | 1 | 20,0 | 4 | 40,0 | 1 | 25,0 | 3 | 75,0 | 7 | 14,9 | 40 | 85,1 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| ABBEY pain scale | 2 | 12,5 | 14 | 87,5 | 2 | 13,3 | 13 | 86,7 | 0 | 0 | 7 | 100 | 1 | 20,0 | 4 | 80,0 | 1 | 25,0 | 3 | 75,0 | 6 | 12,8 | 41 | 87,2 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| PAINAD | 2 | 12,5 | 14 | 87,5 | 2 | 13,3 | 13 | 86,7 | 0 | 0 | 7 | 100 | 1 | 20,0 | 4 | 80,0 | 1 | 25,0 | 3 | 75,0 | 6 | 12,8 | 41 | 87,2 |

4.4.2.4.1 The CNPI (N = 47)

Two of the four respondents reported that the CNPI is available in cardiac wards, six out of 16 in medical wards, and five out of 15 in surgical wards (see Table 4.20). None of the respondents in paediatric wards indicated the CNPI tool's availability, probably due to the reasons indicated by literature that rationalize its availability within wards caring for elderly patients (Luks, Hagg-Gün, Mayer, Fisher & Schuler 2019:747).

4.4.2.4.2 The NOPPAIN tool (N = 47)

Table 4.20 shows that in cardiac wards, only one out of the four respondents reported NOPPAIN to be available (see Table 4.20). In obs-gynae wards, one out of the five respondents indicated the tool's availability, three out of 16 in medical wards, and two out of 15 in surgical wards (see Table 4.20). The scale is meant to be used for elderly patients with dementia or cognitive impairment (Rababa 2018:65). However, this study found that none of the respondents in paediatric wards reported its availability.

4.4.2.4.3 The ABBEY pain scale (N = 47)

In cardiac wards, one out of the four respondents reported the ABBEY pain scale to be available, one out of five in obs-gynae wards, two out of 16 in medical wards, and two out of the 15 in surgical wards (see Table 4.20).

4.4.2.4.4 The PAINAD scale (N = 47)

One respondent (n = 4) in cardiac reported the PAINAD scale to be available, one in obs-gynae wards (n = 5), two in medical wards (n = 16), and two in surgical wards (n = 15) (see Table 4.20). None of the respondents indicated the availability of the tool in paediatric wards, an obvious rationale since the tool is used to assess pain for advanced dementia patients (Lukas et al. 2019:742).

The availability of human resources essential to assess pain to enhance the transfer of learning of pain management was identified.

4.4.2.5 Human resources (N = 47)

Table 4.9 portrays the data received from the 385 nurse respondents, while Table 4.21 illustrates that of the 47 clinical facilitators relating to the availability of human resources (see Annexure 8). The two groups of respondents indicated different opinions on what was most available. This is a concern as within the very same wards, the two groups of participants perceived the availability of resources differently (see Tables 4.9 and 4.21). The different "types" of human resources identified to be available to conduct pain assessment were (1) the pain nurses working in acute or chronic pain services (n = 47), (2) pain nurse specialists (n = 45), and (3) the registered nurses with pain management training (n = 44), (4) the clinical facilitators (n = 44), (5) the pain management physicians (n = 44), (6) the ward nurse managers with pain management training (n = 44), (7) the nurse educators (n = 42), and (8) the nurse supervisors with pain management training (n = 39) (see Table 4.21). In Tables 4.9 and 4.21, the nurse facilitator respondents confirmed similarly to the nurse respondents as they indicated that nurse supervisors were least available. The clinical facilitator respondents indicated pain nurses working in acute or chronic pain services to be available most, contrary to nurse respondents who indicated that registered nurses with pain management training would be available the most. The availability of the nurse supervisors with pain management training information is similar according to the two groups of participants. Both groups said they were the least available (see Tables 4.9 and 4.21).

209

| Table 4.21: Availability of the human reso | urces (N = 47) | |
|--|----------------|--|
|--|----------------|--|

| | Nursing W | /ards | | | | | | | | | | | | | | | | | | | | | | |
|--|-----------|-------|----|------|-----|--------|----|------|----|---------|----|------|-----|--------|-----|------|-----|--------|----|------|-----|------|----|----------|
| | Medical | | | | Su | rgical | | | Pa | aediatr | ic | | 0 | bs-gyr | nae | | Ca | ardiac | | | то | TAL | | |
| Human Resources | n = 16 | | | | n = | 15 | | | n | = 7 | | | n : | = 5 | | | n : | = 4 | | | N = | : 47 | | |
| | Yes | | No | C | Yes | S | No |) | Ye | es | No |) | Ye | es | No |) | Υe | es | No |) | Yes | 6 | Nc | > |
| | F | % | F | % = | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n | % | n | % |
| Pain Nurses working in acute or chronic pain services | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100 | 0 | 0 |
| Pain Nurse specialists | 15 | 93,8 | 1 | 6,2 | 15 | 100 | 0 | 0 | 6 | 85,7 | 1 | 14,3 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 45 | 95,7 | 2 | 4,3 |
| Registered nurse with pain manage- ment training | 15 | 93,8 | 1 | 6,2 | 14 | 93,3 | 1 | 6,7 | 6 | 85,7 | 1 | 14,3 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 44 | 93,6 | 3 | 6,4 |
| Clinical facilitators | 15 | 93,8 | 1 | 6,2 | 14 | 93,3 | 1 | 6,7 | 6 | 85,7 | 1 | 14,3 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 44 | 93,6 | 3 | 6,4 |
| Pain management physicians | 15 | 93,8 | 1 | 6,2 | 15 | 100 | 0 | 0 | 6 | 85,7 | 1 | 14,3 | 5 | 100 | 0 | 0 | 3 | 75,0 | 1 | 25,0 | 44 | 93,6 | 3 | 6,4 |
| Ward Nurse managers with pain man- agement training | 15 | 93,8 | 1 | 6,2 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 44 | 93,6 | 3 | 6,4 |
| Nurse educators | 14 | 87,5 | 2 | 12,5 | 13 | 86,7 | 2 | 13,3 | 5 | 71,4 | 2 | 28,6 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 41 | 87,2 | 6 | 12, 8 |
| Nurse supervisors with pain manage- ment training | 15 | 93,8 | 1 | 6,2 | 12 | 80,0 | 3 | 20,0 | 5 | 71,4 | 2 | 28,6 | 4 | 80,0 | 1 | 20,0 | 3 | 75,0 | 1 | 25,0 | 39 | 83,0 | 8 | 17, 0 |

4.4.2.5.1 Pain nurses working in acute or chronic pain services (N = 47)

The acute and chronic pain team nurses are well known for their collaborative work within wards (Rockett et al. 2017:1239). All the respondents who participated in this study reported these nurses to have been available (see Table 4.21).

4.4.2.5.2 Pain nurse specialists (N = 47)

All respondents who participated reported that pain nurse specialists were available, except in medical wards, as all but one respondent (n = 16; F = 15) indicated that pain nurse specialists were available, and six out of the seven respondents were in paediatric wards (see Table 4.21).

4.4.2.5.3 Registered nurses with pain management training (N = 47)

All respondents in obs-gynae wards (n = 5) and cardiac wards (n = 4) reported that registered nurses with pain management training were available (see Table 4.21). However, in medical wards, 15 out of 16 of the respondents, in surgical wards, 14 out of 15, and in paediatric wards, six out of seven indicated these nurses were available (see Table 4.21).

4.4.2.5.4 The clinical facilitators (N = 47)

All respondents in all the wards reported the clinical facilitators to be available, except for in the medical wards (F = 15; n = 16), the surgical wards (F = 14; n = 15), and the paediatric wards (F = 6; n = 7) (see Table 4.21).

4.4.2.5.5 The pain management physicians (N = 47)

All respondents in surgical and obs-gynae wards indicated pain management physicians to be available. In the medical wards (F = 15; n = 16), paediatric wards (F = 6; n

= 7), and cardiac wards (F = 3; n = 4), they were available as indicated (see Table 4.21).

4.4.2.5.6 The nurse managers with pain management training (N = 47)

The nurse managers with training in pain management were reported to be available by 15 respondents in medical wards (f = 93,8%; n = 16) and by four in obs-gynae wards (f = 80,0%; n = 5). All surgical, paediatric, and cardiac ward respondents indicated these nurse managers to be available (see Table 4.21).

4.4.2.5.7 Nurse educators (N = 47)

Another role player in pain management facilitation is the nurse educator in an organisation who plays a role in educating nurses on how to conduct pain assessments, collaborating in this regard with clinical facilitators (Shoqirat, Mahasneh, Singh & Hadid 2019:6). As portrayed in Table 4.21, all of the respondents in obs-gynae and cardiac wards reported that nurse educators were available. In the medical wards (F = 14; n =16), surgical wards (F = 13; n = 15), and paediatric wards (F = 5; n = 7), the nurse educators were available, although not all participants indicated them to be available (see Table 4.21).

4.4.2.5.8 Nurse supervisors with pain management training (N = 47)

Table 4.21 indicates that in medical wards, all respondents but one (F = 15; n = 16) reported nurse supervisors with pain management training available, 12 out of the 15 respondents in surgical wards, four out of the five respondents in obs-gynae wards, three out of the four respondents in cardiac wards, and five out of the seven respondents in paediatric wards reported availability (see Table 4.21).

Other types of support are available for the patients to manage their pain; hence, the clinical facilitators were asked to indicate their availability.

| | - | | | | | | | | | | | | | | | | | | | | | | | |
|---|-------|---------|-----|------|-------|-------|----|------|-----|----------|----|------|-----|---------|----------|------|-----|-------|----|------|-------|------|----|------|
| | Nurs | sing Wa | rds | | | | | | | | | | | | | | | | | | | | | |
| | Med | ical | | | Surg | gical | | | Pa | ediatric | | | Ob | s-gynae | ; | | Ca | rdiac | | | тот | AL | | |
| Patient sup- ports | n = ′ | 16 | | | n = 1 | 15 | | | n = | 7 | | | n = | 5 | | | n = | 4 | | | N = - | 47 | | |
| | Yes | | No | | Yes | | No | | Ye | S | No | | Yes | 6 | No | | Yes | 6 | No | | Yes | | No | |
| | F | % | F | % = | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n | % | n | % |
| Patients or (other people or patients with pain) | 12 | 75,0 | 4 | 25,0 | 11 | 73,3 | 4 | 26,7 | 5 | 71,4 | 2 | 28,6 | 2 | 40,0 | 3 | 60,0 | 3 | 75,0 | 1 | 25,0 | 33 | 70,2 | 14 | 29,8 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Pain man- agement sup- port groups | 10 | 62,5 | 6 | 37,5 | 10 | 66,7 | 5 | 33,3 | 5 | 71,4 | 2 | 28,6 | 1 | 20,0 | 4 | 80,0 | 2 | 50,0 | 2 | 50,0 | 28 | 59,6 | 19 | 40,4 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Patient pain management websites | 8 | 50,0 | 8 | 50,0 | 6 | 40,0 | 9 | 60,0 | 4 | 57,1 | 3 | 42,9 | 1 | 20,0 | 4 | 80,0 | 4 | 100 | 0 | 0 | 19 | 40,4 | 28 | 59,6 |

Table 4.22: Availability of patient support (N = 47)

| Patient pain management hotlines | 6 | 37,5 | 10 | 62,5 | 6 | 40,0 | 9 | 60,0 | 4 | 57,1 | 3 | 42,9 | 1 | 20,0 | 4 | 80,0 | 4 | 100 | 0 | 0 | 17 | 36,2 | 30 | 63,8 |
|--|---|------|----|------|---|------|---|------|---|------|---|------|---|------|---|------|---|-----|---|---|----|------|----|------|

4.4.2.6 Other types of support (N=47)

4.4.2.6.1 The "peer" patients (other people or patients with pain) (N = 47)

Peer patients or other people with pain were reported to be available to support patients within wards, as illustrated in Table 4.22. In medical wards, 12 out of the 16 respondents reported peer patients to be available, three out of four in cardiac wards, 11 out of 15 in surgical wards, five out of seven in paediatric wards, and two out of the five in obs-gynae wards (see Table 4.22).

4.4.2.6.2 Pain management support groups (N = 47)

Five (n = 7) respondents in paediatric wards reported support groups to be available, 10 (n = 15) respondents in surgical wards, 10 (n = 16) respondents in medical wards, two (n = 4) respondents in cardiac wards, and one (n = 5) respondent in obs-gynae wards (see Table 4.22).

4.4.2.6.3 Pain management websites (N = 47)

Pain management websites were reported to be available by all respondents in cardiac wards (see Table 4.22). In paediatric wards, four respondents (n = 7) reported the websites to be available, eight in medical wards (n = 16), six in surgical wards (n = 15), and all but one respondent in obs-gynae wards (n = 5) (see Table 4.22).

4.4.2.6.4 Pain management hotlines (N = 47)

All the respondents (n = 4) in cardiac wards reported that patient pain management hotlines were available (see Table 4.22). Four out of the seven respondents in paediatric wards reported the hotlines to be available, six out of 15 in surgical wards, six out of 16 in medical wards, and one out of five in obs-gynae wards (see Table 4.22).

4.4.2.7 Publications and electronic resources (N = 47)

Publications and electronic resources which are essential to be employed during pain assessments were reported by respondents to be available, as illustrated in Table 4.23. Respondents within the wards identified the following publications and electronic resources to be utilised during pain assessment: (1) e-learning modules (n = 45), (2) the best clinical practice guidelines for pain assessment (n = 44), (3) electronic flow sheets (n = 43), (4) clinical updates or journals 72,3% (n = 34), (5) the pain toolkit (n = 33), (6) printed reference books (n = 27), (7) the fact sheets (n = 19), (8) the videos on pain management 38,3% (n = 18), and (9) the e-newsletters (n = 14). The two groups of respondents, namely the nurse facilitator respondents and the nurse respondents, both reported e-learning modules to be available the most, with e-newsletters the least available (compare Tables 4.11 and 4.23).

The availability of the publications and electronic resources in different wards, as identified by the facilitator respondents, is illustrated in Table 4.23.

| Publications and electronic resources | Nurs | sing Wa | rds | | | | | | | | | | | | | | | | | | | | | |
|--|-------|---------|-----|------|-------|--------------------|----|------|-----|----------|----|------|-----------|------|----|------|-----|-------|----|-------|------|------|-----|------|
| | Med | lical | | | Surg | Surgical n = 15 | | | | ediatric | | | Obs-gynae | | | | | rdiac | | | ΤΟΤΑ | L | | |
| | n = ' | 16 | | | n = ' | | | | | n = 7 | | | | = 5 | | n = | : 4 | | | N = 4 | | | | |
| | Yes | | No | No | | | No | | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | |
| | F | % | F | % = | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | N = | % | n = | % |
| E-learning modules | 15 | 93,8 | 1 | 6,3 | 15 | 100 | 0 | 0 | 6 | 85,7 | 1 | 14,3 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 45 | 95,7 | 2 | 4,3 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Best clinical practice guide- lines for pain assessment | 15 | 93,8 | 1 | 6,3 | 13 | 86,7 | 2 | 13,3 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 44 | 93,6 | 3 | 6,4 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Electronic flow sheets | 14 | 87,5 | 2 | 12,5 | 14 | 93,3 | 1 | 6,7 | 6 | 85,7 | 1 | 14,3 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 43 | 91,5 | 4 | 8,5 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Clinical updates or journals | 8 | 50,0 | 8 | 50,0 | 14 | 93,3 | 1 | 6,7 | 5 | 71,4 | 2 | 28,6 | 4 | 80,0 | 1 | 20,0 | 3 | 75,0 | 1 | 25,0 | 34 | 72,3 | 13 | 27,7 |

Table 4.23: Availability of the publications and electronic resources (N = 47)

| Pain toolkit | 9 | 56,3 | 7 | 43,7 | 12 | 80,0 | 3 | 20,0 | 5 | 71,4 | 2 | 28,6 | 4 | 80,0 | 1 | 20,0 | 3 | 75,0 | 1 | 25,0 | 33 | 70,2 | 14 | 29,8 |
|------------------------------|---|------|----|------|----|------|---|------|---|------|---|------|---|------|---|------|---|------|---|------|----|------|----|------|
| Printed refer- ence books | 8 | 50,0 | 8 | 50,0 | 12 | 80,0 | 3 | 20,0 | 4 | 57,1 | 3 | 42,9 | 2 | 49,0 | 3 | 60,0 | 1 | 25,0 | 3 | 75,0 | 27 | 57,4 | 20 | 42,6 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Fact sheets | 5 | 31,3 | 11 | 68,7 | 9 | 60,0 | 6 | 40,0 | 2 | 28,6 | 5 | 71,4 | 1 | 20,0 | 4 | 80,0 | 2 | 50,0 | 2 | 50,0 | 19 | 40,4 | 28 | 59,6 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Videos on pain management | 5 | 31,3 | 11 | 68,7 | 8 | 53,3 | 7 | 46,7 | 3 | 42,9 | 4 | 57,1 | 2 | 40,0 | 3 | 60,0 | 0 | 0 | 4 | 100 | 18 | 38,3 | 29 | 61,7 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| E-newsletters | 2 | 12,5 | 14 | 87,5 | 6 | 40,0 | 9 | 60,0 | 4 | 57,1 | 3 | 42,9 | 1 | 20,0 | 4 | 80,0 | 1 | 25,0 | 3 | 75,0 | 14 | 29,8 | 33 | 70,2 |

4.4.2.7.1 *E-learning modules (N = 47)*

The e-learning modules were reported to be available by all respondents in all the wards, except for the medical wards (F = 15; n = 16) and the paediatric wards (F = 6; n = 7) (see Table 4.23).

4.4.2.7.2 Best clinical practice guidelines for pain assessment (N = 47)

In Table 4.23 it is illustrated that all the respondents in paediatric wards, obs-gynae wards and cardiac wards indicated that the best clinical practice guidelines were available, except for those working in medical wards (F = 5; n = 16) and surgical wards (F = 13; n = 15).

4.4.2.7.3 Electronic flow sheets (N = 47)

The electronic flow sheets were reported to be available by all respondents in other wards (obs-gynae and cardiac), but some disagreed, that is, 14 in medical wards (f = 87,5%; n = 16), and 14 in surgical wards (f = 93,3%; n = 15), and six in paediatric wards (f = 91,5%; n = 7) see Table 4.23.

4.4.2.7.4 Clinical updates or journals (N = 47)

Fourteen out of the 15 facilitators responsible for surgical wards reported clinical updates or journals to be available. Four out of the five facilitators were responsible for obs-gynae wards, three out of four in cardiac wards, five out of seven in paediatric wards, and eight out of 16 facilitators were responsible for medical wards (see Table 4.23).

4.4.2.7.5 Pain toolkit (N = 47)

Four out of the five respondents from obs-gynae wards reported that the pain toolkit was available, and it was the same with 12 out of the 15 respondents from surgical

wards, three out of four from cardiac wards, five out of seven from paediatric wards, and nine out of the 16 respondents from medical wards (see Table 4.23).

4.4.2.7.6 Printed reference books (N = 47)

Table 4.23 illustrates that 12 respondents from surgical wards (n = 15) reported the books to be available, four respondents from paediatric wards (n = 7), eight respondents from medical wards (n = 16), two respondents from obs-gynae wards (n = 5), and one respondent from cardiac wards (n = 4) reported the books to be available (see Table 4.23).

4.4.2.7.7 Fact sheets (N = 47)

In surgical wards, nine respondents (n = 15) indicated the sheets to be available, two in cardiac wards (n = 4), five in medical wards (n = 16), two in paediatric (n = 7), and one in obs-gynae wards (n = 5) (see Table 4.23).

4.4.2.7.8 Videos on pain management (N = 47)

Eight out of the 15 respondents from surgical wards reported the videos to be available, three out of the seven respondents from paediatric wards, two out of five in obsgynae wards, and five out of 16 in medical wards. None of the respondents in paediatric wards indicated the availability of the videos (see Table 4.23).

4.4.2.7.9 *E-newsletters* (*N* = 47)

Respondents indicated e-newsletters to be available in paediatric wards (f = 57,1%; F = 4; n = 7), in surgical wards (f = 40%; F = 6; n = 15), in cardiac wards (f = 25,0%; F = 1; n = 4), in obs-gynae wards (f = 20,0%; F = 1; n = 5), and in medical wards (f = 12,5%; F = 2; n = 16) (see Table 4.23).

The clinical facilitators also identified and indicated the available organisations that specialise in pain management.

4.4.2.8 Organisations that specialise in pain management (N = 47).

Of the 47 clinical facilitator respondents, 37 (f = 78,7%) within the wards reported the availability of organisations specialising in pain management (see Table 4.24). The two groups of respondents, namely the nurse facilitator respondents (f = 78,7%; n = 37) and nurse respondents (f = 78,4%; n = 302), both indicated similar thoughts on the availability of organisations that specialise in pain management (compare Tables 4.12 and 4.24). In surgical wards, 13 of the 15 respondents (n = 5; F = 4) indicated the organisations to be available, 12 out of 16 in medical wards, three out of four in cardiac wards, and five out of seven in paediatric wards (see Table 4.24).

| Organisa- tions that | Nur | Nursing Wards | | | | | | | | | | | | | | | | | | | | | | | | |
|-------------------------|--------|---------------|---|------|----------|------|---|------|-----|---------|---|------|-----------|------|---|------|-----|-------|---|------|--------|------|-----|------|--|--|
| specialise in | Ме | dical | | | Surgical | | | | Ра | ediatri | с | | Obs-gynae | | | | | rdiac | | | TOTAL | | | | | |
| search, | n = | 16 | 1 | | n = 15 | | | | n = | - 7 | | | n = | : 5 | | | n = | : 4 | | | N = 47 | | | | | |
| treatment, clinical | Yes No | | | Yes | | No | | Yes | | No | | Yes | | No | | Ye | S | No | | Yes | | No | | | | |
| practice, | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n = | % | n = | % | | |
| tion | 12 | 75,0 | 4 | 25,0 | 13 | 86,7 | 2 | 13,3 | 5 | 71,4 | 2 | 28,6 | 4 | 80,0 | 1 | 20,0 | 3 | 75,0 | 1 | 25,0 | 37 | 78,7 | 10 | 21,3 | | |

Table 4.24: Availability of the organisations that specialise in pain management (N = 47)

4.4.2.9 Policies for pain management (N = 47)

The respondents indicated their awareness of the availability of policies for pain management, as indicated in Table 4.25. Fourty-six (f = 97,9%) of the respondents reported these policies as available (see Table 4.25). This information about the available policies for pain management confirmed what 384 (f = 99,7%) nurse respondents reported, as illustrated in Table 4.13. Table 4.25 illustrates that all respondents from surgical wards (n = 15), paediatric wards (n = 7), obs-gynae wards (n = 5), and cardiac wards (n = 4) reported the policies' availability. In medical wards, 15 out of the 16 respondents indicated the policies to be available (see Table 4.25).

| | Nur | sing W | ards | ; | | | | | | | | | | | | | | | | | | | | | |
|----------|--------|--------|------|-----|------|----------|---|-----|---|---------|----|----|---------|-------|---|-----|------|-------|---|-------|--------|------|-----|-----|--|
| Policios | Мес | dical | | | Surç | Surgical | | | | diatric | | Ob | s-gynae | è | | Car | diac | | | TOTAL | | | | | |
| | n = | 16 | | | n = | n = 15 | | | | n = 7 | | | | = 5 | | | n = | 4 | | | N = 47 | | | | |
| r uncles | Yes No | | | Yes | | No | | Yes | | Nc | No | | s | No | | Yes | | No | | Yes | | No | | | |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n = | % | n = | % | |
| | 15 | 93,8 | 1 | 6,2 | 15 | 100,0 | 0 | 0 | 7 | 100,0 | 0 | 0 | 5 | 100,0 | 0 | 0 | 4 | 100,0 | 0 | 0 | 46 | 97,9 | 1 | 2,1 | |

Table 4.25: Availability of the policies for pain management (N = 47)

The respondents confirmed the teaching approaches they employed in the pain management education of the nurses.

4.4.3 Teaching approaches employed in pain management education (N = 47)

Teaching approaches are sets of principles, beliefs, or ideas about the nature of learning which are translated into the classroom (Experts 2020:498). Exploring the teaching approaches or strategies employed by the clinical facilitators in pain management education of nurses forms the foundation for understanding their teaching styles, abilities, and knowledge levels (Herrman 2020:7). To accommodate all types of learning styles of nurses, the clinical facilitators must employ different teaching strategies or approaches in pain management education to enhance the transfer of learning. Table 4.26 illustrates the teaching approaches reported to be employed by the respondents in the pain management education of nurses. Of the 47 respondents, the teaching approaches identified as being utilised during training, were (1) pain management elearning modules (n = 46), (2) assessment of prior knowledge about pain management as a basis for each individual teaching occasion (n = 45), (3) engaging hands-on activities to learn about pain assessment (n = 42), (4) using practice simulations to assess registered nurses' skills about pain assessment (n = 41), (5) allowing debriefing sessions about pain management after assessing individual nurse's learning needs which each was reported as requiring (n = 41), (6) forming a discussion group to learn about pain management (n = 40), (7) preventing interruptions during teaching sessions (n = 40), (8) providing assignments about pain management to apply at work (n = 40), (9) the use of role modelling to learn about pain assessment (n = 38), (10) the use of case studies to learn how to manage pain (n = 38), (11) the use of role play activities to learn about pain assessment (n = 37), (12) engaging nurses in focus groups to learn about pain management (n = 36), (13) using nursing grand rounds to learn directly at patient bedside how to assess and manage individual patient's pain (n = 35), and (14) asking nurses to write a reflective journal about pain management (n = 17) (see Table 4.26).

The respondents indicated the employment of the teaching approaches in various wards, as illustrated in Table 4.26.

| | Nursing Wards | | | | | | | | | | | | | | | | | | | | | | | | |
|---|---------------|--------|---|------|-----|--------|---|------|----|--------|----|----|-----------|-----|---|----|---|-------|----|------|--------|------|----|------|--|
| Teaching approaches | Ме | dical | | | Sur | gical | | | Pa | ediati | ic | | Obs-gynae | | | | | rdiac | | | TOTAL | | | | |
| employed during pain management education | n = | n = 16 | | | | n = 15 | | | | = 7 | | | n = 5 | | | | | = 4 | | | N = 47 | | | | |
| of nurses | Yes | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | | S | No | | Yes | | No | | |
| | F | % | F | % = | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n | % | n | % | |
| Pain management e- learning modules | 15 | 93,8 | 1 | 6,3 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 46 | 97,9 | 1 | 2,1 | |
| Assessment of prior knowledge about pain management | 15 | 93,8 | 1 | 6,3 | 14 | 93,3 | 1 | 6,7 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 45 | 95,7 | 2 | 4,3 | |
| Hands-on activities | 14 | 87,5 | 2 | 12,5 | 13 | 86,7 | 2 | 13,3 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 3 | 75,0 | 1 | 25,0 | 42 | 89,4 | 5 | 10,6 | |
| Simulations | 14 | 87,5 | 2 | 12,5 | 12 | 80,0 | 3 | 20,0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 3 | 75,0 | 1 | 25,0 | 41 | 87,2 | 6 | 12,8 | |

Table 4.26: Employment of the teaching approaches (N = 47)
| Debriefing sessions | 14 | 87,5 | 2 | 12,5 | 12 | 80,0 | 3 | 20,0 | 6 | 85,7 | 1 | 14,3 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 41 | 87,2 | 6 | 12,8 |
|-----------------------------------|----|------|---|------|----|------|----|------|---|------|---|------|---|------|---|------|---|------|---|------|----|------|----|------|
| Discussion groups | 14 | 87,5 | 2 | 12,5 | 11 | 73,3 | 4 | 26,7 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 40 | 85,1 | 7 | 14,9 |
| Preventing interruptions | 14 | 87,5 | 2 | 12,5 | 11 | 73,3 | 4 | 26,7 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 3 | 75,0 | 1 | 25,0 | 40 | 85,1 | 7 | 14,9 |
| Assignments about pain management | 15 | 93,8 | 1 | 6,3 | 11 | 73,3 | 4 | 26,7 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 3 | 75,0 | 1 | 25,0 | 40 | 85,1 | 7 | 14,9 |
| Role modelling | 15 | 93,8 | 1 | 6,3 | 10 | 66,7 | 5 | 33,3 | 6 | 85,7 | 1 | 14,3 | 4 | 80,0 | 1 | 20,0 | 3 | 75,0 | 1 | 25,0 | 38 | 80,9 | 9 | 19,1 |
| Case studies | 14 | 87,5 | 2 | 12,5 | 11 | 73,3 | 4 | 26,7 | 6 | 85,7 | 1 | 14,3 | 4 | 80,0 | 1 | 20,0 | 3 | 75,0 | 1 | 25,0 | 38 | 80,9 | 9 | 19,1 |
| Role play-activities | 11 | 68,8 | 5 | 31,3 | 11 | 73,3 | 4 | 26,7 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 37 | 78,7 | 10 | 21,3 |
| Focus groups | 13 | 81,3 | 3 | 18,8 | 12 | 80,0 | 3 | 16,8 | 6 | 85,7 | 1 | 14,3 | 4 | 80,0 | 1 | 20,0 | 1 | 25,0 | 3 | 75,0 | 36 | 76,6 | 11 | 23,4 |
| Nursing grand rounds | 11 | 68,8 | 5 | 31,3 | 10 | 66,7 | 5 | 33,3 | 6 | 85,7 | 1 | 14,3 | 5 | 100 | 0 | 0 | 3 | 75,0 | 1 | 25,0 | 35 | 74,5 | 12 | 25,5 |
| Reflective journals | 9 | 56,3 | 7 | 43,8 | 4 | 26,7 | 11 | 73,3 | 4 | 57,1 | 3 | 42,9 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 17 | 36,2 | 30 | 63,8 |

4.4.3.1 Pain management e-learning modules (N = 47)

Using available pain management e-learning modules for the professional development of nurses enhances the transfer of learning of pain management competencies (Vu, Fredrickson & Moore 2017:375). Shipton et al. (2018:152), and Rouleau, Gagnon, Côté, Payne-Gagnon, Hudson, Dubois and Bouix-Picasso (2019:14) indicate that the use of e-learning modules was mentioned the most often among common teaching strategies for pain management. In this study, all the respondents from surgical wards (n = 15), paediatric wards (n = 7), obs-gynae wards (n = 5), and cardiac wards (n = 4) reported the modules to have been employed (see Table 4.26). In medical wards, all but one respondent (F = 15; n = 16) reported the modules to be employed.

4.4.3.2 Assessment of prior knowledge about pain management (N = 47)

Assessment of prior knowledge is used to build new information or knowledge about pain management upon already-known information by increasing engagement to enhance the transfer of learning (Carver & Atkins 2021:77; Billings & Halstead 2019:529; Chu, Wang, Lin, Lee, Lin, Chieh, Sung & Lin 2019:5). All the respondents from paediatric wards (n = 7), obs-gynae wards (n = 5), and cardiac wards (n = 4) reported this teaching approach to have been employed (see Table 4.26). In medical wards, 15 out of the 16 respondents reported using this assessment.

4.4.3.3 Hands-on activities (N = 47)

Hands-on activities relate to a teaching approach that allows engagement or active participation during training sessions that provide direct practical experience to complete the learning task, such as the use of pain management devices, materials and activities that the hands can touch, manipulate, and practice the skills for better retention (Verkuyl, Romaniuk, Atack & Mastrilli 2017:5; Sumil 2016:5). All the respondents of the two wards (paediatric and obs-gynae) indicated they were in favour of hands-on-activities, although only 14 respondents in medical wards (n = 16), 13 in surgical

wards (n = 15), and three in cardiac wards (n = 4) indicated having employed handson activities (see Table 4.26).

4.4.3.4 Simulations (N = 47)

Simulation is a teaching strategy that mimics a situation or process of evidence-based practice using dummies during training sessions to educate and assess skills (Hermann 2020:398; Allred & Gerardi 2017: Gaberson, Oermann & Shellenbarge 2015:187). As illustrated in Table 4.26, all the respondents in obs-gynae (n = 5) and paediatric wards (n = 7) reported simulation as a teaching strategy used to enhance the transfer of learning of pain management competencies of nurses (see Table 4.26). Not all participants in the wards agreed they were using simulation. In medical wards, 14 respondents (n = 16); in surgical wards, 12 respondents (n = 15); and in cardiac wards, three respondents (n = 4) (see Table 4.26) reported utilizing simulation in pain management training.

4.4.3.5 Debriefing sessions (N = 47)

A debriefing session is a teaching approach employed by allowing or engaging the trainees in a reflective discussion about their performances related to learning objectives (Sawhney, Wong, Luctkar-Flude, Jussaume, Eadie, Bowry & Wilson 2018:249; Oriot & Alinier 2017:9). All respondents from obs-gynae (n = 5) and cardiac wards (n = 4) indicated that they were allowing debriefing sessions during their pain management training sessions (see Table 4.26). In medical wards, 14 out of the 16 respondents indicated allowing debriefing sessions, 12 out of the 15 respondents in surgical wards, and six out of the seven respondents in paediatric wards (see Table 4.26).

4.4.3.6 Discussion groups (N = 47)

A discussion group is a trainee-centred democratic teaching strategy embracing an orderly process of face-to-face interaction to exchange ideas about topics. In this study context, discussion groups were suggested to be formed to discuss pain management

(Hermann 2020:163; Suresh 2016:416; Shipton 2018:153; Drake and Williams 2017:9). All respondents in cardiac wards (n = 4) and paediatric wards (n = 7) indicated that they made use of discussion groups during their pain management education of nurses (see Table 4.26). Fourteen out of 16 respondents in medical wards reported discussion groups were suggested to be formed, four out of five in obs-gynae, and 11 out of 15 in surgical wards (see Table 4.26).

4.4.3.7 Preventing interruptions during teaching sessions (N = 47)

According to the National Association of Emergency Medical Services (EMS) of educators (2020:128), preventing interruptions during teaching sessions involves using a learning physical environment that avoids sources of distractions to accommodate the instruction for that teaching session. All respondents in obs-gynae wards (n = 5) and paediatric wards (n = 7) reported that interruptions during teaching sessions were prevented (see Table 4.26). Allred and Gerardi (2017:281) suggest that a teaching session for pain management be conducted in a pleasant, comfortable atmosphere without interruptions as it enhances nurses to apply what they have learned. In medical wards, 14 respondents (n = 16) reported that preventing interruptions during the teaching practice of pain management was established, three in cardiac wards (n = 4), and 11 in surgical wards (n = 15) (see Table 4.26).

4.4.3.8 Assignments about pain management (N = 47)

Assignments are teaching approaches that extend learning after training sessions to apply at work to improve skills proficiency at the workplace (McCune & Alexander 2020:122; Germossa, Sjetne & Hellesø 2018:1). All respondents in paediatric wards (n = 7) indicated that they were providing assignments to nurses to further learn about pain management (see Table 4.26). Respondents in medical wards (F = 15; n = 16), obs-gynae wards (F = 4; n = 5), cardiac wards (F = 3; n = 4), and surgical wards (F = 11; n = 15) reported that assignments about pain management were provided during their teaching (see Table 4.26).

4.4.3.9 Role modelling (N = 47)

Achinstein & Ogawa (2015: 57) and Gaberson et al. (2015:73) describe role modelling as a teaching method, a form of learning from experiences that uses humanist and social learning theories to allow trainees to learn the set examples of new knowledge, skills behaviours observed, and imitate the trainers or mentors. In medical wards, 15 respondents (n = 16) indicated that they were adopting role modelling during pain management education for nurses; six in paediatric wards (n = 7), four in obs-gynae wards (n = 5), three in cardiac (n = 4), and 10 in surgical wards (n = 15) (see Table 4.26).

4.4.3.10 Case studies (N = 47)

The case study teaching approach is a type of problem-based learning that is a participatory, discussion-based way of learning where trainees gain skills in critical thinking, communication, and group dynamics that enable them to recommend care for patients (Gaberson et al. 2015:237). Fourteen out of 16 respondents in medical wards indicated they had used case studies as a method during their pain management education for nurses, six out of seven in paediatric wards, four out of five in obs-gynae wards, three out of four in cardiac wards, and 11 out of 15 in surgical wards (see Table 4.26).

4.4.3.11 Role-play activities (N = 47)

Role-play is a teaching approach in which trainees are required to take on a role and pretend to be someone or something to perform part of activities to learn a realistic or hypothetical situation (Sit 2017:31). All respondents indicated that they were engaging in role-playing activities during their pain management training sessions for nurses. In obs-gynae wards, four out of the five respondents indicated they have engaged in role-playing activities to teach pain management, 11 out of 15 in surgical wards, and 11 out of 16 in medical wards (see Table 4.26).

4.4.3.12 Focus groups (N = 47)

A focus group is a teaching approach that allows a representative group of participants to reflect on their experiences in more detail in interview discussions about a given topic to be learned (Billings & Halstead 2019: 529). Table 4.26 illustrates the respondents who reported engaging nurses in focus groups as a teaching strategy. Six (n = 7) in paediatric wards, 13 (n = 16) in medical wards, four in obs-gynae wards (n = 5), 12 in surgical wards (n = 15) and one in cardiac wards, (n = 4) (see Table 4.26) reported to engage in focus groups.

4.4.3.13 Nursing grand rounds (N = 47)

Nursing grand rounds involve observation and discussions about patient assessments and interventions in a clinical setting to improve knowledge and attitudes (Hamoen, van Blankenstein, De Jong, Ray & Reinders 2020: 81; Hermann 2020;195; Brant, Mohr, Coombs, Finn & Wilmarth 2017:222; Gaberson et al. 2015:250). All respondents from obs-gynae wards (n = 5) reported that nursing grand rounds were implemented as a teaching strategy to provide opportunities for nurses to learn about pain management. All but one respondent (F = 6; n = 7) in paediatric wards, 3 (n = 4) in cardiac wards, 11 (n = 16) in medical wards, and 10 (n = 15) in surgical wards indicated implementing this teaching strategy (see Table 4.26).

4.4.3.14 Reflective journals (N = 47)

Reflective journaling is a teaching approach that provides the trainees an opportunity to write about their learning experiences by sharing their insights into their strengths and weaknesses, the understanding, interrelation of knowledge and emotion, critical thinking, and applying what was learned during training (Hermann 2020:351; Roca, Reguant, Tort & Cancet 2020:5). Reflective journal writing by nurses to share their learning experiences about pain management was reported to be implemented by all respondents in all the wards included in the study, except for some of the respondents

in the paediatric wards (F = 4; n = 7), in medical wards (F = 9; n = 16), and in surgical wards (F = 4; n = 15) (see Table 4.26).

4.4.4 Learning content included in pain management education (N = 47)

The learning content refers to topics, themes, beliefs, behaviours, concepts, and facts, often grouped within each subject, developing skills, knowledge, and attitudes that are expected to be learned and form the basis of teaching and learning (Mishra & Ghosh 2021:215). Employing the teaching approaches during pain management education was required to incorporate the learning content to enhance the transfer of learning of pain management competencies of nurses. The responses regarding the learning content included in pain management education were grouped using the steps of the nursing process, namely (1) **pain assessment**, (2) **strategies to plan** for pain management, (3) **pain intervention strategies** implemented, and (4) **nursing actions** for pain management **evaluation**.

4.4.4.1 Pain assessment

The learning content of pain management education of nurses that covered pain assessment as reported by respondents was (1) physiological pain indicators, (2) behavioural indicators of pain, (3) pain screening, (4) patients' self-report of pain, (5) proxy reported pain, (6) the systematic pain guide tool for history taking, (7) valid and reliable pain rating scales, (8) acute types of pain, (9) chronic types of pain, (10) factors related to the impact of pain on daily living (ADLs) and quality of life (QOL) (see Tables 4.28 to 4.36).

(1) The physiological pain indicators (N = 47)

The physiological pain indicators are responses of sympathetic activation that may infer that pain is present as observed during pain assessment mainly for non-communicative, sedated or critically ill patients (Jiang, Mieronski, Syrjälä, Anzanpour, Terävä, Rahmani, Salnterä, Aantaa, Hagelberg & Liljeberg 2019:500). The physiologic pain

indicators included in the learning content of pain management education were **in-creased blood pressure**, **tachycardia and tachypnea** (see Table 4.27) as is supported by Ghezeljeh, Nasari, Haghani and Loieh (2017:147), and Dongara et al. (2017:27).

| | Nui | sing \ | Wai | rds | | | | | | | | | | | | | | | | | | | | |
|---|-----|--------|-----|-----|-----|-------|----|---|-----|----------|----|---|-----|--------|----|---|-----|-------|----|---|-----|-------|----|---|
| Physiologic | Me | dical | | | Sur | gical | | | Pae | ediatric | ; | | Ob | os-gyn | ae | | Ca | rdiac | | | тот | ΓAL | | |
| pain indica- tors during pain assess- | n = | 16 | | | n = | 15 | | | n = | 7 | | | n = | = 5 | | | n = | 4 | | | N = | 47 | | |
| nent P | Yes | 6 | Nc |) | Yes | 5 | No | | Yes | 5 | No |) | Ye | S | Nc |) | Ye | 6 | No |) | Yes | | No | |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n | % | n | % |
| Increased blood pres- sure | | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100,0 | 0 | 0 |
| | | | | | | | | | | | | | | | | | | | | | | | | |

 Table 4.27: The physiological pain indicators (N = 47)

| Tachycardia | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100,0 | 0 | 0 |
|-------------|----|-----|---|---|----|-----|---|---|---|-----|---|---|---|-----|---|---|---|-----|---|---|----|-------|---|---|
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Tachypnea | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100,0 | 0 | 0 |

a) Increased blood pressure (N = 47)

Increased blood pressure is used as a cue for pain assessment as it occurs more commonly in acute pain and is considered a factor that affects both systolic and diastolic blood pressure by increasing sympathetic activity, leading to vasoconstriction (Nugent & Vitale 2017:387). Table 4.27 shows that all respondents from all nursing wards included increased blood pressure as part of the learning content for pain management education (see Table 4.27).

b) Tachycardia (increased heart rate) (N = 47)

Tachycardia is a cue for pain assessment as it occurs more commonly in acute pain as a factor influencing heart rate by stimulating sympathetic nervous system response and producing vasoconstriction (Potter, Perry, Stockert & Hall 2020: 480). As shown in Table 4.27, all respondents in all the wards reported having included tachycardia in their teaching content during nurses' pain management education.

c) Tachypnoea (increased respiratory rate) (N = 47)

Increased respiratory rate forms part of pain assessment in acute pain as a factor that influences respiration by increasing its flow, frequency, and volume due to sympathetic nervous system response increasing oxygen intake demand (James et al. 2013:527). All respondents indicated that tachypnoea was included during their teaching sessions on pain management (see Table 4.27).

(2) The behavioural indicators of pain (N = 47)

The behavioural or nonverbal indicators of pain are responses to pain stimuli that result in a change in the behaviour of an individual and include facial expressions, verbal responses, and body movements (Chang & Jonhson 2018:275; Costa, Rossato, Bueno, Secco, Sposito, Harrison & De Freitas 2017:5). Respondents in this study identified the behavioural indicators of pain to have been included in the learning content of pain management education of nurses. These were (1) facial expressions (n = 47), verbal responses (n = 47), and (3) body movements (n = 47) (see Table 4.28).

| Table 4.28: The | behavioural pain indicators | (N=47) | |
|-----------------|-----------------------------|--------|--|
|-----------------|-----------------------------|--------|--|

| | | | | | | | | | | | | Nursi | ng \ | Nards | | | | | | | | | | |
|-------------------------|--|-----|------|---|----|------|------|----|---|-------|------|-------|------|-------|-----|----|---|------|------|----|----|-------|------|---|
| Behavioural | | Med | ical | | | Surg | ical | | | Paedi | atri | C | | Obs-g | yna | ie | | Card | liac | | | то | TAL | |
| pain during | | n= | 16 | | | n = | 15 | | | n = | 7 | | | n = | =5 | | | n= | 4 | | | N | = 47 | |
| ment | n= 16 Yes No F % F % | | | | 1 | /es | 1 | No | ١ | (es | | No | | Yes | | No | | res | | No | | Yes | No | D |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n | % | n | % |
| Facial expres- sions | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100,0 | 0 | 0 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Verbal re- sponses | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100,0 | 0 | 0 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| The body movements | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100,0 | 0 | 0 |

(a) Facial expressions (N=47)

Facial expressions such as frowning or grimacing are reliable nonverbal signals of pain useful for assessing pain in patients who are cognitively impaired or unable to self-report pain due to limited communicative ability (Perry, Potter, Ostendorf & Laplante 2021:443; Pölkki, Korhonen & Laukkala 2018:732). Table 4.28 shows that facial expressions as a way of assessing pain were reported to be included as learning content of pain management education for nurses by all respondents in all the wards included in the study.

(b) Verbal responses (N=47)

Verbal responses (such as crying) are automatic verbal responses suggestive of potential indicators for pain that are observed despite a self-report given during pain assessment (Gawlik, Melnyk & Teall 2020:94). All respondents in all the wards indicated that verbal expressions were included as learning content of pain management training sessions on how to conduct pain assessment (see Table 4.28).

(c) The body movements (N=47)

The body movements such as "kicking" primarily serve as protective or pain management functions from further noxious input and promote pain relief observed during pain assessment (Vervoort, Karos, Trost & Prkachin 2018:104; Björn, Pudas-Tähkä, Salanterä & Axelin 2017:72). Observing the body movements used to estimate the presence and intensity of the pain in patients experiencing the pain, and as indicated in table 4.28, all of the respondents in all the wards agreed that they included them as learning content of pain management education for nurses (see Table 4.28). The pain screening was indicated to be included as learning content for pain management education for nurses to conduct pain assessments.

(3) Pain screening (N=47)

Pain screening is a process that includes the initial screening interview and ongoing assessment of the presence of pain by directly asking the patients, who can communicate, using a comprehensive pain assessment guide (Wyatt, Taylor, de Wit & Hotton 2020:282; Roller-Wirnsberger, Singler & Polidori 2018:200). As indicated in Table 4.37, 46 respondents (f=97,9%; N=47) within the wards agreed that their pain management teaching content for nurses included a process of pain screening. Only one respondent from the paediatric wards (F=6; n=7) did not include pain screening as the pain management teaching content for nurses. It is, however, difficult to perform pain screening on paediatric patients due to their inability to communicate or report their pain (Cascella, Bimonte, Saettini & Muzio 2019:133).

| | | | | | | | | | | | | Nursi | ng | Wards | | | | | | | | | | |
|---------------------|----|-----|------|----|----|-------|------|----|---|-------|-------|-------|----|-------|-----|----|---|------|------|----|----|------|-------|-----|
| | | Med | ical | | | Surgi | ical | | | Paedi | atrio | 5 | | Obs-g | yna | e | | Card | liac | | | т | OTAL | |
| Pain screen- | | n= | 16 | | | n =1 | 15 | | | n = | 7 | | | n = | =5 | | | n= | 4 | | | N | l= 47 | |
| Pain screen- ing | Y | ′es | | No | ١ | ′es | | No | ١ | /es | | No | | Yes | | No | , | /es | | No | Y | es | N¢ |) |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n | % | n | % |
| | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 6 | 85.7 | 1 | 14,3 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 46 | 97,9 | 1 | 2,1 |

Table 4.29: Pain screening (N=47)

(4) Obtaining the patient's self-report of pain (N=47)

Obtaining the patient's self-report of pain is considered the golden standard of pain assessment and the most reliable conversational approach to pain assessment by cognitively intact patients able to communicate pain experienced (Kang & Demiris 2018:4; Tidwell 2017:207; Topham & Drew 2017:366). The responses in Table 4.30 indicate that 46 (N=47) respondents within the wards included obtaining the patient's self-report to assess the pain as a learning content during their pain management education for nurses. On the contrary, only one respondent (F=15; n=16) in medical wards did not agree that obtaining the patient's self-report of pain was covered as learning content of pain management education for nurses (see Table 4.30).

| | | | | | | | | | | | | Nursi | ng | Wards | | | | | | | | | | |
|-----------------|----|------|------|-----|----|------|------|----|---|-------|------|-------|----|-------|-----|----|---|------------|-----|----|----|------|------|-----|
| | | Medi | ical | | | Surg | ical | | | Paedi | atri | C | | Obs-g | yna | e | | Card | iac | | | т | DTAL | |
| Patient's self- | | n= ' | 16 | | | n =′ | 15 | | | n = | 7 | | | n = | :5 | | | n= | 4 | | | N | = 47 | |
| report of pain | Y | (es | | No | ١ | (es | | No | ١ | /es | | No | | Yes | | No | , | Yes | | No | Y | es | N | D |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n | % | n | % |
| | 15 | 93,8 | 1 | 6,3 | 15 | 100 | 0 | 0 | 7 | 100 | 1 | 14,3 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 46 | 97,9 | 1 | 2,1 |

Proxy-reported pain was indicated to be included in the learning content for pain management education for nurses to conduct pain assessments.

(5) Proxy-reported pain (N=47)

Proxy-reported pain (obtaining pain reports from parents or family members) is essential in pain assessment for patients with intellectual and developmental disabilities as information about the pain experienced by the patient must be obtained from nurses, parents, or family members (Alotaibi et al. 2018:530; Mamhidir, Sjölund, Fläckman, Wimo, Sköldunger & Engström 2017:6). Of the 47 respondents, 46 (f=97,9%; N=47) respondents reported that proxy reported pain was part of the pain module to assess pain. However, even though proxy-reported pain was reported to have been included by all respondents in all the wards, 15 out of 16 respondents in the medical wards agreed to have included it as learning content.

 Table 4.31: Proxy-reported pain (N=47)

| | Nu | rsing V | Var | ds | | | | | | | | | | | | | | | | | | | | |
|---------------------|------|---------|-----|-----|-----|-------|----|---|----|-------|-----|---|-----|--------|-----|---|----|-------|----|---|-----|------|----|-----|
| | Ме | dical | | | Sur | gical | | | Pa | ediat | ric | | Oł | os-gyı | nae | | Ca | rdiac | ; | | то | TAL | | |
| Proxy reported pain | n= ' | 16 | | | n = | 15 | | | n | =7 | | | n : | =5 | | | n= | 4 | | | N= | 47 | | |
| | Yes | 5 | No |) | Yes | 5 | Nc |) | Ye | es | No |) | Ye | es | Nc |) | Ye | S | No |) | Yes | 5 | Nc |) |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % |
| | 15 | 93.8 | 1 | 6.3 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 46 | 97.9 | 1 | 2.1 |

| | Nurs | ing Wa | rds | | | | | | | | | | | | | | | | | | | | | |
|--|-------|--------|-----|-----|-------|------|----|---|-----|----------|----|---|-----|---------|----|------|-----|------|----|---|-------|------|----|-----|
| | Medi | ical | | | Surg | ical | | | Pae | ediatric | | | Ob | s-gynae | • | | Car | diac | | | тот | AL | | |
| temic pain guide | n = 1 | 6 | | | n = 1 | 5 | | | n = | 7 | | | n = | 5 | | | n = | 4 | | | N = 4 | 47 | | |
| | Yes | | No | | Yes | | No | | Yes | 3 | No | | Yes | 3 | No | | Yes | 3 | No | | Yes | | No | |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n | % | n | % |
| The intensity of pain using the pain scale | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 46 | 97,9 | 1 | 2,1 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| The location of pain | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 46 | 97,9 | 1 | 2,1 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| The duration of pain | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 46 | 97,9 | 1 | 2,1 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Aggravating or alle- viating factors | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 46 | 97,9 | 1 | 2,1 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Description of the type of pain | 15 | 93,8 | 1 | 6,3 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 45 | 95,7 | 2 | 4,3 |

Table 4.32: The WILDA systematic pain assessment guide (N = 47)

Respondents indicated that for taking the history of a patient's pain, the systematic pain guide was included in the learning content of pain management education for nurses.

(6) Systematic pain assessment guides for pain history taking (N=47)

For pain history taking, a patient's self-report of pain is the most important component of comprehensive pain assessment; thus, asking questions that follow five key components of the WILDA pain assessment guide is essential (McDaniel, Caspersen, Crumpton, Galbraith, Hall, Huddleston, Puckett & Williams 2021:5; Fink & Brant 2018:356). To assess pain comprehensively, the five domains of WILDA needed to be included in the learning content of pain management education (see Table 4.32). The domains are (1) asking about the intensity of pain using the pain scale (n=46), (2) the location of pain (n=46), (3) the duration of pain (n=46), aggravating or alleviating factors (n=46), and (5) asking the patient to describe the type of pain in his/her own words (n=45). The domains were reported to be included in the pain management learning content, as illustrated in Table 4.32.

(a) The intensity of pain using the pain scales (N = 47)

According to the literature, the pain management learning content must include the domain by asking the patient to rate the intensity or magnitude of experienced pain using the pain scales (Peate 2019: 510). Asking about the intensity of their pain by using the pain scales was reported to be included in the learning content by all of the respondents in all wards, except for the one respondent in an obs-gynae ward (F = 4; n = 5).

(b) The location of the pain (N = 47)

Inglis and Kenneally (2020:194) indicated that it is essential to include educational content on pain management and how to conduct pain assessment by asking the patient to point to the location of the painful sites, assisting in determining nursing interventions. Questions about the location of pain were indicated to be part of the learning content by all respondents from medical wards, surgical wards, paediatric wards, and cardiac wards. Only one respondent in an obs-gynae ward (F = 4; n = 5) did not include it in the learning content (see Table 4.32).

(c) The duration of pain (N = 47)

The pain module learning content includes a comprehensive pain assessment to distinguish between acute and chronic pain that is determined by the onset of pain, its duration, as well as how long the nonverbal patients exhibit any nonverbal cues of pain (Lewis et al. 2017:108). As portrayed in Table 4.32, in obs-gynae wards, four out of five of the respondents indicated that they were including the domain of asking about the duration of pain as learning content for pain management education, while all respondents from other nursing wards indicated that they were including this domain (see Table 4.40).

(d) Aggravating or alleviating factors (N = 47)

Aggravating or alleviating factors refer to the influences or factors that increase pain as well as activities and situations that decrease pain (Fink and Brant 2018:358; Lewis et 2016:108). All respondents from all nursing wards apart from the one but the same respondent from the obs-gynae ward (F = 4; n = 5) included this domain (see Table 4.32).

(e) Description of the type of pain (N = 47)

According to the participants, the importance of asking patients to describe their pain formed part of the content of the pain management education for nurses. Part of the pain management training program for nurses included a description of the type of pain as reported by the respondents (see Table 4.32). All respondents in all nursing wards, exclusive of one participant in a medical ward (F = 15; n = 16) and one participant in a gynae ward (F = 4; n = 5), indicated that they include this domain (see Table 4.32).

Valid and reliable pain rating scales must be included in the learning content of pain management education for nurses.

(7) The valid and reliable pain rating scales (N = 47)

As part of learning about pain assessment, valid and reliable pain rating scales did form part of the pain management training program's content for nurses, as illustrated in Table 4.33. The scales used to rate pain mentioned by respondents were (1) the Numeric Rating Scale (n = 45), (2) the FLACC pain scale (n = 44), (3) the Wong-Baker Faces scale (n = 43), and (4) the CRIES pain scale (n = 39) (see Table 4.33). The respondents from the various wards in the study indicated that the teaching content on pain management included these scales, as illustrated in Table 4.33.

| Table 4.33: The valid a | and reliable pain | rating scales (N = 47) |
|-------------------------|-------------------|------------------------|
|-------------------------|-------------------|------------------------|

| | Nu | rsing \ | Wa | rds | | | | | | | | | | | | | | | | | | | | |
|--------------------------------|-----|---------|----|------|-----|-------|----|----|-----|--------|-----|---|-----|-------|-----|----|-----|--------|----|------|-----|------|----|------|
| Valid and reliable pain rating | Me | dical | | | Sur | gical | | | Pa | aediat | ric | | Oł | os-gy | nae |) | Ca | ardiac | ; | | то | TAL | | |
| scales | n = | 16 | | | n = | 15 | | | n = | = 7 | | | n : | = 5 | | | n : | = 4 | | | N = | : 47 | | |
| | Yes | 5 | No |) | Yes | 3 | No |) | Ye | es | No |) | Ye | es | No |) | Ye | es | No |) | Yes | 5 | No | > |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n | % | n | % |
| Numeric Rating scale | 15 | 93,8 | 1 | 6,3 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 4 | 80 | 1 | 20 | 4 | 100 | 0 | 0 | 45 | 95,7 | 2 | 4,3 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| FLACC pain scale | 13 | 81,3 | 3 | 18,8 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 44 | 93,6 | 3 | 6,4 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Wong-Baker Faces scale | 13 | 81,3 | 3 | 18,8 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 4 | 80 | 1 | 20 | 4 | 100 | 0 | 0 | 43 | 91,5 | 4 | 8,5 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| CRIES scale | 12 | 75,0 | 4 | 25 | 12 | 80 | 3 | 20 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 3 | 75 | 1 | 25,0 | 39 | 83,0 | 8 | 17,0 |

(a) Numeric rating scale (N = 47)

Cianfrini, Doleys and Richardson (2021:45) recommend that the learning content of pain management education for nurses and patients should include training on how to use NRS to assess and rate pain. The majority of respondents in all wards indicated that they included training on the NRS rating scale in the study content, with two exceptions: one respondent in the medical ward (F = 15; n = 16) and one respondent in the gynae ward (F = 4; n = 5). (see Table 4.33).

(b) FLACC scale
$$(N = 47)$$

Only three respondents from medical wards (F = 13, n = 16) indicated that they did not include FLACC during pain management training sessions (see Table 4.33). On the other hand, all respondents in the other wards included it in the study.

(c) Wong-Baker FACES scale (N = 47)

As illustrated in Table 4.33, the respondents in all the wards, except two in a medical ward (F = 13; n = 16) and one in an obs-gynae ward (F = 4; n = 5) agreed that they include the Wong-Baker FACES scale when training nurses regarding pain management (see Table 4.33). The authors Erogan and Celik (2020:1020), as well as Ziyaeifard, Azarfarin, Zamani, Alizadehasi, Khalili, Moradian, Koleini and Pouraliakbar (2018:13), recommend that the Wong-Baker FACES scale to rate the severity of pain must be part of the content of an education program to teach pain management.

(d) CRIES scale (N = 47)

It is important to include in the learning content of pain management education the topic of how to use the CRIES scale, as recommended by Cameron, Browne, Mitra, Dalziel and Craig (2018:453) as well as by Costa et al. (2017:6). Both in obs-gynae wards and paediatric wards all respondents agreed that they do include training of the CRIES rating scale in the study content (see Table 4.33). Two respondents from

medical wards (F = 12; n = 16), three from surgical wards (F = 12; n = 15), and one from cardiac wards (F = 3; n = 4) indicated that they do *not* include CRIES as part of the pain management training program (see Table 4.33).

The respondents indicated that the topic of the types of acute pain was included as part of the pain management training program.

(8) Acute types of pain (N = 47)

It is well-known that acute pain has a short duration and lasts less than 3 to 6 months, and it may be due to trauma, disease process, injury, or surgery with three general types: nociceptive, inflammatory, and neuropathic (Buckenmaier, Kent, Mariano & Brookman 2019:216). To manage pain effectively, the acute types of pain need to form part of the learning content of pain management training (see Table 4.34). The pain types are (1) post-procedural pain (n = 47), (2) procedural pain (n = 46), (3) acute postoperative pain (n = 46), (4) the acute disease process (n = 45), traumatic pain (n = 44), and labour pain (n = 35) (see Table 4.34).

The acute types of pain covered in pain management learning content were indicated in various wards by respondents in the study, as illustrated in Table 4.34.

| Acute types of pain | Nur | sing \ | Nar | ds | | | | | | | | | | | | | | | | | | | | | | |
|--------------------------------|--------|--------|-----|-----|----------|-----|----|---|------------|-----|----|---|-----|--------|----|------|---------|-----|----|--------|-----|-------|-----|-----|--|--|
| | Me | dical | | | Surgical | | | | Paediatric | | | | | os-gyn | ae | | Cardiac | | | | | TOTAL | | | | |
| | n = 16 | | | | n = 15 | | | | n = 7 | | | | n = | = 5 | | | n = 4 | ļ | | N = 47 | | | | | | |
| | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | | | |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n = | % | n = | % | | |
| Post-procedural pain | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100 | 0 | 0 | | |
| | | | | | | | | | | | | | | | | | | | | | | | | | | |
| Procedural pain | 15 | 93,8 | 1 | 6,3 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 46 | 97,8 | 1 | 2,1 | | |
| | | | | | | | | | | | | | | | | | | | | | | | | | | |
| Acute post-oper- ative pain | 15 | 93,8 | 1 | 6,3 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 46 | 97,8 | 1 | 2,1 | | |
| | | | | | | | | | | | | | | | | | | | | | | | | | | |

Table 4.34: Acute types of pain (N = 47)

| Acute diseases process | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 45 | 95,7 | 2 | 4,3 |
|------------------------|----|------|---|------|----|------|---|------|---|------|---|------|---|------|---|------|---|------|---|------|----|------|----|------|
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Traumatic pain | 15 | 93,8 | 1 | 6,3 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 44 | 93.6 | 3 | 6,4 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Labour pain | 13 | 81,3 | 3 | 18,8 | 10 | 66,7 | 5 | 33,3 | 5 | 71,4 | 2 | 28,6 | 4 | 80,0 | 1 | 20,0 | 3 | 75,0 | 1 | 25,0 | 35 | 74,5 | 12 | 25,5 |

(a) Post-procedural pain (N = 47)

Post-procedural pain is known as the pain inflicted after any procedure performed by a healthcare provider (Ahmed 2020:24; Gooden, Lowrie & Jackson 2018:128; Bice 2018:231). All respondents in the study agreed that they include this type of pain as part of the pain management training program (see Table 4.34).

(b) Procedural pain (N = 47)

Procedural pain is a type of acute pain potentially triggered during procedures performed by a healthcare worker (Sumser, Leimena & Altillo 2019:54; Bice 2018: 231). All respondents from all the wards included in the study indicated to have included this type of pain in the pain management training program (see Table 4.34)

(c) Acute postoperative pain (N = 47)

Acute postoperative pain is known to be the type of pain patients experience immediately and for up to seven days after surgery (Fleisher & Rosenbaum 2017:185). Table 4.34 shows that acute postoperative pain as a type of pain experienced by patients was reported to be part of the teaching content of pain management training sessions for nurses by all respondents in the wards, except for one respondent in an obs-gynae ward (F = 4; n = 5) (see Table 4.34).

(d) Acute disease process (N = 47)

Craft, Gordon, Huether, McCane and Brashers (2018:5) described the acute disease process as an acute type of pain associated with pathophysiological processes that produce temporarily deferred discomforts and occur for a short duration. All respondents in all the wards, apart from one respondent in an obs-gynae ward (F = 4; n = 5), indicated that they included training on the acute diseases process in the study content.

(d) Traumatic pain (N = 47)

Traumatic pain is any pain that occurs following injuries from physical trauma (American Academy of Orthopaedic Surgeons 2021:280). There was consensus among all respondents that traumatic pain was a part of the pain management training program. However, one respondent from a medical ward (F = 15; n = 16) and one from a gynae ward (F = 4; n = 5) indicated that this type of pain is not covered by the pain management training program (see Table 4.34).

(e) Labour pain (N = 47)

Labour pain is a recurrent pain associated with the contraction of the uterus in labour (Mosby's Dictionary 2013:1002). Thirteen (n = 16) respondents in medical wards, 10 (n = 15) in surgical wards, five (n = 7) in paediatric wards, four (n = 5) in obs-gynae wards, and three (n = 4) in cardiac wards indicated that they include labour pain as part of pain management training program (see Table 4.34).

Part of the pain management program covered chronic pain types during education for nurses.

(9) Chronic pain types (N = 47)

Chronic pain is the type of pain that lasts longer than 12 weeks and can be broken into three major types, namely, pain from tissue damage (nociceptive pain), pain from somatosensory damage (neuropathic pain), or a mixture of both nociceptive and somatosensory (Shah, Kodack & Walker 2020:1). Chronic pain types needed to be included as part of the pain management training program (see Table 4.35). The chronic pain types are (1) low back pain (n = 43), (2) neuropathic pain (n = 42), (3) headache (n = 42), and (4) cancer pain (n = 40) (see Table 4.35).

The chronic pain types that form part of the pain management learning content were indicated in various wards by respondents in the study, as illustrated in Table 4.35.

| Chronic | Nur | sing W | /ard | ls | | | | | | | | | | | | | | | | | | | | |
|------------------------------------|-----|--------|------|------|--------------------|------|----|------|-------|---------|----|------|-----------|------|----|------|-----|-------|----|------|--------|------|----|------|
| | Mec | dical | | | Surgical n = 15 | | | | Ра | ediatri | с | | Obs-gynae | | | | | rdiac | | | TOTAL | | | |
| types of | n = | 16 | | | | | | | n = 7 | | | | n = 5 | | | | | - 4 | | | N = 47 | | | |
| pain | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | | Yes | | No | |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n | % | n | % |
| Low back pain | 16 | 100 | 0 | 0 | 14 | 93,3 | 1 | 6,7 | 6 | 85,7 | 1 | 14,3 | 3 | 60,0 | 2 | 40,0 | 4 | 100 | 0 | 0 | 43 | 91,5 | 4 | 8,5 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Neuro- pathic pain | 16 | 100 | 0 | 0 | 13 | 86,7 | 2 | 13,3 | 6 | 85,7 | 1 | 14,3 | 3 | 60,0 | 2 | 40,0 | 4 | 100 | 0 | 0 | 42 | 89,4 | 5 | 10,6 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Headache related to migraine | 14 | 87,5 | 2 | 12,5 | 14 | 93,3 | 1 | 6,7 | 7 | 100 | 0 | 0 | 3 | 60,0 | 2 | 40,0 | 4 | 100 | 0 | 0 | 42 | 89,4 | 5 | 10,6 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Cancer pain | 15 | 93,8 | 1 | 6,3 | 13 | 86,7 | 2 | 13,3 | 6 | 85,7 | 1 | 14,3 | 3 | 60,0 | 2 | 40,0 | 3 | 75,0 | 1 | 25,0 | 40 | 85,1 | 7 | 14,9 |

 Table 4.35: The chronic pain types (N = 47)

(a) Low back pain (N = 47)

Low back pain (LBP) is a type of pain that presents with muscle tension or stiffness localized in the margin and above the inferior gluteal folds, with or without sciatica (leg pain) related to sciatic nerve compression resultant from trauma or pathology and persists for more than three months (Noormohammadpour, Kordi, Masournia, Akbari-Fakhrabadi & Kordi 2018:492; NSCA-National Strength and Conditioning Association & Jacobs 2018:68). Table 4.35 illustrates the respondents that reported including the chronic type of pain as teaching content of pain management program. Fourteen (n = 15) in surgical wards, six in paediatric wards (n = 7), and three in obs-gynae wards (n = 5) indicated that low back pain formed part of pain management education for nurses.

(b) Neuropathic pain (N = 47)

Neuropathic pain is caused by a lesion or disease of the somatosensory nervous system (Benson, Raja, Fishman, Liu & Cohen 2017:252). The patients suffering from this type of pain need to be assessed and managed for pain relief. This is essential as it was reported in the study by all of the respondents in all wards that neuropathic pain is included in the learning content of pain management education for nurses, apart from two respondents in surgical wards (F = 13; n = 15), one respondent in a paediatric ward (F = 6; n = 7), and two respondents in obs-gynae wards (F = 3; n = 5) (see Table 4.35).

(c) Headache related to migraine (N = 47)

Headaches related to migraines are recurring symptoms of pain associated with abnormal sensory perception (Hwang; Tsai, Liu, Chen & Lai 2018:713; Mosby Dictionary 2013:1143). Part of the pain management teaching content included headaches related to migraine as indicated by all of the respondents in all wards, except for one respondent in a surgical ward (F = 14; n = 15), two in medical wards (F = 14; n = 16), and two in obs-gynae wards (F = 3; n = 5) (see Table 4.35).

(d) Cancer pain (N = 47)

Cancer pain is a dull aching, well localised deep squeezing pressure, burning, or tingling pain caused by damage to bones, pressing on the nerves, or internal organs (Burton & Bejarano 2021:58). Alnajar, Darawad, Alshahwan and Samarkandi (2019:186) recommend that it is essential to include cancer pain as the teaching content of pain management programs for nurses. Table 4.35 illustrates that the respondents in the study indicated that cancer pain was included as part of the pain management teaching program. Respondents from medical wards (F = 14; n = 16), surgical wards (F = 13; n = 15), paediatric wards (F = 6; n = 7), cardiac wards (F = 3; n = 4), and obs-gynae wards (F = 3; n = 5) responded that cancer pain was included in pain management teaching program (see Table 4.35).

Respondents indicated that the factors related to the impact of pain on activities of daily living and quality of life should be included in the learning content of pain management education.

(10) Factors related to the impact of pain on activities of daily living (ADLs) and quality of life (QOL) (N = 47)

Pain is multidimensional and contributes to the biopsychosocial experiences of pain impacting activities of daily living (ADLs) and quality of life (Fundaskowski 2020:161; Gagliese, Gauthier, Narain & Freedman 2018:208). Activities of daily living (ADLs) are self-care tasks required on a day-to-day basis that indicate the functional status (Ra-machandra 2018:27). At the same time, the quality of life (QOL) is individuals' perceptions of their position in life in the context of the culture and value systems in which they live and about their goals, expectations, standards, and concerns (Ragsdale & Miller 2020:6). It is essential to identify the factors related to the impact of pain on

ADLs and QOL needed to be included in the learning content of pain management. The factors are (1) the physical impact of pain (n = 44), (2) the psychological impact of pain (n = 42), and (3) the social impact of pain (n = 41). As suggested by Alnajar et al. (2019:186), the factors related to the impact of pain need to form part of the learning content of pain management education.

The respondents from various wards indicated that the teaching content on pain management included these factors, as illustrated in Table 4.36.

| Factors re- lated to the impact of pain on ADLs and QOL | Nur | sing V | Varo | ds | | | | | | | | | | | | | | | | | | | | | |
|--|-----|--------|------|----------|--------|------|----|------------|-------|-----|----|-----------|-----|------|---|------|-------|------|---|-------|--------|------|----|------|--|
| | Me | dical | | Surgical | | | | Paediatric | | | | Obs-gynae | | | | | rdiac | | | TOTAL | | | | | |
| | n = | 16 | | | n = 15 | | | | n = 7 | | | | n = | = 5 | | | n = 4 | | | | N = 47 | | | | |
| | Yes | | No | | Yes | | No | | Yes | | No | No | | Yes | | No | | Yes | | No | | Yes | | | |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n | % | n | % | |
| Physical impact of pain | 16 | 100 | 0 | 0 | 14 | 93,3 | 1 | 6,7 | 7 | 100 | 0 | 0 | 3 | 60,0 | 0 | 0 | 4 | 100 | 0 | 0 | 44 | 93,6 | 3 | 6,4 | |
| | | | | | | | | | | | | | | | | | | | | | | | | | |
| Psychologi- cal impact of pain | 14 | 87,5 | 2 | 13,3 | 13 | 86,7 | 2 | 13,3 | 7 | 100 | 0 | 0 | 4 | 80,0 | 1 | 20,0 | 4 | 100 | 0 | 0 | 42 | 89,4 | 42 | 89,4 | |
| | | | | | | | | | | | | | | | | | | | | | | | | | |
| Social im- pact of pain | 16 | 100 | 0 | 0 | 13 | 86,7 | 2 | 13,3 | 7 | 100 | 0 | 0 | 3 | 60,0 | 2 | 40,0 | 2 | 50.0 | 2 | 50,0 | 41 | 87,2 | 6 | 12,8 | |
| | | | | | | | | | | | | | | | | | | | | | | | | | |

Table 4.36: The factors related to the impact of pain on ADLs and QOL (N = 47)
(a) Physical impact of pain (N = 47)

The physical impact of pain includes any physical disability which influences the experience of pain and has a negative effect on reducing the patient's ability to perform ADLs and QOL (Chang & Johnson 2021:420; Ugur & Erci 2019:330). As illustrated in Table 4.36, all respondents in all the wards agreed that they include the physical impact of pain when training nurses about pain management except for one respondent in a surgical ward (F = 14; n = 15) and two respondents in obs-gynae wards (F = 3; n = 5) who did not agree.

(b) Psychological impact of pain (N = 47)

The psychological impact of pain includes any mental disorders which influence pain experience that contribute to disability in patients' ADLs and QOL (Fundaskowski 2020:161). All respondents in cardiac and paediatric wards indicated that they include the psychological impact in the learning content of pain management training (see Table 4.36). In medical wards, 14 out of the 16 respondents, 13 out of 15 in surgical wards, and 4 out of 5 in obs-gynae wards agreed that they include the psychological impact in the learning content training (see Table 4.36).

(c) Social impact of pain (N = 47)

The social impact of pain is considered to be any strained social relationship that influences an individual's pain experience (Fundaskowski 2020:161). In both wards, namely, medical wards and paediatric wards, all respondents reported that the social impact of pain forms part of the teaching of pain management to nurses, apart from 13 of the respondents in surgical wards (n = 15), three in obs-gynae wards (n = 5), and two in cardiac wards (n = 4) (see Table 4.36).

As part of the nursing plan of care, strategies of planning for pain management were identified to be educated during pain management education.

4.4.4.2 Strategies to plan for pain management (N = 47)

The strategies to plan for pain management are the components of the nursing process and involve actions designed to achieve the target goal regarding pain management (Miftah, Tilahun, Fantahun, Adulkadir & Gebrkirstos 2017:2). As illustrated in Table 4.37, there are several strategies to plan for pain management as indicated by respondents which they do include in the learning content of pain management education, namely (1) development of a nursing care plan (n = 46), (2) setting expected outcomes to enhance a patient's comfort (n = 46), (3) identifying patients' pain management goals (n = 45), and (4) setting measurable goals to achieve a satisfactory level of pain along with the patient (n = 45) (see Table 4.37).

The strategies to plan for pain management indicated to be included in the learning content of pain management education by the respondents in various wards are illustrated in Table 4.37.

| | Nur | rsing V | Nar | ds | | | | | | | | | | | | | | | | | | | | |
|--|-----|---------|-----|-----|-----|-------|----|-----|-----|-------|-----|---|-----|--------|----|------|-----|-------|----|---|-----|------|----|-----|
| Otrata aire ta alea faa | Me | dical | | | Sur | gical | | | Ра | ediat | ric | | Ob | os-gyn | ae | | Ca | rdiac | | | то | TAL | | |
| pain management | n = | 16 | | | n = | 15 | | | n = | = 7 | | | n = | = 5 | | | n = | - 4 | | | N - | = 47 | | |
| | Yes | 6 | Nc |) | Yes | 5 | No |) | Ye | S | No |) | Ye | S | No |) | Ye | s | No |) | Yes | 5 | No | , |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n | % | n | % |
| Development of a nurs- ing care plan | 15 | 93,8 | 1 | 6,3 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 46 | 97,9 | 1 | 2,1 |
| | 0 | | | | | | | | | | | | | | | | | | | | | | | |
| Setting expected out- comes to enhance pa- tients' comfort | 16 | 100 | 0 | 0 | 14 | 93,3 | 1 | 6,7 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 46 | 97,9 | 1 | 2,1 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Identifying patient's pain management goals | 15 | 93,8 | 1 | 6,3 | 14 | 93,3 | 1 | 6,7 | 7 | 100 | 0 | 0 | 3 | 60,0 | 2 | 40,0 | 4 | 100 | 0 | 0 | 45 | 95,7 | 2 | 4,3 |
| | | | | | | | | | | | | | | | | | | | | | | | | |

Table 4.37: The strategies to plan for pain management (N = 47)

| Setting measurable goals to achieve a sat- isfactory level of pain along with the patient | 15 | 93,8 | 1 | 6,3 | 14 | 93,3 | 1 | 6,7 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 45 | 95,7 | 2 | 4,3 | |
|--|----|------|---|-----|----|------|---|-----|---|-----|---|---|---|-----|---|---|---|-----|---|---|----|------|---|-----|--|
|--|----|------|---|-----|----|------|---|-----|---|-----|---|---|---|-----|---|---|---|-----|---|---|----|------|---|-----|--|

1) Development of the nursing care plan (N = 47)

The nursing care plan addresses patients' pain management needs and provides communication among nurses and other healthcare providers about patients experiencing pain to achieve a satisfactory outcome for the level of pain (Ballantyne 2017:79; Fitzpatrick 2017:531). To manage pain effectively, it is essential to include how to develop a nursing care plan in the learning content of pain management education, as it was indicated and agreed by all respondents in all the wards, exclusive of only one respondent in a medical ward (F = 15; n = 16) that did not agree (see Table 4.45).

2) Setting expected outcomes to enhance patients' comfort (N = 47)

Setting expected outcomes to enhance patients' comfort is part of the nursing care process related to pain management that indicates the result of the care implemented to patients to relieve the pain and maintain patients' functional status, safety, and satisfaction (Baraki, Girmay, Kidanu, Gerensea, Gezehgne & Teklay 2017:8; Fairchild, O'Shea & Washington 2017:329). Table 4.37 indicates that all respondents in all the wards agreed that setting expected outcomes to enhance patients' comfort formed part of the learning content of pain management training, but only one respondent in a surgical ward (F = 14; n = 15) did not agree.

3) Identifying patients' pain management goals (N = 47)

Patients' pain management goals are the aims of the nursing care process used to direct pain management that is patient-centred and to assist the patients to use measurable, attainable, realistic, and timely (SMART) goals to participate in their pain management care, readiness to change their pain management beliefs, self-efficacy and achieve outcomes that are satisfactory to them (Suresh 2017:229). All respondents in all the wards except one respondent in a medical ward (F = 14; n = 15), one in a surgical ward (F = 14; n = 15), and three in obs-gynae wards (F = 2; n = 5) indicated

that they include identification of patients' pain management goals in the study content (see Table 4.37).

4) Setting measurable goals to achieve a satisfactory level of pain along with the patient (N = 47)

Setting measurable goals to achieve a satisfactory level of pain along with the patient are the standards against which to measure or observe a patient's response to pain nursing interventions if a change takes place in a patient's pain experience and severity (Suresh 2017:229). As illustrated in Table 4.37, all respondents in all the wards, excluding only one respondent in a medical ward (F = 15; n = 16) and one in a surgical ward (F = 14; n = 15), agreed that setting measurable goals to achieve a satisfactory level of pain along with the patient formed part of the learning content of their pain module.

The pain intervention strategies implemented to manage the pain were essentially covered in the learning content of pain management training for nurses.

4.4.4.3 Pain intervention strategies implemented (N = 47)

The pain intervention strategies to be implemented form a component of the nursing care process essential to managing patients' pain that must be included in the learning content of pain management education for nurses. The pain intervention strategies reported by respondents to be included, are (1) non-pharmacological interventions (n = 47), (2) pharmacological interventions (n = 47), and (3) patient and family education about pain management (n = 47) (see Table 4.38).

Table 4.38: The pain intervention strategies implemented (N = 47)

| | Nur | sing V | Var | ds | | | | | | | | | | | | | | | | | | | | |
|--|-----|--------|-----|----|-----|-------|----|---|-----|---------|----|---|-----|--------|-----|---|-----|--------|----|---|-----|-------|----|---|
| Pain intervention strat- | Me | dical | | | Sur | gical | | | Pa | aediati | ic | | Oł | os-gyr | nae | | Ca | ardiac | | | TO | TAL | | |
| egies | n = | 16 | | | n = | 15 | | | n = | = 7 | | | n : | = 5 | | | n = | = 4 | | | N - | = 47 | | |
| | Yes | \$ | No | D | Yes | 6 | No |) | Ye | es | No |) | Ye | es | No |) | Ye | es | No |) | Yes | 6 | No | D |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n | % | n | % |
| Non-pharmacological pain interventions | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100,0 | 0 | 0 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Pharmacological pain interventions | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100,0 | 0 | 0 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Patient and family edu- cation about pain man- agement | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100,0 | 0 | 0 |

1) Non-pharmacological interventions (N = 47)

All respondents from all nursing wards indicated that the pain management teaching content included non-pharmacological pain interventions (see Table 4.38).

2) Pharmacological interventions (N = 47)

Pharmacological interventions were included in the learning content of pain management education, as indicated by all study respondents from all nursing wards (see Table 4.38).

3) Patient and family education about pain management (N = 47)

Patient and family education about pain management is the practice of informing patients and families about their health, wellness, treatment plans, potential outcomes, and other information, such as skills specific to self-managing pain, maintaining, coping with new situations, and improving the quality of life (Fereidouni, Sarvestani, Hariri, Kuhpaye, Amirkhani & Kalyani 2019:5). All respondents from all nursing wards in the study agreed that they include patient and family education in the learning content of pain management education for nurses (see Table 4.38).

4.4.4.4 Nursing actions for pain management evaluation (N = 47)

Pain management evaluation relates to the reassessment of the effectiveness or outcomes of the pain interventions implemented (Malik 2020:3; Pickering et al. 2018:97). The nursing actions for pain management evaluation that were reported by respondents to be relevant and included in the learning content of pain management education were (1) assessment of the effectiveness of pain nursing care interventions implemented (n = 47), (2) monitoring of adverse effects of pain medications (n = 47), and (3) pain reassessment after interventions of pain (n = 47) (see Table 4.39).

Table 4.39: The nursing actions for pain management evaluation (N = 47)

| | Nur | sing v | varc | ls | | | | | | | | | | | | | | | | | | | | |
|--|-----|--------|------|----|-----|-------|----|---|-----|--------|----|---|-----|--------|-----|---|-----|--------|----|---|-----|-------|----|---|
| Doin monogoment | Me | dical | | | Sur | gical | | | Ра | ediati | ic | | Ot | os-gyr | nae | | Ca | ardiac | | | то | TAL | | |
| evaluations | n = | 16 | | | n = | 15 | | | n = | = 7 | | | n = | = 5 | | | n = | = 4 | | | N - | = 47 | | |
| | Yes | 6 | No |) | Yes | ; | No |) | Ye | s | No |) | Ye | es | No |) | Ye | s | No |) | Yes | 5 | No |) |
| | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | F | % | n | % | n | % |
| Assessment of the effec- tiveness of pain nursing care interventions | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100,0 | 0 | 0 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Monitoring of adverse ef- fects of pain medications | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100,0 | 0 | 0 |
| | | | | | | | | | | | | | | | | | | | | | | | | |
| Pain reassessment after interventions | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100,0 | 0 | 0 |

1) Assessment of the effectiveness of pain nursing care interventions (N = 47)

Assessment of the effectiveness of pain nursing care interventions implemented is a form of the cycle of evaluation at an appropriate interval after pain nursing interventions by comparing the result of the previous pain rating or situation that indicates the effectiveness or outcome of the pain relief measures applied (Naqib, Purvin, Prasad, Hanna, Dimitri, Llufrio & Hanna 2018: 449; Stannard & Krenzischek 2016:209). All respondents in all the wards reported that the assessment of the effectiveness of pain nursing care interventions implemented was included in the teaching content of pain management training sessions for nurses (see Table 4.39).

2) Monitoring adverse effects of pain medications (N = 47)

Adverse effects related to pain medications refer to any undesired abnormal, harmful, or unpleasant effects resulting from pain medications (Drew, Gordon, Morgan & Manworren 2018:209; Brown, Edwards, Seaton & Buckley 2017:65; Ray 2017:116). Table 4.39 illustrates that all respondents in all the wards reported that monitoring of adverse effects related to pain medications administered to the patients was included in the learning content of the pain management training program.

3) Pain reassessment after interventions (N = 47)

Pain reassessment after pain interventions is part of the constant cycles of pain assessment that include evaluating the effectiveness and outcome of pain interventions to achieve a goal desired during planning with the patient (Brown et al. 2017:73; Keen et al. 2017:138). All respondents in the study agreed that constant pain reassessment after pain interventions was covered in the learning content of pain management education for nurses (see Table 4.39).

4.4.5 Transfer of learning climate within the hospital nursing care

It is essential to identify the characteristics of transfer of learning that support and motivate nurses to apply knowledge in the practice of what was learned before in pain management education (Ma et al. 2018:2). Table 4.40 illustrates the characteristics of transfer of learning climate created within the hospital nursing care and identified as supportive by the respondents to motivate the nurses to apply knowledge in the practice of what was learned before in pain management education. The supportive transfer of a learning climate within hospital nursing care indicated by respondents to be available were: (1) the nursing leadership values the learning needs (n = 47), (2) a pain management orientation program is available (n = 47), (3) clinical facilitators offer constructive feedback to registered nurses (n = 47), (4) there is continuous education by the nurse managers (n = 46), (5) trained nurse preceptors orientate newly employed registered nurses (n = 46), (6) there is support by nurse managers (n = 46), (7) there are dedicated clinical facilities for pain management available (n = 45), (8) registered nurses are freely allowed to apply pain management (n = 45), (9) there is peer support among registered nurses (n = 44), (10) there is support from nursing supervisors (n = 44)40), and (11) non-trained nurse preceptors orientate newly employed registered nurses (n = 33) (see Table 4.40).

| | Nurs | sing Wa | ards | | | | | | | | | | | | | | | | | | | | | |
|---|------|---------|--------|-----|-------|-------|--------|-----|--------|----------|--------|-----|--------|--------|--------|-----|--------|------|--------|-----|-----|------|----|-----|
| | Mec | lical | | | Surg | gical | | | Pae | ediatric | | | Ob | s-gyna | е | | Car | diac | | | TOT | AL | | |
| Transfer of learn- | | | | | | | | | | | | | | | | | | | | | | | | |
| ing climate | n = | 16 | | | n = 1 | 15 | | | n = | 7 | | | n = | 5 | | | n = | 4 | | | N = | 47 | | |
| | Yes | | No | | Yes | | No | | Yes | 6 | No | | Yes | 6 | No | | Yes | 6 | No | | Yes | | No | |
| | F = | % = | F = | % = | F = | % | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | F = | % = | n | % | n | % |
| Nursing leadership values learning needs | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100 | 0 | 0 |
| A pain manage- ment orientation program is availa- ble | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100 | 0 | 0 |
| Constructive feed- back to registered nurses | 16 | 100 | 0 | 0 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 47 | 100 | 0 | 0 |
| Continuous edu- cation by nurse managers | 16 | 100 | 0 | 0 | 14 | 93.3 | 1 | 6.7 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 46 | 97,9 | 1 | 2,1 |
| Trained nurse pre- ceptors orientate new RNs | 16 | 100 | 0 | 0 | 14 | 93.3 | 1 | 6.7 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 46 | 97,9 | 1 | 2,1 |
| Support by nurse managers | 16 | 100 | 0 | 0 | 14 | 93.3 | 1 | 6.7 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 46 | 97,9 | 1 | 2,1 |

Table 4.40: The transfer of learning climate within the hospital nursing care (N = 47)

| Dedicated clinical facilities for pain management are available | 15 | 93.8 | 1 | 6.3 | 15 | 100 | 0 | 0 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 3 | 75.0 | 1 | 25.0 | 45 | 95,7 | 2 | 4,3 |
|--|----|------|---|------|----|------|---|------|---|------|---|------|---|------|---|------|---|------|---|------|----|------|----|------|
| RNs freely allowed to apply pain man- agement skills | 16 | 100 | 0 | 0 | 13 | 86.7 | 2 | 13.3 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 4 | 100 | 0 | 0 | 45 | 95,7 | 2 | 4,3 |
| Peer support of RNs | 15 | 93.8 | 1 | 6.3 | 14 | 93.3 | 1 | 6.7 | 7 | 100 | 0 | 0 | 5 | 100 | 0 | 0 | 3 | 75.0 | 1 | 25.0 | 44 | 93,6 | 3 | 6,4 |
| Support by nurse supervisors | 15 | 93.8 | 1 | 6.3 | 12 | 80.0 | 3 | 20.0 | 7 | 100 | 0 | 0 | 4 | 80.0 | 1 | 20.0 | 2 | 50.0 | 2 | 50.0 | 40 | 85,1 | 7 | 14,9 |
| Non-trained nurse preceptors orien- tate RNs | 13 | 81.3 | 3 | 18.8 | 13 | 86.7 | 2 | 13.3 | 3 | 42.9 | 4 | 57.1 | 2 | 40.0 | 3 | 60.0 | 2 | 50.0 | 2 | 50.0 | 33 | 70,2 | 14 | 29,8 |

1) Nursing leadership values learning needs (N = 47)

The support from nursing leadership was essential to value the learning needs of the nurses regarding the pain management training program to enhance the transfer of learning of pain management competencies of nurses, as illustrated in Table 4.40. All respondents in the study indicated that their nursing leadership valued the learning needs of nurses about pain management to apply knowledge that was learned before in practice (see Table 4.40).

2) A pain management orientation program is available (N = 47)

The availability of a pain management orientation program was essential to support newly employed registered nurses in acquainting themselves with the pain management practice of the hospital (see Table 4.40). This is known as a supportive, transformative process to introduce new incoming nurses to important information and a work environment about pain management (Miller & Geahart 2021:269; Kundu & Lata 2017:5). All respondents in all the wards agreed that a pain management orientation program was available that created a positive learning climate to support newly employed registered nurses to be acquainted with the pain management practice of the hospitals (see Table 4.40).

3) Constructive feedback to registered nurses (N = 47)

Constructive feedback refers to a formal or informal acknowledgement of performance or success and areas for improvement related to pain management provided privately and maintained confidentially (Dwyer & Hopwood 2019:159). Constructive feedback to registered nurses regarding their performance in pain management was reported as support available, as illustrated in Table 4.40. As illustrated in Table 4.40, all respondents in the study agreed that registered nurses were receiving supportive, constructive feedback that created a positive transfer of learning climate within their wards.

4) Continuous education by nurse managers (N = 47)

Continuous education by nurse managers is a constant state of supporting registered nurses to remain current in knowledge or keeping abreast with the latest developments, learning new skills, and new changes in technologies required for pain management (Hughes 2019:166; Wei, Roberts, Strickler & Corbett 2018:685). All respondents in all the wards indicated that the nurse managers offered continuous education to support nurses on pain management, except for one respondent in a surgical ward (F = 14; n = 15) (see Table 4.40).

5) Trained nurse preceptors orientate newly employed registered nurses (N = 47)

Trained preceptors are nurses who consistently demonstrate evidence-based nursing practices to support other nurses after receiving a preceptor education program (Ciocco 2020:6; Maryniak, Markantes & Murphy 2017:6). All respondents in all the wards indicated that the newly employed registered nurses were receiving an orientation about pain management in their wards, except for one respondent in a medical ward (F = 14; n = 15) who disagreed (see Table 4.40).

6) Support by nurse managers (N = 47)

Nurse managers' role is to support nurses in achieving evidence-informed nursing practice regarding pain management (McCleary & McParland 2021:154; Peterson, Berggården, Schaller & Larsson 2019:138). As portrayed in Table 4.40, all respondents in all the wards indicated that nurse managers supported nurses in transferring what they learned from the pain management training program into the workplace, except one participant in a surgical ward (F = 14; n = 15).

7) Dedicated clinical facilities for pain management are available (N = 47)

Dedicated clinical facilities for pain management denote a positive, supportive working environment with available resources to facilitate clinical training of nurses at patient care training sites to learn about pain management (Huber & Joseph 2021:332; Lewis et al. 2017:50; Veal, Williams Bereznicki, Cummings, Thompson & Winzenberg 2017:184). Only one respondent from a medical ward (F = 15; n = 16) and one in a cardiac ward (F = 3; n = 4) did not agree that clinical facilities for pain management were available to support the nurses in applying their knowledge in the practice of what was learned in pain management, contrary to all respondents in the other wards who agreed (see Table 4.40).

8) Registered nurses are freely allowed to apply pain management skills (N = 47)

According to Disha Experts (2021:10) and Na-an, Chaipraist and Pukkeeree (2017:303), registered nurses who are autonomous at work are more motivated to apply their knowledge. According to all respondents in all wards, nurses can apply pain management skills freely (see Table 4.40), except for one respondent in a surgical ward (F = 13; n = 15).

9) Peer support of registered nurses (N = 47)

Peers or co-workers can support registered nurses to apply the new skills gained in pain management from learning programs to the workplace (Chatterjee, Pereira & Bates 2018:14; Ma et al. 2018:4). Table 4.40 illustrates that all respondents from paediatric wards (n = 7) and obs-gynae wards (n = 5) agreed that peers or co-workers can support registered nurses in implementing pain management skills from learning programs to their workplace. However, only one respondent in a medical ward (F = 15; n = 16), one in a surgical ward (F = 14; n = 15), and three respondents in cardiac wards (F = 3; n = 4) disagreed that peers or coworkers could support nurses (see Table 4.40). Table 4.40 illustrates that all respondents from paediatric wards (n = 7) and obs-gynae departments (n = 5) agreed that peers or coworkers could support registered nurses in implementing pain management skills from learning programs.

10) Support by nurse supervisors (N = 47)

The support by nurse supervisors is known to provide a link between hospital management and clinical care as it promotes and encourages the continuity of professional development of registered nurses at the workplace (Ma et al. 2018:4; Mather & Cummings 2017:4). Support by nurse supervisors to nurses about pain management was reported to be available by all respondents in paediatric wards (see Table 4.40). Twelve (n = 16) respondents in medical wards, 12 (n = 15) in surgical wards, four (n = 5) in obs-gynae wards, and two (n = 5) in cardiac wards agreed that nurse supervisors were offering support to the nurses to apply knowledge about what they have learned regarding pain management (see Table 4.40).

11) Non-trained nurse preceptors orientate newly employed registered nurses (N = 47)

As illustrated in Table 4.40, the newly employed registered nurses were reported to receive orientation from non-trained preceptors regarding pain management: 13 (n = 15) respondents in surgical wards, 13 (n = 16) in medical wards, two (n = 4) in cardiac wards, three (n = 7) in paediatric wards, and two (n = 5) in obs-gynae wards (F = 2; n = 5). The study by Innes and Calleja (2018: 71) indicated that, based on experience and practice, the non-trained preceptors could orientate new nurses regarding pain management. However, several studies indicated that nurse preceptors who did not attend preceptor training programs ineffectively and poorly orientate new nurses (L'Ecuyer, Hyde & Shatto 2018:239), contrary to the findings in this study.

12) Non-trained nurse preceptors orientate newly employed registered nurses (N = 47)

As illustrated in Table 4.40, the newly employed registered nurses were reported to receive orientation from non-trained preceptors regarding pain management: 13 (n = 15) respondents in surgical wards, 13 (n = 16) in medical wards, two (n = 4) in cardiac wards, three (n = 7) in paediatric wards, and two (n = 5) in obs-gynae wards (F = 2; n = 5). The study by Innes and Calleja (2018: 71) indicated that, based on experience and practice, the non-trained preceptors could orientate new nurses regarding pain management. However, several studies indicated that nurse preceptors who did not attend preceptor training programs ineffectively and poorly orientate new nurses (L'Ecuyer et al. 2018:239), contrary to the findings in this study.

Respondents further commented on resources available to conduct pain assessment, teaching approaches employed in pain management education, and transfer of learning climate with the hospital nursing care areas.

4.4.6 Thematic content analysis

Eleven respondents provided additional answers to the open-ended question for qualitative enhancement that allowed for written comments (see Annexure 5). After reading through all the narrative data, themes emerged, namely: (1) the availability of pain management resources, (2) a teaching pain management culture, (3) a pain management refresher program, and (4) positive feedback.

1) Availability of pain management resources

The following narratives, as direct quotations, emphasize that there are pain management resources available to assess and manage pain:

"Different resources are easily accessible to staff electronically in the unit".

"Pain management is a well-stabilised program in our organization with highly qualified personnel".

"Pain resource[s] online readily available for staff including documentation parameters".

According to Grossman and Burke-Smalley (2018:242), a facilitated training transfer climate is enhanced if there are sufficient pain management resources available at the workplace to be engaged.

2) A teaching pain management culture

According to respondents, pain management is taught in their wards (see examples below):

"Doing rounds in the morning and ask[ing] each patient for pain as CRN [clinical facilitator], discuss with [the] primary nurse what medications given, [and] educate primary nurses about pain".

"I will be grateful if there is a Pain Management Nurse with direct care in each unit that is dedicated to support both patient and all health care."

This finding is confirmed as it was indicated that further learning about pain management in a practice setting was required to increase learner participation through a collaborative interprofessional team approach to enhance the transfer of learning of pain management competencies (Gordon, Watt-Watson & Hogans 2018:2).

3) Pain management refresher program

Participants mentioned the need for a pain management refresher program as a way to stay current with pain management. The narrative below illustrates this subheading:

"I suggest having from time-to-time refreshment and update sessions for pain assessment and management".

"Nurse Specialist to come quarterly to the unit to give latest updates regarding pain management".

This postulates that providing a pain management refresher program is an effective ongoing post-training booster training strategy that substantially enhances the transfer of learning to sustain recall, retention, and application of the important concepts learned during training in the workplace (O'Donovan, O'Donovan, Kuhn, Sachs & Winters 2018:5; Stuns & Heaslip 2019:198).

4) Positive feedback

Receiving positive feedback about pain management performance can contribute to a positive learning climate in nursing care settings, as reported by one participant:

"Motivate staff and provide feedback to them if they are doing correct documentation about pain".

The study of Hughes, Zajac, Spencer and Salas (2018:4) recommends that providing feedback (for example, performance about pain management) on progress related to how the trained skills were performed motivates the transfer of learning and plays a role in a supportive transfer climate.

4.5 CONCLUSION

This chapter analysed and interpreted data from Phases 1 to 3.

Chapter 5 will discuss Phase 4 – the literature review on action plan development and a draft action plan.

CHAPTER 5: PHASE 4: LITERATURE REVIEW ON ACTION PLAN DEVELOPMENT

5.1 INTRODUCTION

Chapter 5, as illustrated in Table 5.1, describes Phase 4 of the study, which consists of a literature review on action plan development and the process followed to develop the first draft action plan. This phase covered the study's objective of developing an action plan to enhance the transfer of learning of pain management competencies of nurses.

The first draft action plan was developed utilising the analysed data from Phases 1, 2, and 3 of this study, as well as a literature review on the **principles** for drafting an action plan, the **steps to follow in action plan development** and the **Systemic Model of Transfer of Learning**, applicable to this study.

| | Organisation and | structure of the study |
|-------------------|---------------------------------------|---|
| Chapter number | Chapter outline | Chapter content |
| Chapter 1 | Overview of the study. | Contains the introduction, background of the study, the problem statement, research purpose and objec- tives, research question, theoretical framework, key theoretical and operational concepts, the research design and methodology and ethical considerations. |
| Chapter 2 | Literature review. | Consists of the literature review related to: 1) Systemic Model of Transfer of Learning by Donovan and Darcy, 2) Transfer of learning and 3) Pain management and tools. |
| Chapter 3 | Research design and methodol- ogy. | Illuminates the overarching research design. Phases 1, 2, and 3 (quantitative phases): Methodology and Data gathering. |

Table 5.1 Organisation and structure of the study

| Chapter 4 | Data analysis and interpretation. | Presents the data analysis and interpretation of the findings from Phases 1 to 3. |
|-----------|---|---|
| Chapter 5 | Phase 4. | Includes a description of Phase 4 of the study:a) Literature review on action plan development.b) Development of the draft action plan. |
| Chapter 6 | Phase 5. | Outlines and describes Phase 5 of the study (quali- tative phase): a) Methodology. b) Validation of the action plan: c) The action plan. |
| Chapter 7 | Conclusion, recommendations, and limitations. | Deals with the conclusion, recommendations, and limitations of the study. |

5.2 AN ACTION PLAN

An action plan is defined as a written document or systematic way that describes specific actions or steps that must be performed to achieve set goals by identifying all key stakeholders who must execute the activities. The stakeholders can be supervisors, trainers, clinical facilitators, trainees, and co-workers who will implement and sustain the set goals that help to enhance the transfer of learning to performance (Gliva-McConvey, Nicholas & Clark 2020:172; Grisold, Struhal & Grisold 2019:259; Abrahamson & Langston 2017:128; Cunningham & Bennett 2017:22; Saunders 2015:127). In this study, the context was the transfer of learning to improve the pain management competencies of nurses that were addressed.

According to UN-Habitat (2017:5), an action plan is an output-oriented actor-specific plan for achieving the objectives of an issue-specific strategy. It specifies details of inputs and actions by various implementers and or stakeholders, with concrete work programmes, time schedules, resources, key measurables and a timeframe of inputs and outputs, negotiated and agreed upon by the key implementers or stakeholders.

In this context, the goal would be to develop an action plan which would be a helpful tool or method to enhance the transfer of learning of pain management competencies

of nurses. This action plan aimed to achieve positive transfer of learning outcomes that would help nurses function effectively in applying the experience of knowledge and skills gained during pain management training into practice.

This action plan was based on the outcomes of the findings from Phases 1, 2, and 3, and therefore, it would comprehensively provide instructional support and structure to assist and monitor change and progress. It includes a list of steps to be taken by clinical facilitators and nurses to achieve a specific target or goal (pain management skills), identifying the resources required as well as the timescale needed to enhance transfer of learning of pain management competencies as is suggested by Rumble (2019:17) as well as Phillips and Phillips (2016:197).

Various types of action plans could be utilised.

5.2.1 Types of action plans

The literature indicates that there are three sequential categories of types of plans that organisations may use, which are closely linked and have various types of goals, namely strategic plans, tactical plans, and operational plans (Adenov 2021:152; Haux, Winter, Ammenwerth & Brigl 2013:179). To achieve the objective of this study, a mixed type of action plan was developed, as it will be performed by all key stakeholders in nursing, who will be identified to execute the planned activities.

- A strategic plan entails the reasons why things need to happen as considered by the management to maintain the vision and mission of the organisation and covers the setting of overall, long-range goals for an extended period of between five and fifteen years (Gliva-McConvey et al. 2020:172; Griffin 2016:179; Lincke 2015:107; Haux et al. 2013:179). The action plan developed in this study intends to enhance the transfer of learning of pain management competencies of nurses by including the nurse managers on the facility level, thus top management, at the two selected hospitals.
- A tactical plan is a plan related to identifying specific, short-range objectives developed by middle-level managers in different units and sub-units to achieve and execute organisational strategic goals (Griffin 2016:179; Lincke 2015:107). This is

applicable in this developed action plan, which incorporates the relevant stakeholders, such as the nurse managers and supervisors in different nursing care areas responsible for educating and transferring knowledge about pain management competencies to the nurses.

An operational plan is a plan about the "how" that things need to happen and is derived from a tactical plan developed by the lower-level managers stated in specific, quantitative terms related to normal departmental day-to-day activities, tasks, and actions to achieve operational goals and are of a shorter time frame and relatively narrow in scope (Gliva-McConvey et al. 2020:17; Griffin 2016:179). Each element includes the action, task, or change, the people who will carry out these changes, when the deadline is, what resources are needed, and who needs to know (Gliva-McConvey et al. 2020:172). A table format illustrates the developed draft action plan as it outlines the goals and expected outcomes. The table consists of five descriptive columns focusing on the action, and the time frames in which an action needs to be completed. In this illustrative way, all actions could be organised to achieve the objective of this study, namely to enhance the transfer of learning of pain management competencies of nurses.

The development of the action plan, according to the definitions in this context, was guided by the principles and steps for action plan development.

5.2.2 Principles of action plan development

UN-Habitat (2017:6) describes six principles underpinning and guiding the development of action plans. These principles characterise the preparation of an action plan that must be: (1) problem or priority-based, (2) realistic and based on achievable actions, (3) participatory, (4) inclusive, (5) reliant on local resources as well as (6) tangible and concrete. These principles were considered as described below.

5.2.2.1 Problem or priority-based

Establishing what the real problem is, thus the problem statement, means that specific issues and their main causes must be briefly described and addressed as priorities in

an action plan (Haigney 2021:54; O'Cathain, Croot, Duncan, Rousseau, Sworn, Turner, Yardley & Hoddinott 2019:3; UN-Habitat 2017:6). In this study, the general problem statement was described (see Section 1.2). The specific challenges or aspects relevant to the transfer of identified learning problems were targeted and prioritised, as was attained from the findings of Phases 1, 2, and 3.

5.2.2.2 Realistic and based on achievable actions

The action statements in the draft action plan were the specific challenges or aspects which could deter transfer of learning that needed to be addressed with specific actions or activities to enhance the likelihood of transfer of learning of pain management competencies of nurses as was identified by the analysis of data from Phases 1, 2, and 3.

The specific actions or activities nominated for inclusion in the planned draft action plan were to be within the competencies and capabilities of the stakeholders involved (UN-Habitat 2017:6). To achieve the objectives of the study, a specific set of specific, measurable, attainable, realistic, and time-bound (SMART) goals were developed to specify the expected outcomes of the action plan to enhance the transfer of learning of pain management competencies of nurses as explained by various authors (Gliva-McConvey et al. 2020:172; O'Cathain et al. 2019:3 Covaerts, Kyndt, Vreye & Dochy 2017:543).

5.2.2.3 Participatory

Authors emphasize the active participation of all stakeholders involved in the development of action plans, as their full engagement as well as their claim of ownership of developed action plans (UN-habitat 2017:7; O'Cathain et al. 2019:3). A participatory approach ensures the development of realistic and achievable objectives that consider the capacities of the participants, determine a realistic timeframe, and instil a sense of ownership and commitment of all involved (Whicher, Harris, Beverley & Swiatek 2018:3247; UN-habitat 2017:7). In this study context the following stakeholders, relevant to the transfer of learning of pain management skills and competencies, actively participated. In Phases 1 and 2, the registered nurses from the participating hospitals provided data, while in Phase 3, clinical facilitators of the same two hospitals participated and provided data. In Phase 4, the panellists consisting of nurses and clinical facilitators from all the participating hospitals are indicative of the active involvement of all stakeholders in developing and agreeing on the content of the action plan, ensuring taking ownership of the developed action plan.

5.2.2.4 Inclusive

Inclusive implies that the planned activities and actions included in the action plan must be viewed from diverse perspectives. In this study context, diverse perspectives refer to the transfer of learning, considering the characteristics, transfer-oriented educational design, and learning needs of all trainees (as nurses) and the work context (Fauth & González-Martinez 2021:11; UN-Habitat 2017:7; Donavan & Darcy 2011:123). Inclusivity in this context entails the utilisation of the questionnaires to gather first, second, and third-phase data and a thorough literature review. The findings of Phases 1, 2, and 3 and the literature reviews were used to develop the action plan. As a result, it was developed based on the dimensions that could enhance the transfer of learning. These factors include the characteristics of nurses, the design of pain management education programs, the climate of transfer of learning within hospital nursing care, and the working environments of nurses.

5.2.2.5 Reliant on local resources

Action plans should make the best possible use of the human, technical and financial resources that are locally relevant and available (O'Cathain et al. 2019:3; UN-Habitat 2017:7). In this context, the availability of resources and the existing training in the selected hospitals are vital to effective pain management. It is also crucial to align an action plan with the institution's master strategic plan (Gaddy 2014:6). By including stakeholders knowledgeable about designing pain management education programs, ensuring the transfer of a learning climate within the hospital nursing care, and ensuring that nurses' workplace environment create opportunities for them to practice the skills acquired during training of pain management due to their involvement in facilitating training opportunities in pain management.

The developed action plan aims to enhance the transfer of learning of pain management competencies of nurses in alignment with the five-year General Strategic Plan for the two hospitals published for the years 2019-2023 as cited from the Ministry of National Guard Health Affairs' Strategic Plan (2019:25). The general strategic plan of the two hospitals aimed to:

- Create an environment that promotes the highest quality patient care with the highest quality care within resources, as per the draft plan, to provide adequate resources for nurses to implement new learning in pain management that is transferable and usable in their work areas effectively.
- Promote clinical excellence in patient safety and satisfaction by aligning the learning content of the short course program in pain management to reflect all types of pain patients may experience.
- Ensure a competent workforce by developing a diverse and inclusive workforce through motivation, innovation, and skill improvement while drafting an action plan to improve nurses' knowledge and competencies in managing pain and developing a pain management short course program incorporating the different learning styles of nurses and the learning content that covers all types of pain experienced by patients.
- Create an interactive environment that attracts nurses by encouraging the pursuit of national and international competencies with highly competitive benefits, in alignment with this draft action plan that promotes a supportive learning environment where nurses of all categories value pain management education designed and delivered by nurses.
- Achieve financial sustainability. This aligns with this draft action plan's objective of providing pain management resources affordably to enhance the transfer of learning (Ministry of National Guard Health Affairs' Strategic Plan 2019:25) (see Tables 5.2 and 5.3).

5.2.2.6 Tangible and concrete

A tangible and concrete action plan that includes clearly defined specific expected outcomes and quantifiable measures by which the progress would be easily evaluated was developed as suggested by UN-Habitat (2017:7) and Whicher et al. (2017:3237). Based on the developed action plan, specific, measurable goals and objectives were identified to anticipate the results of the actions planned. Furthermore, the draft action plan aims to enhance nursing competency transfer in pain management by achieving the actual expected outcomes (see Table 5.3.).

5.2.3 Steps in developing an action plan

The steps adopted from McGurk and Mueser (2021:115) were followed to develop an action plan to enhance the transfer of learning of pain management competencies of nurses. The steps followed were: (1) the description of the problem; (2) the identification of the best possible solutions to the problem; (3) the identification of the people responsible to address each action statement contained in the plan; (4) determination of the specific resources needed; (5) the implementation of the action plan (validation) and (6) a follow-up meeting or evaluation of the action plan (not done in this study context).

5.2.3.1 Step 1: Description of the problem

The first step in developing the first draft action plan was to describe and define the problem identified as specifically as possible, as explained by McGurk and Mueser (2021:115) and Brown-Chidsey and Bickford (2015:77). The problem was described (see Section 1.2), and the data to describe the problem was obtained from (i) nurses and clinical facilitators in Phase 1, namely the identification of the resources available to conduct pain assessments. (ii) In Phase 2, data from nurses who identified the characteristics and learning styles that enhance pain management competency transfer were obtained. (iii) In Phase 3, data from the clinical facilitators revealed the teaching approaches employed during nurses' pain management education, the learning content regarding pain assessment and management, and the transfer of a learning climate within the hospitals' nursing care areas. The data analysed from Phases 1, 2, and 3 addressed the specific challenges or problems relevant to the transfer of learning that had to be addressed in the action plan to enhance the transfer of learning skills to improve pain management as suggested by Pilcher (2016:185), Babu and Gayathri (2018:102) as well as by Brown-Chidsey and Bickford (2015:73), The specific

challenges or problems relevant to the transfer of learning were categorised under the dimensions of the Systemic Model of Transfer of learning applicable to this study (see Table 5.2).

5.2.3.2 Step 2: Identification of the best possible solutions to the problem

The second step was to identify the best possible solutions to the identified problems in developing a **complete**, **clear**, **and specific** action plan as recommended by McGurk and Mueser (2021:115), Kirwan (2016: 59), as well as Brown-Chidsey and Bickford (2015:75). The specific problems to be mitigated in the draft action plan, were identified from the analysed data from Phases 1, 2, and 3. The best possible solutions to be implemented were written as action statements relevant to the transfer of learning. The e-Delphi method (a "collective agreement and consensus method") discussed in Chapter 6 was used to validate the draft action plan as suggested by Keeney, Hasson and Mckenna (2011:43). To achieve the ultimate expected outcomes as the best solution to the identified problems, all the action statements were aligned with the detailed specific actions, set of SMART goals, methods to be used, and all stakeholders to participate as is recommended by Covaerts et al. (2017:543), also by Ford, Baldwin and Prasad (2018:202) (see Table 5.2 and 5.3).

5.2.3.3 Step 3: Identification of the people responsible to implement the plan

For the third step, McGurk and Mueser (2021:115) and Brown-Chidsey and Bickford (2015:79) suggest that the stakeholders responsible for implementing the action plan must be identified. Hence, the draft action plan identified the clear responsibilities of responsible persons such as nurse managers, nurse supervisors, clinical facilitators, nurse educators, pain nurses, and clinical nurse specialists as relevant to implement the action plan to enhance transfer of learning of pain management of nurses' competencies as is indicated in Table 5.3.

5.2.3.4 Step 4: Determine the specific resources needed

According to McGurk and Mueser (2021:116), the specific resources needed to develop an action plan must be determined to accomplish its implementation. A panel of relevant stakeholders, two clinical facilitators (one from each hospital), and ten nurses (five from each nursing care division in the two hospitals) participated in the validation process. The final action plan was based on consensus, including consensus regarding the resources required (people, finances, and pain tools) and the individuals or groups who needed to take responsibility for each action in the action plan.

5.2.3.5 Step 5: Implementation of the action plan (In the study context, the validation of the action plan)

The fifth step is the implementation of the final action plan drafted, as indicated by Brown-Chidsey and Bickford (2015:78) as well as DiPaola and Hoy (2013:100). In this study context, the action plan was not implemented since a longitudinal study would have been needed, not compatible with the time limitations of a PhD study. The panel members validated the proposed action plan (see Chapter 6, Section 6.4) and took three e-Delphi rounds to reach a consensus of 75% among themselves (see Chapter 6, Section 6.6). The final action plan to enhance the transfer of learning of pain management competencies of nurses is presented in Chapter 6, Section 6.7.

5.2.3.6 Step 6: Follow up on the action plan

Step 6 entails follow-up meetings to evaluate whether the action plan was effective, modifying it as needed (McGurk and Mueser 2021:116). This step would require a follow-up study, as Brown-Chidsey & Bickford (2015:80) advise.

5.2.4 Application of the Systemic Model of Transfer of Learning

The Systemic Model of Transfer of Learning adopted as the framework of this study was applied to guide the development of an action plan to enhance the transfer of learning of nurses' pain management competencies. For this purpose, it was used to enhance the nurses' transfer of learning of pain management competencies as a response to identified problems possibly affecting the transfer of learning.

Illustrated in Figure 5.1 (discussed in Chapter 2 Section 2.4), the Systemic Model of Transfer of Learning designates four dimensions of factors that influence the transfer of learning in the workplace. These dimensions are (1) trainee characteristics, (2) training design, (3) training transfer climate and (4) workplace environment characteristics

(Donovan and Darcy 2011:125; Ma et al. 2018:2; Botma & MacKenzie 2016;105; Schneider 2014:68). Higher nursing education institutions have also adapted and utilised this model to achieve their principal goal, to promote the integration of theory and practice (Donovan & Darcy 2011:125; Botma et al. 2015:499; Botma & MacKenzie 2016:105).



Figure 5.1 Systemic Model of Transfer of Learning

Source: Adopted from Donovan and Darcy (2011), cited in Donovan and Darcy (2011:125).

The action plan development process was aligned with the four dimensions of factors of the Systemic Model of Transfer of Learning. What was considered in this study were (1) nurses' characteristics, (2) pain management education program design, (3) transfer of learning climate, and 4) nurses' working environment (Donovan and Darcy 2011:125; Ma et al. 2018:2; Botma & MacKenzie 2016;105; Schneider 2014:68; Boonpektrakul 2014:10).



Figure 5.2 Application of Systemic Model of Transfer of Learning as aligned from the citation by Donovan and Darcy (2011:125)

5.2.4.1 Nurses' characteristics

Nurses' characteristics were presented in Chapter 4 based on an analysis of data collected during Phase 2. Discussions include demographics, application of what has been learned previously, learner types, motivation to learn, and motivation to apply knowledge. All the mentioned characteristics, identified as affecting the transfer of learning of pain management competencies, were considered when drafting the action plan's first draft. This draft action plan aimed to enhance nurses' abilities to learn, analyze, and apply what has been learned at work, as suggested by Boonpektrakul (2014:10).

5.2.4.2 Pain management education program design

Findings from Phase 3, the perspectives of clinical facilitators, were aligned with the Systemic Model of Transfer of Learning training design recommended by Donovan and Darcy (2011:130), Herr et al. (2015:319), and gil, Mataveli and Garcia-Alcaraz (2021:2). Figure 5.2 illustrates the incorporation of the pain management education program design to the systemic model. Therefore, the action plan in this study

considered the design of pain management education programs to enhance the transfer of pain management competencies, including learning styles, teaching approaches, and program content. Cartaxo and Simes (2014:0460) and Schneider (2014:86) recommend that a draft action plan include learning that corresponds to a nursing situation and nurse needs in real-time.

5.2.4.3 Transfer of learning climate within hospital nursing care

As defined by the Systemic Model of Transfer of Learning, the transfer of learning within the hospital nursing care climate is determined by the relationship between the nurses' motivation to transfer learning in their working environment (Donavan & Darcy 2011:123). The clinical facilitators identified a transfer of learning climate across various wards in Phase 3. Furthermore, this draft action plan is aligned with the systemic model, which is based on clinical facilitators' perspectives across various wards, as indicated in this study's findings regarding the types of learning climate that contribute to the transfer of pain management skills (suggested by Boonpektrakul 2014:11). Data analysed from Phase 3 were used to develop an action plan aimed at ensuring a supportive transfer of learning climate for pain management training.

5.2.4.4 Working environment

It is important to consider workplace environment factors when formulating part of the action plan that motions trainees to have opportunities to practice new skills acquired during training, as suggested by Donovan and Darcy (2011:123), Suleiman et al. (2016:23), and Yusin et al. (2014:184). The draft action plan was aligned with the model by ensuring that nurses' work environments were supportive and conducive to conducting pain assessments, as highlighted in Phases 1 and 3 by the nurse and clinical facilitator responders. The analysed data from Phases 1, 2, and 3, as summarized in Table 5.2, were integrated with the available literature to identify possible action statements for the action plan to address the mentioned challenges.

5.2.5 Development of the draft action plan

The action plan was developed after reviewing the literature on how an action plan should be developed, distinguishing the types of action plans, the principles to be applied, and steps to be followed from the analysed data from Phases 1, 2, and 3, integration of data from Phases 1, 2, and 3, and using the System Model of Transfer of learning to guide the process of goal attainment while developing the draft action plan. The actual draft action plan was developed based on the identified needs and recommendations obtained during the integration of findings from Phases 1, 2, and 3 (see Table 5.2). Results from Table 5.2 were used to inform the preliminary draft action plan (see Table 5.3).

Table 5.2: Integration of Phases 1, 2, 3, and literature support

| RESEARCH FINDINGS: PHASE 1,2 and 3: Demography | Literature support | Action state- ment number | Action Statement |
|---|---|------------------------------------|---|
| Only 23,6% of nurses (see Figure 4.3) and 10,6% of nurse facilitators were diploma-prepared (see Figure 4.11). | Nurses with Bachelor's degrees prove to have significantly higher pain management knowledge and attitude levels than those at the Diploma and Certificate level (Kahsay & Pitkäjärvi 2019:1; Alotaibi, Higgins & Chan 2019:123). | 1 | Motivate nurses to further their studies. |
| RESEARCH FINDINGS: PHASE 1 & 3: Available pain assessment tools | Literature Support | | Action statement |
| The QUEST pain assessment guide tool utilised in paediatric wards was reported to be available by only 9,4% of nurses (see Table 4.5) and by 25,5% of nurse facilitator respondents (see Table 4.17). | To identify pain in children, the QUEST pain assessment tool is recommended by Nori, Benjamin, Alshalji and Izakovic (2019:148) to utilise in pediatric wards. | 2 | Make appropriate and relevant pain management tools acces- sible to the nursing team in every clinical area. |
| The COLDSPA assessment guide tool was reported to be available by 41,8% of nurses (see Table 4.5) and by 29,8% of nurse facilitators (see Table 4.17). | COLDSPA is an available and useful mnemonic tool to interview patients about pain (Shorey, Ang, Yap, Ng, Lau & Chui 2019:5). | | |
| 41,9% of nurses (see Table 4.5), and 34,0% of nurse facilitators reported the OPQRSTUV assessment guide tool available (see Table 4.17). | The OPQRSTUV instrument is relevant to collecting pain management information systematically for postoperative pain assessment (Permana and Widagdo 2019:13). | | |

| Only 69,1% of nurses indicated the PQRST pain as- sessment guide tool available (see Table 4.5). | For pain self-reporting, nurses can use mnemonics of PQRST to help them remember key aspects of pain assessment (Fink & Gallagher 2019:231). | |
|--|---|--|
| Only 17,1% of nurses (see Table 4.6) and 23,4% of nurse facilitators confirmed the BPI available (Table 4.18). | BPI is highly reliable for use with dementia patients, yet nurses do not adequately use it in clinical settings (Guetti, Paesani, Colavincenzo, Ciccozzi & Angeletti 2018:3; Liao, Parajuli, Jao, Kitko & Berish 2021:11). | |
| 35,6% of nurses (see Table 4.6) and 23,4% of nurse facilitators (see Table 4.18) reported the VDS available. | VDS tool is not as regularly used as other pain scales (Res- nick, Boltz, Galik, Holmes, Vigne, Fix & Shijun 2019:194). | |
| 23,4% of nurse facilitators indicated the VAS available (see Table 4.18). | Nurses know that the VAS could be used for pain assessment (Pölkki, Korhonen, and Laukkala 2018:729). | |
| 21,0% of nurses (see Table 4.7) and 23,4% of nurse facilitators (see Table 4.19) reported the NPASS available. | The NPASS is a commonly used scale to rate pain in neona- tal patients (Carcia-Rodrigues, Bujan-Bravo, Seijo-Bestlleiro & Gozalez-Martin 2021:5929). | |
| 21,8% of nurses (see Table 4.7) and 25,5% of nurse facilitators (see Table 4.19) confirmed the CPOT available. | CPOT is feasible and useful in critical care (Maatouk, Tassi, Fawaz, and Itani 2019:6). | |
| 24,7% of nurses (see Table 4.7) and 27,7% of nurse facilitators (see Table 4.19) reported that NIPS was available. | NIPS is a tool that is available and commonly used by nurses (Obiedat and Al-Maaitah 2020:4). | |
| 27,8% of nurses (see Table 4.7) and 36,3% of nurse facilitators (see Table 4.19) confirmed the BPS available. | The BPS is used in the intensive care unit for patients who are unable to self-report their pain (Hermes, Acevedo-Nuevo, Berry, Kjellgren, Negro & Massarotto 2018: 55). | |
| 30,1% of nurses (see Table 4.7) and 23,4% of nurse facilitators (see Table 4.19) confirmed the COM-FORT B scales available. | The COMFORT-B pain scale is used for preverbal children younger than three years of age, according to Andersen, Nakstad, Jilli, Cambell-Yeo and Anderzen-Carlsson (2019:341). | | |
|---|--|--|--|
| 65,2% of nurses (see Table 4.7) and 76,6% of nurse facilitators (see Table 4.19) confirmed the CRIES pain scale available. | The CRIES scale can be used to assess acute and postop- erative pain in neonates (Popowicz, Kwiecie-Jagu, Ol- szewska & Mdrzycka-Dbrowska 2020:1895). | | |
| The ABBEY pain tool was reported available by 17.4% of nurses (see Table 4.11) and 12,8% of nurse facilitators (see Table 4.20). | ABBEY pain tools are used to assess elderly patients with dementia or cognitive impairment in palliative situations, whether all scale items apply or not (Tegenborg, Fransson & Martinson 2020:9). | | |
| The PAINAD scale was reported to be available by 19,5% of nurses (see Table 4.11) and 12,8% of nurse facilitators (see Table 4.20). | PAINAD scale is used to assess pain in patients with demen- tia and cognitive impairment (Natavio, McQuillen, Dietrich, Wells, Rhoten, Vallerand & Monroe 2020:507). | | |
| 20,8% of nurses (see Table 4.11) and 14,9% of nurse facilitators reported the NOPPAIN tool available (see Table 4.20). | The NOPPAIN is a commonly used observation tool available for assessing pain in patients with dementia (Rababa 2018:64). | | |
| 26,5% of nurses (see Table 4.11) and 29,8% of nurse facilitators (see Table 4.20) indicated the CNPI scale available. | CNPI is used to assess pain in dementia patients (Smith & Harvey 2022:7). | | |

| 76,6% of nurses reported that nurse supervisors with pain management training were available (see Table 4.9). | Nurse supervisors trained in pain management are regarded as a resource for introducing nurse-based pain management programs (Germossa, Hellos & Sjetne 2019:3). | |
|--|--|--|
| 58,7% of nurses (see Table 4.10) and 40,6% of nurse facilitators (see Table 4.22) confirmed the availability of patient pain management websites. | Websites, including the British Pain Society, European Pain Society, and Irish Pain Society, provide information on pain management (Schofield 2018: i7). | |
| 63,4% of the nurses (see Table 4.10) and 36,2% of nurse facilitators (see Table 4.22) confirmed the availability of patient pain management hotlines. | Nursing teams utilise pain management hotlines to support patients and families in managing their pain (Chan, Lin, George & Liu 2020:672). | |
| Pain management support groups were reported available by 76,4% of nurses (see Table 4.10) and 59,6% of nurse facilitators (see Table 4.22). | Support groups that assist other patients with persistent pain, including online adolescent support groups, are important in pain management (Tolley, Michel, Williams, and Renschler 2020:2). | |
| 70.2% of nurse facilitators (see Table 4.22) reported the availability of "peer" support groups. | Adolescent support groups can be useful in the treatment of chronic pain in patients with pain or peers (Tolley et al. 2020:2). | |
| 44,4% of the nurses (see Table 4.11) and 29,8% of the nurse facilitator respondents (see Table 4.23) confirmed the e-newsletters from the selected publi- cations and electronic resources available. | Patients with pain want to receive information using e-news- letters (Goldstein, Majdi, Bocher-Planka, Watts, and Khan 2018:299). | |

| 50,4% of nurses (see Table 4.11) and 38,3% of nurse facilitators (see Table 4.23) reported the vide eos on pain management available. | f A video on pain management is a useful resource for im- proving health professionals' pain management competency and evidence-based practices (Hurley-Wallace, Wood, Franck, Howard & Liossi 2019:43). | |
|---|--|--|
| 51,4% of nurses (see Table 4.11) and 40,4% of nurse facilitators (see Table 4.23) reported the fac sheets available. | f Fact sheets to advise on pain management in children with intellectual disabilities are useful (Genik, McMurtry, Breau, Lewis & Freedman-Kalchman 2018:434). | |
| Printed reference books aimed at pain managemer were reported available by 57% of nurse facilitator (see Table 4.11). | t Printed reference books are used as scientific resources for pain management (Smeland, Twycross, Lundeberg &Rutten 2018:549). | |
| A pain toolkit was reported available by 70,2% on nurse facilitators (see Table 4.23). | f QuantiPain is effective in clinical settings (Izumi, Hayashi, Saito, Petersen, Arendt-Nielsen & Ikeuchi 2022:8). | |
| 74,8% of nurses (see Table 4.11) and 72,3% of nurse facilitators (see Table 4.23) reported the clinical updates available. | f Evidence-based sources aimed at supporting patients in pain management assist in navigating pain management nursing practice in global settings (Rosa 2018:27). | |
| Organizations that specialise in pain managemer were reported available by 78.4% of nurses (see Ta ble 4.12) and by 78.7% of nurse facilitators (see Ta ble 4.24). | t The World Health Organization, American Pain Society, Na- tional Cancer Network, and other organisations that special- ise in pain management provide pain management guide- lines (Germossa, Sjetne & Helles 2018:3). | |

| RESEARCH FINDINGS: PHASE 2: Application of what was learned before. | Literature support | | Action statement |
|--|--|---|---|
| Nurses applied pain management training in their clinical practice and were able to: Promptly assess patients' pain (31,51%, see Figure 4.4). | Nurses trained in pain management can effectively assess pain by applying what they have learned (Chu, Wang, Lin, Lee, Lin, Chieh, Sung & Lin 2019:6; Malones, Kallmyr, Hage & Eines 2021:202). | 3 | Develop a practice-oriented, content-specific short pain management training program. |
| Accurately assess pain (38.8%, see Figure 4.4). | Nurses trained in pain management are able to assess pain accurately (Chu et al. 2019:6). | | |
| Select appropriate pain intervention strategies for in- dividual patient's pain levels (45,05%, see Figure 4.4). | Nurses who had received pain education were able to imple- ment pain treatment and medication measures effectively (Chu et al. 2019:6). | | |
| Understand and apply the content information taught (57,03%, see Figure 4.4). | As a result of pain management training, nurses displayed the ability to comprehend and apply the program's contents (Germossa et al. 2018:4). | | |
| Train new colleagues in pain management (57,81%, see Figure 4.4). | Nurses who receive pain management training can train other nurses (Osongo 2020:285). | | |

| Recall information about pain management from previous experience (60,16%, see Figure 4.4). | Nurses trained in pain management training have the ability to recall the information (Grenning, Nøst, Rannestad & Bratts 2018:8). | | |
|---|---|---|---|
| Attend pain management learning/training sessions (75,00%, see Figure 4.4). | Nurses with previous training in pain management have con- siderable knowledge applicable to clinical settings (Kahsay & Pitkäjärvi 2019:10). | | |
| Reassess pain after interventions (75,52%, see Figure 4.4). | Pain management training enhances nurses' competencies (Bonkowski, Gagne, Cade & Bulla 2018). | | |
| Concentrate well during pain management learn- ing/training sessions (78,13%, see Figure 4.4). | Nurses can pay attention during pain management training sessions (Waszak, Mitchell, Ren & Fennimore 2018:343). | | |
| Nurse facilitators included labour pain as part of the pain management training program learning content (74,5%, see Table 4.34). | Labor pain education improves pain alleviation (McCauley, Danna, Mrema & Van den Broek 2018:8). | | |
| Nurses described the type of learners they were: Creative learners (53.25%, see Figure 4.5). | Nurses who are creative learners can apply new ideas in practical settings to overcome complex challenges (Yang, Zhou, Chung, Tang, Jiang & Wong 2018:43) | 4 | Develop a pain management short program that accommo- dates all learning types. |
| | | | |

| Enthusiastic thinking learners (56.25%, see Figure 4.5). | Learning environments that incorporate daily activities motivate nurses with a desire to learn (Habes, Jepma, Parlevliet, Bakker & Buurman 2020:5). | |
|--|--|--|
| Self-confident thinkers (56.25%, see Figure 4.5). | Critical thinking assists self-confident nurses in applying their knowledge (Liu, Yu, Wang, Zhu & Yang 2021:4). | |
| Organised thinking learners (58.59%, see Figure 4.5). | Organized thinkers are able to apply information in practice (Forneris & Peden-McAlpine 2021:4). | |
| Truth-seeking learners (62.00%, see Figure 4.5). | Nurses who seek the truth have critical thinking and clinical competence (Tajvidi & Hanjani 2019:5). | |
| Inquisitive thinkers (66.40%, see Figure 4.5). | Inquisitiveness boosts the probability that nurses will be able to apply their knowledge to challenging clinical situations (Yang et al. 2018:43). | |
| Diligent, inquisitive learners (67.19%, see Figure 4.5). | Inquisitive nurses have a wide range of experience, enthusi- asm, curiosity, and resilience for gaining knowledge and de- veloping skills that they then apply to become experts (Si- klander & Impio 2019:1247). | |
| Self-directed learners (77,60%, see Figure 4.5). | Self-directed nurses are able to remain current and apply their learning (Curran, Gustafson, Simmons, Lannon, Wang & Garmsiri 2019:85). | |

| Nurses described their learning styles and indicated that they learn if: They could question the information they have ob- tained (13,80% see Figure 4.8). | Nurses with over 26 years of experience were shown to pre- fer verbal learning strategies such as questioning or explain- ing knowledge and were more likely to transfer their learning (Mangold, Kunze, Quinonez, Taylor & Tenison 2018:212). | |
|--|---|--|
| They could record the lecture (14.58%, see Figure 4.8). | As a result of using video, audio, and animation in the inter- vention group, nurses were able to transfer more information (Farshbaf-Khalili, Jasemi & Seyyedzavvar 2021:4). | |
| They could learn from the internet (16, 93%, see Figure 4.8). | Multimedia-assisted training for nurses enhanced learning transfer by aligning with their preferred learning style (Chu et al. 2019:7). | |
| They could take control of their own learning (20,05%, see Figure 4.8). | Nurses who preferred self-directed learning as their pre- ferred learning method enhanced their learning transfer (Alharbi 2018:233). | |
| They could generate creative ideas in a group (21,88%, see Figure 4.8). | Learners participating in a group setting are stimulated to generate creative ideas (Stefan 2021:38). | |
| They could be in a silent environment (23,18%, see Figure 4.8). | Learners participating in quiet group settings have a higher concentration on learning (Yang, Zhao, Tian & Xing 2021:407), | |
| They could connect the information they already know (35,16%, see Figure 4.8). | Student nurses' learning improves when new information is linked to previous knowledge (Walter 2018:26). | |
| They were able to write down the information (37,76%, see Figure 4.8). | Writing down text, although a learning style, is the least pre- ferred by degree nurses (Zhu, Zeng, Zhang, Zhang, Wan, Guo & Zhang 2018:164). | |

| They could personally take part in the activity to practice the skill (42,45%, see Figure 4.8). | Nurses who attend pain management training that focuses on practical application enhance the transfer of learning (Mi- naya-Freire, Ramon-Aribau, Pou-Pujol, Fajula-Bonet & Subirana-Cascuberta 2020:499). | | |
|--|---|---|--|
| They could participate in the group discussion (51,04%, see Figure 4.8). | Participating in group discussions is among the factors that enhance learning transfer (Brion & Cordeiro (2018:6). | | |
| They could solve different real-life pain management problems (51,04%, see Figure 4.8). | Transfer of learning is enhanced when nurses apply their pain management abilities in actual clinical settings (Sardo, Galletta, Coni, Gonzalez, Piras, Pia, Evangelista, Musu & Finco 2020:2356). | | |
| They could listen to the information such as during a lecture (77,08%, see Figure 4.8). | Nurses have indicated that unimodal methods of learning, such as lectures, cassettes, and stories, are their preferred learning styles (Zhu et al. 2018:165). | | |
| RESEARCH FINDINGS: PHASE 3: | | | |
| Teaching approaches employed | Literature support | | Action statement |
| Nurses employed the following teaching approaches: Writing reflective journals for sharing their pain management experiences (36,2%, see Table 4.26). | Pain management competencies can be applied through writing reflection journals (Carvalho, Pereira, Jácomo, Magalhäes, Araŭjo, Hernández-Marrero, Gomes & Schatman 2018:972). | 5 | Incorporate different teaching approaches to accommodate diverse learners and facilita- tors. |

| Nursing grand rounds to provide pain management training to nurses (74,5% see Table 4.26). | Conducting nursing grand rounds enhances the transfer of learning (Salinas, Johnson, Conrardy, Adams & Brown 2019:41). | | |
|--|---|---|---|
| Focus groups are a teaching strategy to help them manage pain (76,6%, see Table 4.26). | Focus group discussions can enhance the transfer of learn- ing (Chicca & Shellenbarger 2018:183). | | |
| Role-playing activities to teach nurses pain management (78%, see Table 4.26). | Role play can enhance the transfer of learning (Pilnick, Trusson, Beeke, O'Brien, Goldberg & Harwood 2018:3). | | |
| RESEARCH FINDINGS: PHASE 2: Motivations to participate in the training program. | Literature support | | Action statement |
| Nurses were motivated to participate in the training program: Due to their goals and desire to know about pain management (63,28%, see Figure 4.6). | Nurses' goals motivate them to feel competent and use their new skills in the workplace (Ahlstedt, Lindvall, Holmström & Athlin 2019:36). | 1 | Develop strategies to motivate nurses to participate in the short training program. |

| If they know their learning goals (65,10%, see Figure 4.6). Because of the ability to take the initiative without assistance (71,86%, see Figure 4.6). | The knowledge of the learning goals and learning outcomes motivates students and nurses to learn and apply what they have learned before (Guswara & Purwanto 2020:429). Individuals who take the initiative without assistance are self- directed learners motivated to learn and meet their objectives to apply what they have learned (Zhoc, Chung & King 2018:997). | - | |
|--|--|---|--|
| Nurse facilitators indicated that newly hired regis- tered nurses received pain management orientation from non-trained preceptors (70,2%, see Table 4.40). | Preceptor-training programs help preceptors communicate knowledge to nurses (Hugo 2018:131). | | |
| RESEARCH FINDINGS: PHASE 2: | | | |
| Motivations to apply knowledge in practice. | Literature support | | Action statement |
| Because they were motivated to apply what they had learned in practice (58.33%, see Figure 4.7). | Student nurses are motivated to obtain knowledge and skills in pain management in order to become competent nursing professionals in the future (Damsgaard, Solgaard, Johan- nessen, Wennevold, Kvarstein, Pettersen & Garcia 2018:523). | 7 | Motivate nurses to apply the knowledge gained in the training program into practice. |

5.3 THE FIRST DRAFT ACTION PLAN

The first draft of the action plan is displayed in Table 5.3. The draft action plan includes the following primary categories: action statements, specific actions, methods suggested to assist in achieving the action aim and expected outcome, the responsible person(s), and suggested time frames. The embedded validation assessment instrument was also tested for validity and reliability, as discussed in detail in Chapter 6 subsection 6.2.4

Table 5.3: First Draft action plan with embedded validation tool.

Instructions to Panellists

Please indicate your answer by selecting either "Agree" or "Disagree" next to the options provided to indicate your agreement or disagreement with the inclusion of the (1) action statements as well as the (2) methods suggested.

Inclusion of each action statement in the action plan.

The methods suggested to assist in achieving the action goal and expected outcome.

Please choose from the list provided the best possible responsible person/ persons that you recommend must take responsibility for every method.

Please choose from the list provided the best possible timeframe within which every method must be completed once the action plan is accepted.

Please provide detailed suggestions and comments in all the suggestion/comment boxes for improvement of any of the items/aspects in the draft action plan. Action statement 1: Motivate nurses to further their studies

| Agree | Disagree | | |
|---|----------|--|--|
| Method 1.1 Develop a policy to motivate nurses to improve their nursing qualifications. | | | |
| | | | |
| Agree | Disagree | | |
| | | | |

If you agree that a policy should be developed, please tick all that should be included in the policy to motivate the nurses.

| Paid Full-time study leave for 1 year | |
|---|--|
| 20 hours paid study leave and 20 hours full-time work for 1 year | |
| Free accommodation for the period of study leave | |
| A monetary Incentive after completion of a new formal qualification (degree or diploma) | |
| One day off for attending a one-day pain management program | |
| A monetary incentive after completion of a pain management program | |
| A certificate issued as an acknowledgement of nurses pursuing distance learning | |
| A monetary incentive after completion of distance learning programs | |

| Responsible person(s): Please select (check the box(es)) next to the best person(s) to be responsible for the development of a policy that motivates furthe | r |
|---|---|
| studies. | |

 Five Members of the Central Region Nursing Governance and Accountability Board appointed by the Chief Executive Director for Central Region, Riyadh

 Ad hoc committee appointed by the heads of the Human Resource Department Central Region, Riyadh

 Associate Executive Directors of Nursing for King Abdulaziz Medical City (KAMC) and King Abdullah Specialist Children's Hospital (KASCH)

 Nursing policy committee representatives appointed by the Associate Executive Directors of Nursing for KAMC and KASCH

 Clinical Directors of Nursing Operations for KAMC and KASCH

 Director of Postgraduate Center of Nursing Education for KAMC and KASCH

| Time frame: Select the most appropriate timeframe after approval of the action plan, within which time the policy should be developed and finalised. | | | |
|--|--|--|--|
| | | | |
| 1 1–3 months | | | |
| 2 4–6 months | | | |
| 3 7–9 months | | | |
| | | | |
| | | | |
| Mathematic Descent and a matical for the implementation of the network | te nesti eta numera de inconse de inconseiro e configuratione de de a Ministro e Albediane I | | |
| Method 1.2 Present and negotiate for the implementation of the policy i | to motivate nurses to improve their nursing qualifications to the Ministry of National | | |
| Guard Health Affairs (MINGHA) through the Central Region Nursing Gov | ernance and Accountability Board | | |
| | | | |
| | | | |
| Agree | Disagree | | |
| | | | |
| | | | |
| Responsible person(s): Please select (check the box(es)) next to the beat | st person(s) to present and negotiate for the implementation of the policy to motivate | | |
| nurses to improve their nursing qualifications. | | | |
| | | | |
| Five Members of the Central Region Nursing Governance and | Accountability Board appointed by the Chief Executive Director in Central | | |
| Region, Riyadh | | | |
| Ad hoc committee appointed by the heads of the Human Res | source Department Central Region, Riyadh | | |
| Associate Executive Directors of Nursing for KAMC and KAS | CH | | |
| Nursing policy committee representatives appointed by the A | ssociate Executive Director of Nursing for KAMC and KASCH | | |
| | | | |
| | | | |
| | | | |

| 13 months26 months39 months | | | |
|--|--|--|--|
| 2 6 months 3 9 months | | | |
| | | | |
| | | | |
| Method 1.3 Include the policy in all hospitals` policies after approval by the Ministry of National Guard Health Affairs (MNGHA) | | | |
| Agree Disagree | | | |
| Responsible person(s): Please select (check the box(es)) next to the best person(s) to be responsible for the inclusion of the policy in the hospitals' policies to motivate nurses to improve their nursing qualifications. | | | |
| Associate Executive Directors of Nursing for KAMC and KASCH Clinical Directors of Nursing Operations in every facility for KAMC and KASCH Nurse Managers in all KAMC and KASCH nursing care areas | | | |

Time frame: Select the most appropriate timeframe within which time the policy should be included in the hospital policies after approval by the Ministry of National Guard Health Affairs

| 1 | 1 month |
|---|-----------|
| 2 | Six weeks |
| 3 | 2 months |
| 4 | 3 months |

Suggestions/comments

Action statement 2: Make appropriate and relevant pain management tools accessible to the nursing team in every clinical area

| Agree | Disagree |
|---|--|
| Method 2.1 Include pain assessment tools in an elec | tronic patient record system so that the nursing team can choose and access the most appropriate. |
| tool in all pursing care areas | |
| | |
| Agree | |
| If you agree that appropriate pain assessment tools | should be included, please tick all that should be accessible on the electronic patient record system. |
| | |
| | |
| | |
| QUEST | |
| COLDSPA | |
| OPQSTUV | |
| PQRST | |
| BPI | |
| VDS | |
| VAS | |
| NPASS | |

| _ | | | | |
|----|------------------------------------|-------------------|--|-----------------------|
| | CPOT | | | |
| | BPS | | | |
| | COMFORT B | | | |
| | CRIES | | | |
| | ABBEY | | | |
| | PAINAD | | | |
| | NOPPAIN | | | |
| | CNPI | | | |
| | | | | |
| F | Responsible person(s): Please | select (check t | the box(es)) next to the best person(s) to be responsible to provide the electronic format o | f the pain assessment |
| | tools for inclusion in the electro | onic patient rec | ord system to be accessible to the nursing team in every nursing care area. | |
| | | | | |
| | Two computer eng | ineers appointe | ed by the HR manager, each assigned for KAMC and KASCH | |
| | Five Nursing health | informatics ap | pointed by Associate Directors of Nursing for KAMC and KASCH | |
| | Clinical Director of | Nursing Operat | tions appointed by Associate Executive Directors in every facility for KAMC and KASCH | |
| | One Pain nurse sp | ecialist appointe | ed by Clinical Directors of Nursing Operations in every facility for KAMC and KASCH | |
| | | | | |
| | | | | |
| | | | | |
| | Time from a Cale at the most of | | | |
| | Time frame: Select the most a | ppropriate time | frame within which time the pain assessment tools should be included in the electronic p | atient record system. |
| | | | | |
| | 1 1–3 months | | | |
| | 2 4–6 months | | | |
| | 3 7–9 months | | | |
| с. | | | | |

| Method 2.2 Involve nurse supervisors with pain management training to provide post-pain management training supervisory support to the nursing team on how to conduct pain assessments in all nursing care areas. | | | |
|---|--|--|--|
| | | | |
| Agree | Disagree | | |
| Responsible person(s): Please select (check the box(es)) the best per | rson(s) responsible to involve nurse supervisors with pain management training to | | |
| provide post-pain management training and supervisory support to the r | ursing team on how to conduct pain assessments in all nursing care areas. | | |
| | | | |
| Clinical Directors of Nursing Operations for KAMC and KASC | Эн сама сама сама сама сама сама сама сам | | |
| Nurse Managers in all KAMC and KASCH nursing care area Clinical facilitators in all KAMC and KASCH nursing care are | | | |
| Charge nurses in all KAMC and KASCH nursing care areas | | | |
| Time frame: Select the most appropriate timeframe within which time the nurse supervisors should be involved to provide pain management training. | | | |
| supervisory support to the nursing team on how to conduct pain assessment in all nursing care areas. | | | |
| | | | |
| | | | |
| 1 Every shift when the need arises | | | |
| 2 Every patient round when the need arises | | | |
| Method 2.3 Make hospitals' internet-based resources accessible to the patients and family members to obtain other support about pain management | | | |
| Agree | Disagree | | |
| | | | |
| 2.3.1 If you agree that hospitals' internet-based resources should be acc be available. | essible to provide support to patients and family members, please tick all that should | | |
| | | | |

| 1 Patient pain management websites | |
|---|---------------------------------|
| 2 Patient pain management hotlines | |
| 3 Pain management support groups | |
| 4 Peer support groups | |
| Responsible person(s): Please select (check the box(es)) the best person(s) responsible to ensure internet-based resources sh | nould be accessible to patients |
| and family members to provide support about pain management. | |
| | |
| | |
| | |
| Chief Executive Director for KAMC and KASCH | |
| Two health information technologists appointed by the HR manager each assigned for KAMC and KASCH | |
| Two communication and information administrators appointed by the HR manager for KAMC and KASCH | |
| Five Nursing health informatics appointed by Associate Directors of Nursing for KAMC and KASCH | |
| Associate Executive Directors of Nursing for KAMC and KASCH | |
| Clinical Directors of Nursing Operations for KAMC and KASCH | |
| Nurse supervisors appointed by Executive Associate Directors of Nursing in every facility for KAMC and KASCH | |
| Nurse Managers in all KAMC and KASCH nursing care areas | |
| Charge nurses in all KAMC and KASCH nursing care areas | |
| Registered nurses in all KAMC and KASCH nursing care areas | |
| | |
| | |
| | |

Time frame: Select the most appropriate timeframe within which time the hospitals' internet-based resources should be accessible to provide support to patients and family members.

Every day at a convenient time 1 24-hour access 7 days a week 2

Method 2.4 Make hospitals' internet-based resources on pain management publications and electronic materials accessible to the nursing team in all nursing care areas.

Agree

Disagree

2.4.1 If you agree that hospitals' internet-based resources on pain management publications, electronic materials, and organisations that specialises in pain management should be accessible to the nursing team in all nursing care areas, please tick all that should be available.

| 1 | E-newsletters | |
|---|---|--|
| 2 | Videos on pain management | |
| 3 | Facts sheets | |
| 4 | Pain toolkits | |
| 5 | Clinical updates | |
| 6 | The World Health Organization | |
| 7 | American Pain Society | |
| 8 | International Association for Study of Pain | |

Responsible person(s): Please select (check the box(es)) the best person(s) responsible for ensuring internet-based resources on pain management publications, electronic materials, and organisations that specialise in pain management should be accessible to the nursing team in all nursing care areas.

| Two computer engineers appointed by the HR manager each assigned for KAMC and KASCH | |
|--|---|
| Two librarians appointed by the HR manager each assigned for KAMC and KASCH | |
| Nursing health informatics appointed by Associate Directors of Nursing for KAMC and KASCH | |
| Nurse Managers in all KAMC and KASCH nursing care areas | |
| Nurse supervisors appointed by Executive Associate Directors of Nursing for KAMC and KASCH | |
| One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility for KAMC and KASCH | 1 |
| One Nurse educator appointed by the Director of the Postgraduate Nursing Education Center for KAMC and KASCH | |

Time frame: Select the most appropriate timeframe within which time the hospital's internet-based resources on organisations that specialise in pain management, pain management publications, and electronic materials should be accessible to the nursing team in all nursing care areas.

| 1 | Every nursing shift | |
|---|------------------------------|--|
| 2 | 24-hour access 7 days a week | |

Suggestions/comments

Action statement 3: Develop a practice-oriented content content-specific short pain management training program

| Agree | Disagree | | |
|--|----------|--|--|
| Method 3.1 Include practice-oriented pain management training content for all nursing care areas in the pain management program. | | | |
| Agree | Disagree | | |
| | | | |

| If you agree that specific practice-oriented pain management content should be included in all nursing care areas, please tick all that should be included. | | | | |
|---|--|--|--|--|
| Methods to promptly assess a patient's pain in all nursing areas | | | | |
| Assessment of patients' pain in all nursing care areas | | | | |
| The advantages and disadvantages of all pain management scales | | | | |
| Labour pain as a type of pain to be assessed | | | | |
| The selection of appropriate pain intervention strategies based on the pain levels assessed. | | | | |
| Responsible percent(c): Please select (check the bex(es)) the best percent(c) to be responsible for including specific practice criented pain management | | | | |
| training content for all nursing care areas. | | | | |
| Disaster of Depteroducts Contex of Numing Education for KANC and KACCU | | | | |
| Director of Postgraduate Center of Nursing Education for KAMC and KASCH | | | | |
| One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Education for KAMC and KASCH | | | | |
| One Clinical lacinitator appointed by Clinical Directors of Nursing Operations in every fursing care area | | | | |
| | | | | |
| Time frame: Select the appropriate timeframe within which time the specific practice-oriented pain management training content should be provided inclusion in the program. | | | | |
| 1 Week before the due date of the training program | | | | |
| 2 1 month before the due date of the training program | | | | |
| 3 3 months before the due date of the training program | | | | |
| | | | | |
| Suggestions/comments | | | | |

| Action statement 4: Develop a pain management short program that accommodates all learning types | | | | |
|---|---|--|--|--|
| Agree | Disagree | | | |
| Method 4.1 Incorporate different learner types during learning/training sessions. | | | | |
| Agree | Disagree | | | |
| If you agree that different learner types of nurses should be included du included during learning/training sessions. | ring pain management learning/training sessions, please tick all types that should be | | | |
| Creative Learners | | | | |
| Enthusiastic thinking learners | | | | |
| Self-confident thinkers | | | | |
| Organised thinking learners | | | | |
| Truth-seeking learners | | | | |
| Inquisitive thinkers | | | | |
| Diligent inquisitive learners | | | | |
| Self-directed learners | | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |

| | Creative Learners (generate creative ideas in a group) | |
|----------------|---|------------|
| | Enthusiastic thinking learners (listen to the information actively, take part in the activity to practice the skill and participat | e in the |
| | group discussion) | |
| | Self-confident thinkers (take control of their learning) | |
| | Organised thinking learners (solve different real-life problems) | |
| | Truth-seeking learners (question the information obtained) | |
| | Inquisitive thinkers (inquire about the information obtained) | |
| | Diligent inquisitive learners (take the initiative in their learning) | |
| | Calf directed learners (take, control of their learning) | |
| oons ent | ble person(s): Please select (check the box(es)) the best person(s) to be responsible for ensuring that the learning objectives earner types of nurses during pain management learning/training sessions. | accommodat |
| pons | ble person(s): Please select (check the box(es)) the best person(s) to be responsible for ensuring that the learning objectives earner types of nurses during pain management learning/training sessions. | accommodat |
| pons rent | ble person(s): Please select (check the box(es)) the best person(s) to be responsible for ensuring that the learning objectives earner types of nurses during pain management learning/training sessions. Director of Postgraduate Center of Nursing Education for KAMC and KASCH One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Education for KAMC and KASCH Clinical facilitators in all areas of nursing care for KAMC and KASCH | accommodat |
| pons rent | Self-directed learners (take control of their learning) ble person(s): Please select (check the box(es)) the best person(s) to be responsible for ensuring that the learning objectives earner types of nurses during pain management learning/training sessions. Director of Postgraduate Center of Nursing Education for KAMC and KASCH One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Education for KAMC and KASCH Clinical facilitators in all areas of nursing care for KAMC and KASCH One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility | accommoda |
| pons erent | ble person(s): Please select (check the box(es)) the best person(s) to be responsible for ensuring that the learning objectives earner types of nurses during pain management learning/training sessions. Director of Postgraduate Center of Nursing Education for KAMC and KASCH One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Education for KAMC and KASCH Clinical facilitators in all areas of nursing care for KAMC and KASCH One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility ne: Select the appropriate timeframe within which time the learning types must be shared for inclusion within the training program | accommodat |
| pons rent | Self-directed learners (take control of their learning) ble person(s): Please select (check the box(es)) the best person(s) to be responsible for ensuring that the learning objectives earner types of nurses during pain management learning/training sessions. Director of Postgraduate Center of Nursing Education for KAMC and KASCH One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Education for KAMC and KASCH Clinical facilitators in all areas of nursing care for KAMC and KASCH One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility ne: Select the appropriate timeframe within which time the learning types must be shared for inclusion within the training program | accommodat |
| pons rent l | Self-directed learners (take control of their learning) ble person(s): Please select (check the box(es)) the best person(s) to be responsible for ensuring that the learning objectives earner types of nurses during pain management learning/training sessions. Director of Postgraduate Center of Nursing Education for KAMC and KASCH One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Education for KAMC and KASCH Clinical facilitators in all areas of nursing care for KAMC and KASCH One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility ne: Select the appropriate timeframe within which time the learning types must be shared for inclusion within the training program 2 weeks before the due date for finalisation of the training program 1 month before the due date of the training program | accommodat |

| Suggestions/comments | | | | |
|--|---|--|--|--|
| Action statement 5: Incorporate different teaching approaches to accom | modate diverse learners and facilitators in the training of pain management | | | |
| Agree | Disagree | | | |
| Method 5.1 Ensure the inclusion of different teaching approaches in the | offering of the training program. | | | |
| Agree | Disagree | | | |
| 5.1.1 If you agree that different teaching approaches should be utilised during pain management training that accommodates diverse learners and facilitators, please tick all that should be utilized during training. 1 Writing reflective journals 2 Conducting grand rounds 3 Engaging in focus groups 4 Using role-playing activities | | | | |
| Responsible person(s): Please select (check the box(es)) next to the best person(s) to be responsible to ensure that teaching approaches are included during pain management training. Director of Postgraduate Center of Nursing Education for KAMC and KASCH One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Education for KAMC and KASCH Clinical facilitators in all areas of nursing care for KAMC and KASCH One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility | | | | |
| | | | | |

| Time frame: Select the appropriate timeframe within which time different teaching approaches be part of the teaching program before implementation. | | | | |
|--|--|--------------------------|--|--|
| | | | | |
| | | | | |
| 1 | 2 weeks before the due date for finalisation of the training prov | gram | | |
| 2 | 1 month before the due date of the training program | | | |
| 3 | 3 months before the due date of the training program | | | |
| | | | | |
| | | | | |
| | | | | |
| Sugge | stions/comments | | | |
| Cuyye | | | | |
| | | | | |
| Action | statement 6: Develop strategies to motivate nurses to participate in the | e short training program | | |
| Aaree | Disa | 20166 | | |
| / groo | | | | |
| 6.1 If you agree that strategies that motivate the nurses to participate in the pain management training program should be developed, please tick all strategies | | | | |
| that yo | ou think will motivate nurses. | | | |
| | | | | |
| 1 | Conduct a situation analysis to assess the pain management needs of | of the nurses. | | |
| 2 | Involve nurses in the development of the content of the training progra | am. | | |
| Involve nurses in the development of learning goals and learning outcomes for the pain management training program relevant to their | | | | |
| nursing care areas. | | | | |
| 3 | | | | |
| 4 Communicate the advantages of pain management competencies on (on what platform can this be done. | | | | |
| 5 Create a supportive learning environment in nursing care areas. | | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |

| I nurses to participate in the short pain management training program related to their nursing care areas. | | | | |
|--|--|--|--|--|
| nurses to participate in the short pain management training program related to their nursing care areas. | | | | |
| | | | | |
| | | | | |
| | | | | |
| Director of Postgraduate Center of Nursing Education for KAMC and KASCH | | | | |
| One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Education for KAMC and KASCH | | | | |
| Clinical facilitators in all areas of nursing care for KAMC and KASCH | | | | |
| One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility | | | | |
| | | | | |
| | | | | |
| | | | | |
| Time frame: Select the appropriate timeframe within which the strategies that motivate nurses to participate in the short pain management training program | | | | |
| related to their nursing care areas should be developed. | | | | |
| | | | | |
| 1 2 weeks before the due date for the finalisation of the training program | | | | |
| 2 1 month before the training program starts | | | | |
| | | | | |
| | | | | |
| | | | | |
| Suggestions/comments: | | | | |
| Action statement 7: Motivate nurses to apply the knowledge gained in the training program into practice | | | | |
| | | | | |
| | | | | |
| | | | | |

| Method 7. | 1 If you agree that nurses should be motivated, kindly tick all you think that can be done to motivate them to apply their I | nowle | dge in practice. | |
|--|--|----------|------------------|----|
| | | | | |
| | | | | |
| | | | | |
| 1 | Provide nurses with a certificate to recognize their application of pain management knowledge in their respective nursir | ng care | e areas. | |
| 2 | Offer nurses the opportunity to take on the role of pain management experts who are competent in their field. | | | |
| 3 | Allow the nurses to take part in planning outcomes of a pain management training program. | | | |
| 4 | Support nurses' SMART goals and pain management learning. | | | |
| 5 | Support what drives individual nurses to apply what they have learned about pain management. | | | |
| 6 | Assign grades for applying pain management knowledge in practice based on annual performance. | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| | | | | |
| Responsit | ale person(s): Please select (check the box(es)) pext to the best person(s) to be responsible for facilitating the implem | entati | on of the aspect | to |
| motivate r | urses to apply their knowledge in practice | icitiati | | 10 |
| motivater | | | | |
| | Associate Executive Directors of Nursing for KAMC and KASCH | | 1 | |
| | Clinical Directors of Nursing Operations for KAMC and KASCH | | | |
| | Nurse supervisors appointed by Executive Associate Directors of Nursing in every facility for KAMC and KASCH | | | |
| - | Nurse Managers in all pursing care areas for KAMC and KASCH | | | |
| | Director of Destaroducts Conter of Nursing Education for KANC and KASCH | | | |
| One Nurse educator eppeinted by the Director of the Destanduate Center of Nursing Education for KAMC and KASCH | | | | |
| | One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Education for KAIVIC and KASCH | | | |
| | One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility | | | |
| | | | | |
| | | | | |
| | | | | |

| Time fra practice. | nme: Select the appropriate timeframe withir | n which time the | "incentives" 1 | for motivation | should be | provided t | o nurses to | apply their | knowledge in |
|-----------------------|--|------------------|----------------|----------------|-----------|------------|-------------|-------------|--------------|
| 1 | 1 week after the training program | | | | | | | | |
| 2 | 1-3 months after the training program | | | | | | | | |
| 3 | 4-6 months after the training program | | | | | | | | |
| 4 | 7-9 months after the training program | | | | | | | | |
| | | | | | | | | | |
| | | | | | | | | | |
| | | | | | | | | | |
| | | | | | | | | | |
| Suggest | ions/comments | | | | | | | | |

5.4 CONCLUSION

The chapter outlined the literature review on types of action plans, the principles of action plan development, steps in developing an action plan, the application of the Systemic Model of Transfer of Learning, the development of the draft action plan integrating the data from Phases 1, 2, and 3, and the first draft of an action plan with the validation tool embedded. The validation tool, embedded in the draft action plan, was presented to explain how panellists were allowed to directly comment on each action, method, responsible persons, and the time frames allocated to each action.

Chapter 6 will entail a discussion of Phase 5, the validation process of the draft action plan.

CHAPTER SIX: PHASE 5: METHODOLOGY, VALIDATION PROCESS AND FINAL ACTION PLAN

6.1 INTRODUCTION

Chapter 6, as illustrated in Table 6.1, outlines Phase 5 of the study. This chapter describes the qualitative component of the explanatory sequential mixed method approach, thus the methodology used (e-Delphi), data collection and analysis, interpretation of the validation findings, and the final validated action plan.

| Organisation and structure of the study | | | | |
|---|---------------------------------------|--|--|--|
| Chapter number | Chapter outline | Chapter content | | |
| Chapter 1 | Overview of the study. | Contains the introduction, background of the study, the problem statement, research purpose and ob- jectives, research question, theoretical framework, key theoretical and operational concepts, the re- search design and methodology and ethical consid- erations. | | |
| Chapter 2 | Literature review. | Consists of the literature review related to: Systemic Model of Transfer of Learning by Do- novan and Darcy, Transfer of learning and Pain management and tools. | | |
| Chapter 3 | Research design and methodol- ogy. | Illuminates the overarching research design. Phases 1, 2 and 3 (quantitative phases): Methodol- ogy and Data gathering. | | |
| Chapter 4 | Data analysis and interpretation. | Presents the data analysis and interpretation of the findings from Phases 1 to 3. | | |
| Chapter 5 | Phase 4. | Includes a description of Phase 4 of the study: Literature review on action plan development. | | |

Table 6.1 Organisation and structure of the study

| | | Development of the draft action plan. |
|-----------|--|---|
| Chapter 6 | Phase 5. | Outlines and describes Phase 5 of the study (quali- tative phase): Methodology. Validation of the action plan: The action plan. |
| Chapter 7 | Conclusion, recommendations, and limitations | It deals with the conclusion, recommendations and limitations of the study. |

6.2 METHODOLOGY

In Phase 5 of the study, a qualitative design was used to collect data from a panel of professional nurses and clinical facilitators using the e-Delphi technique. The purpose was to validate the draft action plan in order to formulate the final action plan that contributes to enhancing the transfer of learning of pain management competencies of nurses.

6.2.1 The Delphi technique

The Delphi method, developed in the 1950s by Rand Corporation, is generally used to gain consensus among experts on a specific topic (Gluszel 2021:2). It can be defined as a multi-staged or iterative survey used to collect and distil knowledge from anonymous opinions among a group of experts or panel of informed individuals in a specific field (stakeholders) of application with controlled feedback to reach consensus on an important issue (Keeney et al. 2011:3).

This iterative survey method uses a questionnaire, in this case, a validation tool (see Annexure 13) to collect and analyse data, interspersed with controlled feedback with each stage building on the results of the previous one (Izaryk & Skarakis-Doyle 2017:1225; Lloyd & Stirling 2015:3; Meskell et al. 2014:32; Chung et al. 2014:6145;

Grove et al. 2013:691; Polit and Beck 2021:236; Keeney et al. 2011:3). Consensus is reached by, firstly, generating ideas, secondly, by organising and structuring these ideas, and by combining and or addressing individual judgements should there be a lack of agreement (Gruszek 2021), until a 75 % agreement is reached.

A panel of registered nurses (n = 10) and clinical facilitators (n = 2), identified as knowledgeable about pain management, participated as panellists. Through their independent and anonymous inputs in various rounds, the panellists reached a consensus after three rounds. The purpose of this was to validate the action plan to enhance the transfer of learning of pain management competencies among Saudi Arabian nurses.

The Delphi technique belongs to dialectic consensus development methods that resolve differences between views. It is employed to seek an improved position satisfactory to all participants (Grove et al. 2013:435; Keeney et al. 2011:7). The types of Delphi's are (1) Classical or consensus Delphi; (2) Modified Delphi; (3), Decision Delphi (4) Policy Delphi; (5) Real-Time Delphi; (6) e-Delphi; (7) Technological Delphi; (8) Online Delphi; (9) Argument Delphi and (10) Disaggregative Delphi (Grove et al. 2013:435; Keeney et al. 2011:7). For this study, the **e-Delphi technique** was employed.

The electronic-Delphi ("e-Delphi") survey involves the administration of the Delphi by e-mail or through completion of an online form (Hall, Smith, Heffernan, Fackrell (2018:2), Keeney et al. (2011:7). In this study context the panellists gained access to the draft action plan with an embedded validation tool, through a link accessible from the recruitment letters (see Annexures 10, 14). This allowed panellists to anonymously provide their inputs, opinions, and suggestions on the validation tool.

The following "four key characteristics" as stated by Gluszek (2021:3), Hall et al. (2018:2), Rai et al. (2017:3230), Skinner et al. (2015:33), Chung et al. (2014:6145), and Keeney et al. (2011:7) were maintained during the e-Delphi validation process:

6.2.1.1 The anonymity of Delphi participants was maintained. They could freely express their opinions because a gatekeeper shared the recruitment letter with the eligible experts (panellists) via e-mail. They could access the

action plan and validation tool via a link in the recruitment letter on Google Forms, after which the researcher received raw data via the software program Google Form[®]. Thus, the e-mail addresses of the panellists were not known to the researcher, and the raw data could not be linked to the specific panellist.

- **6.2.1.2** Iteration allowed the panellists to refine their views considering the progress of the group's feedback. However, they did this individually, from one round to the next.
- **6.2.1.3 Controlled feedback** informed the participants of the other panellists' combined perspectives by sharing a revised action plan and validation tool with panellists after each round.
- 6.2.1.4 Statistical aggregation of group responses allowed for quantitative analysis and interpretation of data until all panellists reached a consensus of 75%. Items rated below this level were altered or discarded depending on the inputs received in each round until **75%** on every item was reached.

6.2.2 Advantages of e-Delphi technique.

The e-Delphi technique for obtaining group consensus was the method of choice to validate the action plan based on the following advantages, as described by Izaryk and Skarakis-Doyle (2017:1226) and Gluszek (2021: 23):

6.2.2.1 Collecting opinions from a wide geographical area.

The opinions of nurses and clinical facilitators were gathered employing an electronic Delphi tool that was distributed via email and did not require in-person interaction. This enhanced accessibility for panellists from any location and enhanced the response rate to the Delphi tool. The response rate for Rounds 1 and 2 was 100%, while in Round 3 it was 83.3%.

6.2.2.2 Anonymity.

The Delphi participants' identity was kept anonymous. The recruitment letter was shared with the eligible experts (panellists) via e-mail by a gatekeeper. By clicking on

the link in the recruitment letter, they would be able to access the action plan and validation tool, following which analysed data (quantitative opinions) was received through the software program Raw Narrative Data (in bulk) as answers to open-ended questions were received from the software program. Data could, therefore, not be linked to any individual panellist.

6.2.2.3 The iterative process

In an iterative process, the panellists could refine their views based on the group's feedback from round to round (Hall et al. 2018:3).

6.2.3 Criticisms and limitations of the Delphi technique

Even though evidence from the literature indicates that the Delphi method is susceptible to criticisms and limitations as described by Gerrish and Lathlean (2015: 274) and Keeney et al. (2011:20), these criticisms were addressed in this study as follows:

- 6.2.3.1 Lack of universal guidelines by allowing flexibility and gaining consensus. In this study, the rounds continued until a consensus was reached. The panellists knew the rounds would continue until consensus (see the recruitment letters, Annexure 9, 10, 14 and 19) and agreed to participate in all rounds. In this study, three rounds were conducted to gain a consensus.
- 6.2.3.2 The existing literature does not provide a specific or predetermined sample size for the Delphi panel to constitute a representative sample. Keeny et al. (2011:53) and Rai et al. (2017:3233) suggest that if the sample is a homogenous panel, as it is in this study, a sample size of 10-15 is adequate. Therefore, a panel of twelve individuals participated as panellists in the study.
- 6.2.3.3 A lack of true anonymity. Anonymity was ensured in this study. The panellists were recruited by gatekeepers, who shared the recruitment letter with the eligible experts (panellists) via e-mail. The link to the validation tool was included in the recruitment letter, and the inputs of panellists were received in bulk via the software program. The researcher

had no contact with any individual panellist, and the gatekeeper did not have access to raw data. It was not possible to link any data with an individual.

- **6.2.3.4 Expert opinions** are difficult to obtain when selecting the Delph panellists, as identifying them is not always easy. In this study, the gatekeepers received a letter explaining their role as gatekeepers and the requirements about who will be eligible to be an expert panellist (see Annexure 16).
- 6.2.3.4 Misconceptions related to identifying and measuring consensus are contentious within Delphi literature. This research gave panellists ten days to assess the plan and provide feedback and recommendations. After receiving feedback, the researcher assembled and analysed it. The amended drafts of the action plan were shared exclusively with the panellists through gatekeepers, who sent them (with a new recruitment letter) with a link. This distribution method was based on analysing and validating the action plan in each round of the Delphi process. The process was continued until there was a 75% consensus, as recommended by Keeney et al. (2011:27). The researcher conducted three rounds. 6.2.3.5 There is a view that the Delphi technique is demanding by nature as it is time-consuming to complete the process. To overcome time constraints, all panellists were given ten days for each round to respond and were motivated by effectively communicating the objectives and their contribution to the study, leading to participation until the last round.

6.2.4 The Validation Instrument

It was impossible to separate the draft action plan from the validation instrument as each aspect of the action plan had to be validated. Therefore, the validation instrument, embedded as part of the action plan, was developed to ensure a rigorous validation process.
The draft action plan was developed after reviewing the literature on how an action plan should be developed, distinguishing the types of action plans, the principles to be applied, and the steps to be followed. The action plan was informed by the analysed data from Phases 1, 2, and 3 and using the System Model of Transfer of Learning to guide the process of goal attainment. The draft action plan was therefore developed based on the identified needs and recommendations obtained during the integration of findings from Phases 1, 2, and 3 (see Table 5.3).

It was necessary to develop a validation instrument embedded in the draft action plan to validate the action plan (see Table 5.3). The embedded validation assessment instrument allowed the panellists to reflect on and provide opinions on every aspect (see Table 5.3).

6.2.4.1 The characteristics of the validation instrument

The design of the validation instrument was compiled and embedded in the action plan. It was carefully considered and built into the Google Forms® software program. This is a free online toolkit for gathering and managing data based on research aims to be achieved. Google Forms[®] was selected since it does not require login details. Thus, access was easy for all panellists who clicked on the link in the recruitment letter. This speeds up the time for the feedback collection process as Google Forms is a free online tool from Google that provides several shortcuts that allow panellists to navigate and complete the form faster. The validation instrument, embedded in the action plan, consisted of:

- Section A: Contains **biographical information** of the panellists who participated in the validation process.
- Section B: Entails the action statements, methods to revive the actions, persons responsible for the actions, timeframes, and a space for comments. Panellists could agree or disagree with any statement or action and could tick appropriate boxes where they had to indicate their choice (see Figure 6.1, a copy of part of a Google Form, illustrative of the embedded validation instrument). At the end of each sub-item, there was space for

| comments in | the form | of an | open-ended | question | for | qualitative | enhance- |
|---------------|------------|-------|------------|----------|-----|-------------|----------|
| ment (see Fig | gure 6.1). | | | | | | |

| Action statement 1: Motivate nurses to further their studies * |
|--|
| Agree |
| O Disagree |
| |
| Please add any comment pertaining to this action statement |
| Short answer text |
| |
| Method 1.1 Develop a policy to motivate nurses to improve their nursing qualifications * |
| O Agree |
| O Disagree |
| |

Figure 6.1: A copy of a part of the Google Form illustrative of a draft action plan with the embedded validation instrument Before data collection, the draft action plan, with the embedded validation tool, was subjected to pre-testing to determine if the panellists understood the wording, whether and where clarity was needed, or where any other amendments were required (see Section 6.2.5 for the pre-test results). To enhance the validation instrument, it was also tested for the draft action plan.

6.2.5 Population

From the target population as described in Section 3.5, 12 participants, consisting of two clinical facilitators from both hospitals (see Table 6.2), were selected. The population was deemed appropriate due to its composition of nurses with significant pain management experience.

6.2.6 Sample

A **purposive non-probability sampling** strategy was used to select the 12 Delphi panellists who qualified as **experts** in the subject matter.

"Experts" are a group of "informed individuals" and "specialists in the field" with knowledge of a particular field (Keeney et al. 2011:7). In this context, they were the nurses and clinical facilitators who complied with the following criteria:

- a) Must have had an interest in pain management.
- b) Nurses who had attended at least one pain management workshop within the past two years.
- c) Nurses who had attended ward in-service training about pain management in the past 12 months.
- d) Clinical facilitators (clinical resource nurses) responsible for the pain management training of nurses in those nursing care divisions mentioned above.
- e) Be committed to at least three rounds of Delphi (see Annexure 16)

The panel recruited by the gatekeepers consisted of two clinical facilitators (one from each hospital) and ten nurses (five from each hospital) (see Table 6.2). The

purposively selected sample of panellists was, as suggested by various authors (Baran & Galka 2016:117; Rubin & Babbie 2014:669; Carter, Lubinsky & Domholdt 2013:99), selected because they were able to provide high-quality responses and information because of their comprehensive understanding of the subject.

As indicated by Rai et al. (2017:3233), Keeney et al. (2011:48), Gluszek (2021:5), Izaryk and Skarakis-Doyle (2017:1228), as well as Da Silva and Montilha (2021:3), there is no restriction on the number of panellists as it depends significantly on the research objectives. This study's research question and objectives guided the composition of this study's panel.

| Panel members | Nursing Care Divi- sions | Number in Hospital A | Number in Hos- pital B |
|-------------------|-----------------------------|-------------------------|---------------------------|
| Clinical Eacilita | Modical word | 1 | |
| | | 1 | |
| tors | Paediatric ward | | 1 |
| Registered | Medical ward | 1 | 1 |
| Nurses | Surgical ward | 1 | 1 |
| | Paediatric ward | 1 | 2 |
| | Cardiac ward | 1 | |
| | Obs-gynae ward | 1 | 1 |
| Subtotal | | 6 | 6 |
| Total | | 12 | <u>.</u> |

Table 6.2: Sample of e-Delphi panellists

6.2.7 Pre-testing of the e-Delphi validation instrument

A pre-test is a process in which a small group of participants is given a data collection instrument to determine the content and validity of the instrument before the actual study is carried out (Management Association, Information Resources 2019:1206; Keeney et al. 2011:144).

Using the recommendation of Wang and Reio Jr. (2017:244) and Keeney et al. (2011:144), a pre-test was administered before round one of the main e-Delphi. This assisted the researcher in cross-checking that the questions asked, the items included, and the instructions were properly understood and correctly interpreted, thus identifying whether modifications of the draft action plan and embedded validation tool were needed, as suggested by Keeney et al. (2011:63).

Ethics approval to conduct the study was received from The Health Studies Research Ethics Committee, Department of Health Studies, Unisa REC-012714-039 (see Annexure 6a), REC-240816-052 (see Annexure 6b). Nursing Services Permission was obtained to conduct research (see Annexure 7a), as well as from the KAIMRC and the Institutional Review Boards (IRB) of study hospitals A and B, approval study number SP 18/036/R (see Annexure 7b); IRB Annual Extension SP 18/036/R (see Annexure 7c); and IRB 6 Months Extension SP 18/036/R (see Annexure 7d as an extension of the ordinal ethics certificate which was needed due to the time lapse between the different phases of the study). Approval was given by the supervisor, who also pretested the action plan and validation instrument online. The pre-test was conducted between **22 June and 03 July 2023**.

Using the same criteria as explained (see Section 6.2.4), the gatekeepers selected a 33,3% sample. Thus, from the non-sampled nursing divisions with similar characteristics, the gatekeepers recruited two registered nurses, one from each of the Hospitals A and B, and two clinical facilitators, one from each of Hospitals A and B. The panellists involved in the pre-test were not included in the main study.

The gatekeepers emailed the recruitment letter, with the link to the pretest available in the letter, to the four eligible experts (panellists) (two clinical facilitators, one for critical care from each hospital, and two nurses, one for critical care from each hospital). Ten days were allowed to complete the pre-test and submit the answers online.

All four recruited panellists volunteered to participate and submit their inputs (100% response rate). All panellists found the validation instrument's content to be clear and familiar. No suggestions were made by any of the four panellists (N = 4; f = 100%).

6.2.8 Ethical considerations

According to Keeney et al. (2011:105), the ethical principles to focus on, include human dignity, non-maleficence, and the role of the researcher. The ethical principles, as described in detail in Section 3.12, were applied. In addition, the fundamental aspects of ethical principles that must be emphasized in Phase 5 (the validation process), are informed consent, confidentiality, and anonymity.

6.2.8.1 Informed consent

Informed consent is defined as a clear, voluntary agreement given by an autonomous individual capable of making sound decisions to participate in a study after the researcher has presented sufficient and understandable information about potential risks and benefits (Reed, Bohlander, Wake & Smith 2016:283; Polit & Beck 2014:382). To avoid a potential conflict of interest, the gatekeepers shared the recruitment letter (see Annexures 9 and 10) with each purposively selected panellist via e-mail. The information letter informed all participants fully before they decided to participate.

6.2.8.2 Confidentiality

Confidentiality was maintained throughout the Delphi process as follows:

- The link to the e-Delphi validation tool was included in the recruitment letter. Panellists could click the link to obtain access; thus, no identity was required.
- (2) As data were transferred to spreadsheets for statistical analysis, the Google Form software provided bulk raw data with no link to any individual participant.
- (3) The researcher could not link specific responses with specific participants as only sets of raw were received via the software program.

6.2.8.3 Anonymity

The anonymity of panel members provides the opportunity for each panel member to respond to the ideas unbiased by the identities of their colleagues (Bradley 2013:241; Keeney et al. 2011:9). The Google Form software allowed the Delphi participants to

remain anonymous as there was no link between the raw data and the individual participants (see Section 6.2.2.2).

6.2.9 Data gathering

The researcher e-mailed a letter to the gatekeepers to explain what is expected from them as gatekeepers (see Annexures 15, 17, and 21). The gatekeepers then sent the recruitment letters (see Annexures 10, 14 and 19) to the selected panellists.

The data for validation of the draft action plan was gathered as discussed in Section 6.2.1. The researcher received all responses from panellists directly from the software program for data analysis as the action plan and validation tool were uploaded on Google Forms by the researcher (see Annexures 13, 16, and 20).

6.2.10 Data analysis

The e-Delphi data analysis involved both qualitative and quantitative data, as recommended by Keeny et al. (2011: 84).

Closed-ended questions: Each panellist's responses for each option in the validation tool and the summary for all panel members were calculated on Google Forms and presented as frequencies and percentages. The frequencies were calculated to identify the acceptable level of consensus. Even though there is no standard threshold for consensus, for this study, the consensus level was decided to be 75%, as suggested by Keeney et al. (2011:46).

Open-ended questions: For the comments and suggestions made in the open spaces provided, a thematic analysis of the narrative data was done using Clarke and Braun's (2013:121) six-phase framework, as described in Section 3.14.

The results and suggestions of all panel members were integrated, resulting in the development of an updated action plan and validation tool. These updates were sub-sequently communicated to participants in each subsequent round using a hyperlink in the recruitment letter.

6.2.11 Trustworthiness

Qualitative researchers' trustworthiness refers to the degree of confidence they have in their data and analyses following the principles of credibility, confirmability, transferability, dependability and authenticity, which are discussed by Zhang (2021:60), Hall & Roussel (2014:38), Polit & Beck (2014:394) as well as Keeney et al. (201:103).

This qualitative phase's rigour was evaluated using the concept of trustworthiness. The study adhered to the principles of credibility, confirmability, transferability, dependability and authenticity to serve as a validation strategy for a Delphi study. Using these principles, the validation tool allowed each panel member to provide feedback on every component of the draft action plan until there was 75% consensus (Keeney et al. 2011:27).

The trustworthiness, as well as the reliability of the validation instrument, ensured the quality of the data, as described below.

6.2.11.1 Credibility

Credibility refers to the consistency of participants' views with the researcher's views and interpretations to give confidence in the truth value of the data and findings (Zhang 2021:60; Polit & Beck 2014:323). To ensure the credibility of the action plan, nurses and clinical facilitators were invited to participate as expert panellists by receiving recruitment letters shared through emails by gatekeepers who have judged them to be knowledgeable about pain management (see Section 6.2.4). The action plan and validation tool were pretested (see 6.2.5), and the panellists were recruited due to their competencies and skills as experts (see 6.2.4). The validation tool facilitated consensus among panellists, with 75% agreement, about the various components of the draft action plan. These components encompass action statements, methodologies, responsible individuals, and timeframes.

6.2.11.2 Confirmability

Confirmability refers to objectivity or neutrality, the potential for congruence between two or more independent people about the data's accuracy, relevance, or meaning (Polit & Beck 2014:323).

To ensure the validity of the data, Delphi participants were kept anonymous to ensure they were free to express themselves and to ensure that the data were reflective of the information they provided (see Section 6.2.5), and three rounds were completed until a **consensus of 75% was reached.** The researcher kept a comprehensive data trail to ensure that others could repeat the process.

6.2.11.3 Transferability

Transferability is the extent to which qualitative findings can be transferred to or have applicability in other settings or groups (Polit & Beck 2014:323). The researcher ensured a complete data trail to allow other researchers to transfer the findings to a similar context if they found it appropriate.

6.2.11.4 Authenticity

Authenticity refers to the extent to which qualitative researchers fairly and faithfully show a range of different realities in the collection, analysis, and interpretation of a report when it conveys the feeling tone of participants (Polit & Beck 2014:323). The achievement of consensus among panel members indicated the authenticity of this study.

6.2.12 Validity

Since validity was defined in Section 3.11, the researcher adhered to the definition and ensured that the Delphi validation instrument collected pertinent and accurate data purporting to measure the attribute of the construct under investigation. This was determined by evaluating the Delphi validation tool's content and face validity for this study phase.

6.2.12.1 Content validity

The embedded validation instrument's content was derived from the literature review and in consultation with the supervisor, statistician, and scientific review panel to guarantee that the study's objectives could be met. The content of the draft action plan was pre-tested, allowing modifications to be made to the final structure of the validation instrument based on feedback from the pre-testing group.

6.2.12.2 Face validity

The Delphi validation instrument underwent pre-testing before distribution to the panellists of the main study, and the items were factually designed, considering the thorough literature review specific to the draft action plan. The supervisor and the Unisa scientific review committee evaluated and provided feedback on whether the embedded Delphi validation instrument evaluated what it was designed to measure. Thus, it was found that the items of the validation instrument had face validity, as they were sufficiently well-considered to validate the target construct and validate the draft action plan.

The reliability of the Delphi validation instrument was ensured before the validation process as it is a prerequisite for validity.

6.2.13 Reliability

Reliability in this context was assessed to ensure *consistency* in employing the e-Delphi technique. This is analogous to dependability to ensure trustworthiness, which refers to the consistency of data over time and conditions (Polit & Beck 221:316).

6.2.13.1 Consistency

The Google Forms[®] software program was used in this study for e-Delphi data analysis. The pre-testing of the validation instrument enhanced the reliability of the developed tool and draft action plan. No amendments were requested by the pre-test panellists (see Section 6.2.5). Three rounds of e-Delphi were completed until 75% consensus was reached, ensuring data consistency is proof of the reliability of this phase of the study. The researcher kept a trail of the collected data as evidence if supervisors or examiners needed it or if the results were questioned.

Through a pre-test, panellists were allowed to provide feedback on the validation tool before conducting the main study for data collection. The supervisor reviewed the researcher's activities comprehensively and checked that every stage of the validation process adhered to the established procedures. In addition, using the tool contributed to increased credibility, while the process effectively determined the dependability of the action plan.

6.3 FINDINGS

The initial round of analysis yielded the subsequent findings, which will now be expounded upon by a comprehensive analysis of the data and subsequent interpretation.

6.4 ROUND 1

Keeney et al. (2011:13) state that achieving a consensus of 75% does not mean obtaining the correct answer. Rather, a panel of participants has agreed that they are satisfied with the product. Round 1 revealed the following:

6.4.1 Biographical data

6.4.1.1 Gender of panellists (N = 12)

All 12 participants (N = 12; f = 100%) were female. In this study, gatekeepers invited nurses regardless of gender. The participating nurses were mostly female because they volunteered to participate, and there were few male nurses in the hospitals under study, as noted in Sections 4.2.1.2 and 4.4.1.1. Terry, Peck, Carden, Perkins, and Smith (2020:701) report that female nurses outnumber male nurses worldwide. Therefore, the recruitment and participation of only female participants are considered reasonable within the scope of this study.

6.4.1.2 Age of panellists (N = 12)

The mean age of the panellists was 37.58 years. As of the time of data collection, the youngest respondent was 26 years old, and the oldest was 47. In this case, the standard deviation was 5.822. Most of the panellists (n = 7; f = 58.4%) were between 36 and 44 years old (see Table 6.3). The present literature analysis provides evidence that emphasizes the significance of an individual's age and length of clinical experience in determining their expertise in pain management (Varndell, Fry & Elliott 2021:10). This study evaluates the role of the gatekeeper in the recruitment of panellists, specifically focusing on their competence in pain management knowledge and experience.

| Age | n | f = % |
|--------------|----|-------|
| 26- 35 years | 4 | 33.2 |
| 36- 44 years | 7 | 58.4 |
| 45 -50 years | 1 | 8.3 |
| Total | 12 | 100 |

| Table 6.3: Panellists' age (N | = 12) |
|-------------------------------|-------|
|-------------------------------|-------|

6.4.1.3 Nationality of Panellists (N = 12)

Panellists were from four different nationalities; 50% (n = 6) were Filipinos, followed by 25% (n = 3) South Africans, 16.7% (n = 2) Malaysians, and 8.3% (n = 1) participants were from Saudi Arabia (see Table 6.4). The finding proves the movement of nurses around the globe in response to the prevailing shortage of nurses in affluent nations, which actively recruit healthcare professionals from less economically privileged countries (Kingma 2018:299).

| Nationality | n | f = % |
|---------------|----|-------|
| Filipino | 6 | 50 |
| South African | 3 | 25 |
| Malaysian | 2 | 16.7 |
| Saudi | 1 | 8.3 |
| Total | 12 | 100 |

Table 6.4: Panellists' nationality (N = 12)

6.4.1.4 Highest education qualification (N = 12)

Figure 6.2 indicates that the majority of the panellists held Bachelor's degrees (n = 7; f = 58.3%), followed by a Diploma in Nursing (n = 3; f = 25%), while a Master's degree was held by two participants (n = 2; f = 16.7%). The finding above serves as evidence and support that many nurses globally have a bachelor's degree (Suliman and Aljezawi (2018:527)



Figure 6.2 Highest level of nursing education (N = 12)

6.4.1.5 Nursing care areas where the panellists worked (N = 12).

Six panellists were from the surgical wards (N = 12; f = 50%), three from paediatric wards (N = 12; f = 25%), two from the surgical wards (N = 12; f = 16.7), and one from the obs-gynae (labour and delivery) ward (N = 12; f = 8.3%) (see Table 6.5).

| Nursing Wards | n = | <i>f</i> = % |
|---------------|-----|--------------|
| Medical | 6 | 50 |
| Paediatric | 3 | 25 |
| Surgical | 2 | 16.7 |
| Obs-gynae | 1 | 8.3 |
| Total | 12 | 100.0 |

Table 6.5: Distribution of panellists within wards (N = 12)

6.4.1.6 Current positions of the panellists (N = 12)

Table 6.6 illustrates that the current positions of the Delphi panellists were eight (f = 66.7%) registered nurses (staff nurses), two (f = 16.7%) clinical facilitators (clinical resource nurses), and two (f = 16.7%) charge nurses. Alnajar et al. (2019:186) high-light that nurses, as healthcare professionals, play a significant role in pain management as they possess the necessary competence to assess and reassess pain, employ both pharmacological and non-pharmacological interventions for pain management, and provide education to patients and their families regarding treatment options. In this context, each panellist made a substantial contribution, as evidenced by their involvement in pain management, as indicated in Section 6.2.4.

Table 6.6: Panellists' positions (N = 12)

| Positions | n = | <i>f</i> = % |
|----------------------|-----|--------------|
| Registered nurse | 8 | 66.7 |
| Clinical Facilitator | 2 | 16.7 |
| Charge nurse | 2 | 16.7 |
| Total | 12 | 100.0 |

6.4.1.7 The duration of the current position held (N = 12)

An analysis of the panellists' tenure in their respective positions reveals that, on average, they have held their current positions for 7.6 years. As of the time of data collection, the shortest duration in the current position held by the panellists was five months, and the longest was 18 years. In this case, the standard deviation was 6.4. It was evident from Table 6.7 that most of the panellists had 4 to 18 years (n = 9; f = 75%) of substantial experience in their positions. According to Petersson and Elgán (2020:1651), the time nurses spend with patients experiencing pain is crucial for providing attention, engagement, and discourse. This interaction is vital for developing competence in pain management for the patient and the nurses' ongoing education and self-empowerment.

| Table 6.7: The duration of the panellists | ' current positions (N = | 12) |
|---|--------------------------|-----|
|---|--------------------------|-----|

| Years in the position held | n | f = % |
|----------------------------|----|-------|
| 0–3 years | 3 | 25 |
| 4–6 years | 4 | 33.3 |
| 7 years or more | 5 | 41.7 |
| Total | 12 | 100 |

All twelve panellists' responses were analysed using Google Forms, and the findings were described under the headings (1) Action statements, (2) Method(s), (3) Responsible person(s) for methods, and (4) Timeframe required to achieve the methods suggested.

ANALYSIS OF THE VALIDATION:

If 75% or more of the panellists agreed to the specific action statement, the responsible persons for the actions, methods, and time frames within which the outcome must be reached, it was seen as a consensus.

In cases where a selection of items was required or suggested within various methods, a 60% agreement for inclusion of the items indicated inclusion. The following sub-methods applied to the 60% agreement.

Method 1.1.1: Items to be included in the policy to motivate the nurses to improve their qualifications.

Method 2.1.1: Pain assessment tools to be accessed on the electronic patient record system.

Method 2.3.1: Internet-based resources to be accessed by the patients and family members.

Method 2.4.1: Internet-based resources on pain management publications, electronic materials, and organisations that specialise in pain management to be accessed by the nursing team in all nursing care.

Method 3.1.1: Specific practice-oriented pain management training to be content included in all nursing care areas.

Method 4.1.1: Different learner types of nurses to be incorporated during learning and training sessions.

Method 4.1.2: How nurses achieve different learning types.

Method 5.1.1: Different teaching approaches to be utilised during pain management training.

Method 6.1.1: Strategies to be employed to motivate nurses to participate in the pain management training program.

Method 7.1.1: Methods to be implemented to motivate nurses to apply their knowledge in practice.

6.4.2 Action statement 1: Motivate nurses to further their studies (N = 12)

As illustrated in Table 6.8, all twelve (f = 100%) panellists agreed; thus, a consensus of 100% was reached that nurses should be motivated to continue their education as part of the action plan.

| Table 6.8: Action statement 1 | : Motivate nurses to f | further their studies | (N = 12) |
|-------------------------------|------------------------|-----------------------|----------|
|-------------------------------|------------------------|-----------------------|----------|

| | | RESPONSES | | | CONSENSUS |
|--------------------|--|-----------|----|-------|-----------|
| | | | | | REACHED |
| | | | | | (≥ 75%) |
| Action statement 1 | | | = | f = % | Yes/No |
| (n = 12; N = 12) | Motivate nurses to further their studies | | n | . – " | |
| | | Agree | 12 | 100 | YES |
| | | Disagree | 0 | 0 | .20 |

| Method 1.1 | | Develop a policy to motivate nurses to improve their nursing qualifica- | Agree | 12 | 100 | VES |
|------------------------|--|--|-----------|-----|-------|------------------------|
| | | tions. | Disagree | 0 | 0 | 120 |
| 1.1.1 | 1.1.1 The following items must be included in the policy to motivate the nurses to improve their qualifications. | | RESPONSES | | | Agreement of (≥60%) |
| | | | = n | | f = % | Yes/No |
| 1. | Paid full-time | study leave 1 year. | 7 | | 58.3 | NO |
| 2. | 20 hours paic work for 1 yea | I study leave and 20 hours full-time ar. | 6 | | 50 | no |
| 3. | Free accommodation for the period of study leave. | | 8 | | 66.7 | yes |
| 4. | A monetary in mal qualificati | centive after completion of a new for- on (degree or diploma). | 9 | | 75 | yes |
| 5. | One day off to program. | attend a one-day pain management | 10 | | 83.3 | yes |
| 6. | A monetary in management | ncentive after completion of a pain program. | 7 | | 58.3 | no |
| 7. | A certificate inurses pursui | issued as an acknowledgement of ng distance learning. | 12 | | 100 | yes |
| 8. | A monetary in learning progr | centive after completion of distance ams. | 9 | | 75 | yes |
| Respo nurses | nsible person to improve the | (s) to develop a policy to motivate ir nursing qualifications. | RESP | ONS | ES | CONSENSUS REACHED |

| (n : | = 12; N = 12) | | | (≥ 75%) |
|--|--|-----------|-------|---------------------------------|
| | | = n | f = % | YES/NO |
| 1. | Five Members of the Central Region Nursing Govern- ance and Accountability Board appointed by the Chief Executive Director for Central Region, Riyadh. | 6 | 50 | |
| 2. | Ad hoc committee appointed by the heads of the Hu- man Resource Department Central Region, Riyadh. | 7 | 58.3 | |
| 3. | Associate Executive Directors of Nursing for King Ab- dulaziz Medical City (KAMC) and King Abdullah Spe- cialist Children Hospital (KASCH). | 7 | 58.3 | |
| 4. | Nursing policy committee representatives appointed by the Associate Executive Directors of Nursing for KAMC and KASCH. | 9 | 75 | NO |
| 5. | Clinical Directors of Nursing Operations for KAMC and KASCH. | 10 | 83.3 | |
| 6. | Director of Postgraduate Center of Nursing Education for KAMC and KASCH. | 10 | 83.3 | |
| Time frame within which to develop a policy to motivate nurses to improve their nursing qualifications must. (n = 12; N = 12 | | RESPONSES | | CONSENSUS REACHED (≥ 75%) |
| | | = n | f = % | YES/NO |
| | 1. 1-3 months | 3 | 25 | VEQ |
| | 2. 4-6 months | 9 | 75 | 123 |

| 3. 7-9 months | | 0 | | 0 | |
|---|--|-----------|-----|-------|---------------------------------|
| Method 1.2 | Present the policy and negotiate for the implementation to motivate | Agree 11 | | 91.7 | |
| (ii = 12, N = 12) | nurses to improve their nursing qualifications to the Ministry of Na- tional Guard Health Affairs (MNGHA) through the Central Re- gion Nursing Governance and Ac- countability Board. | | 1 | 8.3 | YES |
| Responsible person(s) to present and negotiate for the implementation of the policy to motivate nurses to improve their nursing qualifications to the Ministry of National Guard Health Affairs (MNGHA) through the Central Region Nurs- | | RESPONSES | | | CONSENSUS REACHED (≥ 75%) |
| (n = 12; N = 12) | | = n | | f = % | YES/NO |
| Five Meml Governanc by the Chi gion, Riyac | pers of the Central Region Nursing e and Accountability Board appointed ef Executive Director in Central Re- h. | 7 58 | | 58.3 | |
| 2. Ad hoc con Human Re Riyadh. | nmittee appointed by the heads of the source Department Central Region, | 7 | | 58.3 | |
| Associate KAMC and | Executive Directors of Nursing for KASCH. | 7 | | 58.3 | YES |
| 4. Nursing po pointed by Nursing for | blicy committee representatives ap- the Associate Executive Director of KAMC and KASCH. | 9 | | 75 | |
| | | RESP | ONS | ES | CONSENSUS REACHED |

| Time frame within | n which the policy should be | | | | (≥ 75%) | |
|--|---|------------------------------|------------|---|---|--|
| presented, and the implementation negotiated to the MNGHA .(n = 12; N = 12) . | | = n | | f = % | YES/NO | |
| 1. 3 months | | 4 | | 33.3 | | |
| 2. 6 months | | 7 | | 58.3 | NO | |
| 3. 9 months | | 1 | | 8.3 | | |
| Method 1.3 | Include the policy as part of the hospitals` policies after approval | Agree | 12 100 | | | |
| (n = 12; N = 12) | by the Ministry of National Guard Health Affairs (MNGHA) | Disagree | 0 | 0 | TES | |
| | | | | | | |
| Responsible person(hospitals' policies to m ing qualifications. | s) for the inclusion of the policy in all otivate nurses to improve their nurs- | RESP | ONS | ES | CONSENSUS REACHED (≥ 75%) | |
| Responsible person(hospitals' policies to m ing qualifications. (n = 12; N = 12) | s) for the inclusion of the policy in all otivate nurses to improve their nurs- | RESP = n | ONS | ES f = % | CONSENSUS REACHED (≥ 75%) YES/NO | |
| Responsible person(hospitals' policies to m ing qualifications. (n = 12; N = 12) 1. Associate Ex KAMC and KA | s) for the inclusion of the policy in all otivate nurses to improve their nurs- ecutive Directors of Nursing for ASCH. | RESP = n | ONS | ES f = % 50 | CONSENSUS REACHED (≥ 75%) YES/NO | |
| Responsible person(hospitals' policies to m ing qualifications. (n = 12; N = 12) 1. Associate Ex KAMC and KA 2. Clinical Direct facility for KAM | s) for the inclusion of the policy in all otivate nurses to improve their nurs- ecutive Directors of Nursing for SCH. ors of Nursing Operations in every //C and KASCH. | RESP = n 6 10 | ONS | FS f = % 50 83.3 | CONSENSUS REACHED (≥ 75%) YES/NO | |
| Responsible person(hospitals' policies to m ing qualifications. (n = 12; N = 12) 1. Associate Ex KAMC and KA 2. Clinical Direct facility for KAM 3. Nurse Manage care areas. | s) for the inclusion of the policy in all otivate nurses to improve their nurs- ecutive Directors of Nursing for ASCH. ors of Nursing Operations in every AC and KASCH. ers in all KAMC and KASCH nursing | RESP = n 6 10 10 | ons | ES f = % 50 83.3 83.3 | CONSENSUS REACHED (≥ 75%) YES/NO NO | |

| (n = 12; N = 12) | | | (≥ 75%) |
|------------------|-----|-------|---------|
| | = n | f = % | YES/NO |
| a) 1 month | 3 | 25 | |
| b) Six weeks | 1 | 8.3 | |
| c) 2 months | 1 | 8.3 | NO |
| d) 3 months | 7 | 58.3 | |

A consensus was reached about all the methods presented to achieve action statement 1, as indicated in Table 6.8.

6.4.2.1 Method 1.1: Develop a policy to motivate nurses to improve their nursing qualifications (N = 12)

As illustrated in Table 6.8, a 100% consensus (n = 12; N = 12) was reached that a policy must be developed to motivate nurses to improve their nursing qualifications. Some panellists recommended that a policy must be devised to encourage nurses to pursue further education. One panellist commented:

"When there's a policy, the organization will encourage nurses by giving them a day off or education day on the day of education".

Table 6.8 summarises the items selected by 60% or more of the panellists to be included in the policy to motivate nurses to pursue advanced studies. These items were an acknowledgement certificate for nurses pursuing distance learning (100%; n = 12; N = 12), one day off for attending a one-day pain management program (83.3%; n = 10; N = 12), while 75% (n = 9; N = 12) agreed to include a monetary incentive after

the successful completion of a new formal qualification (degree or diploma) and distance learning programs should be included in the policy; and 66.7% (n = 8; N = 12) agreed on free accommodation for the period of study leave.

One of the panellists indicated that:

"A policy should specify that an employee must have worked for the organization for at least two years before he or she can apply for this course".

According to a study conducted by Panlican, Pasay, Gonzales, Alreshidi and Alenzi (2020:100), it is recommended that hospitals implement a policy for the initial assessment of nurses' ability to manage pain as part of their orientation process. The nursing education department should closely monitor this policy.

a) Responsible person(s) to develop a policy to motivate nurses to improve their nursing qualifications (N = 12)

An overall rate of 83.3% (n = 10; N = 12) of panellists indicated that the clinical director of nursing operations for KAMC and KASCH and the director of the postgraduate centre of nursing education for KAMC and KASCH should be responsible for the development of a policy that motivates nurses for further studies. 75% (n = 9; N = 12) indicated that the nursing policy committee representatives appointed by the Associate Executive Directors of Nursing for KAMC and KASCH would be the appropriate persons responsible (see Table 6.8; method 1.1).

b) Time frame required to develop a policy to motivate nurses to improve their nursing qualifications (N = 12)

A majority of 75% (n = 9; N = 12) of panellists reached an agreement that 4–6 months was the best timeframe for policy development to motivate nurses to improve their nursing qualifications (see Table 6.8; method 1.1).

6.4.2.2 Method 1.2: Present the policy and negotiate for the implementation to motivate nurses to improve their nursing qualifications to the Ministry of

National Guard Health Affairs (MNGHA) through the Central Region Nursing Governance and Accountability Board (N = 12)

As indicated in Table 6.8, panellists reached a consensus of 91.7% (n = 11; N = 12) that the policy must be presented and negotiated for implementation to motivate nurses to improve their nursing qualifications to the Ministry of National Guard Health Affairs (MNGHA) through the Central Region Nursing Governance and Accountability Board.

a) Responsible person(s) to present the policy and negotiate for the implementation to motivate nurses to improve their nursing qualifications to the Ministry of National Guard Health Affairs (MNGHA) through the Central Region Nursing Governance and Accountability Board (N = 12)

There was a 75% consensus (n = 9; N = 12) that the nursing policy committee representative appointed by the Associate Executive Director of Nursing for KAMC and KASCH is best qualified to present and negotiate the policy's implementation (see Table 6.8; method 1.2).

 b) Time frame required to present the policy and negotiate for the implementation to the MNGHA (N = 12)

A consensus was not reached on the time frame within which the policy should be presented, and the implementation should be negotiated with the MNGHA (see Table 6.8; method 1.2).

6.4.2.3 Method 1.3: Include the policy as part of the hospitals` policies after approval of the action plan by the Ministry of National Guard Health Affairs (MNGHA) (N = 12)

As illustrated in Table 6.8, a 100% consensus (n = 12; N = 12) was reached that the policy to motivate nurses to improve their nursing qualifications must be included in all hospital policies following MNGHA approval.

a) Responsible person(s) to include the policy in all hospitals' policies to motivate nurses to improve their nursing qualifications (N = 12)

As illustrated in Table 6.8 (method 1.3), 83.3% (n = 10; N = 12) of panellists indicated that both the clinical directors of nursing operations in every facility for KAMC and KASCH, as well as nurse managers in all KAMC and KASCH nursing care areas will be the appropriate persons to include the policy in all hospitals' policies to motivate nurses to improve their nursing qualifications after being approved by MNGHA (see Table 6.8.

b) Time frame required to include the policy in all hospitals' policies after approval of the action plan by MNGHA (N = 12)

As indicated in Table 6.8, consensus was not reached as to when the policy should be included in all hospitals' policies following its approval by the MNGHA (method 1.3).

6.4.3 Action statement 2: Make appropriate and relevant pain management tools accessible to the nursing team in every clinical area (N = 12)

Panellists reached a 91.7% consensus (n = 11; N = 12) that the action statement to make appropriate and relevant pain management tools accessible to the nursing team in every clinical area should be included in the action plan (see Table 6.9).

Table 6.9: Action statement 2: Make appropriate and relevant pain management tools accessible to the nursing team in every clinical area (N = 12)

| | | RESPONSES | CONSEN- |
|-------------------------|---|-----------|--------------------|
| | | | SUS |
| Action state- ment 2 | Make appropriate and rele- vant pain management tools accessible to the nurs- | | REACHED (≥ 75%) |
| (n = 12; N = | ing team in every clinical ar- | | |
| 12) | eas. | | |

| | | | = n | f = | Yes/No |
|-----------------|--------------------------------|-------|-----|------|-----------|
| | | | | % | |
| | | Agree | 11 | 91.7 | |
| | | Disa- | 1 | 0.2 | YES |
| | | gree | 1 | 0.3 | |
| Method 2.1 | Include pain assessment | Agree | 0 | | |
| | tools in an electronic patient | | | | |
| | record system so that the | | | | |
| | nursing team can choose | Disa- | 0 | | NO |
| | and access the most appro- | gree | 0 | | |
| | priate. tool in all nursing | | | | |
| | care areas. | | | | |
| 2.1.1 The follo | owing pain assessment tools | | | | Agreement |
| should be acc | cessible on the electronic pa- | | | f _ | of ≥60% |
| tient record sy | /stem | = n | | · – | Yes/No |
| | | | | 70 | 163/110 |
| | | | | | |
| QUEST | | 0 | | 0 | |
| COLDSPA | | 0 | | 0 | |
| OPQSTUV | | 0 | | 0 | |
| PQRST | | 8 | | 66.7 | Yes |
| BPI | | 0 | | 0 | |
| VDS | | 0 | | 0 | |
| VAS | | 0 | | 0 | |
| NPASS | | 1 | | 8.3 | No |
| СРОТ | | 1 | | 8.3 | No |
| BPS | | 0 | | 0 | |
| COMFORT B | | 2 | | 16.7 | No |
| CRIES | | 8 | | 66.7 | Yes |
| ABBEY | | 1 | | 8.3 | No |
| PAINAD | | 3 | | 25 | No |

| NOPPAIN | 0 | 0 | |
|---|-----------|----------|------------|
| CNPI | 0 | 0 | |
| Responsible person(s) to include the pain | RESPONSES | | CONSEN- |
| assessment tools in the electronic patient | | | SUS |
| record system | | | |
| (r. 40. N. 40) | | | REACHED |
| (n = 12; N = 12) | | | (≥ 75%) |
| | | | 、 , |
| | | | |
| | = n | f = % | YES/NO |
| Two computer engineers appointed by the | | | |
| HR manager, each assigned for KAMC | 6 | 50 | |
| and KASCH. | | | |
| Five nursing health informatics specialists | | | |
| appointed by Associate Directors of Nurs- | 9 | 75 | |
| ing for KAMC and KASCH. | | | NO |
| Clinical Director of Nursing Operations ap- | | | NO |
| pointed by Associate Executive Directors | 6 | 50 | |
| in every facility for KAMC and KASCH. | | | |
| One pain nurse specialist appointed by | | | |
| Clinical Directors of Nursing Operations in | 11 | 91.7 | |
| every facility for KAMC and KASCH. | | | |
| Time frame within which to include the pain | | | CONSEN- |
| assessment tools in the electronic patient | DEODONOEO | | SUS |
| record system. | RESPONSES | | REACHED |
| (n = 12; N = 12) | | | (≥ 75%) |
| | = n | f = | YES/NO |
| | | % | |
| 1- 3 months | 4 | 33.3 | |
| 4- 6 months | 7 | 58.3 | |

| 7-9 months | | 1 | | 8.3 | NO |
|---|--|---------------|------|---------|---------|
| Method 2.2 | Involve nurse supervisors | Agree | 10 | 83.3 | |
| (n = 12; N = 12) | with pain management training to provide post-pain management training su- pervisory support to the nursing team on how to conduct pain assessments in all nursing care areas. | Disa- gree | 2 | 16.7 | YES |
| | | | | | CONSEN- |
| Responsible p | person(s) to involve nurse su- | | | | |
| pervisors with | n pain management training | RESPON | ISES | | REAGHED |
| to provide po | st-pain management training | | | | (≥ 75%) |
| and supervisory support to the nursing | | | | | |
| team on how | to conduct pain assessments | | |)/50/10 | |
| in all nursing | care areas (n = 12; N = 12). | = n f = % | | YES/NO | |
| Clinical Direct | ors of Nursing Operations for | 9 7 | | 75 | |
| KAMC and K | ASCH. | | | 10 | |
| Nurse Manag | ers in all KAMC and KASCH | 8 66 | | 66.7 | |
| nursing care a | areas. | | | | NO |
| Clinical facil KASCH nursi | itators in all KAMC and ng care areas. | 9 75 | | 75 | |
| Charge nurse | es in all KAMC and KASCH | | | 66.7 | |
| nursing care areas. | | ð | | 66.7 | |
| Time frame | within which to involve the | | | | CONSEN- |
| nurse supervisors to provide pain manage- | | | | | SUS |
| ment training | supervisory support to the | RESPON | ISES | | REACHED |
| | | | | (≥ 75%) | |

| nursing team on how to conduct pain as- | | = n | f = | | YES/NO |
|--|--|---------------|-------|------|---------------------------|
| sessment in all nursing care areas | | | | % | |
| (n = 12; N = 12) | | | | | |
| Every shift wh | nen the need arises | 5 | | 41.7 | NO |
| Every patient | round when the need arises | 8 | | 66.7 | |
| Method 2.3 | Make hospitals' internet- | Agree | 11 | 91.7 | |
| (n = 12; N = 12) | to the patients and family members to obtain other support about pain man- agement. | Disa- gree | 1 | 8.3 | YES |
| The following Internet-based resources should be accessible. | | RESPONSES | | | Agreement of (≥60%) |
| | | = n | f = % | | YES/NO |
| Patient pain n | nanagement websites | 10 | 83.3 | | yes |
| Patient pain n | nanagement hotlines | 9 | 75 | | yes |
| Pain manage | ment support groups | 10 | 83.3 | | yes |
| Peer support | groups | 8 | 66.7 | | yes |
| Responsible person(s) to ensure internet- based resources should be accessible to patients and family members to provide | | RESPONSES | | | CONSEN- SUS REACHED |
| | , pain managementi | | | | (≥ 75%) |
| (n = 12; N = 1 | 2) | = n | f = % | | YES/NO |
| Chief Executive Director for KAMC and KASCH. | | 5 | 41.7 | | |
| Two health in | nformation technologists ap- | | | | |
| pointed by the | he HR manager, each as- | 7 | 58.3 | | |
| signed for KA | MC and KASCH. | | | | |

| Two communication and information ad- | | | |
|--|--------|-------|---------|
| ministrators appointed by the HR manager | 5 | 41.7 | |
| for KAMC and KASCH. | | | NO |
| Five nursing health informatics specialists | | | |
| appointed by Associate Directors of Nurs- | 6 | 50 | |
| ing for KAMC and KASCH. | | | |
| Associate Executive Directors of Nursing | 2 | 25 | |
| for KAMC and KASCH. | 5 | 20 | |
| Clinical Directors of Nursing Operations for | e | 50 | |
| KAMC and KASCH. | 0 | 50 | |
| Nurse supervisors appointed by Executive | | | |
| Associate Directors of Nursing in every fa- | 6 | 50 | |
| cility for KAMC and KASCH. | | | |
| Nurse Managers in all KAMC and KASCH | 8 | 66.7 | |
| nursing care areas. | 0 | 00.7 | |
| Charge nurses in all KAMC and KASCH | 8 | 66.7 | |
| nursing care areas. | 0 | 00.7 | |
| Registered nurses in all KAMC and | Q | 66.7 | |
| KASCH nursing care areas. | 0 | 00.7 | |
| Time frame within which to make the hos- | | | CONSEN- |
| pitals' internet-based resources accessible | | | SUS |
| to the patient and family members to ob- | RESPON | ISES | REACHED |
| tain other support about pain management | | | (|
| • | | 6 04 | (≥ /5%) |
| | = n | f = % | YES/NO |
| (n = 12; N = 12) | | | |
| Every day at a convenient time | 3 | 25 | YES |
| 24-hour access 7 days a week | 9 | 7 | |
| | | | |
| | | | |

| | | | | | CONSEN- |
|---|---|--------|--------|----------|-----------|
| | | RESPON | ISES | | SUS |
| | | | | | REACHED |
| | | = n | | f = % | YES/NO |
| Method 2.4 | Make hospitals' internet- | Agree | 12 | 100 | |
| | based resources on pain | | | | |
| (n = 12; N = | management publications | Dist | | | VES |
| 12) | and electronic materials ac- | Disa- | 0 | 0 | TLO |
| | cessible to the nursing team | gree | | | |
| | in all nursing care areas. | | | | |
| The following | Internet-based resources on | DESDON | | <u> </u> | Agreement |
| pain manage | ment publications, electronic | RESPON | 1959 | | (≥60%) |
| materials, an | d organisations specialising | _ n | f _ 0/ | | |
| in pain manag | ement should be accessible. | = 11 | 1 = 70 | | TES/INU |
| E-newsletters | i de la companya de l | 6 | 50 | | no |
| Videos on pai | n management | 9 | 75 | | yes |
| Facts sheets | | 3 | 25 | | no |
| Pain toolkits | | 10 | 83.3 | | yes |
| Clinical updat | es | 9 | 75 | | yes |
| The World He | alth Organization | 4 | 33.3 | | no |
| American Pai | n Society | 5 | 41.7 | | no |
| International / | Association for Study of Pain | 7 | 58.3 | | no |
| Responsible | person(s) to ensure internet- | | | | CONSEN- |
| based resou | rces on pain management | | | | SUS |
| publications, electronic materials, and or- | | RESPON | ISES | | REACHED |
| ganisations that specialise in pain man- | | | | | (|
| agement should be accessible to the nurs- | | | | | (≥ 75%) |
| ing team in al | I nursing care areas. | = n | f = % | | YES/NO |
| (n = 12; N = 1 | 2) | | | | |

| Two computer engineers appointed by the | | | |
|---|-----------|-------|---------|
| HR manager, each assigned for KAMC | 3 | 25 | |
| and KASCH | | | |
| Two librarians appointed by the HR man- | | | |
| ager each assigned for KAMC and | 6 | 50 | |
| KASCH. | | | |
| Nursing health informatics appointed by | | | |
| Associate Directors of Nursing for KAMC | 11 | 91.7 | |
| and KASCH. | | | |
| Nurse Managers in all KAMC and KASCH | 7 | 58.3 | NO |
| nursing care areas. | 1 | 30.5 | |
| Nurse supervisors appointed by Executive | | | |
| Associate Directors of Nursing for KAMC | 4 | 33.3 | |
| and KASCH. | | | |
| One pain nurse specialist appointed by | | | |
| Clinical Directors of Nursing Operations in | 9 | 75 | |
| every facility for KAMC and KASCH. | | | |
| One nurse educator appointed by the Di- | | | |
| rector of the Postgraduate Nursing Educa- | 7 | 58.3 | |
| tion Center for KAMC and KASCH. | | | |
| Time frame within which to make the hos- | | | CONSEN- |
| pitals' internet-based resources accessible | RESPONSES | | SUS |
| to the nursing team in all nursing care ar- | | | REACHED |
| eas on pain management publications and | | | |
| electronic materials. | | | (≥ 75%) |
| | = n | f = % | YES/NO |
| (n = 12; N = 12) | | | |
| Every nursing shift | 3 | 25 | |
| Continuously available | 9 | 75 | YES |

6.4.3.1 Method 2.1: Include pain assessment tools in an electronic patient record system so that the nursing team can choose and access the most appropriate tool in all nursing care areas (N = 12)

Panellists did not agree or disagree as to whether pain management tools must be accessible on the electronic patient record system so that the nursing team can choose the most appropriate tool in all nursing care areas. This important question was included in the draft action plan in round 2 questions to indicate agreement or disagreement.

 a) Responsible person(s) to ensure that an electronic format of pain assessment tools is available for inclusion in the electronic patient record system so that they can be accessible to the nursing team in every nursing care area (N = 12)

Consensus was not reached regarding who the responsible person(s) should be to provide the electronic format of the pain assessment tools for inclusion in the electronic patient record system (see Table 6.9; method 2.1).

91.7% (n = 11; N = 12) of panellists indicated that clinical directors of nursing operations must appoint one pain nurse specialist in every facility for KAMC and KASCH, and 75% (n = 9; N = 12) of panellists indicated that five nursing health informatics specialists must be appointed by associate directors of nursing for KAMC and KASCH as responsible persons to provide the electronic format of the pain assessment tools (see Table 6.9; method 2.1).

b) Time frame required to include the pain assessment tools in the electronic patient record system (N = 12)

Panellists did not reach a consensus on the time frame within which pain assessment tools must be accessible in the electronic patient record system (see Table 6.9; method 2.1).

6.4.3.2 Method 2.2 Involve nurse supervisors with pain management training to provide post-pain management training supervisory support to the

nursing team on how to conduct pain assessments in all nursing care areas (N = 12)

83.3% (n = 10; N = 12) of panellists reached a consensus that the nurse supervisors should be involved with pain management training and supervisory support to the nursing team on how to conduct pain assessments (see Table 6.9).

 a) Responsible person(s) to involve nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessments (N = 12)

Consensus was not reached regarding the person(s) responsible for ensuring that nurse supervisors with pain management training are involved in providing post-pain management training and supervisory support to the nursing team (see Table 6.9; method 2.2). 75% (n = 9; N = 12) of panellists selected both clinical directors of nursing operations for KAMC and KASCH, as well as clinical facilitators in all KAMC and KASCH nursing care areas, as the best responsible persons to involve nurse supervisors with pain management training and supervisory support

b) Time frame required to involve the nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessment in all nursing care areas (N = 12)

Consensus was not attained on the time frame to involve the nurse supervisors in providing pain management training and supervisory support to the nursing team on how to conduct pain assessment in all nursing care areas (see Table 6.9; method 2.2).

6.4.3.3 Method 2.3: Make hospitals' internet-based resources accessible to the patients and family members to obtain other support about pain management (N = 12)

A 91.7% (n = 11; N = 12) consensus was reached that hospitals' internet-based resources should be made accessible to the patients and family members to obtain other support about pain management (see Table 6.9). 83.3% (n = 10; N = 12) of panellists indicated that patient pain management websites and pain management support groups should be hospitals' internet-based resources accessible to the patients and family members to obtain other support about pain management (see Table 6.9; method 2.3). There was 75% consensus (n = 9; N = 12) that the patient pain management hotlines should be made available to both patients and their family members in conjunction with hospitals' existing internet-based resources (see Table 6.9; method 2.3).

a) Responsible person(s) to ensure that internet-based resources are accessible to patients and their families to provide support about pain management (N = 12)

Consensus was not reached regarding who the responsible person(s) should be to ensure that internet-based resources are accessible to patients and families to provide support about pain management (see Table 6.9; method 2.3).

b) Time frame required to make hospitals' internet-based resources accessible to the patients and family members to obtain other types of support about pain management (N = 12)

75% (n = 9; N = 12) consensus was reached that the hospitals' internet-based resources should be accessible to the patients and family members to ensure that they can obtain other support about pain management, 24-hour hours per day and seven days a week (see Table 6.9; method 2.3).

6.4.3.4 Method 2.4: Make hospitals' internet-based resources on pain management publications and electronic materials accessible to the nursing team in all nursing care areas (N = 12)

Panellists all agreed (f = 100%; n = 12; N = 12) that hospitals' internet-based pain management publications and electronic resources on pain management publications must be accessible to the nursing team in all nursing care areas (see Table 6.9). The hospitals' internet-based resources on pain management publications and electronic materials mentioned were: pain toolkits (f = 75; n = 9; N = 12), videos on pain management (f = 83.3%; n = 10; N = 12), and clinical updates (f = 75%; n = 9; N = 12) (see Table 6.9).

c) Responsible person(s) to ensure that internet-based resources on pain management publications, electronic materials, and organisations that specialise in pain management are made accessible to the nursing team in all nursing care areas (N = 12)

Consensus was not reached (see Table 6.9; method 2.4). 91.7% (n = 11; N = 12) of panellists indicated that associate directors of nursing should appoint nursing health informatics for KAMC and KASCH, and 75% (n = 9; N = 12) indicated that one pain nurse specialist appointed by clinical directors of nursing operations in each facility for KAMC and KASCH should be the be responsible for making internet-based resources on pain management publications and electronic materials accessible to the nursing team in all nursing care areas.

d) Time frame required to make pain management publications and electronic materials available via the hospitals' internet to the nursing team in all nursing care areas (N = 12)

A consensus of 75% (n = 9; N = 12) was reached that access via the hospital internet should be continuously available to make pain management publications and electronic materials accessible to the nursing team (see Table 6.9; method 2.4).

6.4.4 Action statement 3: Develop a practice-oriented content-specific short pain management training program (N = 12).

A hundred per cent consensus (f = 100%; n = 12; N = 12) was reached that developing a practice-oriented content-specific short pain management program must be included in the action plan (see Table 6.10).

Table 6.10: Action statement 3: Develop a practice-oriented content-specific short pain management training program.

| | RESPONSES | CONSEN- |
|---------------|-----------|---------|
| | | SUS |
| Action state- | | |
| ment 3 | | REACHED |
| | | |
| (n = 12; N = | Develop a practice-oriented, | | | | (≥ 75%) |
|-------------------|--------------------------------------|--------|-----|-------|-----------|
| 12) | content-specific short pain | | | | |
| | management training program. | | = | f = | Yes/No |
| | | | n | % | |
| | | Agree | 12 | 100 | |
| | | Disa- | 0 | 0 | YES |
| | | gree | | | |
| Method 3.1 | Include practice-oriented pain | Agree | 12 | 100 | |
| | management training content | | | | VES |
| (n = 12; N = | for all nursing care areas in the | droo | 0 | 0 | 120 |
| 12) | pain management program. | gree | | | |
| The following | specific practice-oriented pain | | | | Agreement |
| management | training content should be in- | RESPON | ISE | S | by (≥60%) |
| cluded in all nu | ursing care areas. | | | - | |
| | | | | | |
| | | = n | | f = % | Yes/No |
| Methods to pro | omptly assess a patient's pain in | | | | |
| all nursing area | as. | 11 | | 91.7 | yes |
| Assessment of | f patients' pain in all nursing care | 9 | | 75 | yes |
| areas. | | | | | |
| The advantage | es and disadvantages of all pain | 9 | | 75 | yes |
| management s | scales. | | | | |
| Labour pain is | a type of pain to be assessed. | 6 | | 50 | no |
| The selection | of appropriate pain intervention | 10 | | 83.3 | yes |
| strategies base | ed on the pain levels assessed. | | | | |
| Responsible p | erson(s) to include specific prac- | RESPON | ISE | S | CONSEN- |
| tice-oriented p | pain management training con- | | | | SUS |
| tent for all nurs | sing care areas (n = 12; N = 12). | | | | |
| | | | | | REACHED |
| | | | | | (≥ 75%) |
| | | = n | , | f = % | Yes/No |
| | | | | ,,, | |

| Director of Postgraduate Center of Nursing Ed- | Λ | 22.2 | |
|---|--------------------|----------------------------|-------------------------------------|
| ucation for KAMC and KASCH. | 4 | 33.3 | |
| One Nurse educator appointed by the Director | | | |
| of the Postgraduate Center of Nursing Educa- | 8 | 66.7 | |
| tion for KAMC and KASCH. | | | |
| One Clinical facilitator appointed by Clinical Di- | | | |
| rectors of Nursing Operations in every nursing | 7 | 58.3 | |
| care area. | | | YES |
| One Pain nurse specialist appointed by Clinical | | | |
| Directors of Nursing Operations in every facil- | 11 | 91.7 | |
| ity. | | | |
| Time frame within which the specific practice- | RESPONSE | S | CONSEN- |
| Time name within which the specific practice- | | -0 | 00110211 |
| oriented pain management training content | | -0 | SUS |
| oriented pain management training content should be provided for inclusion in the pro- | | | SUS |
| oriented pain management training content should be provided for inclusion in the pro- gram. | | | SUS |
| oriented pain management training content should be provided for inclusion in the pro- gram. | | | SUS REACHED (≥ 75%) |
| oriented pain management training content should be provided for inclusion in the pro- gram. (n = 12; N = 12) | = n | f = % | SUS REACHED (≥ 75%) Yes/No |
| oriented pain management training content should be provided for inclusion in the pro- gram. (n = 12; N = 12) | = n | f = % | SUS REACHED (≥ 75%) Yes/No |
| oriented pain management training content should be provided for inclusion in the program. (n = 12; N = 12) 1 week before the due date of the training program. | = n 0 | f = % 0 | SUS REACHED (≥ 75%) Yes/No |
| oriented pain management training content should be provided for inclusion in the program. (n = 12; N = 12) 1 week before the due date of the training program. 1 month before the due date of the training program. | = n 0 | f = % 0 | SUS REACHED (≥ 75%) Yes/No |
| oriented pain management training content should be provided for inclusion in the program. (n = 12; N = 12) 1 week before the due date of the training program. 1 month before the due date of the training program. | = n 0 8 | f = % 0 66.7 | SUS REACHED (≥ 75%) Yes/No |
| oriented pain management training content should be provided for inclusion in the program. (n = 12; N = 12) 1 week before the due date of the training program. 1 month before the due date of the training program. | = n 0 8 | f = % 0 66.7 | SUS REACHED (≥ 75%) Yes/No |
| oriented pain management training content should be provided for inclusion in the program. (n = 12; N = 12) 1 week before the due date of the training program. 1 month before the due date of the training program. 3 months before the due date of the training | = n 0 8 4 | f = % 0 66.7 33.3 | SUS REACHED (≥ 75%) Yes/No |

6.4.4.1 Method 3.1: Include practice-oriented pain management training content for all nursing care areas in the pain management program (N = 12)

A hundred per cent consensus (n = 12; N = 12) was reached that practice-oriented pain management training content for nursing care areas should be included in the pain management program (see Table 6.10). 83,3% (n = 10; N = 12) of panellists

indicated that the selection of appropriate pain intervention strategies based on the pain levels assessed, and 75% (n = 9; N = 12) indicated for both the assessment of patients' pain in all nursing care areas and the advantages and disadvantages of all pain management scales to be specific practice-oriented pain management content to be included in all nursing care areas (see Table 6.10).

a) Responsible person(s) to include specific practice-oriented pain management training content for all nursing care areas (N = 12)

As illustrated in Table 6.10, 91.7% (n = 11; N = 12) of panellists agreed that one pain nurse specialist appointed by clinical directors of nursing operations in every facility will be the best person to assume responsibility (method 3.1)

 b) Time frame required to provide for the inclusion of the specific practice-oriented pain management training content in the program (N = 12)

Consensus was not reached regarding the time frame within which the specific practice-oriented pain management training content should be provided for inclusion in the program (see Table 6.10; method 3.1).

6.4.5 Action statement 4: Develop a pain management short program that accommodates all learning types (N = 12)

The panellists were in agreement (f = 100%; n = 12; N = 12) that a short pain management program that accommodates all learning types must be included in the action plan and that it must be developed (see Table 6.11).

Table 6.11: Action statement 4: Develop a pain management short programthat accommodates all learning types (N = 12)

| | RESPONSES | CONSEN- |
|--|-----------|---------|
| | | SUS |
| | | REACHED |
| | | (≥ 75%) |

| | | | = | f = % | Yes/No |
|--|------------------------------------|-----------|------|-------|-----------|
| | | | n | | |
| Action state- | Develop a pain management | Agree | 12 | 100 | |
| ment 4 | short program that accommo- | | | | |
| | dates all learning types. | Disa- | 0 | 0 | YES |
| (n = 12; N = | | gree | | | |
| 12) | | | | | |
| Method 4.1 | Incorporate different learner | Agree | 12 | 100 | |
| (n. 10; N | types during learning/training | Disa- | | | YES |
| $(\Pi = 12, \Pi = 12)$ | sessions. | gree | 0 | 0 | |
| 12) | | | | | A |
| | | RESPO | NSE | s | Agreement |
| The following o | lifferent learner types of nurses | | | | by (≥60%) |
| should be inc | luded during learning/training | = n | 1 | f = % | Yes/No |
| sessions. | | | | | |
| Creative Learn | ers. | 11 | | 91.7 | yes |
| Enthusiastic th | inking learners. | 8 | (| 66.7 | yes |
| Self-confident t | hinkers. | 3 | 2 | 25 | no |
| Organised thin | king learners. | 9 | | 75 | yes |
| Truth-seeking I | earners. | 5 | 4 | 41.7 | no |
| Inquisitive think | kers. | 4 | | 33.3 | no |
| Diligent inquisit | tive learners. | 6 | | 50 | no |
| Self-directed le | arners. | 5 41.7 | | 41.7 | no |
| The following | different learning types that | DESDONSES | | 9 | Agreement |
| should be incorporated during pain manage- | | | NOL. | U III | by (≥60%) |
| ment learning/training sessions are achieved | | = n | 1 | f = % | Yes/No |
| by using the following learning styles: | | | | | |
| Creative Learners (generate creative ideas in | | 12 | | 100 | yes |
| a group). | | 12 | | .00 | |
| Enthusiastic th | inking learners (listen to the in- | 8 | | 66 7 | VAS |
| formation actively, take part in the activity to | | 0 | | 00.7 | yes |

| practice the skill and participate in the group | | | |
|---|------------------------------|-------------------------------------|-------------------------------------|
| discussion). | | | |
| Self-confident thinkers (take control of their | 3 | 25 | 20 |
| learning). | 5 | 20 | 110 |
| Organised thinking learners (solve different | 8 | 66.7 | VAS |
| real-life problems). | 0 | 00.7 | yes |
| Truth-seeking learners (question the infor- | 4 | 33 3 | no |
| mation obtained). | - | 00.0 | |
| Inquisitive thinkers (inquire about the infor- | 6 | 50 | no |
| mation obtained). | Ŭ | 00 | |
| Diligent inquisitive learners (take the initiative | 7 | 58 3 | no |
| in their learning). | ' | 00.0 | |
| Self-directed learners (take control of their | 6 | 50 | no |
| learning). | Ŭ | 00 | |
| Responsible person(s): the best person(s) to | RESPONSES | | CONSEN- |
| be responsible to ensure that the learning ob- | | | SUS |
| jectives accommodate the different learner | | | |
| | | | |
| types of nurses during pain management | | | REACHED |
| types of nurses during pain management learning/training sessions. ($n = 12$; $N = 12$) | | | REACHED (≥ 75%) |
| types of nurses during pain management learning/training sessions. ($n = 12$; $N = 12$) | = n | f = % | REACHED (≥ 75%) Yes/No |
| types of nurses during pain management learning/training sessions. (n = 12; N = 12) Director of Postgraduate Center of Nursing | = n | f = % | REACHED (≥ 75%) Yes/No |
| types of nurses during pain management learning/training sessions. (n = 12; N = 12) Director of Postgraduate Center of Nursing Education for KAMC and KASCH. | = n 8 | f = % 66.7 | REACHED (≥ 75%) Yes/No |
| types of nurses during pain management learning/training sessions. (n = 12; N = 12) Director of Postgraduate Center of Nursing Education for KAMC and KASCH. One Nurse educator appointed by the Director | = n 8 | f = % 66.7 | REACHED (≥ 75%) Yes/No |
| types of nurses during pain management learning/training sessions. (n = 12; N = 12) Director of Postgraduate Center of Nursing Education for KAMC and KASCH. One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Educa- | = n 8 7 | f = % 66.7 58.3 | REACHED (≥ 75%) Yes/No |
| types of nurses during pain management learning/training sessions. (n = 12; N = 12) Director of Postgraduate Center of Nursing Education for KAMC and KASCH. One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Educa- tion for KAMC and KASCH. | = n 8 7 | f = % 66.7 58.3 | REACHED (≥ 75%) Yes/No |
| types of nurses during pain management learning/training sessions. (n = 12; N = 12) Director of Postgraduate Center of Nursing Education for KAMC and KASCH. One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Educa- tion for KAMC and KASCH. Clinical facilitators in all areas of nursing care | = n 8 7 | f = % 66.7 58.3 | REACHED (≥ 75%) Yes/No |
| betwee decommodate the universit loaner types of nurses during pain management learning/training sessions. (n = 12; N = 12) Director of Postgraduate Center of Nursing Education for KAMC and KASCH. One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Educa- tion for KAMC and KASCH. Clinical facilitators in all areas of nursing care for KAMC and KASCH. | = n 8 7 7 | f = % 66.7 58.3 58.3 | REACHED (≥ 75%) Yes/No YES |
| types of nurses during pain management learning/training sessions. (n = 12; N = 12) Director of Postgraduate Center of Nursing Education for KAMC and KASCH. One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Educa- tion for KAMC and KASCH. Clinical facilitators in all areas of nursing care for KAMC and KASCH. One Pain nurse specialist appointed by Clini- | = n 8 7 7 7 | f = % 66.7 58.3 58.3 | REACHED (≥ 75%) Yes/No YES |
| types of nurses during pain management learning/training sessions. (n = 12; N = 12) Director of Postgraduate Center of Nursing Education for KAMC and KASCH. One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Educa- tion for KAMC and KASCH. Clinical facilitators in all areas of nursing care for KAMC and KASCH. One Pain nurse specialist appointed by Clini- cal Directors of Nursing Operations in every | = n 8 7 7 7 9 | f = % 66.7 58.3 58.3 75 | REACHED (≥ 75%) Yes/No YES |

| Time frame within which time the learning | RESPONS | ES | CONSEN- |
|---|---------|-------|---------|
| types must be shared for inclusion within the | | | SUS |
| training program. (n = 12; N = 12). | | | REACHED |
| | | | (≥ 75%) |
| | = n | f = % | Yes/No |
| 2 weeks before the due date for finalisation of | 2 | 16.7 | |
| the training program. | 2 | 10.1 | |
| 1 month before the due date of the training | 5 | 41.7 | NO |
| program. | | | |
| 3 months before the due date of the training | 5 | 41.7 | |
| program. | | | |

6.4.5.1 Method 4.1: Incorporate different learner types during learning/training sessions (N = 12)

Panellists unanimously (100%; n = 12; N = 12) agreed that different learner types must be incorporated during learning and/or training sessions (see Table 6.11). In light of the result of the options for different learner types of nurses that should be incorporated during learning or training sessions, as illustrated in Table 6.11, 91.7% (n = 11; N = 12) of panellists selected creative learners, and 75% (n = 9; N = 12) selected organised thinking learners type of nurses. 100% consensus (n = 12; n = 12) was accomplished for those creative learners as one of the different learner types achieved by generating creative ideas in a group (see Table 6.11; method 4.1).

 Responsible person(s) to ensure that the learning objectives accommodate the different learner types of nurses during pain management learning/training sessions (N = 12) 75% (n = 9; N = 12) of panellists indicated that one pain nurse specialist appointed by the nursing operations director in every facility would best incorporate different learner types during learning/training sessions (see Table 6.11; method 4.1).

 Time frame required to ensure that the learning types are shared and included within the training program (N = 12)

The panellists indicated diverse opinions on the time frame by which the learning types must be included within the training program (see Table 6.11; method 4.1).

6.4.6 Action statement 5: Incorporate different teaching approaches to accommodate diverse learners and facilitators in the training of pain management (N = 12)

All panellists (f = 100%; n = 12; N = 12) agreed that different teaching approaches must be incorporated into the learning of pain management to accommodate diverse learners and facilitators.

Table 6.12: Action statement 5: Incorporate different teaching approaches to accommodate diverse learners and facilitators in the training of pain management (N = 12)

| Action state- ment 5 (n = 12; N = 12) | Develop a pain man- agement short course that moti- vates nurses to ap- ply knowledge | RESPON | SES | | CONSEN- SUS REACHED (≥ 75%) |
|---|---|---------------|-----|-------|--------------------------------------|
| | program to practico | | n = | f = % | Yes/No |
| | | Agree | 12 | 100 | |
| | | Disa- gree | 0 | 0 | YES |
| | | Agree | 12 | 100 | |

| Method 5.1 (n = 12; N = 12) 5.1.1 Different | Ensure the inclusion of different teaching approaches in the of- fering of the training program. teaching approaches | Disa- gree | 0 | 0 | YES Agreement of |
|---|---|---------------|-----|-----|--------------------------------------|
| that should be utilised during pain | | RESPON | SES | | (≥60%) |
| management tra | aining. | = n | f = | = % | Yes/No |
| Writing reflective | e journals. | 7 | 58 | 3.3 | no |
| Conducting grai | nd rounds. | 7 | 58 | 3.3 | no |
| Engaging in foc | us groups. | 9 | 75 | 5 | yes |
| Using role-play | activities. | \8 | 66 | 6.7 | yes |
| Responsible person(s) to ensure that teaching approaches are in- cluded during pain management train- ing. (n = 12; N = 12) | | RESPONSES | | | CONSEN- SUS REACHED (≥ 75%) |
| | | = n | f = | = % | Yes/No |
| Director of Pos Nursing Educa KASCH. | stgraduate Center of ition for KAMC and | 7 | 58 | 3.3 | |
| One nurse educ Director of the P Nursing Educa KASCH. | cator appointed by the Postgraduate Center of ation for KAMC and | 8 | 66 | 5.7 | YES |
| Clinical facilitate | ors in all areas of nurs- MC and KASCH. | 6 | 50 |) | |

| One pain nurse specialist appointed | | | |
|---|----------|--------------|---------|
| by Clinical Directors of Nursing Oper- | 9 | 75 | |
| ations in every facility. | | | |
| Time frame within which time should | | | CONSEN- |
| different teaching approaches should | | | SUS |
| be part of the teaching program be- | RESPONSE | S | |
| fore implementation. | | | REACHED |
| (n = 12; N = 12) | | | (≥ 75%) |
| | = n | f = % | Yes/No |
| 2 weeks before the due date for finali- | 1 | 83 | |
| sation of the training program. | | 0.0 | |
| 1 month before the due date of the | 6 | 50 | NO |
| training program. | 0 | 50 | |
| 3 months before the due date of the | 5 | <i>A</i> 1 7 | |
| training program. | 0 | | |

6.4.6.1 Method 5.1: Ensure the inclusion of different teaching approaches in the offering of the training program (N = 12)

A 100% consensus (n = 12; N = 12) was reached that different teaching approaches must be included in the training program (see Table 6.12). 75% (n = 9; N = 12) of panellists indicated that focus groups are a teaching approach that should be included in the training program (see Table 6.12).

 a) Responsible person(s) to ensure that teaching approaches are included during pain management training (N = 12)

As illustrated in Table 6.12, 75% (n = 9; N = 12)) a consensus was reached that one pain nurse specialist, appointed by the clinical director of operations in every facility, will be the best responsible person to ensure that the different teaching approaches will be included in the offering of the training program (method 5.1).

b) Time frame within which different teaching approaches should be made part of the teaching program before implementation (N = 12)

Panellists did not reach a consensus on the time frame within which the different teaching approaches should be incorporated as part of the teaching program before implementation (see Table 6.12; method 5.1).

6.4.7 Action statement 6: Develop strategies to motivate nurses to participate in the short training program (N = 12)

Panellists were 100% (n = 12; N = 12) in agreement that strategies should be developed to motivate nurses to participate in the short training program to be included in the action plan (see Table 6.13).

Table 6.13: Action statement 6: Develop strategies to motivate nurses to par-
ticipate in the short training program (N = 12)

| | | RESPONS | SES | | CONSENSUS |
|-------------------|--|----------|-------|-------|--------------|
| | Develop strategies to motivate nurses | | | | REACHED |
| Action state- | gram. | | | | (≥ 75%) |
| ment 6 | | | = | f = % | Yes/No |
| | | | n | | |
| (n = 12; N = 10) | | Agree | 12 | 100 | YES |
| 12) | | Disagree | 0 | 0 | |
| 6.1 The following | g are strategies that will motivate nurses | RESPONS | ES | | Agreement by |
| to participate in | the pain management training program. | Iram. | | | (≥60%) |
| | | = n | f = % | 6 | YES/NO |
| Conduct a situa | ation analysis to assess the pain man- | 6 | 50 | | no |
| agement needs | of the nurses. | | | | |
| Involve nurses i | n the development of the content of the | 9 | 75 | | yes |
| training program | ۱. | | | | |
| Involve nurses i | n the development of learning goals and | 10 | 83.3 | } | Yes |
| learning outcon | nes for the pain management training | | | | |
| program relevar | nt to their nursing care areas. | | | | |

| Communicate the advantages of pain management | 8 | 66.7 | yes |
|--|---------|----------|-----------|
| competencies (on what platform can this be done). | | | |
| Create a supportive learning environment in nursing | 8 | 66.7 | yes |
| care areas. | | | |
| Responsible person(s) to develop the mentioned strat- | RESPONS | SES | CONSENSUS |
| egies that will motivate nurses to participate in the short | | | |
| pain management training program related to their nurs- | | | REACHED |
| ing care areas. (n = 12; N = 12). | | | (> 759/) |
| | | 1 | (275%) |
| | = n | t = % | YES/NO |
| Director of Postgraduate Center of Nursing Education | 8 | 66.7 | |
| for KAMC and KASCH. | | | |
| One Nurse educator appointed by the Director of the | | | |
| Postgraduate Center of Nursing Education for KAMC | 8 | 66.7 | |
| and KASCH. | | | NO |
| Clinical facilitators in all areas of nursing care for KAMC | g | 75 | |
| and KASCH. | | | |
| One pain nurse specialist to be appointed by Clinical Di- | | | |
| rectors of Nursing Operations in every facility. | | | |
| | 10 | 83.3 | |
| Time frame required to develop the strategies that will | RESPONS | SES | CONSENSUS |
| motivate nurses to participate in short pain management | | | |
| training program related to their nursing care areas.(n = | | | REACHED |
| 12; N = 12) | | | (> 75%) |
| | | f _ 0/ | |
| Queeke before the due date for the finalization of the | = 11 | 1 = 70 | |
| 2 weeks before the due date for the finalisation of the | 3 | 25 | NEO. |
| | | | YES |
| 1 month before the training program starts | 9 | 75 | |

6.4.7.1 Strategies 6.1: Types of strategies that will motivate the nurses to participate in the pain management training program (N = 12)

As illustrated in Table 6.13, 83.3% (n-10; N = 12) of panellists indicated the importance of involving nurses in the development of learning goals and learning outcomes for the pain management training program relevant to their nursing care areas as a strategy to motivate them, and 75% (n-9; N = 12) indicated that to involve nurses in the

development of the content of the training program (see Table 6.13; method 6.1) will be motivational.

 a) Responsible person(s) to develop strategies that will motivate nurses to participate in the short pain management training program related to their nursing care areas (N = 12)

83.3% (n = 10; N = 12) of panellists indicated that one pain nurse specialist appointed by the clinical director of nursing operations in every facility, and 75% (n = 9; N = 12) preferred clinical facilitators in all areas of nursing care for KAMC and KASCH (see Table 6.13; strategies 6.1).

 b) Time frame required to develop the strategies that will motivate nurses to participate in the short pain management training program related to their nursing care areas (N = 12)

9 (f = 75%; N = 12) panellists agreed that one month before the training program starts would be the appropriate timeframe for the development of the strategies to motivate nurses to participate in the pain management training program related to their nursing care areas (see Table 6.13; method 6.1).

6.4.8 Action statement 7: Motivate nurses to apply the knowledge gained in the training program in practice (N = 12)

A consensus (f = 100%, n = 12; N = 12) was reached that the action to motivate nurses to apply the knowledge gained in the training program in practice be included in the action plan (see Table 6.14).

Table 6.14: Action statement 7: Motivate nurses to apply the knowledge gainedin the training program in practice (N = 12)

| | RESPONSES | CONSENSUS |
|--------------------|-----------|-----------|
| Action statement 7 | | REACHED |
| (n = 12; N = 12) | | (≥ 75%) |

| | Motivate nurses to apply the | | = | f = | Yes/No |
|----------------------------|---------------------------------------|----------|-----|----------|--------------|
| | knowledge gained in the training | | n | % | |
| | program in practice. | Agree | 12 | 100 | VES |
| | | Disagree | 0 | 0 | TES |
| 7.1 Method: The follow | ving methods can be implemented | RESPONS | SES | | Agreement of |
| to motivate nurses to a | pply their knowledge in practice. | | | | (≥60%) |
| | | = n | | f = % | Yes/No |
| Provide nurses with a | certificate to recognize their appli- | 8 | | 66.7 | yes |
| cation of pain manager | nent knowledge in their respective | | | | |
| nursing care areas. | | | | | |
| Offer nurses the oppo | rtunity to take on the role of pain | 12 | | 100 | yes |
| management experts w | /ho are competent in their field. | | | | |
| Allow the nurses to tal | ke part in planning outcomes of a | 8 | | 66.7 | yes |
| pain management train | ing program. | | | | |
| Support nurses' SMA | RT goals and pain management | 9 | | 75 | yes |
| learning. | | | | | |
| Support aspects that di | ive individual nurses to apply what | 9 | | 75 | yes |
| they have learned abou | ut pain management. | | | | |
| Assign grades for appl | ying pain management knowledge | 8 | | 66.7 | yes |
| in practice based on ar | inual performance. | | | | |
| Responsible person(s) | to facilitate the implementation of | RESPONS | SES | | CONSENSUS |
| the aspect to motivate | nurses to apply their knowledge in | | | | |
| practice (n = 12; N = 12 | 2). | | | | REACHED |
| | | | | | (≥ 75%) |
| | | = n | | f = % | Yes/No |
| | | | | 22.2 | |
| Associate Executive D | irectors of Nursing for KAMC and | 4 | | 33.3 | |
| KASCH. | | | | | |
| Clinical Directors of N | ursing Operations for KAMC and | 9 | | 75 | |
| KASCH. | | | | | |
| Nurse supervisors app | ointed by Executive Associate Di- | 5 | | 41.7 | NO |
| rectors of Nursing in ev | ery facility for KAMC and KASCH. | | | | |

| Nurse Managers in all | nursing care areas for KAMC and | 11 | 91.7 | |
|---|--------------------------------------|-----------|------|--------------|
| KASCH. | | | | |
| Director of Postgraduat | te Center of Nursing Education for | Λ | 33.3 | |
| KAMC and KASCH. | | 4 | 33.3 | |
| One Nurse educator ap | pointed by the Director of the Post- | | | |
| graduate Center of N | ursing Education for KAMC and | 8 | 66.7 | |
| KASCH. | | | | |
| One Pain nurse specia | list appointed by Clinical Directors | 8 | 66.7 | |
| of Nursing Operations i | n every facility. | 0 | 00.7 | |
| Time frame required to | provide the aspects for motivation | RESPONSES | 1 | CONSENSUS |
| to nurses to apply their knowledge in practice (n = 12; N = | | | | |
| 12). | | | | REACHED |
| | | | | (() |
| | | | | (≥ 75%) |
| | | = n | f = | Yes/No |
| | | | % | |
| Time frame required | 1 week after the training program | 2 | 16.7 | |
| to provide the aspects | 1-3 months after the training pro- | Б | 11 7 | |
| for motivating nurses | gram | 5 | 41.7 | |
| to apply their | 4-6 months after the training pro- | e | 50 | |
| knowledge in practice | gram | 0 | 50 | NO |
| | 7-9 months after the training pro- | 0 | 0 | |
| (n = 12; N = 12) | gram | 0 | 0 | |

6.4.8.1 Methods 7.1: Methods that can be implemented to motivate nurses to apply their knowledge in practice (N = 12)

According to Table 6.14, all panellists unanimously agreed (100%; n = 12; N = 12) that nurses should be offered the opportunity to take on the role of pain management experts who are competent in their field. 75% (n = 9; N = 12) of panellists reached a consensus that supporting nurses' SMART goals and pain management learning, as well as supporting what motivates individual nurses to apply what they have learned about pain management, will be implemented to motivate nurses to apply their knowledge in practice.

a) Responsible person(s) to facilitate the implementation of methods to motivate nurses to apply their knowledge in practice (N = 12)

91.7% (n = 11; N = 12) of panellists indicated that the nurse managers in all nursing care areas for KAMC and KASCH, and 75% (n = 9; N = 12) of the clinical directors of nursing operations must be responsible for facilitating the aspect to motivate nurses to apply their knowledge in practice (see Table 6.14; method 7.1).

b) Time frame required to implement the methods to motivate nurses to apply their knowledge in practice (N = 12)

Panellists did not reach a consensus on the timeframe for implementing the aspects that motivate nurses to apply their knowledge in practice (see Table 6.14; method 7.1).

To develop a second draft of the action plan with an embedded validation tool, all the comments and suggestions received from the panellists were incorporated into the validation tool, along with all the items on which consensus was not reached in the first round. A recruitment letter (see Annexure 16) similar to the procedure described in 6.2.7 was utilised for the second round. All panellists from the first round were invited through the gatekeepers. The second round's findings are discussed considering the Google Forms analysis.

6.5 FINDINGS FROM THE SECOND ROUND

For round 2, the validation instrument has been amended as follows: All items where consensus had been reached were indicated as "**consensus reached**" within the validation instrument, as illustrated in Annexure 15. Panelists were requested to respond only to those items on which consensus was *not* reached.

To validate the online survey, the validation instrument was again loaded onto Google Forms, as was the case in Round 1. As outlined in Annexure 16, the recruitment letter to the gatekeepers was shared with 12 panellists with clear instructions on what would be expected in the second round.

In the second round of data collection, a 100% response rate was achieved, which is noteworthy considering that existing research suggests a significant likelihood of

experts withdrawing from the study (Keeney et al. 2011:29). Nevertheless, the absence of any panellists being lost in the second phase of the study ensured that this was not an issue in this study. During the second round, a consensus was reached on certain items, albeit not all, as seen in the tables below (see Annexure 17 for the full findings).

Note: <u>Only</u> the findings for action statements 1 to 7 about those aspects where consensus was <u>not reached</u> in round one are discussed in full in the below Tables 6.15 to 6.28 to prevent repetition of findings (see Annexure 17).

6.5.1 Action statement 1: Motivate nurses to further their studies (N = 12)

6.5.1.1 Responsible person(s) to develop a policy to motivate nurses to improve their nursing qualifications (for method 1.1) (N = 12)

Table 6.15 (method 1.1) illustrates that consensus was not reached due to divergent perspectives on the person(s) responsible for developing a policy to motivate nurses to improve their nursing qualifications.

Table 6.15:Responsible person(s) to develop a policy to motivate nurses to
improve their nursing qualifications (N = 12)

| Responsible person(s) to develop a policy to motivate nurses to im- | Respo | onses | Consensus |
|---|-------|---------|-----------|
| ve their nursing qualifications. | | Reached | |
| | | | (≥ 75%) |
| | n = | f = % | Yes/No |
| Nursing policy committee representatives appointed by the Associate | 5 | 41.7 | |
| Executive Directors of Nursing for KAMC and KASCH. | | | |
| Clinical Directors of Nursing Operations for KAMC and KASCH. | 4 | 33.3 | No |
| Director of Postgraduate Center of Nursing Education for KAMC and | 3 | 25 | |
| KASCH. | | | |

6.5.1.2 Time frame required to present the policy and negotiate the implementation to the MNGHA) (N = 12)

As illustrated in Table 6.16 (method 1.1), panellists did not reach a consensus on the time frame within which the policy should presented and the implementation negotiated with the MNGHA.

Table 6.16: Time frame required to present the policy and negotiate the imple-
mentation to the MNGHA) (N = 12)

| Time frame required to present the policy and negotiate | Re- | | Consen- |
|---|------|------|---------|
| the implementation to the Ministry of National Guard | spon | ses | sus |
| Health Affairs (MNGHA) | | | |
| | | | Reached |
| | | | (> 75%) |
| | | | (27570) |
| | | | |
| | n = | f = | Yes/No |
| | | % | |
| 6 months | 8 | 66.7 | |
| 9 months | 4 | 33.3 | |
| | | | No |

6.5.1.3 Responsible person(s) to include the policy in all hospitals' policies to motivate nurses to improve their nursing qualifications (for Method 1.3) (N = 12)

Table 6.17 (method 1.3) illustrates that the panellists did not reach a consensus as they reflected different views on who the responsible person(s) must be that should ensure the inclusion of the policy in all hospitals' policies to motivate nurses to improve their nursing qualifications.

Table 6.17:Responsible person(s) to include the policy in all hospitals' policiesto motivate nurses to improve their nursing qualifications (N = 12)

| Responsible person(s) to include the policy in all hospitals' policies to motivate nurses to improve their nursing qualifications | Responses | | Responses | | Consen- sus Reached (≥ 75%) |
|--|-----------|-------|-----------|--|--------------------------------------|
| | n = | f = % | Yes/No | | |
| Clinical Directors of Nursing Operations in every facil- ity for KAMC and KASCH. | 4 | 33.3 | No | | |
| Nurse Managers in all KAMC and KASCH nursing care areas. | 8 | 66.7 | | | |

6.5.1.4 Time frame required to include the policy in all hospitals' policies after approval of the action plan by MNGHA (for method 1.3) (N = 12)

Panellists reached a consensus (n = 9; f = 75%) that the policy must be included within three months in all the hospitals' policies after approval of the action plan by the Ministry of National Guard Health Affairs (see Table 6.18; method 1.3).

Table 6.18: Time frame required to include the policy in all hospitals' policiesafter approval of the action plan by MNGHA (N = 12)

| Responses | Consensus |
|-----------|-----------|
| | Reached |

| Time frame required to include the policy in all hospi- | | | (≥ 75%) |
|--|-----|-------|---------|
| tals' policies after approval of the action plan by the Ministry of National Guard Health Affairs. | n = | f = % | Yes/No |
| 1 month | 3 | 25 | |
| 3 months | 9 | 75 | Yes |

6.5.2 Action statement 2: Make appropriate and relevant pain management tools accessible to the nursing team in every clinical area (N = 12)

6.5.2.1 **Responsible person(s)** to ensure that an electronic format of pain assessment tools are available for inclusion in the electronic patient record system so that they can be accessible to the nursing team in every nursing care area (for method 2.1) (N = 12).

Seventy-five per cent (n = 9; N = 12) consensus was reached that one pain nurse specialist appointed by clinical directors of nursing operations in every facility for KAMC and KASCH must provide the electronic format of the pain assessment tools for inclusion in the electronic patient record system to be accessible to the nursing team in every nursing care area (see Table 6.19; method 2.1).

Table 6.19: Responsible person(s) to ensure that an electronic format of the pain assessment tools for inclusion in the electronic patient record system is accessible to the nursing team in every nursing care area (N = 12)

| Responsible person(s) to ensure that an elec- | Responses | Consensus |
|--|-----------|-----------|
| tronic format of the pain assessment tools for in- | | Decelerat |
| clusion in the electronic patient record system is | | Reached |
| elusion in the electronic patient record system is | | |

| accessible to the nursing team in every nursing care area. | | | (≥ 75%) |
|---|-----|-------|---------|
| | n = | f = % | Yes/No |
| Five nursing health informatics specialists appointed by Associate Directors of Nursing for KAMC and KASCH. | 3 | 25 | Yes |
| One pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility for KAMC and KASCH. | 9 | 75 | |

6.5.2.2 Time frame required to include the pain assessment tools in the electronic patient record system (Method 2.1) (N = 12)

As illustrated in Table 6.20, consensus was not reached in round 2 on the time frame within which the pain assessment tools should be available in the electronic patient record system.

Table 6.20: Time frame required to include the pain assessment tools in theelectronic patient record system (N = 12)

| Time frame required to include the pain as- sessment tools in the electronic patient record system. | Respons | es | Consensus Reached (≥ 75%) |
|---|-----------|----|---------------------------------|
| | n = f = % | | Yes/No |

| 1- 3 months | 7 | 58.3 | |
|-------------|---|------|----|
| 4- 6 months | 5 | 41.7 | No |

6.5.2.3 Responsible person(s) to involve nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessments (for method 2.2) (N = 12)

Consensus was not reached about who should be responsible for involving the nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessments (see Table 2.1).

Table 6.21: Responsible person(s) to involve nurse supervisors with pain man-
agement training to provide supervisory support to the nursing
team on how to conduct pain assessments (N = 12)

| Responsible person(s) to involve nurse super- visors with pain management training to provide post-pain management training and supervisory support to the nursing team on how to conduct pain assessments | Responses | | Consensus Reached (≥ 75%) |
|---|-----------|-------|---------------------------------|
| | n = | f = % | Yes/No |
| Clinical Directors of Nursing Operations for KAMC and KASCH | 6 | 50 | No |
| Clinical facilitators in all KAMC and KASCH nursing care areas | 6 | 50 | |

6.5.2.4 Time frame required to involve the nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessments ((for Method 2.2) (N = 12)

Consensus was not reached on the time frame within which the nurse supervisors should be involved in providing pain management training and supervisory support to the nursing team on how to conduct pain assessment (see Table 6.22; method 2.2).

Table 6.22: Time frame required to involve the nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessment in all nursing care areas (N = 12)

| Time frame required to involve the nurse su- pervisors with pain management to provide su- pervisory support to the nursing team on how to conduct pain assessment in all nursing care | Responses | | Consensus Reached (≥ 75%) |
|--|-----------|-------|---------------------------------|
| areas | n = | f = % | Yes/No |
| Every shift when the need arises | 50 | 50 | |
| Every patient round when the need arises | 50 | 50 | No |

6.5.2.5 Responsible person(s) to ensure that internet-based resources are accessible to patient and family to provide support about pain management (for method 2.3) (N = 12)

75 % (n = 9; N = 12) of panellists indicated that nurse managers in all KAMC and KASCH nursing care areas would be the best persons to ensure that internet-based resources are accessible to patients and families to provide support about pain management (see Table 6.23; method 2.3).

Table 6.23: Responsible person(s) to ensure that internet-based resources are
accessible to patient and family to provide support about pain man-
agement (N = 12)

| Responsible person(s) to ensure internet- based resources are accessible to patients and family members to provide support about pain management | Responses | | Consensus Reached (≥ 75%) |
|--|-----------|-------|---------------------------------|
| | n = | f = % | Yes/No |
| Nurse Managers in all KAMC and KASCH nurs- ing care areas | 9 | 75 | Yes |
| Charge nurses in all KAMC and KASCH nursing care areas | 0 | 0 | |
| Registered nurses in all KAMC and KASCH nursing care areas | 3 | 25 | |

6.5.2.5 Responsible person(s) to ensure that internet-based resources on pain management publications, electronic materials, and organisations that specialise in pain management are made accessible to the nursing team in all nursing care areas (for method 2.4) (N = 12)

Consensus was not reached regarding who the responsible person(s) should be to ensure that internet-based resources on pain management publications, electronic materials, and organisations that specialise in pain management are made accessible to the nursing team in all nursing care areas (see Table 6.24; method 2.4).

Table 6.24Responsible person(s) to ensure that internet-based resources on
pain management publications, electronic materials, and organisa-
tions that specialise in pain management are made accessible to
the nursing team in all nursing care areas (N = 12)

| Responsible person(s) to ensure that internet-based resources on pain management publications, electronic materials, and organisations that specialise in pain management are made accessible to the nursing team in all pursing care areas | Responses | | Consensus Reached (≥ 75%) |
|--|-----------|-------|---------------------------------|
| | n = | f = % | Yes/No |
| Nursing health informatics appointed by Associated Di- rectors of Nursing for KAMC and KASCH | 5 | 41.7 | No |
| One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility for KAMC and KASCH | 7 | 58.3 | |

6.5.3. Action statement 3: Develop a practice-oriented content-specific short pain management training program (N = 12)

6.5.3.1 Time frame required to provide for inclusion of the specific oriented pain management training in the program (for Method 3.1) (N = 12)

83.3% (n = 10; N = 12) of panellists reached a consensus that the specific practiceoriented pain management training content should be included in the program three months before the due date of the training program (see Table 6.25; method 3.1).

Table 6.25: Time frame required to provide for inclusion of the specific orientedpain management training in the program (for method 3.1) (N = 12)

| Time frame required to provide for the inclusion of | Responses | Consensus |
|---|-----------|-----------|
| the specific oriented pain management training in the program | | Reached |
| | | (≥ 75%) |

| | n = | f = % | Yes/No |
|---|-----|-------|--------|
| 1 month before the due date of the training program | 2 | 16.7 | |
| 3 months before the due date of the training pro- gram | 10 | 83.3 | Yes |

6.5.4 Action statement 4: Develop a pain management short program that accommodates all learning types (N = 12)

6.5.4.1 Time frame required to ensure that the learning types are shared and included within the training program (for method 4.1) (N = 12)

83.3% (n = 10; N = 12) of panellists reached a consensus that the appropriate time frame to ensure sharing and inclusion of the learning types within the training program must be three months before the due date of the training program (see Table 6.26; method 4.1).

Table 6.26: Time frame required to ensure that the learning types are sharedand included within the training program (N = 12)

| Time frame required to ensure that the learn- ing types are shared and included within the training program | Respons | es | Consensus Reached (≥ 75%) |
|--|---------|-------|---------------------------------|
| | n = | f = % | Yes/No |

| 1 month before the due date of the training program | 2 | 16.7 | Yes | |
|--|----|------|-----|--|
| 3 months before the due date of the training program | 10 | 83.3 | | |

- 6.5.5 Action statement 5: Incorporate different teaching approaches to accommodate diverse learners and facilitators in the training of pain management (N = 12)
- 6.5.5.1 Time frame within which time should different teaching approaches be part of the teaching program before implementation (for method 5.1 (N = 12)

As illustrated in Table 6.27 (method 5.1), 75% (n = 9; N = 12) of panellists reached a consensus that different teaching approaches should be included as part of the teaching program three months before the due date of when the training program is to be implemented.

Table 6.27: Time frame within which time should different teaching approachesshould be part of the teaching program before implementation (N =12)

| Time frame within which time should different teaching | Responses | Consen- |
|--|-----------|---------|
| approaches be part of the teaching program before im- | | sus |
| plementation. (n = 12; N = 12) | | Reached |
| | | (≥ 75%) |
| | | |

| | n = | f = % | Yes/No |
|--|-----|-------|--------|
| 1 month before the due date of the training program | 3 | 25 | |
| 3 months before the due date of the training program | 9 | 75 | Yes |

6.5.6 Action statement 6: Develop strategies to motivate nurses to participate in the short training program (N = 12)

6.5.6.2 Responsible person(s) to develop strategies that will motivate nurses to participate in the short pain management training program related to their nursing care areas (for strategies 6.1) (N = 12)

A consensus of 75% (n = 9; N = 12) was reached that one pain nurse specialist appointed by the Clinical Directors of Nursing Operations in every facility will be the best person to take responsibility for developing strategies to motivate nurses to participate in the short pain management training program (see Table 6.28; strategies 6.1).

Table 6.28: Responsible person(s) to develop the mentioned strategies thatwill motivate nurses to participate in the short pain managementtraining program related to their nursing care areas (N = 12)

| Responsible person(s) to be responsible | Responses | Consensus | |
|---|-----------|-----------|--------|
| for developing the mentioned strategies | | Reached | |
| that will motivate nurses to participate in the | | | |
| short pain management training program | | (≥ 75%) | |
| related to their nursing care areas. | | | |
| | n = | f = % | Yes/No |

| Clinical facilitators in all areas of nursing care for KAMC and KASCH. | 3 | 25 | Yes |
|--|---|----|-----|
| One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility. | 9 | 75 | |

- 6.5.7 Action statement 7: Motivate nurses to apply the knowledge gained in the training program in practice (N = 12)
- 6.5.7.1 Responsible person(s) to facilitate the implementation of methods to motivate nurses to apply their knowledge in practice (for method 7.1) (N = 12)

Table 6.29 (method 7.1) illustrates that 75% (n = 9; N = 12) of panellists agreed that directors of nursing operations for KAMC and KASCH must be responsible for facilitating the implementation of methods motivating nurses to apply their knowledge in practice.

Table 6.29: Responsible person(s) to facilitate the implementation of the meth-
ods to motivate nurses to apply their knowledge in practice (N = 12)

| Responsible person(s) to facilitate the implementa- tion of the methods to motivate nurses to apply their knowledge in practice. | Responses | | Consensus Reached (≥ 75%) |
|--|-----------|-------|---------------------------------|
| | n = | f = % | Yes/No |

| Nurse Managers in all nursing care areas for KAMC and KASCH | 3 | 25 | Yes |
|---|---|----|-----|
| Clinical Directors of Nursing Operations for KAMC and KASCH | 9 | 75 | |

6.5.7.2 Time frame required to implement the methods to motivate nurses to apply their knowledge in practice (for method 7.1) (N = 12)

Consensus was not reached on the applicable time frame regarding the implementation of the suggested methods that will motivate nurses to apply their knowledge in practice (see Table 6.30; method 7.1).

Table 6.30: Time frame required to implement the methods to motivate nursesto apply their knowledge in practice (N = 12)

| Time frame required to implement the methods to motivate nurses to apply their knowledge in practice | Responses | | Consensus Reached (≥ 75%) |
|--|-----------|-------|---------------------------------|
| | n = | f = % | Yes/No |
| 1–3 months after the training program | 6 | 50 | |
| 4–6 months after the training program | 6 | 50 | No |

6.6 FINDINGS FROM ROUND THREE

6.6.1 Demographic characteristics

During the third round of data collection in the Delphi process, two out of the twelve panellists did not participate, resulting in a response rate of 83.3% (n = 10; N = 12). This size of 10 panel members was still acceptable, as suggested by Touma (2022: 750) that the size of ample usually varies from ten to eighteen experts. After comparing all the demographic information from the three rounds of data sets, it was clear that the two panellists lost in this round were both registered nurses from the Philipines, having Bachelor's degrees at the highest level of education. This is noteworthy, as Touma (2022:758) indicated that certain experts are likely to withdraw at some stage of the study. As seen by the tables below, a consensus was reached among the ten panel members on all items not reached during Rounds 1 and 2.

Note: <u>Only</u> the findings for action statements 1 to 7 of those aspects where consensus was <u>not reached</u> in Round 2 are discussed below in Tables 6.31 to 6.39 in full to eliminate the repetition of findings (see Annexure 21).

6.6.2 Action statement 1: Motivate nurses to further their studies (N = 10)

6.6.2.1 Responsible person(s) to develop a policy to motivate nurses to improve their nursing qualifications (for method 1.1) (N = 10)

All agreed (f = 100%; n = 10; N = 10) that nursing policy committee representatives appointed by the associate executive committee directors of nursing for KAMC and KASCH will be suitable persons responsible for developing a policy intended at motivating nurses to improve their nursing qualifications (see Table 6.31, method 1.1).

Table 6.31: Responsible person(s) to develop a policy to motivate nurses toimprove their nursing qualifications (N = 10)

| Responsible person(s) to develop a policy to motivate nurses to improve their nursing qualifications. | Responses | | Consen- sus Reached (≥ 75%) Yes/No |
|--|-----------|----------|--|
| | n = | f = % | Yes/No |
| Nursing policy committee representatives appointed by the Associate Executive Directors of Nursing for KAMC and KASCH. | 10 | 100 | Yes |
| Clinical Directors of Nursing Operations for KAMC and KASCH. | 0 | 0 | |

6.6.2.2 Time frame required to present the policy and negotiate for the implementation to the MNGHA (for method 1.2) (N = 10)

The panellists reached a consensus (f = 90%; n = 9; N = 10) that the policy should be presented and the implementation negotiated to the Ministry of National Guard Health Affairs within six months (see Table 6.32; method 1.2).

Table 6.32: Time frame required to present the policy and negotiate for the implementation to the MNGHA (for method 1.2) (N = 10)

| Time frame required to present the policy and negoti- ate for the implementation to the MNGHA | Responses | | Consensus Reached (≥ 75%) |
|---|-----------|----------|---------------------------------|
| | n = | f = % | Yes/No |
| 1. 6 months | 9 | 90 | Yes |
| 2. 9 months | 1 | 10 | |

6.6.2.4 Responsible person(s) to include the policy in all hospitals' policies to motivate nurses to improve their nursing qualifications (for method 1.3) (N = 10)

As illustrated in Table 6.33 (method 1.2), the panellists reached a consensus that clinical directors of nursing operations in every facility for KAMC and KASCH must be the best responsible persons to incorporate the policy in all hospitals' policies to motivate nurses to improve their nursing qualifications.

Table 6.33:Responsible person(s) to include the policy in all hospitals' policiesto motivate nurses to improve their nursing qualifications (N = 10)

| Responsible person(s) to include the policy in all | Responses | Consen- |
|---|-----------|---------|
| hospitals' policies to motivate nurses to improve their | | sus |
| nursing qualifications (n = 10; N = 10). | | Reached |
| | | (≥ 75%) |

| | n = | f = % | Yes/No |
|---|-----|----------|--------|
| Clinical Directors of Nursing Operations in every facil- ity for KAMC and KASCH. | 9 | 90 | Yes |
| Nurse Managers in all KAMC and KASCH nursing care areas. | 1 | 10 | |

6.6.3 Action statement 2: Make appropriate and relevant pain management tools accessible to the nursing team in every clinical area (N = 10)

6.6.3.1 Time frame required to include the pain assessment tools in the electronic patient record system (for method 2.1) (N = 10)

90% (n = 9; N = 10) of panellists reached a consensus that the pain assessment tools should be incorporated into the electronic patient record system within 1-3 months after implementing the action plan (see Table 6.34; method 2.1).

Table 6.34: Time frame required to include the pain assessment tools in theelectronic patient record system (N = 10)

| Time frame required to include the pain assess- | Respo | Responses Consensus | | sponses Consensus |
|--|-------|---------------------|--------------------|-------------------|
| ment tools in the electronic patient record system | | | Reached (≥ 75%) | |
| | | | | |
| | n = | f = % | Yes/No | |

| 1–3 months | 9 | 90 | Yes |
|------------|---|----|-----|
| 4–6 months | 1 | 10 | |

6.6.3.1 Responsible person(s) to involve nurse supervisors with pain management training to provide post-pain management training and supervisory support to the nursing team on how to conduct pain assessments (for method 2.2) (N = 10)

Consensus was reached (n = 9; N = 10) that clinical directors of nursing operations for KAMC and KASCH are suitable persons for involving the nurse supervisors with pain management training in providing post-pain management training and supervisory support to the nursing team on how to conduct pain assessments (see Table 6.35; method 2.2).

Table 6.35:Responsible person(s) to involve nurse supervisors with pain man-
agement training to provide post-pain management training and
supervisory support to the nursing team on how to conduct pain
assessments (N = 10)

| Responsible person(s) to involve nurse supervisors with pain management training to provide post-pain management training and supervisory support to the nursing team on how to conduct pain assessments. | Responses | | Consensus Reached (≥ 75%) |
|--|-----------|-------|---------------------------------|
| | n = | f = % | Yes/No |
| Clinical Directors of Nursing Operations for KAMC and KASCH. | 9 | 90 | Yes |

| Clinical facilitators in all KAMC and KASCH nurs- | 1 | 10 | |
|---|---|----|--|
| ing care areas. | | | |

6.6.3.2 Time frame required to involve the nurse supervisors to provide pain management training supervisory support to the nursing team on how to conduct pain assessment in all nursing care areas (for method 2.2) (N = 10)

Panellists all agreed (f = 100%; n = 10; N = 10) that the nurse supervisors should be involved in every shift when the need arises to provide pain management training and supervisory support to the nursing team on how to conduct pain assessment (see Table 6.36; method 2.2).

Table 6.36: Time frame within which the nurse supervisors should be involved in providing pain management training supervisory support to the nursing team on how to conduct pain assessment in all nursing care areas (N = 10)

| Time frame required to involve the nurse supervisors to provide pain management training supervisory support to the nursing team on how to conduct pain assessment in all nursing care areas. | Responses | | ervi- Responses per- con- s. | Consensus Reached (≥ 75%) |
|--|-----------|-------|---------------------------------------|---------------------------------|
| | n = | f = % | Yes/No | |
| Every shift, when the need arises. | 10 | 100 | Yes | |
| Every patient round when the need arises. | 0 | 0 | | |

6.6.3.4 Responsible person(s) to ensure internet-based resources on pain management publications, electronic materials, and organisations that specialise in pain management should be accessible to the nursing team in all nursing care areas (for method 2.4) (N = 10)

90% (n = 9; N = 10) of panellists reached a consensus that nursing health informatics appointed by associated directors of nursing for KAMC and KASCH will be the best responsible persons to ensure internet-based resources on pain management publications, electronic materials, and organisations that specialise in pain management should be accessible to the nursing team in all nursing care areas (see Table 6.37; method 2.4).

Table 6.37: Responsible person(s) to ensure internet-based resources on pain management publications, electronic materials, and organisations that specialise in pain management should be accessible to the nursing team in all nursing care areas (N = 10)

| Responsible person(s): The best possible individual/individuals to ensure internet-based resources on pain management publications, electronic materials, and organisations that specialise in pain management should be ac- | Responses | | Consensus Reached (≥ 75%) |
|---|-----------|--------|---------------------------------|
| cessible to the nursing team in all nursing care areas ($N = 10$). | f = % | Yes/No | f = % |
| Nursing health informatics appointed by Asso- ciated Directors of Nursing for KAMC and KASCH. | 9 | 90 | Yes |
| One pain nurse specialist is appointed by the | 1 | 10 | |
|---|---|----|--|
| Clinical Directors of Nursing Operations in | | | |
| every facility for KAMC and KASCH. | | | |
| | | | |

6.6.4 Action statement 7: Motivate nurses to apply the knowledge gained in the training program in practice (N = 10)

6.6.4.1 Time frame required to implement the methods to motivate nurses to apply their knowledge in practice (N = 10)

As illustrated in Table 6.38 (method 7.1), 90% (n = 9; N = 10) consensus was reached that the methods to motivate nurses to apply their knowledge in practice must be in place within 1-3 months after the acceptance of the action plan.

Table 6.38: Time frame required to implement the methods to motivate nursesto apply their knowledge in practice (N = 10)

| Time frame required to implement the methods to motivate nurses to apply their knowledge in prac- tice. | Respo | nses | Consensus Reached (≥ 75%) |
|---|-------|-------|---------------------------------|
| | n = | f = % | Yes/No |
| 1–3 months after the training program. | 9 | 90 | Yes |
| 4–6 months after the training program. | 1 | 10 | |

6.7 THE VALIDATED FINAL ACTION PLAN

In Phase 5 of this study, the panellists validated the draft action plan and, following three rounds of Delphi, reached a consensus on the action plan intended to enhance the transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals. Table 6.39 illustrates the final validated action plan to enhance the transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals.

| Table 6.39: | The validated action plan to enhance the transfer of learning of pa | |
|-------------|---|--|
| | management competencies of nurses in Saudi Arabian teaching | |
| | hospitals. | |

| Action state- ment | Methods | Responsible person(s) | Timeframe |
|---|--|---|--|
| 1. Motivate nurses to fur- ther their studies. | 1.1 Develop a policy to motivate nurses to improve their nursing qualifica- tions by including a certificate issued as an acknowledgement of nurses pursuing distance learning, one day off for attending pain management programs, monetary incentive after com- pletion of a pain man- agement program (degree or diploma) and after completion | Nursing policy committee rep- resentatives, appointed by the associate executive direc- tors for KAMC and KASCH. | The policy to mo- tivate nurses to improve their nursing qualifica- tions must be de- veloped and fi- nalised 4-6 months after ap- proval of the ac- tion plan by the MNGHA. |

| of distance learning programs, and offer free accommodation for the period of study leave 1.2 Present and negotiate for implementing the policy to the Ministry of National Guard Health Affairs (MNGHA) through the Central Region Nurs- ing Governance and Accountability Board. | Nursing policy committee rep- resentatives ap- pointed by the Associate Exec- utive Director of Nursing for KAMC and KASCH | The policy should be presented, and the imple- mentation thereof negotiated with the Ministry of National Guard Health Affairs within six months after ap- proval of the ac- tion plan. |
|--|--|--|
| 1.3 Include the policy as part of the policies of all hospi- tals. | Clinical direc- tors of nursing operations in every facility for KAMC and KASCH | The policy should be included within three months after ap- proval of the ac- tion plan by the Ministry of Na- tional Guard Health Affairs. |

| 2. Make appropriate and relevant pain management tools accessible to the nursing team in every clinical area. | 2.1 Include all pain assessment tools, including PQRST and CRIES, in an electronic patient record system so the nursing team can choose and access the most appropriate tool in all nursing care areas. 2.2 Involve nurse supervisors with pain management training to provide post-pain management training supervisory support to the nursing team on how to conduct pain assessments in all nursing care areas. | One pain nurse specialist ap- pointed by clini- cal directors of nursing in every facility for KAMC and KASCH Clinical Direc- tors of Nursing Operations for KAMC and KASCH | The pain assess- ment tools should be in- cluded within 1–3 months in the electronic patient record system af- ter implementing the action plan in KAMC and KASCH. Nurse supervi- sors must be available on every shift to provide pain management training and su- pervisory support to the nursing team (when the need arises) af- |
|---|---|---|---|
| | | | ter approval of the action plan. |
| | 2.3 Make all hospitals' internet-based re- sources, including websites, support groups, hotlines, and | Nurse Manag- ers in all KAMC and KASCH nursing care ar- eas. | The hospitals' in- ternet-based re- sources must be made accessible to the patients |

| | peer support groups, accessible to pa- tients and family members to obtain other support about pain management. | | and family mem- bers to obtain other support about pain man- agement, 24 hours per day and seven days a week, after ap- proval of the ac- tion plan. |
|---|--|--|--|
| | 2.4 Make all hospitals' internet-based re- sources, including publications, elec- tronic materials, and organizations that specialize in pain management, acces- sible to the nursing team in all nursing care areas that con- sist of pain toolkits, videos on pain man- agement, and clinical updates. | Nursing health informatics ap- pointed by As- sociate Direc- tors of Nursing for KAMC and KASCH. | The hospitals' in- ternet-based re- sources, publica- tions, electronic materials, and or- ganizations that specialize in pain management should be contin- uously available. |
| Develop a practice-ori- ented con- tent-specific short pain | a. Include practice-ori- ented pain management training content, in- clusive of *the | One pain nurse specialist ap- pointed by Clini- cal Directors of Nursing | The practice-ori- ented content- specific short pain |

| manage- | methods to promptly | Operations in | management |
|--------------|-------------------------|-------------------|---------------------|
| ment train- | assess a patient's | every facility. | training program |
| ing program. | pain in all nursing ar- | | must be available |
| | eas; | | three months |
| | * The selection of | | before the due |
| | appropriate pain in- | | date for the of- |
| | tervention strategies | | fering of the |
| | based on the pain | | training pro- |
| | levels assessed; | | gram. |
| | *the advantages and | | |
| | disadvantages of all | | |
| | pain management | | |
| | scales and *assess- | | |
| | ment of patients' | | |
| | pain in all nursing | | |
| | care areas. | | |
| | | | |
| 4. Develop a | a. Accommodate differ- | One pain nurse | |
| pain | ent | specialist ap- | Adventise the in- |
| manage- | learning types and | pointed by Clini- | the different |
| ment short | learning styles | cal Directors of | |
| program | nurses used to | Nursing Opera- | he accommo |
| that accom- | achieve them when | tions in every | dated in the short |
| modates all | developing the train- | facility | program three |
| learning | ing program, specifi- | | months before |
| types. | cally | | the due date of |
| | *creative learners | | the offering of |
| | (generate creative | | the training pro- |
| | ideas in a group), | | aram |
| | *enthusiastic thinking | | gram. |
| | learners (listen to the | | |
| | information actively, | | |

| | take part in the act ity to practice the s and participate in the group discussion), well as * organised thinking learners (solve different real-life prob- lems). | tiv- skill the as ig fer- | |
|--|---|--|--|
| 5. Incorporate different teaching ap- proaches to accommo- date diverse learners and facilitators in pain man- agement training. | a. Ensure the inclusion of different teaching approaches, including focus groups a role play, in the offing of the training program. | on One pain nurse specialist ap- pointed by Clini- and cal Directors of fer- Nursing opera- tions in every facility. | Different teach- ing approaches must be offered in the teaching program and must be available to learners and facilitators three months before the due date of when the training program is to be implemented. |
| 6. Develop strat- egies to moti- vate nurses to participate in the short | a. Involve nurses in t *development of th content, goals, and outcomes of the training program, *communicate the | he One pain nurse specialist ap- pointed by Clini- cal Directors of nursing | Nurses must be invited to partici- pate in the devel- opment of the content, goals, |

| training pro- | advantages of pain | operations in | and outcomes of |
|---------------|----------------------------|------------------|-------------------------|
| gram. | management compe- | every facility. | the training one |
| | tencies (the platform | | month before |
| | that can be utilized), | | the training pro- |
| | and | | gram is to be of- |
| | b. *create a supportive | | fered. |
| | learning environ- | | |
| | ment. | | |
| 7. Motivate | a. *Offer nurses the op- | Nurse Manag- | |
| nurses to ap- | portunity to take on | ers in all nurs- | The implementa- |
| ply the | the role of a pain | ing care areas | tion of methods |
| knowledge | management expert | for KAMC and | to motivate |
| gained in the | who is competent in | KASCH are the | nurses to apply |
| training pro- | the field, *support | best persons to | their knowledge |
| gram in prac- | nurses' SMART | be responsible | in practice must |
| tice. | goals and pain man- | for facilitating | be within 1-3 |
| | agement learning, | the implementa- | months after the |
| | *support what drives indi- | tion of the as- | training pro- |
| | vidual nurses to apply | pect to motivate | gram. |
| | what they have learned | nurses to apply | |
| | about pain management | their knowledge | |
| | | in practice. | |
| | * and assign grades for | | |
| | applying pain manage- | | |
| | ment knowledge in prac- | | |
| | tice based on annual per- | | |
| | formance | | |
| | | | |

6.8 CONCLUSION

This chapter discussed the research methodology and designs used in Phase 5. The embedded draft action plan was validated by the panel members of clinical facilitators and professional registered nurses in three e-Delphi rounds to reach a consensus on every action statement, the method to be employed to achieve the objectives of the specific actions, the responsible person(s), and the timeframes within which the actions/objectives should be reached. The final action plan to enhance the transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals was validated.

Chapter 7 describes the study's conclusion, final recommendations, and limitations.

CHAPTER SEVEN: CONCLUSIONS, RECOMMENDATIONS, AND LIMITATIONS

7.1 INTRODUCTION

Chapter 7 presents the conclusions derived from the study findings, as well as the recommendations and limitations of the study.

Table 7.1 Organisation and structure of the study

| Organisation | and structure of the study | |
|-------------------|---------------------------------------|--|
| Chapter number | Chapter outline | Chapter content |
| Chapter 1 | Overview of the study | Contains the introduction, background of the study, the problem statement, research purpose and objectives, research question, theoretical framework, key theoretical and operational con- cepts, the research design and methodology and ethical considerations. |
| Chapter 2 | Literature review | Consists of the literature review related to: Systemic Model of Transfer of Learning by Do- novan and Darcy, Transfer of learning and Pain management and tools. |
| Chapter 3 | Research design and method- ology | Illuminates the overarching research design. Phase 1, 2 and 3 (quantitative phases): Meth- odology and Data gathering |
| Chapter 4 | Data analysis and interpreta- tion | Presents the data analysis and interpretation of the findings from Phases 1 to 3. |
| Chapter 5 | Phase 4 | Included a description of Phase 4 of the study: Literature review on action plan development |

| | | Development of the draft action plan. |
|-----------|---|--|
| Chapter 6 | Phase 5 | Outlines and describes Phase 5 of the study (qualitative phase): Methodology Validation of the action plan: The action plan. |
| Chapter 7 | Conclusion, recommenda- tions, and limitations | Conclusions, recommendations and limitations of the study. |

7.2 CONCLUSIONS

This explanatory sequential mixed-method study aimed to develop an action plan to enhance the transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals. The action plan (see Table 6.39) was validated by a panel of registered nurses and clinical facilitators identified as knowledgeable about pain management. The study was conducted over five phases:

- 1. Phase 1 aimed to identify and describe the resources available to conduct pain assessments.
- 2. Phase 2 aimed to identify and describe the nurses' characteristics and learning styles that enhance the transfer of pain management competencies.
- 3. Phase 3 explored the teaching approaches employed by the clinical facilitators during nurses' pain management education, describing the learning content regarding pain assessment and management and the climate of transfer of learning within the hospitals' nursing care areas.

- 4. Phase 4 combined the data from Phases 1, 2, and 3 and utilised a literature review to develop an action plan to enhance the transfer of learning of pain management competencies.
- 5. Phase 5 involved the validation of the developed action plan to enhance the transfer of learning of pain management competencies of nurses by a purposive selected panel of clinical facilitators and nurses for implementation using the e-Delphi technique by seeking to reach a consensus of 75% on action statements, methods, the responsible person(s) as well as the time frame included in the action plan.

To ensure that the finalisation of the action plan and its implementation achieved the study's objectives, the research supervisor ensured that it was completed under those objectives. Throughout the following sub-sections, the study's objectives are described as follows:

7.2.1 Objective 1: Identify and describe the resources available to conduct pain assessments.

The data from Phases 1 and 3 revealed the resources (pain assessment guides) available to nurses and clinical facilitators to conduct pain assessments in the wards. The available resources were in nine categories, namely:

- (1) The WILDA was most accessed by nurses as well as clinical facilitators (see Tables 4.5 and 4.17), as supported by Fink and Gallagher (2019:231), who indicate WILDA as the most commonly used **pain assessment guide tool**.
- (2) The Numeric Rating Scale (NRS) emerged as the most accessible pain rating assessment tool for patients who can self-report their pain in alignment with what is indicated in the literature (Kim & Jung 2020:2951).
- (3) Nurses reported the FLACC pain scale to be primarily available as the **pain rating assessment tool for patients who cannot self-report their pain.**
- (4) Nurse respondents agreed with the nurse facilitator respondents as they furnished similar information that the CNPI (Checklist for Non-verbal Pain Behavior)

was most available to assess the pain of elderly patients with dementia or cognitive impairment.

- (5) **Nurse respondents** indicated **that** the most **available human resources** for pain assessment were the registered nurses with pain management training in the wards. In contrast, most nurse facilitators indicated that pain nurses working in acute or chronic pain services were the most available human resource.
- (6) Other types of support assisting with pain assessments were indicated by both groups of respondents as "other patients" or "other people with pain" (see Tables 4.10 and 4.22).
- (7) Both the clinical nurse facilitators and nurses indicated that e-learning modules and organisations specialising in pain management were commonly available as pain management resources listed in the **publications and electronic resources** to conduct pain assessments.
- (8) Nurse and clinical nurse facilitators indicated that organisations that specialise in pain management were available, including research, treatment, clinical practice, and education to support pain strategies (see Tables 4.12 and 4.24).
- (9) Nurses and nurse facilitators confirmed that policies for pain management were available for utilisation (see Tables 4.12 and 4.25).

7.2.2 Objective 2: Identify and describe nurses' characteristics and learning styles that enhance the transfer of pain management competencies.

Phase 2 of the study revealed the nurses' three top-rated characteristics and learning styles that can enhance the transfer of learning of pain management competencies of nurses in the context of two Saudi Arabian teaching hospitals.

 Participants identified the characteristics as being able to apply what was learned before (see Figure 4.4) as the ability to think rationally to assess a patient experiencing pain, concentrate well during pain management learning /training, and reassess pain after interventions.

- As to describing what type of learners they were (see Figure 4.5), they indicated themselves as curious thinking, self-directed, and hardworking inquiring learners.
- Participants indicated that what motivated them to participate in the pain management training program (see Figure 4.6) was to learn pain management skills relevant to their working areas, attend pain management training programs, and gain new knowledge about pain management.
- What primarily motivated them to apply knowledge in practice (see Figure 4.7) was the desire to perform a pain management skill successfully, increase their work performance in pain management, and receive positive feedback about their performance.

In Section 4.3.3, Figure 4.8, the preferred learning styles of nurses were identified to enhance the transfer of learning of pain management competencies. The data analysis, as portrayed in Figure 4.8, uncovered nurses' five top-rated preferred learning styles that would enhance their transfer of learning as (1) watching an activity such as a demonstration, (2) reading through the information, (3) listening to information such as during a lecture, (4) solving different pain management real-life problems and (5) participating in group discussions (see Tables 5.2 and 5.3). The learning styles identified were (1) creative learners who can generate creative ideas in a group, (2) enthusiastic thinkers who listen to the information actively, take part in the activity, practice a skill and participate in the group discussions as well as (3) organised thinkers who solve different real-life problems (see Table 6.37)

7.2.3 Objective 3: Explore the teaching approaches employed by the clinical facilitators during pain management education of nurses.

The study explored the teaching approaches employed by the clinical facilitators during the pain management education of nurses. The most commonly suggested teaching strategies were pain management e-learning modules followed by an assessment of prior knowledge of pain (see Table 4.26). The last choice of teaching approach (see Table 4.26) mentioned was writing reflective journals for sharing pain management experiences. Nursing grand rounds to provide pain management training, focus groups to help nurses manage pain, and role-play activities to teach pain management were to be addressed in the action plan to facilitate implementation.

7.2.4 Objective 4: Describe the learning content regarding pain assessment and management.

Participants agreed that a short pain management training program must be contentspecific and practical. The most relevant content-specific practice-oriented information that should be included must be (1) methods to assess a patient's pain in all nursing areas promptly, (2) the selection of appropriate pain intervention strategies based on the pain levels assessed, (3) the advantages and disadvantages of all pain management scales.

7.2.5 Objective 5: Describe the transfer of learning climate within the hospitals' nursing care areas.

The elements of the transfer of learning climate for nursing care in two Saudi Arabian hospitals identified by clinical facilitators are presented in Table 4.40. They include: (1) nursing leadership values the learning needs, (2) a pain management orientation program was available, (3) clinical facilitators offer constructive feedback to registered nurses, (4) there is continuous education by the nurse managers, (5) trained nurse preceptors orientate newly employed registered nurses, (6) there is support by nurse managers, (7) there are dedicated clinical facilities for pain management available, (8) registered nurses are allowed to apply pain management freely, (9) there is peer support of registered nurses, and (10) there is support from nursing supervisors.

7.2.6 Objective 6: Develop and validat an action plan that can be implemented to enhance the transfer of learning of pain management competencies of nurses.

The validated and approved action plan to enhance the transfer of learning of pain management competencies of nurses is illustrated in Table 6.37. The specific action statements addressed are: (1) motivate nurses to further their studies, (2) make appropriate and relevant pain management tools accessible to the nursing team in all clinical areas, (3) develop a practice-oriented content-specific short pain management training program, (4) develop a pain management short program that accommodates all learning types, (5) incorporate different teaching approaches to accommodate diverse learners and facilitators in the training of pain management, (6) develop strategies to motivate nurses to participate in the short training program, (7) and motivate nurses to apply the knowledge gained in the training program in practice.

7.3 RECOMMENDATIONS

The improvement of pain management skills relies highly on the implementation of the developed action plan. The plan is suitable to be implemented in the two teaching hospitals that participated in the study due to the buy-in of the stakeholders who participated in its development. In other similar contexts, the action plan can be adopted or adapted depending on the similarity of context. The action plan must be systematically shared and fully implemented by adopting the realisation process recommended by Schaeffer, Gille and Hurrelmann (2020:12), which consists of the three steps of diffusion, dissemination, and implementation.

7.3.1 First step: Diffusion

Diverse channels must be utilized to make this action plan known and available to others (Schaeffer et al. 2020:12). The final examined thesis will be electronically published in the Unisa repository for other scholars to access. The complete research report will be electronically shared with the KAIMRC IRB for archiving and information sharing.

7.3.2 Second step: Dissemination

The action plan must be shared and thus disseminated to specific target groups who will be important for facilitating the implementation, as indicated by Schaeffer et al. (2020:12). Due to the active participation of the stakeholders within the two teaching hospitals, already taking ownership, the validated plan will be electronically shared with the nursing education centres of the two hospitals as well as pain management nursing teams within the two hospitals. An appointment will be secured with the associate nursing directors of the two hospitals and the director of nursing education in

KAMC and KASCH nursing administration conference rooms to share the action plan, explain the development thereof, and motivate implementation.

The following are recommended to motivate the implementation of the action plan. The action plan and how to facilitate implementation will be presented at the monthly leadership forums of both hospitals, as all nurses in leadership roles attend the forums. The action plan will be shared in the form of an abstract with the members of the editorial board of the two hospitals responsible for promoting awareness of hospitalwide nurses' abstracts on quality key performance indicators, as well as research projects conducted by the two hospitals to motivate and promote nurses' studies. An abstract will be submitted to national pain management conferences, such as the Saudi Pain Society, and to international pain management conferences, such as IASP, for possible inclusion in the conference program where the study findings will be presented.

The process followed to develop the action plan and the action plan itself will be published in national and international peer-reviewed journals such as the *Journal of Pain Management Nursing*.

7.3.3 The third step: collaborative Implementation

Collaborative implementation implies working with other stakeholders who can distribute the action plan and recommend implementation as suggested by Schaeffer et al. (2020:12). An appointment will be secured with the two study hospitals' nursing clinical governance and policy committees. They will be requested via e-mail to secure a meeting to share the action plan and motivate for its implementation. A presentation outlining the action plan will be delivered to the director of nursing education centres at both hospitals to facilitate its implementation.

7.3.4 Recommendations for further research

A follow-up study can be conducted to describe and assess the challenges and opportunities for implementing the action plan in all teaching hospitals. In another study, the action plan can be tested for applicability and adaptability in other contexts. Furthermore, further research is recommended to assess the effect of the implementation of the action plan on the pain management skills of nurses in specific settings.

7.4 LIMITATIONS

The study was conducted in two teaching hospitals in Riyadh, Kingdom of Saudi Arabia. It is possible that the study would have revealed different findings if other teaching hospitals were included. However, a proportionate stratum was used to ensure representativeness. However, It is possible for any hospital or entity that wants to enhance the transfer of learning of pain management skills to implement (test) the action plan and adopt or adapt it according to their unique context.

7.5 SUMMARY

The transfer of learning pain management skills remains relevant to all nurses taking care of patients in healthcare facilities. Effective pain management influences patient satisfaction and patient outcomes. Therefore, this study aimed to develop an action plan to improve the transfer of learning of pain management competencies among nurses to improve patient satisfaction and outcomes. Table 6.37 illustrates the action plan developed and validated in the context of two Saudi Arabian teaching hospitals, which was presented and recommended for implementation.

"Knowing Is not enough; we must apply. Wishing is not enough; we must do." (Quote from Johann Wolfgang von Goethe).

REFERENCES

Books:

- Abrahams, H. 2010. *Rebuilding lives after domestic violence: understanding long-term outcomes*. Philadelphia: Jessica Kingsley Publishers.
- Abrahamson, E & Langston, J. 2017. *Making sense of human anatomy and physiology: a learner-friendly approach.* Berkley: North Atlantic Books.
- Achinstein, B & Ogawa, RT. 2015. *Changed(d) agents: new teachers of color in urban schools.* New York: Teachers College Press Learning.
- Ackley, BJ, Ladwig, GB & Makic, MBF. 2017. *Nursing diagnosis handbook e-book: an evidence-based guide to planning care*. 11th edition. St Louis: Elsevier Health Sciences.
- Ackley, BJ, Ladwig, GB. 2014. *Nursing diagnosis handbook e-book: an evidence-based guide to planning care*. 10th edition. St Louis: Elsevier Health Sciences.
- Acton, QA. 2013. *Pain: new insights for the healthcare professional:* 2013 edition. Atlanta: Scholarly Editions.
- Alexander, M, Corrigan, A, Gorski, L, Hankins, J & Perucca, R. 2010. Infusion nursing: an evidence-based approach. 3rd edition. St. Louis: Saunders Elsevier.
- Allen, EK & Cowdery, GE. 2015. *The exceptional child: inclusion in early childhood education*. 8th edition. Stamford: Cengage Learning.
- Alvord, B. 2010. Advanced instructional design: designing effective training. Wyomissing. ALERA Publishing Group.
- American Academy of Orthopaedic Surgeons. 2021. AAOS essentials of musculoskeletal care. Burlington: Jones & Bartlett.

- Amichai-Hamburger, Y. 2013. *The social net: understanding our online behaviour*. 2nd edition. Oxford: Oxford University Press.
- Andonov, S. 2021. Safety in risky industries: black swans, gray rhinos and other adverse events. 1st edition. Boca Raton: CRC Press.
- Andrade, MS & Evans, NW. 2013. *Principles and practices for responses in second language writing: developing self-regulated learners*. New York: Routledge.
- Arai, K. ed. 2022. Advances in Information and Communication: Proceedings of the 2022 Future of Information and Communication Conference (FICC), Volume 2 (Vol. 439). Cham: Springer Nature.
- Ares, G & Varela, P. 2018. *Methods in consumer research, volume 1: new approaches to classic methods*. Cambridge: Woodhead Publishing.
- Arnstein, P. 2010. *Clinical coach for effective pain management*. Philadelphia: F.A. Davies.
- Arnstein, P. 2011. *Clinical coach for effective pain management.* Philadelphia: Davis Company.
- Ashcroft, W & Delloso, AM. 2014. *Autism: classroom success with ABA*. New York: National Professional Resources.
- Atkinson, SJW, Armbruster, CK & Evans, EM. 2021. *Fitness and wellness in Canada: a way of life*. Champaign: Human Kinetics.
- Attri, RK. 2023. The Craft of Business Research: How Practitioners Conduct High Value Corporate and Doctoral Studies. Speed To Proficiency Research: S2Pro©.
- Auzer, KA. 2017. Institutional design and capacity to enhance effective governance of oil and gas wealth: the case of Kurdistan Region. Gateway East: Springer Nature.

- Avis, J. 2016. Social justice, transformation and knowledge, policy, workplace learning and skills. New York: Routledge.
- Babu, S & Gayathri, T. 2018. *Psychology for nurses*. 2nd edition. St. Louis: Elsevier Health Sciences.
- Babbie, ER. 2020. The practice of social research. Boston:Cengage AU.
- Baird, MS. 2016. *Manual of critical care nursing: nursing interventions and collaborative management.* 7thedition. St Louis: Elsevier Health Sciences.
- Baldwin, L. 2018. Research concepts for the practitioner of educational leadership.In *Research Concepts for the Practitioner of Educational Leadership*. Boston:Brill Sense.
- Ballweg, R, Brown, D, Vetrosky, DT & Ritsema, TS. 2018. *Physician assistant: a guide to clinical practice e-book.* 6th edition. Philadelphia: Elsevier Health Sciences.
- Baran, RJ & Galka, RJ. 2016. Customer Relationship Management: the foundation of contemporary marketing strategy. Taylor & Francis.
- Bastable, SB. 2014. *Nurse as educator: principles of teaching and learning for nursing practice*. 4th edition. New York: Jones & Bartlett Learning.
- Bazeley, P. 2013. Qualitative data analysis: practical strategies. Thousand Oaks: SAGE.
- Bazeley, P. 2018. Integrating analyses in mixed methods research. Thousand Oaks: SAGE.
- Beevi, A. 2012. *Pediatric nursing care plans*. New Delhi: Jaypee Brothers Medical Publishers.
- Behrens, BJ & Beinert, H. 2014. *Physical agents' theory and practice*. 3rd edition. Philadelphia: FA Davies.
- Beins, BC. 2017. Research method. A tool for life. 3rd edition. Clays: Cambridge University Press.

Bennett, M. 2010. *Neuropathic pain*. 2nd edition. Oxford: Oxford University Press.

- Benson, H, Raja, SN, Fishman, SE, Liu, SS & Cohen, SP. 2011. *Essentials of pain medicine e-book*. 3rd edition. Philadelphia: Elsevier Health Sciences.
- Benson, H, Raja, SN, Fishman, SM, Liu, SS & Cohen, SP. 2017. *Essentials of pain medicine e-book*. 4th edition. St. Louis: Elsevier Health Sciences.
- Benuto, LT & Leany, BD. 2015. *Guide to psychological assessment with African Americans.* New York: Springer Science & Business Media.
- Bergin, T. 2018. An introduction to data analysis: quantitative, qualitative and mixed methods. Thousand Oaks: SAGE.
- Billings, DM & Halstead, JA. 2019. *Teaching in nursing: a guide for faculty*. 6th edition.
 Missouri: Elsevier Health Science.
- Black, B. 2017. *Professional nursing: concepts & challenges*. 8th edition. St Louis: Elsevier Health Sciences.
- BLAGA, S. 2020. Social Entrepreneurship Motivation. Editura Universității din București-Bucharest University Press.
- Blumberg, FC. 2014. *Learning by playing video gaming in education*. Oxford: Oxford University Press.
- Boltz, M. 2012. *Evidence-based geriatric nursing protocols for best practice*. 4th edition. New York: Springer Publishing.
- Bone, A & Maharg, P. 2019. Critical perspectives on scholarship of assessment and learning in law: volume 1: England volume 1 of assessment in legal education. Acton: ANU Press.
- Botma, Y, Greeff, M, Mulaudzi, FM & Wright, SCD. 2010. *Research in health sciences*. Cape Town: Clyson Printers.

Boxwell, G. 2010. *Neonatal intensive care nursing*. 2nd edition. New York: Routledge.

- Bradley, C. 2013. Handbook of psychology and diabetes: a guide to psychological measurement in diabetes research and practice. London: Routledge.
- Bradley, N. 2013. *Marketing research: tools and techniques*. Oxford: Oxford University Press.
- Brown, D, Edwards, H & Seaton, L. 2017. *Lewis' medical-surgical nursing: assessment and management of clinical problems.* 4thedition. St Louis: Elsevier Health Sciences.
- Brown-Chidsey, R & Bickford, R. 2015. Practical handbook of multi-tiered systems of support: building academic and behavioural success in schools. New York: Guilford Publications.
- Bruckenthal, P & Quinlan-Colwell, A. 2012. Compact clinical guide to geriatric pain management: an evidence-based approach for nurses. New York: Springer Publishing Company.
- Bruera, E, Higginson, T, Von Gunten, CF & Morita, T. 2016. *Textbook of palliative medicine and supportive care.* 2nd edition. Boca Raton: CRC Press.
- Bruera, ED & Portenoy, RK. 2010. *Cancer pain: assessment and management*. New York: Cambridge University Press.
- Bryant, A & Charmaz, K. eds. 2019. *The SAGE handbook of current developments in grounded theory*. Sage.
- Bryant, R & Nix, D. 2016. *Acute & chronic wounds: current management concepts*. 5th edition. St Louis: Elsevier Health Sciences.
- Buckenmaier, CC, Kent, M, Mariano, ER & Brookman, JC. 2019. *Acute pain medicine.* Oxford: Oxford University Press.
- Buckley, R & Caple, J. 2009. *The theory and practice of training*. 6th edition. Philadelphia: Kogan Page Publishers.

- Buonocore, G & Bellieni, CV. 2017. *Neonatal pain: suffering, pain, and risk of brain damage in the fetus and newborn*. 2nd edition. Manhattan: Springer Cham International Publishing AG.
- Burns, D. 2015. Foundations of adult nursing. Thousand Oaks: SAGE.
- Burns, RB & Dobson, CB. 2012. *Introductory psychology*. Berlin: Springer Science & Business Media.
- Cameron, P, Browne, GJ, Mitra, B, Dalziel, S & Craig, S. 2018. *Text of paediatric emergency medicines*. 3rd edition. St. Louis: Elsevier Health Sciences.
- Cardenas, D.D. and Hooton, T.M. 2015. *Medical complications in physical medicine and rehabilitation*. New York: Demos Medical Publishing.
- Carliner, S. 2015. *Training design basics*. 2nd edition. Alexandria: Association for Talent Development.
- Carter, B & Simons, J. 2014. *Stories of children's pain: linking evidence to practice*. Thousand Oaks: SAGE.dare.
- Carter, R, Lubinsky, J & Domholdt, E. 2013. *Rehabilitation research principles and applications*. 4th edition. St. Louis: Elsevier Saunders.
- Carver, L & Atkins, HS. 2021. *Preparing to lead in a digital environment: what all educators need to know*. London: Rowman & Littlefield.
- Cash, JC, Glass, CA, Fraser, D, Corcoran, L & Edwards, M. 2019. *Canadian family practice guidelines*. New York: Springer Publishing Company.
- Chang, E & Johnson, A. 2018. *Living with chronic illness and disability: principles for nursing practice*. 3rd edition. Chatswood: Elsevier Health Sciences.
- Chang, E & Johnson, A. 2021. *Living with chronic illness and disability: principles for nursing practice*. 4th edition. Chatswood: Elsevier Health Sciences.
- Cheatle, MD & Fine, PG. 2017. *Facilitating treatment adherence in pain medicine.* Oxford: Oxford University Press.

- Chen, F & Terken, J. 2022. *Automotive Interaction Design: From Theory to Practice*. Beijing: Springer Nature.
- Chila, AG. 2010. Foundation of osteopathic medicine. Philadelphia: Lippincott Williams & Wilkins.
- Chisholms-Burns, MA, Schwihammer, TL, Malone, PM, Kolesar, JM, Lee, KC & Bookstave, PB. 2019. *Pharmacotherapy principles and practice,* 5th edition. Blacklick: McGraw Hill Professional.
- Cianfrini, LR, Doleys, DM & Richardson, EJ. 2021. Pain psychology for clinicians: a practical guide for the non-psychologist managing patients with chronic pain. New York: Oxford University Press.
- Ciocco, M. 2020. Fast facts for the nurse preceptor, second edition: keys to providing a successful preceptorship. New York: Springer Publishing Company.
- Cohen, L, Manion, L & Morrison, K. 2011. *Research methods in education*. 7th edition. New York: Routledge.
- Collins, D. 2015. Nursing care plans for home health care. Bloomington: LTCS Books.
- Cooper, K & Gosnell, K. 2015. *Foundations and adult health nursing*. 7th edition. St. Louis: Elsevier Health Sciences.
- Coté, CJ, Lerman, J & Anderson, BJ. 2013. *A practice of anaesthesia for infants and children: expert consult online and print.* Philadelphia: Elsevier Health Sciences.
- Cox, E. 2013. Coaching understood: a pragmatic inquiry into the coaching process. Thousand Oaks: SAGE.
- Craft, J, Gordon, C, Huether, SE, McCane, KL & Brashers, VL. 2018. *Understanding pathophysiology*. 3rd edition. Philadelphia: Elsevier Health Sciences.
- Cranton, P. 2016. Understanding and promoting transformative learning: a guide to theory and practice. 3rd edition. Sterling: Stylus Publishing.

- Creswell, JW & Clark, VLP. 2011. *Designing and conducting mixed methods research.* 2nd edition. Thousand Oaks: SAGE.
- Creswell, JW. 2015. A concise introduction to mixed methods research. Thousand Oaks: SAGE.
- CTET (Central Teacher Eligibility Test) 2020. Social science/studies paper-2 for class 6 to 8 2020. New Delhi: Arihant Publications India Limited.
- Cunningham, I & Bennett, B. 2017. Self-managed learning in action: putting SML into practice. London: Taylor & Francis.
- Curry, L & Nunez-Smith, M. 2015. *Mixed methods in health sciences research*. A practical primer. Thousand Oaks: SAGE.
- Czaja, SJ & Sharit, J. 2016. *Designing training and instructional programs for older adults*. New York: CRS Press.
- Czarnecki, ML & Turner, HN. 2018. Core curriculum for pain management nursing ebook. 3rd edition. St Louis: Elsevier Health Sciences.
- Cypress, BS. 2021. Fundamentals of qualitative phenomenological nursing research. John Wiley & Sons.
- D'Arcy, Y. 2011. Compact clinical guide to acute pain management. An evidencebased approach for nurses. New York: Springer Publishing Company.
- D'Arcy, Y. 2013. Compact clinical guide to critical care, trauma, and emergency pain management: an evidence-based approach for nurses. New York: Springer Publishing Company.
- Daffron, SR & North, MW. 2011. *Successful transfer of learning*. Malabar: Krieger Publishing Company.
- Dagenias, S & Haldeman, S. 2012. *Evidence-based management of low back pain ebook*. St Louis: Elsevier Health Sciences.

- Dains, JE, Baumann, LC & Scheibel, P. 2015. Advanced health assessment & clinical diagnosis in primary care e-book. 5th edition. St Louis: Elsevier Health Sciences.
- Davies, A. 2012. *Cancer-related breakthrough pain*. 2nd edition. Oxford: Oxford University Press.
- Davies, PS & D'Arcy, YM. 2013. Compact clinical guide to cancer pain management: an evidenced-based approach for nurses. New York: Springer Publishing Company.
- Davis, PJ, Cladis, FP & Motoyama, EK. 2011. *Smith's anaesthesia for infants and children e-book: expert consult premium.* 8th edition. Philadelphia: Elsevier Health Sciences.
- De Vet, HCW, Terwee, CB, Monkkink, LB & Knol, DL. 2011. *Measurement in medicine: a practical guide.* Cambridge: Cambridge University Press.
- DeWit, SC & Kumagai, CK. 2013. *Medical-surgical nursing: concepts and practice*. 2nd edition. St Louis: Elsevier Health Sciences.
- DeWit, SC & O'Neil, PA. 2014. *Fundamental concepts and skills for nursing*. 4thedition. St. Louis: Elsevier Health Sciences.
- DeWit, SC & Williams, PA. 2013. *Fundamental concepts and skills for nursing. e-book.* 4th edition. St Louis: Elsevier Health Sciences.
- DeCuir-Gunby, JT & Schutz, PA. 2017. *Developing a mixed methods proposal. A practical guide for beginning researchers.* Thousand Oaks: SAGE.
- Deer, TR, Pope, JE, Lamer, TJ, Provenzano, D. 2019. *Deer's treatment of pain: an illustrated guide for practitioners*. Sewickley: Springer.
- DePoy, E & Gitlin, LN. 2011. Introduction to research understanding and applying multiple strategies. 4th edition. St Louis: Elsevier Health Sciences.

- Dhinu, MR. 2021. *Applied Statistics in Physical Education and Sports*. New Dehli: Friends Publications (India).
- Dillman, DA, Smyth, JD & Christian, LM. 2014. *Internet, phone, mail, and mixed-mode surveys. The tailored design method.* New Jersey: John Wiley & Sons, Inc.
- DiPaola, M & Hoy, WK. 2013. Principals improving instruction: supervision, evaluation, and professional development. Charlotte: Information Age Publishing, Inc.
- Disha Experts. 2021. NTA UGC NET Paper 1 Topic-Wise 50 solved papers (2019 to 2004). New Delhi: Disha Publication Inc.
- Dolan, B & Holt, L. 2013. Accident & emergency theory into practice. 3rd edition. Philadelphia: Bailliere Tindall Elsevier.
- Donovan, P & Townsend, J. 2011. *Transfer of learning pocketbook*. Hants: Management Pocketbooks Ltd.
- Dougherty, L. and Lister, S. (eds.) 2015. *The Royal Marsden manual of clinical nursing procedures*. John Wiley & Sons.
- Duhme, L. 2012. Cultural tourism: case study Portugal. Hamburg: Diplomica Verlag.
- Dwyer, J & Hopwood, N. 2019. *The business communication handbook*. Victoria: Cengage AU.
- Dziedzic, K & Hammond, A. 2010. *Rheumatology: evidenced-based practice for physiotherapists and occupational therapists.* Churchill Livingstone.
- Ebert, MH & Kems, RD. 2011. *Behavioural and psychopharmacologic pain management*. Cambridge: Cambridge University Press.
- Edwards, SL & Coyne, I. 2013. A survival guide to children's nursing. St Louis: Elsevier Health Sciences.
- Elsevier, Inc. 2014. Job readiness for health professionals e-book: soft skills strategies for success. St Louis: Elsevier Health Sciences.

- Emergency Nurses Association (ENA). 2013. *Sheehy's manual of emergency care*. 7th edition. St Louis: Elsevier Health Sciences.
- Engel, RJ & Schutt, RK. 2014. *Fundamentals of social work research*. Thousand Oaks: SAGE.
- Erford, B. 2013. Assessment for counselors. 2nd edition. Belmont: Cengage Learning.
- Fairchild, SL, O'Shea, R & Washington, R. 2017. Pierson and Fairchild's principles & techniques of patient care e-book. 6th edition. St Louis. Elsevier Health Sciences.
- Fanaroff, AA. 2011. Yearbook of neonatal and perinatal medicine 2011. Philadelphia: Elsevier Health Sciences.
- Fell, DW, Lunnen, KY & Rauk, RP. 2018. Lifespan neurorehabilitation: a patient-centered approach from examination to intervention and outcomes. Philadelphia: A Davies.
- Ferrell, BR, Coyle, N & Paice, JA. 2015. *Oxford textbook of palliative nursing*. Oxford: Oxford University Press.
- Findlay, JW. 2015. *Play intelligence: from IQ to PIQ*. New York: University Press of America.
- Fishman, S, Ballantyne, J & Rathmell, JP. 2010. *Bonica's management of pain*. Philadelphia: Lippincott Williams & Wilkins.
- Fishman, SM. 2012. *Bonica's management of pain*. 4th edition. Philadelphia: Lippincott Williams & Wilkins.
- Fitzpatrick, J. 2017. *Encyclopedia of nursing research*. 4th edition. New York: Springer Publishing Company.
- Fleisher, LA & Rosenbaum, SH. 2017. *Complications in anesthesia.* 3rd edition. Philadelphia: Elsevier Health Sciences. Forbes, H & Watt, E. 2016. *Jarvis' physical*

examination and health assessment. 2nd edition. Chatswood: Elsevier Health Sciences.

- Flick, U. 2022. The SAGE handbook of qualitative research design. Thousand Oaks: SAGE
- Flynn, SV. ed. 2022. Research design for the behavioral sciences: An applied approach. Springer Publishing Company.
- Forbes, H & Watt, E. 2016. Jarvis's health assessment and physical examination. 2nd edition. Chatswood: Elsevier Health Science.
- Fortune, AE, Reid, WJ & Miller, RL. 2013. *Qualitative research in social work*. 2nd edition. New York: Columbia University Press.
- Frasard, S & Prasuhn, FC. 2016. *Training initiative and strategies for the modern workforce.* Hershey: IGI Global.
- French, LA & Nikolic-Novakovic, L. 2012. *War trauma and its aftermath: an international perspective on the Balkan and Gulf Wars*. Lanham: University Press of America.
- Fritz, S & Chaitow, L. 2011. *The Massage Therapist's guide to pain management ebook*. Elsevier Health Sciences.
- Fundakowski, CE. 2020. *Head and neck cancer: psychological and psychosocial effects*. Philadelphia: Springer Nature.
- Gaberson, K, Oermann, MH, Shellenbarger, T. 2015. *Clinical teaching strategies in nursing*. 4th edition. New Jersey: Springer Publishing Company.
- Gandevani, N. 2017. *How to become a successful trader: the trading personality profile: your key to maximizing profit with any system.* Morrisville: Lulu.com.
- Gardener, SL, Carter, BS, Enzman-Hines, MI & Hernandez, JA. 2011. *Merenstein and Gardener's handbook of neonatal intensive care e-book*. 7th edition St Louis: Elsevier Health Sciences.

- Gawlik, KS, Melnyk, BM & Teall, AM. 2020. *Evidence-based physical examination: best practice for health & well-being assessment*. New Jersey: Springer Publishing Company.
- Gavin Ware, D & Johnson, T. eds. 2018. The Mental Health Effects of Informal Caregiving: Emerging Research and Opportunities: Emerging Research and Opportunities.Hershey: IGI Global
- Gerrish, K & Lathlean, J. 2015. *The research process in nursing*. 7th edition. West Sussex: John Wiley.
- Gitlin, L & Czaja, S. 2016. *Behavioural intervention research designing, evaluating, and implementing.* New York: Springer Publishing Company.
- Gleason, CA & Devaskar, MD. 2012. *Avery's disease of the newborn*. 9th edition. Philadelphia: Elsevier Saunders.
- Gliva-McConvey, G, Nicholas, GF & Clark, L. 2020. Comprehensive healthcare simulation: implementing best practices in standardized patient methodology. Cham: Springer.
- Gómez-Galán, J. 2016. *Educational research in higher education: methods and experiences.* Delft: River Publishers.
- Gooden, CK, Lowrie, L & Jackson, BF. 2018. *The pediatric procedural sedation handbook*. New York: Oxford University Press.
- Gravetter, FJ & Forzano, LB. 2018. *Research methods for the behavioural sciences*. 6th edition. Boston: Cengage Learning.
- Gray, J & Grove, SK. 2021. The practice of nursing research appraisal, synthesis and generation of evidence. 9th. edition. St Louis: Elsevier.
- Griffin, R. 2014. Complete training evaluation: *the comprehensive guide to measuring return on investment.* Philadelphia: Kogan Page Publishers.

Griffin, RW. 2016. *Management*. 12th edition. Boston: Cengage Learning.

Grima, S, Sood, K. & Özen, E. eds. 2023. *Contemporary Studies of Risks in Emerging Technology, Part A.* Bingley:Emerald Publishing Limited.

- Grisold, W, Struhal, W & Grisold, T. 2019. *Advocacy in neurology*. Oxford: Oxford University Press.
- Grove, SK, Burns, N & Gray, J. 2013. The practice of nursing research: appraisal, synthesis, and generation of evidence. 7th edition. St Louis: Elsevier Health Sciences.
- Grove, SK. Gray, J.R & Faan, PR. 2019. Understanding nursing research: First South Asia edition, e-book: Building an evidence-based practice. St. Louis: Elsevier India.
- Grove, SK, Burns, N & Gray, J. 2015. *Understanding nursing research: building an evidence-based practice.* St Louis: Elsevier Health Sciences.
- Grove, SK. & Gray, JR. 2023. Understanding nursing research e-book: Building an evidence-based practice. 8th edition. St Louis: Elsevier Health Sciences
- Grove, SK. & Gray, JR. 2018. Understanding nursing research e-book: Building an evidence-based practice. St Louis: Elsevier Health Sciences.
- Grundström, H. 2018. *Disclosing the invisible: experiences, outcomes and quality of endometriosis healthcare* (Vol. 1621). Linköping: Linköping University Electronic Press.
- Guccione, AA, Avers, D, Wong, R. 2012. *Geriatrics physical therapy e-book*. 3rd edition. St Louis: Elsevier Health Sciences.
- Guralnick, D, Auer, ME & Poce, A. eds. 2022. Innovative Approaches to Technology-Enhanced Learning for the workplace and Higher education: Proceedings of 'The Learning Ideas Conference'2022 (Vol. 581). Paris:Springer Nature.
- Hale, KS & Stanney, KM. 2015. *Handbook of vertical environments design, implementation, and applications*. 2nd edition. Boca Raton: CRC Press.

- Hall, RP. 2020. *Mixing methods in social research: qualitative, quantitative and combined methods*.Thousand Oaks:SAGE
- Hall, HR & Roussel, L. 2014. *Evidence-based practice: an integrative approach to research, administration, and practice.* Burlington: Jones & Bartlett Publishers.
- Hardin, SR & Kaplow, R. 2016. *Cardiac surgery essential for critical care nursing*. Burlington: Jones & Bartlet Learning.
- Harris, MJ & Muvuka, B. 2023. Integrated Research Methods in Public Health. Hoboken: John Wiley & Sons.
- Hatzipanagos, S & Rochon, R. 2012. Approaches to assessment that enhance learning in higher education. New York: Routledge.
- Haugen, N & Galura, SJ. 2011. *Ulrich & Canale's nursing care planning guides e-book.* 7th edition. Philadelphia: Elsevier Health Sciences.
- Herrmann, JW. 2020. *Creative teaching strategies for the nurse educator*. 3rd edition. Philadelphia: F.A. Davis Company.
- Hochberg, MC, Silman, AJ, Smolen, JS, Weinblatt, ME & Weisman, MH. 2015. *Rheumatology*. 6th edition. Volume 1. Philadelphia: Elsevier Health Sciences.
- Hockenberry, MJ & Wilson, D. 2018. *Wong's nursing care of infants and children. Ebook*. Amsterdam: Elsevier Health Sciences.
- Hogan, TP. 2019. *Psychological testing: A practical introduction*. Hoboken: John Wiley & Sons.
- Holton, JA & Walsh, I. 2017. Classic grounded theory application with qualitative and quantitative data. Thousand Oaks: SAGE.
- Houser, J. 2021. Nursing Research: Reading, Using and Creating Evidence: Reading, Using and Creating Evidence. Jones & Bartlett Learning.
- Hoyle, R. 2015. Informed learning in organizations: how to create a continuous learning culture. Philadelphia: Kogan Page Limited.

- Huber, D & Joseph, L. 2021. *Leadership and nursing care management e-book.* St. Louis. Elsevier Health Sciences.
- Hughes, C. (ed.) 2019. Strategies for attracting, maintaining, and balancing a mature workforce. IGI Global.
- ICN (International Council of Nurses). 2019. International council of nurses calls for ethical recruitment process to address critical shortage of nurses. Geneva: ICN.
- Ignatavicius, DD & Workman, ML. 2016. *Medical-surgical nursing: patient-centred collaborative care.* 8th edition. St Louis: Elsevier Health Sciences.
- Information Resources Management Association ed. 2021. Research Anthology on Developing Effective Online Learning Courses. Hershey: IGI Global.
- Inglis, D & Kenneally, J. 2020. *Clinical skills for paramedic practice ANZ 1e*. St Louis: Elsevier Health Sciences.
- Institute of Medicine. 2011. *The future of nursing leading change, advancing health.* Washington: National Academies Press.
- Iphofen, R. 2017. *Finding common ground, consensus in research ethics across social sciences: advances in research ethics and integrity*. Bingley: Emerald Group Publishing.
- James, SR, Nelson, K & Ashwill, J. 2013. *Nursing care of children: principles and practice*. 4th edition. St Louis: Elsevier Health Sciences.
- Janet, H. 2016. Wearable technology and mobile innovations for next-generation education. Hershey: IGI Global.
- Jankovic, D & Peng, P. 2015. *Regional nerve blocks in anesthesia and pain therapy: traditional and ultrasound guided techniques*. 4th edition. New York: Springer International Publishing.
- Jha, A. 2023. Social Research Methodology: Qualitative and Quantitative Designs. Taylor & Francis.

Jha, AS. 2014. Social research methods. New Delhi: McGraw Hill Education.

- Johnson, RB & Christensen, L. 2019. *Educational research: Quantitative, qualitative, and mixed approaches*. Thousand and Oaks: SAGE publications.
- Johnson, RB & Christensen, L. 2012. *Educational research: quantitative, qualitative, and mixed approaches.* 4th edition. Thousand Oaks: SAGE.
- Johnson, JA & Davey, KS. 2019. *Essentials of managing public health organizations.* Burlington: Jones & Bartlett Learning.
- Kaiser, MR, Kaminski, K & Foley, JM. 2013. *Learning transfer in adult education*. New Jersey: Wiley Periodicals, Inc.
- Kaplan, JS & McCune, SL. 2018. *Cliffs Notes FTCE general knowledge test.* 4th edition. New York: Houghton Mifflin Harcourt Publishing Company.
- Kakshooy, AM & Chiappelli, F. 2018. *Practical biostatistics in translational healthcare*.Los Angeles:Springer.
- Karishree, R, Sudha, K, Soundarya, MS & Ramya, M.2023. Data analytics and modelling techniques. Puducherry: Quing Publications.
- Kawachi, I, Lang, I & Ricciardi, W. eds. 2020. Oxford handbook of public health practice 4e. Oxford University Press.
- Kazdin, AE. 2021. *Research design in clinical psychology*. Cambridge: Cambridge University Press.
- Kearney-Nunnery, R. 2016. Advancing your career concepts in professional nursing. 6th edition. Philadelphia: FA Davis.
- Keeney, S, Hasson, F & McKenna, H. 2011. *The Delphi technique in nursing and health research.* New Delhi: Wiley-Blackwell.
- Keogh, J. 2012. Schaums's outline of medical charting: 300 review questions and answers. New York: McGraw Hill.

- Khosrow-Pour, M., Clarke, S., Jennex, M.E., Anttiroiko, A.V., Kamel, S., Lee, I.,
 Kisielnicki, J., Gupta, A., Van Slyke, C., Wang, J. and Weerakkody, V., 2022.
 Research anthology on innovative research methodologies and utilization across multiple disciplines.
- Kingma, M. 2018. *Nurses on the move: Migration and the global health care economy*. New York: Cornell University Press.
- Kirwan, C. 2016. Improving learning transfer: a guide to getting more out of what you put into your training: New York: CRC Press.
- Kirwan, C. 2016. Making sense of organizational learning: putting theory into practice. New York: Routledge.
- Kline, RB. 2023. *Principles and practice of structural equation modeling*. 5th edition.New York: Guilford publications.
- Kolb, DA. 2015. *Experiential learning: experience as the source of learning and development*. 2nd edition. Harlow: Pearson Education, Inc.
- Krau, SD & Overstreet, M. 2017. Pain management, an issue of critical nursing clinics, e-book volume 29, issue 4 of the clinics: nursing. St Louis: Elsevier Health Sciences.
- Kronenberge, J & Ledbetter. 2020. *Comprehensive medical assisting*. 5th edition. Burlington: Jones & Bartlett Learning.
- Kumar, R. 2019. *Nursing research & statistics*. Panama City: Jaypee Brothers Medical Publishers.
- Landy, FJ & Conte, JM. 2010. Work in the 21st century: an introduction to industrial and organizational psychology. 3rd edition. California: John Wiley & Sons Inc.
- Landy, FJ & Conte, JM. 2016. Work in the 21st century, binder ready version: an introduction to industrial and organizational psychology. 5th edition. New Jersey: John Wiley & Sons Inc.
- Leavy, P. 2022. Research design: Quantitative, qualitative, mixed methods, artsbased, and community-based participatory research approaches. Guilford Publications.
- Leavy, P. 2017. Research design: quantitative, qualitative, mixed methods, artsbased, and community-based participatory research approaches. New York: The Guilford Press.
- Leberman, S & McDonald, L. 2016. *The transfer of learning: participants' perspectives of adult education and training*. New York: CRC Press.
- Leedy, PD & Ormrod, JE. 2015. *Practical research planning and design*. 11th edition. Harlow: Pearson Education Limited.
- Leifer, G. 2011. *Introduction to maternity and pediatric nursing*. 6th edition. St Louis: Elsevier Health Sciences.
- Leslie, RA, Johnson, EK, Thomas, G & Goodwin, APL. 2011. *Dr Podcast scripts for the final FRCA*. Cambridge: Cambridge University Press.
- Lewis, SL, Bucher, L, Heitkemper, MM, Harding, MM, Kwong, J & Roberts, D. 2017. Medical-surgical nursing e-book: assessment and management of clinical problems, single volume. 10th edition. St. Louis: Elsevier Health Sciences.
- Liamputtong, P. 2013. *Qualitative research methods*. 4th edition. South Melbourne, Victoria: Oxford University Press.
- Lim, P & Parker, A. 2020. *Mentoring millennials in an Asian context: talent management insights from Singapore*. Bingley: Emerald Group Publishing.
- Lin, G & Qu, M. 2016. Smart use of state public health data for health data for health disparity assessment. Boca Raton: CRS Press.
- LoBiondo-Wood, G & Haber, J. 2010. *Nursing research methods and critical appraisal for evidence-based practice*. 7th edition. St. Louis: Mosby Elsevier.

Lohr, SL. 2021. Sampling: design and analysis. 2nd edition . Chapman and Hall/CRC.

- Long, CO. 2013. Pain management education in long-term care: it can make a difference. *Pain Management Nursing*. Vol 14 (4). 220-227.
- Lopez, AL. 2010. Flying tigers aim high and think big. Bloomington: Xlibris Corporation.
- Lussier, RN & Achua, CF. 2016. *Leadership: Theory, application & skill development*. 6th edition. Boston: Cengage Learning.
- Lynch, SM. 2013. Using statistics in social research: a concise approach. New Jersey: Springer Publishing Company.
- Macintyre, PE & Schug, SA. 2015. *Acute pain management: a practical guide*. 4th edition. Boca Raton: Taylor & Francis Group, LLC.
- Mackintosh, N & Mackintosh, NJ. 2011. *IQ and human intelligence*. Oxford University Press.
- Malik, T. 2020. *Practical chronic pain management: a case-based approach*. Cham: Springer Nature.
- Management Association, Information Resources. 2018. *Global business expansion: concepts, methodologies, tools and applications*. Hershey: IGI Global.
- Management Association, Information Resources. 2019. Pre-service and in-service teacher education: concepts, methodologies, tools, and applications, tools, and applications. Hershey: GI Global.
- Mangal, SK & Mangal, S. 2013. *Research methodology in behavioural sciences*. Delhi: PH Learning Pvt. Ltd.
- Manichander, T. 2016. Personality. Lulu. com.
- Marchiori, M. 2018, July. ECRM 2018 *17th European Conference on Research Meth*ods in Business and Management. Academic Conferences and publishing limited.
- Marko, T & Dickerson, M. 2017. *Clinical handbook of neonatal pain management for nurses*. New York: Springer Publishing Company.

- Marmo, L & D'Arcy, Y. 2013. Compact clinical guide to critical care, trauma, and emergency pain management: an evidence-based approach for nurses. New York: Demos Medical Publishing.
- Marx, J, Walls, R & Hockerberger, R. 2013. *Rosen's emergency medicine- concepts and clinical practice e-book*. 8th edition. St Louis: Elsevier Health Sciences.
- Mason, KP. 2015. *Pediatric sedation outside of the operating room: a multispecialty international collaboration*. 2nd edition. New York: Springer Science & Business Media.
- Matzo, M & Sherman, DW. 2015. *Palliative care nursing, quality care of end of life.* 4th edition. New York: Springer.
- Maxfield, MG & Babbie, ER. 2018. *Research methods for criminal justice and criminology*. 8th edition. Boston: Cengage Learning.
- McClain, BC & Suresh, S. 2011. Handbook of pediatric chronic pain: current science and integrative practice. New York: Springer Science & Business.
- McCune, SL & Alexander, VC. 2020. *CliffsNotes TExEs PPR EC-12(160).* Boston: Houghton Mifflin Harcourt Publishing Company.
- McGurk, SR & Mueser, KT. 2021. Cognitive remediation for successful employment and psychiatric recovery: The Thinking Skills for Work program. Guilford Publications.
- McLeary, L & McParland, T. 2021. Ross-Kerr and Wood's Canadian nursing issues & perspectives e-book. Toronto: Elsevier Health Sciences.
- Mellenbergh, GJ. 2019. *Counteracting methodological errors in behavioral research*. Paris: Springer International Publishing.
- Mertens, DM. 2015. Research and evaluation in education and psychology: integrating diversity with quantitative, qualitative, and mixed methods. Thousand Oaks: SAGE.

- Messinger, AM. & Guadalupe-Diaz, X.L. eds. 2020. *Transgender intimate partner violence: A comprehensive introduction*. NYU Press.
- Miller, CA. 2012. Fast facts for dementia care: what nurses need to know in a nutshell. New York: Springer Publishing Company.
- Minton, CAB & Lenz, AS. 2019. Practical approaches to applied research and program evaluation for helping professionals. London:Routledge.
- Mishra, BK. 2016. *Psychology: the study of human behaviour*. 2nd edition. Delhi: PHI Learning Private Ltd.
- Mishra, SP & Ghosh, B. 2021. An introduction to sports management and curriculum design in physical education. New Delhi: Friends Publications.
- Moore, RJ. 2013. Handbook of pain and palliative care: biobehavioral approaches for the life. New York: Springer Science & Business Media.
- *Mosby's Dictionary of Medical, Nursing & Health Professionals.* 2013. 9th edition. St Louis: Mosby
- Mukul, G & Deepa, G. 2011. *Research methodology*. New Delhi: PHI Learning Private Limited.
- Munkvold, R & Kolãs, L. 2015. ECGBL 2015 9th European conference on gamesbased learning: ECGBL 2015. Reading: Academic Conferences and Publishing International Limited.
- Nagelhout, JJ & Plaus, K. 2010. Nurse anaesthesia. 4th edition. St Louis: Saunders Elsevier.
- National Association of EMS of educators. 2020. *Foundations of education: an EMS approach*. 3rd edition. Pittsburgh. Jones & Bartlett Learning.
- Ngulube, P. ed. 2021. Handbook of research on mixed methods research in information science. Hershey: IGI Global.

- Nicol, M, Bavin, C, Cronin, P, Rawlings-Anderson, K, Cole, E, Hunter, J. 2012. *Nursing skills clinical skills for caring*. 4th edition. St Louis: Elsevier Health Sciences.
- Nori, K, Benjamin, L, Alshalji, J & Izakovic, J. 2019. *Pediatric dermatologic surgery.* Oxford. John Wiley & Sons.
- NSCA-National Strength and Conditioning Association & Jacobs, PL. 2018. NSCA's essentials of training special populations. Champaign: Human Kinetics.
- Nugent, PM & Vitale, BA. 2017. Fundamentals Davis essential nursing content and practice questions. Philadelphia: F.A. Davis.
- O'Connor, AB. 2015. Clinical instruction and evaluation. A teaching resources. 3rd edition. Burlington: Jones & Barret Learning.
- Oakes, LL. 2011. Compact clinical guide to infant and child pain management: an evidence-based approach for nurses. New York: Springer Publishing Company.
- Oakland, J & Oakland, JS. 2019. *Statistical process control.* 7th edition. New York Routledge.
- Ømgreen, R, Meyer, B & Buhl, M. 2019. ECL 2019 18th European Conference on elearning. Copenhagen: Academic Conferences and publishing limited.
- Onwuegbuzie, AJ. & Johnson, RB. eds. 2021. *The Routledge reviewer's guide to mixed methods analysis*.New York: Routledge.
- Orey, M. 2015. Communication skills training. Alexandria: ADT Press.
- Oriot, D & Alinier, G. 2017. *Pocketbook for simulation debriefing in healthcare*. Springer. Cham: Springer International Publishing AG.
- Owen, J. 2014. Behavioral aspects of sleep problems in childhood and adolescence, an issue of sleep medicine clinics, e-book. Philadelphia: Elsevier Health Science.
- Pace, V, Treloar, A & Scott, S. 2011. *Dementia: from advanced disease to bereavement.* Oxford: Oxford University Press.

- Pachana, NA & Laidlaw, K. 2014. *The Oxford handbook of clinical geropsychology*. Oxford: Oxford University Press.
- Paice, JA. 2015. *Physical aspects of care: pain and gastrointestinal symptoms*. Oxford: Oxford University Press.
- Pajo, B. 2017. *Introduction to research methods:* a hands-on approach. Thousand Oaks: SAGE.
- Pallant, J., 2020. SPSS survival manual: A step by step guide to data analysis using IBM SPSS. New York: Routledge.
- Parsons, G & Preece, W. 2010. *Principles and practice of managing pain: a guide for nurses and allied health professionals.* New York: Mc Graw Hill-Open University Press.
- Pasero, C & McCaffery, M. 2011. *Pain assessment and pharmacological management*. St Louis: Mosby Elsevier.
- Patel, M. 2018. CRM integrate with ERP. 1st edition. Lunawada: RED'SHINE Publication.
- Paternoster, R & Bachman, RD. 2018. Essentials of statistics for criminology and criminal justice. Thousand Oaks: SAGE.
- Paz, JC & West, MP. 2013. Acute care handbook for physical therapists e-book. 4th edition. Philadelphia: Elsevier Health Sciences.
- Peate, I. 2019. *Alexander's nursing practice e-book: hospital and home*. 5th edition. St. Louis: Elsevier Health Sciences.
- Pecora, PJ. ed. 2018. Evaluating family-based services. New York: Routledge.
- Perry, AG, Potter, PA & Ostendorf, W. 2015. *Nursing interventions and clinical skills e-book*. 6th edition. Philadelphia: Elsevier Health Sciences.
- Perry, AG, Potter, PA, Ostendorf, W & Laplante, N. 2021. *Clinical nursing skills and techniques e-book*. 10th edition. St. Louis: Elsevier Health Sciences.

- Perry, SE, Hockenberry, MJ, Lowdermilk, DL & Wilson, D. 2014. Maternal child nursing care. 5th edition. Philadelphia: Elsevier Health Sciences.
- Phillips, JJ & Phillips, PP. 2016. *Handbook of training evaluation and measurement methods*. 4th edition. New York: Routledge.
- Pickering, G & Gibson, S. 2015. *Pain, emotion and cognition: a complex nexus.* New York: Springer.
- Pickering, G, Zwakhalen, S & Kaasalainen, S. 2018. *Pain management in older adults: a nursing perspective*. Cham: Springer Nature.
- Pillitteri, A. 2010. *Maternal and child health nursing: Care of the childbearing family*. Philadelphia: Lippincott Williams & Wilkins.
- Pillitteri, A. 2014. *Maternal and child health nursing care of the childbearing and childrearing family*. 7th edition. Philadelphia: Lippincott Williams & Wilkins.
- Polit, DF & Beck, CT. 2021. *Nursing research generating and assessing evidence evidence for nursing practice*. 11th edition. Philadelphia: Wolters Kluwer Health
- Poretsky, L & Liao, EP. 2013. Acute and chronic complications of diabetes, an issue of endocrinology and metabolism clinics, e-book. Philadelphia: Elsevier Health Sciences.
- Potter, PA, Perry, AG, Stockert, P & Hall, A. 2013. *Fundamentals of nursing*. 8th edition. St Louis: Elsevier Health Sciences.
- Potter, PA, Perry, AG, Stockert, PA & Hall, A. 2020. *Fundamentals of nursing*. 10th edition. St. Louis: Elsevier Health Sciences.
- Pozgar, GD. 2019. Legal and ethical issues for health professionals. 5th edition. Burlington: Jones & Bartlett Learning.
- Privitera, GJ. 2022. *Research methods for the behavioral sciences*. Thousand Oaks: SAGE Publications.

- Profetto-McGrath, J, Polit, DF & Beck, CT. 2010. *Canadian essentials of nursing research*. Philadelphia: Wolters Kluwer Heatlh.
- Quinlan-Colwell, A & D'Arcy, YM. 2012. Compact clinical guide to geriatric pain management: an evidence-based approach for nurses. New York: Springer Publishing Company.
- Qureshi, AI & Georgiadis, AL. 2011. *Textbook of interventional neurology*. Cambridge: Cambridge University Press.
- Ramachandra, DL. 2018. *Concise book on geriatric medicine*. New Delhi: Educreation Publishing.
- Ray, SD. 2017. Side effects of drugs annual: a worldwide yearly survey of new data in adverse drug reactions. St Louis: Elsevier Health Sciences.
- Rebeiro, G, Jack, L, Scully, N & Wilson, D. 2013. *Fundamentals of nursing: clinical skills workbook*. Philadelphia: Elsevier Health Sciences.
- Reed, A, Bohlander, M, Wake, N & Smith, E. 2016. *Consent: domestic and comparative perspectives*. New York: Routledge.
- Rees, C. 2011. *Introduction to research for midwives*. 3rd edition. St. Louis: Elsevier Health Sciences.
- Regis, TR. 2018. Doctoral students: Attrition, retention rates, motivation, and financial constraints. Lulu. com.
- Richards, KA., Hemphill, MA & Wright, PM. 2023. *Qualitative research and evaluation in physical education and sport pedagogy*. Jones & Bartlett Learning.

Richards, L. 2021. Handling qualitative data a practical guide. Thousand Oaks: SAGE.

- Rizzo, M. 2018. *The Wiley handbook on the aging mind and brain*. New Jersey: John Wiley & Sons.
- Roberts, JR, Custalow, CB & Thomsen, TW. 2017. *Clinical procedure in emergency medicine and acute care.* 7th edition. Philadelphia: Elsevier, Inc.

Rolf, HL. 2014. Finite mathematics. 8th edition. Boston: Cengage Learning.

- Roller-Wirnsberger, R, Singler, K & Polidori, MC. 2018. *Learning geriatric medicine: a study guide for medical students*. Cham: Springer International Publishing AG.
- Rose, M. 2013. Oncology in primary care. Philadelphia: Lippincott Williams & Wilkins.
- Rubin, A. 2020. *Pragmatic Program Evaluation for Social Work*. Cambridge: Cambridge University Press.
- Rubin, A & Babbie, ER. 2014. *Research methods for social work*. 8th edition. Boston: Cengage Learning.
- Rumble, S. 2019. Prepare operational budgets. Victoria: Cengage AU.
- Saks, AM & Haccoun, RR. 2010. *Managing performance through training and development*. 5th edition. Toronto: Cengage Learning.
- Salhan, S. 2011. *Textbook of gynaecology*. London: JP Medical Ltd.
- Salmons, JE. 2016. Doing qualitative research online. Thousand Oaks: SAGE.
- Salvendy, S. 2012. *Handbook of human factors and ergonomics.* 4th edition. New Jersey: John Wiley & Sons Inc.
- Sangster, M. 2016. *Engaging primary children in mathematics*. New York: Bloomsbury Publishing.
- Sankaranarayanan, B & Sindhu, B. 2012. *Learning and teaching nursing*. 4th edition. New Delhi: JP Medical Ltd.
- Saunders. RP. 2015. *Implementation monitoring and process evaluation*. Thousand Oaks: SAGE Publications.
- Schaller, AS. 2018. Impact of pain and evaluation of education and self-care in patients with head and neck cancer. Linköping: Linköping University Electronic Press.

Schneider, K. 2014. Transfer of learning in organizations. New York: Springer.

Seel, NM. 2011. Encyclopedia of the sciences of learning. New York: Springer.

- Shaffer, DR & Kipp, K. 2014. Developmental psychology: childhood and adolescence. Belmont: Cengage Learning.
- Shalin, HJ. 2014. *Remote workforce training: effective technologies and strategies*. Hershey: IGI Global.
- Sharma, S. 2022. *Nursing Research and Statistics-E-Book*. St. Louis: Elsevier Health Sciences.
- Silvestri, LA. 2012. Saunders Q & A review for the NCLEX-R[®] examination. St Louis: Elsevier Health Sciences.
- Sit, HW. 2017. Inclusive teaching strategies for discipline-based English studies: enhancing language attainment and classroom interaction in a multicultural learning environment. Newcastle: Springer.
- Skeel, RT & Khleif, SN. 2011. *Handbook of cancer chemotherapy.* 8th edition. Philadelphia: Lippincott Williams & Wilkins.
- Slutskiy, P. 2021. Communication and Libertarianism. Philadelphia: Springer.
- Snowman, J & McCown, R. 2015. *Psychology applied to teaching*. 14th edition. Stamford: Cengage Learning.
- Snowman, J, McCown, R & Biehler, R. 2012. *Psychology applied to teaching*. 13th edition. Belmont: Cengage Learning.
- Speedie, L & Middleton, A. 2021. Wong's nursing care of infants and children: Australia and New Zealand edition e-book: for professionals. St Louis: Elsevier Health Science.
- Stacey, A. ed. 2019. June. ECRM 2019 18th European conference on research methods in business and management. Academic Conferences and publishing limited.

- Stanhope, M & Lancaster, J. 2012. *Public health nursing population-centered health care in the community*. 8th edition. Philadelphia: Elsevier Mosby.
- Stannard, C, Coupe, M & Pickering, T. 2013. *Opioids in non-cancer pain*. Oxford: Oxford University Press.
- Stannard, D & Krenzischek, D. 2018. *Perianesthesia nursing care*. 2nd edition. Burlington. Jones & Bartlett Learning.
- Stauss, B. and Seidel, W. 2019. *Effective complaint management: The business case for customer satisfaction*. 2nd edition. Cham: Springer Publishing Company.
- Stefan, DC & Rodriquez-Galindo, C. 2014. *Pediatric haematology-oncology in countries with limited resources: a practical manual*. New York: Springer Science & Business Media.
- Stefan, K. (ed.) 2021. Innovation management: insights by young business developers volume 3. Deutsches Institut für Ideen- und Innovationsmanagment.
- Still, C. Sarwer, DB & Blankenship, J. 2015. *The ASMBS textbook of bariatric surgery: Volume 2: Integrated health.* Philadelphia: Springer.
- Sumil, N. 2016. The minds-on hearts-on hands-on learning engagements. GRIN Verlag.
- Sumser, B, Leimena, M & Altillo, T. 2019. *Palliative care: a guide for health social workers.* New York: Oxford University Press.
- Sun, P. 2011. *The treatment of pain with Chinese herbs and acupuncture e-book*. 2nd edition. Philadelphia: Churchill Livingstone.
- Suresh, S. 2016. *Communication and educational technology: contemporary pedagogy for health care professionals*. 2nd edition. Philadelphia: Elsevier Health Sciences.
- Suresh, S. 2017. *Potter and Peery's fundamentals of nursing: second South Asia edition e-book.* 2nd edition. St Louis. Elsevier Health Sciences.

- Swearingen, PL. 2016. *All-in-one care planning resource e-book.* 4th edition. St Louis: Elsevier Health Sciences.
- Tashakkori, A & Teddlie, C. 2021. Sage handbook of mixed methods in social & behavioral research. Thousand Oaks: SAGE publications
- Taylor, RR. 2017. *Kielhofner's research in occupational therapy: methods of inquiry for enhancing practice*. 2nd edition. Philadelphia: FA Davies.
- Thomas, R. 2011. Pain management task force: final report. Darby: Dianne Publishing.
- Tidwell, J. 2017. *Pediatric critical care, an issue of critical nursing clinics*. Vol. 29. Philadelphia: Elsevier Health Sciences.
- Tiko, I. 2015. Strategic information technology governance and organizational politics in modern business. Hershey: Business Science Reference.
- Tiran, D. 2010. *Reflexology in pregnancy and childbirth e-book*. St Louis: Churchill Livingstone.
- Tollefson, J, Bishop, T, Jelly, E. Watson, G. and Tambree, K. 2011. *Clinical Skills for Enrolled/Division 2 Nurses*. Cengage Learning Australia.
- Torlone, F & Vryonides, M. 2016. *Innovative learning models for prisoners. Volume 4* of studies on adult learning and education. Firenze: Firenze University Press.
- Toth, C & Moulin, DE. 2013. *Neuropathic pain: causes management and understanding*. Cambridge: Cambridge University Press.
- Tsui, BCH & Suresh, S. 2015. Pediatric atlas of ultrasound and nerve stimulationguided regional anesthesia. New York: Springer Science & Business Media.
- Turk, DC & Melzack, R. 2011. *Handbook of pain assessment*. 3rd edition. New York: Guildford Press.
- Urman, RD & Vadivelu, N. 2011. *Pocket pain medicine*. Philadelphia: Lippincott Williams & Wilkins.

- Vadivelu, N, Urman, RD & Hines. RL. 2011. *Essentials of pain management.* Springer Science & Business Media.
- Valencia-Go, GN. 2016. *Research success: a Q & A review applying critical thinking to test taking*. Philadelphia: F.A. Davis.
- Van der Akker, J, Branch, RM, Gustafson, K, Nieveen, N & Plomp, T. 2012. *Design approaches and tools in education and training*. New York: Springer-Science & Business Media.
- Van der Loo, M & De Jong, E. 2018. *Statistical data cleaning with applications in R*. Oxford: John Wiley & Sons.
- Van Griensven, H, Strong, J & Unruh, AM. 2014. *Pain: a textbook for therapists.* 2nd edition. Philadelphia: Elsevier Health Services.
- Various authors. 2018. Routledge library editions: comparative education. New York Routledge
- Vervoort, TV, Karos, K, Trost, Z & Prkachin, KM. 2018. Social and interpersonal dynamics in pain: we don't suffer alone. Cham: Springer International Publishing AG.
- Vu, P, Fredrickson, S & Moore, C. 2017. Handbook of research on innovative pedagogies and technologies for online learning in higher education. Hershey: IGI Global.
- Waldman, SD. 2011. *Pain management*. 2nd edition. Philadelphia: Elsevier Health Sciences.
- Wa-Mbaleka, S. & Rosario, A. eds. 2022. *The SAGE handbook of qualitative research in the Asian context.* Thousand Oaks: SAGE.
- Wan, M. 2013. Incidental trainer: a reference guide for training design, development delivery. New York: CRS Press.

- Wang, VC. ed. 2018. Scholarly publishing and research methods across disciplines. Hershey: IGI Global.
- Wang, VCX. 2017. *Encyclopedia of strategic leadership and management*. Hershey: IGI Global.
- Wang. VCX & Reio Jr. TG. 2017. Handbook of research on innovative techniques, trends, and analysis for optimized research methods. Hershey: GI Global.
- Washington, CM & Leaver, DT. 2015. *Principle and practice of radiation. e-book.* 4th edition Philadelphia: Elsevier Health Services.
- Weber, A. 2014. *Turning learning into action. A proven methodology for effective transfer of learning.* London: Kogan Page Limited.
- Weber, JR & Kelly, JH, Sprengel, AD. 2014. *Lab manual for health assessment in nursing*. 5th edition. Philadelphia: Lippincott Williams & Wilkins.
- Weber, JR & Kelly, JH. 2014. *Health assessment in nursing*. 5th edition. Philadelphia: Lippincott Williams & Wilkins.
- Weigelt, JA. 2012. Surgical critical care, an issue of surgical clinics. E-book. Philadelphia: Elsevier Health Sciences.
- Weir, JP & Vincent, W.J. 2020. *Statistics in kinesiology*. Champaign: Human Kinetics Publishers.
- Werner, JM & DeSimone, RL. 2012. *Human resource development.* 6th edition. Mason: Cengage Learning.
- Werth, Jr, JL. 2013. Counselling clients near the end of life: a practical guide for mental health professionals: New York: Springer Publishing Company.
- White, L, Duncan, G & Baumie, W. 2013. *Medical surgical nursing: an integrated approach*. 3rd edition. Clifton Park: Cengage Learning.
- WHO. 2020. State of the world's nursing 2020: investing in education, jobs and leadership. Geneva: World Health Organisation.

Wright, S. 2014. Pain management in nursing practice. Thousand Oaks: SAGE.

- Wyatt, JP, Taylor, RG, de Wit, K & Hotton, EJ. 2020. *Oxford handbook of emergency medicine*. 5th edition. Oxford: Oxford University Press.
- Yarbro, CH, Wujcik, D & Gobel, BH. 2014. *Cancer symptom management*. Burlington: Jones & Bartlett Learning.
- Zahoor, R. 2021. Reality of Reality Television. London: Blue Rose Publishers.
- Zhang, H. 2021. Learning from experience: The use of structured video-assisted debriefing among nursing students. Volume 1760 of Linköping University Medical Dissertations. Linköping: Linköping University Electronic Press.
- Zickermann, P. 2014. *Co-branding: fit factors between partner brands*. Hamburg: Anchor Academia Publishing.

Journal Articles:

- Ackini, C & Saunders, MNK. 2015. Using questionnaire surveys to gather data for within organisation HRD research. In: Saunders, M & Tosey, P (eds.) *Handbook of Research Methods on HRD.* Cheltenham: Edward Elgar, 353–365.
- Adams, AT, Murname, EL, Adams, P, Elfenbein, M, Chang, PF, Sannon, S, Gay, G & Choudhury, T. 2018. Keppi: A Tangible user interface for self-reporting pain.
 Proceedings of the CHI Conference on Human Factors in Computing Systems.
 Paper no. 502:1–13.
- Adamson, KA & Prion, S. 2013. Reliability: measuring internal consistency using Cronbach's α. *Clinical Simulation in Nursing*, 9:e179-e180.
- Aghera, A, Emery, M, Bounds, R, Bush, C, Stansfield, RB, Gillett, B & Santen, SA.
 2018. A randomized trial of SMART goal-enhanced debriefing after simulation to promote educational actions. *Western Journal of Emergency Medicine*, 9(1):112–120.

- Ahlstedt, C. Lindvall, CE, Holmström, IK & Athlin, AM. 2019. What makes registered nurses remain in work? An ethnographic study. *International Journal of Nursing Studies*, 89:32–38.
- Ahmad, WMRW, Razzaq, ARA, Mustafa, MZ, Ahmad, A, Gessler & Spöttle. 2013. Learning transfer types in national dual training system in Malaysia. *Asian Journal of Humanities and Social Sciences*, 1(2):196–206.
- Ahmed, S.S.S. 2020. Nurses' knowledge regarding neonatal pain management at neonatal care units in Wad Medani Pediatric and Obstetrics and Gynecology Teaching Hospitals, Gezira State, Sudan. Doctoral thesis, University of Gezira.
- Aktas, YY & Karabulut, N. 2016. A survey on Turkish nursing students' perception of clinical learning environment and its association with academic motivation and clinical decision making. *Nurse Education Today*, 36:124–128.
- Al Yami, MS & Watson, R. 2014. An overview of nursing in Saudi Arabia. *Journal of Health Specialties*, 2(1):10–12.
- Alboliteeh, M, Magarey, J & Wiechula, R. 2017. The profile of Saudi nursing workforce: a cross-sectional study. *Nursing Research and Practice*, 2017:1–9.
- Alboliteeh, M, Magarey, J & Wiechula, R. 2018. The professional journey of Saudi nurses graduates: A lived experience. *Clinical Nursing Studies*, 6 (1):76–82.
- Alharbi, HA. 2018. Readiness for self-directed learning: how bridging and traditional nursing students differ? *Nurse Education Today*, 61:231–234.
- Al-Hatem, Al, Masood, M & Al-Samarraie, H. 2018. Fostering student nurses' selfregulated learning with the Second Life environment: an empirical study. *Journal of Information Technology Education*, 17:285–307.
- Ali, M, Sultan, SF, Kumar, R & Ghouri, N. 2020. Knowledge, attitude and practice of labor analgesia amongst healthcare worker and patients: a single center cross sectional study. *Pakistan Journal of Medical Sciences*, 36(1):S4–S8.

- Alizadeh, Z, Paymard, A, Khalili, A & Hejr, H. 2017. A systemic review of pain assessment method in children. *Annals of Tropical Medicine and Public Health*, 10(4):1–4.
- Alizadeh-Khoie, M, Sharifi, Farshad, Akbar, ME, Fadayevatan, R & Haghi, M. 2017. Iranian brief pain inventory: validation and application in elderly people with cancer pain. *Journal of Pain and Symptom Management*, 54 (4):563–569.
- Aljohani, KAS. 2020. Nursing education in Saudi Arabia: history and development. *Cureus* 12(4):e7874. DOI 10.7759/cureus.7874.
- Alkhazim, MA & Althubaiti, A. 2016. Continuing medical education in Saudi Arabia: Experiences and perception of participants. *Journal of Health Specialities*, 2(1):13–19.
- Allred, K & Gerardi, N. 2017. Computer simulation for pain management education: a pilot study. *Pain Management Nursing*, 18 (5):278–287.
- Almasi, K, Bavani, SM & Mohammadpour, Y. 2018. Examining the preferred learning styles (PLSs) of nursing and midwifery students of Urmia University of Medical Sciences. World Family Medicine/Middle East Journal of Medicine, 16 (1):241– 244.
- Alnajar, MK, Darawad, MW, Alshahwan, SS & Samarkandi, OA. 2019. Knowledge and attitudes toward cancer pain management among nurses at oncology units. *Journal of Cancer Education*, 34:186–193.
- Alotaibi, K, Higgins, I, Day, J & Chan, S. 2018. Paediatric pain management: knowledge, attitudes, barriers and facilitators among nurses – integrative review. *International Nursing Review*, 65:524–533.
- Alotaibi, K. Higgins, I & Chan, S. 2019. Nurses' knowledge and attitude toward pediatric pain management: a cross-sectional study. *Pain Management Nursing*, 20(2):118–125.

- Alqahtani, M & Jones, LK. 2015. Quantitative study of oncology nurses' knowledge and attitudes towards pain management in Saudi Arabian hospitals. *European Journal of Oncology*, 19:44–49.
- Alshareef, AA, Wraith, D, Dingle, K & Mays, JM. 2020. Identifying the factors influencing Saudi Arabian nurses' turnover. *Journal of Nursing Management*, 28(5):1030–1040.
- Aluko, FR & Shonubi, OK. 2014. Going beyond Kirkpatrick's training evaluation model: the role of workplace factors in distance learning transfer. *Africa Education Review*, 11(4):638–657.
- Alzghoul, BI, Abdullah, NAC. 2016. Pain management practices by nurses: an application of the knowledge, attitude and practices (KAP) model. *Global Journal of Health Science*, 8(6):154–160.
- Ammaturo, DA, Hadjistavropoulos, MAT & Williams, J. 2017. Pain in dementia: use of observational pain assessment tools by people who are not health professionals. *Pain Medicine*, 18:1895–1907.
- Amonash, AK, Kyei-Dompim, J, Bam, V, Kyei, EF Oduro, E, Ahoto, CK & Axellin, A. 2020. Exploring the educational needs of nurses on children's pain management: A descriptive qualitative study. *Nursing Open*, 7:841–849.
- Anderson, R, Taylor, S, Taylor, T & Virues-Ortega, J. 2022. Thematic and textual analysis methods for developing social validity questionnaires in applied behavior analysis. *Behavioral Interventions*, 37(3):732-753.
- Panlican, A, Passay-an, E & Gonzales, F. 2020. A survey on the knowledge and attitude on pain management among nurses employed in the government hospitals. Saudi Journal for Health Sciences, 9(2):97–101.
- Amris, K, Jones, LE & Williams, AC de C. 2019. Pain from torture: assessment and management. *PAIN Reports,* 4 (e794):1–4.
- Arifin, SRM. 2018. Ethical considerations in qualitative study. *International Journal of Care Scholars*, 1(2):30–33.

- Andersen, RA, Nakstad, B, Jylli, L, Cambell-Yeo, M & Anderzen-Carlsson, A. 2019.
 The complexities of nurses' pain assessment in hospitalized preverbal children.
 Pain Management Nursing, 20:337–334.
- Andersen, RD, Munsters, JMA, Vederhus, BJ & Gradin, M. 2018. Pain assessment practices in Swedish and Norwegian neonatal care units. *Scandinavian Journal of Caring Sciences*, 32(3):1074–1082.
- Angeletti, C, Guetti, C, Paesani, M, Colavincenzo, S, Ciccozzi, A & Angeletti, PM. 2018. An analysis of Italia nurses' approach to patients' pain: a nationwide online survey. *Hindawi Pain Research and Management*, April 2018:1–8.
- Arbour, C & Gélinas, C. 2014. Behavioral and physiologic indicators of pain in nonverbal patients with a traumatic brain injury: an integrative review. *Pain Management Nursing*, 15(2):506–518
- Asadi-Noghabi, AA, Gholizadeh, M, Zolfaghan, M, Mehran, A & Sohrabi, M. 2015. Nurses use of critical care pain observational tool in patients with low consciousness. *Oman Medical Journal*, 30(4):276–282.
- Atee, M, Hoti, K, Parsons, R, Hughes, JD. 2017. Pain assessment in dementia: evaluation of a point–of–care technological solution. *Journal of Alzheimer's Disease*, 60:137–150.
- Auerbach, DI, Buerhaus, PI & Staiger, DO. 2017. Millennials almost twice as likely to be registered nurses as baby boomers were. *Health Affairs*, 36(10):1804–1807.
- Aydede, M. 2017. Defending the IASP definition of pain. *The Monist*, 100(4):1–31.
- Aziato, L & Adejumo, O. 2013. Determinants of nurses' knowledge gap on pain management in Ghana. *Nurse Education in Practice*, 14(2):195-199.
- Badu, HI, Abalo, EM, Bam, VB, Agyemang, DO, Noi, S, Budu, FA & Peprah, P. 2019.
 "I prefer a male nurse to a female nurse" patients' preference for and satisfaction with nursing care provided by male nurses at the Komfo Anokye Teaching Hospital. *BMC Nursing*, 18:1–9.

- Banerjee, P, Gupta, R & Bates, R. 2017. Influence of organisational learning culture on knowledge worker's motivation to transfer training: testing moderating effects of learning transfer climate. *Current Psychology*, 36(3):606–617.
- Bansal, A & Thakur, M. 2013. The impact of perception of organizational transfer climate factors and trainees, characteristics on training transfer: the context of mergers and acquisitions. *Journal of International Business and Economics*, 1(1):50–66.
- Baraki, Z, Girmay, F, Kidanu, K, Gerensea, H, Gezehgne, D & Teklay, H. 2017. A cross-sectional study on nursing process implementation and associated factors among nurses working in selected hospitals of Central and Northwest zones, Tigray Region, Ethiopia. *BioMed Central Nursing*, 16(54):1–9.
- Barros, SRAF & Albuquerque, APS. 2014. Nursing approaches for pain diagnosis and classification of outcomes. *Revista Dor,* 15(2):107–111.
- Beek, K, Dawson, A & Whelan, A. 2017. A review of factors affecting the transfer of sexual and reproductive health training into practice in low and lower-middle-income country humanitarian settings. *Conflict and Health*, 11(16):1–12.
- Bement, MKH & Sluka, KA. 2015. The current state of physical pain curricula in the United States: A faculty survey. *The Journal of Pain*, 16(2):144–152.
- Berg-Weger, M & Stewart, DB. 2017. Non-pharmacologic interventions for persons with dementia. *Science of Medicine*, 114 (2):116–119.
- Bhandari, RP, Goddard, J, Campbell, F, Sangster, M & Stevens, B. 2019. Becoming a pediatric pain specialist: training opportunities to advance the science and practice of pediatric pain treatment. *Pediatric Pain Letter*, 21(1):1–10.
- Bice, AA. 2018. Cognitive dissonance and pediatric procedural pain management: A concept clarification. *Pain Management Nursing*, 19(3):230–237.
- Björn, A, Pudas-Tähkä, SM, Salanterä, S & Axelin, A. 2017. Video education for critical care nurses to assess pain with behavioural pain assessment tool: a descriptive comparative study. *Intensive and Critical Care Nursing*, 42:68–74.

- Blau, I & Shamir-Inbal, T. 2017. Re-deigned flipped learning model in an academic course: the role of co-creation and co-regulation. *Computers & Education*, 115:69–81.
- Blume, BD, Ford, JK, Surface, EA & Olenick, J. 2017. A dynamic model of training transfer. *Human Resource Management Review*, 29(2):1–14.
- Boekel, LM, Vissers, KCP, van der Sande, R, Bronkhorst, E, Lerou, JGC & Steegers, MAH. 2017. Moving beyond pain scores: multidimensional pain assessment is essential for adequate pain management after surgery. *PLoS ONE*, 12(5):1–16.
- Bonkowski, SL, De Gange, JC, Cade, MB & Bulla, SA. 2018. Evaluation of a pain management education program and operational guideline on nursing practice, attitudes, and pain management. *The Journal of Continuing Education in Nursing*, 49 (4):178–185.
- Booker, SQ & Haedtke, C. 2016. Assessing pain in verbal older adults. *Nursing*, 46(2):65–68.
- Boonpektrakul, OA. 2013. Developing Action-Plan Training Model to enhance employee's competencies: a hotel case study. *Stamford Journal*, 5 (2):9–16.
- Bostick, GP, Norman, KE, Sharma, A, Toxopeus, R, Irwin, G & Dhillon, R. 2021. Improving cultural knowledge to facilitate cultural adaptation of pain management in a culturally and linguistically diverse community. *Physiotherapy Canada*, 73 (1):19–25.
- Botma, Y & MacKenzie, MJ. 2016. Perspective on transfer of learning by nursing students in primary healthcare facilities. *Journal of Nursing Education and Practice*, 6(11):104–110.
- Botma, Y, Van Rensburg, Van Rensburg, GH, Coetzee, IM & Heyns, T. 2015. A conceptual framework for educational design at modular level to promote transfer of learning. *Innovations in Educational and Teaching International*, 52(5):499– 509.

- Bowman, RL, Davis, DL, Ferguson, RM & Taylor, J. 2018. Women's motivation, perception and experience of complementary and alternative medicine in pregnancy: a meta-synthesis. *Midwifery*, 59:81–87.
- Boyle, GJ, Boerresen, BH & Jang, DM. 2015. Factor analysis of the McGill Pain Questionnaire (MPQ) in acute and chronic patients. *Psychological Reports*, 116(3):797–820.
- Bradley-Ingle, A. 2019. More nurse supervisors wanted. *Kai Tiaki Nursing: New Zealand*, 25(9):34-35.
- Brant, JM, Mohr, C, Coombs, NC, Finn, S & Wilmarth, E. 2017. Nurses' knowledge and attitudes about pain: personal and professional characteristics and patientreported pain satisfaction. *Pain Management Nursing*, 18(4):214–223.
- Braun, V & Clarke, V. 2006. Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3(2):77–101.
- Brion, C & Codeiro, PA. 2018. The missing link to learning among school leaders in Burkina Faso and Ghana. *Frontiers in Education*, 2(69):1–12.
- Burger, U & Trehan, K. 2018. Action learning in East Africa: new encounters or impossible challenges? *Action Learning: Research and Practice*, 15(2):126–138.
- Burton, AW & Bejarano, PF. 2021. Should know about cancer pain management.
- Carcia-Rodrigues, MT, Bujan-Bravo, S, Seijo-Bestlleiro, R & Gozalez-Martin, C. 2021. Pain assessment and management in the newborn: a systematized review. *World Journal of Clinical Cases*, 9(21) 5921–5931.
- Cartaxo, M & Simões, E. 2014. Effects of the transfer design on post-training performance in eLearning. *Proceeding of INTED 2014 Conference*, 0458–0465.
- Carvalho, AS, Pereira, SM, Jacomo, A, Magalhães, S, Araŭjo, J, Hernandez-Marrero,
 P, Gomes, CC & Schatman, ME. 2018. Ethical decision making in pain management: a conceptual framework. *Journal of Pain Research*, 11:967–976.

- Cascella, M, Bimonte, S, Saettini, F& Muzio, M.R. 2019. The challenge of pain assessment in children with cognitive disabilities: Features and clinical applicability of different observational tools. *Journal of Paediatrics and Child Health*, 55(2):129-135.
- Chaghari, M, Saffari, M, Ebadi, A & Ameryoun, A. 2017. Empowering education: a new model for in-service training of nursing staff. *Journal of Advances in Medical Education & Professionalism*, 5(1):26–32.
- Chan, DXH, Lin, XF, George, JM & Liu, CW. 2020. Clinical challenges and considerations in management of chronic pain patients during a COVID-19 pandemic. *Annals Academy of Medicine*, 49:669–673.
- Chatterjee, A, Pereira, A & Bates, R. 2018. Impact of individual perception of organizational culture on the learning transfer environment. *International Journal of Training and Development*, 22(5):1–19.
- Chen, HJ & Chen, YM. 2015. Pain assessment: validation of the physiologic indicators in the ventilated adult patient. *Pain Management Nursing*, 16(2):105–111.
- Chicca, J & Shellenbarger, T. 2018. Connecting with Generation Z: Approaches in Nursing Education. *Teaching and Learning in Nursing*, 13(3):180–184.
- Chin, T, Caputo, F, Lin, CL & Hu, F. 2022. Understanding cognitive differences across cultures: Integrating neuroscience and cultural psychology. *Frontiers in Psychology*, 13:1041734.
- Chotolli, MR & Luize, PB. 2015. Non-pharmacological approaches to control pediatric cancer pain: nursing team view. *Rev Dor,* 16(2):109–113.
- Choueiry, J, Reszel, J, Hamid, JS, Wilding, J, Martelli, J & Harrison, D. 2020. Development and pilot evaluation of an educational tool for the FLACC pain scale. *Pain Management Nursing*, 21:523–529.
- Chow, KM & Chan, JCY. 2015. Pain knowledge and attitudes of nursing students: a literature review. *Nurse Education Today*, 35:366–372.

- Chu, TL, Wang, J, Lin, HL, Lee, HF, Lin, Chieh, CT, Sung, YC, and Lin, YE. 2019. Multimedia-assisted instruction on pain assesses learning of new nurses: a quasi-experimental study. *BMC Medical Education*, 19(68):1–8.
- Chung, ES, Won, K, Kim, Y & Lee, H. 2014. Water resource vulnerability characteristics by district's population size in a changing climate using subjective and objective weights. *Sustainability*, 6:6141–6157.
- Cimermanová, I. 2018. The Effect of learning styles on academic achievement in different forms of teaching. *International Journal of Instruction*, 11(3):219–232.
- Clarke, V & Braun, V. 2013. Teaching thematic analysis: overcoming challenges and developing strategies for effective learning. *The Psychologist*, 26(2):120–123.
- Clauw, DJ, Essex, MN, Pitman, V & Jones, KD. 2019. Reframing chronic pain as a disease, not a symptom: rationale and implications for pain management. *Post-graduate Medicine*, 131(3):185–198.
- Cole, LC & LoBiondo-Wood, G. 2014. Music as an adjuvant therapy in control of pain symptoms in hospitalized adults: a systematic review. *Pain Management Nurs-ing*, 15(1):406–425.
- Cook, DA & Artino Junior, AR. 2016. Motivation to learn: an overview of contemporary theories. *Medical Education*, 50:997–1014.
- Cooney, MF. 2016. Postoperative pain management clinical practice guidelines. *Journal of PeriAnaesthesia Nursing*, 31(5):445–451.
- Corbett, A, Achterberg, W, Husebo, B, Lobbezoo, F, de Vet, H, Kunz, M, Strand, L, Constantinou, M, Tudose, C, Kappesser, J, de Waal, M & Latenbacher, S. 2014. An international road map to improve pain assessment in people with impaired cognition: the development of the Pain Assessment in Impaired Cognition (PAIC) meta-tool. *BMC Neurology*, 14:2–14.
- Costa, T, Rossato, LM, Bueno, IL, Secco, Sposito, NPB, Harrison, D & De Freitas, JS. 2017. Nurses' knowledge and practices regarding pain management in newborns. *Rev Esc Enferm USP*, 51:1–7.

- Covaerts, N, Kyndt, E., Vreye, S & Dochy, F. 2017. A supervisors' perspective on their role in transfer of training. *Human Resource Development Quarterly,* 28(4):515–552.
- Coyne, P, Mulvenon, C & Paice, JA. 2018. American Society for Pain Management and Hospice and Palliative Nurses Association position statement: pain management at the end of life. *Pain Management Nursing*, 19(1)3–7.
- Craig, P, Hall, S & Phillips, C. 2016. Using the Freeth/Kirkpatrick model to evaluate interprofessional learning outcomes in a rural setting. *Focus on Health Professional Education: A Multi-Disciplinary Journal*, 17(1):84–99.
- Cronfalk, BS, Ternestedt, BM, Larrson, LLF, Henriksen, E Norberg, A & Osterlind, J. 2015. Utilization of palliative care principles in nursing home care: educational interventions. *Palliative and Supportive Care*, 13:1745–1753.
- Curran, V, Gustafson, DL, Simmons, K, Lannon, H, Wang, C. & Garmsiri, M. 2019. Adult learners' perceptions of self-directed learning and digital technology usage in continuing professional education: an update for the digital age. *Journal* of Adult and Continuing Education, 25(1):74–93.
- Da Silva, MR & Montilha, RCI. 2021. Contribution of the Delphi technique to the validation of an occupational therapy assessment in the visual impairment field. *Brazilian Journal of Occupational Therapy*, 29(e2863):1–15.
- Daffron, S, Moore, S & Chicovsky, T. 2015. Transfer of Learning for Health Care Workers. *PAACE Journal of Lifelong Learning*, 24:49–66.
- Daher, A, Versloot, J & Costa, LR. 2014. The cross-cultural process of adapting observational tools for pediatric pain assessment: the case of the dental discomfort questionnaire. *BioMed Central Research Notes*, 7:1–6.
- Damico, V, Cazzaniga, F, Murano, L, Nattino, G & Molin, AD. 2016. Impact of a clinical therapeutic intervention on pain assessment, management, and nursing practices in an intensive care unit: a before-and-after study. *Pain Management Nursing*, 19(3):256–266.

- Danielson, J, Spector, J, Ballweg, R, Vorvick, L & Loeser, JD. 2013. A blueprint of pain curriculum across prelicensure health sciences programs: one NIH pain consortium center of excellence in pain education (CoEPE) experience. *The Journal of Pain*, 14(12):1533–1538.
- Day, L & Beard, KV. 2019. Meaningful inclusion of diverse voices: the case for culturally responsive teaching in nursing education. *Journal of Professional Nursing*, 35:277–281.
- De Freitas, GRM, De Castro, CG, Castro, SMJ & Heineck. 2014. Degree of knowledge of health care professionals about pain management and use of opioids in pediatrics. *Pain Medicine*, 15:807–819.
- Deng, FF. 2016. Comparison of nursing among different countries. *Chinese Nursing Research*, 2:96–98.
- Deni, F, Greco, M, Turi, S, Meani, R, Comotti, L, Perotti, V, Mello, A, Colnaghi, N, Pasculli, N, Nardelli, P, Landoni, G, Beretta, L. 2019. Acute Pain Service: a 10– year experience. *World Institute of Pain*, 19(6):586–593.
- Desai, A, Aucott, S, Frank, K, Silber-Flagg, J. 2018. Comparing N-PASS and NIPS improving pain measurement in the neonate. *Clinical Issues in Neonatal Care*, 18(4):260–266.
- Desai, SA, Nanavati, RN, Jasani, BB & Kabra, N. 2017. Comparison on neonatal pain, agitation, and sedation scale with premature infant pain profile for the assessment of acute prolonged pain in neonate on assisted ventilation: a prospective observational study. *Indian Journal of Palliative Care*, 23(3):287–292.
- Devan, H, Perry, M, van Hattem, A, Thurlow, G, Shepherd, S, Muchemwa, C & Grainger, R. 2019. Do pain management websites foster self-management support for people with persistent pain? A scoping review. *Patient Education and Counseling*, 102(9):1590–1601.

- DeVore, J, Clontz, A, Ren, D, Cairns, L & Beach, M. 2016. Improving satisfaction with better pain management in hospitalized patients. *The Journal for Nurse Practitioner*, 13(1):e23–e27.
- Di, W, Danxia, X & Chun, L. 2019. The effects of learner factors on higher-order thinking in the smart classroom environment. *Journal of Computer in Education*, 6(4):483–498.
- Dijk, JFM, Schuurmans, MJ, Alblas, EE, Kalkman, CJ & van Wijck, AJM. 2017. Postoperative pain: knowledge and beliefs of patients and nurses. *Journal of Clinical Nursing*, 26:3500–3510.
- Dirani. K. 2017. Understanding the process of transfer of training in a military context: marching into new roles. *Advance in Developing Human Resources*, 19(1):101–112.
- Dithole, KS, Thupayagale-Tsheneagae, G, Akpor, OA & Moleki, MM. 2017. Communication skills intervention: promoting effective communication between nurses and mechanically ventilated patients. *BioMed Central Nursing*, 16(1):1–6.
- Dolin, CD, Deierlein, AL and Evans, MI. 2018. Folic acid supplementation to prevent recurrent neural tube defects: 4 milligrams is too much. *Fetal diagnosis and therapy*, 44(3):161–165.
- Dongara, AR, Nimbalkar, SM, Phatak, AG, Patel, DV & Nimbalkar, AS. 2017. An educational intervention to improve nurses' understanding of pain in children in Western India. *Pain Management Nursing*, 18(1):24–32.
- Donovan, P & Darcy, DP. 2011. Learning transfer: the views of practitioners in Ireland. International Journal of Training and Development, 15(2):121–139.
- Doorenbos, AZ, Gordon, DB, Tauben, D, Palisoc, J, Drangsholt, M, Lindhorst, T, Danielson, J, Spector, J, Ballweg, R, Vorvick, L & Loeser, JD. 2013. A blueprint of pain curriculum across prelicensure health sciences programs: one NIH Pain Consortium Center of Excellence in Pain Education (CoEPE) experience. *The Journal of Pain*, 14(12):1533–1538.

- Drake, G & Williams, ACDC. 2017. Nursing education interventions for managing acute pain in hospital settings: a systematic review of clinical outcomes and teaching methods. *Pain Management Nursing*, 18(1):3–15.
- Drennan, VM & Ross, F. 2019. Global nurses shortages: the facts, the impact and action for change. *British Medical Bulletin*, 130(1):25–37.
- Drew, DJ, Gordon, DB, Morgan, B & Mánworrren, RCB. 2018. "As needed" range orders for opioid analgesics in the management of pain: a consensus statement of the American Society for Pain Management Nursing and the American Pain Society. *Pain Management Nursing*, 19(3):207–207.
- Dueñas, M, Ojeda, B, Salazar, A, Mico, JA & Failde, I. 2016. A review of chronic pain impact on patients, their social environment and the health care system. *Journal* of Pain Research, 9:457–467.
- Duffield, C, Gardener, G, Doubrovsky, A & Wise, S. 2019. Manager, clinician, or both? Nurse managers' engagement in clinical care activities. *Journal of Nursing Management*, 27(7):1538–1545.
- Dunlap, JJ & Patterson, S. 2020. Assessing abdominal pain. *Gastroenterology Nursing*, 43(3):267–270.
- Egede, J, Valstar, M, Torres, MT & Sharkey, D. 2019. Automatic neonatal pain estimation: an acute pain in neonates database. 8th International Conference on Affective Comparing and Intelligent Interaction, 475–481.
- Eid, T, Manias, E, Bucknall, T & Almazzrooa, A. 2014. Nurses' knowledge and attitudes regarding pain in Saudi Arabia. *Pain Management Nursing*, 15(4):e25– e36.
- Ekim, A & Ocakci, AF. 2013. Knowledge and attitude regarding pain management of paediatric nurses in Turkey. *Pain Management Nursing*, 14(4):e262–e267.
- El Rahi, C, Zaghloul, H & Murillo, JR. 2017. Pain assessment practices in patients with cancer admitted to the oncology floor. *Journal of Hematology Oncology Pharmacy*, 7(3):109–113.

- Elmali, H & Akpinar, RB. 2017. The effect of watching funny and unfunny videos on post-surgical pain levels. *Complementary Therapies in Clinical Practice*, 26:36–41.
- Emsden, C, Schäfer, US, Denhaerynck, K, Grossmann, Frei, IA, Kirsch, M. 2020. Validating a pain assessment tool in heterogeneous ICU patients: Is it possible? *Nursing Critical Care*, 25:8–15.
- Erogan, G & Celik, S. 2020. Assessment of postoperative pain by the parent, nurse and an independent observer among 1–7-year-old children. *International Journal of Caring Science*, 13(2):1013–1022.
- Evans, M & Boucher, AR. 2015. Optimizing the power of choice: supporting student autonomy to foster motivation and engagement in learning. *International Mind, Brain, and Education Society,* 9(2):87–91.
- Faizal, ANY, Saiful, HM, Bekri, R, Jamil, AB, Amiruddin, MH, Ruhizan, MY & Arasinah, K. 2017. Impact of work environment on learning transfer of skills. *Social Sciences & Humanities*, 25(S):33–40.
- Farshbaf-Khalili, A, Jasemi, M & Seyyedzavvar, A. 2021. Comparing the effect of electronic and lecture education on pain management on the knowledge, attitude, and practice of nurses: a randomized-controlled trial. *Journal of Education and Health Promotion*, October 2021:1–10.
- Fauth, F & González-Martinez, J. 2021. Trainee perceptions of instructional design in continuous online training and learning transfer. *Educational Research International*, 1–12.
- Felix, MMS, Ferreira, MBG, Cruz, LF & Barbosa, MH. 2017. Relaxation therapy with guided imagery for postoperative pain management: an integrative review. *Pain Management Nursing*, 1–10.
- Fereidouni, Z, Sarvestani, RS, Hariri, G, Kuhpaye, SA, Amirkhani, M & Kalyani, MN. 2019. Moving into action: the master key to patient education. *The Journal of Nursing*, 27(1):1–8.

- Ferreira, KASL, De Andrade, DC & Teixeira, MJ. 2013. Development and validation of a Brazilian version of the short-form McGill Pain Questionnaire (SF-MPQ). *Pain Management Nursing*, 14(4):210–219.
- Findley, G, Ryan, C, Cartwright, A & Martin, D. 2019. Study protocol for an investigation of the effectiveness of the pain toolkit for people with low back pain: doubleblind randomised controlled trial. *British Medical Journal*, 9:1–5.
- Fink, RM & Brant, JM. 2018. Complex cancer pain assessment. *Hematology/Oncology Clinics*, 32(3):353-369.
- Fink, RM & Gallagher, E. 2019, June. Cancer pain assessment and measurement. *Seminars in Oncology Nursing*, 35(3):229–234. WB Saunders.
- Finlay, KA & Elander, J. 2016. Reflecting the transition from pain management services to chronic pain support group attendance: an interpretative phenomenological analysis. *British Journal of Health Psychology*, 21:660–667.
- Fischer, T, Hosie, A, Luckett, T, Agar, M & Phillips, J. 2019. Strategies for pain assessment in adult patients with delirium: a scoping review. *Journal of Pain and Symptom Management,* 58(3):487–502.
- Fishman, SM, Young, HM, Arwood, EL, Chou, R, Herr, K, Murinson, BB, Watt-Watson, Carr, DB, Gordon, DB, Stevens, BJ, Bakerjian, D, Ballantyne, JC, Courtenay, M, Koebner, Djukic, M, Koebner, IJ, Mongoven, JM, Prasad, R, Singh, N, Freitas, AC & Silva, AS. 2017. Exploring OHS trainers' role in the transfer of training. *Safety Science*, 91:310–319.
- Fitzgerald, S, Tripp, H & Halksworth-Smith, G. 2017. Assessment and management of acute pain in older people: barriers and facilitators to nursing practice. *Australian Journal of Advanced Nursing*, 35(1):48–57.
- Ford, JK, Baldwin, TT & Prasad, J. 2018. Annual review of organizational psychology and organizational behaviour. Transfer of training known and the unknown. *Annual Reviews*, 5:201–225.

- Forneris, SG & Peden-McAlpine, C. 2021. Evaluation of a reflective learning intervention to improve critical thinking in novice nurses. *Journal of Advance Nursing*, 57(4):410–421.
- Freitas, AC and Silva, SA. 2017. Exploring OHS trainers' role in the transfer of training. Safety Science, 91:310–31
- Friedrichsdorf, SJ & Goubert, L. 2020. Pediatric pain treatment and prevention for hospitalized children. *PAIN Reports*, 5:1–13.
- Fry, M, Arendts, G & Chenoweth, L. 2016. Emergency nurses' evaluation of observational pain assessment tools for older people with cognitive impairment. *Journal of Clinical Nursing*, 26:1281–1290.
- Fujimoto, Y & Härtel, CEJ. 2017. Organizational diversity learning framework: going beyond diversity training programs. *Personnel Review*, 46(6):1120–1141.
- Gaddy, S. 2014. Develop a strategic plan for your disability service unit. *Disability Compliance for Higher Education,* 19(6):1–6.
- Gagliese, L, Gauthier, LR, Narain, N & Freedman, T. 2018. Pain, aging and dementia: towards a biopsychosocial model. *Progress in Neuropsychopharmacology & Biological Psychiatry*, 87:207–215.
- Garcia, JBS, Bonilla, P, Kraychete, DC, Flores, FC, de Valtolina, EDP & Guerreo, C.
 2017. Optimizing post-operating pain management in Latin America. *Revista Brasileria De Anestesiologia*, 67(4):395–403.
- Gegenfurtner, A, Könings, KD, Kosmajac, N & Gebhardt, M. 2016. Voluntary or mandatory training participation as a moderator in the relationship between goal orientation and transfer of training. *International Journal of Training and Development*, 20(4):290–301.
- Gélinas, C, Arbour, C, Michaud, C, Robar, L & Côté, J. 2015. Patients and ICU nurses' perspectives of non-pharmacological interventions for pain management. *British Association of Critical Care Nurses*, 18(6):307–318.

- Genik, LM, McMurtry, CM, Breau, LM, Lewis, SP & Freedman-Kalchman, T. 2018. Pain in children with developmental disabilities: development and preliminary effectiveness of a pain training workshop for respite workers. *Clinical Journal* of Pain, 34(5):428–437.
- Georgiou, E, Hadjibalassi, M, Lambrinou, E, Andreou, P & Papathanassoglou, EDE. 2015. The impact of pain assessment on critically ill patients' outcomes: a systemic review. *BioMed Research International*, 2015:1–18.
- Gerçeker, GÖ, Binay, S, Bilsin, E, Kabraman, A & Yilmaz, HB. 2018. Effects of virtual reality and external cold and vibration on pain in 7 to 12-year-old children during phlebotomy: a randomized controlled trial. *Journal of Perianesthesia Nursing*, 33(6):981–989.
- Germossa, GN, Hellosø, R & Sjetne, IS. 2019. Hospitalized patients' pain experience before and after the introduction of a nurse-based pain management programme: a separate sample pre and post study. *BMC Nursing*, 18(40):1–9.
- Germossa, GN, Sjetne, IS & Hellesø, R. 2018. The impact of an in-service educational program on nurses' knowledge and attitudes regarding pain management in an Ethiopian University Hospital. *Frontiers in Public Health*, 6(299):1–7.
- Ghanbari-Homayi, S, Fardiazar, Z, Meedya, S, Mohammad-Alizade-Charandabi, S, Asghari-Jafarabadi, M, Mohammadi, E & Mirghafouvard, M. 2019. Predictors of traumatic birth experience among a group of Iranian primipara women: a cross-sectional study. *BMC Pregnancy and Childbirth*, 19(182):1–9.
- Ghazali, DA, Kenway, P, Clery, R, Choquet, C and Casalino, E. 2019. A multicenter randomized control trial evaluating professional practice assessment of patient pain management after simulation training course: Study protocol. *Contemporary Clinical Trials Communications*, 14:1–10.
- Ghezeljeh, TN, Nasari, M, Haghani, H & Loieh, HR. 2017. The effect of nature sounds on physiological indicators among patients in the cardiac care unit. *Complementary Therapies in Clinical Practice*, 29:147–152.

- Gil, AJ, Mataveli, M & Garcia-Alcaraz, JL. 2021. Towards an analysis of the transfer of training: empirical evidence from schools in Spain. *European Journal of Training and Development*, 1–15.
- Gill, M. 2018. Anatomy and pathophysiology of chronic pain and the impact of hypnotherapy. A Journal of Clinical Neuroscience and Psychopathology, 20(2):85– 90.
- Gladston, S, Emmanuel, NM & Prasad, L. 2016. Pain assessment: a key to pain alleviation among children. *IOSR Journal of Nursing and Health Science*, 5(2):54– 57.
- Glässel, A, Kirchberger, I, Kollerits, B, Amann, E & Cieza, A. 2011. Content validity of the extended IDV Cors Set for stroke: an international Delphi survey of Physical Therapists. *Physical Therapy*, 91(8):1211–1222.
- Glowacki, D. 2015. Effective pain management and improvements in patients' outcomes and satisfaction. *Critical Care Nurses*, 35(3):33–43.
- Gluszek, E. 2021 Use of the e-Delphi method to validate the corporate reputation management maturity mode (CR3M). *Sustainability*, 13:1–28.
- Goldstein, G, Majdi, A, Bocher-Planka, I, Watts, R & Khan, M. 2018. Physical, emotional, and informational challenges of patients living with endocrine tumours in the United States: understanding their unmet needs. *Journal of Clinical Oncology*, 36(4):299–299.
- Gordon, DB, Rees, SM, McCausland, MP, Pellino, TA, Sanford-Ring, S, Smith-Helmenstine, J & Danis, DM. 2008. Improving reassessment and documentation of pain management. *The Joint Commission Journal on Quality and Patient Safety*, 34(9):509–517.
- Gordon, DB, Watt-Watson, J & Hogans, BB. 2018. Interprofessional pain education with, from, and about competent, collaborative practice teams to transform pain care. *PAIN Reports*, 3(3):1–6.

- Gorrie, F, Goodall, K, Rush, R & Ravenscroft, J. 2019. Towards population screening for cerebral visual impairment: validity of the five questions and the CVI questionnaire. *PloS One*, 14(3):e0214290.
- Govaerts, N, Kyndt, E & Dochy, F. 2017. The influence of specific supervisor support types on transfer of training: examining the mediating effect of training retention. *Vocations and Learning*, 11:265-288.
- Gregory, J & Richardson, C. 2014. The use of pain assessment tools in clinical practice: a pilot survey. *Journal of Pain & Relief*, 3(2):1–5.
- Grenning, K, Nøst, TH, Rannestad, T & Bratås, O. 2018. Participants and developers' experiences with a chronic pain self-management intervention under development: a qualitative study. SAGE Open Medicine, 6:1–9.
- Grossman, R & Burke-Smalley, LA. 2018. Context-dependent accountability strategies to improve the transfer of training: a proposed theoretical model and research propositions. *Human Resource Management Review*, 28:234–247.
- Guswara, AM & Purwanto, W. 2020. The contribution of google classroom application and motivation to the learning outcomes of web programming. *Journal of Education Technology*, 4(4):424–432.
- Gularso D & Okti Purwaningsih, AW. 2023, December. Development of an Instrument to Measure Food Security Literacy Capability of Primary School Students. In Proceedings of the 2nd UPY International Conference on Education and Social Science (UPINCESS 2023), 812: 95. Paris:Springer Nature.
- Haanpää, M. 2014. Clinical examination of a patient with possible neuropathic pain. *IASP Press*, 201–206.
- Habes, EV, Jepma, P, Parlevliet, A, Bakker, A, Buurman, BM. 2020. Video-based tools to enhance nurses' geriatric knowledge: a development and pilot study. *Nurse Education Today*, 90:1–7.

- Hadjistavropoulos, T, Herr, K, Prkachin, KM, Craig, KD, Gibson, SJ, Lukas, A, Smith, JH. 2014. Pain assessment in elderly adults with dementia. *The Lancet Neurology*, 13(12):1216–1227.
- Haigney, S. 2021. Responding to FDA CAPA requests. *Pharmaceutical Technology*, 45(5):52–56.
- Hall, DA, Smith, H, Heffernan, E, Fackrell, K & The Core Outcome Measures in Tinnitus International Delphi (COMiT'ID) Research Steering Group. 2018. Recruiting and retaining participants in e-Delphi surveys for core outcome set development: evaluating the COMiT'ID study. *PLoS One*, 13(7):1–22.
- Hall, H, Leach, M, Brosnan, C & Collins, M. 2017. Nurses' attitudes towards complementary therapies: a systematic review and meta-synthesis. *International Journal of Nursing Studies*, 69:47–56.
- Hall, HG, Griffiths, D & McKenna, LS. 2015. Complementary and alternative medicine: interaction and communication between midwives and women. *Women and Birth*, 28:137–142.
- Hall, RW & Anand, KJS. 2014. Pain management in newborns. *Clinics in Perinatology,* 41:895–924.
- Halstead, JA. & Billings, DM. 2023. Getting Started in Teaching for Nursing and the Health Professions-E-Book: Getting Started in Teaching for Nursing and the Health Professions-E-Book.St. Louis: Elsevier Health Sciences.
- Hamoen, EC, Van Blankenstein, FM, De Jong, PGM, Ray, A & Reinders, ME. 2020.
 Development of a clinical teaching unit in internal medicine to promote interprofessional and multidisciplinary learning: a practical intervention. *Teaching and Learning in Medicine*, 33(1):78–88.
- Hassan, MA, Habiba, U, Majeed, F & Shoaib, M. 2019. Adaptive gamification in elearning based on students' learning styles. *Interactive Learning Environments*, 29 (2019):545–565.

- Hermes, C, Acevedo-Nuevo, M, Berry, A, Kjellgren, T, Negro, A & Massarotto, P. 2018. Gaps in pain, agitation and delirium management in intensive care: outputs from a nurse workshop. *Intensive and Critical Care Nursing*, 48:52–60.
- Herr, K, Coyne, PJ, Ely, EE, Gélinas, C & Manworren, RCB. 2019. Pain assessment in the patient unable to self-report: clinical practice recommendations in support of the ASPMN 2019 position statement. *Pain Management Nursing*, 20:404– 417.
- Herr, K, St Marie, B, Gordon, DB, Paice, JA, Watt-Watson, J, Stevens, BJ, Bakerjian, D, Young, HM. 2015. An interprofessional consensus of core competencies for prelicensure education in pain management: Curriculum application for nursing. *Journal of Nursing Education*, 54(6):317–327.
- Hillman, BA, Tabrizi, MN, Carson, KA & Aucott, SW. 2015. The Neonatal Pain, Agitation and Sedation Scale and the bedside nurse's assessment of neonates. *Journal of Perinatology*, 35:128–131.
- Hla, TK, Hegarty, M, Russell, P, Drake-Brockman, TF & Ramgolam, A. 2014. Perception of pediatric pain: a comparison of postoperative pain assessments between child, parent, nurse, and independent observer. *Pediatric Anesthesia*, 24(11):1127–1131.
- Hoeger Bement, MK, St. Marie, BJ, Nordstrom, TM, Christensen, N, Mongoven, JM, Koebner, IJ, Fishman, SM & Sluka, KA. 2014. An interprofessional consensus of core competencies for prelicensure education in pain management: curriculum application for physical therapy. *Physical Therapy*, 94(4):451–465.
- Hökkä, M, Kaakinen, P & Pölkki, T. 2014. A systematic review: non-pharmacological interventions in treating pain in patients with advanced cancer. *Journal of Advanced Nursing*, 70(9):1954-1969.
- Hollenbaugh, EE & Everett, MR. 2013. The effects of anonymity on self-disclosure in blogs: an application of the online disinhibition effect. *Journal of Computer Me-diated Communication*, 283–302.
- Holliday-Welsh, DM, Gessert, CE & Renier, CM. 2018. Massage in the management of agitation in nursing home residents with cognitive impairment. *Geriatric Nursing*, 30(2):108–117.
- Howard, RF & Liossi, C. 2014. Pain assessment in children. *Archives of Diseases in Childhood*, 1–2.
- Huber, J, Muck, T, Maatz, P, Keck, B, Enders, P, Maatouk, I & Ihrig, A. 2018. Faceface vs. online peer support groups for prostate cancer: A cross-sectional comparison study. *Journal of Cancer Survivorship*, 12(1):1–9.
- Hughes, AM, Zajac, S, Spencer, JM & Salas, E. 2018. A checklist for facilitating transfer in organizations. *Journal of Training and Development*, 1–13.
- Hughes, DM, Breatnach, C, van Dijk, M, Magner, C & Paul, G. 2014. Assessing the validity and reliability of the COMFORT-Behaviour scale in children three years of age and over. *Archives of Diseases in Childhood*, 99(2):A1–A620.
- Hugo, L. 2018. Development and implementation of a training programme from preceptors: a realist evaluation. *Dissertation, University of the Free State*.
- Hummel, P. 2017. Psychometric evaluation of the neonatal pain, agitation, and sedation (N-PASS) scale in infants and children up to age 36 months. *Paediatric Nursing*, 43(4):175–184.
- Hurley-Wallace, A, Wood, C, Franck, LS, Howard, RF & Liossi, C. 2019. Paediatric pain education for health care professionals. *PAIN reports*, 4(1):1–5.
- Hwang, J-H, Tsai, S-J, Liu, T-C, Chen, Y-C & Lai, J-T. 2018. Association of tinnitus and other disorders with a history of migraines. *JAMA Otolaryngology – Head & Neck Surgery*, 144(8):712–717.
- Hylands-White, N, Duarte, R & Raphael, JH. 2017. An overview of treatment approaches for chronic pain management. *Rheumatology International*, 37:29–42.

- Innes, T & Calleja, P. 2018. Transition support for new graduate and novice nurses in critical care settings: an integrative review of the literature. *Nurse Education in Practice*, 30:62–72.
- Ismail, AK, Ghafar, MAA, Shamsuddin, NSA, Roslan, NA, Kaharuddin, H & Muhamad, NAN. 2015.The assessment of acute pain in pre-hospital care using verbal analogue scales. *The Journal of Emergency Medicine*, 49(3):287–293.
- Izaryk, K & Skarakis-Doyle, E. 2017. Using the Delphi technique to explore complex concepts in speech-language pathology: an illustrative example from children's social communication. *American Journal of Speech-Language*, 26(4):1225–1235.
- Izumi, M, Hayashi, Y, Saito, R, Petersen, KK, Arendt-Nielsen, LA & Ikeuchi, M. 2022. Detection of altered pain facilitatory and inhibitory mechanisms in patients with knee osteoarthritis by using a simple bedside tool kit (QuantiPain). *Pain Reports*, 7(e998):1–10.
- Jacklyn, J. 2019. The impact of a computerised pain reassessment tool. *Clarkson College*, 1–18.
- Jai, PN, Bakshi, SG & Thota, RS. 2020. Acute pain service in India: a glimpse of the current scenario. *Journal of Anaesthesiology Clinical Pharmacology*, 31(4):554–557.
- Jang, JH, Park, WH, Kim, HI & Chang, SO. 2020. Ways of reasoning used by nurses in postoperative pain assessment. *Pain Management Nursing*, 21:379–385.
- Jayasekara, R, Smith, C, Hall, C, Rankin, E, Smith, M, Visvanathan, V & Friebe, T. 2018. The effectiveness of clinical education models for undergraduate nursing programs: a systematic review. *Nurse Education in Practice*, 29:116–126.
- Jiang, M, Mieronski, R, Syrjälä, E, Anzanpour, A, Terävä, V, Rahmani, AM, Salnterä, Aantaa, R, Hagelberg, N & Liljeberg, P. 2019. Acute pain intensity monitoring with the classification of multiple physiological parameters. *Journal of Clinical Monitoring and Computing*, 33:493–507.

- Johansen, ML & O'Brien, L. 2016. Decision making in nursing practice: a concept analysis. *Nursing Forum*, 51(1):40–48.
- Jokonya, O. 2016. The significance of mixed methods research in information systems research. *Proceedings of the Eleventh Midwest Association for Information Systems Conference*, 1–5.
- Joly, BM, Coranado, F, Bickford, BC, Leider, JP, Alford, A, Mickeever, J & Harper, E. 2018. A review of public health training needs assessment approaches opportunities to move forward. *Journal of Public Health Management and Practice*, 1–7.
- Jufri, J, Permana, RA & Widagdo, I. 2019. Autogenic relaxation for postoperative caesarean section pain in RSAD Kodam V Brawijaya Surabaya. *International Conference of Kerta Cendekia Nursing Academy*, 1:12–18.
- Kaddoura, M, Vandyke, O, Smallwood, C & Gonzalez, KM. 2016. Perceived benefits and challenges of repeated exposure to high fidelity simulation experiences of first degree accelerated bachelor nursing students. *Nurse Education Today*, 36:298–303.
- Kahsay, DT & Pitkäjärvi, M. 2019. Emergency nurses' knowledge, attitude and perceived barriers regarding pain management in resource-limited settings: crosssectional study. *BMC Nursing*, 18(56):1–13.
- Kamariannaki, D, Alikari, V, Sachlas, A, Stathoulis, J, Fredelos, EC & Zyga, S. 2016. Motivations for the participation of nurses in continuing nursing education programs. *Archives of Hellenic Medicine*, 34(2):229–235.
- Kang, Y & Demiris, G. 2018. Self-report pain assessment tools for cognitively intact older adults: integrative review. *International Journal of Older People Nursing*, 13(2):1–19.
- Karabulut, N, Aktas, YY, Gürcayir, D, Yilmaz, D & Gökmen, V. 2015. Patient satisfaction with their pain management and comfort level after open heart surgery. *Australian Journal of Advanced Nursing*, 32(3):16–24.

- Karcioglu, Ö. 2017. Time to write assessment and documentation of acute pain in the emergency setting. Emergency Medicine, *Openventio*, 3(2):e1–e3.
- Kasurkar, RA, Cate, JTJ, Vos, CMP, Westers, P, Croiset, G. 2013. How motivation affects academic performance: a structural equation modelling analysis. *Advances in Health Sciences Education*, 18:57–69.
- Kayalar, F & Kayalar, F. 2017. The effects of auditory learning strategy on learning skills of language learners (students' views). *IOSR Journal of Humanities and Social Science*, 22(10):4–10.
- Kee, A, McCrate, B, McLennon, S, Wall, D & Jones, S. 2017. Influencing nursing knowledge and attitudes to positively affect care of patients with persistent pain in the hospital setting. *Pain Management Nursing*, 18(3):137–143.
- Keefe, G & Wharrad, HJ. 2012. Using e-learning to enhance nursing students' pain management education. *Nurse Education Today*, 32:e66–e72.
- Keen, A, McCrate, B, McLennon, S, Ellis, A, Wall, D & Jones, S. 2017. Influencing nursing knowledge and attitudes to positively affect care of patients with persistent pain in the hospital setting. *Pain Management Nursing*, 18(3):137–143.
- Khalil, NS. 2018. Critical care nurses' use of non-pharmacological pain management methods in Egypt. *Applied Nursing Research*, 44:33–38.
- Khan, I, Mufti, S & Nazir, NA. 2015. Transfer of training: a recognised review on work environment and motivation to transfer. *International Journal of Management, Knowledge and Learning*, 4(2):197–219.
- Khatib, SK & Razvi, SS. 2018. Nurses' role in acute postoperative pain management: a survey of 16 tertiary hospitals of Maharashtra. *International Journal of Nursing Education,* 10(1):49–54.
- Khoie, MR, Tabrizi, TS, Khorasani, ES, Rahimi, S & Marhamati, N. 2017. A hospital recommendation system based on patient satisfaction survey. *Applied Sciences*, 1–24.

- Kibel, A. 2012. Castrate-resistant prostate cancer. *The Urologic Clinics of North America*, 39(4):15.
- Kim, HJ & Jung, SO. 2020. Comparative evaluations of single-item pain-intensity measures in cancer patients: Numeric rating scale vs. verbal rating scale. *Journal of Clinical Nursing*, 29(15–16):2945–2952.
- Kim, JH & Callahan, JL. 2013. Finding the intersection of the learning organization and learning transfer. *European Journal of Training and Development*, 37(2):183– 200.
- Kim, S. 2020. Journal statistics, Scopus pre-evaluation, and appreciation for 2019 reviewers of the Korean Journal of Women Health Nursing. Korean Journal of Women Health Nursing, 26(1):1–3.
- Kiser-Larson, N. 2017. The experience of intense pain nursing management and interventions. *Journal of Christian Nursing*, 34(2):88–96.
- Kizza, IB, Muliira, JK, Kohi, TW, Nabirye, RC. 2016. Nurses' knowledge of the principles of acute pain assessment in critically ill adult patients who are able to selfreport. *International Journal of Africa Nursing Sciences*, 4:20–27.
- Kliger, M, Stahl, S, Haddad, M, Suzan, E, Adler, R & Eisenberg, E. 2015. Measuring the intensity of chronic pain: are the visual analogue scale and the verbal rating scale interchangeable? *Pain Practice*, 15(15):538–547.
- Koc, T & Goza, D. 2015. The effect of reflexology on acute pain team in infants: randomized controlled trial. Worldviews on Evidence-Based Nursing, 12(5):289– 296.
- Kotfis, K, Zegan-Barańska, M, Szydłowski, Ł, Żukowski, M, Ely. EW. 2017. Methods of pain assessment in adult intensive care unit patients – Polis version of the CPOT (Critical Care Pain Observation Tool) and BPS (Behavioral Pain Scale). *Anaesthesiology Intensive Therapy*, 49(1):66–72.

- Krokmyrdal, KA & Andenaes, R. 2015. Nurses' competence in pain management in patients with opioid addiction: a cross-sectional survey study. *Nurse Education Today*, 35:789–794.
- Kulshrestha, A & Bajwa, SJS. 2021. Management of acute postoperative pain in pediatric patients. *Anaesthesia, Pain & Intensive Care*, 2021:101–107.
- Kundu, SC & Lata, K. 2017. Effects of supportive work environment on employee retention: mediating organizational engagement. *International Journal of Organizational Analysis*, 1–29.
- Kuokkanen, L, Leino-Kilpi, HL, Numminen, O, Isoaho, H, Flinkman, M & Meretoja, R.
 2016. Newly graduated nurses' empowerment and other work-related factors.
 BioMed Central Nursing, 15(22):1–8.
- L'Ecuyer, KM, Hyde, MJ & Shatto, BJ. 2018. Preceptors' perception of role competency. *The Journal of Continuing Education in Nursing*, 49(5):233–240.
- Laguna, M & Purc, E. 2016. Personality traits and training initiation process: intention, planning, and action initiation. *Frontiers in Psychology*, 7(1767):1–11.
- Latchman, J. 2014. Improving pain management at nursing education level: evaluating knowledge and attitudes. *Journal of Advanced Practitioner*, 5(1):0–16.
- LeBaron, VT, Palat, G, Sinha, G, Chinta, SK, Jamima, BK, Pilla, UL, Podduturi, N, Shapuram, Y, Vennela, P, Lalani, Z & Beck, SL. 2017. Recommendations to support nurses and improve the delivery of oncology and palliative care in India. *Indian Journal of Palliative Care*, 23(2):188–198.
- Lee, C, Lee, H, Lee, J & Park, J. 2014. A multiple group analysis of the training transfer model: exploring the differences between high and low performers in a Korean insurance company. *The International Journal of Human Resource Management*, 25(20):2837–2857.
- Lemmetty, S & Collin, K. 2019. Self-directed learning as a practice of workplace learning: interpretative repertoires of self-directed learning in ICT work. *Vacations and Learning*, 13:47–70.

- Leppink, J & Pérez-Fuster, P. 2017. We need more replication research a case for test-retest reliability. *Perspectives on Medical Education*, 6:158–164.
- Liao, YJ, Parajuli, J, Jao, YL, Kitko, L & Berish, D. 2021. Non-pharmacological interventions for pain in people with dementia: a systemic review. *International Journal of Nursing Studies*, 124:1–19.
- Lichtner, V, Dowding, D, Esterhuizen, P, Closs, J, Long, AF, Corbett, A & Briggs, M.
 2014. Pain assessment for people with dementia: a systemic review of systematic reviews of pain assessment tools. *BioMed Central Geriatrics*, 14(138):1–19.
- Lim, SN, Han, HS, Lee, KH, Lee, SC, Kim, JH, Yun, JN, Park, SG, Park, MD, Choe, YH, Ryoo, HM, Lee, KH, Cho, DY, Zang, DY & Choi, JH. 2015. A satisfaction survey on cancer pain management using a self-reporting pain assessment tool. *Journal of Palliative Medicine*, 18(3):225–231.
- Lin, FF, Del Fabbro, L, Needham, J, Sidwell, D & Shaw, J. 2021. Supporting culturally and linguistically diverse (CALD) undergraduate nursing students undertaking clinical placements in Australia: an exploratory qualitative study of clinical facilitator and CALD student perceptions. *Nurse Education Today*, 97:1–9.
- Lin, I, Wiles, L, Waller, R, Goucke, R, Nagree, Y, Gibberd, M, Straker, L, Maher, CG & O'Sullivan, PPB. 2020. What does best practice care for musculoskeletal pain look like? Eleven consistent recommendations from high-quality clinical practice guidelines: systematic review. *British Journal of Sports Medicine*, 54:79– 86.
- Lin, MH, Chen, HC & Liu, KS. 2017. A study of the effects of digital learning on learning motivation and learning outcome. *EURASIA Journal of Mathematics Science and Technology Education*, 13(7):3553–3564.
- Lloyd, B & Stirling, C. 2015. A tool to support meaningful person-centred activity for clients with dementia a Delphi study. *BMC Nursing*, 14(10):1–8.

- Lukas, A, Hagg-Grün, U, Mayer, B, Fisher, T & Schuler, M. 2019. Pain assessment in advanced dementia. Validity of the German PAINAD a prospective double-blind randomised placebo-controlled trial. *PAIN*, 160:742–753.
- Ma, F, Bai, Y, Bai, Y, Ma, W, Yang, X & Li, J. 2018. Factors influencing transfer in nursing profession: a qualitative study. *BMC Medical Education*, 18(44):2–9.
- Ma, X,, Y, Wang, X & Zang, Y. 2018. An integrative review: developing and measuring creativity in nursing. *Nurse Education Today*, 62. 1–8.
- Maatouk, H, Tassi, A, Fawaz, MA & Itani, MS. 2019. Nurses' evaluation of critical care patients. *Data in Brief,* 25:1–7.
- Madagamage, GT, Warnakulasooriya, BNF & Wickramasuriya, HVA. 2014. Factors influencing motivation to transfer training: An empirical study of a government sector training program in Sri Lanka. *Tropical Agricultural Research*, 26(1):12– 25.
- Madagamage, GT, Warnakulasooriya, BNF & Wickramasuriya, HVA. 2014. The impact of trainee characteristics and work environmental factors on motivation to learn. *Asian Journal of Management Sciences & Education*, 3(4):60–73.
- Magana, AJ, Vieira, C & Boutin, M. 2018. Characterizing engineering learners' preferences for active and passive learning methods. *IEEE Transactions on Education*, 61(1):46–54.
- Maguire, M & Delahunt, B. 2017. Doing a thematic analysis: a practical, step-by-step guide for learning and teaching scholars. *AISHE-J (All Ireland Journal of Teach-ing and Learning in Higher Education)*, 8(3):3351–33514.
- Mahama, F & Ninnoni, JPK. 2019. Assessment and management of postoperative pain among nurses at a resource-constraint teaching hospital in Ghana. *Nurs-ing Research and Practice*, 1–7.
- Malakian, A, Dehdashtiyan, M, Aramesh, MR, Aletayeb, SMH & Ghazanfari, F. 2017. Assessment of coorg effect on neonatal pain during heel prick blood sampling: a randomized clinical trial. *Biomedical Research*, 28(15):6880–6883.

- Malara, A, De Biase, GA, Bettarini, F, Ceravolo, F, Di Cello, S, Garo, M, Praino, F, Settembrini, V, Sgro, Spadea, F & Rispoli, V. 2016. *Journal of Alzheimer's Dis*ease, 50:1217–1225.
- Malones, BD, Kallmyr, SS, Hage, V & Eines, TF. 2021. How hospitalized patients evaluate and report their pain together with nurses: A scoping review. *Nordic Journal of Nursing Research*, 41(4):197–206.
- Mamhidir, AG, Sjölund, BM, Fläckman, B, Wimo, A, Sköldunger, A & Engström, M. 2017. Systematic pain assessment in nursing homes: a cluster-randomized trial using mixed-methods approach. *BioMed Central Geriatrics*, 17(61):1–16
- Mandysova, P, Nedvědová, A & Ehler, E. 2017. A comparison of three self-report pain scales in Czech patients with stroke. *Central European Journal of Nursing and Midwifery*, 8(1):572–579.
- Mangold, KL, Kunze, KL, Quinonez, MM, Taylor, LM & Tenison, AJ. 2019. Learning style preferences of practicing nurses. *Journal for Nurses in Professional Development*, 34(4):212–218.
- Manwere, A, Chipfuwa, T, Mukwamba, MM, Chironda, C. 2015. Knowledge and attitudes of registered nurses towards pain management of adult medical patients: a case of Bindura Hospital. *Health Science Journal*, 9(43):1–6.
- Martin, LK. 2015. Implementing a pain care toolkit to improve patient perception of pain care. *The Journal of Continuing Education in Nursing*, 46(7):295–296.
- Marutha, NG & Ngulube, PG. 2012. Electronic records management in the public health sector of the Limpopo province in South Africa. *Journal of the South African Society of Archivists*, 45:39–67.
- Maryniak,K, Markantes,T & Murphy,C. 2017. Enhancing the new nurse experience: creation of new employee training unit. *Nursing Economics*, 35(6):322–326.
- Mather, C & Cummings, E. 2017. Modelling digital knowledge transfer: nurse supervisor transforming learning at point of care to advance nursing practice. *Journal Informatics,* 1–14.

- Mathias, MS, Kukla, M, McGuire, AB & Bair, MD. 2016. How do patients with chronic pain benefit from a peer-supported pain self-management intervention? A qualitative investigation. *Pain Medicine*, 17:2247–2255.
- Matsuishi, Y, Hoshino, H, Shimojo, N, Enomoto, Y, Kido, T, Hoshino, T, Sumitani, M
 & Inoue. 2018. Verifying the validity and reliability of the Japanese version of the Face, Legs, Activity, Cry, Consolability (FLACC) Behavioral Scale. *PLOS One*, 1–9.
- Mazara, L, Zareel, MA, Gharib, A & Aljazzazi, HS. 2016. A study to assess the compliance rate with pain assessment and reassessment by nurses in ED, HGH – a quality improvement initiative. *Journal of Emergency Medicine, Trauma and Acute Care*, (2016):121.
- McCauley, M, Danna, VA, Mrema, D & van den Broek, N. 2018. "We know it's labour pain, so we don't do anything'': healthcare provider's knowledge and attitudes regarding the provision of pain relief during labour and after childbirth. *BMC Pregnancy and Childbirth,* 18(444):1–9.
- McComb, SA, Kirkpatrick, JM. 2015. Impact of pedagogical approaches on cognitive complexity and motivation to learn: comparing nursing and engineering undergraduate students. *Nursing Outlook*, 64:37–48.
- McDaniel, C, Caspersen, A, Crumpton, S, Galbraith, D, Hall, B, Huddleston, P, Puckett, M & Williams, R. 2021. Medical-surgical patients' and registered nurses' satisfaction and comprehensiveness of patient assessment using the clinically aligned pain assessment tool. *American Society for Pain Management Nursing*, 12(19):1–8.
- Mcdowell, J. 2015. Talk in feminised occupations: exploring male nurses' linguistic behaviour'. *Gender and Language*, 9(3):365–389.
- McGuire, DB, Kaiser, KS, Haisfield-Wolfe, ME & Iyamu, F. 2016. Pain assessment in non-communicative adult palliative care patients. *Nursing Clinics North America*, 51(3):397–431.

- McKenna, L, Connell, B, Bultler, AE & Lau, R. 2018. Learning style preference of Australian accelerated postgraduate pre-registration nursing students: a cross-sectional survey. *Nurse Education in Practice*, 28:280–284.
- McKim, CA. 2017. The value of mixed methods research: a mixed methods study. Journal of Mixed Methods Research, 11(2):202–222.
- Meissner, W, Huygen, F, Neugebauer, EAM, Osterbrink, J, Benhamou, D, Betteridge, N, Coluzzi, F, De Andres, J, Fawcett, W, Fletcher, D, Kalso, E, Kehlet, H, Morlion, B, Pérez, J, Pergolizzi, J, Schäfer, M. 2018. Management of acute pain in the postoperative setting: the importance of quality indicators. *Current Medical Research and Opinion*, 34(1):187–196.
- Meskell, P, Murphy, K, Shaw, D, Casey, D. 2014. Insights into the use and complexities of the policy Delphi technique. *Nurse Researcher*, 2(3):32–39.
- Miftah, R, Tilahun, W, Fantahun, A, Adulkadir, S & Gebrkirstos, K. 2017. Knowledge and factors associated with pain management for hospitalized children among nurses working in public hospitals in Mikelle city, North Ethiopia: cross-sectional study. *BioMed Central Research Notes*, 10(122):1–6.
- Miller, MT & Gearhart, GD. eds. 2021. Handbook of research on the changing role of college and university leadership. IGI Global.
- Minaya-Freire, A, Ramon-Aribau, A, Pou-Pujol, G, Fajula-Bonet, M & Subirana-Cascuberta, M. 2020. Facilitators, barriers, and solutions in pain management for older adults with dementia. *Pain Management Nursing*, 21:495–501.
- Mohanty, PC, Dash, M & Dash, M. 2017. Impact of trainee characteristics and organizational climate on training effectiveness with special reference to financial organization in India. *International Journal of Applied Business Economic Research*, 15(18):233–241.
- Moizes, JS & Wichert-Ana, L. 2017. Pain scales for use in non-communicating people in an experience reported: newborns, babies, children, adults and elderly. *International Journal of Psychology and Neuroscience*, 1–15.

- Morris, WW & Roques, CJ. 2018. Pain management in low- and middle-income countries. *British Journal of Anaesthesia*, 18(9):265–270.
- Mwachiro, M, Mwachiro, E, Wachu, MA, Koske, W, Thure, L, Parker, RK & White, RE. 2020. Assessing post-operative pain with self-report via the Jerrycan pain scale in rural Kenya. *World Journal of Surgery*, 44:3636–3642.
- Na-an, K, Chaipraist, K & Pukkeeree, P. 2017. Influences of workplace environment factors on employees' training transfer". *Industrial and Commercial Training*, 49(6):303–314.
- Nafukho, FM, Alfred, M, Chakraborty, M, Johnson, M & Cherrstrom, CA. 2017. Predicting workplace transfer of learning: a study of adult learners enrolled in a continuing professional education training program. *European Journal of Training and Development*, 41(4):327–353.
- Nair, S & Neil, MJE. 2013. Paediatric pain: physiology assessment and pharmacology. *Anaesthesia Tutorial of the Week,* 289:1–10.
- Naqib, D, Purvin, M, Prasad, R, Hanna, IM, Dimitri, S, Llufrio, A & Hanna, MN. 2018. Quality improvement to improve postoperative pain with a clinical pathway and nursing education program. *Pain Management Nursing*, 19(5):447–455.
- Natavio, T, McQuillen, E, Dietrich, MS, Wells, N, Rhoten, B, Vallerand, AH & Monroe, TB. 2020. A comparison of the pain assessment checklist for seniors with limited ability to communicate (PACSLAC) and pain assessment in advanced dementia scale (PAINAD). *Pain Management Nursing*, 21:502–509.
- Nazli, NNNN, Sipon, S, Zumrah, AR and Abdullah, S. 2015. The factors that influence the transfer of training in disaster preparedness training: A review. *Procedia-Social and Behavioral Sciences*, 192:54–58.
- Newton, PM., Da Silva, A & Berry, S. 2020. December. The case for pragmatic evidence-based higher education: A useful way forward?. In *Frontiers in Education*,5:583157. Frontiers Media SA.

- Ngoepe, M & Ngulube, P. 2013. An exploration of the role of record management in corporate governance in South Africa. *SA Journal of Information Management*, 15(2)1–8.
- Niderberger, M & Spranger, J. 2020. Delphi technique in health sciences: A map. *Frontiers in Public Health*, 8(457):1–10.
- Nik Nazli, N.N.N. & Sheikh Khairudin, S.M.H. 2018. The factors that influence transfer of training and its effect on organizational citizenship behaviour: Evidence from Malaysia civil defence force. *Journal of Workplace Learning*, 30(2):121–146.
- Noone, T & Seery, A. 2018. Critical thinking dispositions in undergraduate nursing students: a case study approach. *Nurse Education Today*, 68:203–207.
- Noormohammadpour, P, Kordi, M, Masournia, MA, Akbari-Fakhrabadi, M & Kordi, R. 2018. The role of a multi-step core stability exercise program in the treatment of nurses with chronic low back pain: a single-blinded randomized controlled trial. *Asian Spine Journal*, 12(3):490–502.
- Nowak, T, Neumann-Podczadka, A, Deskur-Smilecka, E, Styszyński & Wieczorowska-Tobis, K. 2018. *Clinical Interventions in Aging*, 13:1045–1051.
- O'Cathain, A, Croot, L, Duncan, E, Rousseau, N, Sworn, K, Turner, KM, Yardley, L & Hoddinott, P. 2019. Guidance on how to develop complex interventions to improve health and healthcare. *British Medical Journal Open*, 1–9.
- O'Donovan, J, O'Donovan, C, Kuhn, I, Sachs, SE & Winters, N. 2018. Ongoing training of community health workers in low-income and middle-income countries: a systematic scoping review of the literature. *British Medical Journal Open*, 1–10
- Obiedat.H & Al-Maaitah, El. 2020. Critique of the use of neonatal infant pain scale (NIPS). *Neonatal and Pediatric Medicine*, 6:1–5.
- Ogbolu, Y, Scrandis, DA & Fitzpatrick, G. 2018. Barriers and facilitators of care for diverse patients: nurse leader perspectives and nurse manager implications. *Journal of Nursing Management,* 26:3–10.

- Ojong, IN, Ojong-Alasia, M & Nlumanze, FF. 2014. Nurses' assessment and management of pain among surgical patients in a secondary health facility in Calabar Metropolis, Cross River State, Nigeria. *European Journal of Experimental Biology*, 4(1):315–320.
- Olsson, E, Ahi, H, Bengtsson, K, Vejayaram, DN, Norman, E, Bruschettini, M & Eriksson, M. 2021. The use and reporting of neonatal pain scales: a systematic review of randomized trials. *PAIN*, 162(2):353–360.
- Olusadum, NJ & Anulika, NJ. 2018. Impact of motivation on employee performance: a study of Alvan Ikoku Federal College of Education. *Journal of Management and Strategy*, 9(1):53–65.
- Öqvist, A & Malmström, M. 2017. What motivates the students? A study on the effects of teacher leadership and student self-efficacy. *International Journal of Leadership in Education*, 1–21.
- Osongo, LN. 2020. Barriers to cancer pain management among nurses in Kenya: a focused ethnography. *Pain Management Nursing*, 21:283–289.
- Oulton, JA & Caldwell, P. 2008. Nurses. International Council of Nurses, Geneva, Switzerland, 564–572.
- Panlican, AS, Pasay-An, EA, Gonzales, FM, Alreshidi, MS, Ibno, NL and Alenzi, SS. 2020. A survey on the knowledge and attitude on pain management. *Saudi Journal for Health Sciences*, 9(2):97–101.
- Patino, CM & Ferreira, JC. 2018. Inclusion and exclusion criteria in research studies: definitions and why they matter. *Jornal Brasileiro de Pneumologia*, 44(2):84– 84.
- Permana, JRA & Widagdo, I. 2019. Autogenic relaxation for postoperative caesarean section, pain in RSAD Kodam V Brawijaya. *Surabaya. International Conference of Kerta Cendekia Nursing Academy*, 1:12–18

- Peterson, A, Berggården, M, Schaller, AS & Larsson, S. 2019. Nurses' advocacy of clinical pain management in hospitals: a qualitative study. *Pain Management Nursing*, 2:133–139.
- Peterson, A, Carlfjord, S, Schaller, A, Gerdle, B & Larrsson, B. 2017. Using education and support strategies to improve the way nurses assess regular and transient pain-a quality improvement study of three hospitals. *Scandinavian Journal of Pain*, 16:15–21.
- Phillips, JL, Heneka, N, Hickman, L, Lam, L & Shaw, T. 2017. Can a complex online intervention improve cancer nurses' pain screening and assessment practices?
 Results from a multicenter, pre-post test pilot study. *Pain Management Nursing*, 18(2):75–89.
- Phillips, NM, Duke, MM & Weerasuriya, R. 2017. Questioning skills of clinical facilitators supporting undergraduate nursing students. *Journal of Clinical Nursing*, 26:4344–4352.
- Pilcher, J. 2016. Learning needs assessment. *Journal for Nurses in Professional Development,* 32(4):185–191.R
- Pilnick, A, Trusson, D, Beeke, S, O'Brien, Goldberg, S & Harwood, RH. 2018. Using conversation analysis to inform role play and simulated interaction in communications skills training for healthcare professionals: identifying avenues for further development through a scoping review. *BMC Medical Education*, 18(267):1–10.
- Ploeg, J, Davies, B, Edwards, N, Gifford, W & Miller PE 2007. Factors influencing bestpractice guideline implementation: lessons learned from administrators, nursing staff, and project leaders. *Worldviews on Evidence-Based Nursing, Sigma Theta Tau International,* 4(4):210–219.
- Pölkki, T, Korhonen, A, Laukkala, A. 2018. Nurses' perceptions of pain assessment and management practices in neonates: a cross-sectional survey. *Scandinavian Journal of Caring Sciences*, 32:725–733.

- Popowicz, H, Kwiecień-Jaguś, K, Olszewska, J, Mędrzycka-Dąbrowska, AW. 2020. Pain scales in neonates receiving mechanical ventilation in neonatal intensive care units systematic review. *Journal of Pain Research*, 13:1883–1897.
- Pretorius, A & Searle, JS. 2015. Barriers and enablers to emergency department nurses: management of patients' pain. *Pain Management Nursing*, 16(3):372– 379.
- Punshon, G, Maclaine, K, Trevatt, P, Radford, M, Shanley, O & Leary, A. 2019. Nursing pay by gender distribution in the UK: does the glass escalator still exist? *International Journal of Nursing Studies*, 93:21–29.
- Qureshi, QA, Bhutto, A & Memon, ZA. 2015. Factors affecting transfer of training at workplace: study of SSGC, Hyderabad. *IBT, Journal of Business Studies*, 11(1):102–118.
- Qureshi, QA, Bhutto, A & Tunio, RA. 2017. Factor affecting the transfer at the workplace: case study of SSGC Ltd, Pakistan. *International Journal of Academic Research in Business and Social Sciences*, 7(2):357–370.
- Rababa, M. 2017. The association of nurses' assessment and certainty to pain management and outcomes for nursing home residents in Jordan. *Geriatric Nursing*, 39(1):66-71.
- Rababa, M. 2018. Pain assessment in people with dementia: remaining controversies. *Global Journal of Health Science*, 10(5):62–69.
- Rafii, F, Ghezeljeh, TN & Nasrollah, S. 2019. Design and implementation of clinical competency evaluation for nursing students in medical-surgical wards. *Journal of Family Medicine and Primary Care*, 8(4):1408–1413.
- Rai, A, Gupta, GP & Kumar, P. 2017. Estimation of software development efforts using improved Delphi technique: a novel approach. *International Journal of Applied Engineering Research*, 12(12):3228–3236.
- Reinhold, S, Gegenfurtner, A & Lewalter, D. 2018. Social support and motivation to transfer as predictors of training transfer: testing full and partial mediation using

meta-analytic structural equation modelling. *International Training and Devel*opment, 1–14.

- Resnick, B, Boltz, M, Galik, E, Holmes, S, Vigne, E, Fix, S & Shijun, Z. 2019. Pain assessment, management, and impact among older adults in assisted living. *Pain Management Nursing*, 20:192–197.
- Rey, L, Pena, M & Neto, F. 2020. Protective resources for psychological well-being of adolescents. *Frontiers in psychology*, 11:541151.
- Rhame, K, Le, DT, Horner, A, Thomas, A, Foreman, B, Kreitzer, NP, Ngwenya, LB. 2019. Implementation of a neurotrauma hotline for post-hospital continuity of care. *Acta Neurologica Scandinavica*, 141:351–354.
- Roca, J, Reguant, M, Tort, G & Cancet, O. 2020. Developing reflective competence between simulation and clinical practice through a learning transference model: A qualitative study. *Nurse Education Today*, 92:1–6.
- Rockett, M, Vanstone, J, Chand, Waeland, D. 2017. A survey of acute pain services in the UK. *Anesthesia*, 72:1237–1242.
- Roets, L, Botma, Y & Grobler, C. 2016. Scholarship in nursing degree-prepared nurses versus diploma prepared. *Health SA Gesondheid*, 21:422–430.
- Rosa, WE. 2018. Transcultural pain management: theory, practice, and nurse-client partnerships. *Pain Management Nursing*, 19(1):23–33.
- Rose, L, Haslam, L, Dale, C, Knechtel, L & McGillion, M. 2013. Behavioural pain assessment tool for critically ill adults unable to self-report pain. *American Journal* of Critical Care, 22(3):246–254.
- Ross, JG. 2015. The effect of simulation training on baccalaureate nursing students competes in performing intramuscular injection. *Nursing Education Perspectives*, 36(1):48–49.

- Rostad, HM, Utne, I, Grove, EK, Puts, M & Halvorsrud, L. 2017. Measurement properties, feasibility and clinical utility of the Doloplus-2 pain scale in older adults with cognitive impairment: a systematic review. *BMC Geriatrics*, 17(1):1-28.
- Rouleau, G, Gagnon, MP, Côté, J, Payne-Gagnon, J, Hudson, E, Dubois, CA & Bouix-Picasso, J. 2019. Effects of e-learning in a continuing education context on nursing care: systematic review of systematic qualitative, quantitative, and mixedstudies reviews. *Journal of Medical Internet Research*, 21(10):1–19.
- Royal College of Nursing (RCN). 2018. RCN pain knowledge and skills framework for the nursing team. *Royal College of Nursing*, 1–15.
- Rozario, D. 2018. Optimization of communication in the surgical program via instant messaging, web-based surveys, newsletters, websites, smartphones and telemedicine: the experience of Oakville Trafalgar Memorial Hospital. *Canadian Journal of Surgery*, 61(4):E4–E6.
- Sacco, S, Braschinsky, M, Ducros, A, Lampi, C, Little, P, Van den Brink, AM, Pozo-Rosich, P, Reuter, U, De la Torre, E, Del Rio, MS, Sinclair, AJ, Katsarava, Z & Martelletti, P. 2020. European Headache Federation consensus on the definition of resistant and refractory migraine. *The Journal of Headache and Pain*, 21(76):1–13.
- Saelim, K, Chavananon, S, Ruangnapa, K, Prasertsan, P & Anuntaseree, W. 2019. Effectiveness of protocolized sedation utilizing the COMFORT-B scale in mechanically ventilated children in a pediatric intensive care unit. *Journal of Pediatric Intensive Care*, 8:156–163.
- Saha, FJ, Brüning, A, Barcelona, C, Büssing, A, Langhorst, J, Dobos, G, Lauche, R & Cramer, H. 2016. Integrative medicine for chronic pain. *Medicine*, 95(27):1–7.
- Salinas, D, Johnson, SC, Conrardy, JA, Adams, TL & Brown, JD. 2019. Sustaining nursing grounds through interdisciplinary teamwork and interorganizational partnership. *American Journal of Nursing*, 119(4):41–48.

- Santos, AM, Machado, RR, Ribeiro, CJN, Neto, JMM, Ribeiro, MCO & Menezes, MGV. 2018. Nursing students' knowledge about pain assessment. *British Journal of Pain*, 1(4):325–330.
- Sardo, S Galletta, M, Coni, E, Gonzalez, CIA, Piras, I, Pia, G, Evangelista, M, Musu, M & Finco, G. 2020. Nurses' behavior regarding pain treatment in an emergency department: a single-center observational study. *Journal of Pain Research*, 13:2355–2359.
- Sawhney, M, Wong, M, Luctkar-Flude, M, Jussaume, L, Eadie, C, Bowry, R & Wilson.
 R. 2018. Using simulation to enhance education regarding epidural analgesia for registered nurses. *Pain Management Nursing*, 19(3):246–255.
- Schaeffer, D, Gille, S. & Hurrelmann, K. 2020. Implementation of the national action plan health literacy in Germany—lessons learned. *International Journal of Environmental Research and Public Health*, 17(12):1–12.
- Schofield, P. 2018. The assessment of pain in older: UK national guidelines. *Age and Aging*, 47:i1–i22.
- Seixas-Moizes, J & Wichert-Ana, L. (2017). Pain scales for use in non-communicating people in an experience reported: Newborns, babies, children, adults and elderly. *International Journal of Psychology and Neuroscience*, 3(1):1-15.
- Shah, HR, Kodack, E & Walker, J. 2020. Continuing education activity neuropathic pain: a review of pathophysiology, presentation, and management. *Topics in Pain Management*, 35(8):1–12.
- Shipton, EE, Bate, F, Garrick, R, Steketee, C, Shipton, EA & Visser, EJ. 2018. Systematic review of pain medicine content, teaching, and assessment in medical school curricula internationally. *Pain Therapy*, 7:139–161.
- Shoqirat, N, Mahasneh, D, Singh, C & Hadid, L. 2019. Do surgical patients' characteristics and behaviours affect nurses' pain management decisions? A qualitative inquiry. *International Journal of Nursing Practice*, 25:1–8.

- Shorey, S, Ang, E, Yap, J, Ng, ED, Lau, ST & Chui, CK. 2019. A virtual counselling application using intelligence for communication skills training in nursing education: development study. *Journal of Medical Internet Research*, 21(10), 1–13.
- Shougaard, LMV, de Thurah, A, Bech, P, Hjollund, NH & Christiansen, DH. 2018. Testretest reliability and measurement error of the Danish WHO-5 well-being index in outpatients with epilepsy. *Health and Quality of Life Outcomes*, 16:1–6.
- Siagian, MV, Saragih, S & Sinaga, B. 2019. Development of learning materials oriented on problem-based learning model to improve students' mathematical problem-solving ability and metacognition ability. *International Electronic Journal of Mathematics Education*, 14(2):331–340.
- Siklander, P & Impiö, N. 2019. Common features of expertise in working life: implications for higher education. *Journal of Further and Higher Education*, 43(9):1239–1254.
- Singh, S. 2017. Trainee characteristics and transfer of training: effect of supervisory support (a study of public managers in Nepal). *Journal of Business and Management Research*, 2(1&2):1–13.
- Shrotryia, VK & Dhanda, U. 2019. Content validity of assessment instrument for employee engagement. *Sage Open*, 9(1):2158244018821751.
- Skinner, R, Nelson, RR, Chin, WW & Land, L. 2015. The Delphi method research strategy in studies of information systems. *Communications of the Association for Information Systems*, 37(2):31–63.
- Sluka, KA & George, SZ. 2021. A new definition of pain: update and implications for physical therapist and rehabilitation science. *Physical Therapy & Rehabilitation Journal*, 101:1–3.
- Sluka, KA, St. Marie, B & Strassels, SA. 2013. Core competencies for pain management: results of an interprofessional consensus summit. *Pain Medicine*, 14:971–798.

- Smeland, AH, Twycross, A, Lundeberg, S & Rustøen, T. 2018. Nurses' knowledge, attitudes and clinical practice in pediatric postoperative pain management. *Pain Management Nursing*, 19(6):585–598.
- Smith, T & Harvey, K. 2022. Psychometric properties of pain measurements for people living with dementia: a COSMIN systemic review. *European Geriatric Medicine*, 1–17.
- Snoek, M & Volman, M. 2014. The impact of the organizational transfer climate on the use of teacher leadership competencies developed in a post-initial Master's program. *Teacher and Teacher Education*, 37:91–100.
- Song, W, Eaton, LH, Gordon, DB, Hoyle, C and Doorenbos, AZ. 2015. Evaluation of evidence-based nursing pain management practice. *Pain Management Nurs-ing*, 16(4):456–463.
- Srimannarayana, M. 2016. An exploratory study of training transfer climate in India. International Journal of Business and Management, 11(8):263–272.
- Steyn, R. 2017. How many items are too many? An analysis of respondent disengagement when completing questionnaires. *African Journal of Hospitality, Tourism and Leisure,* 6(2):1–11.
- Stirling, BV & Alquraini, WA. 2017. Using VARK to assess students' learning style preferences: do they differ from other health professionals? *Journal of Taibah University Medical Sciences*, 12(2):125–130.
- Stites, M. 2013. Observational pain scales in critically ill adults. *Critical Care Nurse,* 33(3): 68–79.
- Stone, AL & Walker, LS. 2017. Adolescents' observations of parent pain behaviors: preliminary measure validation and test of social learning theory in pediatric chronic pain. *Journal of Pediatric Psychology*, 42(1):65–74.
- Storsveen, AM & Hall-Lord, ML. 2016. The CPOT a tool for pain assessment for intensive care patients. *Opphavstrett Sykepleien*, 1–11.

- Stuebe, AL, McKenzie, CP, Tucker, C, Tully, K, Bryant, K & Verbiest, S. 2018. Using electronic medical record data to quantify racial and ethnic differences in pain management following caesarean birth. *American Journal of Medicine*, 1–1.
- Stuns, KK & Heaslip, G. 2019. Effectiveness of humanitarian logistics training: the Finnish Red Cross (FRC) emergency response unit (ERU). *Journal of Humanitarian Logistics and Supply Chain Management*, 9(2):196–220.
- Subedi, D. 2016. Explanatory sequential mixed method design as the third research community of knowledge claim. *American Journal of Educational Research*, 4(7):570–577.
- Suleiman, W, Dassanayake, MS & Othman, AEA. 2016. Roles of Trainee Characteristics and Work Environment in Training Transfer: A Conceptual Extension of Baldwin and Ford Model with Job Attitude Factors in Nigerian Context. *Amity Journal of Training and Development*, 1(1):20–31.
- Suliman, M. and Aljezawi, M. 2018. Nurses' work environment: indicators of satisfaction. *Journal of Nursing Management*, 26(5):525–530.
- Sunderland, S, Yarnold, CH, Head, SJ, Osborn, JA, Purssell, A, Peel, JK & Schwarz, KW. 2016. Regional versus general anesthesia and the incidence of unplanned health care resource utilisation for postoperative pain after wrist fracture surgery. *Regional Anesthesia and Acute Pain*, 41:22–27.
- Suzuki, T. 2017. Does the combination use of two pain assessment tools have a synergistic effect? *Journal of Intensive Care*, 5(1):1–3.
- Svendsen, ED & Bjørk, IT. 2014. Experienced nurses' use of non-pharmacological approaches comprises more than relief from pain. *Journal of Pediatric Nursing*, 29:e19–E28.
- Sweet, L & Broadbent, J. 2017. Nursing students' perceptions of the qualities clinical facilitators that enhance learning. *Nurse Education in Practice*, 22:30–36.

- Taber, KS. 2018. The use of Cronbach's alpha when developing and reporting research instruments in science education. *Research Science Education*: 48:1273–1296.
- Taguma, M. 2018. Future education and skills 2030. Conceptual learning framework. Organisation for Economic Co-operation and Development, 1–29.
- Taherdoost, H. 2016. Validity and reliability of the research instrument; how to test the validation of a questionnaire/survey in research. *International Journal of Academic Research in Management*, 5(3):28–36.
- Takai, Y, Yamamoto-Mitani, N, Chiba, Y & Kato, A. 2014. Feasibility and clinical utility of the Japanese version of the Abbey Pain Scale in Japanese aged care. *Pain Management Nursing*, 15(2):439–448.
- Takai, Y, Yamamoto-Mitani, N, Ko, A & Heilemann, MV. 2014. Difference in pain measures by mini-mental state examination scores of residents in aged care facilities: examining the usability of the Abbey Pain Scale-Japanese. *Pain Management Nursing*, 15(1):236–245.
- Tegenborg, S, Fransson, P & Martinsson, L. 2020. Developing the APS-SE: translation and cultural adaptation of the Abbey Pain Scale (APS) to Swedish care context. *Research Square*, 1–12.
- Teo, K, Johnson, MH, Truter, S, Pandanaboyana, S & Windsor, JA. 2016. Pain assessment in chronic pancreatitis: a comparative review of methods. *Pancreatology*, 1–9.
- Terry, D, Peck, B, Carden, C, Perkins, AJ & Smith, A. 2020. Traversing the Funambulist's fine line between nursing and male identity: a systematic review of the factors that influence men as they seek to navigate the nursing profession. *European Journal of Investigation in Health, Psychology and Education*, 10(3):691–703.
- Testers, L, Gegenfurtner, A, Van Geel, R, Brand-Gruwel, S. 2019. From monocontextual to multicontextual transfer: organisational determinants to transfer generic

information literacy competences to multiple contexts. *Frontline Learning Research*, 7(1):23–42.

- Thirion, J, O'Riordan, MA & Stormorken, A. 2015. Revisiting the pieces of hurt pain assessment tool do the pieces matter? *Pediatric Pain Letter*, 17(1):1–3.
- Thomas, CM & Kellgren, M. 2017. Benner's novice to expert model: An application for simulation facilitators. *Nursing Science Quarterly*, 30(3):227-234.
- Thompson, K, Johnson, MI, Milligan, J & Briggs, M. 2018 Twenty-five years of pain education research what we have learned? Findings from a comprehensive scoping review of research into pre-registration pain education for health professionals. *PAIN*, 159(11):2146–2158.
- Tolley, JA, Michel, MA, Willams, AE & Renschler, JS. 2020. Peer support in the treatment of chronic pain in adolescents: a review of the literature and available resources. *Children*, 7(129):1–10.
- Tonhäuser, C & Bürker, L. 2016. Determinants of transfer of training: A comprehensive literature review. *International Journal for Research in Vocational Education and Training*, 3(2)127–165.
- Toode, K, Routasalo, P, Helminen, M & Suominen, T. 2015. Hospital nurses' work motivation. *Scandinavian Journal of Caring Sciences*, 29:248–257.
- Topham, D & Drew, D. 2017. Quality improvement project: replacing the numeric rating scale with a clinically aligned pain assessment (CAPA) tool. *Pain Management Nursing*, 18(6):363–371.
- Toumia, A. 2022. Delphi technique between the easy theoretic and the difficulty application in scientific research and future studies. *Technium Social Sciences Journal*, 37:749-759.
- Tse, MMY & Ho, SSK. 2014. Enhancing knowledge and attitudes in pain management: a pain management education program for nursing home staff. *Pain Management Nursing*, 15(1):2–11.

- Ufashingabire, CM, Nsereko, E, Njunwa, KJ, Brysiewicz, P. 2016. Knowledge and attitudes of nurses regarding pain in the intensive care unit patient in Rwanda. *Rwanda Journal Series F: Medicine and Health Sciences*, 3(1):21–26.
- Ugur, HG & Erci, B. 2019. The effect of home care for stroke patients and education of caregivers on the caregiver burden and quality of life. *Acta Clinica Croatia*, 58(2):321–332.
- UN-Habitat. 2017. Guidelines for drafting and implementation of the national action plan on housing, urban development and land management in transition economies on the basis of the Geneva UN charter on sustainable housing. *Nairobi UN-Habitat*, 1–28.
- Upadhyay, C, Cameron, K, Murphy, K & Battistella, M. 2014. Measuring pain in patients undergoing hemodialysis review of pain assessment tools. *Clinical Kidney Journal*, 7:367–372.
- Vaajoki, A. 2013. We have to take pain definition, pain management, and the results of non-pharmacological studies seriously. *Alternative and Integrative Medicine*, 2(7):1–2.
- Vael, A & Whitted, K. 2014: An educational intervention to improve pain assessment in preverbal children. *Pediatric Nursing*, 40(6):302–307.
- Van Rosse, F, de Bruijne, M, Suurmond, J, Essink-Bot, ML & Wagner, C. 2015. Language barriers and patient safety risks in hospital care. A mixed methods study. *International Journal of Nursing Studies*, 1–9.
- Varndell, W, Fry, M & Elliott, D. 2016. A systemic review of observational pain assessment instruments for use with nonverbal intubated critically ill adult patients in the emergency department: an assessment of their suitability and psychometric properties. *Journal of Clinical Nursing*, 26(1), Accepted:1–37.
- Veal, F, Williams, M, Bereznicki, L, Cummings, Thompson, P & Winzenberg, T. 2017. Barriers to optimal pain management in aged care facilities: an Australian qualitative study. *Pain Management Nursing*, 19(2):177–185.

- Verkuyl, M, Romaniuk, D, Atack, L & Mastrilli, P. 2017. Virtual gaming simulation for nursing education: an experiment. *Clinical Simulation in Nursing*, 13:238–244.
- Vrouva, S, Batistaki, C, Koutsioumpa, E, Kostopoulos, D, Stamoulis, E & Kostopanagiotou, G. 2016. The Greek version of Shoulder Pain and Disability Index (SPADI): translation, cultural adaptation, and validation in patients with rotator cuff tear. *Journal of Orthopaedics and Traumatology*, 17(4):315–326.
- Walter, JS. 2018. Global perspectives: making the shift from multiculturalism to culturally responsive teaching. *National Association for Music Education*, 31(2):24– 28.
- Wang, PF, Shen, LQ, Hong-Jun-Zhang, HJ, Li, BH & Ji, H. 2017. A nursing pain assessment and record information system. *Computers Informatics Nursing*, 35(12):647–652.
- Wariaghli, G, Allali, F, Berrada, K, Idrissi, Z, Hmamouch, I, Abouqal, R & Hajjaj-Hassouni, N. 2013. The patient acceptable symptom state of chronic musculoskeletal pain measured on a visual analog scale in Moroccan patients. *Pain Medicine*, 14:103–109.
- Warlow, TA & Hain, RDW. 2018. "Total pain" in children with severe neurological impairment. *Journal of Children*, 5(13):1–10.
- Waszak, DL, Mitchell, AM, Ren, D & Fennimore, LA. 2018. A quality improvement project to improve education provided by nurses to ED patients prescribed opioid analgesics at discharge. *Journal of Emergency Nursing*, 44(4):336–344.
- Watt-Watson, J, McGillion, M, Lax, L, Oskarsson, J, Hunter, J, MacLennan, C, Knickle, K & Victor, JC. 2019. Evaluating an innovative eLearning pain education interprofessional resource: a pre-post study. *Pain Medicine*, 20(1):37–49.
- Wei, H, Roberts, P, Strickler, J & Corbett, RW. 2018. Nurse leaders' strategies to foster nurse resilience. *Journal of Nursing Management*, 27:681–687.

- Weissman-Fogel, I, Roth, A, Natan-Raav, K & Lotan, M. 2015. Pain experience of adults with intellectual disabilities caregiver reports. *Journal of Intellectual Disability Research*, 59(10):914–924.
- Wen, MLY & Lin, DYC. 2014. How supportive transfer climate affects individual's motivation to training transfer. *International Journal of Learning & Development*, 4(1):83–97.
- Wen, MLY & Lin, DYC. 2014. Trainees' characteristics in training transfer: the relationship among self-efficacy, motivation to learn, motivation to transfer and training transfer. *International Journal of Human Resource Studies*, 4(1):114–129.
- Wenzel, R& Cordery, J. 2014. Training transfer research. A manager's guide and bibliography. Australian Institute of Management – Western Australia, Perth, 1– 79.
- Whicher, A, Harris, C Beverley, K & Swiatek. 2018. Design for circular economy: developing an action plan for Scotland. *Journal of Cleaner Production*, 172:3237– 3248.
- Wooldridge, S & Branney, J. 2020. Congruence between nurses' and patients' assessment of postoperative pain: a literature review. *British Journal of Nursing*, 29(4):212–220.
- Wooldridge, S & Branney, J. 2020. Congruence between nurses' and patients' assessment of postoperative pain: a literature review. *British Journal of Nursing*, 29(4):212–220.
- Yamitsky, D & Keefe, FJ. 2016. Access to *PAIN Reports* is now open: IASP inaugurates a new journal. *PAIN Reports*, e563:1–2.
- Yanaprasasart, P & Lüdi, G. 2018. Diversity and multilingual challenges in academic settings. International Journal of Bilingual Education and Bilingualism, 21(7):825–840.

- Yang, X, Zhao, X Tian, X & Xing, B. 2021. Effects of environment and posture on the concentration and achievement of students in mobile learning. *Interactive Learning Environments*, 29(3):400–413.
- Yang, Z, Zhou, Y, Chung, JWY, Tang, Q, Jiang, L & Wong, TKS. 2018. Challenge based learning nurtures creative thinking: an evaluative study. *Nurse Education Today*, 71:40–47.
- Yaripoor, S, Khalili, A, Joobakhsh, F, Talebiyanpour, MS & Almasi, S. 2016. Systematic review of pain assessment scales in newborns under maxillofacial surgery admitted to the surgical ward. *International Journal of Medical Research & Health Sciences*, 5(10):41–44.
- Yasin, RM, Nur, YFA, Ridzwan, CR, Bekri, RM, Arif, ARA, Mahazir, II & Ashikin, HT. 2014. Learning transfer at skill institutions and workplace environment: a conceptual framework. *Asian Social Science*, 10(1):179–188.
- Yeh, ST, Lee CY & Chou, WH. 2019. Recommendation of learning pain website for pain physicians in clinical practice. *Taiwan Journal of Pain*, 29(1):27–37.
- Yiengprugsawan, V & Steptoe, A. 2018. Impacts of persistent general and site-specific pain on activities of daily living and physical performance: a prospective analysis of the English longitudinal study of ageing. *Geriatric Gerontology International*, 18:1051–1057.
- Yilmaz, E, Karakaya, E, Baydur, H & Tekin, I. 2018. Effect of transcutaneous electrical nerve stimulation on postoperative pain and patient satisfaction. *Pain Management Nursing*, 1–9.
- Yoo, J, De Gange, JC, Kim, HJ & Oh, J. 2019. Development and evaluation of a webbased acute pain management education program for Korean registered nurses: a randomized controlled trial. *Nurse Education in Practice*, 38:7–13.
- Young, K, Godbold, R & Wood. 2018. How do student nurses learn to care? An analysis of pre-registration adult nursing practice assessment documents. *Nurse Education In Practice*, 28:168–174.

- Yu, Y, Hu, L, Chen, X, Ge, M, Zhu, H & Yan , Y. 2017. The impact of the predictive nursing education process on degree and quality of life for patients in the oncology department. *Iran Journal of Public Health*, 46(9):1231–1236.
- Yue, M, Zhang, M, Zhang, C, Jin, C. 2017. The effectiveness of concept mapping on development of critical thinking in nursing education: a systematic review and meta-analysis. *Nurse Education Today*, 52:87–94.
- Zainuddin, Z. 2018. Students' learning performances and perceived motivation in gamified flipped-class instruction. *Computer & Education*, 126:75–88.
- Zhoc, KCH, Chung, TSH & King, RB. 2018. Emotional intelligence (EI) and self-directed learning: examining their relation and contribution to better student learning outcomes in higher education. *British Educational Research Journal*, 44(6):982–1004.
- Zhu, H, Zeng, H, Zhang, H, Zhang, H, Wan, F, Guo, H & Zhang, C. 2018. The preferred learning styles utilizing VARK among nursing students with bachelor degrees and associate degrees in China. *Acta Paul Enferm*, 31(2):162–9.
- Zitzmann, NU, Matthisson, L, Ohla, H & Joda, T. 2020. Digital undergraduate education in dentistry: a systematic review. *International Journal of Environmental Research and Public Health*, 17(9):1–23.
- Ziyaeifard, M, Azarfarin, R, Zamani, K, Alizadehasi, A, Khalili, Y, Moradian, M, Koleini & Pouraliakbar, H. 2018. Imperative role education of ICU nurses regarding postoperative pain management after pediatric cardiac surgery. *Iranian Heart Journ*al, 19(2):13–19.
- Zizile, T. and Tendai, C. 2018. The importance of entrepreneurial competencies on the performance of women entrepreneurs in South Africa. *Journal of Applied Business Research (JABR)*, 34(2):223–236.

Internet Sources:

- General Authority for Statistics. 2016. Demography survey.Stats.gov.sa. From:<u>https://www.stats.gov.sa/sites/default/files/en-demographic-research-</u> 2016_2.pdf) [1-208]. (accessed 30 May 2018).
- 2012. *IASP Taxonomy*. Retrieved from: <u>https://www.iasp-pain.org/Taxonomy</u>. (accessed 20 October 2017).
- IASP. 2017. Assessment of musculoskeletal pain: experimental and clinical. Fact sheet no. 3. International Association of Study of Pain. From: <u>https://www.iasppain.org/Advocacy/Content.aspx?ItemNumber = 1101</u> (accessed 11 September 2020).
- R Core Team. 2018. *R a language and environment for statistical computing. R Foundation for Statistical Computing*. Vienna: Austria. Available from: <u>https://www.R-</u> <u>project.org/.(accessed 28 January 2019)</u>.
- Saudi Arabia. From: <u>https://en.wikipedia.org/wiki/Saudi Arabia (</u>accessed 31 May 2018).
- WHO. 2020. State of the world's nursing. World Health Organisation. From: <u>https://apps.who.int/nhwaportal/Sown/Files?name = SAU&lang = EN</u>. (accessed 28 August 2020).

Guidelines and Others:

- American Association of Critical-Care Nurses (AACN) Practice Alert. 2014. Assessing pain in the critically ill adult. *Critical Care Nurse*, 34(1):81–83.
- Joint Commission International (JCI). 2017. *Joint Commission International Accreditation standards for hospitals*. Oak Brook: Joint Commission Resources, Inc.
- Ministry of National Guard Health Affairs (MNGHA), Administrative policy and procedures. 2017. *Pain management* (APP No: 1430-07 of November 2017). Riyadh: NGHA Printing Press.

- Ministry of National Guard Health Affairs. 2019. *Strategic Plan 2019–2023.* Riyadh: MNGHA.
- 2017. King Abdulaziz Medical City Riyadh and Nursing Services. *Pain Management Program Quality Data.*
- Registered Nurses' Association of Ontario (RNAO). 2013. Clinical best practice guidelines. Assessment and management of pain. 3rd edition. Toronto, ON: Registered Nurses' Association of Ontario (RNAO).
- Saudi Central Board for Accreditation of Healthcare Institutions (CBAHI). 2015. *National hospital standards*. 3rd edition, Jeddah: Department of Public Relations.
- 2015. United Kingdom. Core standards for pain management services in the UK. *Royal College of Anesthetists*, 1–161.

ANNEXURES

ANNEXURE 1: Recruitment letter: Request of names list of registered nurses and clinical resource nurses from Nurse Managers (gatekeepers) KAMC

| Image: Services Reprint Provide Control of the reprint Provided Type of the control of the cont | | Ministry of National Guard Health Affairs | 回路回 |
|--|---|--|--|
| RYD-18-6020-67517 Date: G. 08 April 2018 H 22 Rajab 1439 Memorandum: Assistance with recruitment of participants TO : Nurse Managers (Medical, Surgical, Cardiac and Oby-Gynae In-patients Wards at KAMC-F. Thru : Ms. Nabeeha Tashkandi Director of Nursing Oncology FROM : Litaba Efraim Kolobe CRN, Pain Management Feam, Badge Number 47295 SUBJECT : Request for a list of Registered Nurses and Clinical Resource Nurses working in the wards at King Abdulaziz Medical City, Riyadh Study Title : Antexion Plan to enhance Transfer of Learning of Pain Management Competencies of nurses of Saudi Arabian teaching hospitals My name is Litaba Efraim Kolobe and I am a registered Doctoral student at the University of South Africa. The aim of my study is to develop an action plan to enhance transfer of learning of pain management Competencies of nurses in Saud Arabian teaching hospitals. I already received approval from Research Ethics Committee of the Department of Heath Studies, University of South Africa. Swell as an ethics approval certificate from King Abdullah International Medical Research Center (KAIMKC). Institutional Review Board (IRB) and Nursing Service Research Committee. I arder y received approval from Research ethics from King Abdullah International Medical Research Nurses who attended ward in-service trainings about pain management the past 3 years. | | Nursing Services | 11 A A A A A A A A A A A A A A A A A A |
| Date: G. 08 April 2018 H. 22 Rajab 1439 Annexure 1 Memorandum: Assistance with recruitment of participants TO : Nurse Managers (Medical, Surgical, Cardiac and Oby-Gynae In-patients Wards at KAMC-F. Thru : Ms. Nabeeha Tashkandi Associate Executive Director, Nursing Service, MMC-R ⁻¹ APR 2018 Ms. Andrea Doherty Director of Nursing Oncology FROM : Litaba Efraim Kolobe CRN, Pain Management Jeam, Badge Number 47295 SUBJECT : Request for a list of Registered Nurses and Clinical Resource Nurses working in the wards at King Abdulaziz Medical City, Riyadh Study Title : An action Plan to enhance Transfer of Learning of Pain Management Competencies of nurses of Saudi Arabian teaching hospitals My name is Litaba Efraim Kolobe and I am a registered Doctoral student at the University of South Africa. Th aim of my study is to develop an action plan to enhance transfer of learning of pain management Competencies of nurses in Saudi Arabian teaching hospitals. I aiready received approval from Research Ethics Committee of the Department of Health Studies, Universi of South Africa. Swell as an ethics approval certificate from King Adullah International Medical Researce Center (KAIMRC), Institutional Review Board (IRB) and Nursing Service Research Committee. In order for me to collect data, your assistance to inform Volunteer nurses to partake in the study wil b appreciated. Will you please send the attached rec | | | RYD-18-6020-67517 |
| Memorandum: Assistance with recruitment of participants TO : Nurse Managers (Medical, Surgical, Cardiac and Oby-Gynae In-patients Wards at KAMC-F Thru : Ms. Nabeeha Tashkandi Associate Executive Director, Nursing Service, MMC-R 1 APR 2018 Ms. Andrea Doherty Director of Nursing Oncology : FROM : Litaba Efraim Kolobe CRN, Pain Management Jeam, Badge Number 47295 SUBJECT : Request for a list of Registered Nurses and Clinical Resource Nurses working in the wards at King Abdulaziz Medical City, Riyadh Study Title : An action Plan to enhance Transfer of Learning of Pain Management Competencies of nurses of Saudi Arabian teaching hospitals My name is Litaba Efraim Kolobe and I am a registered Doctoral student at the University of South Africa. Tr aim of rmy study is to develop an action plan to enhance transfer of learning of pain manageme competencies of nurses in Saudi Arabian teaching hospitals. My name is Litaba Efraim Kolobe and I am a registered Doctoral student at the University of South Africa. Tr aim of rm to collect data, your assistance to inform volunteer nurses to partake in the study will the appreciated. Will you please send the attached recruitment letter and a list to provide names of all Registere Nurses and Clinical Resource Nurses interested in research and who meet the following inclusion/eligibili criteria. 1. Nurses who attended at least one pain management workshop within the past 12 wonths. 2. Nurses who attended ward in-service trainings | Date: <u>G</u> 04 <u>H</u> 22 | 8 April 2018 2 Rajab 1439 | Annexure 1 |
| TO : Nurse Managers (Medical, Surgical, Cardiac and Oby-Gynae In-patients Wards at KAMC-F (Medical, Surgical, Cardiac and Oby-Gynae In-patients Wards at KAMC-F Thru : Ms. Nabeeha Tashkandi Associate Executive Director, Nursing Service, MMC-F 1 APR 2018 Ms. Andrea Doherty Director of Nursing Oncology FROM : Litaba Efraim Kolobe CRN, Pain Management Team, Badge Number 47295 SUBJECT : Request for a list of Registered Nurses and Clinical Resource Nurses working in the wards at King Abdulaziz Medical City, Riyadh Study Title : An action Plan to enhance Transfer of Learning of Pain Management Competencies of nurses of Saudi Arabian teaching hospitals My name is Litaba Efraim Kolobe and I am a registered Doctoral student at the University of South Africa. Th aim of my study is to develop an action plan to enhance transfer of learning of pain management Competencies of nurses in Saudi Arabian teaching hospitals. Ialeady received approval from Research Ethics Committee of the Department of Health Studies, Universi of South Africa as well as an ethics approval certificate from King Abdullah International Medical Researc Center (KAIMC), Institutional Review Board (IRB) and Nursing Service Research Committee. In order for me to collect data, your assistance to inform volunteer nurses to partake in the study will b appreciated. Will you please send the attached recruitment letter and a list to provide names of all Register Nurses and Clinical Resource Nurses interested in research and who meet the following inclusion/eligibili criteria:- | Me | morandum: Assistance with recruitment of | participants |
| Invalidation Ms. Nabeeha Tashkandi Associate Executive Director, Nursing Service, KMMC-R1 APR 2018 Ms. Andrea Doherty Director of Nursing Oncology Ms. Andrea Doherty Director of Nursing Oncology FROM : Litaba Efraim Kolobe CRN, Pain Management Team, Badge Number 47295 SUBJECT : Request for a list of Registered Nurses and Clinical Resource Nurses working in the wards at King Abdulaziz Medical City, Riyadh Study Title : An action Plan to enhance Transfer of Learning of Pain Management Competencies of nurses of Saudi Arabian teaching hospitals My name is Litaba Efraim Kolobe and I am a registered Doctoral student at the University of South Africa. Th aim of my study is to develop an action plan to enhance transfer of learning of pain manageme competencies of nurses in Saudi Arabian teaching hospitals. I already received approval from Research Ethics Committee of the Department of Heatth Studies, Universi of South Africa as well as an ethics approval certificate from King Abdullah International Medical Researc Center (KAIMRC), Institutional Review Board (IRB) and Nursing Service Research Committee. In order for me to collect data, your assistance to inform volunteer nurses to partake in the study will ta appreciated. Will you please send the attached recruitment letter and a list to provide names of all Registere Nurses who attended at least one pain management workshop within the past 3 years. 1. Nurses who attended at least one pain management workshop within the past 3 years. 2. Nurses who attended at least one pain management workshop within the past 3 years. <td>то</td> <td>: Nurse Managers</td> <td>- In a stimute Manda at KAMO I</td> | то | : Nurse Managers | - In a stimute Manda at KAMO I |
| Thru : Ms. Nabeeha Tashkandi Associate Executive Director, Nursing Service, KAMC-R 1 APR 2018 Associate Executive Director, Nursing Service, KAMC-R 1 APR 2018 Director of Nursing Oncology Director of Nursing Oncology FROM : Litaba Efraim Kolobe CRN, Pain Management Team, Badge Number 47295 SUBJECT : Request for a list of Registered Nurses and Clinical Resource Nurses working in the wards at King Abdulaziz Medical City, Riyadh Study Title : An action Plan to enhance Transfer of Learning of Pain Management Competencies of nurses of Saudi Arabian teaching hospitals My name is Litaba Efraim Kolobe and I am a registered Doctoral student at the University of South Africa. The aim of my study is to develop an action plan to enhance transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals. I already received approval from Research Ethics Committee of the Department of Health Studies, University of South Africa as well as an ethics approval certificate from King Abdulah International Medical Researc Center (KAIMRC), Institutional Review Board (IRB) and Nursing Service Research Committee. In order for me to collect data, your assistance to inform volunteer nurses to partake in the study will be appreciated. Will you please send the attached recruitment letter and a list to provide names of all Registere Nurses and Clinical Resource Nurses interested in research and who meet the following inclusion/eligibili criteria. Nurses who attended at least one pain management workshop within the past 3 years. Nurses who attended at least one pain management workshop within the past 3 years. Participants willing and comfortable to be interviewed in English. Clinical facilitators (clinical resource nurses) who educates nurses about pain management in thos nursing care divisions mentioned above. I will appreciate the list of names and contact details from those who volunteer to participa | | (Medical, Surgical, Cardiac and Oby-Gyna | e in-patients wards at KAMC-F |
| Ms. Andrea Doherty Director of Nursing Oncology FROM : Litaba Efraim Kolobe CRN, Pain Management Team, Badge Number 47295 SUBJECT : Request for a list of Registered Nurses and Clinical Resource Nurses working in the wards at King Abdulaziz Medical City, Riyadh Study Title : An action Plan to enhance Transfer of Learning of Pain Management Competencies of nurses of Saudi Arabian teaching hospitals My name is Litaba Efraim Kolobe and I am a registered Doctoral student at the University of South Africa. Th aim of my study is to develop an action plan to enhance transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals. I already received approval from Research Ethics Committee of the Department of Health Studies, Universit of South Africa as well as an ethics approval certificate from King Abdullah International Medical Researce Center (KAIMRC), Institutional Review Board (IRB) and Nursing Service Research Committee. In order for me to collect data, your assistance to inform volunteer nurses to partake in the study will b appreciated. Will you please send the attached recruitment letter and a list to provide names of all Registere Nurses and Clinical Resource Nurses interested in research and who meet the following inclusion/eligibili criteria:. 1. Nurses who attended at least one pain management workshop within the past 3 years. 2. Nurses who attended at least one pain management workshop within the past 3 years. 3. Participants willing and comfortable to be interviewed in English. 4. Clinical facilitators (clinical reso | Thru | : Ms. Nabeeha Tashkandi Associate Executive Director, Nursing Se | Vice, KAMC- & 1 APR 2018 |
| FROM : Litaba Efraim Kolobe CRN, Pain Management Learn, Badge Number 47295 SUBJECT : Request for a list of Registered Nurses and Clinical Resource Nurses working in the wards at King Abdulaziz Medical City, Riyadh Study Title : An action Plan to enhance Transfer of Learning of Pain Management Competencies of nurses of Saudi Arabian teaching hospitals My name is Litaba Efraim Kolobe and I am a registered Doctoral student at the University of South Africa. The aim of my study is to develop an action plan to enhance transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals. I already received approval from Research Ethics Committee of the Department of Health Studies, Universi of South Africa as well as an ethics approval certificate from King Abdullah International Medical Researc Center (KAIMRC), Institutional Review Board (IRB) and Nursing Service Research Committee. In order for me to collect data, your assistance to inform volunteer nurses to partake in the study will b appreciated. Vill you please send the attached recruitment letter and a list to provide names of all Registere Nurses and Clinical Resource Nurses interested in research and who meet the following inclusion/eligibili criteria. Nurses who attended at least one pain management workshop within the past 3 years. Nurses who attended at least one pain management workshop within the past 12 months. Participants willing and comfortable to be interviewed in English. Clinical facilitators (clinical resource nurses) who educates nurses about pain management in thos nursing care divisions mentioned above. I will appreciate the list of names and contact details from those who volunteer to participate to enable me i meet them, do a selection and then I will provide the selected ones with the questionnaire that they need to to complete. They will be asked to post the completed questionnaires in an envelope that will be p | | Ms. Andrea Doherty | |
| CRN, Pain Management Team, Badge Number 47295 SUBJECT : Request for a list of Registered Nurses and Clinical Resource Nurses working in the wards at King Abdulaziz Medical City, Riyadh Study Title : An action Plan to enhance Transfer of Learning of Pain Management Competencies of nurses of Saudi Arabian teaching hospitals My name is Litaba Efraim Kolobe and I am a registered Doctoral student at the University of South Africa. Th aim of my study is to develop an action plan to enhance transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals. I already received approval from Research Ethics Committee of the Department of Health Studies, Universi of South Africa as well as an ethics approval certificate from King Abdullah International Medical Researc Center (KAIMRC), Institutional Review Board (IRB) and Nursing Service Research Committee. In order for me to collect data, your assistance to inform volunteer nurses to partake in the study will b appreciated. Will you please send the attached recruitment letter and a list to provide names of all Registere Nurses who attended at least one pain management workshop within the past 3 years. Nurses who attended at least one pain management workshop within the past 12 months. Names who attended at least one pain management workshop within the past 12 months. Chical facilitators (clinical resource nurses) who educates nurses about pain management in thos nursing care divisions mentioned above. I will appreciate the list of names and contact details from those who volunteer to participate to enable me 1 meet them, do a selection and then I will provide the selected ones with the questionnaire that they need to b complete. They will be asked to post the completed questionnaires in an envelope that will be provided for th purpose. I appreciate your time and willingness to assist. Yours Sincerely Litaba Efraim Kolobe | FROM | Litaba Efraim Kolobe | |
| SUBJECT : Request for a list of Registered Nurses and Clinical Resource Nurses working in the wards at King Abdulaziz Medical City, Riyadh Study Title : An action Plan to enhance Transfer of Learning of Pain Management Competencies of nurses of Saudi Arabian teaching hospitals My name is Litaba Efraim Kolobe and I am a registered Doctoral student at the University of South Africa. The aim of my study is to develop an action plan to enhance transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals. I already received approval from Research Ethics Committee of the Department of Health Studies, Universi of South Africa as well as an ethics approval certificate from King Abdullah International Medical Research Center (KAIMRC), Institutional Review Board (IRB) and Nursing Service Research Committee. In order for me to collect data, your assistance to inform volunteer nurses to partake in the study will be appreciated. Will you please send the attached recruitment letter and a list to provide names of all Registere Nurses and Clinical Resource Nurses interested in research and who meet the following inclusion/eligibilit criteria: Nurses who attended at least one pain management workshop within the past 3 years. Nurses who attended at least one pain management workshop within the past 12 months. Participants willing and comfortable to be interviewed in English. Clinical facilitators (clinical resource nurses) who educates nurses about pain management in thos nursing care divisions mentioned above. I will appreciate the list of names and contact details from those who volunteer to participate to enable me t meet them, do a selection and then I will provide the selected ones with the questionnaire that they need to b complete. They will be asked to post the completed questionnaires in an envelope that will be provided for th purpose. I appreciate your time and willingness to assist. | | CRN, Pain Management Team, Badge Nu | mber 47295 |
| Study Title : An action Plan to enhance Transfer of Learning of Pain Management Competencies of nurses of Saudi Arabian teaching hospitals My name is Litaba Efraim Kolobe and I am a registered Doctoral student at the University of South Africa. Th aim of my study is to develop an action plan to enhance transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals. I already received approval from Research Ethics Committee of the Department of Health Studies, Universi of South Africa as well as an ethics approval certificate from King Abdullah International Medical Research Center (KAIMRC), Institutional Review Board (IRB) and Nursing Service Research Committee. In order for me to collect data, your assistance to inform volunteer nurses to partake in the study will b appreciated. Will you please send the attached recruitment letter and a list to provide names of all Registere Nurses and Clinical Resource Nurses interested in research and who meet the following inclusion/eligibili criteria:- Nurses who attended at least one pain management workshop within the past 3 years. Nurses who attended ward in-service trainings about pain management the past 12 months. Participants willing and comfortable to be interviewed in English. Clinical facilitators (clinical resource nurses) who educates nurses about pain management in thos nursing care divisions mentioned above. I will appreciate the list of names and contact details from those who volunteer to participate to enable me for meet them, do a selection and then I will provide the selected ones with the questionnaire that they need to b complete. They will be asked to post the completed questionnaires in an envelope that will be provided for th purpose. I appreciate your time and willingness to assist. Yours Sincerely Litaba Efraim Kolobe | SUBJECT | : Request for a list of Registered Nurses | and Clinical Resource |
| I already received approval from Research Ethics Committee of the Department of Health Studies, Universion South Africa as well as an ethics approval certificate from King Abdullah International Medical Research Center (KAIMRC), Institutional Review Board (IRB) and Nursing Service Research Committee. In order for me to collect data, your assistance to inform volunteer nurses to partake in the study will the appreciated. Will you please send the attached recruitment letter and a list to provide names of all Registere Nurses and Clinical Resource Nurses interested in research and who meet the following inclusion/eligibilit criteria:- Nurses who attended at least one pain management workshop within the past 3 years. Nurses who attended ward in-service trainings about pain management the past 12 months. Participants willing and comfortable to be interviewed in English. Clinical facilitators (clinical resource nurses) who educates nurses about pain management in thos nursing care divisions mentioned above. I will appreciate the list of names and contact details from those who volunteer to participate to enable me meet them, do a selection and then I will provide the selected ones with the questionnaire that they need to the complete. They will be asked to post the completed questionnaires in an envelope that will be provided for the purpose. I appreciate your time and willingness to assist. Yours Sincerely Litaba Efraim Kolobe | | Competencies of nurses of Saudi Arabian teaching | of Pain Management ng hospitals |
| Center (KAIMRC), Institutional Review Board (IRB) and Nursing Service Research Committee. In order for me to collect data, your assistance to inform volunteer nurses to partake in the study will the appreciated. Will you please send the attached recruitment letter and a list to provide names of all Registeres. Nurses and Clinical Resource Nurses interested in research and who meet the following inclusion/eligibilit criteria:- Nurses who attended at least one pain management workshop within the past 3 years. Nurses who attended ward in-service trainings about pain management the past 12 months. Participants willing and comfortable to be interviewed in English. Clinical facilitators (clinical resource nurses) who educates nurses about pain management in thos nursing care divisions mentioned above. I will appreciate the list of names and contact details from those who volunteer to participate to enable me meet them, do a selection and then I will provide the selected ones with the questionnaire that they need to t complete. They will be asked to post the completed questionnaires in an envelope that will be provided for the purpose. | My name is Li aim of my s | Competencies of nurses of Saudi Arabian teaching itaba Efraim Kolobe and I am a registered Doctoral studen itudy is to develop an action plan to enhance transf | of Pain Management ng hospitals t at the University of South Africa. The er of learning of pain manageme |
| In order for me to collect data, your assistance to inform volunteer nurses to partake in the study will the appreciated. Will you please send the attached recruitment letter and a list to provide names of all Registeres. Nurses and Clinical Resource Nurses interested in research and who meet the following inclusion/eligibilit criteria:- Nurses who attended at least one pain management workshop within the past 3 years. Nurses who attended ward in-service trainings about pain management the past 12 months. Participants willing and comfortable to be interviewed in English. Clinical facilitators (clinical resource nurses) who educates nurses about pain management in thos nursing care divisions mentioned above. I will appreciate the list of names and contact details from those who volunteer to participate to enable me i meet them, do a selection and then I will provide the selected ones with the questionnaire that they need to be complete. They will be asked to post the completed questionnaires in an envelope that will be provided for the purpose. I appreciate your time and willingness to assist. | My name is Li aim of my s competencies I already rece of South Afric | Competencies of nurses of Saudi Arabian teachin itaba Efraim Kolobe and I am a registered Doctoral studen itudy is to develop an action plan to enhance transf of nurses in Saudi Arabian teaching hospitals. itived approval from Research Ethics Committee of the De a as well as an ethics approval certificate from King Ab | of Pain Management ng hospitals t at the University of South Africa. Ther of learning of pain manageme partment of Health Studies, Universidullah International Medical Researd |
| Nurses who attended at least one pain management workshop within the past 3 years. Nurses who attended ward in-service trainings about pain management the past 12 months. Participants willing and comfortable to be interviewed in English. Clinical facilitators (clinical resource nurses) who educates nurses about pain management in thos nursing care divisions mentioned above. I will appreciate the list of names and contact details from those who volunteer to participate to enable me meet them, do a selection and then I will provide the selected ones with the questionnaire that they need to be complete. They will be asked to post the completed questionnaires in an envelope that will be provided for the purpose. I appreciate your time and willingness to assist. Yours Sincerely Litaba Efraim Kolobe | My name is Li aim of my s competencies I already rece of South Afric Center (KAIM | Competencies of nurses of Saudi Arabian teaching itaba Efraim Kolobe and I am a registered Doctoral studen tudy is to develop an action plan to enhance transf of nurses in Saudi Arabian teaching hospitals. ived approval from Research Ethics Committee of the De ca as well as an ethics approval certificate from King Ab RC), Institutional Review Board (IRB) and Nursing Service | at the University of South Africa. The r of learning of pain management partment of Health Studies, Universi dullah International Medical Research Research Committee. |
| Nurses who attended ward in-service trainings about pain management the past 12 months. Participants willing and comfortable to be interviewed in English. Clinical facilitators (clinical resource nurses) who educates nurses about pain management in thos nursing care divisions mentioned above. I will appreciate the list of names and contact details from those who volunteer to participate to enable mere them, do a selection and then I will provide the selected ones with the questionnaire that they need to be complete. They will be asked to post the completed questionnaires in an envelope that will be provided for the purpose. I appreciate your time and willingness to assist. Yours Sincerely Litaba Efraim Kolobe | My name is Li aim of my s competencies I already rece of South Afric Center (KAIM In order for r appreciated. V Nurses and C | Competencies of nurses of Saudi Arabian teaching itaba Efraim Kolobe and I am a registered Doctoral studen itudy is to develop an action plan to enhance transf of nurses in Saudi Arabian teaching hospitals. inved approval from Research Ethics Committee of the De ca as well as an ethics approval certificate from King Ab RC), Institutional Review Board (IRB) and Nursing Service ne to collect data, your assistance to inform volunteer r Will you please send the attached recruitment letter and a Clinical Resource Nurses interested in research and who | of Pain Management ing hospitals t at the University of South Africa. The or of learning of pain manageme partment of Health Studies, Universi dullah International Medical Research Research Committee. urses to partake in the study will the ist to provide names of all Registered meet the following inclusion/eligibili |
| 3. Participants willing and controtable to be interviewed in English. 4. Clinical facilitators (clinical resource nurses) who educates nurses about pain management in thos nursing care divisions mentioned above. I will appreciate the list of names and contact details from those who volunteer to participate to enable me i meet them, do a selection and then I will provide the selected ones with the questionnaire that they need to b complete. They will be asked to post the completed questionnaires in an envelope that will be provided for th purpose. I appreciate your time and willingness to assist. Yours Sincerely Litaba Efraim Kolobe | My name is Li aim of my s competencies I already rece of South Afric Center (KAIM In order for r appreciated. V Nurses and C criteria:- 1. Nurse | Competencies of nurses of Saudi Arabian teaching itaba Efraim Kolobe and I am a registered Doctoral studen itudy is to develop an action plan to enhance transf of nurses in Saudi Arabian teaching hospitals. ived approval from Research Ethics Committee of the De as as well as an ethics approval certificate from King Ab RC), Institutional Review Board (IRB) and Nursing Service ne to collect data, your assistance to inform volunteer r Will you please send the attached recruitment letter and a Clinical Resource Nurses interested in research and who es who attended at least one pain management workshop | of Pain Management ng hospitals t at the University of South Africa. The or of learning of pain manageme partment of Health Studies, Universidullah International Medical Research Research Committee. urses to partake in the study will the ist to provide names of all Registered meet the following inclusion/eligibility within the past 3 years. |
| nursing care divisions mentioned above. I will appreciate the list of names and contact details from those who volunteer to participate to enable me meet them, do a selection and then I will provide the selected ones with the questionnaire that they need to to complete. They will be asked to post the completed questionnaires in an envelope that will be provided for the purpose. I appreciate your time and willingness to assist. Yours Sincerely Litaba Efraim Kolobe | My name is Li aim of my s competencies I already rece of South Afric Center (KAIM In order for r appreciated. V Nurses and C criteria:- 1. Nurse 2. Nurse | Competencies of nurses of Saudi Arabian teaching itaba Efraim Kolobe and I am a registered Doctoral studen study is to develop an action plan to enhance transf of nurses in Saudi Arabian teaching hospitals. Seved approval from Research Ethics Committee of the De ca as well as an ethics approval certificate from King Ab RC), Institutional Review Board (IRB) and Nursing Service ne to collect data, your assistance to inform volunteer r Will you please send the attached recruitment letter and a Clinical Resource Nurses interested in research and who es who attended at least one pain management workshop es who attended ward in-service trainings about pain mana | of Pain Management ing hospitals t at the University of South Africa. The or of learning of pain manageme partment of Health Studies, Universi dullah International Medical Researd Research Committee. urses to partake in the study will the ist to provide names of all Registered meet the following inclusion/eligibility within the past 3 years. gement the past 12 months. |
| I will appreciate the list of names and contact details from those who volunteer to participate to enable me meet them, do a selection and then I will provide the selected ones with the questionnaire that they need to t complete. They will be asked to post the completed questionnaires in an envelope that will be provided for th purpose. I appreciate your time and willingness to assist. Yours Sincerely Litaba Efraim Kolobe | My name is Li aim of my s competencies I already rece of South Afric Center (KAIM In order for r appreciated. V Nurses and C criteria:- 1. Nurse 2. Nurse 3. Partie 4. Clinic | Competencies of nurses of Saudi Arabian teaching itaba Efraim Kolobe and I am a registered Doctoral studen itudy is to develop an action plan to enhance transf of nurses in Saudi Arabian teaching hospitals. ived approval from Research Ethics Committee of the De as as well as an ethics approval certificate from King Ab RC), Institutional Review Board (IRB) and Nursing Service ne to collect data, your assistance to inform volunteer r Will you please send the attached recruitment letter and a Clinical Resource Nurses interested in research and who es who attended at least one pain management workshop es who attended ward in-service trainings about pain mana cipants willing and comfortable to be interviewed in English cal facilitators (clinical resource nurses) who educates nur | of Pain Management ng hospitals t at the University of South Africa. The or of learning of pain manageme partment of Health Studies, Universidullah International Medical Research Research Committee. urses to partake in the study will the ist to provide names of all Registered meet the following inclusion/eligibility within the past 3 years. gement the past 12 months. ses about pain management in those |
| I appreciate your time and willingness to assist. Yours Sincerely Litaba Efraim Kolobe | My name is Li aim of my s competencies of South Afric Center (KAIM In order for r appreciated. \ Nurses and C criteria:- 1. Nurse 2. Nurse 3. Partia 4. Clinic nursi | Competencies of nurses of Saudi Arabian teaching itaba Efraim Kolobe and I am a registered Doctoral studen to drurses in Saudi Arabian teaching hospitals. wived approval from Research Ethics Committee of the Dec as as well as an ethics approval certificate from King Ab RC), Institutional Review Board (IRB) and Nursing Service ne to collect data, your assistance to inform volunteer of Will you please send the attached recruitment letter and a Clinical Resource Nurses interested in research and who es who attended at least one pain management workshop es who attended ward in-service trainings about pain mana cipants willing and comfortable to be interviewed in English cal facilitators (clinical resource nurses) who educates nu- ng care divisions mentioned above. | of Pain Management ing hospitals t at the University of South Africa. The or of learning of pain manageme partment of Health Studies, Universi dullah International Medical Researd Research Committee. urses to partake in the study will the ist to provide names of all Registered meet the following inclusion/eligibili within the past 3 years. gement the past 12 months. ses about pain management in those |
| Yours Sincerely Litaba Efraim Kolobe | My name is Li aim of my s competencies I already rece of South Afric Center (KAIM In order for r appreciated. V Nurses and C criteria:- 1. Nurse 2. Nurse 3. Partic 4. Clinic nursi I will apprecia meet them, do complete. The purpose. | Competencies of nurses of Saudi Arabian teaching itaba Efraim Kolobe and I am a registered Doctoral studen study is to develop an action plan to enhance transf of nurses in Saudi Arabian teaching hospitals. sived approval from Research Ethics Committee of the De ca as well as an ethics approval certificate from King Ab RC), Institutional Review Board (IRB) and Nursing Service ne to collect data, your assistance to inform volunteer of Will you please send the attached recruitment letter and a Clinical Resource Nurses interested in research and who es who attended at least one pain management workshop es who attended ward in-service trainings about pain mana cipants willing and comfortable to be interviewed in English cal facilitators (clinical resource nurses) who educates nur ng care divisions mentioned above. | of Pain Management ng hospitals t at the University of South Africa. The r of learning of pain manageme partment of Health Studies, Universi dullah International Medical Researd Research Committee. urses to partake in the study will b ist to provide names of all Registere meet the following inclusion/eligibil within the past 3 years. gement the past 12 months. ses about pain management in those bunteer to participate to enable me the questionnaire that they need to b nenvelope that will be provided for the |
| Litaba Efraim Kolobe | My name is Li aim of my s competencies I already rece of South Afric Center (KAIM In order for r appreciated. V Nurses and C criteria:- 1. Nurse 3. Partic 4. Clinic nursi I will apprecia meet them, de complete. The purpose. | Competencies of nurses of Saudi Arabian teaching itaba Efraim Kolobe and I am a registered Doctoral student study is to develop an action plan to enhance transf of nurses in Saudi Arabian teaching hospitals. Sived approval from Research Ethics Committee of the Deve as as well as an ethics approval certificate from King Ab RC), Institutional Review Board (IRB) and Nursing Services ne to collect data, your assistance to inform volunteer of Will you please send the attached recruitment letter and a Clinical Resource Nurses interested in research and who es who attended at least one pain management workshop es who attended ward in-service trainings about pain mana- cipants willing and comfortable to be interviewed in English cal facilitators (clinical resource nurses) who educates nur- ing care divisions mentioned above. | of Pain Management ing hospitals at the University of South Africa. The of learning of pain manageme partment of Health Studies, Universi- dullah International Medical Researce Research Committee. urses to partake in the study will the ist to provide names of all Registered meet the following inclusion/eligibile within the past 3 years. gement the past 12 months. ses about pain management in those blunteer to participate to enable me the questionnaire that they need to be envelope that will be provided for the |
| | My name is Li aim of my s competencies I already rece of South Afric Center (KAIM In order for r appreciated. V Nurses and C criteria:- 1. Nurse 2. Nurse 3. Partia 4. Clinic nursi I will appreciat meet them, do complete. The purpose. I appreciate y Yours Sincer | Competencies of nurses of Saudi Arabian teaching itaba Efraim Kolobe and I am a registered Doctoral studen study is to develop an action plan to enhance transf of nurses in Saudi Arabian teaching hospitals. sived approval from Research Ethics Committee of the De ca as well as an ethics approval certificate from King Ab RC), Institutional Review Board (IRB) and Nursing Service ne to collect data, your assistance to inform volunteer of Will you please send the attached recruitment letter and a Clinical Resource Nurses interested in research and who es who attended at least one pain management workshop es who attended ward in-service trainings about pain mana cipants willing and comfortable to be interviewed in English cal facilitators (clinical resource nurses) who educates nur ng care divisions mentioned above. | of Pain Management ing hospitals t at the University of South Africa. The or of learning of pain manageme partment of Health Studies, Universi dullah International Medical Researd Research Committee. urses to partake in the study will the ist to provide names of all Registered meet the following inclusion/eligibil within the past 3 years. gement the past 12 months. ses about pain management in those bunteer to participate to enable me the questionnaire that they need to the newelope that will be provided for the |

ANNEXURE 2: Recruitment letter: Request of names list of registered nurses and clinical resource nurses from Nurse Managers (gatekeepers) KASCH

| | King Abdulaziz Medical City in R Nursing Services | iyadh |
|------------------|--|---|
| Date: G 0 H 2 | April 2018 Rajab 1439 | Annexure 2 |
| Me | norandum: Assistance with | recruitment of participants |
| то | : Nurse Managers (Pediatrics and Adult In-) | patients Wards at KASCH) |
| Thru | : Ms. Angela Caswell Associate Executive Dir | ector, Nursing Service, KASCH-R |
| | Director of Nursing Onco | logy |
| FROM | : Litaba Efraim Kolobe CRN, Pain Managemen | Team, Badge Number 47295 |
| SUBJECT | : Request for a list of R Nurses working in the Hospital, Riyadh | egistered Nurses and Clinical Resource wards at King Abdullah Specialist Childro |
| Study Title | : An action Plan to enhance T Competencies of Nurses in | ransfer of Learning of Pain Management Saudi Arabian Teaching Hospitals |

I already received approval from Research Ethics Committee of the Department of Health Studies, University of South Africa as well as an ethics approval certificate from King Abdullah International Medical Research Center (KAIMRC), Institutional Review Board (IRB) and Nursing Service Research Committee.

In order for me to collect data, your assistance to inform volunteer nurses to partake in the study will be appreciated. Will you please send the attached recruitment letter and a list to provide names of all Registered Nurses and Clinical Resource Nurses interested in research and who meet the following inclusion/eligibility criteria:-

- 1. Nurses who attended at least one pain management workshop within the past 3 years.
- 2. Nurses who attended ward in-service trainings about pain management the past 12 months.
- 3. Participants willing and comfortable to be interviewed in English.
- Clinical facilitators (Clinical Resource Nurses) who educates nurses about pain management in those nursing care divisions mentioned above.

I will appreciate the list of names and contact details from those who volunteer to participate to enable me to meet them, do a selection and then I will provide the selected ones with the questionnaire that they need to be complete. They will be asked to post the completed questionnaires in a envelope that will be provided for this purpose.

I appreciate your time and willingness to assist.

Yours Sincerely Litaba Efraim Kolobe

ANNEXURE 3: Information letter and consent form

Annexure 3

P.O.BOX 22490 Riyadh 11426 03 December 2019

Dear Colleague

My name is Litaba Efraim Kolobe and I am a registered doctoral student at the University of South Africa, The title of my intended study is 'An action to enhance transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals'. The aim of my study is to develop an action plan to enhance the transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals.

I hereby request you to volunteer to participate in the study. I intend to share the action plan with the nursing administrations, Center of nursing education and the pain management team to enhance the implementation thereof and to contribute to positive patient outcomes.

If you agree to participate it will be expected from you to complete a questionnaire anonymously. This questionnaire will only take 10 to 20 minutes of your time. Your name will not be on the questionnaire and the information that you provide will be kept confidential and no data will be linked to any individual. You can complete the questionnaire in your private time and put it back in the sealed box within one week after receiving it.

You will not be remunerated as participation is voluntary. The action plan however will benefit other students and patients in future. The results might be published, but your personal information and inputs will not be able to be tracked back to you.

You may withdraw from the study at any time, without fear of being victimised. If you agree to participate please communicate your willingness to your nurse manager so that your contact details can be shared with me. I will then provide you with the questionnaire to complete. After completion please place the questionnaire into the specifically marked survey box that will be placed in your ward's main nurses' station.

Please feel free to contact me at (Mobile +966503920421 or Tel. Ext 12864) if you have any questions regarding this research study.

Yours Sincerely

Litaba Efraim Kolobe

| Primary Investigator | Litaba Efraim Kolobe | | Supervisor | Prof. Lizeth Roets |
|------------------------------|----------------------|--|---------------------------|--------------------|
| Tel :2520088 Ext 1 | 2864 Pager : 3613 | | Tel: 012 429 2226 | |
| E-mail : kolobel@ngha.med.sa | | | Email: roestl@unisa.ac.za | |

P.O.BOX 22490

Riyadh

11426

03 December 2019

Dear Colleague

My name is Litaba Efraim Kolobe and I am a registered doctoral student at the University of South Africa, The title of my intended study is 'An action to enhance transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals'. The aim of my study is to develop an action plan to enhance the transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals.

I hereby request you to volunteer to participate in the study. I intend to share the action plan with the nursing administrations, Center of nursing education and the pain management team to enhance the implementation thereof and to contribute to positive patient outcomes.

If you agree to participate it will be expected from you to complete a questionnaire anonymously. This questionnaire will only take 10 to 20 minutes of your time. Your name will not be on the questionnaire and the information that you provide will be kept confidential and no data will be linked to any individual. You can complete the questionnaire in your private time and put it back in the sealed box within one week after receiving it.

You will not be remunerated as participation is voluntary. The action plan however will benefit other students and patients in future. The results might be published, but your personal information and inputs will not be able to be tracked back to you.

You may withdraw from the study at any time, without fear of being victimised. If you agree to participate please communicate your willingness to your nurse manager so that your contact details can be shared with me. I will then provide you with the
questionnaire to complete. After completion please place the questionnaire into the specifically marked survey box that will be placed in your ward's main nurses' station.

Please feel free to contact me at (Mobile +966503920421 or Tel. Ext 12864) if you have any questions regarding this research study.

Yours Sincerely

Litaba Efraim Kolobe

| Primary Invest gator | ti- Litaba Kolobe | Efraim | Supervisor | Prof. Lizeth Roets | | | |
|-------------------------|----------------------|----------|-------------------|--------------------|--|--|--|
| Tel :2520088 Ex | xt 12864 Page | r : 3613 | Tel: 012 429 2226 | | | | |
| E-mail : <u>kolobel</u> | engha.med.sa | 2 | Email: roestl@un | isa.ac.za | | | |

CONSENT FORM

Title: AN ACTION PLAN TO ENHANCE TRANSFER OF LEARNING OF PAIN MAN-AGEMENT COMPETENCIES OF NURSES IN SAUDI ARABIAN TEACHING HOS-PITALS

Principal investigator: Litaba Efraim Kolobe

I,.....the undersigned, agree to participate in the above-mentioned research study. I confirm that I received a letter that confirms that I can withdraw at any stage without penalty. I am aware that

- I will not be paid for my participation, and if I feel uncomfortable in any way during the completion of a questionnaire, I have the right to decline to answer any further questions.
- Participation involves completing the questionnaire.
- The researcher will not identify me by name in any of the reports and that my confidentiality as a participant in the study will remain secured.
- All materials containing identifying information will be destroyed once the completed study is accepted.

I have read and understood the information provided to me and had all my questions answered to my satisfaction, and I voluntarily agree to participate in the study.

If I have any questions about the research, I will contact the person mentioned below:

| Participant's Name Participant's Signature | Date | |
|--|----------|--|
| Witness Name Witness Signature | Date | |

ANNEXURE 4 : Questionnaire 1: Phase 1 (professional nurses) Questionnaire: Resources available to conduct a pain assessment

Dear Registered Nurse

Thank you for your willingness to voluntary participate in the research.

It is essential to complete the questionnaire as honest as possible. Please indicate your answer with a tick (\checkmark) in the appropriate box OR by answering the questions in the spaces provided:

Example

I like apples Yes No√

SECTION A: Biographical information

Please answer the following questions by placing a tick (\checkmark) in the appropriate box or by answering the questions in the spaces provided.

| 1. What is your gender? | For office use |
|--|----------------|
| Male1 Female2 2. What is your age? | 1 |
| [In complete years (e.g. 24)]1 | 2 |
| 3. To which nationality do you belong? | 3 |
| 4. What is your highest education qualification? | |
| Master degree1 | 4 |
| 518 | |



SECTION B: RESOURCES AVAILABLE TO DO PAIN ASSESSMENT

Please answer all the questions. Please choose "Yes" or "No" as your response to ALL statements below that describes your choice with a tick (\checkmark).

| Item | Item content | Yes | No | For of- |
|------|--------------|-----|----|---------|
| | | | | fice |
| No | | | | use |
| | | | | |

| 1 | The systematic pain assessment guide tools for pain history taking included below are available in your context to conduct pain assessments: | | | |
|-----|---|-----|----|------|
| 1.1 | QUEST (question the child, uses pain rating tools, evaluates behaviour, sensitise parents, and take ac- tion) approach is available to assess pain. | Yes | No | |
| 1.2 | WILDA (words to describe pain, intensity, location, duration, and aggravating or alleviating factors) ap- proach is available to assess pain. | Yes | No | |
| 1.3 | PQRST (p rovoking/ p alliation factors, q uality of pain, r egion of pain, s everity and timing) approach is available to assess pain. | Yes | No | |
| 1.4 | OPQRSTUV (onset of pain, provoking/palliating, quality, region/radiation of pain, severity of pain, timing/treatment, understanding/impact on you and values) approach is available to assess pain. | Yes | No | |
| 1.5 | COLDSPA (character, onset, location, duration, severity, pattern and associated factors) approach is available to assess pain | Yes | No | |
| 2 | Pain rating assessment tools of patients who can self-report their pain included below are available in your context to rate pain during pain assessments: | | | 7-11 |

| 2.1 | The Wong-Baker FACES pain scale is available to rate pain in children who can report their pain. | Yes | No | |
|-----|---|-----|----|--|
| 2.2 | The Numeric Rating Scale (NRS) is available to rate pain in children and adults who can report their pain. | Yes | No | |
| 2.3 | The Verbal Analogue Scale (VAS) is available to rate pain in adults who can report their pain. | Yes | No | |
| 2.4 | The Verbal Descriptor Scale (VDS) is available to rate pain in adults who can report their pain. | Yes | No | |
| 2.5 | The Brief Pain Inventory (BPI) is a questionnaire available to assess pain in cancer patients. | Yes | No | |
| 3 | Pain rating assessment tools of patients who cannot self-report their pain included below are available in your context to rate pain during pain assessments: | | | |
| 3.1 | A CRIES (Crying, Required oxygen, Increased vital signs, Expressions, Sleeplessness) pain scale is available to rate pain in premature and neonates during pain assessments. | Yes | No | |
| 3.2 | The Neonatal Pain, Agitation and Sedation Scale (N-PASS), is available to rate pain in premature and neonates during pain assessments. | Yes | No | |

| 3.3 | The Neonatal Infant Pain Scale (NIPS) pain scale is available to rate pain in premature and neonates during pain assessments. | Yes | No | |
|-----|---|-----|----|--|
| 3.4 | A FLACC (Faces, Legs, Activity, Crying, and Con- solability) pain scale is available to rate pain for pa- tients who cannot verbalise their pain during pain assessments. | Yes | No | |
| 3.5 | The COMFORT-Behaviour pain scale (COM- FORT-B) is available to rate pain for patients unable to verbalise their pain during pain assessments. | Yes | No | |
| 3.6 | The Critical Care Pain Observational Tool (CPOT) is available to asses and manage pain in adult non-conscious, critically-ill ventilated or non- ventilated patients in intensive care unit | Yes | No | |
| 3.7 | The Behavioural Pain Scale (BPS) is available for assessing pain in uncommunicative, critically-ill, se- dated and intubated patients in intensive care units. | Yes | No | |
| 4 | Pain rating assessment tools for elderly patients with dementia or cognitive impairment included below are available in your context to rate pain during pain assessment: | | | |
| 4.1 | The Abbey Pain Scal e (ABBEY) is available to measure acute, chronic and acute-on chronic pain intensity in patients with late-stage dementia. | Yes | No | |

| 4.2 | The Checklist on Nonverbal Pain Indicators (CNPI) is available to measure pain behaviours in cognitively impaired older adults. | Yes | No | | |
|-----|---|-----|----|--|--|
| 4.3 | The Pain Assessment in Advanced Dementia Scale (PAINAD) is available to assess pain in pa- tients with advanced moderate to severe dementia. | Yes | No | | |
| 4.4 | The Non-communicative Patient's Assessment Instrument (NOPPAN) pain scale is available to as- sess pain in demented and cognitively impaired pa- tients. | Yes | No | | |
| 5 | Human resources included below are available in your context to conduct pain assessments: | | | | |
| 5.1 | Competent registered nurse who previously re- ceived pain management training. | Yes | No | | |
| 5.2 | Clinical facilitators. | Yes | No | | |
| 5.3 | Pain nurses working in acute or chronic pain ser- vices | Yes | No | | |
| 5.4 | Pain nurse specialists. | Yes | No | | |

| 5.5 | Nurse educators. | Yes | No | |
|------------------------|---|-------------------------|---------------------|-------|
| 5.6 | Pain management physicians (such as acute pain physician, chronic pain physician) | Yes | No | |
| 5.7 | Ward nurse managers who received pain manage- ment training used to conduct pain assessment. | Yes | No | |
| 5.8 | Nurse supervisors who received pain management training. | Yes | No | |
| _ | Patient support as a resource included below are available in your context to conduct pain assessments: | | | |
| 6 | Patient support as a resource included below an your context to conduct pain assessments: | re availa | able in | 28-35 |
| 6 6.1 | Patient support as a resource included below as your context to conduct pain assessments: Patients or (other people or patients with pain). | re availa Yes | able in No | 28-35 |
| 6 6.1 6.2 | Patient support as a resource included below at your context to conduct pain assessments: Patients or (other people or patients with pain). Pain management support groups. | Yes | able in No No | 28-35 |
| 6 6.1 6.2 6.3 | Patient support as a resource included below as your context to conduct pain assessments: Patients or (other people or patients with pain). Pain management support groups. Patient pain management websites. | re availa Yes Yes | able in No No | 28-35 |

| 7 | Publications and electronic resources available in your context to conduct pain assessments: | | | |
|-----|--|-----|----|--|
| 7.1 | Clinical updates or journals maybe resources used to conduct pain assessment. | Yes | No | |
| 7.2 | Videos on pain management. | Yes | No | |
| 7.3 | E-newsletters. | Yes | No | |
| 7.4 | Fact sheets. | Yes | No | |
| 7.5 | E-learning modules. | Yes | No | |
| 7.6 | Pain toolkit. | Yes | No | |
| 7.7 | Printed reference books. | Yes | No | |

| 7.8 | Electronic flow sheets. | Yes | No | |
|-----|---|-----|----|-------|
| | | | | |
| 7.9 | Best clinical practice guidelines about pain assess- ment. | Yes | No | |
| 8 | Organisations that specialise with pain research, treatment, clinical practice and education (such as Saudi Pain Society, World Health Organisation, and International Association Study for Pain). | Yes | No | |
| 9 | Policies. | Yes | No | |
| | | | | 40-50 |

Would you like to add any comments?

THANK YOU FOR TAKING THE TIME TO COMPLETE THIS QUESTIONNAIRE

ANNEXURE 5 : Questionnaire 2: Phase 2 (professional nurses)

Questionnaire : Nurses' characteristics and learning styles seen to enhance transfer of pain management compétencies.

Dear Registered Nurse.

Thank you for your willingness to voluntary participate in the research.

It is essential to complete the questionnaire as honest as possible. Please indicate your answer with a tick (\checkmark) in the appropriate box, circle or in the spaces provided:

SECTION A : Biographical information

Please answer the following questions by placing a tick (\checkmark) in appropriate box or by answering the questions in the spaces provided.





SECTION B: CHARACTERISTICS SEENTO ENHANCE TRANSFER OF LEARNING OF PAIN MANAGEMENT COMPETENCIES

| ltem | Item Content | For Office Use |
|------|---|----------------------|
| Q1 | How do you apply what you have learned? | |
| | Rate yourself on a scale from 1 (this does not describe me at all) to 10 (this describes me perfectly). | |
| | 1236789 | |
| | | |

| 1.1 | I remember the pain management infor- mation from past experience. | Û23456789® | |
|------|---|-------------------------|--|
| 1.2 | I learn during pain management learn- ing/training sessions. | Û@③④⑤⑦®⑨⑩ | |
| 1.3 | I concentrate well during pain manage- ment learning/training sessions. | Û23456789W | |
| 1.4 | I understand the content of information taught during pain management learn-ing/training sessions. | D23456789W | |
| 1.5 | I choose an appropriate pain intervention strategy for every individual patient's pain level. | Û@③④⑤⑦®⑨₪ | |
| 1.6 | I think rationally to assess a patient experi- encing pain. | Û23456789W | |
| 1.7 | I perform an accurate pain assessment. | Û23456789W | |
| 1.8 | I assess pain on time. | Û23 4 56789® | |
| 1.9 | I reassess pain after interventions. | Û23 4 56789® | |
| 1.10 | I apply knowledge by orientating new col- leagues on how to assess and manage pain. | Û23456789® | |

| | Please choose from High to Low ONLY 3 (<u>questions 1.1 to 1.10</u> that best describe yo ment question number in the box below: 1 st 2 nd 3 rd | three) statements from above ou: Please write only the state- | |
|-----|---|--|------|
| | | | 7-13 |
| Q 2 | How does the following describes Rate yourself on a scale from 1 (this does not describes me perfectly). (1(2(3)(4)(5)(6)(5)) | e you as a learner? ot describe me at all) to 10 (this ⑦⑧⑨⑩ | |
| 2.1 | Self-directed learner.(Learner taking charge of his/her learning) | Û23456789® | |
| 2.2 | Inquisitive thinking learner.(Learner who is inclined to ask questions or eager for knowledge) | Û23456789W | |

| 2.3 | Curious thinking learner.(Learner having desire to learn or know more about some-thing) | Û23456789® | |
|-----|---|------------|--|
| 2.4 | Enthusiastic thinking learner.(Learner showing interest or excitement about learn-ing and doing something) | ÛQ3456789® | |
| 2.5 | Truth seeking learner.(Learner who ask challenging questions or ask truth, reasons and evidence about something he/she is learning) | ÛQ3456789® | |
| 2.6 | Organised thinking learner.(Learner who is able to think carefully to plan about some- thing to learn or do) | ÛQ3456789® | |
| 2.7 | Hard-working in enquiring learner.(Leaner putting efforts in doing a lot of work to know about something) | Û23456789® | |
| 2.8 | Self-confidence thinking learner. (Leaner who belief in oneself and know about own ability to learn and do something) | Û23456789W | |
| 2.9 | Creative learner. (Learner having courage to try to learn new things, feeling to pro- duce ideas, like to be the first do something and open to share his/her experience to others) | Û23456789® | |

| | Please choose from High to Low <u>ONLY 3 (reduces in the second s</u> | three) statements from <u>above</u> ou: Please write only the state- elow: | | |
|-----|---|--|-------|--|
| | | | 14-25 | |
| Q3 | How does the following describe how you are motivated to learn? Rate yourself on a scale from 1 (this does not describe me at all) to 10 (this describes me perfectly). 1-2-3-4-5-6-7-8-9-10 | | | |
| 3.1 | Attending pain management training pro- grams motivates me to learn. | | | |
| 3.2 | The ability to take initiative without assis- tance of others motivates me to learn | | | |
| 3.3 | Learning of pain management skills relevant to my working area motivates me to learn | | | |
| 3.4 | Gaining new knowledge about pain man- agement motivates me to learn | D234567890 | | |

| 3.5 | Knowing the desired goals for learn management motivates me to learn | 1234567890 | | | | |
|-----|--|--------------------------------|---|-------|--|--|
| 3.6 | My own goals about knowing pain man- | | | | | |
| | Please choose from High to Low <u>Q</u> <u>questions 3.1 to 3.6</u> that best des ment question number in the box 1 st 2 nd 3 rd | ONLY 3 (cribe yo below: | three) statements from <u>above</u> ou: Please write only the state- | | | |
| | | | | 26-34 | | |
| Q4 | How does the following describe how you are motivated to apply in practise what you have learned? Rate yourself on a scale from 1 (this does not describe me at all) to 10 (this describes me perfectly). 1-2-3-4-5-6-7-8-9-10 | | | | | |

| 4.1 | The desire to successfully perform a pain management skill. | Û23 4 56789® | | | | |
|-----|---|---|-------|--|--|--|
| 4.2 | The aim to increase my work perfor- mance about pain management. | ÛQ3 4 56789® | | | | |
| 4.3 | Receiving positive feedback about my 0234567890 performance. | | | | | |
| 4.4 | Taking part in planning outcomes of a pain management training program. | ÛQ3456789@ | | | | |
| 4.5 | The intention to function as a competent expert nurse in pain management skills. | | | | | |
| | Please choose from High to Low <u>ONLY 3</u> <u>questions 4.1 to 4.5</u> that best describe y ment question number in the box below | (three) statements from above ou: Please write only the state- | | | | |
| | | | 35-43 | | | |

SECTION C: LEARING STYLES SEEN TO ENHANCE TRANSFER OF LEARNING OF PAIN MANAGEMENT COMPETENCIES

| ltem No | Item Con | tent | For of- fice use |
|------------|---|---|------------------------|
| | How does the following descr Rate yourself on a scale from 1 (this o (this describes me perfectly). ①②③④⑤⑥ | ibe your learning styles? loes not describe me at all) to 10 | |
| 1 | By reading through the information. | ÛQ3 4 56789® | |
| 2 | By watching the activity such as a demonstration. | D23456789W | |
| 3 | By listening to the information such as during a lecture. | Û23456789W | |
| 4 | By participating in the group discus- sion. | Û23456789W | |

| 5 | By generating creative ideas in a group. | ①②③④⑤⑦⑧⑨⑩ | |
|----|---|-----------------------------------|--|
| 6 | By recording the lectures. | Û23456789W | |
| 7 | By connecting the information I al- ready know to new information. | Û23\$\$89® | |
| 8 | By taking control of my own learning. | Û23456789₩ | |
| 9 | By writing down the information. | ①234567890 | |
| 10 | By questioning the information that I have obtained. | Û23\$\$89® | |
| 11 | By personally taking part in the activity to practice the skill. | Û23456789® | |
| 12 | By learning from the internet. | Û23456789₩ | |
| 13 | In a silent environment. | Û23456789W | |
| 14 | By solving different pain management real life problems. | ①②③④⑤⑦⑧⑨⑩ | |
| | Please choose from High to Low <u>ONL</u>Y | (3 (three) statements from above | |
| | items 1 to 14 that best describe you | : Please write only the statement | |
| | question number in the box below: | | |
| | | | |
| | | | |
| | | | |

| 2 nd 3 rd | |
|------------------------------------|-------|
| | 44-60 |

Would you like to add any comments?



THANK YOU FOR TAKING THE TIME TO COMPLETE THIS QUESTIONNAIRE

ANNEXURE 6a: UINISA Ethical approval





University of South Africa Drater Street, Mucdeneuk Ringer, City of Triwane PO Pro 392, UNISA (2003 South Africa Telephone +27, 12, 429, 311) Facial rife, -27, 12, 429, 4, 00 www.unisa.ac.ac.ac.

ANNEXURE 6b : Ethical extension UNISA REC



COLLEGE OF HUMAN SCIENCES RESEARCH ETHICS REVIEW COMMITTEE

30 June 2023

Dear Mr Kolobe Litaba Efraim

NHREC Registration # : Rec-240816-052 CREC Reference # : 36640085_CREC_CHS_2023

Decision: Ethics Approval from 30 June 2023 to 30 June 2024

Researcher(s): Name: Mr. K. L. Efraim Contact details: <u>36640085@mylife.unisa.ac.za</u> Supervisor(s): Name: Prof L. Roets Contact details: <u>roetsl@unisa.ac.za</u>

TITLE: AN ACTION PLAN TO ENHANCE THE TRANSFER OF LEARNING OF PAIN MANAGEMENT COMPETENCIES OF NURSES IN SAUDI ARABIAN TEACHING HOSPITALS

Degree Purpose: PhD

Thank you for the application for research ethics clearance by the Unisa College of Human Science Ethics Committee. Ethics approval is granted for one year.

The *negligible risk application was reviewed by* College of Human Sciences Research Ethics Committee, in compliance with the Unisa Policy on Research Ethics and the Standard Operating Procedure on Research Ethics Risk Assessment.

The proposed research may now commence with the provisions that:

- The researcher(s) will ensure that the research project adheres to the values and principles expressed in the UNISA Policy on Research Ethics.
- Any adverse circumstance arising in the undertaking of the research project that is relevant to the ethicality of the study should be communicated in writing to the College Ethics Review Committee.
- The researcher(s) will conduct the study according to the methods and procedures set out in the approved application.



University of South Africa Prefer Street, McGlanauk Ridge City of Tahware PO Box 392 UNISA 0003 South Africa Telephone: +27 12 429 31 11 Resemble: +27 12 429 41 50 www.unisa.ac.za

ANNEXURE 7a: Nursing Services Permission to conduct research



ANNEXURE 7b:IRB approval SP 18/036/R



The approval includes the following related documents:

| Document/Title | Version | Date |
|---------------------------------------|---------|-------------|
| Research Proposal | 01 | 20 Mar 2018 |
| Questionnaires – Annexure 5 and 8 | 01 | 20 Mar 2018 |
| Cross-Sectional Informed Consent Form | 01 | 20 Mar 2018 |

The approval of the research study is valid for **one year** from the above approval to expiration date.

Terms of Approval:

 Annual Reports: An Annual report must be submitted for approval to avoid termination/suspension of your research.

 Financial report: If your study is funded project, details financial report should be submitted with the scientific report.

Final Report: After completion of the study, a final report must be forwarded to the IRB.

Retention of original data: The PL is responsible for the storage and retention of original data
pertaining to the project for a minimum of five years.

 Reporting of adverse events or unanticipated problems: The PI is responsible to report any serious or unexpected adverse events or unanticipated problems, which could involve a risk to participants or others.

 Biological samples: No biological samples to be shipped out of the Kingdom of Saudi Arabia without prior IRB approval.

 Participant incentives: No financial compensation or gifts to be given to participants without prior IR8 approval.

 Storage of biological samples: All biological samples collected for the purpose of this research must be stored in the KAIMRC related repository.

2 5 MAR 2018 Dr. Abdallah Adlan

Chaliman, Institutional Review Board (IRB) Ministry of National Guard - Health Affairs AA/GA/crid

P.O. Box 22490, Riyadh 11426 Tel 8011111 Telex : 403450 INGHMED SJ IORACLE 29795) H4 Thirting Press 17137

س ، بد ۲۲۶۹ الرياش ۱۹۶۹ تلفون ، ۲۰۱۱ م تلفس ، ۲۰۱۹

ANNEXURE 7c:IRB Annual Extension SP 18/036/R

| Kingdom of linistry of National (| Saudi A Guard - H | rabla ealth Alfairs | | Dy | غه | المغلكة العربية المسعودية ورارة الحرس الوطني - المُنوّون الصح |
|---|--|--|---|--|--|--|
| Z | | King Abd | lullah Internatio (KG Ref 1515 | nal Medical Re IMRC) | search Con | ter nano.med.sa |
| IRB Of | fice | Memo Ref.no. | IRB/1552/23 | E-CTS Ref. N | (). (). () () () () () () () () () () () () () | RYD-23-7810-94444 |
| Date: | (G) 27 (H) 07 | JUNE 2023 Thu Alhijjah 144 | 14 | | | |
| То: | DR. N/ Principa Consuli Munistr | ASSER TAWFEE al Investigator ant, Department y of National Gua | 2 0 SP18/036/R t of Anesthesia ard Health Affairs | | | |
| From: | PROF. Chairm Ministr | HAMDAN AL J. an, Institutional y of National Gui | AHDALI Raview Board (IR and Health Affairs | B) | By | 27 JUN 2023 |
| Subject | RESE/ LEAR/ TEACH | RCH PROTOCO | DL: SP18/036/R MANAGEMENT C LS | - AN ACTION F OMPETENCIES | LAN TO EN S OF NURSE | HANCE TRANSFER OF S OF SAUDI ARABIAN |
| This is to regarding Kindly be expire on | acknowle the <u>requ</u> informed DECEME | idge receipt of <u>est for extensi</u> that your reque: ER 27, 2023 | your memorandur on of the above-m st for IRB extension | n on June 15, entioned resear ion has been a | 2023 via E- ch project. pproved for | CTS# RYD-23-7810-94444, |
| IRB office | expects a | a final report who | an the project cond | ludes. | | |
| Thank you | J. | | | | | |
| Best rega | ids, | | | | | |
| H (A)) | | | | | | |
| | | | | | | |
| | | | | | | |

| istry of | National Gu | ard - Health Affairs | K | Jy | الشؤون الصحية. ا | إره الحرس الوطشي - |
|----------|--|--|---|-------------------------------------|--|-----------------------|
| | () | King Abd | ullah Internation (KA) | nal Medical Res IMRC) | earch Center | |
| 3 | IRB Offi | CE Memo Ref.ne. 1 | « (RB/2957/23 | E-CTS Ref. r | ■第回 no. ぞ天子の RYD-23-781 回惑業 | 0-94444 |
| C | Pate: | (G) 30 NOVEMBER 20 (H) 16 Jurnada I 1445 | 23 | | | |
| 8 | Fo: | DR. NASSER TAWFEED Principal Investigator, Si Section Head Obstetrics Department of Anesthes Ministry of National Guar | Q P18/036/R & Gynecology Ane ia, KAMC rd Health Affairs | sthesla | | • |
| , | rom: | PROF. HAMDAN AL JA Chairman, Institutional R Ministry of National Guar | H DALI Review Board (I RB rd Health Affairs | | Johdan 3 U NOV | 2023 |
| s | ubject: | RESEARCH PROTOCO Pain Management Co | L: SP18/036/R mpetencies of N | - An Action Pla arses of Saudi J | n to Enhance Transfer o Arablan Teaching Hospit | f Learning of tals |
| k v | Girdly be inf vill expire or IRB office ex Best regards | ermed that your reques 1 May 30, 2024. spects a final report who s. | t for IRB extens | kon has heen ap udes. | proved for six (6) mont | hs only and it |
| | | | | | | |
| | 10/04/90 | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |

ANNEXURE 8: Questionnaire 3: Phase 3 (clinical facilitators)

Questionnaire: Resources available to conduct a pain assessment

Dear Clinical Facilitator ('Clinical Resource Nurse)

Thank you for your willingness to voluntary participate in the research.

It is essential to complete the questionnaire as honest as possible. Please indicate your answer with a tick (\checkmark) in the appropriate box OR by answering the questions in the spaces provided:

Example





SECTION A: Biographical information

Please answer the following questions by placing a tick (\checkmark) in appropriate box or by answering the questions in the spaces provided.





SECTION B: RESOURCES AVAILABLE TO DO PAIN ASSESSMENT

Please answer all the questions. Please choose "Yes" or "No" as your response to ALL statements below that describes your choice with a tick (\checkmark).

| ltem No | Item content | Yes | No | For of- fice use |
|------------|---|-----|----|------------------------|
| 1 | The systematic pain assessment guide tools for pain history taking included below are available in your context to conduct pain assessments: | | | |
| 1.1 | QUEST (question the child, uses pain rating tools, evaluates behaviour, sensitise parents, and take ac- tion) approach is available to assess pain. | Yes | No | |
| 1.2 | WILDA (words to describe pain, intensity, location, duration, and aggravating or alleviating factors) approach is available to assess pain. | Yes | No | |
| 1.3 | PQRST (p rovoking/ p alliation factors, q uality of pain, r egion of pain, s everity and timing) approach is available to assess pain. | Yes | No | |
| 1.4 | OPQRSTUV (onset of pain, provoking/palliating, quality, region/radiation of pain, severity of pain, timing/treatment, understanding/impact on you and values) approach is available to assess pain. | Yes | No | |

| 1.5 | COLDSPA (character, onset, location, duration, severity, pattern and associated factors) approach is available to assess pain | Yes | No | |
|-----|---|-----|----|-------|
| 2 | Pain rating assessment tools of patients who can self-report their pain included below are available in your context to rate pain during pain assessments: | | | |
| 2.1 | The Wong-Baker FACES pain scale is used to rate pain children who can report their pain. | Yes | No | |
| 2.2 | The Numeric Rating Scale (NRS) is available to rate pain in children and adults who can report their pain. | Yes | No | |
| 2.3 | The Verbal Analogue Scale (VAS) is available to rate pain in adults who can report their pain. | Yes | No | |
| 2.4 | The Verbal Descriptor Scale (VDS) is available to rate pain in adults who can report their pain. | Yes | No | |
| 2.5 | The Brief Pain Inventory (BPI) is a questionnaire used to assess pain in cancer patients. | Yes | No | |
| 3 | Pain rating assessment tools of patients who cannot self-report their pain included below are available in your context to rate pain during pain assessments: | | | 12-16 |

| | - | - | - | |
|-----|---|-----|----|--|
| 3.1 | A CRIES (Crying, Required oxygen, Increased vital signs, Expressions, Sleeplessness) pain scale is available to rate pain in premature and neonates during pain assessments. | Yes | No | |
| 3.2 | The Neonatal Pain, Agitation and Sedation Scale (N-PASS), is available to rate pain in premature and neonates during pain assessments. | Yes | No | |
| 3.3 | The Neonatal Infant Pain Scale (NIPS) pain scale is available to rate pain in premature and neonates during pain assessments. | Yes | No | |
| 3.4 | AFLACC (Faces, Legs, Activity, Crying, and Con- solability) pain scale is available to rate pain for pa- tients who cannot verbalise their pain during pain assessments. | Yes | No | |
| 3.5 | The COMFORT-Behaviour pain scale (COM- FORT-B) is available to rate pain for patients unable to verbalise their pain during pain assessments. | Yes | No | |
| 3.6 | The Critical Care Pain Observational Tool (CPOT) is available to asses and manage pain in adult non-conscious, critically-ill ventilated or non-ventilated patients in intensive care unit | Yes | No | |
| 3.7 | The Behavioural Pain Scale (BPS) is available for assessing pain in uncommunicative, critically-ill, se- dated and intubated patients in intensive care units. | Yes | No | |

| 4 | Pain rating assessment tools for elderly patients with dementia or cognitive impairment included below are available in your context to rate pain during pain assessment: | | | |
|-----|---|-----|----|--|
| 4.1 | The Abbey Pain Scal e (ABBEY) is available to measure acute, chronic and acute-on chronic pain intensity in patients with late-stage dementia. | Yes | No | |
| 4.2 | The Checklist on Nonverbal Pain Indicators (CNPI) is available to measure pain behaviours in cognitively impaired older adults. | Yes | No | |
| 4.3 | The Pain Assessment in Advanced Dementia Scale (PAINAD) is available to assess pain in pa- tients with advanced moderate to severe dementia. | Yes | No | |
| 4.4 | The Non-communicative Patient's Assessment Instrument (NOPPAN) pain scale is available to as- sess pain in demented and cognitively impaired pa- tients. | Yes | No | |
| 5 | Human resources included below are available in your context to conduct pain assessments: | | | |
| 5.1 | Competent registered nurse who previously re- ceived pain management training. | Yes | No | |
| 5.2 | Clinical facilitators. | Yes | No | |

| 5.3 | Pain nurses working in acute or chronic pain ser- vices | Yes | No | |
|-----|---|-----|----|--|
| 5.4 | Pain nurse specialists. | Yes | No | |
| 5.5 | Nurse educators. | Yes | No | |
| 5.6 | Pain management physicians (such as acute pain physician, chronic pain physician) | Yes | No | |
| 5.7 | Ward nurse managers who received pain manage- ment training used to conduct pain assessment. | Yes | No | |
| 5.8 | Nurse supervisors who received pain management training. | Yes | No | |
| 6 | Patient support as a resource included below are available in your context to conduct pain assessments: | | | |
| 6.1 | Patients or (other people or patients with pain). | Yes | No | |
| 6.2 | Pain management support groups. | Yes | No | |

| 6.3 | Patient pain management websites. | Yes | No | |
|-----|--|-----|----|--|
| 6.4 | Patient pain management hotlines. | Yes | No | |
| 7 | Publications and electronic resources available in your context to conduct pain assessments: | | | |
| 7.1 | Clinical updates or journals maybe resources used to conduct pain assessment. | Yes | No | |
| 7.2 | Videos on pain management. | Yes | No | |
| 7.3 | E-newsletters. | Yes | No | |
| 7.4 | Fact sheets. | Yes | No | |
| 7.5 | E-learning modules. | Yes | No | |
| 7.6 | Pain toolkit. | Yes | No | |
|-----|---|-----|----|-------|
| 7.7 | Printed reference books. | Yes | No | |
| 7.8 | Electronic flow sheets. | Yes | No | |
| 7.9 | Best clinical practice guidelines about pain assess- ment. | Yes | No | |
| 8 | Organisations that specialise with pain research, treatment, clinical practice and education (such as Saudi Pain Society, World Health Organisation, and International Association Study for Pain). | Yes | No | |
| 9 | Policies. | Yes | No | |
| | | | | 40-50 |

SECTION C: TEACHING APPROACHES EMPLOYED DURING PAIN MANAGE-MENT EDUCATION OF NURSES

Please answer all the questions. Please choose "Yes" or "No" as your response to ALL statements below that describes your choice with a tick (\checkmark).

| ltem No | Item content | Yes | Νο | For of- fice use | |
|------------|---|-----|----|------------------------|--|
| | Teaching approaches included below are employed in your con- text during pain education of nurses: | | | | |
| 1 | Assessing prior knowledge about pain management as a base to each individual teaching lesson. | Yes | No | | |
| 2 | Asking nurses to write a reflective journal about pain management. | Yes | No | | |
| 3 | Allowing debriefing sessions about pain management after assessing individual nurse's learning need. | Yes | No | | |
| 4 | Using role modelling to learn about pain assessment. | Yes | No | | |
| 5 | Using case studies to learn how to manage pain. | Yes | No | | |

| 6 | Using available pain management e-learning mod- ules. | Yes | No | |
|----|---|-----|----|--|
| 7 | Engaging nurses in focus groups to learn about pain management. | Yes | No | |
| 8 | Forming a discussion group to learn about pain man- agement. | Yes | No | |
| 9 | Engaging hands-on activities to learn about pain as- sessment. | Yes | No | |
| 10 | Using role play activities to learn about pain assess- ment. | Yes | No | |
| 11 | Using practice simulations to assess registered nurses' skills about pain assessment. | Yes | No | |
| 12 | Using nursing grand rounds to learn directly at patient bedside how to assess and manage individual patient's pain. | Yes | No | |
| 13 | Preventing interruptions during teaching sessions. | Yes | No | |

| 14 | Providing assignment about pain management to apply it at work. | Yes | No | |
|----|---|-----|----|-------|
| | | | | 51-64 |

SECTION D: LEARNING CONTENT REGARDING PAIN ASSESSMENT AND MAN-AGEMENT

Please answer all the questions. Please choose "Yes" or "No" as your response to ALL statements below that describes your choice with a tick (\checkmark).

| ltem No | Item content | Yes | Νο | For of- fice use |
|------------|--|-----|----|---------------------|
| | ASSESSMENT | | | |
| 1 | The physiologic pain indicators stated below are included as learning content in your context regarding pain assessment: | | | |
| 1.1 | An increase in blood pressure. | Yes | No | |
| 1.2 | An increase in heart rate. | Yes | No | |

| 1.3 | Increased respiratory rate. | Yes | No | |
|--|---|-----|----|--|
| 2 The behavioural indicators of pain stated below are included as learning content in your context regarding pain assessment: | | | | |
| 2.1 | Facial expressions (such as frowning or grimacing). | Yes | No | |
| 2.2 | Verbal responses (such as crying). | Yes | No | |
| 2.3 | Body movements (such as kicking). | Yes | No | |
| 3 | Pain screening is included for registered nurses' knowledge of pain assessment. | Yes | No | |
| 4 | Obtaining the patient's self report of pain is included for registered nurses' knowledge of pain assessment. | Yes | No | |
| 5 | Obtaining a pain reported by parents or family mem- bers of the patient (proxy-reported) is included for registered nurses' knowledge of pain assessment. | Yes | No | |

| 6 | The systematic pain guide tool for pain history taking (namely WILDA) described below is included as learning content in your context regarding pain assessment: | | | | |
|-----|--|-----|----|--|--|
| 6.1 | Asking the patient to describe type of pain in his/her own words. | Yes | No | | |
| 6.2 | Asking about intensity of pain using pain scale | Yes | No | | |
| 6.3 | Asking about location of pain. | Yes | No | | |
| 6.4 | Asking about d uration of pain. | Yes | No | | |
| 6.5 | Asking about a ggravating or alleviating factors. | Yes | No | | |
| 7 | 7 The valid and reliable pain rating scales listed below are included as learning content in your context regarding pain assessment: | | | | |
| 7.1 | Using of Numeric Rating scale. | Yes | No | | |
| 7.2 | Using of Wong-Baker Faces scale. | Yes | No | | |

| 7.3 | Using of CRIES (crying, required oxygen, increased vital signs, expressions and sleeplessness) scale. | Yes | No | | |
|-----|---|-----|----|-------|--|
| 7.4 | Using of FLACC (Face, Legs, Activity, Crying and Consolability) pain scale. | Yes | No | | |
| | | | | 79-82 | |
| | NURSING DIAGNOSES | | | | |
| 8 | Acute types of pain stated below are included as learning con- tent in your context regarding pain assessment and manage- ment: | | | | |
| 8.1 | Labour pain (related to uterine muscles contractions) | Yes | No | | |
| 8.2 | Procedural pain (related to insertion of intravenous cannula). | Yes | No | | |
| 8.3 | Post-procedural pain (related to post peg tube inser- tion). | Yes | No | | |
| 8.4 | Acute post-operative pain (related to tissue injury secondary to surgical intervention). | Yes | No | | |

| 8.5 | Acute disease process (related to Sickle Cell anae- mia disease, abdominal pain due to intestinal cramps) | Yes | No | | |
|-----|---|-------|-------------------|-------|--|
| 8.6 | Traumatic pain (related to tissue injuries, shifting bone fragments) | Yes | No | | |
| 9 | Chronic types of pain stated below are included as tent in your context regarding pain assessment ment: | and m | ig con- anage- | 83-88 | |
| 9.1 | Cancer pain (related to tumour pressing on bones, nerves or organs). | Yes | No | | |
| 9.2 | Neuropathic pain (related to nerve damage or nerve degeneration). | Yes | No | | |
| 9.3 | Headache (related to migraine) | Yes | No | | |
| 9.4 | Low back pain (related to sciatic nerve compression). | Yes | No | | |
| 10 | 10 Factors related to impact of pain on activities of daily livings (ADLs) and quality of life (QOL) stated below are included as learning content in your context regarding pain assessment and management: | | | | |

| 10.1 | The physical impact of pain (indicated by muscle stiffness, difficulty in walking, fatigue, difficulty to sit or stand) | Yes | No | |
|---------------------------|--|-----------------------------------|-------------------------|-------|
| 10.2 | The psychological impact of pain (indicated by depression, emotional stress, anxiety, agitation, frustration) | Yes | No | |
| 10.3 | The social impact of pain (indicated by strained so- cial relationships, frequent absenteeism) | Yes | No | |
| | | | | 94-96 |
| | PLANNING | | | |
| | | | | |
| 11 | Strategies to plan for pain management stated cluded as learning content in your context regard agement: | below a ing pair | are in- n man- | |
| 11 11.1 | Strategies to plan for pain management stated cluded as learning content in your context regard agement: Developing the nursing care plan that addresses pa- tient's pain management needs. | below a ing pair Yes | are in- n man- | |
| 11 11.1 11.2 | Strategies to plan for pain management stated cluded as learning content in your context regard agement: Developing the nursing care plan that addresses pa- tient's pain management needs. Identifying patient's pain management goals. | below a ing pair Yes Yes | are in- n man- No | |

| 11.4 | Setting expected outcomes to enhance patient's comfort. | Yes | No | |
|------|--|-----|----|---------|
| | | | | 97-100 |
| | IMPLEMENTATION | | | |
| 12 | 2 Pain intervention strategies implemented below are included as learning content in your context regarding pain assessment and management: | | | |
| 12.1 | Non-pharmacological pain interventions. | Yes | No | |
| 12.2 | Pharmacological pain interventions. | Yes | No | |
| 12.3 | Patient and family education about pain manage- ment. | Yes | No | |
| | | | | 101-103 |
| | EVALUATION | | | |
| 13 | Nursing actions for pain management evaluation stated below are included as learning content in your context regarding pain assessment and management: | | | |

| 13.1 | Assessing the effectiveness of pain nursing care in- terventions implemented | Yes | No | |
|------|---|-----|----|---------|
| 13.2 | Monitoring of adverse effects related to pain medica- tions administered to the patient | Yes | No | |
| 13.3 | Constant pain reassessment after interventions of pain (toward achieving desired pain management goals and outcomes). | Yes | No | |
| | | | | 104-106 |

SECTION E: TRANSFER OF LEARNING CLIMATE WITHIN THE HOSPITAL NURS-ING CARE

Please answer all the questions, choose "Yes" or "No" by marking the statement that describe your choice with a tick (\checkmark).

| ltem No | Item content | Yes | Νο | For of- fice use |
|------------|---|-----|----|---------------------|
| 1 | The nursing leadership value the learning needs of the professional nurses by considering pain manage- ment training program. | Yes | No | |

| 2 | The ward nurse managers value continuous educa- tion to the registered nurses about pain management by allowing unit in-service trainings. | Yes | No | |
|---|--|-----|----|--|
| 3 | Pain management orientation program is available to support newly employed registered nurses to ac- quaint them to pain management practice of the hos- pital. | Yes | No | |
| 4 | The dedicated clinical facilities for pain assessment and management are available in the working areas for registered nurses to learn about pain manage- ment. | Yes | No | |
| 5 | Nurse preceptors who did not attend preceptor train- ing program also effectively orientate newly em- ployed registered nurses regarding pain manage- ment. | Yes | No | |
| 6 | Nurse preceptors who attended preceptor training program effectively orientate newly employed regis- tered nurses regarding pain management. | Yes | No | |
| 7 | Registered nurses support each other to apply what they learnt about pain management. | Yes | No | |
| 8 | Nursing supervisors support registered nurses when they need it bout pain management. | Yes | No | |

| 9 | Ward nurse managers support registered nurses to apply what they learnt about pain management. | Yes | No | |
|----|---|-----|----|---------|
| 10 | Clinical facilitators offer constructive feedback to reg- istered nurses regarding their performance about pain management. | Yes | No | |
| 11 | Registered nurses are given freedom to apply their pain management skills in their working areas. | Yes | No | |
| | | | | 107-117 |

Would you like to add any comments?



THANK YOU FOR TAKING THE TIME TO COMPLETE THIS QUESTIONNAIRE.

ANNEXURE 9: INVITATION FOR PARTICIPATION IN THE PRE-TEST FOR THE VALIDATION OF THE ACTION PLAN

Dear Panellist,

My name is Litaba Efraim Kolobe, and I am currently a doctoral student at the University of South Africa. Currently, I am conducting a study entitled 'AN ACTION PLAN **TO ENHANCE TRANSFER OF LEARNING OF PAIN MANAGEMENT COMPETEN-CIES OF NURSES IN SAUDI ARABIAN HOSPITALS'.** The purpose of my study is to develop an action plan for enhancing the transfer of pain management competencies among nurses in Saudi Arabian teaching hospitals. Ethics approval to conduct this study was obtained from the Research Ethics Committee, College of Human Sciences, UNISA with reference number REC-012714-039.

You are cordially invited to participate in the pre-testing of the developed action plan with the embedded validation tool Kindly assist **me in cross-checking that the questions asked, the items included, and the instructions are properly understood and will be correctly interpreted. Please provide me with comments regarding any issues you may encounter while completing the instrument. Your participation will remain anonymous and confidential. It will not be possible to share your individual contributions with other panellists as all information will be received via the software program in bulk.**

To determine whether the action plan is appropriate for implementation, you should evaluate each statement contained in the plan, but also assess the appropriateness of the validation instrument. If you agree to take part in the pre-test, kindly click on the link via the **Google Forms software** provided at the end of the recruitment letter. Instructions are provided as you go along completing the pre-test.

Your participation is voluntary, and you may choose not to take part, but just ignoring the invitation and not clicking on the link to access the action plan and validation tool. You still have a chance to withdraw even after completing the validation tool, by not submitting your answers. If you agree to participate, you will be required to complete and comment on this anonymous online action plan

and validation instrument. By clicking the link provided below, you will be able to access the instrument. It will only take 30 to 45 minutes for you to complete this questionnaire. I would greatly appreciate it if you would complete the questionnaire at your convenience by opening the link below.

https://docs.google.com/forms/d/1OJManLl1eKrqfL8v4Jp34Ojj4fHKtHiizGAe_hBOZ8 c/edit

Because participation in this pre-test is voluntary, you will not be remunerated. Your inputs will however be beneficial to allow me to adapt the action plan and instruments before sharing them with the stakeholders for final inputs. The results of the study may be published, but your personal information and inputs will not be able to be traced back to you.

Remember to click the submit button after you have completed the instrument.

Please feel free to contact me by at Mobile +966503920421 or by email at <u>kolobe66@yahoo.com</u> if you have any questions regarding this research study. You may also contact my research supervisor Prof Lizeth Roets at +27 12 429 2226 or <u>roetsl@unisa.ac.za</u>. If you have any other concerns, you may also contact the College research ethics committee at Khankb@unisa.ac.za, the chairperson of the committee.

Your contribution and time is appreciated.

Kind regards.

Litaba Efraim Kolobe

ANNEXURE 10: RECRUITMENT LETTER: ROUND 1 (e-Delphi)

Dear Panellist,

My name is Litaba Efraim Kolobe, and I am currently a doctoral student at the University of South Africa. I am conducting a study entitled 'AN ACTION PLAN TO EN-HANCE TRANSFER OF LEARNING OF PAIN MANAGEMENT COMPETENCIES OF NURSES IN SAUDI ARABIAN HOSPITALS'. The purpose of my study is to develop an action plan for enhancing the transfer of pain management competencies among nurses in Saudi Arabian teaching hospitals. Ethics approval to conduct this study was obtained from the Research Ethics Committee, College of Human Sciences, UNISA with reference number REC-012714-039.

You are cordially invited to participate in the first round of the validation of the developed action plan by using the embedded validation tool. Kindly assist **me in crosschecking whether you agree or disagree with the action statements, the methods used to achieve the actions, the persons who need to take responsibility as well as the appropriate time frames. Please also provide me with any comments or suggestions to improve the action plan.** Your participation will remain anonymous and confidential, and your specific contributions will not be shared with other panellists as all information will be received via the software program in bulk.

If you agree to take part in the first round of the validation of the action plan, kindly click on the link via the **Google Forms software** provided at the end of this letter. Instructions are provided as you go along completing the first round.

Your participation is voluntary, and you may choose not to take part, by just ignoring the invitation and not clicking on the link to access the action plan and validation tool. You still have a chance to withdraw even after completing the validation tool, by not submitting your answers. All comments and suggestions for improvement will be incorporated and forwarded to a second round until there is a 75% consensus among all panellists. By clicking the link provided below, you will be able to access the instrument. It will only take 30 to 45 minutes for you to complete

this questionnaire. I would greatly appreciate it if you would complete the questionnaire at your convenience by opening the link below.

https://docs.google.com/forms/d/15wodxUyvL8Lgz3LYZuTAfR85xiceSbqSdSvzObMWIAc/edit

Please submit your answers within 10 days after receiving this invitation.

Because participation in this study is voluntary, you will not be remunerated. Your input will, however, be beneficial to allow me to revise the action plan by implementing all suggested changes by all panellists and share the revised action plan with you again and again until consensus amongst all panellists is achieved. The results of the study may be published, but your personal information and input will not be able to be traced back to you.

After completion, please click the submit button.

Please feel free to contact me by at Mobile +966503920421 or by email at <u>kolobe66@yahoo.com</u> if you have any questions regarding this research study. You may also contact my research supervisor Prof Lizeth Roets at +27 12 429 2226 or <u>roetsl@unisa.ac.za</u>. If you have any other concerns, you may also contact the College research ethics committee at <u>Khankb@unisa.ac.za</u>, the chairperson of the committee.

Your contribution and time will be appreciated.

Kind regards.

Litaba Efraim Kolobe

ANNEXURE 11: Storage and management of data collected

Study Title: An Action Plan to enhance Transfer of Learning of Pain Management competencies of nurses in Saudi Arabian Teaching Hospitals.

- **1.** To protect identity by giving each subject a code number to anonymise the data.
- 2. Master list of the subjects' names and personal information will be kept under lock and key.
- **3.** Only the researcher for the study will be the person with access to the data/information generated by the study.
- **4.** The copies of the research methods, coded data and the reports will be kept locked in safe place for a period of 3 years.

| Primary Investi- gator | Litaba Kolobe | Efraim | Supervisor | Prof Roets | Lizeth |
|---------------------------|------------------|-------------|------------------|-----------------|--------|
| Conta | act details: | | Contac | t details: | |
| King Abdulaziz Me yadh | edical City Hos | spital, Ri- | Tel: 012 429 222 | 26 <u>za</u> | |
| Tel :2520088 ext. | 12864 Mail co | de:1242 | | | |
| Pager :3613 | | | | | |
| E-mail : <u>36640085</u> | i@mylife.unisa | .ac.za | | | |
| kolobel@ngha.me | ed.sa | | | | |

ANNEXURE 12: Draft 1: Action plan with embedded validation tool

0

X

:

:

Section 1 of 3

AN ACTION PLAN TO ENHANCE THE TRANSFER OF LEARNING OF PAIN MANAGEMENT COMPETENCIES OF NURSES IN SAUDI ARABIAN TEACHING HOSPITALS

Thank you for reading the recruitment letter and for volunteering to participate in the ROUND 1. Remember you can still decide not to take part, by not submitting your answers once you have completed the pre-test.

Your valuable inputs will however be greatly appreciated.

Please follow the instructions carefully

Kind regards Efraim Kolobe

SECTION A: BIOGRAPHICAL DATA OF PANELLISTS

Description (optional)

1. What is your gender? *

) Male

Female

) Other...

2. How old are you? *

1/5/24, 2:46 AM AN ACTION PLAN TO ENHANCE THE TRANSFER OF LEARNING OF PAIN MANAGEMENT COMPETENCIES OF NURSES IN ...

3. 3. To which nationality do you belong? *

Check all that apply.

| British | | |
|---------------|--|--|
| Czech | | |
| Filipino | | |
| Indian | | |
| Jordanian | | |
| Malaysian | | |
| Portuguese | | |
| Saudi | | |
| South African | | |
| Slovak | | |
| Spanish | | |
| Other: | | |

4. 4. What is your highest education qualifications?

Check all that apply.

- Master's degree Bachelor's degree
- Diploma in Nursing

Other:

| 5. In which of the following nursing care areas do you work? * |
|--|
| Cardiac Ward |
| Obs-Gynae Ward |
| Medical Ward |
| Paediatric Ward |
| Surgical Ward |
| Other |
| |

| 6. Please indicate your specific position in the nursing care areas do you work * |
|--|
| Clinical Facilitator (Clinical Resource Nurse) |
| Charge Nurse |
| Registered Nurse |
| |
| 7. How long have been in this current position? * |
| Short answer text |
| |

Section 3 of 3

SECTION B: ACTION PLAN

Instructions to Panellists

Please indicate your answers by selecting either 'Agree' or 'Disagree' next to the options provided to indicate your agreement or disagreement with the inclusion of the (1) action statement as well as the (2) methods suggested.

ž

:

1. Inclusion of each action statement in the action plan

2. The methods suggested to assist in achieving the action goal and expected outcome

Please choose from the list provided the **best possible responsible person(s)** that you recommend must take responsibility for every method.

Please choose from the list provided the **best possible timeframe** within which every method must be completed once the action plan is accepted.

Please provide detailed suggestions and comments in all the **suggestion /comment** boxes for improvement of any of the items/aspects in the draft action plan.

Action statement 1: Motivate nurses to further their studies *

Agree

Disagree

Please add any comment pertaining to this action statement

Short answer text

Method 1.1 Develop a policy to motivate nurses to improve their nursing qualifications *

Agree

Disagree

Please add any comment pertaining to this method

| If you agree that a policy should be developed, please tick all that should be included in the * policy to motivate the nurses |
|--|
| Paid full-time study leave for 1 year |
| 20 hours paid study leave and 20 hours full-time work for 1 year |
| Free accommodation for the period of study leave |
| A monetary incentive after completion of a new formal qualification (degree or diploma) |
| One day off for attending a one-day pain management program |
| A monetary incentive after completion of a pain management program |
| A certificate issued as an acknowledgment of nurses pursuing distance learning |
| A monetary incentive after completion of distance learning programs |
| |

Please add any other **suggestion** on what can be included in such a policy

| Responsible person(s) Please select (check the box(es a policy that motivates further s | s)) next to the best person/s to be responsible for the development of studies. |
|---|---|
| Five Members of the Central F | Region Nursing Governance and Accountability Board appointed by the Chie |
| Ad hoc committee appointed | by the heads of the Human Resource Department Central Region, Riyadh |
| Associate Executive Directors | of Nursing for King Abdulaziz Medical City (KAMC) and King Abdullah Spe |
| Nursing policy committee rep | resentatives appointed by the Associate Executive Directors of Nursing for |
| Clinical Directors of Nursing C | Operations for KAMC and KASCH |
| Director of Postgraduate Cent | ter of Nursing Education for KAMC and KASCH |
| Short answer text | |
| Time frame | * |
| Select the most appropriate tin policy should be developed and | neframe, after approval of the action plan, within which time the d finalised. |
| 1-3 months | |
| 4- 6 months | |
| 7-9 months | |

| Please ac finalisatic | Id any other suggestion regarding the most appropriate timeframe for development and on of such a policy |
|-------------------------------------|--|
| Short ansv | ver text |
| Method 1 to improv (MNGHA) | .2 Present and negotiate for the implementation of the policy to motivate nurses * re their nursing qualifications to the Ministry of National Guard Health Affairs) through the Central Region Nursing Governance and Accountability Board |
| Agree | |
| Disag | ree |
| ease add nd negoti nort answe | any other suggestion on who would be the best person(s) to be responsible to present ate for the implementation of such a policy |
| me frame | * a: Select the most appropriate timeframe within which time the policy should be and the implementation negotiated to the Ministry of National Guard Health Affairs |
| (NGHA) | |
| /INGHA) | hs |
| /INGHA)] 3 mont] 6 mont | hs |

| Please add any comment pertaining to this method |
|---|
| Short answer text |
| |
| |
| * Responsible person(s): Please select (check the box(es)) next to the best person/s to be responsible for the inclusion of the policy in the hospitals policies to motivate nurses to improve their nursing qualifications. |
| Associate Executive Directors of Nursing for KAMC and KASCH |
| Clinical Directors of Nursing Operations in every facility for KAMC and KASCH |
| Nurse Managers in all KAMC and KASCH nursing care areas |
| |
| Please add any other suggestion regarding the most appropriate timeframe for presentation of such a policy |
| Short answer text |
| |
| Method 1.3 Include the policy in all hospitals` policies after approval by the Ministry of * National Guard Health Affairs (MNGHA) |
| Agree |
| Disagree |

| Short answer text |
|---|
| |
| |
| * Responsible person(s): Please select (check the box(es)) next to the best person/s to be responsible for the inclusion of the policy in the hospitals policies to motivate nurses to improve their nursing qualifications. |
| Associate Executive Directors of Nursing for KAMC and KASCH |
| Clinical Directors of Nursing Operations in every facility for KAMC and KASCH |
| Nurse Managers in all KAMC and KASCH nursing care areas |
| Short answer text |
| |
| |
| Time frame : Select the most appropriate timeframe within which time the policy should be included in the hospital policies after approval by the Ministry of National Guard Health Affairs |
| Time frame: Select the most appropriate timeframe within which time the policy should be included in the hospital policies after approval by the Ministry of National Guard Health Affairs |
| Time frame: Select the most appropriate timeframe within which time the policy should be included in the hospital policies after approval by the Ministry of National Guard Health Affairs 1 month Six weeks |
| Time frame: Select the most appropriate timeframe within which time the policy should be included in the hospital policies after approval by the Ministry of National Guard Health Affairs 1 month Six weeks 2 months |

Please add any other **suggestion** regarding the most appropriate timeframe to include such a policy in the hospital policies

Short answer text

Action statement 2: Make appropriate and relevant pain management tools accessible to the nursing team in every clinical area

Agree

Disagree

Please add any comment pertaining to this action statement

Short answer text

Method 2.1 Include pain assessment tools in an electronic patient record system so that the nursing team can choose and access the most appropriate tool in all nursing care areas.

QUEST
COLDSPA
OPQSTUV
PQRST
BPI
VDS
VAS

| □ NPASS |
|-----------|
| СРОТ |
| BPS |
| COMFORT B |
| CRIES |
| ABBEY |
| PAINAD |
| NOPPAIN |
| CNPI |

Please add any comment pertaining to this method

Short answer text

Responsible person(s): Please select (check the box(es)) next to the best person/s to be responsible to provide the electronic format of the pain assessment tools for inclusion in the electronic patient record system to be accessible to the nursing team in every nursing care area.

Two computer engineers appointed by the HR manager each assigned for KAMC and KASCH

Five Nursing health informatics appointed by Associate Directors of Nursing for KAMC and KASCH

Clinical Director of Nursing Operations appointed by Associate Executive Directors in every facility for KA...

One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility for KAMC...

| Please add any other suggestion on who would be the best person(s) to be responsible to provide such an electronic format Short answer text |
|--|
| Time frame: Select the most appropriate timeframe within which time the pain assessment * tools should be included in the electronic patient record system 1-3 months 4-6 months 7-9 months |
| |
| Please add any suggestion regarding the most appropriate timeframe to include such a pain assessment tools Short answer text |
| * Method 2.2 Involve nurse supervisors with pain management training to provide post-pain management training supervisory support to the nursing team on how to conduct pain assessments in all nursing care areas. |
| Agree |
| Disagree |

| Flease and any comment benchmind to this method | Please add | anv comment | pertaining to | this method |
|--|------------|-------------|---------------|-------------|
|--|------------|-------------|---------------|-------------|

Short answer text

Responsible person(s): Please select (check the box(es)) the best person/s responsible to involve nurse supervisors with pain management training to provide post-pain management training and supervisory support to the nursing team on how to conduct pain assessments in all nursing care areas

Clinical Directors of Nursing Operations for KAMC and KASCH

Nurse Managers in all KAMC and KASCH nursing care areas

Clinical facilitators in all KAMC and KASCH nursing care areas

Charge nurses in all KAMC and KASCH nursing care areas

Please add any other **suggestion** on who may be the best person(s) to be responsible for providing such an electronic format

Short answer text

Time frame: Select the most appropriate timeframe within which time the nurse supervisors should be involved to provide pain management training supervisory support to the nursing team on how to conduct pain assessment in all nursing care areas.

Every shift when the need arises

Every patient round when the need arises

| Please add any s be involved in su Short answer text | suggestion regarding the most appropriate time frame the nurse supervisors sho uch a supervisory support t | uld |
|--|--|-----|
| Method 2.3 Mai | ke hospitals' internet-based resources accessible to the patients and family * | |
| Agree | | |
| Disagree | | |
| | | |
| Please add any | other comment pertaining to this method | |
| Short answer text | t | |
| If you agree that to patients and t | * It hospitals' internet-based resources should be accessible to provide support family members, please tick all that should be available. | |
| Patient pain ı | management websites | |
| Patient pain r | management hotlines | |
| Pain manage | ement support groups | |
| | | |
| Please add any s | suggestion on accessibility of such an internet-based resources | |

| * Responsible person(s): Please select (check the box(es)) the best person/s responsible to ensure internet-based resources should be accessible to patients and family members to provide support about pain management. |
|--|
| Chief Executive Director for KAMC and KASCH |
| Two health information technologists appointed by the HR manager each assigned for KAMC and KASCH |
| Two communication and information administrators appointed by the HR manager for KAMC and KASCH |
| Five Nursing health informatics appointed by Associate Directors of Nursing for KAMC and KASCH |
| Associate Executive Directors of Nursing for KAMC and KASCH |
| Clinical Directors of Nursing Operations for KAMC and KASCH |
| Nurse supervisors appointed by Executive Associate Directors of Nursing in every facility for KAMC and |
| Nurse Managers in all KAMC and KASCH nursing care areas Charge nurses in all KAMC and KASCH nursing care areas Registered nurses in all KAMC and KASCH nursing care areas |
| |
| Please add any suggestion on who may be the best person(s) to be responsible for ensuring accessibility of such internet-based resources Short answer text |
| |
| * Time frame: Select the most appropriate timeframe within which time the hospitals' internet-based resources should be accessible to provide support to patients and family members. |
| Every day at a convenient time |
| 24-hour access 7 days a week |

| Please add any suggestion regrading the most appropriate time frame for accessibility of such internet-based resources |
|---|
| Short answer text |
| |
| Method 2.4 Make hospitals' internet-based resources on pain management publications and * electronic materials accessible to the nursing team in all nursing care areas |
| Agree |
| Disagree |
| |
| |
| Please add any other comment pertaining to this method |
| Long answer text |

| *** | |
|--|--|
| * If you agree that hospitals' internet-based resources on pain management publications, electronic materials, and organisations that specialises in pain management should be accessible to the nursing team in all nursing care areas, please tick all that should be available. | |
| E-newsletters | |
| Videos on pain management | |
| Facts sheets | |
| Pain toolkits | |
| Clinical updates | |
| The World Health Organization | |
| | |

| America | n Pain Society |
|---|--|
| Internati | onal Association for Study of Pain |
| | |
| Please add about sucł | any suggestion on what internet-based resources should be available to nursing teams n pain management organisations |
| Long answe | rtext |
| | |
| Responsible | person(s): Please Select (cneck the box(es)) the best person/s responsible to |
| Responsible ensure interi organisation in all nursing | person(s): Please Select (check the box(es)) the best person/s responsible to net-based resources on pain management publications, electronic materials, and s that specialises in pain management should be accessible to the nursing team g care areas |
| Responsible ensure intern organisation in all nursing Two com | person(s): Please Select (check the box(es)) the best person/s responsible to net-based resources on pain management publications, electronic materials, and is that specialises in pain management should be accessible to the nursing team g care areas |
| Hesponsible ensure internorganisation organisation in all nursing Two com Two libra | person(s): Please select (check the box(es)) the best person/s responsible to het-based resources on pain management publications, electronic materials, and is that specialises in pain management should be accessible to the nursing team g care areas nputer engineers appointed by the HR manager each assigned for KAMC and KASCH irians appointed by the HR manager each assigned for KAMC and KASCH |
| Responsible ensure internorganisation in all nursing Two com Two libra Nursing I | person(s): Please Select (Check the box(es)) the best person/s responsible to net-based resources on pain management publications, electronic materials, and is that specialises in pain management should be accessible to the nursing team g care areas nputer engineers appointed by the HR manager each assigned for KAMC and KASCH irians appointed by the HR manager each assigned for KAMC and KASCH mealth informatics appointed by Associate Directors of Nursing for KAMC and KASCH |
| Responsible ensure internorganisation in all nursing Two com Two libra Nursing I Nurse Mage | person(s): Please Select (check the box(es)) the best person/s responsible to het-based resources on pain management publications, electronic materials, and is that specialises in pain management should be accessible to the nursing team g care areas apputer engineers appointed by the HR manager each assigned for KAMC and KASCH arians appointed by the HR manager each assigned for KAMC and KASCH health informatics appointed by Associate Directors of Nursing for KAMC and KASCH anagers in all KAMC and KASCH nursing care areas |
| Responsible ensure internorganisation in all nursing Two com Two libra Nursing I Nurse Mail Nurse Su | person(s): Please Select (CRECK the Dox(es)) the Dest person/s responsible to net-based resources on pain management publications, electronic materials, and is that specialises in pain management should be accessible to the nursing team of care areas aputer engineers appointed by the HR manager each assigned for KAMC and KASCH arians appointed by the HR manager each assigned for KAMC and KASCH mealth informatics appointed by Associate Directors of Nursing for KAMC and KASCH anagers in all KAMC and KASCH nursing care areas pervisors appointed by Executive Associate Directors of Nursing for KAMC and KASCH |
| Responsible ensure internorganisation in all nursing Two com Two libra Nursing I Nurse Mage Nurse su One Pain | person(s): Please Select (Check the box(es)) the best person/s responsible to net-based resources on pain management publications, electronic materials, and is that specialises in pain management should be accessible to the nursing team g care areas nputer engineers appointed by the HR manager each assigned for KAMC and KASCH rians appointed by the HR manager each assigned for KAMC and KASCH nealth informatics appointed by Associate Directors of Nursing for KAMC and KASCH anagers in all KAMC and KASCH nursing care areas pervisors appointed by Executive Associate Directors of Nursing for KAMC and KASCH |

Please add any **suggestion** on who may be the best person(s) to be responsible for ensuring accessibility of internet-based resources to such pain management organisations

Short answer text

Time frame: Select the most appropriate timeframe within which time the hospitals' internetbased resources on organisations that specialises in pain management, pain management publications, and electronic materials should be accessible to the nursing team in all nursing care areas.

Every nursing shift

24-hour access 7 days a week

Please add any **suggestion** regarding the most appropriate time frame for accessing internet-based resources to such pain management organisations

Short answer text

Action statement 3: Develop a practice-oriented content specific short pain management training program

Agree

Disagree
| Please add any comment pertaining to this action statement | |
|--|--|
| Short answer text | |
| | |
| *** Method 3.1 Include practice-oriented pain management training content for all nursing care * areas in the pain management program | |
| Agree | |
| Disagree | |
| | |
| Please add any comment pertaining to this method | |
| Short answer text | |
| | |
| If you agree that specific practice-oriented pain management content should be included in * all nursing care areas, please tick all that should be included | |
| Methods to promptly assess a patients' pain in all nursing areas | |
| Assessment of patients' pain in all nursing care areas | |
| The advantages and disadvantages off all pain management scales | |
| Labour pain as a type of pain to be assessed | |
| The selection of appropriate pain intervention strategies based on the pain levels assessed | |
| | |

| Please a | dd any suggestion on what can be included in such a pain management training content |
|---------------------------------|--|
| Short ans | wer text |
| | |
| | |
| Respons to include areas. | * ible person(s): Please select (check the box(es)) the best person/s to be responsible e specific practice-oriented pain management training content for all nursing care |
| Direc | tor of Postgraduate Center of Nursing Education for KAMC and KASCH |
| One l | Nurse educator appointed by the Director of the Postgraduate Center of Nursing Education for KAM |
| One (| Clinical facilitator appointed by Clinical Directors of Nursing Operations in every nursing care area |
| One F | Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility |
| | |
| Please ad a pain ma | d any suggestion on who may be the best person(s) to be responsible for including such nagement training content |
| Short answ | rer text |
| | |
| | |
| Time fram pain mana | IE: Select the appropriate timeframe within which time the specific practice-oriented * agement training content should be provided for inclusion in the program |
| 1 weel | k before the due date of the training program |
| 1 mon | th before the due date of the training program |
| | |
| 3 mon | ths before the due date of the training program |

Please add any **suggestion** regarding the most appropriate time frame to provide for inclusion such a pain management training content in the program

:::

| Action statement 4: learning types | Develop a pain management short program that accommodate all | * |
|---------------------------------------|--|---|
| Agree | | |
| Disagree | | |
| | | |
| Please add any comr | ::: ment pertaining to this action statement | |
| Short answer text | | |
| | | |
| | | |
| | | |

| Method 4.1 Incorporate different learner types during learning/training sessions * |
|--|
| Agree |
| Disagree |
| |
| Please add any comment pertaining to this method |
| Short answer text |
| |

| If you agree that different learner types of nurses should be included during pain |
|---|
| management learning/training sessions, please tick all types that should be included during |
| learning/training sessions. |

Creative Learners

| - E | Enthusiastic | thinking | learners |
|-----|--------------|----------|----------|
|-----|--------------|----------|----------|

Self-confident thinkers

Organised thinking learners

Truth-seeking learners

Inquisitive thinkers

Diligent inquisitive learners

Self-directed learners

Please add any suggestion on what can be included in such learning/training sessions

Short answer text

Responsible person(s): Please select (check the box(es)) the best person/s to be responsible to ensure that the learning objectives accommodate the different learner types of nurses during pain management learning/training sessions.

Director of Postgraduate Center of Nursing Education for KAMC and KASCH

One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Education for KAM...

Clinical facilitators in all areas of nursing care for KAMC and KASCH

One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility

 $\begin{smallmatrix} 0 & 0 & 0 \\ 0 & 0 & 0 \end{smallmatrix}$

Please add any **suggestion** on who may be the best person(s) to be responsible for ensuring that the learning objectives accommodate these learner types

| shared for inclusion within the training program. |
|---|
| 2 weeks before the due date for finalisation of the training program |
| 1 month before the due date of the training program |
| 3 months before the due date of the training program |
| Please add any suggestion regarding the most appropriate time frame to share for inclusion of these learning types into the training program Short answer text |
| |
| |
| Action statement 5: Incorporate different teaching approaches to accommodate diverse * learners and facilitators in the training of pain management Agree |
| Action statement 5: Incorporate different teaching approaches to accommodate diverse * learners and facilitators in the training of pain management Agree Disagree |

| Method 5.1 Ensure the inclusion of different teaching approaches in the offering of the training program | * |
|--|---|
| Agree | |
| Disagree | |
| ::: Please add any comment pertaining to this method | |
| Short answer text | |
| | |
| If you agree that different teaching approaches should be utilised during pain management training that accommodates diverse learners and facilitators, please tick all that should be utilized during training. | * |
| Writing reflective journals | |





Using role-playing activities

Please add any other suggestion for teaching approaches that can be utilized in such training

Responsible person(s): Please select (check the box(es)) next to the best person/s to be responsible to ensure that teaching approaches are included during pain management training.

Director of Postgraduate Center of Nursing Education for KAMC and KASCH

One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Education for KAM...

Clinical facilitators in all areas of nursing care for KAMC and KASCH

One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility

| program | jestion regarding the most appropriate time frame to be part of such a teaching |
|--|---|
| Short answer text | |
| Action statement 6: training program | Develop strategies to motivate nurses to participate in the short * |
| Agree | |
| Disagree | |
| | |
| Short answer text | |
| | |
| | |
| Method 6.1 If you a management trainir motivate nurses. | gree that strategies that motivate the nurses to participate in the pain ng program should be developed, please tick all strategies that you think will |
| Method 6.1 If you as management trainin motivate nurses. | gree that strategies that motivate the nurses to participate in the pain ng program should be developed, please tick all strategies that you think will on analysis to assess the pain management needs of the nurses |
| Method 6.1 If you as management trainin motivate nurses. | gree that strategies that motivate the nurses to participate in the pain ng program should be developed, please tick all strategies that you think will on analysis to assess the pain management needs of the nurses the development of the content of the training program |
| Method 6.1 If you a management trainin motivate nurses. | gree that strategies that motivate the nurses to participate in the pain ng program should be developed, please tick all strategies that you think will ion analysis to assess the pain management needs of the nurses the development of the content of the training program the development of learning goals and learning outcomes for the pain management trai |
| Method 6.1 If you a management trainin motivate nurses. | gree that strategies that motivate the nurses to participate in the pain ng program should be developed, please tick all strategies that you think will ion analysis to assess the pain management needs of the nurses the development of the content of the training program the development of learning goals and learning outcomes for the pain management trai e advantages of pain management competencies on (on what platform can this be done. |

| Please | add | any | comment | regarding | to | this | method |
|--------|-----|-----------|---------|-----------|----|------|--------|
| | | Sec. 1. J | | rogaranig | ~~ | | |

Short answer text

Responsible person(s): Please select (check the box(es)) the best person/s to be responsible * to the develop the mentioned strategies that will motivate nurses to participate in the short pain management training program related to their nursing care areas

Director of Postgraduate Center of Nursing Education for KAMC and KASCH

One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Education for KAM...

Clinical facilitators in all areas of nursing care for KAMC and KASCH

One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility

Please add any **suggestion** on who may be the best person(s) to be responsible to develop such strategies that would motivate nurses' participation

Short answer text

Time frame: Select the appropriate timeframe within which time the strategies should be developed that motivate nurses to participate in the short pain management training program related to their nursing care areas.

2 weeks before the due date for the finalisation of the training program

1 month before the training program starts

Please add any **suggestion** regarding the most appropriate time frame to develop such strategies motivating nurses' participation

| | î |
|---|---|
| Action statement 7: Motivate nurses to apply the knowledge gained in the training program into practice | |
| | |
| Agree | |
| Disagree | |
| | |
| | |
| Please add any comment pertaining to this action statement | |
| Short answer text | |
| | |
| | |
| | |
| | * |

Method 7.1 If you agree that nurses should be motivated kindly tick all you think that can be done to motivate them to apply their knowledge in practise.

Provide nurses with a certificate to recognize their application of pain management knowledge in their re...

Offer nurses the opportunity to take on the role of pain management experts who are competent in their ...

Allow the nurses to take part in planning outcomes of a pain management training program

Support nurses' SMART goals and pain management learning

Support what drives individual nurses to apply what they have learned about pain management

Assign grades for applying pain management knowledge in practice based on annual performance

Please add any suggestion regarding to this method

| * Responsible person(s): Please select (check the box(es)) next to the best person/s to be responsible to facilitate the implementation of the aspect to motivate nurses to apply their knowledge in practice. |
|---|
| Associate Executive Directors of Nursing for KAMC and KASCH |
| Clinical Directors of Nursing Operations for KAMC and KASCH |
| Nurse supervisors appointed by Executive Associate Directors of Nursing in every facility for KAMC and |
| Nurse Managers in all nursing care areas for KAMC and KASCH |
| Director of Postgraduate Center of Nursing Education for KAMC and KASCH |
| One Nurse educator appointed by the Director of the Postgraduate Center of Nursing Education for KAM |
| One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility |
| |
| Please add any suggestion on who may be the best person(s) to be responsible to facilitate the implementation of such motivation to nurses |
| Short answer text |
| |
| Time frame: Select the appropriate timeframe within which time the motivation should be * provided to nurses to apply their knowledge in practice |
| 1 week after the training program |
| 1-3 months after the training program |
| |
| 4-6 months after the training program |

Please add any **suggestion** regarding the most appropriate time frame to provide such a motivation to nurses

Short answer text

Thank you for taking the time to share your feedback!

Description (optional)

ANNEXURE 13: Recruitment letter: Round 2 e- Delphi RECRUITMENT LETTER ROUND 2

Dear Panellist,

I am Litaba Efraim Kolobe, a doctorate student at the University of South Africa. 'AN ACTION PLAN TO ENHANCE TRANSFER OF LEARNING OF PAIN MANAGE-MENT COMPETENCIES OF NURSES IN SAUDI ARABIAN HOSPITALS' is my study. The purpose of my study is to develop an action plan for enhancing the transfer of pain management competencies among nurses in Saudi Arabian teaching hospitals. This study was approved by the UNISA College of Human Sciences Research Ethics Committee under reference number REC-012714-039.

You are cordially invited to participate in the **second round** of validation of the developed action plan by using the embedded validation tool. In this second round, all the information received from panellists in the first round was analysed, and their suggestions were incorporated. The items for which consensus has already been reached are included in the action plan. You are therefore requested to respond ONLY to items where no consensus was reached. For example, if all statements indicate **'CONSEN-SUS REACHED**,' you should **disregard these sections** and **ONLY** respond to those that do not contain this statement. Please provide your honest opinion once more regarding whether you agree or disagree with the action statements, the methods used to accomplish the actions, the individuals who need to take responsibility, and the appropriate time frames. Please include any additional comments or suggestions for enhancing the action plan. Your participation will be kept anonymous and confidential, and your specific contributions will not be shared with other panellists, as all information will be gathered via software programs in bulk.

If you agree to take part in the **second round** of the validation of the action plan, kindly click on the link via the **Google Forms software** provided at the end of this letter. Instructions are provided as you go along completing the first round.

Your participation is voluntary, and you may choose not to take part, by just ignoring the invitation and not clicking on the link to access the action plan and

validation tool. You still have a chance to withdraw even after completing the validation tool, by not submitting your answers. All comments and suggestions for improvement will be incorporated and forwarded to a second round until there is a 75% consensus among all panellists. By clicking the link provided below, you will be able to access the instrument. It will only take 30 to 45 minutes for you to complete this questionnaire. I would greatly appreciate it if you would complete the questionnaire at your convenience by opening the link below.

https://docs.google.com/forms/d/1Fb20jtww9i9w36qo2rjpka4xehMbENj9jTZ-Yju_ZU8/edit

Please submit your answers within 10 days after receiving this invitation.

Because participation in this study is voluntary, you will not be remunerated. Your input will, however, be beneficial to allow me to revise the action plan by implementing all suggested changes by all panellists and share the revised action plan with you again and again until consensus amongst all panellists is achieved. The results of the study may be published, but your personal information and input will not be able to be traced back to you.

After completion, please click the submit button.

Please feel free to contact me by at Mobile +966503920421 or by email at <u>kolobe66@yahoo.com</u> if you have any questions regarding this research study. You may also contact my research supervisor Prof Lizeth Roets at +27 12 429 2226 or <u>roetsl@unisa.ac.za</u>. If you have any other concerns, you may also contact the College research ethics committee at <u>Khankb@unisa.ac.za</u>, the chairperson of the committee.

Your contribution and time will be appreciated.

Kind regards.

Litaba Efraim Kolobe

ANNEXURE 14: Gatekeeper letter: Request to recruit Delphi panellists for round 1

Assistance with recruitment of Delphi participants Round 1

TO : Nurse Managers

FROM: Litaba Efraim Kolobe

Pain Management Nurse Specialist Number 47295

SUBJECT : Request to recruit Nurses and CRNS to participate as Delphi panellists

from your ward/unit

My name is Litaba Efraim Kolobe and I am a registered doctoral student at the University of South Africa. The title of my study is 'An Action Plan to Enhance Transfer of Learning of Pain Management Competencies of Nurses in Saudi Arabian **Teaching Hospitals**'. My study aims to develop an action plan to enhance the transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals. I already received approval from the Research Ethics Committee of the Deof South partment of Health Studies, University Africa (Ref # 36640085_CREC_CHS_2023) as well as an ethics approval certificate from King Abdullah International Medical Research Center (KAIMRC) and Institutional Review Board (IRB) (SP18/036/R) attached.

My study is at the final phase of validating the action plan drafted, few rounds will be conducted with the nurses who volunteered to participate in your unit as Delphi panellists. Delphi panel will be a few numbers of nurses and clinical resources who will validate the action plan drafted in this study until a consensus of 75% is reached. Your

602

ward /unit has been selected, therefore for me to collect data, your role will be to assist in selecting volunteer nurses or clinical resource nurses to partake in the study will be appreciated. May you please forward the attached recruitment letter to all registered nurses or clinical resource nurses interested in research and who meet the following inclusion/eligibility criteria:

- 1. Must have an interest in pain management.
- 2. Nurses who attended at least one pain management workshop within the past 3 years.
- Nurses who attended ward in-service training about pain management in the past 12 months.
- 4. Clinical facilitators (clinical resource nurses) responsible for pain management training of nurses in those nursing care divisions mentioned above.
- 5. Be committed to at least 3 rounds of Delphi.

I would appreciate it if you shared the recruitment letter containing the link and IRB approval with the nurses who will voluntarily participate. Participants will complete their opinions on the link provided in the attached recruitment letter. I will also inform you when we go for the next round.

I appreciate your time and willingness to assist.

Yours Sincerely

Litaba Efraim Kolobe

ANNEXURE 15: Draft 2: Action plan with embedded validation tool

Instructions:

Dear Panellist

Thank you for your willingness to participate in the second round of the Delphi. Thank you for your inputs, your time and experience are appreciated. Where consensus was researched between the opinions of all panellists it is indicated as such in this second draft. Where consensus was not reached, you will have the opportunity to provide your inputs again.

Action statement 1: Motivate nurses to further their studies (CONSENSUS REACHED)

Method 1.1: Develop a policy to motivate nurses to improve their nursing qualifications (CONSENSUS REACHED)

1.1.1 The following items must be included in the policy nurses to improve their qualifications

- A certificate issued as an acknowledgment of nurses pursuing distance learning
- One day off for attending a one-day pain management program
- A monetary incentive after completion of a new formal qualification (degree or diploma
- A monetary incentive after completion of distance learning programs
- Free accommodation for the period of study leave

(Agreement by ≥60% of panellists)

Responsible person(s): Please select the BEST possible individual/individuals to take responsibility to develop a policy to motivate nurses to improve their nursing qualifications: **PLEASE ONLY CHOOSE ONE OPTION.**

- 1. Nursing policy committee representatives appointed by the Associate Executive Directors of Nursing for KAMC and KASCH
- 2. Clinical Directors of Nursing Operations for KAMC and KASCH
- 3. Director of Postgraduate Center of Nursing Education for KAMC and KASCH

The policy to motivate nurses to improve their nursing qualifications must be developed within 4-6 months after approval of the action plan (CONSENSUS REACHED)

Method 1.2: Present and negotiate for the implementation of the policy to motivate nurses to improve their nursing qualifications to the Ministry of National Guard Health Affairs (MNGHA) through the Central Region Nursing Governance and Accountability Board (CONSENSUS REACHED)

Responsible person(s): Nursing policy committee representatives appointed by the Associate Executive Director of Nursing for KAMC and KASCH are the BEST possible individual/individuals to present and negotiate for the implementation of the policy to motivate nurses to improve their nursing qualifications to MNGHA through Central Region Nursing Governance and Accountability Board (CONSENSUS REACHED)

Time frame: Please select the most appropriate timeframe within which time the policy should be presented, and the implementation negotiated to the Ministry of National Guard Health Affairs (MNGHA): **PLEASE ONLY CHOOSE ONE OPTION.**

2. 9 months

Method 1.3: Include the policy in all hospitals` policies after approval by the Ministry of National Guard Health Affairs (MNGHA) **(CONSENSUS REACHED)**

Responsible person(s): Please select the best person(s) to be responsible for the inclusion of the policy in all hospitals' policies to motivate nurses to improve their nursing qualifications: **PLEASE ONLY CHOOSE ONE OPTION.**

- 1. Clinical Directors of Nursing Operations in every facility for KAMC and KASCH
- 2. Nurse Managers in all KAMC and KASCH nursing care areas

Time frame: Please select the most appropriate timeframe within which time the policy should be included in all hospitals' policies after approval by the Ministry of National Guard Health Affairs: **PLEASE ONLY CHOOSE ONE OPTION.**

٠

•

- a) 1 month
- b) 3 months

Suggestions/comments:

Action statement 2: Make appropriate and relevant pain management tools accessible to the nursing team in every clinical areas (CONSENSUS REACHED)

| • | | |
|---|-------|--|
| Method 2.1: Include pain assessment tools in an electronic pa- | Agree | |
| tient record system so that the nursing team can choose and ac- | | |
| cess the most appropriate. tool in all nursing care areas. | Disa- | |
| | gree | |

2.1.1 The following pain assessment tools that must be included and accessible on the electronic patient record system:

- PQRST
- CRIES

(Agreement by ≥60% of panellists)

Responsible person(s): Please select the best person(s) to be responsible to provide the electronic format of the pain assessment tools for inclusion in the electronic patient record system to be accessible to the nursing team in every nursing care area: **PLEASE ONLY CHOOSE ONE OPTION.**

- 1. Five Nursing health informatics appointed by Associate Directors of Nursing for KAMC and KASCH
- One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility for KAMC and KASCH

Time frame: Please select the most appropriate timeframe within which time the pain assessment tools should be included in the electronic patient record system: **PLEASE ONLY CHOOSE ONE OPTION**

1. 1-3 months

2. 4-6 months

Method 2.2: Involve nurse supervisors with pain management training to provide post-pain management training supervisory support to the nursing team on how to conduct pain assessments in all nursing care areas. **(CONSENSUS REACHED)**

Responsible person(s): Please select the best person(s) responsible to involve nurse supervisors with pain management training to provide post-pain management training and supervisory support to the nursing team on how to conduct pain assessments in all nursing care areas: **PLEASE ONLY CHOOSE ONE OPTION**

1) Clinical Directors of Nursing Operations for KAMC and KASCH

2) Clinical facilitators in all KAMC and KASCH nursing care areas

Time frame: Please select the most appropriate timeframe within which time the nurse supervisors should be involved to provide pain management training supervisory support to the nursing team on how to conduct pain assessment in all nursing care areas: **PLEASE ONLY CHOOSE ONE OPTION**

- Every shift when the need arises
- Every patient round when the need arises

Method 2.3: Make hospitals' internet-based resources accessible to the patients and family members to obtain other support about pain management **(CONSENSUS REACHED)**

2.3.1 The following Internet-based resources that must be accessible to the patients and family members to obtain other support about pain management:

- Patient pain management websites
- Patient pain management hotlines
- Pain management support groups
- Peer support groups

(Agreement by ≥60% of panellists)

Responsible person(s): Please select the best person(s) to be responsible to ensure internet-based resources should be accessible to patients and family members to provide support about pain management. **PLEASE ONLY CHOOSE ONE OP-TION**

- 1. Nurse Managers in all KAMC and KASCH nursing care areas
- 2. Charge nurses in all KAMC and KASCH nursing care areas
- 3. Registered nurses in all KAMC and KASCH nursing care areas

Time frame: The hospitals' internet-based resources to the patients and family members to obtain other support about pain management must be made accessible within 24-hour access 7 days a week. (CONSENSUS REACHED)

Method 2.4: Make hospitals' internet-based resources on pain management publications and electronic materials accessible to the nursing team in all nursing care areas. (**CONSENSUS REACHED**)

2.4.1 The following Internet-based resources on pain management publications, electronic materials, and organisations that specialises in pain management that must be accessible:

- Pain toolkits
- Videos on pain management
- Clinical updates

(Agreement by ≥60% of panellists)

Responsible person(s): Please select the BEST possible individual/individuals to ensure internet-based resources on pain management publications, electronic materials, and organisations that specialises in pain management should be accessible to the nursing team in all nursing care areas: **PLEASE ONLY CHOOSE ONE OP-TION**.

Nursing health informatics appointed by Associated Directors of Nursing for KAMC and KASCH

One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility for KAMC and KASCH

Time frame: The hospitals' internet-based resources on pain management publications and electronic materials to the nursing team in all nursing care areas must be made accessible within 24-hour access 7 days a week (CON-SENSUS REACHED)

Suggestions/comments:

Action statement 3: Develop a practice-oriented content specific short pain management training program. (CONSENSUS REACHED)

Method 3.1: Include practice-oriented pain management training content for all nursing care areas in the pain management program. **(CONSENSUS REACHED)**

3.1.1 The following specific practice-oriented pain management training content that must be included in all nursing care areas:

- Methods to promptly assess a patient's pain in all nursing areas
- The selection of appropriate pain intervention strategies based on the pain levels assessed
- Assessment of patients' pain in all nursing care areas
- The advantages and disadvantages of all pain management scales

(Agreement by ≥60% of panellists)

Responsible person(s): One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility is the best possible individual/individuals to be responsible to include specific practice-oriented pain management training content for all nursing care areas. **(CONSENSUS REACHED)**

Time frame: Please select the appropriate timeframe within which time the specific practice-oriented pain management training content should be provided for inclusion in the program. **PLEASE ONLY CHOOSE ONE OPTION**

- a) 1 month before the due date of the training program
- b) 3 months before the due date of the training program

Suggestions/comments:

Action statement 4: Develop a pain management short program that accommodate all learning types. (CONSENSUS REACHED)

Method 4.1: Incorporate different learner types during learning/training sessions. (CONSENSUS REACHED)

4.1.1 The following different learner types of nurses that must be included during

pain management learning/training sessions:

- Creative Learners
- Organised thinking learners
- Enthusiastic thinking learners

(Agreement by ≥60% of panellists)

4.1.2 The following different learning types that must be incorporated during pain

management learning/training sessions are achieved by using the following

learning styles:

- Creative learners by generating creative ideas in a group.
- Enthusiastic thinking learners by listening to the information actively, take part in the activity to practice the skill and participate in group discussion
- Organised thinking learners by solving different real-life problems

(Agreement by ≥60% of panellists)

Responsible person(s): One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility is the best possible person(s) to be

responsible for ensuring that the learning objectives accommodate the different learner types of nurses during pain management learning/training sessions. (CON-SENSUS REACHED)

Time frame: Please select the appropriate timeframe within which time the learning types must be shared for inclusion within the training program. **PLEASE ONLY CHOOSE ONE OPTION**

a) 1 month before the due date of the training program

b) 3 months before the due date of the training program

Suggestions/comments:

Action statement 5: Develop a pain management short course that motivates nurses to apply knowledge gained in the training program to practice. (CON-SENSUS REACHED)

Method 5.1: Ensure the inclusion of different teaching approaches in the offering of the training program. **(CONSENSUS REACHED)**

5.1 The following teaching approaches that must be utilised during pain management training that accommodates diverse learners and facilitators:

- Engaging in focus groups
- Using role-playing activities

(Agreement by ≥60% of panellists)

Responsible person(s): One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility is the best possible person(s) to be

responsible to ensure that teaching approaches are included during pain management training. **(CONSENSUS REACHED).**

Time frame: Please select the appropriate timeframe within which time different teaching approaches should be part of the teaching program before implementation. **PLEASE ONLY CHOOSE ONE OPTION**

- 1. 1 month before the due date of the training program
- 2. 3 months before the due date of the training program

Suggestions/comments

Action statement 6: Develop strategies to motivate nurses to participate in the short training program. (CONSENSUS REACHED).

6.1 The following are strategies that will motivate nurses to participate in the pain

management training program:

- 1. Involve nurses in the development of learning goals and learning outcomes for the pain management training program relevant to their nursing care areas.
- 2. Involve nurses in the development of the content of the training program
- 3. Communicate the advantages of pain management competencies
- 4. Create a supportive learning environment in nursing care areas

(Agreement by ≥60% of panellists)

Responsible person(s): Please select the best person(s) to be responsible to develop the mentioned strategies that will motivate nurses to participate in the short pain management training program related to their nursing care areas. **PLEASE ONLY CHOOSE ONE OPTION**

- 1. Clinical facilitators in all areas of nursing care for KAMC and KASCH
- One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility

Time frame: The strategies that motivate nurses to participate in the short pain management training program related to their nursing care areas should be developed within 1 month before the training program starts. (CONSENSUS REACHED).

Suggestions/comments:

Action statement 7: Motivate nurses to apply the knowledge gained in the training program into practice. (CONSENSUS REACHED)

7.1: The following methods that can be implemented to motivate nurses to apply

their knowledge in practice:

- Offer nurses the opportunity to take on the role of pain management experts who are competent in their field.
- Support nurses' SMART goals and pain management learning
- Support drives individual nurses to apply what they have learned about pain management.
- Allow the nurses to take part in planning outcomes of a pain management training program

Assign grades for applying pain management knowledge in practice based
 on annual performance

(Agreement by ≥60% of panellists)

Responsible person(s): Please select the best person(s) to be responsible to facilitate the implementation of the aspect to motivate nurses to apply their knowledge in practice. **PLEASE ONLY CHOOSE ONE OPTION**

1. Nurse Managers in all nursing care areas for KAMC and KASCH

2. Clinical Directors of Nursing Operations for KAMC and KASCH

Time frame: Please select the appropriate timeframe within which time the aspects for motivation should be provided to nurses to apply their knowledge in practice.

- 1. 1-3 months after the training program
- 2. 4-6 months after the training program

Suggestions/comments:

ANNEXURE 16:

Gatekeeper letter: to recruit panellists for round 2

Assistance with recruitment of Delphi participants Round 2

TO : Nurse Managers

FROM: Litaba Efraim Kolobe

Pain Management Nurse Specialist Number 47295

SUBJECT : Request to recruit Nurses and CRNS to participate as Delphi panellists

from your ward/unit

My name is Litaba Efraim Kolobe and I am a registered doctoral student at the University of South Africa. The title of my study is 'An Action Plan to Enhance Transfer of Learning of Pain Management Competencies of Nurses in Saudi Arabian Teaching Hospitals'. My study aims to develop an action plan to enhance the transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals. I already received approval from the Research Ethics Committee of the De-Health partment of Studies, University of South Africa (Ref # 36640085 CREC_CHS_2023) as well as an ethics approval certificate from King Abdullah International Medical Research Center (KAIMRC) and Institutional Review Board (IRB) (SP18/036/R) attached.

My study is at the **second round** of validating the action plan drafted, therefore may you assist in recruiting the same nurses and clinical resource nurses who volunteered

initially in the first-round round to participate in your unit as Delphi panellists for continuity. May you please forward the attached recruitment letter to all registered nurses clinical and clinical resource nurses who participated during the first round as stated in the first round to meet the following inclusion criteria:

- 1. Must have an interest in pain management.
- Nurses who attended at least one pain management workshop within the past 3 years.
- 3. Nurses who attended ward in-service training about pain management in the past 12 months.
- 4. Clinical facilitators (clinical resource nurses) responsible for pain management training of nurses in those nursing care divisions mentioned above.
- 5. Be committed to at least 3 rounds of Delphi.

I would appreciate it if you would once more share the recruitment letter containing the link and IRB approval with the nurses who will voluntarily participate. Participants will complete their opinions on the link provided in the attached recruitment letter. I will also inform you when we go for the **next round**.

I appreciate your time and willingness to assist.

Yours Sincerely

Litaba Efraim Kolobe

ANNEXURE 17:

FINDINGS DURING ROUND 2 FOR ACTION STATEMENTS 1 TO 7

Action statement 1: Motivate nurses to further their studies (CONSENSUS REACHED)

Method 1.1: Develop a policy to motivate nurses to improve their nursing qualifications (CONSENSUS REACHED)

- 1.1.1 The following Items that must be included in the policy nurses to improve their qualifications
 - A certificate issued as an acknowledgment of nurses pursuing distance learning
 - One day off for attending a one-day pain management program
 - A monetary incentive after completion of a new formal qualification (degree or diploma
 - A monetary incentive after completion of distance learning programs
 - Free accommodation for the period of study leave
 - (Agreement by ≥60% of panellists)

| Responsible person(s) to develop a policy to moti- | Re- | | Consen- |
|--|------|----------|--------------------|
| vate nurses to improve their nursing qualifications (n | spon | ses | sus |
| = 12; N = 12) | | | Reached (≥ 75%) |
| | n = | f = % | Yes/No |

| 1. | Nursing policy committee representatives appointed by the Associate Executive Directors of Nursing for KAMC and KASCH | 5 | 41.7 | No |
|----|---|---|------|----|
| 2. | Clinical Directors of Nursing Operations for KAMC and KASCH | 4 | 33.3 | |
| 3. | Director of Postgraduate Center of Nursing Education for KAMC and KASCH | 3 | 25 | |

The policy to motivate nurses to improve their nursing qualifications must be developed and finalised 4-6 months after approval of the action plan (CON-SENSUS REACHED)

Method 1.2: Present and negotiate for the implementation of the policy to motivate nurses to improve their nursing qualifications to the Ministry of National Guard Health Affairs (MNGHA) through the Central Region Nursing Governance and Accountability Board (CONSENSUS REACHED)

Responsible person(s): Nursing policy committee representatives appointed by the Associate Executive Director of Nursing for KAMC and KASCH are the BEST possible individual/individuals to present and negotiate for the implementation of the policy to motivate nurses to improve their nursing qualifications to MNGHA through Central Region Nursing Governance and Accountability Board (CONSENSUS REACHED)

| Time frame required to present the policy and of ne- | Re- | Consen- |
|--|---------|---------|
| gotiate the implementation to the Ministry of National | sponses | sus |
| Guard Health Affairs (MNGHA) (n = 12; N = 12) | | Reached |

| | | | (≥ 75%) |
|--|------------------|------------------|--------------------------------------|
| | n = | f = % | Yes/No |
| 1. 6 months | 8 | 66.7 | |
| 2. 9 months | 4 | 33.3 | Νο |
| Method 1.3: Include the policy in all hospitals` policies after of National Guard Health Affairs (MNGHA) (CONSENSUS | er appr S REA | oval by CHED) | the Ministry |
| Responsible person(s) to be responsible for the inclusion of the policy in all hospitals' policies to motivate nurses to improve their nursing qualifications ($n = 12$; $N = 12$) | Re- sponses | | Consen- sus Reached (≥ 75%) |
| | n = | f = % | Yes/No |
| a) Clinical Directors of Nursing Operations in every facil- ity for KAMC and KASCH | 4 | 33.3 | No |
| b) Nurse Managers in all KAMC and KASCH nursing care areas | 8 | 66.7 | |

| Time frame required to include the policy in all hospitals' policies after approval of the action plan by the Ministry of National Guard Health Affairs ($n = 12$; $N = 12$) | Re- sponses | | Consen- sus Reached (≥ 75%) |
|--|----------------|----------|--------------------------------------|
| | n = | f = % | Yes/No |
| a) 1 month | 3 | 25 | |
| b) 3 months | 9 | 75 | Yes |

Action statement 2: Make appropriate and relevant pain management tools accessible to the nursing team in every clinical area. (CONSENSUS REACHED)

Method 2.1: Include pain assessment tools in an electronic patient record system so that the nursing team can choose and access the most appropriate. tool in all nursing care areas. (n = 12; N = 12)

| F | Response | S | Consen- sus |
|-------|----------|-------|----------------|
| | | | Reached |
| | | | (≥ 75%) |
| | n = | f = % | Yes/No |
| Agree | 12 | 100 | Yes |

| | Disa- gree | 0 | 0 | | | |
|--|---------------|---|---|--|--|--|
| 2.1.1 The following pain assessment tools that must be included and accessible on the electronic patient record system. | | | | | | |
| PQRST | | | | | | |
| CRIES | | | | | | |
| (Agreement by ≥60% of panellists) | | | | | | |
| | | | | | | |

| Responsible person(s) to provide the elec- tronic format of the pain assessment tools for inclusion in the electronic patient record sys- tem to be accessible to the nursing team in every nursing care area | Responses | | Consen- sus Reached (≥ 75%) | | |
|--|-----------|-------|--------------------------------------|--|--|
| | n = | f = % | Yes/No | | |
| Five Nursing health informatics ap- pointed by Associate Directors of Nurs- ing for KAMC and KASCH | 3 | 25 | Yes | | |
| One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility for KAMC and KASCH | 9 | 75 | | | |
| Time frame required to include the pain assessment tools in the electronic patient record system (n = 12; N = 12). | | | | | |
| 1. 1-3 months | 7 | 58.3 | | | |
| | | No | | | |
|--|-----------|-------|--------------------------------------|--|--|
| 2. 4-6 months | 5 | 41.7 | | | |
| Method 2.2: Involve nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessments in all nursing care areas. (CONSENSUS REACHED) | | | | | |
| Responsible person(s) to involve nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessments in all nursing care areas ($n = 12$; $N = 12$). | Responses | | Consen- sus Reached (≥ 75%) | | |
| | n = | f = % | Yes/No | | |
| Clinical Directors of Nursing Operations for KAMC and KASCH | 6 | 50 | Νο | | |
| Clinical facilitators in all KAMC and KASCH nursing care areas | 6 | 50 | INO | | |
| Time frame required to involve the nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessment in all nursing care areas ($n = 12$; $N = 12$). | | | | | |
| • Every shift when the need arises | 50 | 50 | | | |
| • Every patient round when the need | 50 | 50 | No | | |

arises

Method 2.3: Make hospitals' internet-based resources accessible to the patients and family members to obtain other support about pain management **(CONSENSUS REACHED)**

2.3.1 The following Internet-based resources that must be accessible to the patients and family members to obtain other support about pain management:

- Patient pain management websites
- Patient pain management hotlines
- Pain management support groups
- Peer support groups

(Agreement by ≥60% of panellists)

Responsible person(s) to ensure internet-based resources should be accessible to patients and family members to provide support about pain management (n = 12; N = 12).

| Nurse Managers in all KAMC and KASCH nursing care areas | 9 | 75 | Yes |
|--|---|----|-----|
| 2. Charge nurses in all KAMC and KASCH nursing care areas | 0 | 0 | |
| Registered nurses in all KAMC and KASCH nursing care areas | 3 | 25 | |

Time frame: The hospitals' internet-based resources that should be accessible to the patients and family members to obtain other support about pain management must be made **24-hour access 7 days a week. (CONSENSUS REACHED)**

Method 2.4: Make hospitals' internet-based resources on pain management publications and electronic materials accessible to the nursing team in all nursing care areas. (**CONSENSUS REACHED**)

2.4.1 The following internet-based resources on pain management publications, electronic materials, and organisations that specialises in pain management that must be accessible

- Pain toolkits
- Videos on pain management
- Clinical updates

(Agreement by ≥60% of panellists)

Responsible person(s): The best possible individual/individuals to ensure internetbased resources on pain management publications, electronic materials, and organisations that specialises in pain management should be accessible to the nursing team in all nursing care areas (n12 = ; N = 12).

| 1. | Nursing health informatics appointed by | F | 44 7 | | | | | |
|----|---|---|--------|----|---|---|------|--|
| | Associated Directors of Nursing for KAMC | Э | 5 41.7 | 5 | Э | Э | 41.7 | |
| | and KASCH | | | No | | | | |
| 2. | One Pain nurse specialist appointed by | 7 | F0 0 | | | | | |
| | Clinical Directors of Nursing Operations in | 7 | 58.3 | | | | | |
| | every facility for KAMC and KASCH | | | | | | | |
| | | | | | | | | |

Time frame required to make pain management publications and electronic materials available via the hospitals internet to the nursing team in all nursing care areas must be **24-hour access 7 days a week (CONSENSUS REACHED)**

Action statement 3: Develop a practice-oriented content specific short pain management training program. (CONSENSUS REACHED)

Method 3.1: Include practice-oriented pain management training content for all nursing care areas in the pain management program. **(CONSENSUS REACHED)**

3.1.1 The following specific practice-oriented pain management training content that must be included in all nursing care areas

- Methods to promptly assess a patient's pain in all nursing areas
- The selection of appropriate pain intervention strategies based on the pain levels assessed
- Assessment of patients' pain in all nursing care areas
- The advantages and disadvantages of all pain management scales

(Agreement by ≥60% of panellists)

Responsible person(s): One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility is the best possible individual/individuals to be responsible to include specific practice-oriented pain management training content for all nursing care areas. (CONSENSUS REACHED)

| Time frame within which time the specific prac- tice-oriented pain management training content should be provided for inclusion in the program (n = 12; N = 12). | Res | sponses | Consen- sus Reached (≥ 75%) |
|---|--------|----------|--------------------------------------|
| | n = | f = % | Yes/No |

| a) 1 month before the due date of the training program | 2 | 16.7 | Yes |
|--|----|------|-----|
| b) 3 months before the due date of the train- ing program | 10 | 83.3 | |

Action statement 4: Develop a pain management short program that accommodate all learning types. (CONSENSUS REACHED)

Method 4.1: Incorporate different learner types during learning/training sessions. (CONSENSUS REACHED)

4.1.1 The following different learner types of nurses that must be included during pain management

learning/training sessions:

- Creative Learners
- Organised thinking learners
- Enthusiastic thinking learners

(Agreement by ≥60% of panellists)

4.1.2 The following different learning types that must be incorporated during pain management learning/training sessions are achieved by using the following learning styles:

- Creative learners by generating creative ideas in a group.
- Enthusiastic thinking learners by listening to the information actively, take part in the activity to practice the skill and participate in group discussion
- Organised thinking learners by solving different real-life problems

(Agreement by ≥60% of panellists)

Responsible person(s): One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility is the best responsible for ensuring that the learning objectives accommodate the different learner types of nurses during pain management learning/training sessions. (CONSENSUS REACHED)

| Time frame within which time the learning types must be shared for inclusion within the training program. (n = 12; N = 12). | Responses | | Consen- sus Reached (≥ 75%) |
|--|-----------|-------|--------------------------------------|
| | n = | f = % | Yes/No |
| 1 month before the due date of the training program | 2 | 16.7 | Yes |
| 2. 3 months before the due date of the training program | 10 | 83.3 | |

Action statement 5: Develop a pain management short course that motivates nurses to apply knowledge gained in the training program to practice. (CON-SENSUS REACHED)

Method 5.1: Ensure the inclusion of different teaching approaches in the offering of the training program. **(CONSENSUS REACHED)**

5.1.1 The following teaching approaches that must be included during pain management training that accommodates diverse learners and facilitators.

- Engaging in focus groups
- Using role-playing activities

(Agreement by ≥60% of panellists)

Responsible person(s): One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility is the best possible person(s) to be responsible to ensure that teaching approaches are included during pain management training. **(CONSENSUS REACHED).**

| Time frame within which time should different teaching approaches be part of the teaching program before implementation. (n = 12; N = 12) | Responses | | Consen- sus reached |
|--|-----------|----------|---------------------------|
| | n = | f = % | Yes/No |
| 1 month before the due date of the training program | 3 | 25 | Yes |
| 2. 3 months before the due date of the training program | 9 | 75 | |

Action statement 6: Develop strategies to motivate nurses to participate in the short training program. (CONSENSUS REACHED).

6.1 The following are strategies that will motivate nurses to participate in the pain management training program:

 Involve nurses in the development of learning goals and learning outcomes for the pain management training program relevant to their nursing care areas.

- 2. Involve nurses in the development of the content of the training program
- 3. Communicate the advantages of pain management competencies
- 4. Create a supportive learning environment in nursing care areas

(Agreement by ≥60% of panellists)

Responsible person(s) to be responsible to develop the mentioned strategies that will motivate nurses to participate in the short pain management training program related to their nursing care areas. (n = 12; N = 12)

| 1. Clinical facilitators in all areas of nursing care for KAMC and KASCH | 3 | 25 | Yes |
|---|---|----|-----|
| One Pain nurse specialist ap- pointed by Clinical Directors of Nursing Operations in every facility | 9 | 75 | |

Time frame: The strategies that motivate nurses to participate in the short pain management training program related to their nursing care areas should be developed **1 month before the training program starts. (CONSENSUS REACHED).**

Action statement 7: Motivate nurses to apply the knowledge gained in the training program into practice. (CONSENSUS REACHED)

Method 7.1: The following methods that can be implemented to motivate nurses to apply their knowledge in practice:

- 1. Offer nurses the opportunity to take on the role of pain management experts who are competent in their field.
- 2. Support nurses' SMART goals and pain management learning
- 3. Support drives individual nurses to apply what they have learned about pain management.

| 4. Allow the nurses to take part in planning outcomes of a pain management training program 5. Assign grades for applying pain management knowledge in practice based on annual performance (Agreement by ≥60% of panellists) | | | |
|---|----------------|----------|--------------------------------------|
| Responsible person(s) to facilitate the implemen- tation of the aspect to motivate nurses to apply their knowledge in practice | Responses | | Consen- sus Reached (≥ 75%) |
| | n = | f = % | Yes/No |
| 8. Nurse Managers in all nursing care areas for KAMC and KASCH | 3 | 25 | Yes |
| Clinical Directors of Nursing Operations for KAMC and KASCH | 9 | 75 | |
| Time frame required to implement the methods to motivate nurses to apply their knowledge in practice | Re- sponses | | Consen- sus reached |
| | n = | f = % | Yes/No |
| 10.1-3 months after the training program | 6 | 50 | |

| 11. 4-6 months after the training program | 6 | 50 | No |
|--|---|----|----|
|--|---|----|----|

ANNEXURE 18: DRAFT 3 ACTION PLAN WITH EMBEDDED VALIDATION TOOL

Instructions:

Dear Panellist

Thank you for your willingness to participate in the third round of the Delphi. Thank you for your inputs, your time and experience are appreciated. Where consensus was researched between the opinions of all panellists it is indicated as such in this second draft. Where consensus was not reached, you will have the opportunity to provide your inputs again.

Action statement 1: Motivate nurses to further their studies (CONSENSUS REACHED)

Method 1.1: Develop a policy to motivate nurses to improve their nursing qualifications **(CONSENSUS REACHED)**

1.1.1 The following items to be included in the policy nurses to improve their qualifications

- A certificate issued as an acknowledgment of nurses pursuing distance learning
- One day off for attending a one-day pain management program
- A monetary incentive after completion of a new formal qualification (degree or diploma
- A monetary incentive after completion of distance learning programs
- Free accommodation for the period of study leave

(Agreement of ≥60% by panellists)

Please select the BEST possible person(s) to take responsibility to develop a policy that motivates nurses to improve their nursing qualifications. PLEASE ONLY CHOOSE ONE OPTION

- 1. Nursing policy committee representatives appointed by the Associate Executive Directors of Nursing for KAMC and KASCH
- 2. Clinical Directors of Nursing Operations for KAMC and KASCH

Time frame: Policy to motivate nurses to improve their nursing qualifications must be developed and finalised **4-6 months after approval of the action plan (CONSENSUS REACHED)**

Method 1.2: Present and negotiate for the implementation of the policy to motivate nurses to improve their nursing qualifications to the Ministry of National Guard Health Affairs (MNGHA) through the Central Region Nursing Governance and Accountability Board **(CONSENSUS REACHED)**

Responsible person(s): Nursing policy committee representatives appointed by the Associate Executive Director of Nursing for KAMC and KASCH are the BEST possible individual/individuals to present and negotiate for the implementation of the policy to motivate nurses to improve their nursing qualifications to MNGHA through Central Region Nursing Governance and Accountability Board (CONSENSUS REACHED)

Time frame within which time the policy should be presented, and the implementation negotiated to the Ministry of National Guard Health Affairs (MNGHA) PLEASE ONLY CHOOSE ONE OPTION

1. 6 months

2. 9 months

Method 1.3: Include the policy in all hospitals` policies after approval by the Ministry of National Guard Health Affairs (MNGHA) **(CONSENSUS REACHED)**

Responsible person(s): Please select the BEST person(s) to be responsible for the inclusion of the policy in all hospitals' policies to motivate nurses to improve their nursing qualifications. **PLEASE ONLY CHOOSE ONE OPTION**

- 1. Clinical Directors of Nursing Operations in every facility for KAMC and KASCH
- 2. Nurse Managers in all KAMC and KASCH nursing care areas

Time frame: The policy should be included in all hospitals' policies 3 months after approval by the Ministry of National Guard Health Affairs (**CONSENSUS REACHED**)

Please kindly add any suggestions/comments:

Action statement 2: Make appropriate and relevant pain management tools accessible to the nursing team in every clinical areas. (CONSENSUS REACHED)

Method 2.1: Include pain assessment tools in an electronic patient record system so that the nursing team can choose and access the most appropriate. tool in all nursing care areas. (CONSENSUS REACHED)

2.1.1 The following pain assessment tools that must be included and accessible on the electronic patient record system.

- PQRST
- CRIES

(Agreement of ≥60% by panellists)

Responsible person(s): Please select the BEST person(s) to be responsible to provide the electronic format of the pain assessment tools for inclusion in the electronic patient record system to be accessible to the nursing team in every nursing care area. **PLEASE ONLY CHOOSE ONE OPTION**

- Five Nursing health informatics appointed by Associate Directors of Nursing for KAMC and KASCH
- 2. One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility for KAMC and KASCH

Time frame: Please select the most appropriate timeframe within which time the pain assessment tools should be included in the electronic patient record system:

1. 1-3 months

2. 4-6 months

Method 2.2: Involve nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessments in all nursing care areas. **(CONSENSUS REACHED)**

Responsible person(s): Please select the best person(s) responsible to involve nurse supervisors with pain management training to provide supervisory support to the nursing team on how to conduct pain assessments in all nursing care areas. **PLEASE ONLY CHOOSE ONE OPTION**

- 1. Clinical Directors of Nursing Operations for KAMC and KASCH
- 2. Clinical facilitators in all KAMC and KASCH nursing care areas

Time frame: Please select the most appropriate timeframe within which time the nurse supervisors with pain management training should be involved to provide supervisory support to the nursing team on how to conduct pain assessment in all nursing care areas. PLEASE ONLY CHOOSE ONE OPTION

Every shift when the need arises

Every patient round when the need arises

Method 2.3: Make hospitals' internet-based resources accessible to the patients and family members to obtain other support about pain management **(CONSENSUS REACHED)**

2.3.1 The following internet-based resources that must be accessible to the patients and family members to obtain other support about pain management

- Patient pain management websites
- Patient pain management hotlines
- Pain management support groups
- Peer support groups

(Agreement of ≥60% by panellists)

Responsible person(s): Nurse Managers in all KAMC and KASCH nursing care areas are the best persons to ensure internet-based resources are accessible to patients and family members to provide support about pain management **(CONSENSUS REACHED)**

Time frame: The hospitals' internet-based resources accessible to the patients and family members to obtain other support about pain management must be **24-hour access 7 days a week. (CONSENSUS REACHED**

Method 2.4: Make hospitals' internet-based resources on pain management publications and electronic materials accessible to the nursing team in all nursing care areas. (**CON-SENSUS REACHED**)

2.4.1 The following Internet-based resources on pain management publications, electronic materials, and organisations that specialises in pain management that must be accessible

- Pain toolkits
- Videos on pain management
- Clinical updates

(Agreement of ≥60% by panellists)

Responsible person(s): The best possible individual/individuals to ensure internet-based resources on pain management publications, electronic materials, and organisations that specialises in pain management should be accessible to the nursing team in all nursing care areas. **PLEASE ONLY CHOOSE ONE OPTION**

- Nursing health informatics appointed by Associated Directors of Nursing for KAMC and KASCH
- 2. One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility for KAMC and KASCH

Time frame: The hospitals' internet-based resources on pain management publications and electronic materials accessible to the nursing team in all nursing care areas must be made **24-hour access 7 days a week (CONSENSUS REACHED)**

Please kindly add any suggestions/comments:

Action statement 3: Develop a practice-oriented content specific short pain management training program. (CONSENSUS REACHED)

Method 3.1: Include practice-oriented pain management training content for all nursing care areas in the pain management program. **(CONSENSUS REACHED)**

3.1.1 The following specific practice-oriented pain management training content that should be included in all nursing care areas

- Methods to promptly assess a patient's pain in all nursing areas
- The selection of appropriate pain intervention strategies based on the pain levels assessed
- Assessment of patients' pain in all nursing care areas
- The advantages and disadvantages of all pain management scales

(Agreement of ≥60% by panellists)

Responsible person(s): One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility is the best possible individual to be responsible to include specific practice-oriented pain management training content for all nursing care areas. (CONSENSUS REACHED)

Time frame: The specific practice-oriented pain management training content should be provided for inclusion in the program 3 months before the due date of the training program (CONSENSUS REACHED)

Please kindly add any suggestions/comments:

Action statement 4: Develop a short pain management program that accommodate all learning types. (CONSENSUS REACHED)

Method 4.1: Incorporate different learner types during learning/training sessions. (CON-SENSUS REACHED)

4.1.1 The following different learner types of nurses that must be included during pain management

learning/training sessions:

- Creative Learners
- Organised thinking learners
- Enthusiastic thinking learners

(Agreement of ≥60% by panellists)

4.1.2 The following different learning types that should be incorporated during pain management

learning/training sessions are achieved by using the following learning styles:

- Creative learners by generating creative ideas in a group.
- Enthusiastic thinking learners by listening to the information actively, take part in the activity to practice the skill and participate in group discussion
- Organised thinking learners by solving different real-life problems

(Agreement of ≥60% by panellists)

Responsible person(s): One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility is the best responsible person for ensuring that the learning objectives accommodate the different learner types of nurses during pain management learning/training sessions. (CONSENSUS REACHED)

Time frame: The learning types must be shared for inclusion within the training program **3** months before the due date of the training program. (CONSENSUS REACHED)

Please kindly add any suggestions/comments:

Action statement 5: Develop a pain management short course that motivates nurses to apply knowledge gained in the training program to practice. (CONSENSUS REACHED) **Method 5.1:** Ensure the inclusion of different teaching approaches in the offering of the training program. **(CONSENSUS REACHED)**

Teaching approaches that should be utilised during pain management training that accommodates diverse learners and facilitators.

- Engaging in focus groups
- Using role-playing activities

(Agreement of ≥60% by panellists)

Responsible person(s): One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility is the best possible person to be responsible to ensure that teaching approaches are included during pain management training. (CON-SENSUS REACHED).

Time frame: Different teaching approaches must be part of the teaching program before implementation **3 months before the due date of the training program (CONSENSUS REACHED)**

Please kindly add any suggestions/comments:

Action statement 6: Develop strategies to motivate nurses to participate in the short training program. (CONSENSUS REACHED).

6.1 The following are strategies that will motivate nurses to participate in the pain management training program:

- Involve nurses in the development of learning goals and learning outcomes for the pain management training program relevant to their nursing care areas.
- Involve nurses in the development of the content of the training program
- Communicate the advantages of pain management competencies

Create a supportive learning environment in nursing care areas

(Agreement of ≥60% by panellists)

Responsible person(s): One pain nurse specialist appointed by Clinical Directors of nursing operations in every facility is the best person to be responsible for developing the mentioned strategies that will motivate nurses to participate in the short pain management training program related to their nursing care areas. (CONSENSUS REACHED)

Time frame: The strategies that motivate nurses to participate in the short pain management training program related to their nursing care areas should be developed **1 month before the training program starts. (CONSENSUS REACHED).**

Please kindly add any suggestions/comments:

Action statement 7: Motivate nurses to apply the knowledge gained in the training program into practice. (CONSENSUS REACHED)

7.1: The following are methods that can be implemented to motivate nurses to apply their

knowledge in practice:

- Offer nurses the opportunity to take on the role of pain management experts who are competent in their field.
- Support nurses' SMART goals and pain management learning
- Support drives individual nurses to apply what they have learned about pain management.
- Allow the nurses to take part in planning outcomes of a pain management training program
- Assign grades for applying pain management knowledge in practice based on annual performance

(Agreement of ≥60% by panellists)

Responsible person(s): Nurse Managers in all nursing care areas for KAMC and KASCH are the best persons to be responsible to facilitate the implementation of the aspect to motivate nurses to apply their knowledge in practice. **(CONSENSUS REACHED)**

Time frame: Please select the appropriate timeframe within which time the aspects for motivation should be provided to nurses to apply their knowledge in pr Nurse Managers in all KAMC and KASCH nursing care areas are the best persons responsible to ensure internet-based resources are accessible to patients and family members to provide support about pain management practice PLEASE ONLY CHOOSE ONE OPTION

1. 1-3 months after the training program

2. 4-6 months after the training program

ANNEXURE 19: RECRUITMENT LETTER ROUND 3

Dear Panellist,

I am Litaba Efraim Kolobe, a doctorate student at the University of South Africa. 'AN ACTION PLAN TO ENHANCE TRANSFER OF LEARNING OF PAIN MANAGE-MENT COMPETENCIES OF NURSES IN SAUDI ARABIAN HOSPITALS' is my study. The purpose of my study is to develop an action plan for enhancing the transfer of pain management competencies among nurses in Saudi Arabian teaching hospitals. This study was approved by the UNISA College of Human Sciences Research Ethics Committee under reference number REC-012714-039.

Thank you for the time by providing your valuable inputs in the second round. You are now cordially invited to participate in the **third round** of validation of the developed action plan by using the embedded validation tool. In this second round, all the information received from panellists in the first round was analysed, and their suggestions were incorporated. The items for which consensus has already been reached are included in the action plan. You are therefore requested to respond **ONLY** to items where no consensus was reached. For example, if all statements indicate **'CONSEN-SUS REACHED**,' you should **disregard these sections** and **ONLY** respond to those that do not contain this statement.

Please provide your honest opinion once more regarding whether you agree or disagree with the action statements, the methods used to accomplish the actions, the individuals who need to take responsibility, and the appropriate time frames. Please include any additional comments or suggestions for enhancing the action plan. Your participation will be kept anonymous and confidential, and your specific contributions will not be shared with other panellists, as all information will be gathered via software programs in bulk.

If you agree to participate in the **third round** of the validation of the action plan, kindly click on the link via the **Google Forms software** provided at the end of this letter. Instructions are provided as you go along completing the first round.

Your participation is voluntary, and you may choose not to take part, by just ignoring the invitation and not clicking on the link to access the action plan and validation tool. You still have a chance to withdraw even after completing the validation tool, by not submitting your answers. All comments and suggestions for improvement will be incorporated and forwarded to a fourth round until there is a 75% consensus among all panellists. By clicking the link provided below, you will be able to access the instrument. It will only take 30 to 45 minutes for you to complete this questionnaire. I would greatly appreciate it if you would complete the questionnaire at your convenience by opening the link below.

https://docs.google.com/forms/d/13qf4iJtlpSkn1tEqIuwiU1ciZrhGUWNh8hT-Xhj5XOo/edit

Please submit your answers within 10 days after receiving this invitation.

Because participation in this study is voluntary, you will not be remunerated. Your input will, however, be beneficial to allow me to revise the action plan by implementing all suggested changes by all panellists and share the revised action plan with you again and again until consensus amongst all panellists is achieved. The results of the study may be published, but your personal information and input will not be able to be traced back to you.

After completion, please click the submit button.

Please feel free to contact me by at Mobile +966503920421 or by email at <u>kolobe66@yahoo.com</u> if you have any questions regarding this research study. You may also contact my research supervisor Prof Lizeth Roets at +27 12 429 2226 or <u>roetsl@unisa.ac.za.</u> If you have any other concerns, you may also contact the College research ethics committee at <u>Khankb@unisa.ac.za</u>, the chairperson of the committee.

Your contribution and time will be appreciated.

Kind regards.

Litaba Efraim Kolobe

ANNEXURE 20: GATEKEEPER LETTER: ROUND 3

Assistance with recruitment of Delphi participants Round 3

TO : Nurse Managers

FROM: Litaba Efraim Kolobe

Pain Management Nurse Specialist Number 47295

SUBJECT : Request to recruit Nurses and CRNS to participate as Delphi panellists

from your ward/unit

My name is Litaba Efraim Kolobe and I am a registered doctoral student at the University of South Africa. The title of my study is 'An Action Plan to Enhance Transfer of Learning of Pain Management Competencies of Nurses in Saudi Arabian **Teaching Hospitals**'. My study aims to develop an action plan to enhance the transfer of learning of pain management competencies of nurses in Saudi Arabian teaching hospitals. I already received approval from the Research Ethics Committee of the Department of Health Studies. University of South Africa (Ref # 36640085_CREC_CHS_2023) as well as an ethics approval certificate from King Abdullah International Medical Research Center (KAIMRC) and Institutional Review Board (IRB) (SP18/036/R) attached.

My study is at the **third round** of validating the action plan drafted, therefore may you assist in recruiting the same nurses and clinical resource nurses who volunteered in

647

the second-round round to participate in your unit as Delphi panellists for continuity. May you please forward the attached recruitment letter to them for voluntary participation in this **third round**.

I would appreciate it if you would once more share the recruitment letter containing the link and IRB approval with the nurses who will voluntarily participate. Participants will complete their opinions on the link provided in the attached recruitment letter. I will also inform you when we go for the **next round**.

I appreciate your time and willingness to assist.

Yours Sincerely

Litaba Efraim Kolobe

ANNEXURE 21 Findings: during Round 3 from Action Statements 1 to 7

| Action statement 1: Motivate nurses to further their studies (CONSEN | SUS |
|--|-----|
| REACHED) | |

Method 1.1: Develop a policy to motivate nurses to improve their nursing qualifications **(CONSENSUS REACHED)**

1.1.1 The following Items that must be included in the policy nurses to improve their qualifications

- A certificate issued as an acknowledgment of nurses pursuing distance learning
- One day off for attending a one-day pain management program
- A monetary incentive after completion of a new formal qualification (degree or diploma
- A monetary incentive after completion of distance learning programs
- Free accommodation for the period of study leave

(Agreement of ≥60% by panellists)

| Responsible person/s to develop a policy to motivate nurses to improve their nursing qualifications. (n=10; N=10) | Responses | | Consensus Reached (≥ 75%) |
|---|-----------|-----|---------------------------------|
| | n= | f=% | Yes/No |
| Nursing policy committee representatives appointed by the Associate Executive Directors of Nursing for KAMC and KASCH | 10 | 100 | Yes |
| 2. Clinical Directors of Nursing Operations for KAMC and KASCH | 0 | 0 | |

The policy to motivate nurses to improve their nursing qualifications must be developed within 4-6 months after approval of the action plan **(CONSENSUS REACHED)**

Method 1.2: Present and negotiate for the implementation of the policy to motivate nurses to improve their nursing qualifications to the Ministry of National Guard Health Affairs (MNGHA) through the Central Region Nursing Governance and Accountability Board **(CONSENSUS REACHED)**

Responsible person/s: Nursing policy committee representatives appointed by the Associate Executive Director of Nursing for KAMC and KASCH are the best possible individual/individuals to present and negotiate for the implementation of the policy to motivate nurses to improve their nursing qualifications to MNGHA through Central Region Nursing Governance and Accountability Board (CONSENSUS REACHED)

| Time frame required to present the policy and of negoti- | Responses | Consensus |
|--|-----------|-----------|
| ate the implementation to the Ministry of National Guard | | Decebed |
| Health Affairs (MNGHA) (n=10; N=10) | | Reached |
| | | (≥ 75%) |

| | n= | 1 | f=% | Yes/No | |
|--|----------------------------|------------------------------|---------------------------|---------------------------------|--|
| 3. 6 months | 9 | ! | 90 | Yes | |
| 4. 9 months | 1 | , | 10 | | |
| Method 1.3: Include the policy in all hospitals` policies after approval by the Ministry of Na- tional Guard Health Affairs (MNGHA) (CONSENSUS REACHED) | | | | | |
| Responsible person/s to be responsible for the inclusion of the policy in all hospitals' policies to motivate nurses to improve their nursing qualifications (n=10; N=10) | on Re : | sponses | 6 | Consensus Reached (≥ 75%) | |
| | n= | 1 | f=% | Yes/No | |
| 1. Clinical Directors of Nursing Operations in every faci ity for KAMC and KASCH | il- 9 | ! | 90 | Yes | |
| 2. Nurse Managers in all KAMC and KASCH nursing care areas | 1 | | 10 | | |
| Time frame: The policy should be included in all hospital proval of the action plan by the Ministry of National Guar REACHED) | als' poli rd Healt | cies 3 m h Affairs | ionths s (CO) | s after ap- NSENSUS | |
| Action statement 2: Make appropriate and relevant pa to the nursing team in every clinical areas (CONSEN | ain mar ISUS RI | ageme EACHEI | nt too D) | ols accessible | |
| Method 2.1: Include pain assessment tools in an electronic patient record system so that | | | | | |
| the nursing team can choose and access the most appropriate. tool in all nursing care areas. | | | | | |
| (CONSENSUS REACHED) | | | | | |
| 2.1.1 The following are pain assessment tools that r | must be | e includ | led ai | nd accessible | |
| on the electronic patient record system. PQRST CRIES (Agreement of ≥60% by panellists) | | | | | |
| Responsible person(s): One pain nurse specialist appo | ointed b | y clinica | al dire | ctors of nurs- | |
| ing in every facility for KAMC and KASCH will be respon | sible to | provide | the e | electronic for- | |
| mat of the pain assessment tool for inclusion in the elect accessible to the nursing team in every nursing care are | tronic pa a (CON | atient ree I SENSU | cord s IS RE | system to be ACHED) | |
| Time frame within which time the pain assessment | Respor | nses | | Consensus | |
| tools should be included in the electronic patient | | | | Reached | |
| record system (n=10; N=10) | | | | (≥ 75%) | |
| | n= | f=% | | Yes/No | |

| 1. 1- 3 months | 9 | 90 | Yes | |
|--|-----------------|-------------|-------------------|--|
| 2. 4- 6 months | 1 | 10 | | |
| Method 2.2: Involve nurse supervisors with pain mana | gement | training to | provide post-pain | |
| management training supervisory support to the nursi | ng team | on how to | conduct pain as- | |
| sessments in all nursing care areas. (CONSENSUS R | EACHED |) | | |
| Responsible person(s) to involve nurse supervisors | Res | ponses | Consensus | |
| with pain management training to provide post-pain | | | Reached | |
| management training and supervisory support to the | | | (≥ 75%) | |
| nursing team on how to conduct pain assessments in | n= | f=% | Yes/No | |
| all nursing care areas (n=10; N=10) | | | | |
| 1. Clinical Directors of Nursing Operations for | 9 | 90 | Yes | |
| 2. Clinical facilitators in all KAMC and KASCH nurs- | 1 | 10 | | |
| ing care areas | Dee | | 0 | |
| Time frame within which time the nurse supervi- | Res | sponses | Consensus | |
| sors should be involved to provide pain manage- | | | Reached | |
| ment training supervisory support to the nursing | | 6 0(| (275%) | |
| team on now to conduct pain assessment in all | n= | f=% | Yes/No | |
| nursing care areas. (n=10; N=10) | | | | |
| 1. Every shift when the need arises | 10 | 100 | Yes | |
| 2. Every patient round when the need arises | 0 | 0 | | |
| Method 2.3: Make hospitals' internet-based resources accessible to the patients and fam- | | | | |
| ily members to obtain other support about pain manage | ement (C | ONSENS | US REACHED) | |
| 2.3.1 The following Internet-based resources that r | nust be | accessibl | e to the patients | |
| and family members to obtain other support about | pain ma | nagemen | t | |
| Patient pain management websites | | | | |
| Patient pain management hotlines | | | | |
| Pain management support groups Peer support groups | | | | |
| (Agreement of ≥60% by panellists) | | | | |
| Nurse Managers in all KAMC and KASCH nursing care areas are responsible per- | | | | |
| son(s) to ensure internet-based resources should be accessible to patients and family | | | | |
| members to provide support about pain management (CONSENSUS REACHED) | | | | |
| Method 2.4: Make hospitals' internet-based resources | on pain | managem | ent publications | |
| and electronic materials accessible to the nursing team | i in all nu | rsing care | areas. (CON- | |
| SENSUS REACHED) | | | | |

2.4.1 The following Internet-based resources on pain management publications, electronic materials, and organisations that specialises in pain management that must be accessible

- Pain toolkits
- Videos on pain management
- Clinical updates

(Agreement of ≥60% by panellists)

Responsible person/s: The best possible individual/individuals to ensure internet-based resources on pain management publications, electronic materials, and organisations that specialises in pain management should be accessible to the nursing team in all nursing care areas. (n=; N=)

| Nursing health informatics appointed by Asso- ciated Directors of Nursing for KAMC and KASCH | 9 | 90 | Yes |
|--|---|----|-----|
| One Pain nurse specialist appointed by Clini- cal Directors of Nursing Operations in every facility for KAMC and KASCH | 1 | 10 | |

Action statement 3: Develop a practice-oriented content specific short pain management training program. (CONSENSUS REACHED)

Method 3.1: Include practice-oriented pain management training content for all nursing

care areas in the pain management program. (CONSENSUS REACHED)

3.1.1 The following specific practice-oriented pain management training content that must be included in all nursing care areas:

- Methods to promptly assess a patient's pain in all nursing areas
- The selection of appropriate pain intervention strategies based on the pain levels assessed
- Assessment of patients' pain in all nursing care areas
- The advantages and disadvantages of all pain management scales (Agreement of ≥60% by panellists)

Responsible person/s: One Pain nurse specialist appointed by Clinical Directors of

Nursing Operations in every facility is the best person to include specific practice-oriented

pain management training content for all nursing care areas. (CONSENSUS REACHED)

Time frame: The specific practice-oriented pain management training content should be

provided for inclusion in the program 3 months before the due date of the training program (CONSENSUS REACHED)

Action statement 4: Develop a pain management short program that accommodate all learning types. (CONSENSUS REACHED)

Method 4.1: Incorporate different learner types during learning/training sessions. (CON-SENSUS REACHED)

4.1.1 The following different learner types of nurses that must be included during pain

management learning/training sessions:

- Creative Learners
- Organised thinking learners
- Enthusiastic thinking learners

(Agreement of ≥60% by panellists)

4.1.2 The following different learning types that must be incorporated during pain management learning/training sessions are achieved by using the following learning styles:

- Creative learners by generating creative ideas in a group.
- Enthusiastic thinking learners by listening to the information actively, take part in the activity to practice the skill and participate in group discussion

• Organised thinking learners by solving different real-life problems (Agreement of ≥60% by panellists)

Responsible person(s): One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility is the best possible person/s for ensuring that the learning objectives accommodate the different learner types of nurses during pain management learning/training sessions. **(CONSENSUS REACHED)**

Time frame: The learning types must be shared for inclusion within the training program 3 months before the due date of the training program (CONSENSUS REACHED)

Action statement 5: Develop a pain management short course that motivates nurses to apply knowledge gained in the training program to practice. (CONSENSUS REACHED)

Method 5.1: Ensure the inclusion of different teaching approaches in the offering of the training program. **(CONSENSUS REACHED)**

5.1 The following teaching approaches that mudtbe utilised during pain management training that accommodates diverse learners and facilitators.

- Engaging in focus groups
- Using role-playing activities

(Agreement of ≥60% by panellists)

Responsible person/s: One Pain nurse specialist appointed by Clinical Directors of Nursing Operations in every facility is the best person for ensuring that teaching approaches are included during pain management training. (CONSENSUS REACHED).

Time frame: Different teaching approaches must be part of the teaching program before implementation **3 months before due date of the training program. (CONSENSUS REACHED)**

Action statement 6: Develop strategies to motivate nurses to participate in the short training program. (CONSENSUS REACHED).

Involving nurses in the development of the content of the training is the strategy that

will motivate nurses to participate in the pain management training program. (CONSEN-

SUS REACHED)

6.1 The following strategies that will motivate nurses to participate in the pain

management training program:

- Involve nurses in the development of learning goals and learning outcomes for the pain management training program relevant to their nursing care areas.
- Involve nurses in the development of the content of the training program
- Communicate the advantages of pain management competencies
- Create a supportive learning environment in nursing care areas

(Agreement of ≥60% by panellists)

Responsible person(s): One pain nurse specialist appointed by clinical directors of nursing operations in every facility is the best person for developing the strategies that will motivate nurses to participate in the short pain management training program related to their nursing care areas. (CONSENSUS REACHED)

Time frame: The strategies that motivate nurses to participate in the short pain management

training program related to their nursing care areas should be developed **1 month before** the training program starts. (CONSENSUS REACHED).

Action statement 7: Motivate nurses to apply the knowledge gained in the training program into practice. (CONSENSUS REACHED)

Method 7.1: The following ,methods that can be implemented to motivate nurses to apply their knowledge in practice:

- Offer nurses the opportunity to take on the role of pain management experts who are competent in their field.
- Support nurses' SMART goals and pain management learning
- Support drives individual nurses to apply what they have learned about pain management.
- Allow the nurses to take part in planning outcomes of a pain management training program
- Assign grades for applying pain management knowledge in practice based on annual performance

(Agreement of ≥60% by panellists)

Responsible person(s): **Nurse Managers in all nursing care areas for KAMC and KASCH** are the best persons to be responsible to facilitate the implementation of the aspect

to motivate nurses to apply their knowledge in practice. (CONSENSUS REACHED)

Time frame required to implement the methods to motivate nurses to apply their knowledge in practice (n=10; N=10)

| 1. 1-3 months after the training program | 9 | 90 | Yes | |
|--|---|----|-----|--|
|--|---|----|-----|--|

| 2. 4-6 months after the training program | 1 | 10 | |
|--|---|----|--|

| ANNEXURE 22 V | alidated Action Plan |
|----------------------|----------------------|
|----------------------|----------------------|

| Action statement | Methods | Responsible per- son(s) | Timeframe |
|--|--|---|---|
| 1. Motivate nurses to further their studies. | 1.1 Develop a policy to moti- vate nurses to improve their nursing qualifications by including a certificate is- sued as an acknowledge- ment of nurses pursuing distance learning, one day off for attending pain man- agement programs, mone- tary incentive after com- pletion of a pain manage- ment program (degree or diploma) and after com- pletion of distance learn- ing programs, and offer free accommodation for the period of study leave | Nursing policy committee repre- sentatives, ap- pointed by the as- sociate executive directors for KAMC and KASCH. | The policy to moti- vate nurses to im- prove their nursing qualifications must be developed and fi- nalised 4-6 months after approval of the action plan by the MNGHA. |
| | 1.2 Present and negotiate for implementing the policy to the Ministry of National Guard Health Affairs (MNGHA) through the Central Region Nursing Governance and Account- ability Board. | Nursing policy committee repre- sentatives ap- pointed by the As- sociate Executive Director of Nursing for KAMC and KASCH | The policy should be presented, and the implementation thereof negotiated with the Ministry of National Guard Health Affairs within |

| | 1.3 Include the policy as part of the policies of all hospitals. | Clinical directors of nursing operations in every facility for KAMC and KASCH | six months after approval of the ac- tion plan. The policy should be included within three months after approval of the ac- tion plan by the Min- istry of National Guard Health Af- fairs. |
|--|---|---|---|
| 2. Make appropri- ate and rele- vant pain man- agement tools accessible to the nursing team in every clinical area. | 2.1 Include all pain assessment tools, includ- ing PQRST and CRIES, in an electronic patient record system so the nursing team can choose and access the most ap- propriate tool in all nurs- ing care areas. | One pain nurse specialist ap- pointed by clinical directors of nurs- ing in every facility for KAMC and KASCH | The pain assess- ment tools should be included within 1–3 months in the electronic patient record system after implementing the action plan in KAMC and KASCH. |
| | 2.2 Involve nurse supervisors with pain management training to provide post- pain management training supervisory support to the nursing team on how to | Clinical Directors of Nursing Opera- tions for KAMC and KASCH | Nurse supervisors must be available on every shift to pro- vide pain manage- ment training and supervisory support |

| conduct pain assessments in all nursing care areas. 2.3 Make all hospitals' inter- net-based resources, in- cluding websites, support groups, hotlines, and peer support groups, ac- cessible to patients and family members to obtain other support about pain management. | Nurse Managers in all KAMC and KASCH nursing care areas. | to the nursing team (when the need arises) after ap- proval of the action plan. The hospitals' inter- net-based resources must be made ac- cessible to the pa- tients and family members to obtain other support about pain management, 24 hours per day and seven days a week, after ap- proval of the action plan. |
|---|---|---|
| 2.4 Make all hospitals' inter- net-based resources, in- cluding publications, elec- tronic materials, and or- ganizations that special- ize in pain management, accessible to the nursing team in all nursing care areas that consist of pain toolkits, videos on pain management, and clinical updates. | Nursing health in- formatics ap- pointed by Associ- ate Directors of Nursing for KAMC and KASCH. | The hospitals' inter- net-based re- sources, publica- tions, electronic ma- terials, and organi- zations that special- ize in pain manage- ment should be con- tinuously available. |

| 3. | Develop a practice-ori- ented content- specific short pain manage- ment training program. | а. | Include practice-oriented pain management train- ing content, inclusive of *the methods to promptly assess a patient's pain in all nursing areas; * The selection of appro- priate pain intervention strategies based on the pain levels assessed; *the advantages and dis- advantages of all pain management scales and *assessment of patients' pain in all nursing care areas. | One pain nurse specialist ap- pointed by Clinical Directors of Nurs- ing Operations in every facility. | The practice-ori- ented content-spe- cific short pain man- agement training program must be available three months before the due date for the of- fering of the train- ing program. |
|----|---|----|--|---|---|
| 4. | Develop a pain management short program that accommo- dates all learn- ing types. | a. | Accommodate different learning types and learn- ing styles nurses used to achieve them when de- veloping the training pro- gram, specifically *creative learners (gener- ate creative ideas in a group), *enthusiastic thinking learners (listen to the information actively, take part in the activity to practice the skill and par- ticipate in the group dis- cussion), as well as | One pain nurse specialist ap- pointed by Clinical Directors of Nurs- ing Operations in every facility | Advertise the inclu- siveness of the dif- ferent learner types to be accommo- dated in the short program three months before the due date of the of- fering of the train- ing program. |
| 5. Incorporate dif- ferent teaching approaches to accommodate diverse learners and facilitators in pain manage- ment training. | * organised thinking learners (solve differen real-life problems). a. Ensure the inclusion of different teaching ap- proaches, including for groups and role play, i the offering of the train program. | nt One pain nurse specialist ap- cus pointed by Clinical n Directors of Nurs- ing operations in every facility. | Different teaching approaches must be offered in the teach- ing program and must be available to learners and facilita- tors three months before the due date of when the training |
|---|--|--|--|
| 6. Develop strate- gies to motivate nurses to partic- ipate in the short training program. | a. Involve nurses in the * velopment of the contegoals, and outcomes of the training program, *communicate the advantages of pain management competencie (the platform that can butilized), and | de- ont, specialist ap- pointed by Clinical Directors of nurs- ing operations in every facility. | plemented. Nurses must be in- vited to participate in the development of the content, goals, and outcomes of the training one month before the training program is to be |
| 7. Motivate nurses to apply the knowledge gained in the | b. *create a supportive learning environment. a. *Offer nurses the opportunity to take on the roof a pain management expert who is competered. | or- Nurse Managers in le all nursing care ar- eas for KAMC and KASCH are the | offered. The implementation of methods to moti- vate nurses to apply |

| training program | in the field, *support nurs- | best persons to be | their knowledge in |
|------------------|---|-----------------------|--------------------|
| in practice. | es' SMART goals and | responsible for fa- | practice must be |
| | pain management learn- | cilitating the imple- | within 1-3 months |
| | ing, | mentation of the | after the training |
| | *support what drives individ- | aspect to motivate | program. |
| | ual nurses to apply what they | nurses to apply | |
| | have learned about pain man- | their knowledge in | |
| | agement | practice. | |
| | * and assign grades for apply- ing pain management knowledge in practice based on annual performance | | |

ANNEXURE 23: Certificate of Language Editing

CERTIFICATE OF LANGUAGE EDITING

I, the undersigned, declare that I have edited the Doctor of Philosopy (Nursing) Thesis of Litaba Efraim Kolobe, titled:

AN ACTION PLAN TO ENHANCE THE TRANSFER OF LEARNING OF PAIN MANAGEMENT COMPETENCIES OF NURSES IN SAUDI ARABIAN HOSPITALS.

A small part of the thesis, contained in specific figures, the direct quotations of human participants, and the documents included in the annexures, was not edited because these sections contain transcribed verbal communication, were included as complete images or formed part of completed research records which had to remain intact.

As is usual, the editor is not responsible for the correctness of changes made to the thesis after it was edited and before submission or after possible changes suggested by the examiners have been implemented.

Signed:

Prof (emeritus) P.J. Botha

(A member of the South African Translators' Institute, no. 1000048.)

Date: 16 February 2024